SUBCHAPTER B—REGULATIONS UNDER THE FEDERAL POWER ACT

PART 4—LICENSES, PERMITS, EX-EMPTIONS, AND DETERMINATION OF PROJECT COSTS

Subpart A—Determination of Cost of Projects Constructed Under License

Sec.

- 4.1 Initial cost statement.
- 4.3 Report on project cost.
- 4.4 Service of report.
- 4.5 Time for filing protest.
- 4.6 Burden of proof.
- 4.7 Findings.

Subpart B—Determination of Fair Value of Constructed Projects, Under Section 23(a) of the Act

- 4.10 Valuation data.
- 4.11 Reports.
- 4.12 Service of report.
- 4.13 Time for filing protest.
- 4.14 Hearing upon report.

Subpart C—Determination of Cost of Constructed Projects not Subject to Section 23(a) of the Act

- 4.20 Initial statement.
- 4.21 Reports.
- 4.22 Service of report.
- 4.23 Time for filing protest.
- 4.24 Determination of cost.
- 4.25 Findings.

Subpart D—Application for Preliminary Permit, License or Exemption: General Provisions

- 4.30 Applicability and definitions.
- 4.31 Initial or competing application: who may file.
- 4.32 Acceptance for filing or rejection; information to be made available to the public; requests for additional studies.
- 4.33 Limitations on submitting applications.
- 4.34 Hearings on applications; consultation on terms and conditions; motions to intervene; alternative procedures.
- 4.35 Amendment of application; date of acceptance.
- 4.36 Competing applications: deadlines for filing; notices of intent; comparisons of plans of development.
- 4.37 Rules of preference among competing applications.
- 4.38 Consultation requirements.

4.39 Specifications for maps and drawings.

Subpart E—Application for License for Major Unconstructed Project and Major Modified Project

- 4.40 Applicability.
- 4.41 Contents of application.

Subpart F—Application for License for Major Project—Existing Dam

- 4.50 Applicability.
- 4.51 Contents of application.

Subpart G—Application for License for Minor Water Power Projects and Major Water Power Projects 5 Megawatts or Less

- 4.60 Applicability and notice to agencies.
- 4.61 Contents of application.

Subpart H—Application for License for Transmission Line Only

- 4.70 Applicability.
- 4.71 Contents of application.

Subpart I—Application for Preliminary Permit; Amendment and Cancellation of Preliminary Permit

- 4.80 Applicability.
- 4.81 Contents of application.
- 4.82 Amendments.
- 4.83 Cancellation and loss of priority.
- 4.84 Surrender of permit.

Subpart J—Exemption of Small Conduit Hydroelectric Facilities

- 4.90 Applicability and purpose.
- 4.91 [Reserved]
- 4.92 Contents of exemption application.
- 4.93 Action on exemption applications.
- 4.94 Standard terms and conditions of exemption.
- 4.95 Surrender of exemption.
- 4.96 Amendment of exemption.

Subpart K—Exemption of Small Hydroelectric Power Projects of 5 Megawatts or Less

- 4.101 Applicability.
- 4.102 Surrender of exemption.
- 4.103 General provisions for case-specific exemption.
- 4.104 Amendment of exemption.
- 4.105 Action on exemption applications.

- 4.106 Standard terms and conditions of casespecific exemption from licensing.
- 4.107 Contents of application for exemption from licensing.
- 4.108 Contents of application for exemption from provisions other than licensing.

Subpart L—Application for Amendment of license

- 4.200 Applicability.
- 4.201 Contents of application.
- 4.202 Alteration and extension of license.

Subpart M—Fees Under Section 30(e) of the Act

- 4.300 Purpose, definitions, and applicability.4.301 Notice to fish and wildlife agencies and estimation of fees prior to filing.
- 4.302 Fees at filing.
- 4.303 Post-filing procedures.
- 4.304 Payment.
- 4.305 Enforcement.

AUTHORITY: 16 U.S.C. 791a-825r, 2601-2645; 42 U.S.C. 7101-7352

Source: Order 141, 12 FR 8485, Dec. 19, 1947, unless otherwise noted.

Subpart A—Determination of Cost of Projects Constructed Under License

§ 4.1 Initial cost statement.

- (a) Notification of Commission. When a project is constructed under a license issued under the Federal Power Act, the licensee shall, within one year after the original project is ready for service, file with the Commission a letter, in quadruplicate, declaring that the original costs have been booked in compliance with the Commission's Uniform System of Accounts and the books of accounts are ready for audit.
- (b) Licensee's books. The licensee's books of accounts for each project shall be maintained in such a fashion that each year's additions, betterments, and delections to the project may be readily ascertained.
- (c) Availability of information to the public. The information made available to the Commission in accordance with this section must be available to the public for inspection and copying when specifically requested.
- (d) Compliance with the Act. Compliance with the provisions of this section satisfies the filing requirements of sec-

tion 4(b) of the Federal Power Act (16 U.S.C. 797(b)).

[Order 53, 44 FR 61948, Oct. 29, 1979]

§ 4.3 Report on project cost.

- (a) Scheduling an audit. When the original cost declaration letter, filed in accordance with §4.1 is received by the Commission, its representative will schedule and conduct an audit of the books, cost records, engineering reports, and other records supporting the project's original cost. The audit may include an inspection of the project works.
- (b) Project records. The cost records shall be supported by memorandum accounts reflecting the indirect and overhead costs prior to their spread to primary accounts as well as all the details of allocations including formulas utilized to spread the indirect and overhead costs to primary accounts.
- (c) Report by Commission staff. Upon completion of the audit, a report will be prepared for the Commission setting forth the audit findings and recommendations with respect to the cost as claimed.

[Order 53, 44 FR 61948, Oct. 29, 1979]

§ 4.4 Service of report.

Copies of such report will be served upon said licensees, and copies will also be sent to the State public service commission, or if the State has no regulatory agency, to the Governor of the State where such project is located, and to such other parties as the Commission shall prescribe, and the report will be made available for public inspection at the time of service upon the licensee.

(Administrative Procedure Act, 5 U.S.C. 551–557 (1976); Federal Power Act, as amended, 16 U.S.C. 291–628 (1976 & Supp. V 1981), Dept. of Energy Organization Act 42 U.S.C. 7101–7352 (Supp. V 1981); E.O. 12009, 3 CFR 142 (1978))

[Order 141, 12 FR 8485, Dec. 19, 1947, as amended by Order 344, 48 FR 49010, Oct. 24, 1983; Order 737, 75 FR 43402, July 26, 2010]

§ 4.5 Time for filing protest.

Thirty days after service thereof will be allowed to such licensee within which to file a protest to such reports. If no protest is filed within the time allowed, the Commission will issue such order as may be appropriate. If a protest is filed, a public hearing will be ordered in accordance with subpart E of part 385 of this chapter.

[Order 141, 12 FR 8485, Dec. 19, 1947, as amended by Order 225, 47 FR 19056, May 3, 1982]

§4.6 Burden of proof.

The burden of proof to sustain each item of claimed cost shall be upon the licensee and only such items as are in the opinion of the Commission supported by satisfactory proof may be entered in the electric plant accounts of the licensee.

[Order 53, 44 FR 61948, Oct. 29, 1979]

§ 4.7 Findings.

- (a) Commission determination. Final action by the Commission will be in the form of an order served upon all parties to the proceeding. One copy of the order will be furnished to the Secretary of Treasury by the Commission.
- (b) Adjustments to licensee's books. The licensee's books of account for the project shall be adjusted to conform to the actual legitimate cost as revised by the order of the Commission. These adjustments and the project may be audited by Commission representatives, as scheduled.

 $[{\rm Order}\ 53,\ 44\ {\rm FR}\ 61948,\ {\rm Oct.}\ 29,\ 1979]$

Subpart B—Determination of Fair Value of Constructed Projects, Under Section 23(a) of the Act

§4.10 Valuation data.

- (a) Notification of Commission. In every case arising under section 23(a) of the Federal Power Act that requires the determination of the fair value of a project already constructed, the licensee shall, within six months after the date of issuance of a license, file with the Commission a letter, in quadruplicate.
- (b) Contents of letter. The letter referred to in paragraph (a) shall contain a statement to the effect that an inventory and appraisal in detail, as of the effective date of the license, of all property subject thereto and to be so valued has been completed. The letter

shall also include a statement to the effect that the actual legitimate original cost, or if not known, the estimated original cost, and accrued depreciation of the property, classified by prime accounts as prescribed in the Commission's Uniform System of Accounts, have been established.

- (c) *Licensee's books*. The licensee's books of account for each project shall be maintained in such a fashion that each year's additions, betterments, and deletions to the projects may be readily ascertained.
- (d) Availability of information to the public. The information made available to the Commission in accordance with this section must be available for inspection and copying by the public when specifically requested.

[Order 53, 44 FR 61948, Oct. 29, 1979]

§4.11 Reports.

Representatives of the Commission will inspect the project works, engineering reports, and other records of the project, check the inventory and make an appraisal of the property and an audit of the books, records, and accounts of the licensee relating to the property to be valued, and will prepare a report of their findings with respect to the inventory, appraisal, original cost, accrued depreciation, and fair value of the property.

§ 4.12 Service of report.

A copy of such report will be served upon said licensee, and copies will also be sent to the State public service commission, or if the State has no regulatory agency, to the Governor of the State where such project is located. The report will be made available for public inspection at the time of service upon the licensee.

(Administrative Procedure Act, 5 U.S.C. 551–557 (1976); Federal Power Act, as amended, 16 U.S.C. 291–628 (1976 & Supp. V 1981), Dept. of Energy Organization Act 42 U.S.C. 7101–7352 (Supp. V 1981); E.O. 12009, 3 CFR 142 (1978))

[Order 141, 12 FR 8485, Dec. 19, 1947, as amended by Order 344, 48 FR 49010, Oct. 24, 1983; Order 737, 75 FR 43402, July 26, 2010]

§ 4.13 Time for filing protest.

Thirty days after service thereof will be allowed to the licensee within which to file a protest to such report.

§4.14 Hearing upon report.

- (a) Public hearing. After the expiration of the time within which a protest may be filed, a public hearing will be ordered in accordance with subpart E of part 385 of this chapter.
- (b) Commission determination. After the conclusion of the hearing, the Commission will make a finding of fair value, accompanied by an order which will be served upon the licensee and all parties to the proceeding. One copy of the order shall be furnished to the Secretary of the Treasury by the Commission.
- (c) Adjustment to licensee's books. The licensee's books of account for the project shall be adjusted to conform to the fair value of the project as revised by the order of the Commission. These adjustments and the project may be audited by Commission representatives, as scheduled.

[Order 53, 44 FR 61949, Oct. 29, 1979, as amended by Order 225, 47 FR 19056, May 3, 1982]

Subpart C—Determination of Cost of Constructed Projects not Subject to Section 23(a) of the Act

§ 4.20 Initial statement.

- (a) Notification of Commission. In all cases where licenses are issued for projects already constructed, but which are not subject to the provisions of section 23(a) of the Act (49 Stat. 846; 16 U.S.C. 816), the licensee shall, within 6 months after the date of issuance of license, file with the Commission a letter, in quadruplicate.
- (b) Contents of letter. The letter referred to in paragraph (a) of this section shall contain a statement to the effect that an inventory in detail of all property included under the license, as of the effective date of such license, has been completed. The letter shall also include a statement to the effect that actual legitimate original cost, or if not known, the estimated original cost, and accrued depreciation of the property, classified by prime accounts

as prescribed in the Commission's Uniform System of Accounts, have been established.

- (c) Licensee's books. The licensee's books of account for each project shall be maintained in such a fashion that each year's additions, betterments, and deletions to the project may be readily ascertained.
- (d) Availability of information to the public. The information made available to the Commission in accordance with this section must be available for inspection and copying by the public when specifically requested.
- (e) Compliance with the Act. Compliance with the provisions of this section satisfies the filing requirements of section 4(b) of the Federal Power Act (16 U.S.C. 797(b)).

[Order 53, 44 FR 61949, Oct. 29, 1979]

§4.21 Reports.

Representatives of the Commission will inspect the project works, engineering reports, and other records of the project, check the inventory and estimated depreciation, make an audit of the books, records, and accounts of the licensee relating to the property under license, and prepare a report of their findings with respect to the inventory, the original cost of the property, and the estimated accrued depreciation thereon.

§ 4.22 Service of report.

Copies of such report will be served upon said licensees, and copies will also be sent to the State public service commission, or if the State has no regulatory agency, to the Governor of the State where such project is located, and to such other parties as the Commission shall prescribe, and the report will be made available for public inspection at the time of service upon the licensee.

(Administrative Procedure Act, 5 U.S.C. 551–557 (1976); Federal Power Act, as amended, 16 U.S.C. 291–628 (1976 & Supp. V 1981), Dept. of Energy Organization Act 42 U.S.C. 7101–7352 (Supp. V 1981); E.O. 12009, 3 CFR 142 (1978))

[Order 141, 12 FR 8485, Dec. 19, 1947, as amended by Order 344, 48 FR 49010, Oct. 24, 1983; Order 737, 75 FR 43402, July 26, 2010]

§ 4.23 Time for filing protest.

Thirty days after service thereof will be allowed to such licensee within which to file a protest to such reports. If no protest is filed within the time allowed, the Commission will issue such order as may be appropriate. If a protest is filed, a public hearing will be ordered in accordance with subpart E of part 385 of this chapter.

[Order 141, 12 FR 8485, Dec. 19, 1947, as amended by Order 225, 47 FR 19056, May 3, 1982]

§ 4.24 Determination of cost.

The Commission, after receipt of the reports, or after the conclusion of the hearing if one is held, will determine the amounts to be included in the electric plant accounts of the licensee as the cost of the property and the accrued depreciation thereon.

§4.25 Findings.

- (a) Commission determination. Final action by the Commission will be in the form of an order served upon all parties to the proceeding. One copy of the order shall be furnished to the Secretary of Treasury by the Commission.
- (b) Adjustment to licensee's books. The licensee's books of account for the project shall be adjusted to conform to the actual legitimate cost as revised by the order of the Commission. These adjustments and the project may be audited by Commission representatives, as scheduled.

[Order 53, 44 FR 61949, Oct. 29, 1979]

Subpart D—Application for Preliminary Permit, License or Exemption: General Provisions

AUTHORITY: Federal Power Act, as amended, 16 U.S.C. 792–828c; Department of Energy Organization Act, 42 U.S.C. 7101–7352; E.O. 12009, 42 FR 46267; Public Utility Regulatory Policies Act of 1978, 16 U.S.C. 2601–2645; Pub. L. 96–511, 94 Stat. 2812 (44 U.S.C. 3501 et seq.).

§4.30 Applicability and definitions.

- (a) (1) This subpart applies to applications for preliminary permit, license, or exemption from licensing.
- (2) Any potential applicant for an original license for which prefiling consultation begins on or after July 23,

2005 and which wishes to develop and file its application pursuant to this part, must seek Commission authorization to do so pursuant to the provisions of part 5 of this chapter.

(b) For the purposes of this part—

- (1)(i) Competing development application means any application for a license or exemption from licensing for a proposed water power project that would develop, conserve, and utilize, in whole or in part, the same or mutually exclusive water resources that would be developed, conserved, and utilized by a proposed water power project for which an initial preliminary permit or initial development application has been filed and is pending before the Commission.
- (ii) Competing preliminary permit application means any application for a preliminary permit for a proposed water power project that would develop, conserve, and utilize, in whole or in part, the same or mutually exclusive water resources that would be developed, conserved and utilized by a proposed water power project for which an initial preliminary permit or initial development application has been filed and is pending before the Commission.
- (2) Conduit means any tunnel, canal, pipeline, aqueduct, flume, ditch, or similar manmade water conveyance that is operated for the distribution of water for agricultural, municipal, or industrial consumption and not primarily for the generation of electricity. The term not primarily for the generation of electricity includes but is not limited to a conduit:
- (i) Which was built for the distribution of water for agricultural, municipal, or industrial consumption and is operated for such a purpose; and
- (ii) To which a hydroelectric facility has been or is proposed to be added.
- (3) Construction of a dam, for the purposes of provisions governing application for exemption of a small conduit hydroelectric facility, means any construction, repair, reconstruction, or modification of a dam that creates a new impoundment or increases the normal maximum surface elevation or the normal maximum surface area of an existing impoundment.
- (4)(i) Dam, for the purposes of provisions governing application for license

- of a major project—existing dam, means any structure for impounding or diverting water.
- (ii) *Dam*, for the purposes of provisions governing application for exemption of a small conduit hydroelectric facility, means any structure that impounds water.
- (iii) Dam, for the purposes of provisions governing application for exemption of a small hydroelectric power project, means any structure for impounding water, including any diversion structure that is designed to obstruct all or substantially all of the flow of a natural body of water.
- (5) Development application means any application for either a license or exemption from licensing for a proposed water power project.
- (6)(i) Existing dam, for the purposes of provisions governing application for license of a major project—existing dam, means any dam (as defined in paragraph (b)(4)(i) of this section) that has already been constructed and which does not require any construction or enlargement of impoundment structures other than repairs or reconstruction.
- (ii) Existing dam, for the purposes of provisions governing application for exemption of a small hydroelectric power project, means any dam, the construction of which was completed on or before July 22, 2005, and which does not require any construction or enlargement of impoundment structures (other than repairs or reconstruction) in connection with the installation of any small hydroelectric power project.
- (7) Existing impoundment, for the purposes of provisions governing application for license of a major project—existing dam, means any body of water that an existing dam impounds.
- (8) Federal lands, for the purposes of provisions governing application for exemption of a small hydroelectric power project, means any lands to which the United States holds fee title.
- (9)(i) Fish and wildlife agencies means the United States Fish and Wildlife Service, the National Marine Fisheries Service, and the state agency in charge of administrative management over fish and wildlife resources of the state

in which a proposed hydropower project is located.

- (ii) Fish and wildlife recommendation means any recommendation designed to protect, mitigate damages to, or enhance any wild member of the animal kingdom, including any migratory or nonmigratory mammal, fish, bird, amphibian, reptile, mollusk, crustacean, or other invertebrate, whether or not bred, hatched, or born in captivity, and includes any egg or offspring thereof, related breeding or spawning grounds, and habitat. A "fish and wildlife recommendation" includes a request for a study which cannot be completed prior to licensing, but does not include a request that the proposed project not be constructed or operated, a request for additional pre-licensing studies or analysis or, as the term is used in §§ 4.34(e)(1) and 4.34(f)(3), a recommendation for facilities, programs, or other measures to benefit recreation or tourism.
- (10) Indian tribe means, in reference to a proposal to apply for a license or exemption for a hydropower project, an Indian tribe which is recognized by treaty with the United States, by federal statute, or by the U.S. Department of the Interior in its periodic listing of tribal governments in the FEDERAL REGISTER in accordance with 25 CFR 83.6(b), and whose legal rights as a tribe may be affected by the development and operation of the hydropower project proposed (as where the operation of the proposed project could interfere with the management and harvest of anadromous fish or where the project works would be located within the tribe's reservation).
- (11)(i) Initial development application means any acceptable application for either a license or exemption from licensing for a proposed water power project that would develop, conserve, and utilize, in whole or in part, water resources for which no other acceptable application for a license or exemption from licensing has been submitted for filing and is pending before the Commission.
- (ii) Initial preliminary permit application means any acceptable application for a preliminary permit for a proposed water power project that would develop, conserve, and utilize, in whole or

in part, water resources for which no other acceptable preliminary permit application has been submitted for filing and is pending before the Commission.

- (12) Install or increase, for the purposes of provisions governing application for exemption of a small hydroelectric power project, means to add new generating capacity at a site that has no existing generating units, to replace or rehabilitate an abandoned or unused existing generating unit, or to increase the generating capacity of any existing power plant by installing an additional generating unit or by rehabilitating an operable generating unit in a way that increases its rated electric power output.
- (13) Licensed water power project means a project, as defined in section 3(11) of the Federal Power Act, that is licensed under Part I of the Federal Power Act.
- (14) Major modified project means any major project—existing dam, as defined in paragraph (b)(16) of this section, that would include:
- (i) Any repair, modification or reconstruction of an existing dam that would result in a significant change in the normal maximum surface area or the normal maximum surface elevation of an existing impoundment; or
- (ii) Any change in existing project works or operations that would result in a significant environmental impact.
- (15) Major unconstructed project means any unlicensed water power project that would:
- (i) Have a total installed generating capacity of more than 1.5 MW; and
- (ii) Use the water power potential of a dam and impoundment which, at the time application is filed, have not been constructed.
- (16) Major project—existing dam means a licensed or unlicensed, existing or proposed water power project that would:
- (i) Have a total installed generating capacity or more than 2,000 horsepower (1.5 MW); and
- (ii) Not use the water power potential provided by any dam except an existing dam.
- (17) Minor water power project means any licensed or unlicensed, existing or proposed water power project that

would have a total installed generation capacity of 2,000 horsepower (1.5 MW), or less.

- (18) New development, for the purposes of provisions governing application for license of a major project—existing dam, means any construction, installation, repair, reconstruction, or other change in the existing state of project works or appurtenant facilities, including any dredging and filling in project waters.
- (19) New license means any license, except an annual license issued under section 15 of the Federal Power Act, for a water power project that is issued under the Federal Power Act after the initial license for that project.
- (20)(i) Non-Federal lands, for the purposes of provisions governing application for exemption of a small conduit hydroelectric facility, means any lands except lands to which the United States holds fee title.
- (ii) Non-Federal lands, for the purposes of provisions governing application for exemption of a small hydroelectric power project, mean any lands other than Federal lands defined in paragraph (b)(8) of this section.
- (21) Person means any individual and, as defined in section 3 of the Federal Power Act, any corporation, municipality, or state.
- (22) *Project*, for the purposes of provisions governing application for exemption of a small hydroelectric power project, means:
- (i) The impoundment and any associated dam, intake, water conveyance facility, power plant, primary transmission line, and other appurtenant facility if a lake or similar natural impoundment or a manmade impoundment is used for power generation; or
- (ii) Any diversion structure other than a dam and any associated water conveyance facility, power plant, primary transmission line, and other appurtenant facility if a natural water feature other than a lake or similar natural impoundment is used for power generation.
- (23) Qualified exemption applicant means any person who meets the requirements specified in §4.31(b)(2) with respect to a small hydroelectric power project for which exemption from licensing is sought.

- (24) Qualified license applicant means any person to whom the Commission may issue a license, as specified in section 4(e) of the Federal Power Act.
- (25) Ready for environmental analysis means the point in the processing of an application for an original or new license or exemption from licensing which has been accepted for filing, where substantially all additional information requested by the Commission has been filed and found adequate.
- (26) Real property interests, for the purposes of provisions governing application for exemption of a small conduit hydroelectric facility or a small hydroelectric power project, includes ownership in fee, rights-of-way, easements, or leaseholds.
- (27) Resource agency means a Federal, state, or interstate agency exercising administration over the areas of flood control, navigation, irrigation, recreation, fish and wildlife, water resource management (including water rights), or cultural or other relevant resources of the state or states in which a project is or will be located.
- (28) Small conduit hydroelectric facility means an existing or proposed hydroelectric facility that is constructed, operated, or maintained for the generation of electric power, and includes all structures, fixtures, equipment, and lands used and useful in the operation or maintenance of the hydroelectric facility, but excludes the conduit on which the hydroelectric facility is located or the transmission lines associated with the hydroelectric facility and which:
- (i) Utilizes for electric power generation the hydroelectric potential of a conduit:
- (ii) Is located entirely on non-Federal lands, as defined in paragraph (b)(20)(i) of this section;
- (iii) Has an installed generating capacity of 15 MW or less (40 MW in the case of a municipal water supply project):
- (iv) Is not an integral part of a dam;(v) Discharges the water it uses for power generation either:
 - (A) Into a conduit;
- (B) Directly to a point of agricultural, municipal, or industrial consumption; or

- (C) Into a natural water body if a quantity of water equal to or greater than the quantity discharged from the hydroelectric facility is withdrawn from that water body downstream into a conduit that is part of the same water supply system as the conduit on which the hydroelectric facility is located; and
- (vi) Does not rely upon construction of a dam, which construction will create any portion of the hydrostatic head that the facility uses for power generation unless that construction would occur for agricultural, municipal, or industrial consumptive purposes even if hydroelectric generating facilities were not installed.
- (29) Small hydroelectric power project means any project in which capacity will be installed or increased after the date of notice of exemption or application under subpart K of this chapter, which will have a total installed capacity of not more than 5 MW, and which:
- (i) Would utilize for electric power generation the water power potential of an existing dam that is not owned or operated by the United States or by an instrumentality of the Federal Government, including the Tennessee Valley Authority; or
- (ii)(A) Would utilize for the generation of electricity a natural water feature, such as a natural lake, waterfall, or the gradient of a natural stream, without the need for a dam or manmade impoundment; and
- (B) Would not retain water behind any structure for the purpose of a storage and release operation.
- (30) PURPA benefits means benefits under section 210 of the Public Utility Regulatory Policies Act of 1978 (PURPA). Section 210(a) of PURPA requires electric utilities to purchase electricity from, and to sell electricity to, qualifying facilities.

[Order 413, 50 FR 11676, Mar. 25, 1985, as amended by Order 487, 52 FR 48404, Dec. 22, 1987; Order 499, 53 FR 27001, July 18, 1988; Order 503, 53 FR 36567, Sept. 21, 1988; Order 533, 56 FR 23146, May 20, 1991; 56 FR 61154, Dec. 2, 1991; Order 533—A, 57 FR 10809, Mar. 31, 1992; 59 FR 10577, Mar. 7, 1994; Order 2002, 68 FR 51115, Aug. 25, 2003; Order 699, 72 FR 45323, Aug. 14, 2007]

§ 4.31 Initial or competing application: who may file.

- (a) Application for a preliminary permit or a license. Any citizen, association of citizens, domestic corporation, municipality, or state may submit for filing an initial application or a competing application for a preliminary permit or a license for a water power project under Part I of the Federal Power Act.
- (b) Application for exemption of a small conduit hydroelectric facility—(1) Exemption from provisions other than licensing. Any citizen, association of citizens, domestic corporation, municipality, or state that has all of the real property interests in the lands necessary to develop and operate that project, or an option to obtain those interests, may apply for exemption of a small conduit hydroelectric facility from provisions of Part I of the Federal Power Act, other than licensing provisions.
- (2) Exemption from licensing. Any person having all the real property interests in the lands necessary to develop and operate the small conduit hydroelectric facility, or an option to obtain those interests, may apply for exemption of that facility from licensing under Part I of the Federal Power Act.
- (c) Application for case-specific exemption of a small hydroelectric power project—(1) Exemption from provisions other than licensing. Any qualified license applicant or licensee seeking amendment of its license may apply for exemption of the related project from provisions of Part I of the Federal Power Act other than licensing provisions.
- (2) Exemption from licensing—(i) Only Federal lands involved. If only rights to use or occupy Federal lands would be necessary to develop and operate the proposed small hydroelectric power project, any person may apply for exemption of that project from licensing.
- (ii) Some non-Federal lands involved. If real property interests in any non-Federal lands would be necessary to develop and operate the proposed small hydroelectric power project, any person who has all of the real property interests in non-Federal lands necessary to develop and operate that project, or an option to obtain those interests,

may apply for exemption of that project from licensing.

[Order 413, 50 FR 11678, Mar. 25, 1985]

§4.32 Acceptance for filing or rejection; information to be made available to the public; requests for additional studies.

- (a) Each application must:
- (1) For a preliminary permit or license, identify every person, citizen, association of citizens, domestic corporation, municipality, or state that has or intends to obtain and will maintain any proprietary right necessary to construct, operate, or maintain the project:
- (2) For a preliminary permit or a license, identify (providing names and mailing addresses):
- (i) Every county in which any part of the project, and any Federal facilities that would be used by the project, would be located:
- (ii) Every city, town, or similar local political subdivision:
- (A) In which any part of the project, and any Federal facilities that would be used by the project, would be located; or
- (B) That has a population of 5,000 or more people and is located within 15 miles of the project dam;
- (iii) Every irrigation district, drainage district, or similar special purpose political subdivision:
- (A) In which any part of the project, and any Federal facilities that would be used by the project, would be located; or
- (B) That owns, operates, maintains, or uses any project facilities or any Federal facilities that would be used by the project;
- (iv) Every other political subdivision in the general area of the project that there is reason to believe would likely be interested in, or affected by, the application; and
- $\left(v\right)$ All Indian tribes that may be affected by the project.
- (3)(i) For a license (other than a license under section 15 of the Federal Power Act) state that the applicant has made, either at the time of or before filing the application, a good faith effort to give notification by certified mail of the filing of the application to:

- (A) Every property owner of record of any interest in the property within the bounds of the project, or in the case of the project without a specific boundary, each such owner of property which would underlie or be adjacent to any project works including any impoundments; and
- (B) The entities identified in paragraph (a)(2) of this section, as well as any other Federal, state, municipal or other local government agencies that there is reason to believe would likely be interested in or affected by such application.
- (ii) Such notification must contain the name, business address, and telephone number of the applicant and a copy of the Exhibit G contained in the application, and must state that a license application is being filed with the Commission.
- (4)(i) As to any facts alleged in the application or other materials filed, be subscribed and verified under oath in the form set forth in paragraph (a) (3)(ii) of this section by the person filing, an officer thereof, or other person having knowledge of the matters sent forth. Ιf the subscription and verification is by anyone other than the person filing or an officer thereof, it shall include a statement of the reasons therefor
- (ii) This (application, etc.) is executed in the $\,$

State of
County of
by:
(Name)
(Address)
being duly sworn, depose(s) and say(s) that the contents of this (application, etc.) are true to the best of (his or her) knowledge of belief. The undersigned applicant(s) has (have) signed the (application, etc.) this day of, 19
(Applicant(s))
By:
Subscribed and sworn to before me, a [Notary Public, or title of other official authorized by the state to notarize documents, as appropriate] of the State of this day of, 19 /SEAL/ [if any]
, comments and a

(Notary Public, or other authorized official)

- (5) Contain the information and documents prescribed in the following sections of this chapter, according to the type of application:
 - (i) Preliminary permit: §4.81;
- (ii) License for a minor water power project and a major water power project 5 MW or less: §4.61;
- (iii) License for a major unconstructed project and a major modified project: § 4.41;
- (iv) License for a major project—existing dam: §4.51;
- (v) License for a transmission line only: §4.71;
- (vi) Nonpower license for a licensed project: §16.11;
- (vii) Exemption of a small conduit hydroelectric facility: § 4.92;
- (viii) Case-specific exemption of a small hydroelectric power project: §4.107; or
- (ix) License or exemption for a project located at a new dam or diversion where the applicant seeks PURPA benefits: §292.208.
- (b) (1) Each applicant for a preliminary permit, license, and transfer or surrender of license and each petitioner for surrender of an exemption must submit the application or petition to the Secretary of the Commission in accordance with filing procedures posted on the Commission's Web site at http:// www.ferc.gov. The applicant or petitioner must serve one copy of the application or petition on the Director of the Commission's Regional Office for the appropriate region and on each resource agency, Indian tribe, and member of the public consulted pursuant to §4.38 or §16.8 of this chapter or part 5 of this chapter. In the case of an application for a preliminary permit, the applicant must, if the Commission so directs, serve copies of the application on the U.S. Department of the Interior and the U.S. Army Corps of Engineers. The application may include reduced prints of maps and drawings conforming to §4.39(d). The originals (microfilm) of maps and drawings are not to be filed initially, but will be required pursuant to paragraph (d) of this section. The Commission may also ask for the filing of full-sized prints in appropriate cases.

- (2) Each applicant for exemption must submit the application to the Secretary of the Commission in accordance with filing procedures posted on the Commission's Web site at http:// www.ferc.gov. An applicant must serve one copy of the application on the Director of the Commission's Regional Office for the appropriate region and on each resource agency consulted pursuant to §4.38. For each application filed following October 23, 2003, maps and drawings must conform to the requirements of §4.39. The originals (microfilm) of maps and drawing are not to be filed initially, but will be requested pursuant to paragraph (d) of this section.
- (3)(i) An applicant must make information regarding its proposed project reasonably available to the public for inspection and reproduction, from the date on which the applicant files its application for a license or exemption until the licensing or exemption proceeding for the project is terminated by the Commission. This information includes a copy of the complete application for license or exemption, together with all exhibits, appendices and any amendments, and any comments, pleadings, supplementary or additional information, or correspondence filed by the applicant with the Commission in connection with the ap-
- (ii) An applicant must delete from any information made available to the public under this section, specific site or property locations the disclosure of which would create a risk of harm, theft, or destruction of archeological or Native American cultural resources or to the site at which the sources are located, or would violate any federal law, including the Archaeological Resources Protection Act of 1979, 16 U.S.C. 470w-3, and the National Historic Preservation Act of 1966, 16 U.S.C. 470hh.
- (4)(i) An applicant must make available the information specified in paragraph (b)(3) of this section in a form that is readily accessible, reviewable, and reproducible, at the same time as the information is filed with the Commission or required by regulation to be made available.

- (ii) An applicant must make the information specified in paragraph (b)(3) of this section available to the public for inspection:
- (A) At its principal place of business or at any other location that is more accessible to the public, provided that all the information is available in at least one location;
- (B) During regular business hours;
- (C) In a form that is readily accessible, reviewable and reproducible.
- (iii) The applicant must provide a copy of the complete application (as amended) to a public library or other convenient public office located in each county in which the proposed project is located.
- (iv) An applicant must make requested copies of the information specified in paragraph (b)(3) of this section available either:
- (A) At its principal place of business or at any other location that is more accessible to the public, after obtaining reimbursement for reasonable costs of reproduction; or
- (B) Through the mail, after obtaining reimbursement for postage fees and reasonable costs of reproduction.
- (5) Anyone may file a petition with the Commission requesting access to the information specified in paragraph (b)(3) of this section if it believes that an applicant is not making the information reasonably available for public inspection or reproduction. The petition must describe in detail the basis for the petitioner's belief.
- (6) An applicant must publish notice twice of the filing of its application, no later than 14 days after the filing date, in a daily or weekly newspaper of general circulation in each county in which the project is located. The notice must disclose the filing date of the application and briefly summarize it, including the applicant's name and address, the type of facility applied for, its proposed location, the places where the information specified in paragraph (b)(3) of this section is available for inspection and reproduction, and the date by which any requests for additional scientific studies are due under paragraph (b)(7) of this section, and must state that the Commission will publish subsequent notices soliciting

public participation if the application is found acceptable for filing. The applicant must promptly provide the Commission with proof of the publications of this notice.

- (7) If any resource agency, Indian tribe, or person believes that an additional scientific study should be conducted in order to form an adequate factual basis for a complete analysis of the application on its merits, the resource agency, Indian tribe, or person must file a request for the study with the Commission not later than 60 days after the application is filed and serve a copy of the request on the applicant. The Commission will issue public notice of the tendering for filing of each application for hydropower license or exemption; each such applicant must submit a draft of this notice to the Commission with its application. For any such additional study request, the requester must describe the recommended study and the basis for the request in detail, including who should conduct and participate in the study, its methodology and objectives, whether the recommended study methods are generally accepted in the Scientific community, how the study and information sought will be useful in furthering the resource goals that are affected by the proposed facilities, and approximately how long the study will take to complete, and must explain why the study objectives cannot be achieved using the data already available. In addition, in the case of a study request by a resource agency or Indian tribe that had failed to request the study during the pre-filing consultation process under §4.38 of this part or §16.8 of this chapter, the agency or Indian tribe must explain why this request was not made during the pre-filing consultation process and show good cause why its request for the study should be considered by the Commis-
- (8) An applicant may file a response to any such study request within 30 days of its filing, serving a copy of the response on the requester.
- (9) The requirements of paragraphs (b)(3) to (b)(8) of this section only apply to an application for license or exemption filed on or after May 20, 1991. Paragraphs (b)(3) and (b)(4) of this sec-

tion do not apply to applications subject to the requirements of §16.7 of this chapter.

- (c)(1) Every application for a licensee or exemption for a project with a capacity of 80 megawatts or less must include in its application copies of the statements made under §4.38(b)(1)(vi).
- (2) If an applicant reverses a statement of intent not to seek PURPA benefits:
- (i) Prior to the Commission issuing a license or exemption, the reversal of intent will be treated as an amendment of the application under §4.35 and the applicant must:
- (A) Repeat the pre-filing consultation process under §4.38; and
- (B) Satisfy all the requirements in §292.208 of this chapter; or
- (ii) After the Commission issues a license or exemption for the project, the applicant is prohibited from obtaining PURPA benefits.
- (d) When any application is found to conform to the requirements of paragraphs (a), (b) and (c) of this section, the Commission or its delegate will:
- (1) Notify the applicant that the application has been accepted for filing, specifying the project number assigned and the date upon which the application was accepted for filing, and, for a license or exemption application, direct the filing of the originals (microfilm) of required maps and drawings;
- (2)(i) For an application for a preliminary permit or a license, issue public notice of the application as required in the Federal Power Act;
- (ii) For an application for exemption from licensing, publish notice once in a daily or weekly newspaper of general circulation in each county in which the project is or will be located; and
- (3) If the project affects lands of the United States, notify the appropriate Federal office of the application and the specific lands affected, pursuant to section 24 of the Federal Power Act.
- (4) For an application for a license seeking benefits under section 210 of the Public Utility Regulatory Policies Act of 1978, as amended, for a project that would be located at a new dam or diversion, serve the public notice

issued for the application under paragraph (d)(2)(i) of this section to interested agencies at the time the applicant is notified that the application is accepted for filing.

- (e) In order for an application to conform adequately to the requirements of paragraphs (a), (b) and (c) of this section and of §4.38, an application must be completed fully. No blanks should be left in the application. No material or information required in the application should be omitted. If an applicant believes that its application conforms adequately without containing certain required material or information, it must explain in detail why the material or information is not being submitted and what steps were taken by the applicant to provide the material or information. If the Commission finds that an application does not adequately conform to the requirements of paragraphs (a), (b) and (c) of this section and of §4.38, the Commission or its designee will consider the application either deficient or patently deficient.
- (1) Deficient applications. (i) An application that in the judgment of the Director of the Office of Energy Projects does not conform to the requirements of paragraphs (a), (b) and (c) of this section and of §4.38, may be considered deficient. An applicant having a deficient application will be afforded additional time to correct deficiencies, not to exceed 45 days from the date of notification in the case of an application for a preliminary permit or exemption from licensing or 90 days from the date of notification in the case of an application for license. Notification will be by letter or, in the case of minor deficiencies, by telephone. Any notification will specify the deficiencies to be corrected. Deficiencies must be corrected by submitting the specified materials or information to the Secretary of the Commission within the time specified in the notification of deficiency in accordance with filing procedures posted on the Commission's Web site at http://www.ferc.gov.
- (ii) Upon submission of a conforming application, action will be taken in accordance with paragraph (d) of this section.
- (iii) If the revised application is found not to conform to the require-

- ments of paragraphs (a), (b) and (c) of this section and of §4.38, or if the revisions are not timely submitted, the revised application will be rejected. Procedures for rejected applications are specified in paragraph (e)(2)(iii).
- (2) Patently deficient applications. (i) If, within 90 days of its filing date, the Director of the Office of Energy Projects determines that an application patently fails to substantially comply with the requirements of paragraph (a), (b), and (c) of this section and of §4.38 of this part or §16.8 of this chapter, or is for a project that is precluded by law, the application will be rejected as patently deficient with the specification of the deficiencies that render the application patently deficient.
- (ii) If, after 90 days of its filing date, the Director of the Office of Energy Projects determines that an application patently fails to substantially comply with the requirements of paragraphs (a), (b), and (c) of this section and of § 4.38 of this part or § 16.8 of this chapter, or is for a project that is precluded by law:
- (A) The application will be rejected by order of the Commission, if the Commission determines it is patently deficient; or
- (B) The application will be considered deficient under paragraph (e)(1) of this section, if the Commission determines it is not patently deficient.
- (iii) Any application that is rejected may be resubmitted if the deficiencies are corrected and if, in the case of a competing application, the resubmittal is timely. The date the rejected application is resubmitted will be considered the new filing date for purposes of determining its timeliness under §4.36 and the disposition of competing applications under §4.37.
- (f) Any application will be considered accepted for filing as of the application filing date if the Secretary receives all of the information and documents necessary to conform to the requirements of paragraphs (a), (b) and (c) of this section and of §4.38 within the time prescribed by the Commission or its delegate under paragraph (e) of this section.
- (g) An applicant may be required to submit any additional information or

documents that the Commission or its designee considers relevant for an informed decision on the application. The information or documents must take the form, and must be submitted within the time, that the Commission or its designee prescribes. An applicant may also be required to provide within a specified time additional copies of the complete application, or any of the additional information or documents that are filed, to the Commission or to any person, agency, or other entity that the Commission or its designee specifies. If an applicant fails to provide timely additional information, documents, or copies of submitted materials as required, the Commission or its designee may dismiss the application, hold it in abeyance, or take other appropriate action under this chapter or the Federal Power Act.

- (h) A prospective applicant, prior to submitting its application for filing, may seek advice from the Commission staff regarding the sufficiency of the application. For this purpose, five copies of the draft application should be submitted to the Director of the Division of Hydropower Licensing. An applicant or prospective applicant may confer with the Commission staff at any time regarding deficiencies or other matters related to its application. All conferences are subject to the requirements of §385.2201 of this chapter governing ex parte communications. The opinions or advice of the staff will not bind the Commission or any person delegated authority to act on its behalf.
- (i) Intervention in any preliminary permit proceeding will not constitute intervention in any subsequent licensing or exemption proceeding.
- (j) Any application, the effectiveness of which is conditioned upon the future occurrence of any event or circumstance, will be rejected.
- (k) Critical Energy Infrastructure Information. (1) If this section requires an applicant to reveal Critical Energy Infrastructure Information (CEII), as defined in §388.113(c) of this chapter, to any person, the applicant shall omit the CEII from the information made available and insert the following in its place:

- (i) A statement that CEII is being withheld:
- (ii) A brief description of the omitted information that does not reveal any CEII; and
- (iii) This statement: "Procedures for obtaining access to Critical Energy Infrastructure Information (CEII) may be found at 18 CFR 388.113. Requests for access to CEII should be made to the Commission's CEII Coordinator."
- (2) The applicant, in determining whether information constitutes CEII, shall treat the information in a manner consistent with any filings that applicant has made with the Commission and shall to the extent practicable adhere to any previous determinations by the Commission or the CEII Coordinator involving the same or like information.
- (3) The procedures contained in §§ 388.112 and 388.113 of this chapter regarding designation of, and access to, CEII, shall apply in the event of a challenge to a CEII designation or a request for access to CEII. If it is determined that information is not CEII or that a requester should be granted access to CEII, the applicant will be directed to make the information available to the requester.
- (4) Nothing in this section shall be construed to prohibit any persons from voluntarily reaching arrangements or agreements calling for the disclosure of CEH

[Order 413, 50 FR 11678, Mar. 25, 1985]

EDITORIAL NOTE: For FEDERAL REGISTER citations affecting §4.32, see the List of CFR Sections Affected, which appears in the Finding Aids section of the printed volume and at www.fdsvs.gov.

§4.33 Limitations on submitting applications.

- (a) Limitations on submission and acceptance of a preliminary permit application. The Commission will not accept an application for a preliminary permit for project works that:
- (1) Would develop, conserve, and utilize, in whole or in part, the same water resources that would be developed, conserved, and utilized by a project for which there is an unexpired preliminary permit.

- (2) Would interfere with a licensed project in a manner that, absent the licensee's consent, would be precluded by Section 6 of the Federal Power Act.
- (3) Would develop, conserve, and utilize, in whole or in part, the same water resources that would be developed, conserved, and utilized by a project for which an initial development application has been filed unless the preliminary permit application is filed not later than the time allowed under §4.36(a) for the filing of applications in competition against an initial application for a preliminary permit that would develop, conserve, and utilize, in whole or in part, the same resources.
- (b) Limitations on submissions and acceptance of a license application. The Commission will not accept an application for a license or project works that would develop, conserve, or utilize, in whole or part, the same water resources that would be developed, conserved, and utilized by a project for which there is:
- (1) An unexpired preliminary permit, unless the permittee has submitted an application for license; or
- (2) An unexpired license, except as provided for in Section 15 of the Federal Power Act.
- (c) Limitations on submission and acceptance of an application for a license that would affect an exempted project. (1) Except as permitted under §4.33(c)(2), §4.94(d), or §4.106 (c), (e) or (f), the Commission will not accept an application for a license for project works that are already exempted from licensing under this part.
- (2) If a project is exempted from licensing pursuant to §4.103 or §4.109 and real property interests in any non-Federal lands would be necessary to develop or operate the project, any person who is both a qualified license applicant and has any of those real property interests in non-Federal lands may submit a license application for that project. If a license application is submitted under this clause, any other qualified license applicant may submit a competing license application in accordance with §4.36.
- (d) Limitations on submission and acceptance of exemption applications—(1) Unexpired permit or license. (i) If there is

- an unexpired permit in effect for a project, the Commission will accept an application for exemption of that project from licensing only if the exemption applicant is the permittee. Upon acceptance for filing of the permittee's application, the permit will be considered to have expired.
- (ii) If there is an unexpired license in effect for a project, the Commission will accept an application for exemption of that project from licensing only if the exemption applicant is the licensee.
- (2) Pending license applications. If an accepted license application for a project was submitted by a permittee before the preliminary permit expired, the Commission will not accept an application for exemption of that project from licensing submitted by a person other than the former permittee.
- (3) Submitted by qualified exemption applicant. If the first accepted license application for a project was filed by a qualified exemption applicant, the applicant may request that its license application be treated initially as an application for exemption from licensing by so notifying the Commission in writing and, unless only rights to use or occupy Federal lands would be necessary to develop and operate the project, by submitting documentary evidence showing that the applicant holds the real property interests required under §4.31. Such notice and documentation must be submitted not later than the last date for filing protests or motions to intervene prescribed in the public notice issued for its license application under §4.32(d)(2).
- (e) Priority of exemption applicant's earlier permit or license application. Any accepted preliminary permit or license application submitted by a person who later applies for exemption of the project from licensing will retain its validity and priority under this subpart until the preliminary permit or license application is withdrawn or the project is exempted from licensing.

[Order 413, 50 FR 11680, Mar. 25, 1985, as amended by Order 499, 53 FR 27002, July 18, 1988; Order 2002, 68 FR 51116, Aug. 25, 2003; Order 699, 72 FR 45324, Aug. 14, 2007]

§ 4.34 Hearings on applications; consultation on terms and conditions; motions to intervene; alternative procedures.

(a) Trial-type hearing. The Commission may order a trial-type hearing on an application for a preliminary permit, a license, or an exemption from licensing upon either its own motion or the motion of any interested party of record. Any trial-type hearing will be limited to the issues prescribed by order of the Commission. In all other cases the hearings will be conducted by notice and comment procedures.

(b) Notice and comment hearings. All comments (including mandatory and recommended terms and conditions or prescriptions) on an application for exemption or license must be filed with the Commission no later than 60 days after issuance by the Commission of public notice declaring that the application is ready for environmental analysis. All reply comments must be filed within 105 days of that notice. All comments and reply comments and all other filings described in this section must be served on all persons listed in the service list prepared by the Commission, in accordance with the requirements of §385.2010 of this chapter. If a party or interceder (as defined in §385.2201 of this Chapter) submits any written material to the Commission relating to the merits of an issue that may affect the responsibilities of a particular resource agency, the party or interceder must also serve a copy of the submission on this resource agency. The Commission may allow for longer comment or reply comment periods if appropriate. A commenter or reply commenter may obtain an extension of time from the Commission only upon a showing of good cause or extraordinary circumstances in accordance with §385.2008 of this chapter. Late-filed fish and wildlife ommendations will not be subject to the requirements of paragraphs (e), (f)(1)(ii), and (f)(3) of this section, and late-filed terms and conditions or prescriptions will not be subject to the requirements of paragraphs (f)(1)(iv), (f)(1)(v), and (f)(2) of this section. Latefiled fish and wildlife recommendations, terms and conditions, or prescriptions will be considered by the

Commission under section 10(a) of the Federal Power Act if such consideration would not delay or disrupt the proceeding.

(1) Agencies responsible for mandatory terms and conditions and presentations. Any agency responsible for mandatory terms and conditions or prescriptions for licenses or exemptions, pursuant to sections 4(e), 18, and 30(c) of the Federal Power Act and section 405(d) of the Public Utility Regulatory Policies Act of 1978, as amended, must provide these terms and conditions or prescriptions in its initial comments filed with the Commission pursuant to paragraph (b) of this section. In those comments, the agency must specifically identify and explain the mandatory terms and conditions or prescriptions and their evidentiary and legal basis. In the case of an application prepared other than pursuant to part 5 of this chapter, if ongoing agency proceedings to determine the terms and conditions or prescriptions are not completed by the date specified, the agency must submit to the Commission by the due date:

- (i) Preliminary terms and conditions or prescriptions and a schedule showing the status of the agency proceedings and when the terms and conditions or prescriptions are expected to become final; or
- (ii) A statement waiving the agency's right to file the terms and conditions or prescriptions or indicating the agency does not intend to file terms and conditions or prescriptions.
- (2) Fish and Wildlife agencies and Indian tribes. All fish and wildlife agencies must set forth any recommended terms and conditions for the protection, mitigation of damages to, or enhancement of fish and wildlife, pursuant to the Fish and Wildlife Coordination Act and section 10(j) of the Federal Power Act, in their initial comments filed with the Commission by the date specified in paragraph (b) of this section. All Indian tribes must submit recommendations (including fish and wildlife recommendations) by the same date. In those comments, a fish and wildlife agency or Indian tribe must discuss its understanding of the resource issues presented by the proposed facilities and the evidentiary

basis for the recommended terms and conditions.

- (3) Other Government agencies and members of the public. Resource agencies, other governmental units, and members of the public must file their recommendations in their initial comments by the date specified in paragraph (b) of this section. The comments must clearly identify all recommendations and present their evidentiary basis.
- (4) Submittal of modified recommendations, terms and conditions or prescriptions. (i) If the information and analysis (including reasonable alternatives) presented in a draft environmental document, issued for comment by the Commission, indicate a need to modify the recommendations or terms and conditions or prescriptions previously submitted to the Commission pursuant to paragraphs (b)(1), (b)(2), or (b)(3) of this section, the agency, Indian tribe, or member of the public must file with the Commission any modified recommendations or terms and conditions or prescriptions on the proposed project (and reasonable alternatives) no later than the due date for comments on the draft environmental impact statement. Modified recommendations or terms and conditions or prescriptions must be clearly distinguished from comments on the draft document.
- (ii) If an applicant files an amendment to its application that would materially change the project's proposed plans of development, as provided in §4.35, an agency, Indian tribe or member of the public may modify the recommendations or terms and conditions or prescriptions it previously submitted to the Commission pursuant to paragraphs (b)(1), (b)(2), or (b)(3) of this section no later than the due date specified by the Commission for comments on the amendment.
- (5)(i) With regard to certification requirements for a license applicant under section 401(a)(1) of the Federal Water Pollution Control Act (Clean Water Act), an applicant shall file within 60 days from the date of issuance of the notice of ready for environmental analysis:
- (A) A copy of the water quality certification:

- (B) A copy of the request for certification, including proof of the date on which the certifying agency received the request; or
- (C) Evidence of waiver of water quality certification as described in paragraph (b)(5)(ii) of this section.
- (ii) In the case of an application process using the alternative procedures of paragraph 4.34(i), the filing requirement of paragraph (b)(5)(i) shall apply upon issuance of notice the Commission has accepted the application as provided for in paragraph 4.32(d) of this part.
- (iii) A certifying agency is deemed to have waived the certification requirements of section 401(a)(1) of the Clean Water Act if the certifying agency has not denied or granted certification by one year after the date the certifying agency received a written request for certification. If a certifying agency denies certification, the applicant must file a copy of the denial within 30 days after the applicant received it.
- (iv) Notwithstanding any other provision in title 18, chapter I, subchapter B, part 4, any application to amend an existing license, and any application to amend a pending application for a license, requires a new request for water quality certification pursuant to paragraph (b)(5)(i) of this section if the amendment would have a material adverse impact on the water quality in the discharge from the project or proposed project.
- (c) Additional procedures. If necessary or appropriate the Commission may require additional procedures (e.g., a prehearing conference, further notice and comment on specific issues or oral argument). A party may request additional procedures in a motion that clearly and specifically sets forth the procedures requested and the basis for the request. Replies to such requests may be filed within 15 days of the request.
- (d) Consultation procedures. Pursuant to the Federal Power Act and the Public Utility Regulatory Policies Act of 1978, as amended, the Commission will coordinate as appropriate with other government agencies responsible for mandatory terms and conditions for

exemptions and licenses for hydropower projects. Pursuant to the Federal Power Act and the Fish and Wildlife Coordination Act, the Commission will consult with fish and wildlife agencies concerning the impact of a hydropower proposal on fish and wildlife and appropriate terms and conditions for license to adequately and equitably protect, mitigate damages to, and enhance fish and wildlife (including related spawning grounds and habitat). Pursuant to the Federal Power Act and the Endangered Species Act, the Commission will consult with the U.S. Fish and Wildlife Service or the National Marine Fisheries Service, as appropriate, concerning the impact of a hydropower proposal on endangered or threatened species and their critical

(e) Consultation on recommended fish and wildlife conditions; Section 10(j) process. (1) In connection with its environmental review of an application for license, the Commission will analyze all terms and conditions timely recommended by fish and wildlife agencies pursuant to the Fish and Wildlife Coordination Act for the protection, mitigation of damages to, and enhancement of fish and wildlife (including related spawning grounds and habitat) affected by the development, operation, and management of the proposed project. Submission of such recommendations marks the beginning of the process under section 10(j) of the Federal Power Act.

(2) The agency must specifically identify and explain the recommendations and the relevant resource goals and objectives and their evidentiary or legal basis. The Commission may seek clarification of any recommendation from the appropriate fish and wildlife agency. If the Commission's request for clarification is communicated in writing, copies of the request will be sent by the Commission to all parties, affected resource agencies, and Indian tribes, which may file a response to the request for clarification within the time period specified by the Commission. If the Commission believes any fish and wildlife recommendation may be inconsistent with the Federal Power Act or other applicable law, the Commission will make a preliminary determination of inconsistency in the draft environmental document or, if none, the environmental assessment. The preliminary determination, for any recommendations believed to be inconsistent, shall include an explanation why the Commission believes the recommendation is inconsistent with the Federal Power Act or other applicable law, including any supporting analysis and conclusions, and an explanation of how the measures recommended in the environmental document would adequately and equitably protect, mitigate damages to, and enhance, fish and wildlife (including related spawning grounds and habitat) affected by the development, operation, and management of the project.

(3) Any party, affected resource agency, or Indian tribe may file comments in response to the preliminary determination of inconsistency, including any modified recommendations, within the time frame allotted for comments on the draft environmental document or, if none, the time frame for comments on the environmental analysis. In this filing, the fish and wildlife agency concerned may also request a meeting, telephone or video conference, or other additional procedure to attempt to resolve any preliminary determination of inconsistency.

(4) The Commission shall attempt, with the agencies, to reach a mutually acceptable resolution of any such inconsistency, giving due weight to the recommendations, expertise, and statutory responsibilities of the fish and wildlife agency. If the Commission decides, or an affected resource agency requests, the Commission will conduct a meeting, telephone, or video conference, or other procedures to address issues raised by its preliminary determination of inconsistency and comments thereon. The Commission will give at least 15 days' advance notice to each party, affected resource agency, or Indian tribe, which may participate in the meeting or conference. Any meeting, conference, or additional procedure to address these issues will be scheduled to take place within 90 days of the date the Commission issues a preliminary determination of inconsistency. The Commission will prepare a written summary of any meeting held under this subsection to discuss section 10(j) issues, including any proposed resolutions and supporting analysis, and a copy of the summary will be sent to all parties, affected resource agencies, and Indian tribes.

- (5) The section 10(j) process ends when the Commission issues an order granting or denying the license application in question. If, after attempting to resolve inconsistencies between the fish and wildlife recommendations of a fish and wildlife agency and the purposes and requirements of the Federal Power Act or other applicable law, the Commission does not adopt in whole or in part a fish and wildlife recommendation of a fish and wildlife agency, the Commission will publish the findings and statements required by section 10(j)(2) of the Federal Power Act.
- (f) Licenses and exemption conditions and required findings—(1) License conditions. (i) All licenses shall be issued on the conditions specified in section 10 of the Federal Power Act and such other conditions as the Commission determines are lawful and in the public interest.
- (ii) Subject to paragraph (f)(3) of this section, fish and wildlife conditions shall be based on recommendations timely received from the fish and wildlife agencies pursuant to the Fish and Wildlife Coordination Act.
- (iii) The Commission will consider the timely recommendations of resource agencies, other governmental units, and members of the public, and the timely recommendations (including fish and wildlife recommendations) of Indian tribes affected by the project.
- (iv) Licenses for a project located within any Federal reservation shall be issued only after the findings required by, and subject to any conditions that may be timely received pursuant to, section 4(e) of the Federal Power Act.
- (v) The Commission will require the construction, maintenance, and operation by a licensee at its own expense of such fishways as may be timely prescribed by the Secretary of Commerce or the Secretary of the Interior, as appropriate, pursuant to section 18 of the Federal Power Act.
- (2) Exemption conditions. Any exemption from licensing issued for conduit facilities, as provided in section 30 of

- the Federal Power Act, or for small hydroelectric power projects having a proposed installed capacity of 5,000 kilowatts or less, as provided in section 405(d) of the Public Utility Regulatory Policies Act of 1978, as amended, shall include such terms and conditions as the fish and wildlife agencies may timely determine are appropriate to carry out the responsibilities specified in section 30(c) of the Federal Power Act.
- (3) Required findings. If, after attempting to resolve inconsistencies between the fish and wildlife recommendations of a fish and wildlife agency and the purposes and requirements of the Federal Power Act or other applicable law, the Commission does not adopt in whole or in part a fish and wildlife recommendation of a fish and wildlife agency, the Commission will publish the findings and statements required by section 10(j)(2) of the Federal Power Act.
- (g) Application. The provisions of paragraphs (b) through (d) and (f) of this section apply only to applications for license or exemption; paragraph (e) applies only to applications for license.
- (h) Unless otherwise provided by statute, regulation or order, all filings in hydropower hearings, except those conducted by trial-type procedures, shall conform to the requirements of subpart T of part 385 of this chapter.
- (i) Alternative procedures. (1) An applicant may submit to the Commission a request to approve the use of alternative procedures for pre-filing consultation and the filing and processing of an application for an original, new or subsequent hydropower license or exemption that is subject to §4.38 or §16.8 of this chapter, or for the amendment of a license that is subject to the provisions of §4.38.
- (2) The goal of such alternative procedures shall be to:
- (i) Combine into a single process the pre-filing consultation process, the environmental review process under the National Environmental Policy Act and administrative processes associated with the Clean Water Act and other statutes:
- (ii) Facilitate greater participation by and improve communication among

the potential applicant, resource agencies, Indian tribes, the public and Commission staff in a flexible pre-filing consultation process tailored to the circumstances of each case:

- (iii) Allow for the preparation of a preliminary draft environmental assessment by an applicant or its contractor or consultant, or of a preliminary draft environmental impact statement by a contractor or consultant chosen by the Commission and funded by the applicant;
- (iv) Promote cooperative efforts by the potential applicant and interested entities and encourage them to share information about resource impacts and mitigation and enhancement proposals and to narrow any areas of disagreement and reach agreement or settlement of the issues raised by the hydropower proposal; and
- (v) Facilitate an orderly and expeditious review of an agreement or offer of settlement of an application for a hydropower license, exemption or amendment to a license.
- (3) A potential hydropower applicant requesting the use of alternative procedures must:
- (i) Demonstrate that a reasonable effort has been made to contact all resource agencies, Indian tribes, citizens' groups, and others affected by the applicant's proposal, and that a consensus exists that the use of alternative procedures is appropriate under the circumstances;
- (ii) Submit a communications protocol, supported by interested entities, governing how the applicant and other participants in the pre-filing consultation process, including the Commission staff, may communicate with each other regarding the merits of the applicant's proposal and proposals and recommendations of interested entities; and
- (iii) Serve a copy of the request on all affected resource agencies and Indian tribes and on all entities contacted by the applicant that have expressed an interest in the alternative pre-filing consultation process.
- (4) As appropriate under the circumstances of the case, the alternative procedures should include provisions for:

- (i) Distribution of an initial information package and conduct of an initial information meeting open to the public:
- (ii) The cooperative scoping of environmental issues (including necessary scientific studies), the analysis of completed studies and any further scoping; and
- (iii) The preparation of a preliminary draft environmental assessment or preliminary draft environmental impact statement and related application.
- (5)(i) If the potential applicant's request to use the alternative procedures is filed prior to July 23, 2005, the Commission will give public notice in the FEDERAL REGISTER inviting comment on the applicant's request to use alternative procedures. The Commission will consider any such comments in determining whether to grant or deny the applicant's request to use alternative procedures. Such a decision will not be subject to interlocutory rehearing or appeal.
- (ii) If the potential applicant's request to use the alternative procedures is filed on or after July 23, 2005 and prior to the deadline date for filing a notification of intent to seek a new or subsequent license required by §5.5 of this chapter, the Commission will give public notice and invite comments as provided for in paragraph (i)(5)(i) of this section. Commission approval of the potential applicant's request to use the alternative procedures prior to the deadline date for filing of the notification of intent does not waive the potential applicant's obligation to file the notification of intent required by §5.5 of this chapter and Pre-Application Document required by §5.6 of this chap-
- (iii) If the potential applicant's request to use the alternative procedures is filed on or after July 23, 2005 and is at the same time as the notification of intent to seek a new or subsequent license required by §5.5, the public notice and comment procedures of part 5 of this chapter shall apply.
- (6) If the Commission accepts the use of alternative procedures, the following provisions will apply.
- (i) To the extent feasible under the circumstances of the proceeding, the Commission will give notice in the

FEDERAL REGISTER and the applicant will give notice, in a local newspaper of general circulation in the county or counties in which the project is located, of the initial information meeting and the scoping of environmental issues. The applicant will also send notice of these stages to a mailing list approved by the Commission.

(ii) Every six months, the applicant shall file with the Commission a report summarizing the progress made in the pre-filing consultation process and referencing the applicant's public file, where additional information on that process can be obtained. Summaries or minutes of meetings held in the process may be used to satisfy this filing requirement. The applicant must also file with the Commission a copy of its initial information package, scoping document, and the preliminary draft environmental review document. All filings with the Commission under this section must include the number of copies required by paragraph (h) of this section, and the applicant shall send a copy of these filings to each participant that requests a copy.

(iii) At a suitable location, the applicant will maintain a public file of all relevant documents, including scientific studies, correspondence, and minutes or summaries of meetings, compiled during the pre-filing consultation process. The Commission will maintain a public file of the applicant's initial information package, scoping documents, periodic reports on the pre-filing consultation process, and the preliminary draft environmental review document.

(iv) An applicant authorized to use alternative procedures may substitute a preliminary draft environmental review document and additional material specified by the Commission instead of Exhibit E to its application and need not supply additional documentation of the pre-filing consultation process. The applicant will file with the Commission the results of any studies conducted or other documentation as directed by the Commission, either on its own motion or in response to a motion by a party to the licensing or exemption proceeding.

(v) Pursuant to the procedures approved, the participants will set rea-

sonable deadlines requiring all resource agencies, Indian tribes, citizens' groups, and interested persons to submit to the applicant requests for scientific studies during the pre-filing consultation process, and additional requests for studies may be made to the Commission after the filing of the application only for good cause shown.

(vi) During the pre-filing process the Commission may require the filing of preliminary fish and wildlife recommendations, prescriptions, mandatory conditions, and comments, to be submitted in final form after the filing of the application; no notice that the application is ready for environmental analysis need be given by the Commission after the filing of an application pursuant to these procedures.

(vii) Any potential applicant, resource agency, Indian tribe, citizens' group, or other entity participating in the alternative pre-filing consultation process may file a request with the Commission to resolve a dispute concerning the alternative process (including a dispute over required studies), but only after reasonable efforts have been made to resolve the dispute with other participants in the process. No such request shall be accepted for filing unless the entity submitting it certifies that it has been served on all other participants. The request must document what efforts have been made to resolve the dispute.

(7) If the potential applicant or any resource agency, Indian tribe, citizens' group, or other entity participating in the alternative pre-filing consultation process can show that it has cooperated in the process but a consensus supporting the use of the process no longer exists and that continued use of the alternative process will not be productive, the participant may petition the Commission for an order directing the use by the potential applicant of appropriate procedures to complete its application. No such request shall be accepted for filing unless the entity submitting it certifies that it has been served on all other participants. The request must recommend specific procedures that are appropriate under the circumstances.

(8) The Commission may participate in the pre-filing consultation process

§ 4.35

and assist in the integration of this process and the environmental review process in any case, including appropriate cases where the applicant, contractor, or consultant funded by the applicant is not preparing a preliminary draft environmental assessment or preliminary draft environmental impact statement, but where staff assistance is available and could expedite the proceeding.

(9) If this section requires an applicant to reveal Critical Energy Infrastructure Information (CEII), as defined by §388.113(c) of this chapter, to any person, the applicant shall follow the procedures set out in §4.32(k).

[Order 533, 56 FR 23148, May 20, 1991, as amended at 56 FR 61155, Dec. 2, 1991; Order 540, 57 FR 21737, May 22, 1992; Order 596, 62 FR 59810, Nov. 5, 1997; Order 2002, 68 FR 51116, Aug. 25, 2003; Order 643, 68 FR 52094, Sept. 2, 2003; 68 FR 61742, Oct. 30, 2003]

§4.35 Amendment of application; date of acceptance.

- (a) General rule. Except as provided in paragraph (d) of this section, if an applicant amends its filed application as described in paragraph (b) of this section, the date of acceptance of the application under §4.32(f) is the date on which the amendment to the applicant was filed.
- (b) Paragraph (a) of this section applies if an applicant:
- (1) Amends its filed license or preliminary permit application in order to change the status or identity of the applicant or to materially amend the proposed plans of development; or
- (2) Amends its filed application for exemption from licensing in order to materially amend the proposed plans of development, or
- (3) Amends its filed application in order to change its statement of intent of whether or not it will seek benefits under section 210 of PURPA, as originally filed under §4.32(c)(1).
- (c) An application amended under paragraph (a) is a new filing for:
- (1) The purpose of determining its timeliness under §4.36 of this part;
- (2) Disposing of competing applications under §4.37; and
- (3) Reissuing public notice of the application under §4.32(d)(2).

- (d) If an application is amended under paragraph (a) of this section, the Commission will rescind any acceptance letter already issued for the application.
- (e) *Exceptions*. This section does not apply to:
- (1) Any corrections of deficiencies made pursuant to §4.32(e)(1);
- (2) Any amendments made pursuant to §4.37(b)(4) by a State or a municipality to its proposed plans of development to make them as well adapted as the proposed plans of an applicant that is not a state or a municipality;
- (3) Any amendments made pursuant to §4.37(c)(2) by a priority applicant to its proposed plans of development to make them as well adapted as the proposed plans of an applicant that is not a priority applicant;
- (4) Any amendments made by a license or an exemption applicant to its proposed plans of development to satisfy requests of resource agencies or Indian tribes submitted after an applicant has consulted under §4.38 or concerns of the Commission; and
- (5)(i) Any license or exemption applicant with a project located at a new dam or diversion who is seeking PURPA benefits and who:
- (A) Has filed an adverse environmental effects (AEE) petition pursuant to §292.211 of this chapter; and
- (B) Has proposed measures to mitigate the adverse environmental effects which the Commission, in its initial determination on the AEE petition, stated the project will have.
- (ii) This exception does not protect any proposed mitigative measures that the Commission finds are a pretext to avoid the consequences of materially amending the application or are outside the scope of mitigating the adverse environmental effects.
- (f) Definitions. (1) For the purposes of this section, a material amendment to plans of development proposed in an application for a license or exemption from licensing means any fundamental and significant change, including but not limited to:
- (i) A change in the installed capacity, or the number or location of any generating units of the proposed project if the change would significantly modify

the flow regime associated with the project;

- (ii) A material change in the location, size, or composition of the dam, the location of the powerhouse, or the size and elevation of the reservoir if the change would:
- (A) Enlarge, reduce, or relocate the area of the body of water that would lie between the farthest reach of the proposed impoundment and the point of discharge from the powerhouse; or
- (B) Cause adverse environmental impacts not previously discussed in the original application; or
- (iii) A change in the number of discrete units of development to be included within the project boundary.
- (2) For purposes of this section, a material amendment to plans of development proposed in an application for a preliminary permit means a material change in the location of the powerhouse or the size and elevation of the reservoir if the change would enlarge, reduce, or relocate the area of the body of water that would lie between the farthest reach of the proposed impoundment and the point of discharge from the powerhouse.
- (3) For purposes of this section, a change in the status of an applicant means:
- (i) The acquisition or loss of preference as a state or a municipality under section 7(a) of the Federal Power Act; or
- (ii) The loss of priority as a permittee under section 5 of the Federal Power Act.
- (4) For purposes of this section, a change in the identity of an applicant means a change that either singly, or together with previous amendments, causes a total substitution of all the original applicants in a permit or a license application.

[Order 413, 50 FR 11680, Mar. 25, 1985, as amended by Order 499, 53 FR 27002, July 18, 1988; Order 533, 56 FR 23149, May 20, 1991; Order 2002, 68 FR 51115, Aug. 25, 2003]

§ 4.36 Competing applications: deadlines for filing; notices of intent; comparisons of plans of development.

The public notice of an initial preliminary permit application or an initial development application shall pre-

- scribe the deadline for filing protests and motions to intervene in that proceeding (the *prescribed intervention deadline*).
- (a) Deadlines for filing applications in competition with an initial preliminary permit application. (1) Any preliminary permit application or any development application not filed pursuant to a notice of intent must be submitted for filing in competition with an initial preliminary permit application not later than the prescribed intervention deadline.
- (2) Any preliminary permit application filed pursuant to a notice of intent must be submitted for filing in competition with an initial preliminary permit application not later than 30 days after the prescribed intervention deadline.
- (3) Any development application filed pursuant to a notice of intent must be submitted for filing in competition with an initial preliminary permit application not later than 120 days after the prescribed intervention deadline.
- (b) Deadlines for filing applications in competition with an initial development application. (1) Any development application not filed pursuant to a notice of intent must be submitted for filing in competition with an initial development application not later than the prescribed intervention deadline.
- (2) Any development application filed pursuant to a notice of intent must be submitted for filing in competition with an initial development application not later than 120 days after the prescribed intervention deadline.
- (3) If the Commission has accepted an application for exemption of a project from licensing and the application has not yet been granted or denied, the applicant for exemption may submit a license application for the project if it is a qualified license applicant. The pending application for exemption from licensing will be considered withdrawn as of the date the Commission accepts the license application for filing. If a license application is accepted for filing under this provision, any qualified license applicant may submit a competing license application not later than the prescribed intervention deadline set for the license application.

- (4) Any preliminary permit application must be submitted for filing in competition with an initial development application not later than the deadlines prescribed in paragraphs (a)(1) and (a)(2) for the submission of preliminary permit applications filed in competition with an initial preliminary permit application.
- (c) Notices of intent. (1) Any notice of intent to file an application in competition with an initial preliminary permit or an initial development application must be submitted for filing not later than the prescribed intervention deadline for the initial application.
 - (2) A notice of intent must include:
- (i) The exact name, business address, and telephone number of the prospective applicant; and
- (ii) An unequivocal statement of intent to submit a preliminary permit application or a development application (specify which type of application).
- (d) Requirements for competing applications. (1) Any competing application must:
- (i) Conform to all requirements for filing an initial application; and
- (ii) Include proof of service of a copy of the competing application on the person(s) designated in the public notice of the initial application for service of pleadings, documents, or communications concerning the initial application.
- (2) Comparisons of plans of development. (i) After the deadline for filing applications in competition against an initial development application has expired, the Commission will notify each license and exemption applicant of the identity of the other applicants.
- (ii) Not later than 14 days after the Commission serves the notification described in paragraph (d)(2)(i) of this section, if a license or exemption applicant has not already done so, it must serve a copy of its application on each of the other license and exemption applicants.
- (iii) Not later than 60 days after the Commission serves the notification described in paragraph (d)(2)(i) of this section, each license and exemption applicant must file with the Commission a detailed and complete statement of how its plans are as well or better

adapted than are the plans of each of the other license and exemption applicants to develop, conserve, and utilize in the public interest the water resources of the region. These statements should be supported by any technical analyses that the applicant deems appropriate to support its proposed plans of development.

[Order 413, 50 FR 11680, Mar. 25, 1985; 50 FR 23947, June 7, 1985]

§4.37 Rules of preference among competing applications.

Except as provided in §4.33(e), the Commission will select among competing applications on the following bases:

- (a) If an accepted application for a preliminary permit and an accepted application for a license propose project works that would develop, conserve, and utilize, in whole or in part, the same water resources, and the applicant for a license has demonstrated its ability to carry out its plans, the Commission will favor the license applicant unless the permit applicant substantiates in its filed application that its plans are better adapted to develop, conserve, and utilize in the public interest the water resources of the region
- (b) If two or more applications for preliminary permits or two or more applications for licenses (not including applications for a new license under section 15 of the Federal Power Act) are filed by applicants for project works that would develop, conserve, and utilize, in whole or in part, the same water resources, and if none of the applicants is a preliminary permittee whose application for license was accepted for filing within the permit period, the Commission will select between or among the applicants on the following bases:
- (1) If both or neither of two applicants are either a municipality or a state, the Commission will favor the applicant whose plans are better adapted to develop, conserve, and utilize in the public interest the water resources of the region, taking into consideration the ability of each applicant to carry out its plans.

- (2) If both of two applicants are either a municipality or a state, or neither of them is a municipality or a state, and the plans of the applicants are equally well adapted to develop, conserve, and utilize in the public interest the water resources of the region, taking into consideration the ability of each applicant to carry out its plans, the Commission will favor the applicant with the earliest application acceptance date.
- (3) If one of two applicants is a municipality or a state, and the other is not, and the plans of the municipality or a state are at least as well adapted to develop, conserve, and utilize in the public interest the water resources of the region, the Commission will favor the municipality or state.
- (4) If one of two applicant is a municipality or a state, and the other is not, and the plans of the applicant who is not a municipality or a state are better adapted to develop, conserve, and utilize in the public interest the water resources of the region, the Commission will inform the municipality or state of the specific reasons why its plans are not as well adapted and afford a reasonable period of time for the municipality or state to render its plans at least as well adapted as the other plans. If the plans of the municipality or state are rendered at least as well adapted within the time allowed, the Commission will favor the municipality or state. If the plans are not rendered at least as well adapted within the time allowed, the Commission will favor the other applicant.
- (c) If two or more applications for licenses are filed for project works which would develop, conserve, and utilize, in whole or in part, the same water resources, and one of the applicants was a preliminary permittee whose application was accepted for filing within the permit period (priority applicant), the Commission will select between or among the applicants on the following bases:
- (1) If the plans of the priority applicant are at least as well adapted as the plans of each other applicant to develop, conserve, and utilize in the public interest the water resources of the region, taking into consideration the ability of each applicant to carry out

its plans, the Commission will favor the priority applicant.

- (2) If the plans of an applicant who is not a priority applicant are better adapted than the plans of the priority applicant to develop, conserve, and utilize in the public interest the water resources of the region, taking into consideration the ability of each applicant to carry out its plans, the Commission will inform the priority applicant of the specific reasons why its plans are not as well adapted and afford a reasonable period of time for the priority applicant to render its plans at least as well adapted as the other plans. If the plans of the priority applicant are rendered at least as well adapted within the time allowed, then the Commission will favor the priority applicant. If the plans of the priority applicant are not rendered as well adapted within the time allowed, the criteria specified in paragraph (b) will govern.
- (3) The criteria specified in paragraph (b) will govern selection among applicants other than the priority applicant.
- (d) With respect to a project for which an application for an exemption from licensing has been accepted for filing, the Commission will select among competing applications on the following bases:
- (1) If an accepted application for a preliminary permit and an accepted application for exemption from licensing propose to develop mutually exclusive small hydroelectric power projects, the Commission will favor the applicant whose substantiated plans in the application received by the Commission are better adapted to develop, conserve, and utilize in the public interest the water resources of the region. If the substantiated plans are equally well adapted, the Commission will favor the application for exemption from licensing.
- (2) If an application for a license and an application for exemption from licensing, or two or more applications for exemption from licensing are each accepted for filing and each proposes to develop a mutually exclusive project, the Commission will favor the applicant whose plans are better adapted to develop, conserve, and utilize in the public interest the water resources of

the region. If the plans are equally well adapted, the Commission will favor the applicant with the earliest application acceptance date.

(e) A municipal applicant must provide evidence that the municipality is competent under applicable state and local laws to engage in the business of developing, transmitting, utilizing, or distributing power, or such applicant will be considered a non-municipal applicant for the purpose of determining the disposition of competing applications.

[Order 413, 50 FR 11682, Mar. 25, 1985, as amended by Order 2002, 68 FR 51117, Aug. 25,

§ 4.38 Consultation requirements.

- (a) Requirement to consult. (1) Before it files any application for an original license or an exemption from licensing that is described in paragraph (a)(4) of this section, a potential applicant must consult with the relevant Federal, State, and interstate resource agencies, including the National Marine Fisheries Service, the United States Fish and Wildlife Service, the National Park Service, the United States Environmental Protection Agency, the Federal agency administering any United States lands or facilities utilized or occupied by the project, the appropriate State fish and wildlife agencies, the appropriate State water resource management agencies, the certifying agency under section 401(a)(1) of the Federal Water Pollution Control Act (Clean Water Act), 33 U.S.C. §1341(c)(1), and any Indian tribe that may be affected by the proposed project.
- (2) Each requirement in this section to contact or consult with resource agencies or Indian tribes shall be construed to require as well that the potential applicant contact or consult with members of the public.
- (3) If a potential applicant for an original license commences first stage pre-filing consultation on or after July 23, 2005 it shall file a notification of intent to file a license application pursuant to §5.5 and a pre-application document pursuant to the provisions of §5.6.
- (4) The Director of the Office of Energy Projects will, upon request, provide a list of known appropriate Fed-

- eral, state, and interstate resource agencies, Indian tribes, and local, regional, or national non-governmental organizations likely to be interested in any license application proceeding.
- (5) An applicant for an exemption from licensing or an applicant for a license seeking benefits under section 210 of the Public Utility Regulatory Policies Act, as amended, for a project that would be located at a new dam or diversion must, in addition to meeting the requirements of this section, comply with the consultation requirements in §4.301.
- (6) The pre-filing consultation requirements of this section apply only to an application for:
- (i) Original license;
- (ii) Exemption;
- (iii) Amendment to an application for original license or exemption that materially amends the proposed plans of development as defined in §4.35(f)(1):
- (iv) Amendment to an existing license that would increase the capacity of the project as defined in §4.201(b); or
- (v) Amendment to an existing license that would not increase the capacity of the project as defined in §4.201(b), but that would involve:
- (A) The construction of a new dam or diversion in a location where there is no existing dam or diversion:
- (B) Any repair, modification, or reconstruction of an existing dam that would result in a significant change in the normal maximum surface area or elevation of an existing impoundment;
- (C) The addition of new water power turbines other than to replace existing turbines.
- (7) Before it files a non-capacity reamendment as defined in lated §4.201(c), an applicant must consult with the resource agencies and Indian tribes listed in paragraph (a)(1) of this section to the extent that the proposed amendment would affect the interests of the agencies or tribes. When consultation is necessary, the applicant must, at a minimum, provide the resource agencies and Indian tribes with copies of the draft application and allow them at least 60 days to comment on the proposed amendment. The amendment as filed with the Commission must summarize the consultation

with the resource agencies and Indian tribes on the proposed amendment, propose reasonable protection, mitigation, or enhancement measures to respond to impacts identified as being caused by the proposed amendment, and respond to any objections, recommendations, or conditions submitted by the agencies or Indian tribes. Copies of all written correspondence between the applicant, the agencies, and the tribes must be attached to the application.

- (8) This section does not apply to any application for a new license, a nonpower license, a subsequent license, or surrender of a license subject to sections 14 and 15 of the Federal Power Act.
- (9) If a potential applicant has any doubt as to whether a particular application or amendment would be subject to the pre-filing consultation requirements of this section or if a waiver of the pre-filing requirements would be appropriate, the applicant may file a written request for clarification or waiver with the Director, Office of Energy Projects.
- (b) First stage of consultation. (1) A potential applicant for an original license that commences pre-filing consultation on or after July 23, 2005 must, at the time it files its notification of intent to seek a license pursuant to §5.5 of this chapter and a pre-application document pursuant to §5.6 of this chapter and, at the same time, provide a copy of the pre-application document to the entities specified in §5.6(a) of this chapter.
- (2) A potential applicant for an original license that commences pre-filing consultation under this part prior to July 23, 2005 or for an exemption must promptly contact each of the appropriate resource agencies, affected Indian tribes, and members of the public likely to be interested in the proceeding; provide them with a description of the proposed project and supporting information; and confer with them on project design, the impact of the proposed project (including a description of any existing facilities, their operation, and any proposed changes), reasonable hydropower alternatives, and what studies the applicant should conduct. The potential appli-

cant must provide to the resource agencies, Indian tribes and the Commission the following information:

- (i) Detailed maps showing project boundaries, if any, proper land descriptions of the entire project area by township, range, and section, as well as by state, county, river, river mile, and closest town, and also showing the specific location of all proposed project facilities, including roads, transmission lines, and any other appurtenant facilities:
- (ii) A general engineering design of the proposed project, with a description of any proposed diversion of a stream through a canal or penstock;
- (iii) A summary of the proposed operational mode of the project;
- (iv) Identification of the environment to be affected, the significant resources present, and the applicant's proposed environmental protection, mitigation, and enhancement plans, to the extent known at that time:
- (v) Streamflow and water regime information, including drainage area, natural flow periodicity, monthly flow rates and durations, mean flow figures illustrating the mean daily streamflow curve for each month of the year at the point of diversion or impoundment, with location of the stream gauging station, the method used to generate the streamflow data provided, and copies of all records used to derive the flow data used in the applicant's engineering calculations;
- (vi) (A) A statement (with a copy to the Commission) of whether or not the applicant will seek benefits under section 210 of PURPA by satisfying the requirements for qualifying hydroelectric small power production facilities in § 292.203 of this chapter;
- (B) If benefits under section 210 of PURPA are sought, a statement on whether or not the applicant believes diversion (as that term is defined in §292.202(p) of this chapter) and a request for the agencies' view on that belief, if any;
- (vii) Detailed descriptions of any proposed studies and the proposed methodologies to be employed; and
- (viii) Any statement required by §4.301(a) of this part.
- (3) (i) A potential exemption applicant and a potential applicant for an

original license that commences prefiling consultation;

- (A) On or after July 23, 2005 pursuant to part 5 of this chapter and receives approval from the Commission to use the license application procedures of part 4 of this chapter; or
- (B) Elects to commence pre-filing consultation under part 4 of this chapter prior to July 23, 2005; must:
- (I) Hold a joint meeting at a convenient place and time, including an opportunity for a site visit, with all pertinent agencies, Indian tribes, and members of the public to explain the applicant's proposal and its potential environmental impact, to review the information provided, and to discuss the data to be obtained and studies to be conducted by the potential applicant as part of the consultation process;
- (2) Consult with the resource agencies, Indian tribes and members of the public on the scheduling and agenda of the joint meeting; and
- (3) No later than 15 days in advance of the joint meeting, provide the Commission with written notice of the time and place of the meeting and a written agenda of the issues to be discussed at the meeting.
- (ii) The joint meeting must be held no earlier than 30 days, but no later than 60 days, from, as applicable;
- (A) The date of the Commission's approval of the potential applicant's request to use the license application procedures of this part pursuant to the provisions of part 5 of this chapter; or
- (B) The date of the potential applicant's letter transmitting the information required by paragraph (b)(2) of this section, in the case of a potential exemption applicant or a potential license applicant that commences prefiling consultation under this part prior to July 23, 2005.
- (4) Members of the public must be informed of and invited to attend the joint meeting held pursuant to paragraph (b)(3) of this section by means of the public notice provision published in accordance with paragraph (g) of this section. Members of the public attending the meeting are entitled to participate in the meeting and to express their views regarding resource issues that should be addressed in any appli-

- cation for license or exemption that may be filed by the potential applicant. Attendance of the public at any site visit held pursuant to paragraph (b)(3) of this section will be at the discretion of the potential applicant. The potential applicant must make either audio recordings or written transcripts of the joint meeting, and must promptly provide copies of these recordings or transcripts to the Commission and, upon request, to any resource agency, Indian tribe, or member of the public.
- (5) Not later than 60 days after the joint meeting held under paragraph (b)(3) of this Section (unless extended within this time period by a resource agency, Indian tribe, or members of the public for an additional 60 days by sending written notice to the applicant and the Director of the Office of Energy Projects within the first 60 day period, with an explanation of the basis for the extension), each interested resource agency and Indian tribe must provide a potential applicant with written comments:
- (i) Identifying its determination of necessary studies to be performed or the information to be provided by the potential applicant;
- (ii) Identifying the basis for its determination;
- (iii) Discussing its understanding of the resource issues and its goals and objectives for these resources;
- (iv) Explaining why each study methodology recommended by it is more appropriate than any other available methodology alternatives, including those identified by the potential applicant pursuant to paragraph (b)(2)(vii) of this section;
- (v) Documenting that the use of each study methodology recommended by it is a generally accepted practice; and
- (vi) Explaining how the studies and information requested will be useful to the agency, Indian tribe, or member of the public in furthering its resource goals and objectives that are affected by the proposed project.
- (6)(i) If a potential applicant and a resource agency or Indian tribe disagree as to any matter arising during the first stage of consultation or as to the need to conduct a study or gather information referenced in paragraph

(c)(2) of this section, the potential applicant or resource agency or Indian tribe may refer the dispute in writing to the Director of the Office of Energy Projects (Director) for resolution.

(ii) At the same time as the request for dispute resolution is submitted to the Director, the entity referring the dispute must serve a copy of its written request for resolution on the disagreeing party and any affected resource agency or Indian tribe, which may submit to the Director a written response to the referral within 15 days of the referral's submittal to the Director.

(iii) Written referrals to the Director and written responses thereto pursuant to paragraphs (b)(6)(i) or (b)(6)(ii) of this section must be filed with the Commission in accordance with the Commission's Rules of Practice and Procedure, and must indicate that they are for the attention of the Director pursuant to §4.38(b)(6).

(iv) The Director will resolve the disputes by letter provided to the potential applicant and all affected resource agencies and Indian tribes.

(v) If a potential applicant does not refer a dispute regarding a request for a potential applicant to obtain information or conduct studies (other than a dispute regarding the information specified in paragraph (b)(2) of this section), or a study to the Director under paragraph (b)(6) of this section, or if a potential applicant disagrees with the Director's resolution of a dispute regarding a request for information (other than a dispute regarding the information specified in paragraph (b)(2) of this section) or a study, and if the potential applicant does not provide the requested information or conduct the requested study, the potential applicant must fully explain the basis for its disagreement in its application.

(vi) Filing and acceptance of an application will not be delayed, and an application will not be considered deficient or patently deficient pursuant to §4.32(e)(1) or (e)(2) of this part, merely because the application does not include a particular study or particular information if the Director had previously found, under paragraph (b)(6)(iv) of this section, that each study or information is unreasonable

or unnecessary for an informed decision by the Commission on the merits of the application or use of the study methodology requested is not a generally accepted practice.

(7) The first stage of consultation ends when all participating agencies and Indian tribes provide the written comments required under paragraph (b)(5) of this section or 60 days after the joint meeting held under paragraph (b)(3) of this section, whichever occurs first, unless a resource agency or Indian tribe timely notifies the applicant and the Director of Energy Projects of its need for more time to provide written comments under paragraph (b)(5) of this section, in which case the first stage of consultation ends when all participating agencies and Indian tribes provide the written comments required under paragraph (b)(5) of this section or 120 days after the joint meeting held under paragraph (b)(5) of this section, whichever occurs first.

(c) Second stage of consultation. (1) Unless determined to be unnecessary by the Director pursuant to paragraph (b)(6) of this section, a potential applicant must diligently conduct all reasonable studies and obtain all reasonable information requested by resource agencies and Indian tribes under paragraph (b) of this section that are necessary for the Commission to make an informed decision regarding the merits of the application. These studies must be completed and the information obtained:

- (i) Prior to filing the application, if the results:
- (A) Would influence the financial (e.g., instream flow study) or technical feasibility of the project (e.g., study of potential mass soil movement); or
- (B) Are needed to determine the design or location of project features, reasonable alternatives to the project, the impact of the project on important natural or cultural resources (e.g., resource surveys), or suitable mitigation or enhancement measures, or to minimize impact on significant resources (e.g., wild and scenic river, anadromous fish, endangered species, caribou migration routes);

- (ii) After filing the application but before issuance of a license or exemption, if the applicant otherwise complied with the provisions of paragraph (b)(2) of this section and the study or information gathering would take longer to conduct and evaluate than the time between the conclusion of the first stage of consultation and the expiration of the applicant's preliminary permit or the application filing deadline set by the Commission;
- (iii) After a new license or exemption is issued, if the studies can be conducted or the information obtained only after construction or operation of proposed facilities, would determine the success of protection, mitigation, or enhancement measures (e.g., post-construction monitoring studies), or would be used to refine project operation or modify project facilities.
- (2) If, after the end of the first stage of consultation as defined in paragraph (b)(7) of this section, a resource agency or Indian tribe requests that the potential applicant conduct a study or gather information not previously identified and specifies the basis and reasoning for its request, under paragraphs (b)(5) (i)-(vi) of this section, the potential applicant must promptly initiate the study or gather the information, unless the study or information is unreasonable or unnecessary for an informed decision by the Commission on the merits of the application or use of the methodology requested by a resource agency or Indian tribe for conducting the study is not a generally accepted practice. The applicant may refer any such request to the Director of the Office of Energy Projects for dispute resolution under the procedures set forth in paragraph (b)(6) of this section and need not conduct prior to filing any study determined by the Director to be unreasonable or unnecessary or to employ a methodology that is not generally accepted.
- (3)(i) The results of studies and information-gathering referenced in paragraphs (c)(1)(ii) and (c)(2) of this section will be treated as additional information; and
- (ii) Filing and acceptance of an application will not be delayed and an application will not be considered deficient or patently deficient pursuant to §4.32

- (e)(1) or (e)(2) merely because the study or information gathering is not complete before the application is filed.
- (4) A potential applicant must provide each resource agency and Indian tribe with:
- (i) A copy of its draft application that:
- (A) Indicates the type of application the potential applicant expects to file with the Commission; and
- (B) Responds to any comments and recommendations made by any resource agency and Indian tribe either during the first stage of consultation or under paragraph (c)(2) of this section:
- (ii) The results of all studies and information-gathering either requested by that resource agency or Indian tribe in the first stage of consultation (or under paragraph (c)(2) of this section if available) or which pertain to resources of interest to that resource agency or Indian tribe and which were identified by the potential applicant pursuant to paragraph (b)(2)(vii) of this section, including a discussion of the results and any proposed protection, mitigation, or enhancement measures; and
- (iii) A written request for review and comment.
- (5) A resource agency or Indian tribe will have 90 days from the date of the potential applicant's letter transmitting the paragraph (c)(4) information to it to provide written comments on the information submitted by a potential applicant under paragraph (c)(4) of this section.
- (6) If the written comments provided under paragraph (c)(5) of this section indicate that a resource agency or Indian tribe has a substantive disagreement with a potential applicant's conclusions regarding resource impacts or its proposed protection, mitigation, or enhancement measures, the potential applicant will:
- (i) Hold a joint meeting with the disagreeing resource agency or Indian tribe and other agencies with similar or related areas of interest, expertise, or responsibility not later than 60 days from the date of the written comments of the disagreeing agency or Indian tribe to discuss and to attempt to

reach agreement on its plan for environmental protection, mitigation, or enhancement measures;

- (ii) Consult with the disagreeing agency or Indian tribe and other agencies with similar or related areas of interest, expertise, or responsibility on the scheduling of the joint meeting; and
- (iii) At least 15 days in advance of the meeting, provide the Commission with written notice of the time and place of the meeting and a written agenda of the issues to be discussed at the meeting.
- (7) The potential applicant and any disagreeing resource agency or Indian tribe may conclude a joint meeting with a document embodying any agreement among them regarding environmental protection, mitigation, or enhancement measures and any issues that are unresolved.
- (8) The potential applicant must describe all disagreements with a resource agency or Indian tribe on technical or environmental protection, mitigation, or enhancement measures in its application, including an explanation of the basis for the applicant's disagreement with the resource agency or Indian tribe, and must include in its application any document developed pursuant to paragraph (c)(7) of this section.
- (9) A potential applicant may file an application with the Commission if:
- (i) It has complied with paragraph (c)(4) of this section and no resource agency or Indian tribe has responded with substantive disagreements by the deadline specified in paragraph (c)(5) of this section; or
- (ii) It has complied with paragraph (c)(6) of this section and a resource agency or Indian tribe has responded with substantive disagreements.
- (10) The second stage of consultation ends:
- (i) Ninety days after the submittal of information pursuant to paragraph (c)(4) of this section in cases where no resource agency or Indian tribe has responded with substantive disagreements; or
- (ii) At the conclusion of the last joint meeting held pursuant to paragraph (c)(6) of this section in cases where a resource agency or Indian tribe has re-

- sponded with substantive disagreements.
- (d) Third stage of consultation. (1) The third stage of consultation is initiated by the filing of an application for a license or exemption, accompanied by a transmittal letter certifying that at the same time copies of the application are being mailed to the resource agencies, Indian tribes, other government offices, and consulted members of the public specified in paragraph (d)(2) of this section.
- (2) As soon as an applicant files such application documents with the Commission, or promptly after receipt in the case of documents described in paragraph (d)(2)(iii) of this section, as the Commission may direct the applicant must serve on every resource agency, Indian tribes, and member of the public consulted, and on other government offices copies of:
- (i) Its application for a license or an exemption from licensing;
- (ii) Any deficiency correction, revision, supplement, response to additional information request, or amendment to the application; and
- (iii) Any written correspondence from the Commission requesting the correction of deficiencies or the submittal of additional information.
- (e) Waiver of compliance with consultation requirements. (1) If a resource agency or Indian tribe waives in writing compliance with any requirement of this section, a potential applicant does not have to comply with that requirement as to that agency or tribe.
- (2) If a resource agency or Indian tribe fails to timely comply with a provision regarding a requirement of this section, a potential applicant may proceed to the next sequential requirement of this section without waiting for the resource agency or Indian tribe to comply.
- (3) The failure of a resource agency or Indian tribe to timely comply with a provision regarding a requirement of this section does not preclude its participation in subsequent stages of the consultation process.
- (4) Following October 23, 2003, a potential license applicant engaged in pre-filing consultation under part 4 may during first stage consultation request to incorporate into pre-filing

consultation any element of the integrated license application process provided for in part 5 of this chapter. Any such request must be accompanied by a:

- (i) Specific description of how the element of the part 5 license application would fit into the pre-filing consultation process under this part; and
- (ii) Demonstration that the potential license applicant has made every reasonable effort to contact all resource agencies, Indian tribes, non-governmental organizations, and others affected by the applicant's proposal, and that a consensus exists in favor of incorporating the specific element of the part 5 process into the pre-filing consultation under this part.
- (f) Application requirements documenting consultation and any disagreements with resource agencies. An applicant must show in Exhibit E of its application that it has met the requirements of paragraphs (b) through (d) and paragraphs (g) and (h) of this section, and must include a summary of the consultation process and:
- (1) Any resource agency's or Indian tribe's letters containing comments, recommendations, and proposed terms and conditions:
- (2) Any letters from the public containing comments and recommendations:
- (3) Notice of any remaining disagreement with a resource agency or Indian tribe on:
- (i) The need for a study or the manner in which a study should be conducted and the applicant's reasons for disagreement, and
- (ii) Information on any environmental protection, mitigation, or enhancement measure, including the basis for the applicant's disagreement with the resource agency or Indian tribe;
- (4) Evidence of any waivers under paragraph (e) of this section;
- (5) Evidence of all attempts to consult with a resource agency or Indian tribe, copies of related documents showing the attempts, and documents showing the conclusion of the second stage of consultation;
- (6) An explanation of how and why the project would, would not, or should not, comply with any relevant com-

prehensive plan as defined in §2.19 of this chapter and a description of any relevant resource agency or Indian tribe determination regarding the consistency of the project with any such comprehensive plan;

- (7) A description of how the applicant's proposal addresses the significant resource issues raised at the joint meeting held pursuant to paragraph (b)(3) of this section; and
- (8) A list containing the name and address of every federal, state, and interstate resource agency and Indian tribe with which the applicant consulted pursuant to paragraph (a)(1) of this section.
- (g) Public participation. (1) At least 14 days in advance of the joint meeting held pursuant to paragraph (b)(3) of this section, the potential applicant must publish notice, at least once, of the purpose, location, and timing of the joint meeting, in a daily or weekly newspaper published in each county in which the proposed project or any part thereof is situated. The notice shall include a summary of the major issues to be discussed at the joint meeting.
- (2)(i) A potential applicant must make available to the public for inspection and reproduction the information specified in paragraph (b)(2) of this section from the date on which the notice required by paragraph (g)(1) of this section is first published until a final order is issued on any license application.
- (ii) The provisions of §4.32(b) will govern the form and manner in which the information is to be made available for public inspection and reproduction.
- (iii) A potential applicant must make available to the public for inspection at the joint meeting required by paragraph (b)(3) of this section at least two copies of the information specified in paragraph (b)(2) of this section.
- (h) Critical Energy Infrastructure Information. If this section requires an applicant to reveal Critical Energy Infrastructure Information (CEII), as defined by §388.113(c) of this chapter, to

any person, the applicant shall follow the procedures set out in §4.32(k).

[Order 533, 56 FR 23153, May 20, 1991, as amended at 56 FR 61155, Dec. 2, 1991; Order 2002, 68 FR 51117, Aug. 25, 2003; Order 643, 68 FR 52094, Sept. 2, 2003; 68 FR 61742, Oct. 30, 20031

§4.39 Specifications for maps and drawings.

All required maps and drawings must conform to the following specifications, except as otherwise prescribed in this chapter:

- (a) Each original map or drawing must consist of a print on silver or gelatin 35mm microfilm mounted on Type D $(3\frac{1}{4}"$ by $7\frac{3}{8}")$ aperture cards. Full-sized prints of maps and drawings must be on sheets no smaller than 24 by 36 inches and no larger than 28 by 40 inches. A space five inches high by seven inches wide must be provided in the lower right hand corner of each sheet. The upper half of this space must bear the title, numerical and graphical scale, and other pertinent information concerning the map or drawing. The lower half of the space must be left clear. Exhibit G drawings must be stamped by a registered land surveyor. If the drawing size specified in this paragraph limits the scale of structural drawings (exhibit F drawings) described in paragraph (c) of this section, a smaller scale may be used for those drawings. Potential applicants or licensees may be required to file maps or drawings in electronic format as directed by the Commission.
- (b) Each map must have a scale in full-sized prints no smaller than one inch equals 0.5 miles for transmission lines, roads, and similar linear features and no smaller than one inch equals 1,000 feet for other project features, including the project boundary. Where maps at this scale do not show sufficient detail, large scale maps may be required.
 - (1) True and magnetic meridians;
 - (2) State, county, and town lines; and
- (3) Boundaries of public lands and reservations of the United States [see 16 U.S.C. 796 (1) and (2)], if any. If a public land survey is available, the maps must show all lines of that survey crossing the project area and all official subdivisions of sections for the

public lands and reservations, including lots and irregular tracts, as designated on the official plats of survey that may be obtained from the Bureau of Land Management, Washington, DC, or examined in the local land survey office; to the extent that a public land survey is not available for public lands and reservations of the United States, the maps must show the protractions of townships and section lines, which, if possible, must be those recognized by Federal agency administering the those lands.

- (c) Drawings depicting details of project structures must have a scale in full-sized prints no smaller than:
- (1) One inch equals 50 feet for plans, elevations, and profiles; and
- (2) One inch equals 10 feet for sections.
- (d) Each map or drawing must be drawn and lettered to be legible when it is reduced to a print that is 11 inches on its shorter side. Following notification to the applicant that the application has been accepted for filing [see §4.31(c)], prints reduced to that size must be bound in each copy of the application which is required to be submitted to the Commission or provided to any person, agency, or other entity.
- (e) The maps and drawings showing project location information and details of project structures must be filed in accordance with the Commission's instructions on submission of Critical Energy Infrastructure Information in §§ 388.112 and 388.113 of subchapter X of this chapter.

[Order 54, 44 FR 61334, Oct. 25, 1979. Redesignated by Order 413, 50 FR 11678, Mar. 25, 1985; Order 2002, 68 FR 51119, Aug. 25, 2003; 68 FR 61742, Oct. 30, 2003]

Subpart E—Application for License for Major Unconstructed Project and Major Modified Project

§4.40 Applicability.

(a) Applicability. The provisions of this subpart apply to any application for an initial license for a major unconstructed project that would have a total installed capacity of more than 5 megawatts, and any application for an initial or new license for a major

modified project with a total installed capacity more than 5 megawatts. An applicant for license for any major unconstructed or major modified water power project that would have a total installed generating capacity of 5 megawatts or less must submit application under subpart G (§§ 4.60 and 4.61).

(b) Guidance from Commission staff. A prospective applicant for a license for a major unconstructed project or major modified project may seek advice from the Commission's Office of Energy Projects regarding the applicability of this subpart to its project [see § 4.32(h)], including the determinations whether any proposed repair, modification or reconstruction of an existing dam would result in a significant change in the normal maximum surface elevation of an existing impoundment, or whether any proposed change in existing project works or operation would result in a significant environmental impact.

[Order 184, 46 FR 55936, Nov. 13, 1981, as amended by Order 413, 50 FR 11683, Mar. 25, 1985; Order 499, 53 FR 27002, July 18, 1988; Order 2002, 68 FR 51119, Aug. 25, 2003]

§ 4.41 Contents of application.

Any application under this subpart must contain the following information in the form prescribed:

(a) Initial statement.

BEFORE THE FEDERAL ENERGY REGULATORY COMMISSION

Application for License for Major Unconstructed Project or Major Modified Project

- (1) [Name of applicant] applies to the Federal Energy Regulatory Commission for a [license or new license, as appropriate] for the [name of project] water power project, as described in the attached exhibits. [Specify any previous FERC project number designation]
- (2) The location of the proposed project is:

State or territory:
County:
Township or nearby town:
Stream or other body of water:
(3) The exact name, business address, and telephone number of the applicant are:

(4) The applicant is a (citizen of the United States, association of citizens of the United

States, domestic corporation, municipality, or State, as appropriate) and (is/is not) claiming preference under section 7(a) of the Federal Power Act. See 16 U.S.C. 796.

(5)(i) The statutory or regulatory requirements of the state(s) in which the project would be located and that affect the project as proposed with respect to bed and banks and to the appropriation, diversion, and use of water for power purposes, and with respect to the right to engage in the business of developing, transmitting, and distributing power and in any other business necessary to accomplish the purposes of the license under the Federal Power Act, are: [provide citation and brief identification of the nature of each requirement; if the applicant is a municipality, the applicant must submit copies of applicable state or local laws or a municipal charter or, if such laws or documents are not clear, any other appropriate legal authority, evidencing that the municipality is competent under such laws to engage in the business of developing, transmitting, utilizing, or distributing power.]

- (ii) The steps which the applicant has taken, or plans to take, to comply with each of the laws cited above are: [provide brief description for each requirement]
- (b) Exhibit A is a description of the project. If the project includes more than one dam with associated facilities, each dam and the associated component parts must be described together as a discrete development. The description for each development must contain:
- (1) The physical composition, dimensions, and general configuration of any dams, spillways, penstocks, powerhouses, tailraces or other structures proposed to be included as part of the project;
- (2) The normal maximum water surface area and normal maximum water surface elevation (mean sea level), gross storage capacity of any impoundments to be included as part of the project:
- (3) The number, type and rated capacity of any proposed turbines or generators to be included as part of the project;
- (4) The number, length, voltage and interconnections of any primary transmission lines proposed to be included a part of the project [See 16 U.S.C. 796(11)];
- (5) The description of any additional mechanical, electrical, and transmission equipment appurtenant to the project; and

- (6) All lands of the United States, including lands patented subject to the provisions of section 24 of the Act, 16 U.S.C. 818, that are enclosed within the project boundary described under paragraph (h) of this section (Exhibit G), identified and tabulated by legal subdivisions of a public land survey, by the best available legal description. The tabulation must show the total acreage of the lands of the United States within the project boundary.
- (c) Exhibit B is a statement of project operation and resource utilization. If the project includes more than one dam with associated facilities, the information must be provided separately for each discrete development. The exhibit must contain:
- (1) A description of each alternative site considered in selecting of the proposed site:
- (2) A description of any alternative facility designs, processes, and operations that were considered.
- (3) A statement as to whether operation of the power plant will be manual or automatic, an estimate of the annual plant factor, and a statement of how the project will be operated during adverse, mean, and high water years;
- (4) An estimate of the dependable capacity and average annual energy production in kilowatt-hours (or mechanical equivalent), supported by the following data:
- (i) The minimum, mean, and maximum recorded flows in cubic feet per second of the stream or other body of water at the powerplant intake or point of diversion, with a specification of any adjustment made for evaporation, leakage minimum flow releases (including duration of releases) or other reductions in available flow; monthly flow duration curves indicating the period of record and the gauging stations used in deriving the curves; and a specification of the critical streamflow used to determine the dependable capacity:
- (ii) An area-capacity curve showing the gross storage capacity and usable storage capacity of the impoundment, with a rule curve showing the proposed operation of the impoundment and how the usable storage capacity is to be utilized:

- (iii) The estimated minimum and maximum hydraulic capacity of the powerplant in terms of flow and efficiency (cubic feet per second at one-half, full and best gate), and the corresponding generator output in kilowatts;
- (iv) A tailwater rating curve; and
- (v) A curve showing powerplant capability versus head and specifying maximum, normal, and minimum heads;
- (5) A statement of system and regional power needs and the manner in which the power generated at the project is to be utilized, including the amount of power to be used on-site, if any, supported by the following data:
- (i) Load curves and tabular data, if appropriate;
- (ii) Details of conservation and rate design programs and their historic and projected impacts on system loads; and
- (iii) The amount of power to be sold and the identity of proposed purchaser(s); and
- (6) A statement of the applicant's plans for future development of the project or of any other existing or proposed water power project on the affected stream or other body of water, indicating the approximate location and estimated installed capacity of the proposed developments.
- (d) Exhibit C is a proposed construction schedule for the project. The information required may be supplemented with a bar chart. The construction schedule must contain:
- (1) The proposed commencement and completion dates of any new construction, modification, or repair of major project works:
- (2) The proposed commencement date of first commercial operation of each new major facility and generating unit; and
- (3) If any portion of the proposed project consists of previously constructed, unlicensed water power structures or facilities, a chronology of original completion dates of those structures or facilities specifying dates (approximate dates must be identified as such) of:
- (i) Commencement and completion of construction or installation;
- (ii) Commencement of first commercial operation; and

- (iii) Any additions or modifications other than routine maintenance.
- (e) Exhibit D is a statement of project costs and financing. The exhibit must contain:
- (1) A statement of estimated costs of any new construction, modification, or repair, including:
- (i) The cost of any land or water rights necessary to the development;
- (ii) The total cost of all major project works:
- (iii) Indirect construction costs such as costs of construction equipment, camps, and commissaries;
 - (iv) Interest during construction; and
- (v) Overhead, construction, legal expenses, and contingencies;
- (2) If any portion of the proposed project consists of previously constructed, unlicensed water power structures or facilities, a statement of the original cost of those structures or facilities specifying for each, to the extent possible, the actual or approximate total costs (approximate costs must be identified as such) of:
- (i) Any land or water rights necessary to the existing project works;
- (ii) All major project works; and
- (iii) Any additions or modifications other than routine maintenance;
- (3) If the applicant is a licensee applying for a new license, and is not a municipality or a state, an estimate of the amount which would be payable if the project were to be taken over pursuant to section 14 of the Federal Power Act, 16 U.S.C. 807, upon expiration of the license in effect including:
 - (i) Fair value;
 - (ii) Net investment; and
 - (iii) Severance damages;
- (4) A statement of the estimated average annual cost of the total project as proposed, specifying any projected changes in the costs (life-cycle costs) over the estimated financing or licensing period if the applicant takes such changes into account, including:
 - (i) Cost of capital (equity and debt);
 - (ii) Local, state, and Federal taxes;
 - (iii) Depreciation or amortization,
- (iv) Operation and maintenance expenses, including interim replacements, insurance, administrative and general expenses, and contingencies; and

- (v) The estimated capital cost and estimated annual operation and maintenance expense of each proposed environmental measure;
- (5) A statement of the estimated annual value of project power based on a showing of the contract price for sale of power or the estimated average annual cost of obtaining an equivalent amount of power (capacity and energy) from the lowest cost alternative source of power, specifying any projected changes in the costs (life-cycle costs) of power from that source over the estimated financing or licensing period if the applicant takes such changes into account:
- (6) A statement describing other electric energy alternatives, such as gas, oil, coal and nuclear-fueled power-plants and other conventional and pumped storage hydroelectric plants;
- (7) A statement and evaluation of the consequences of denial of the license application and a brief perspective of what future use would be made of the proposed site if the proposed project were not constructed;
- (8) A statement specifying the sources and extent of financing and annual revenues available to the applicant to meet the costs identified in paragraphs (e) (1) and (4) of this section:
- (9) An estimate of the cost to develop the license application; and
- (10) The on-peak and off-peak values of project power, and the basis for estimating the values, for projects which are proposed to operate in a mode other than run-of-river.
- (f) Exhibit E is an Environmental Report. Information provided in the report must be organized and referenced according to the itemized subparagraphs below. See §4.38 for consultation requirements. The Environmental Report must contain the following information, commensurate with the scope of the project:
- (1) General description of the locale. The applicant must provide a general description of the environment of the proposed project area and its immediate vicinity. The description must include location and general information helpful to an understanding of the environmental setting.

- (2) Report on water use and quality. The report must discuss water quality and flows and contain baseline data sufficient to determine the normal and seasonal variability, the impacts expected during construction and operation, and any mitigative, enhancement, and protective measures proposed by the applicant. The report must be prepared in consultation with the state and Federal agencies with responsibility for management of water quality and quantity in the affected stream or other body of water. The report must include:
- (i) A description of existing instream flow uses of streams in the project area that would be affected by construction and operation; estimated quantities of water discharged from the proposed project for power production; and any existing and proposed uses of project waters for irrigation, domestic water supply, industrial and other purposes:
- (ii) A description of the seasonal variation of existing water quality for any stream, lake, or reservoir that would be affected by the proposed project, including (as appropriate) measurements of: significant ions, chlorophyll a, nutrients, specific conductance, pH, total dissolved solids, total alkalinity, total hardness, dissolved oxygen, bacteria, temperature, suspended sediments, turbidity and vertical illumination;
- (iii) A description of any existing lake or reservoir and any of the proposed project reservoirs including surface area, volume, maximum depth, mean depth, flushing rate, shoreline length, substrate classification, and gradient for streams directly affected by the proposed project;
- (iv) A quantification of the anticipated impacts of the proposed construction and operation of project facilities on water quality and downstream flows, such as temperature, turbidity and nutrients;
- (v) A description of measures recommended by Federal and state agencies and the applicant for the purpose of protecting or improving water quality and stream flows during project construction and operation; an explanation of why the applicant has rejected any measures recommended by an agency; and a description of the applicant's alternative measures to pro-

- tect or improve water quality stream flow:
- (vi) A description of groundwater in the vicinity of the proposed project, including water table and artesian conditions, the hydraulic gradient, the degree to which groundwater and surface water are hydraulically connected, aquifers and their use as water supply, and the location of springs, wells, artesian flows and disappearing streams; a description of anticipated impacts on groundwater and measures proposed by the applicant and others for the mitigation of impacts on groundwater; and
- (3) Report on fish, wildlife, and botanical resources. The applicant must provide a report that describes the fish, wildlife, and botanical resources in the vicinity of the proposed project; expected impacts of the project on these resources; and mitigation, enhancement, or protection measures proposed by the applicant. The report must be prepared in consultation with the state agency or agencies with responsibility for these resources, the U.S. Fish and Wildlife Service, the National Marine Fisheries Service (if the proposed project may affect anadromous, estuarine, or marine fish resources), and any state or Federal agency with managerial authority over any part of the proposed project lands. The report must contain:
- (i) A description of existing fish, wildlife, and plant communities of the proposed project area and its vicinity, including any downstream areas that may be affected by the proposed project and the area within the transmission line corridor or right-of-way. A map of vegetation types should be included in the description. For species considered important because of their commercial or recreational value, the information provided should include temporal and spatial distributions and densities of such species. Any fish, wildlife, or plant species proposed or listed as threatened or endangered by the U.S. Fish and Wildlife Service or National Marine Fisheries Service [see 50 CFR 17.11 and 17.12] must be identified;
- (ii) A description of the anticipated impacts on fish, wildlife and botanical resources of the proposed construction

and operation of project facilities, including possible changes in size, distribution, and reproduction of essential population of these resources and any impacts on human utilization of these resources:

- (iii) A description of any measures or facilities recommended by state or Federal agencies for the mitigation of impacts on fish, wildlife, and botanical resources, or for the protection or enhancement of these resources, the impact on threatened or endangered species, and an explanation of why the applicant has determined any measures or facilities recommended by an agency are inappropriate as well as a description of alternative measures proposed by applicant to protect fish, wildlife and botanical resources; and
- (iv) The following materials and information regarding any mitigation measures or facilities, identified under clause (iii), proposed for implementation or construction:
 - (A) Functional design drawings;
- (B) A description of proposed operation and maintenance procedures for any proposed measures or facilities;
- (C) An implementation, construction and operation schedule for any proposed measures or facilities;
- (D) An estimate of the costs of construction, operation, and maintenance of any proposed facilities or implementation of any measures;
- (E) A statement of the sources and amount of financing for mitigation measures or facilities; and
- (F) A map or drawing showing, by the use of shading, crosshatching or other symbols, the identity and location of any proposed measures or facilities.
- (4) Report on historic and archaeological resources. The applicant must provide a report that discusses any historical and archaeological resources in the proposed project area, the impact of the proposed project on those resources and the avoidance, mitigation, and protection measures proposed by the applicant. The report must be prepared in consultation with the State Historic Preservation Officer (SHPO) and the National Park Service of the U.S. Department of Interior. The report must contain:
- (i) A description of any discovery measures, such as surveys, inventories,

and limited subsurface testing work, recommended by the specified state and Federal agencies for the purpose of locating, identifying, and assessing the significance of historic and archaeological resources that would be affected by construction and operation of the proposed project, together with a statement of the applicant's position regarding the acceptability of the recommendations;

- (ii) The results of surveys, inventories, and subsurface testing work recommended by the state and Federal agencies listed above, together with an explanation by the applicant of any variations from the survey, inventory, or testing procedures recommended;
- (iii) An identification (without providing specific site or property locations) of any historic or archaeological site in the proposed project area, with particular emphasis on sites or properties either listed in, or recommended by the SHPO for inclusion in, the National Register of Historic Places that would be affected by the construction of the proposed project;
- (iv) A description of the likely direct and indirect impacts of proposed project construction or operation on sites or properties either listed in, or recommended as eligible for, the National Register of Historic Places;
- (v) A management plan for the avoidance of, or mitigation of, impacts on historic or archaeological sites and resources based upon the recommendations of the state and Federal agencies listed above and containing the applicant's explanation of variations from those recommendations; and
- (vi) The following materials and information regarding the mitigation measures described under paragraph (f)(4)(v) of this section:
- (A) A schedule for implementing the mitigation proposals;
- (B) An estimate of the cost of the measures; and
- (C) A statement of the sources and extent of financing.
- (vii) The applicant must provide five copies (rather than the eight copies required under §4.32(b)(1) of the Commission's regulations) of any survey, inventory, or subsurface testing reports containing specific site and property information, and including maps and

photographs showing the location and any required alteration of historic and archaeological resources in relation to proposed project facilities.

- (5) Report on socio-economic impacts. The applicant must provide a report which identifies and quantifies the impacts of constructing and operating the proposed project on employment, population, housing, personal income, local governmental services, local tax revenues and other factors within the towns and counties in the vicinity of the proposed project. The report must include:
- (i) A description of the socio-economic impact area;
- (ii) A description of employment, population and personal income trends in the impact area;
- (iii) An evaluation of the impact of any substantial in-migration of people on the impact area's governmental facilities and services, such as police, fire, health and educational facilities and programs;
- (iv) On-site manpower requirements and payroll during and after project construction, including a projection of total on-site employment and construction payroll provided by month;
- (v) Numbers of project construction personnel who:
- (A) Currently reside within the impact area:
- (B) Would commute daily to the construction site from places situated outside the impact area; and
- (C) Would relocate on a temporary basis within the impact area;
- (vi) A determination of whether the existing supply of available housing within the impact area is sufficient to meet the needs of the additional population;
- (vii) Numbers and types of residences and business establishments that would be displaced by the proposed project, procedures to be utilized to acquire these properties, and types and amounts of relocation assistance payments that would be paid to the affected property owners and businesses; and
- (viii) A fiscal impact analysis evaluating the incremental local government expenditures in relation to the incremental local government revenues that would result from the construc-

- tion of the proposed project. Incremental expenditures may include, but are not be limited to, school operating costs, road maintenance and repair, public safety, and public utility costs.
- (6) Report on geological and soil resources. The applicant must provide a report on the geological and soil resources in the proposed project area and other lands that would be directly or indirectly affected by the proposed action and the impacts of the proposed project on those resources. The information required may be supplemented with maps showing the location and description of conditions. The report must contain:
- (i) A detailed description of geological features, including bedrock lithology, stratigraphy, structural features, glacial features, unconsolidated deposits, and mineral resources;
- (ii) A detailed description of the soils, including the types, occurrence, physical and chemical characteristics, erodability and potential for mass soil movement;
- (iii) A description showing the location of existing and potential geological and soil hazards and problems, including earthquakes, faults, seepage, subsidence, solution cavities, active and abandoned mines, erosion, and mass soil movement, and an identification of any large landslides or potentially unstable soil masses which could be aggravated by reservoir fluctuation;
- (iv) A description of the anticipated erosion, mass soil movement and other impacts on the geological and soil resources due to construction and operation of the proposed project; and
- (v) A description of any proposed measures or facilities for the mitigation of impacts on soils.
- (7) Report on recreational resources. The applicant must prepare a report containing a proposed recreation plan describing utilization, design and development of project recreational facilities, and public access to the project area. Development of the plan should include consideration of the physically handicapped. Public and private recreational facilities provided by others that would abut the project should be noted in the report. The report must be prepared in

consultation with appropriate local, regional, state and Federal recreation agencies and planning commissions, the National Park Service of the U.S. Department of the Interior, and any other state or Federal agency with managerial responsibility for any part of the project lands. The report must contain:

- (i) A description of any areas within or in the vicinity of the proposed project boundary that are included in, or have been designated for study for inclusion in:
- (A) The National Wild and Scenic Rivers Systems (see 16 U.S.C. 1271);
- (B) The National Trails System (see 16 U.S.C. 1241); or
- (C) A wilderness area designated under the Wilderness Act (see 16 U.S.C. 1132);
- (ii) A detailed description of existing recreational facilities within project vicinity, and the public recreational facilities which are to be provided by the applicant at its sole cost or in cooperation with others no later than 3 years from the date of first commercial operation of the proposed project and those recreation facilities planned for future development based on anticipated demand. When public recreation facilities are to be provided by other entities, the applicant and those entities should enter into an agreement on the type of facilities to be provided and the method of operation. Copies of agreements with cooperating entities are to be appended to the plan:
- (iii) A provision for a shoreline buffer zone that must be within the project boundary, above the normal maximum surface elevation of the project reservoir, and of sufficient width to allow public access to project lands and waters and to protect the scenic, public recreational, cultural, and other environmental values of the reservoir shoreline;
- (iv) Estimates of existing and future recreational use at the project, in daytime and overnight visitation (recreation days), with a description of the methodology used in developing these data;
- (v) A development schedule and cost estimates of the construction, operation, and maintenance of existing, ini-

tial, and future public recreational facilities, including a statement of the source and extent of financing for such facilities:

- (vi) A description of any measures or facilities recommended by the agencies consulted for the purpose of creating, preserving, or enhancing recreational opportunities at the proposed project, and for the purpose of ensuring the safety of the public in its use of project lands and waters, including an explanation of why the applicant has rejected any measures or facilities recommended by an agency; and
- (vii) A drawing or drawings, one of which describes the entire project area, clearly showing:
- (A) The location of project lands, and the types and number of existing recreational facilities and those proposed for initial development, including access roads and trails, and facilities for camping, picnicking, swimming, boat docking and launching, fishing and hunting, as well as provisions for sanitation and waste disposal;
- (B) The location of project lands, and the type and number of recreational facilities planned for future development:
- (C) The location of all project lands reserved for recreational uses other than those included in paragraphs (f)(7)(vii) (A) and (B) of this section; and
- (D) The project boundary (excluding surveying details) of all areas designated for recreational development. sufficiently referenced to the appropriate Exhibit G drawings to show that all lands reserved for existing and future public recreational development and the shoreline buffer zone are included within the project boundary. Recreational cottages, mobile homes and year-round residences for private use are not to be considered as public recreational facilities, and the lands on which these private facilities are to be developed are not to be included within the proposed project boundary.
- (8) Report on aesthetic resources. The applicant must provide a report that describes the aesthetic resources of the proposed project area, the expected impacts of the project on these resources, and the mitigation, enhancement or

protection measures proposed. The report must be prepared following consultation with Federal, state, and local agencies having managerial responsibility for any part of the proposed project lands or lands abutting those lands. The report must contain:

- (i) A description of the aesthetic character of lands and waters directly and indirectly affected by the proposed project facilities;
- (ii) A description of the anticipated impacts on aesthetic resources from construction activity and related equipment and material, and the subsequent presence of proposed project facilities in the landscape;
- (iii) A description of mitigative measures proposed by the applicant, including architectural design, land-scaping, and other reasonable treatment to be given project works to preserve and enhance aesthetic and related resources during construction and operation of proposed project facilities; and
- (iv) Maps, drawings and photographs sufficient to provide an understanding of the information required under this paragraph. Maps or drawings may be consolidated with other maps or drawings required in this exhibit and must conform to the specifications of §4.39.
- (9) Report on land use. The applicant must provide a report that describes the existing uses of the proposed project lands and adjacent property, and those land uses which would occur if the project is constructed. The report may reference the discussions of land uses in other sections of this exhibit. The report must be prepared following consultation with local and state zoning or land management authorities, and any Federal or state agency with managerial responsibility for the proposed project or abutting lands. The report must include:
- (i) A description of existing land use in the proposed project area, including identification of wetlands, floodlands, prime or unique farmland as designated by the Natural Resources Conservation Service of the U.S. Department of Agriculture, the Special Area Management Plan of the Office of Coastal Zone Management, National Oceanic and Atmospheric Administration, and lands

owned or subject to control by government agencies;

- (ii) A description of the proposed land uses within and abutting the project boundary that would occur as a result of development and operation of the project; and
- (iii) Aerial photographs, maps, drawings or other graphics sufficient to show the location, extent and nature of the land uses referred to in this section.
- (10) Alternative locations, designs, and energy sources. The applicant must provide an environment assessment of the following:
- (i) Alternative sites considered in arriving at the selection of the proposed project site:
- (ii) Alternative facility designs, processes, and operations that were considered and the reasons for their rejection;
- (iii) Alternative electrical energy sources, such as gas, oil, coal, and nuclear-fueled power plants, purchased power or diversity exchange, and other conventional and pumped-storage hydroelectric plants; and
- (iv) The overall consequences if the license application is denied.
- (11) List of literature. Exhibit E must include a list of all publications, reports, and other literature which were cited or otherwise utilized in the preparation of any part of the environmental report.
- (g) Exhibit F consists of general design drawings of the principal project works described under paragraph (b) of this section (Exhibit A) and supporting information used as the basis of design. If the Exhibit F submitted with the application is preliminary in nature, applicant must so state in the application. The drawings must conform to the specifications of §4.39.
- (1) The drawings must show all major project structures in sufficient detail to provide a full understanding of the project, including:
 - (i) Plans (overhead view);
 - (ii) Elevations (front view);
 - (iii) Profiles (side view); and
 - (iv) Sections.
- (2) The applicant may submit preliminary design drawings with the application. The final Exhibit F may be submitted during or after the licensing

process and must show the precise plans and specifications for proposed structures. If the project is licensed on the basis of preliminary designs, the applicant must submit a final Exhibit F for Commission approval prior to commencement of any construction of the project.

- (3) Supporting design report. The applicant must furnish, at a minimum, the following supporting information to demonstrate that existing and proposed structures are safe and adequate to fulfill their stated functions and must submit such information in a separate report at the time the application is filed. The report must include:
- (i) An assessment of the suitability of the site and the reservoir rim stability based on geological and subsurface investigations, including investigations of soils and rock borings and tests for the evaluation of all foundations and construction materials sufficient to determine the location and type of dam structure suitable for the site;
- (ii) Copies of boring logs, geology reports and laboratory test reports;
- (iii) An identification of all borrow areas and quarry sites and an estimate of required quantities of suitable construction material;
- (iv) Stability and stress analyses for all major structures and critical abutment slopes under all probable loading conditions, including seismic and hydrostatic forces induced by water loads up to the Probable Maximum Flood as appropriate; and
- (v) The bases for determination of seismic loading and the Spillway Design Flood in sufficient detail to permit independent staff evaluation.
- (4) The applicant must submit two copies of the supporting design report described in paragraph (g)(3) of this section at the time preliminary and final design drawings are submitted to the Commission for review. If the report contains preliminary drawings, it must be designated a "Preliminary Supporting Design Report."
- (h) Exhibit G is a map of the project that must conform to the specifications of §4.39. In addition, to the other components of Exhibit G, the Applicant must provide the project boundary data in a geo-referenced electronic format—such as ArcView shape files,

GeoMedia files, MapInfo files, or any similar format. The electronic boundary data must be positionally accurate to ±40 feet, in order to comply with the National Map Accuracy Standards for maps at a 1:24,000 scale (the scale of USGS quadrangle maps). The electronic exhibit G data must include a text file describing the map projection used (i.e., UTM, State Plane, Decimal Degrees, etc.), the map datum (i.e., feet, meters, miles, etc.). Three sets of the maps must be submitted on compact disk or other appropriate electronic media. If more than one sheet is used for the paper maps, the sheets must be numbered consecutively, and each sheet must bear a small insert sketch showing the entire project and indicate that portion of the project depicted on that sheet. Each sheet must contain a minimum of three known reference points. The latitude and longitude coordinates, or state plane coordinates, of each reference point must be shown. If at any time after the application is filed there is any change in the project boundary, the applicant must submit, within 90 days following the completion of project construction, a final exhibit G showing the extent of such changes. The map must show:

- (1) Location of the project and principal features. The map must show the location of the project as a whole with reference to the affected stream or other body of water and, if possible, to a nearby town or any other permanent monuments or objects, such as roads, transmission lines or other structures, that can be noted on the map and recognized in the field. The map must also show the relative locations and physical interrelationships of the principal project works and other features described under paragraph (b) of this section (Exhibit A).
- (2) Project boundary. The map must show a project boundary enclosing all project works and other features described under paragraph (b) of this section (Exhibit A) that are to be licensed. If accurate survey information is not available at the time the application is filed, the applicant must so state, and a tentative boundary may be submitted. The boundary must enclose only those lands necessary for operation and maintenance of the project

and for other project purposes, such as recreation, shoreline control, or protection of environmental resources (see paragraph (f) of this section (Exhibit E)). Existing residential, commercial, or other structures may be included within the boundary only to the extent that underlying lands are needed for project purposes (e.g., for flowage, public recreation, shoreline control, or environmental protection of sources). If the boundary is on land covered by a public survey, ties must be shown on the map at sufficient points to permit accurate platting of the position of the boundary relative to the lines of the public land survey. If the lands are not covered by a public land survey, the best available legal description of the position of the boundary must be provided, including distances and directions from fixed monuments or physical features. The boundary must be described as follows:

- (i) *Impoundments*. (A) The boundary around a project impoundment must be described by one of the following:
- Contour lines, including the contour elevation (preferred method);
- (2) Specified courses and distances (metes and bounds);
- (3) If the project lands are covered by a public land survey, lines upon or parallel to the lines of the survey; or
- (4) Any combination of the above methods.
- (B) The boundary must be located no more than 200 feet (horizontal measurement) from the exterior margin of the reservoir, defined by the normal maximum surface elevation, except where deviations may be necessary in describing the boundary according to the above methods or where additional lands are necessary for project purposes, such as public recreation, shoreline control, or protection of environmental resources.
- (ii) Continuous features. The boundary around linear (continuous) project features such as access roads, transmission lines, and conduits may be described by specified distances from center lines or offset lines of survey. The width of such corridors must not exceed 200 feet unless good cause is shown for a greater width. Several sections of a continuous feature may be shown on a single sheet with information show-

ing the sequence of contiguous sections.

- (iii) *Noncontinuous features*. (A) The boundary around noncontinuous project works such as dams, spillways, and powerhouses must be described by one of the following:
 - (1) Contour lines;
 - (2) Specified courses and distances;
- (3) If the project lands are covered by a public land survey, lines upon or parallel to the lines of the survey; or
- (4) Any combination of the above methods.
- (B) The boundary must enclose only those lands that are necessary for safe and efficient operation and maintenance of the project or for other specified project purposes, such as public recreation or protection of environmental resources.
- (3) Federal lands. Any public lands and reservations of the United States (Federal lands) [see 16 U.S.C. 796 (1) and (2)] that are within the project boundary, such as lands administered by the U.S. Forest Service, Bureau of Land Management, or National Park Service, or Indian tribal lands, and the boundaries of those Federal lands, must be identified as such on the map by:
- (i) Legal subdivisions of a public land survey of the affected area (a protraction of identified township and section lines is sufficient for this purpose); and
- (ii) The Federal agency, identified by symbol or legend, that maintains or manages each identified subdivision of the public land survey within the project boundary; or
- (iii) In the absence of a public land survey, the location of the Federal lands according to the distances and directions from fixed monuments or physical features. When a Federal survey monument or a Federal bench mark will be destroyed or rendered unusable by the construction of project works, at least two permanent, marked witness monuments or bench marks must be established at accessible points. The maps show the location (and elevation, for bench marks) of the survey monument or bench mark which will be destroyed or rendered unusable, as well as of the witness monuments or bench marks. Connecting courses and distances from the witness

§ 4.50

monuments or bench marks to the original must also be shown.

- (iv) The project location must include the most current information pertaining to affected Federal lands as described under §4.81(b)(5).
- (4) Non-Federal lands. For those lands within the project boundary not identified under paragraph (h)(3) of this section, the map must identify by legal subdivision:
- (i) Lands owned in fee by the applicant and lands that the applicant plans to acquire in fee; and
- (ii) Lands over which the applicant has acquired or plans to acquire rights to occupancy and use other than fee title, including rights acquired or to be acquired by easement or lease.

[Order 184, 46 FR 55936, Nov. 13, 1981; 48 FR 4459, Feb. 1, 1983, as amended by Order 413, 50 FR 11684, Mar. 25, 1985; Order 464, 52 FR 5449, Feb. 23, 1987; Order 540, 57 FR 21737, May 22, 1992; Order 2002, 68 FR 51119, Aug. 25, 2003; 68 FR 61742, Oct. 30, 2003; 68 FR 63194, Nov. 7, 2003; 68 FR 69957, Dec. 16, 2003; Order 699, 72 FR 45324, Aug. 14, 2007]

Subpart F—Application for License for Major Project—Existing Dam

AUTHORITY: Federal Power Act, as amended (16 U.S.C. 792–828c); Public Utility Regulatory Policies Act of 1978 (16 U.S.C. 2601–2645); Department of Energy Organization Act (42 U.S.C. 7101–7352); E.O. 12009, 42 FR 46267; Pub. L. 96–511, 94 Stat. 2812 (44 U.S.C. 3501 $et\ seq.$).

§ 4.50 Applicability.

- (a) Applicability. (1) Except as provided in paragraph (a)(2) of this section, the provisions of this subpart apply to any application for either an initial license or new license for a major project—existing dam that is proposed to have a total installed capacity of more than 5 megawatts.
- (2) This subpart does not apply to any major project—existing dam (see § 4.40) that is proposed to entail or include:
- (i) Any repair, modification or reconstruction of an existing dam that would result in a significant change in the normal maximum surface area or normal maximum surface elevation of an existing impoundment; or
- (ii) Any new development or change in project operation that would result in a significant environmental impact.

- (3) An applicant for license for any major project—existing dam that would have a total installed capacity of 5 megawatts or less must submit application under subpart G (§§ 4.60 and 4.61).
- (b) Guidance from Commission staff. A prospective applicant for a major license-existing dam may seek advice from the Commission staff regarding the applicability of these sections to its project (see §4.32(h)), including the determinations whether any proposed repair or reconstruction of an existing dam would result in a significant change in the normal maximum surface area or the normal maximum surface elevation of an existing impoundment, or whether any proposed new development or change in project operation would result in a significant environmental impact.

[Order 59, 44 FR 67651, Nov. 27, 1979, as amended by Order 184, 46 FR 55942, Nov. 13, 1981; Order 413, 50 FR 11684, Mar. 25, 1985; Order 499, 53 FR 27002, July 18, 1988]

§ 4.51 Contents of application.

An application for license under this subpart must contain the following information in the form specified. As provided in paragraph (f) of this section, the appropriate Federal, state, and local resource agencies must be given the opportunity to comment on the proposed project, prior to filing of the application for license for major project—existing dam. Information from the consultation process must be included in this Exhibit E, as appropriate.

(a) Initial statement.

BEFORE THE FEDERAL ENERGY REGULATORY COMMISSION

Application for License for Major Project— Existing Dam

- (1) (Name of applicant) applies to the Federal Energy Regulatory Commission for a (license or new license, as appropriate) for the (name of project) water power project, as described in the attached exhibits. (Specify any previous FERC project number designation.)
- (2) The location of the project is:

State or territory:	
County:	
Township or nearby town:	
Stream or other body of water:	

Federal Energy Regulatory Commission

(3) The exact name and business address of the applicant are:

The exact name and business address of each person authorized to act as agent for the applicant in this application are:

- (4) The applicant is a [citizen of the United States, association of citizens of the United States, domestic corporation, municipality, or state, as appropriate] and (is/is not) claiming preference under section 7(a) of the Federal Power Act. See 16 U.S.C. 796.
- (5)(i) The statutory or regulatory requirements of the state(s) in which the project would be located that affect the project as proposed, with respect to bed and banks and to the appropriation, diversion, and use of water for power purposes, and with respect to the right to engage in the business of developing. transmitting, and distributing power and in any other business necessary to accomplish the purposes of the license under the Federal Power Act, are: [Provide citation] and brief identification of the nature of each requirement; if the applicant is a municipality, the applicant must submit copies of applicable state and local laws or a municipal charter, or, if such laws or documents are not clear, other appropriate legal authority, evidencing that the municipality is competent under such laws to engage in the business of developing, transmitting, utilizing, or distributing power.]
- (ii) The steps which the applicant has taken or plans to take to comply with each of the laws cited above are: (provide brief description for each law).
- (6) The applicant must provide the name and address of the owner of any existing project facilities. If the dam is federally owned or operated, provide the name of the agency.
- (b) Exhibit A is a description of the project. This exhibit need not include information on project works maintained and operated by the U.S. Army Corps of Engineers, the Bureau of Reclamation, or any other department or agency of the United States, except for any project works that are proposed to be altered or modified. If the project includes more than one dam with associated facilities, each dam and the associated component parts must be described together as a discrete development. The description for each development must contain:

- (1) The physical composition, dimensions, and general configuration of any dams, spillways, penstocks, powerhouses, tailraces, or other structures, whether existing or proposed, to be included as part of the project;
- (2) The normal maximum surface area and normal maximum surface elevation (mean sea level), gross storage capacity, and usable storage capacity of any impoundments to be included as part of the project;
- (3) The number, type, and rated capacity of any turbines or generators, whether existing or proposed, to be included as part of the project;
- (4) The number, length, voltage, and interconnections of any primary transmission lines, whether existing or proposed, to be included as part of the project (see 16 U.S.C. 796(11));
- (5) The specifications of any additional mechanical, electrical, and transmission equipment appurtenant to the project; and
- (6) All lands of the United States that are enclosed within the project boundary described under paragraph (h) of this section (Exhibit G), identified and tabulated by legal subdivisions of a public land survey of the affected area or, in the absence of a public land survey, by the best available legal description. The tabulation must show the total acreage of the lands of the United States within the project boundary.
- (c) Exhibit B is a statement of project operation and resource utilization. If the project includes more than one dam with associated facilities, the information must be provided separately for each such discrete development. The exhibit must contain:
- (1) A statement whether operation of the powerplant will be manual or automatic, an estimate of the annual plant factor, and a statement of how the project will be operated during adverse, mean, and high water years;
- (2) An estimate of the dependable capacity and average annual energy production in kilowatt-hours (or a mechanical equivalent), supported by the following data:
- (i) The minimum, mean, and maximum recorded flows in cubic feet per second of the stream or other body of water at the powerplant intake or

point of diversion, with a specification of any adjustments made for evaporation, leakage, minimum flow releases (including duration of releases), or other reductions in available flow; monthly flow duration curves indicating the period of record and the gauging stations used in deriving the curves; and a specification of the period of critical streamflow used to determine the dependable capacity:

- (ii) An area-capacity curve showing the gross storage capacity and usable storage capacity of the impoundment, with a rule curve showing the proposed operation of the impoundment and how the usable storage capacity is to be utilized:
- (iii) The estimated hydraulic capacity of the powerplant (minimum and maximum flow through the powerplant) in cubic feet per second;
 - (iv) A tailwater rating curve; and
- (v) A curve showing powerplant capability versus head and specifying maximum, normal, and minimum heads;
- (3) A statement, with load curves and tabular data, if necessary, of the manner in which the power generated at the project is to be utilized, including the amount of power to be used on-site, if any, the amount of power to be sold, and the identity of any proposed purchasers; and
- (4) A statement of the applicant's plans, if any, for future development of the project or of any other existing or proposed water power project on the stream or other body of water, indicating the approximate location and estimated installed capacity of the proposed developments.
- (d) $Exhibit\ C$ is a construction history and proposed construction schedule for the project. The construction history and schedules must contain:
- (1) If the application is for an initial license, a tabulated chronology of construction for the existing projects structures and facilities described under paragraph (b) of this section (Exhibit A), specifying for each structure or facility, to the extent possible, the actual or approximate dates (approximate dates must be identified as such) of:
- (i) Commencement and completion of construction or installation;

- (ii) Commencement of commercial operation; and
- (iii) Any additions or modifications other than routine maintenance; and
- (2) If any new development is proposed, a proposed schedule describing the necessary work and specifying the intervals following issuance of a license when the work would be commenced and completed.
- (e) $Exhibit\ D$ is a statement of costs and financing. The statement must contain:
- (1) If the application is for an initial license, a tabulated statement providing the actual or approximate original cost (approximate costs must be identified as such) of:
- (i) Any land or water right necessary to the existing project; and
- (ii) Each existing structure and facility described under paragraph (b) of this section (Exhibit A).
- (2) If the applicant is a licensee applying for a new license, and is not a municipality or a state, an estimate of the amount which would be payable if the project were to be taken over pursuant to section 14 of the Federal Power Act upon expiration of the license in effect [see 16 U.S.C. 807], including:
 - (i) Fair value;
 - (ii) Net investment; and
 - (iii) Severance damages.
- (3) If the application includes proposals for any new development, a statement of estimated costs, including:
- (i) The cost of any land or water rights necessary to the new development; and
- (ii) The cost of the new development work, with a specification of:
 - (A) Total cost of each major item;
- (B) Indirect construction costs such as costs of construction equipment, camps, and commissaries;
- (C) Interest during construction; and
- (D) Overhead, construction, legal expenses, taxes, administrative and general expenses, and contingencies.
- (4) A statement of the estimated average annual cost of the total project as proposed specifying any projected changes in the costs (life-cycle costs) over the estimated financing or licensing period if the applicant takes such changes into account, including:

- (i) Cost of capital (equity and debt);
- (ii) Local, state, and Federal taxes;
- (iii) Depreciation and amortization;
- (iv) Operation and maintenance expenses, including interim replacements, insurance, administrative and general expenses, and contingencies; and
- (v) The estimated capital cost and estimated annual operation and maintenance expense of each proposed environmental measure.
- (5) A statement of the estimated annual value of project power, based on a showing of the contract price for sale of power or the estimated average annual cost of obtaining an equivalent amount of power (capacity and energy) from the lowest cost alternative source, specifying any projected changes in the cost of power from that source over the estimated financing or licensing period if the applicant takes such changes into account.
- (6) A statement specifying the sources and extent of financing and annual revenues available to the applicant to meet the costs identified in paragraphs (e) (3) and (4) of this section.
- (7) An estimate of the cost to develop the license application;
- (8) The on-peak and off-peak values of project power, and the basis for estimating the values, for projects which are proposed to operate in a mode other than run-of-river; and
- (9) The estimated average annual increase or decrease in project generation, and the estimated average annual increase or decrease of the value of project power, due to a change in project operations (*i.e.*, minimum bypass flows; limits on reservoir fluctuations).
- (f) Exhibit E is an Environmental Report. Information provided in the report must be organized and referenced according to the itemized subparagraphs below. See §4.38 for consultation requirements. The Environmental Report must contain the following information, commensurate with the scope of the proposed project:
- (1) General description of the locale. The applicant must provide a general description of the environment of the project and its immediate vicinity. The description must include general infor-

- mation concerning climate, topography, wetlands, vegetative cover, land development, population size and density, the presence of any floodplain and the occurrence of flood events in the vicinity of the project, and any other factors important to an understanding of the setting.
- (2) Report on water use and quality. The report must discuss the consumptive use of project waters and the impact of the project on water quality. The report must be prepared in consultation with the state and Federal agencies with responsibility for management of water quality in the affected stream or other body of water. Consultation must be documented by appending to the report a letter from each agency consulted that indicates the nature, extent, and results of the consultation. The report must include:
- (i) A description (including specified volume over time) of existing and proposed uses of project waters for irrigation, domestic water supply, steamelectric plant, industrial, and other consumptive purposes;
- (ii) A description of existing water quality in the project impoundment and downstream water affected by the project and the applicable water quality standards and stream segment classifications;
- (iii) A description of any minimum flow releases specifying the rate of flow in cubic feet per second (cfs) and duration, changes in the design of project works or in project operation, or other measures recommended by the agencies consulted for the purposes of protecting or improving water quality, including measures to minimize the short-term impacts on water quality of any proposed new development of project works (for any dredging or filling, refer to 40 CFR part 230 and 33 CFR 320.3(f) and 323.3(e))1:
- (iv) A statement of the existing measures to be continued and new measures proposed by the applicant for the purpose of protecting or improving water quality, including an explanation of why the applicant has rejected any measures recommended by

¹33 CFR part 323 was revised at 47 FR 31810, July 22, 1982, and §323.3(e) no longer exists.

an agency and described under paragraph (f)(2)(iii) of this section.

- (v) A description of the continuing impact on water quality of continued operation of the project and the incremental impact of proposed new development of project works or changes in project operation; and
- (3) Report on fish, wildlife, and botanical resources. The report must discuss fish, wildlife, and botanical resources in the vicinity of the project and the impact of the project on those resources. The report must be prepared in consultation with any state agency with responsibility for fish, wildlife, and botanical resources, the U.S. Fish and Wildlife Service, the National Marine Fisheries Service (if the project may affect anadromous fish resources subject to that agency's jurisdiction), and any other state or Federal agency with managerial authority over any part of the project lands. Consultation must be documented by appending to the report a letter from each agency consulted that indicates the nature, extent, and results of the consultation. The report must include:
- (i) A description of the fish, wildlife, and botanical resources of the project and its vicinity, and of downstream areas affected by the project, including identification of any species listed as threatened or endangered by the U.S. Fish and Wildlife Service (See 50 CFR 17.11 and 17.12);
- (ii) A description of any measures or facilities recommended by the agencies consulted for the mitigation of impacts on fish, wildlife, and botanical resources, or for the protection or improvement of those resources;
- (iii) A statement of any existing measures or facilities to be continued or maintained and any measures or facilities proposed by the applicant for the mitigation of impacts on fish, wildlife, and botanical resources, or for the protection or improvement of such resources, including an explanation of why the applicant has rejected any measures or facilities recommended by an agency and described under paragraph (f)(3)(ii) of this section.
- (iv) A description of any anticipated continuing impact on fish, wildlife, and botanical resources of continued operation of the project, and the incre-

- mental impact of proposed new development of project works or changes in project operation; and
- (v) The following materials and information regarding the measures and facilities identified under paragraph (f)(3)(iii) of this section:
- (A) Functional design drawings of any fish passage and collection facilities, indicating whether the facilities depicted are existing or proposed (these drawings must conform to the specifications of §4.39 regarding dimensions of full-sized prints, scale, and legibility);
- (B) A description of operation and maintenance procedures for any existing or proposed measures or facilities;
- (C) An implementation or construction schedule for any proposed measures or facilities, showing the intervals following issuance of a license when implementation of the measures or construction of the facilities would be commenced and completed;
- (D) An estimate of the costs of construction, operation, and maintenance, of any proposed facilities, and of implementation of any proposed measures, including a statement of the sources and extent of financing; and
- (E) A map or drawing that conforms to the size, scale, and legibility requirements of §4.39 showing by the use of shading, cross-hatching, or other symbols the identity and location of any measures or facilities, and indicating whether each measure or facility is existing or proposed (the map or drawings in this exhibit may be consolidated).
- (4) Report on historical and archeological resources. The report must discuss the historical and archeological resources in the project area and the impact of the project on those resources. The report must be prepared in consultation with the State Historic Preservation Officer and the National Park Service. Consultation must be documented by appending to the report a letter from each agency consulted that indicates the nature, extent, and results of the consultation. The report must contain:
- (i) Identification of any sites either listed or determined to be eligible for inclusion in the National Register of Historic Places that are located in the

project area, or that would be affected by operation of the project or by new development of project facilities (including facilities proposed in this exhibit):

- (ii) A description of any measures recommended by the agencies consulted for the purpose of locating, identifying, and salvaging historical or archaeological resources that would be affected by operation of the project, or by new development of project facilities (including facilities proposed in this exhibit), together with a statement of what measures the applicant proposes to implement and an explanation of why the applicant rejects any measures recommended by an agency.
- (iii) The following materials and information regarding the survey and salvage activities described under paragraph (f)(4)(ii) of this section:
- (A) A schedule for the activities, showing the intervals following issuance of a license when the activities would be commenced and completed; and
- (B) An estimate of the costs of the activities, including a statement of the sources and extent of financing.
- (5) Report on recreational resources. The report must discuss existing and proposed recreational facilities and opportunities at the project. The report must be prepared in consultation with local, state, and regional recreation agencies and planning commissions, the National Park Service, and any other state or Federal agency with managerial authority over any part of the project lands. Consultation must be documented by appending to the report a letter from each agency consulted indicating the nature, extent, and results of the consultation. The report must contain:
- (i) A description of any existing recreational facilities at the project, indicating whether the facilities are available for public use;
- (ii) An estimate of existing and potential recreational use of the project area, in daytime and overnight visits;
- (iii) A description of any measures or facilities recommended by the agencies consulted for the purpose of creating, preserving, or enhancing recreational opportunities at the project and in its vicinity (including opportunities for

the handicapped), and for the purpose of ensuring the safety of the public in its use of project lands and waters;

- (iv) A statement of the existing measures or facilities to be continued or maintained and the new measures or facilities proposed by the applicant for the purpose of creating, preserving, or enhancing recreational opportunities at the project and in its vicinity, and for the purpose of ensuring the safety of the public in its use of project lands and waters, including an explanation of why the applicant has rejected any measures or facilities recommended by an agency and described under paragraph (f)(5)(iii) of this section; and
- (v) The following materials and information regarding the measures and facilities identified under paragraphs (f)(5) (i) and (iv) of this section:
- (A) Identification of the entities responsible for implementing, constructing, operating, or maintaining any existing or proposed measures or facilities;
- (B) A schedule showing the intervals following issuance of a license at which implementation of the measures or construction of the facilities would be commenced and completed;
- (C) An estimate of the costs of construction, operation, and maintenance of any proposed facilities, including a statement of the sources and extent of financing:
- (D) A map or drawing that conforms to the size, scale, and legibility requirements of §4.39 showing by the use of shading, cross-hatching, or other symbols the identity and location of any facilities, and indicating whether each facility is existing or proposed (the maps or drawings in this exhibit may be consolidated); and
- (vi) A description of any areas within or in the vicinity of the proposed project boundary that are included in, or have been designated for study for inclusion in, the National Wild and Scenic Rivers System, or that have been designated as wilderness area, recommended for such designation, or designated as a wilderness study area under the Wilderness Act.
- (6) Report on land management and aesthetics. The report must discuss the

management of land within the proposed project boundary, including wetlands and floodplains, and the protection of the recreational and scenic values of the project. The report must be prepared following consultation with local and state zoning and land management authorities and any Federal or state agency with managerial authority over any part of the project lands. Consultation must be documented by appending to the report a letter from each agency consulted indicating the nature, extent, and results of the consultation. The report must contain:

- (i) A description of existing development and use of project lands and all other lands abutting the project impoundment:
- (ii) A description of the measures proposed by the applicant to ensure that any proposed project works, rights-of-way, access roads, and other topographic alterations blend, to the extent possible, with the surrounding environment; (see, e.g., 44 F.P.C. 1496, et seg.):
- (iii) A description of wetlands or floodplains within, or adjacent to, the project boundary, any short-term or long-term impacts of the project on those wetlands or floodplains, and any mitigative measures in the construction or operation of the project that minimize any adverse impacts on the wetlands or floodplains;
- (iv) A statement, including an analysis of costs and other constraints, of the applicant's ability to provide a buffer zone around all or any part of the impoundment, for the purpose of ensuring public access to project lands and waters and protecting the recreational and aesthetic values of the impoundment and its shoreline;
- (v) A description of the applicant's policy, if any, with regard to permitting development of piers, docks, boat landings, bulkheads, and other shoreline facilities on project lands and waters; and
- (vi) Maps or drawings that conform to the size, scale and legibility requirements of §4.39, or photographs, sufficient to show the location and nature of the measures proposed under paragraph (f)(6)(ii) of this section (maps or

drawings in this exhibit may be consolidated).

- (7) List of literature. The report must include a list of all publications, reports, and other literature which were cited or otherwise utilized in the preparation of any part of the environmental report.
- (g) Exhibit F. $See \ \$4.41(g)$ of this chapter.
- (h) Exhibit G. See §4.41(h) of this chapter.

[Order 141, 12 FR 8485, Dec. 19, 1947, as amended by Order 123, 46 FR 9029, Jan. 28, 1981; Order 183, 46 FR 55251, Nov. 9, 1981; Order 184, 46 FR 55942, Nov. 13, 1981; Order 413, 50 FR 11684, Mar. 25, 1985; Order 464, 52 FR 5449, Feb. 23, 1987; Order 540, 57 FR 21737, May 22, 1992; Order 2002, 68 FR 51120, Aug. 25, 2003; 68 FR 61742, Oct. 30, 2003]

Subpart G—Application for License for Minor Water Power Projects and Major Water Power Projects 5 Megawatts or Less

§ 4.60 Applicability and notice to agencies.

- (a) Applicability. The provisions of this subpart apply to any application for an initial license or a new license for:
- (1) A minor water power project, as defined in §4.30(b)(17);
- (2) Any major project—existing dam, as defined in §4.30(b)(16), that has a total installed capacity of 5 MW or less: or
- (3) Any major unconstructed project or major modified project, as defined in §4.30 (b) (15) and (14) respectively, that has a total installed capacity of 5 MW or less.
- (b) Notice to agencies. The Commission will supply interested Federal, state, and local agencies with notice of any application for license for a water power project 5 MW or less and request comment on the application. Copies of the application will be available for inspection at the Commission's Public Reference Room. The applicant shall also furnish copies of the filed application to any Federal, state, or local agency that so requests.

Federal Energy Regulatory Commission

- (c) Unless an applicant for a license for a minor water power project requests in its application that the Commission apply the following provisions of Part I of the Federal Power Act when it issues a minor license for a project, the Commission, unless it determines it would not be in the public interest to do so, will waive:
- (1) Section 4(b), insofar as it requires a licensee to file a statement showing the actual legitimate costs of construction of a project;
- (2) Section 4(e), insofar as it relates to approval by the Chief of Engineers and the Secretary of the Army of plans affecting navigation;
- (3) Section 6, insofar as it relates to the acceptance and expression in the license of terms and conditions of the Federal Power Act that are waived in the licensing order;
- (4) Section 10(c), insofar as it relates to a licensee's maintenance of depreciation reserves;
 - (5) Sections 10(d) and 10(f);
- (6) Section 14, with the exception of the right of the United States or any state or municipality to take over, maintain, and operate a project through condemnation proceedings; and
 - (7) Sections 15, 16, 19, 20 and 22.

[Order 413, 50 FR 11685, Mar. 25, 1985, as amended by Order 513, 54 FR 23806, June 2, 1989; Order 2002, 68 FR 51120, Aug. 25, 2003]

§ 4.61 Contents of application.

- (a) General instructions—(1) Entry upon land. No work may be started on any proposed project works until the applicant receives a signed license from the Commission. Acceptance of an application does not authorize entry upon public lands or reservations of the United States for any purpose. The applicant should determine whether any additional Federal, state, or local permits are required.
- (2) Exhibits F and G must be submitted on separate drawings. Drawings for Exhibits F and G must have identifying title blocks and bear the following certification: "This drawing is a part of the application for license made by the undersigned this day of ,19 ."
- (3) Each application for a license for a water power project 5 megawatts or

less must include the information requested in the initial statement and lettered exhibits described by paragraphs (b) through (f) of this section, and must be provided in the form specified. The Commission reserves the right to require additional information, or another filing procedure, if data provided indicate such action to be appropriate.

(b) Initial statement.

BEFORE THE FEDERAL ENERGY REGULATORY COMMISSION

Application for License for a [Minor Water Power Project, or Major Water Power Project, 5 Megawatts or Less, as Appropriate]

(1)	(Name	of Appli	cant)	appli	es
to the Federal	Energy	Regulat	ory Co	ommi	is-
sion for	(lice	ense or ne	ew lice	nse,	as
appropriate) for	or the		(na	me	of
project) water	power	project,	as des	scrib	ed
hereinafter. (S	pecify	any pre	evious	FEF	łС
project number	designa	ition.)			

(2) The location of the project is:

State or territory:
County:
Fownship or nearby town:
Stream or other body of water:
(3) The exact name address and telephone

(3) The exact name, address, and telephone number of the applicant are:

(4) The exact name, address, and telephone number of each person authorized to act as agent for the applicant in this application, if applicable, are:

(5) The applicant is a [citizen of the United States, association of citizens of the United States, domestic corporation, municipality, or State, as appropriate] and (is/is not) claiming preference under section 7(a) of the Federal Power Act. See 16 U.S.C. 796.

(6)(i) The statutory or regulatory requirements of the state(s) in which the project would be located that affect the project as proposed with respect to bed and banks and the appropriation, diversion, and use of water for power purposes, and with respect to the right to engage in the business of developing, transmitting, and distributing power and in any other business necessary to accomplish the purposes of the license under the Federal Power Act, are: [provide citation and brief identification of the nature of each requirement; if the applicant is a municipality, the applicant must submit copies of applicable state or local laws or a municipal charter or, if such laws or documents are not

clear, any other appropriate legal authority, evidencing that the municipality is competent under such laws to engage in the business of developing, transmitting, utilizing, or distributing power.]

- (ii) The steps which the applicant has taken or plans to take to comply with each of the laws cited above are: [provide brief description for each requirement]
 - (7) Brief project description
- (i) Proposed installed generating capacity
 _____ MW.
- (ii) Check appropriate box:
- \square existing dam \square unconstructed dam \square existing dam, major modified project (see §4.30(b)(14))
- (8) Lands of the United States affected (shown on Exhibit G):

	(Name)	(Acres)
(i) National Forest		
(ii) Indian Reserva-		
tion.		
(iii) Public Lands		
Under Jurisdiction		
of.		
(iv) Other		
(v) Total U.S. Lands		

- (vi) Check appropriate box:
- ☐ Surveyed land ☐ Unsurveyed land
 (9) Construction of the project is planned
 to start within ___ months, and is planned
 to be completed within ___ months, from
 the date of issuance of license.
- (c) Exhibit A is a description of the project and the proposed mode of operation.
- (1) The exhibit must include, in tabular form if possible, as appropriate:
- (i) The number of generating units, including auxiliary units, the capacity of each unit, and provisions, if any, for future units:
 - (ii) The type of hydraulic turbine(s);
- (iii) A description of how the plant is to be operated, manual or automatic, and whether the plant is to be used for peaking;
- (iv) The estimated average annual generation in kilowatt-hours or mechanical energy equivalent;
- (v) The estimated average head on the plant;
- (vi) The reservoir surface area in acres and, if known, the net and gross storage capacity;
- (vii) The estimated minimum and maximum hydraulic capacity of the plant (flow through the plant) in cubic feet per second and estimated average flow of the stream or water body at the plant or point of diversion; for projects

with installed capacity of more than 1.5 megawatts, monthly flow duration curves and a description of the drainage area for the project site must be provided:

- (viii) Sizes, capacities, and construction materials, as appropriate, of pipelines, ditches, flumes, canals, intake facilities, powerhouses, dams, transmission lines, and other appurtenances; and
- (ix) The estimated cost of the project.
- (x) The estimated capital costs and estimated annual operation and maintenance expense of each proposed environmental measure.
- (2) State the purposes of project (for example, use of power output).
- (3) An estimate of the cost to develop the license application; and
- (4) The on-peak and off-peak values of project power, and the basis for estimating the values, for projects which are proposed to operate in a mode other than run-of-river.
- (5) The estimated average annual increase or decrease in project generation, and the estimated average annual increase or decrease of the value of project power due to a change in project operations (i.e., minimum bypass flows, limiting reservoir fluctuations) for an application for a new license:
- (6) The remaining undepreciated net investment, or book value of the project;
- (7) The annual operation and maintenance expenses, including insurance, and administrative and general costs;
- (8) A detailed single-line electrical diagram;
- (9) A statement of measures taken or planned to ensure safe management, operation, and maintenance of the project.
- (d) $Exhibit\ E$ is an Environmental Report.
- (1) For major unconstructed and major modified projects 5 MW or less. Any application must contain an Exhibit E conforming with the data and consultation requirements of §4.41(f) of this chapter, if the application is for license for a water power project which has or is proposed to have a total installed generating capacity greater

than 1.5 MW but not greater than 5 MW, and which:

- (i) Would use the water power potential of a dam and impoundment which, at the time of application, has not been constructed (see §4.30(b)(15)); or
- (ii) Involves any repair, modification or reconstruction of an existing dam that would result in a significant change in the normal maximum surface area or elevation of an existing impoundment or involves any change in existing project works or operations that would result in a significant environmental impact (see §4.30(b)(14)).
- (2) For minor projects and major projects at existing dams 5 MW or less. An application for license for either a minor water power project with a total proposed installed generating capacity of 1.5 MW or less or a major project—existing dam with a proposed total installed capacity of 5 MW or less must contain an Exhibit E under this subparagraph. See §4.38 for consultation requirements. The Environmental Report must contain the following information:
- (i) A description, including any maps or photographs which the applicant considers appropriate, of the environmental setting of the project, including vegetative cover, fish and wildlife resources, water quality and quantity, land and water uses, recreational uses, historical and archeological resources, and scenic and aesthetic resources. The report must include a discussion of endangered or threatened plant and animal species, any critical habitats, and any sites included in, or eligible for inclusion in, the National Register of Historic Places. The applicant may obtain assistance in the preparation of this information from state natural resources agencies, the state historic preservation officer, and from local offices of Federal natural resources agencies.
- (ii) A description of the expected environmental impacts from proposed construction or development and the proposed operation of the power project, including any impacts from any proposed changes in the capacity and mode of operation of the project if it is already generating electric power, and an explanation of the specific measures proposed by the applicant,

the agencies, and others to protect and enhance environmental resources and values and to mitigate adverse impacts of the project on such resources. The applicant must explain its reasons for not undertaking any measures proposed by any agency consulted.

- (iii) A description of the steps taken by the applicant in consulting with Federal, state, and local agencies with expertise in environmental matters during the preparation of this exhibit prior to filing the application for license with the Commission. In this report, the applicant must:
- (A) Indicate which agencies were consulted during the preparation of the environmental report and provide copies of letters or other documentation showing that the applicant consulted or attempted to consult with each of the relevant agencies (specifying each agency) before filing the application, including any terms or conditions of license that those agencies have determined are appropriate to prevent loss of, or damage to, natural resources; and
- (B) List those agencies that were provided copies of the application as filed with the Commission, the date or dates provided, and copies of any letters that may be received from agencies commenting on the application.
- (iv) Any additional information the applicant considers important.
- (e) Exhibit F. See 4.41(g) of this chapter.
- (f) $Exhibit\ G$. See § 4.41(h) of this chapter.

[Order 185, 46 FR 55949, Nov. 13, 1981, as amended by Order 413, 50 FR 11685, Mar. 25, 1985; Order 464, 52 FR 5449, Feb. 23, 1987; Order 513, 54 FR 23806, June 2, 1989; Order 2002, 68 FR 51120, Aug. 25, 2003; 68 FR 61742, Oct. 30, 2003]

Subpart H—Application for License for Transmission Line Only

§ 4.70 Applicability.

This subpart applies to any application for license issued solely for a transmission line that transmits power from a licensed water power project to

the point of junction with the distribution system or with the interconnected primary transmission system.

[Order 184, 46 FR 55942, Nov. 13, 1981, as amended by Order 2002, 68 FR 51120, Aug. 25, 2003; 68 FR 61742, Oct. 30, 2003]

§ 4.71 Contents of application.

An application for license for transmission line only must contain the following information in the form specified.

(a) Initial statement.

BEFORE THE FEDERAL ENERGY REGULATION COMMISSION

Application for License for Transmission Line Only

(1) [Name of applicant] applies to the Federal Energy Regulatory Commission for a [license or new license, as appropriate] for the [name of project] transmission line only, as described in the attached exhibits, that is connected with FERC Project No. ____, for which a license [was issued, or application was made, as appropriate] on the _____ day of _____, 19 .

(2)	The	location	of	the	transmission	line
would	l be:					

State or territory:

County:

Township or nearby town:

- (3) The proposed use or $\overline{\text{market for the}}$ power to be transmitted.
- (4) The exact name, business address, and telephone number of the applicant are:

(5) The applicant is a [citizen of the United States, association of citizens of the United States, domestic corporation, municipality, or State, as appropriate] and (is/is not) claiming preference under section 7(a) of the Federal Power Act. See 16 U.S.C. 796.

(6)(i) [For any applicant which, at the time of application for license for transmission line only, is a non-licensee.] The statutory or regulatory requirements of the state(s) in which the project would be located and that affect the project as proposed with respect to bed and banks and to the appropriation, diversion, and use of water for power purposes, and with respect to the right to engage in the business of developing, transmitting, and distributing power and in any other business necessary to accomplish the purposes of the license under the Federal Power Act, are: [provide citation and brief identification of the nature of each requirement; if the applicant is a municipality, the applicant must submit copies of applicable state or local laws or a municipal charter or, if such laws or documents are not clear, other appropriate legal authority, evidencing that the municipality is competent under such laws to engage in the business of developing, transmitting, utilizing, or distributing power.]

- (ii) [For any applicant which, at the time of application for license for transmission line only, is a licensee.] The statutory or regulatory requirements of the state(s) in which the transmission line would be located and that affect the project as proposed with respect to bed and banks and to the appropriation, diversion, and use of water for power purposes, are: [provide citations and brief identification of the nature of each requirement.]
- (iii) The steps which the applicant has taken or plans to take to comply with each of the laws cited above are: [provide brief descriptions for each law.]
- (b) Required exhibits. The application must contain the following exhibits, as appropriate:
- (1) For any transmission line that, at the time the application is filed, is not constructed and is proposed to be connected to a licensed water power project with an installed generating capacity of more than 5 MW—Exhibits A, B, C, D, E, F, and G under §4.41 of this chapter;
- (2) For any transmission line that, at the time the application is filed, is not constructed and is proposed to be connected to a licensed water power project with an installed generating capacity of 5 MW or less—Exhibits E, F, and G under §4.61 of this chapter; and
- (3) For any transmission line that, at the time the application is filed, has been constructed and is proposed to be connected to any licensed water power project—Exhibits E, F, and G under § 4.61 of this chapter.

[Order 184, 46 FR 55942, Nov. 13, 1981, as amended by Order 413, 50 FR 11685, Mar. 25, 1985; Order 699, 72 FR 45324, Aug. 14, 2007]

Subpart I—Application for Preliminary Permit; Amendment and Cancellation of Preliminary Permit

AUTHORITY: Federal Power Act, as amended 16 U.S.C. 792–828c; Department of Energy Organization Act, 42 U.S.C. 7101–7352; E.O. 12009, 42 FR 46267; Public Utility Regulatory Policies Act of 1978, 16 U.S.C. 2601–2645, unless otherwise noted.

§4.80 Applicability.

Sections 4.80 through 4.83 pertain to preliminary permits under Part I of the Federal Power Act. The sole purpose of a preliminary permit is to secure priority of application for a license for a water power project under Part I of the Federal Power Act while the permittee obtains the data and performs the acts required to determine the feasibility of the project and to support an application for a license.

[Order 54, 44 FR 61336, Oct. 25, 1979, as amended by Order 413, 50 FR 11685, Mar. 25, 1985]

§ 4.81 Contents of application.

Each application for a preliminary permit must include the following initial statement and numbered exhibits containing the information and documents specified:

(a) Initial statement:

State or territory:

BEFORE THE FEDERAL ENERGY REGULATORY COMMISSION

Application for Preliminary Permit

(1) [Name of applicant] applies to the Federal Energy Regulatory Commission for a preliminary permit for the proposed [name of project] water power project, as described in the attached exhibits. This application is made in order that the applicant may secure and maintain priority of application for a license for the project under Part I of the Federal Power Act while obtaining the data and performing the acts required to determine the feasibility of the project and to support an application for a license.

(2) The location of the proposed project is:

County:
Township or nearby town:
Stream or other body of water:
(3) The exact name, business address, and telephone number of the applicant are:
The exact name and business address o each person authorized to act as agent for the applicant in this application are:
(4) [Name of applicant] is a [citizen, asso ciation, citizens, domestic corporation, mu nicipality, or State, as appropriate] and (is/is not) claiming preference under section 7(a)

of the Federal Power Act. [If the applicant is

a municipality, the applicant must submit copies of applicable state or local laws or a municipal charter or, if such laws or documents are not clear, any other appropriate legal authority, evidencing that the municipality is competent under such laws to engage in the business of development, transmitting, utilizing, or distributing power].

(5) The proposed term of the requested permit is [period not to exceed 36 months].

- (6) If there is any existing dam or other project facility, the applicant must provide the name and address of the owner of the dam and facility. If the dam is federally owned or operated, provide the name of the agency.
- (b) Exhibit 1 must contain a description of the proposed project, specifying and including, to the extent possible:
- (1) The number, physical composition, dimensions, general configuration and, where applicable, age and condition, of any dams, spillways, penstocks, powerhouses, tailraces, or other structures, whether existing or proposed, that would be part of the project;
- (2) The estimated number, surface area, storage capacity, and normal maximum surface elevation (mean sea level) of any reservoirs, whether existing or proposed, that would be part of the project;
- (3) The estimated number, length, voltage, interconnections, and, where applicable, age and condition, of any primary transmission lines whether existing or proposed, that would be part of the project [see 16 U.S.C. 796(11)];
- (4) The total estimated average annual energy production and installed capacity (provide only one energy and capacity value), the hydraulic head for estimating capacity and energy output, and the estimated number, rated capacity, and, where applicable, the age and condition, of any turbines and generators, whether existing or proposed, that would be part of the project works;
- (5) All lands of the United States that are enclosed within the proposed project boundary described under paragraph (e)(3) of this section, identified and tabulated on a separate sheet by legal subdivisions of a public land survey of the affected area, if available. If the project boundary includes lands of the United States, such lands must be identified on a completed land description form, provided by the Commission.

The project location must identify any Federal reservation, Federal tracts, and townships of the public land surveys (or official protractions thereof if unsurveyed). A copy of the form must also be sent to the Bureau of Land Management state office where the project is located:

- (6) Any other information demonstrating in what manner the proposed project would develop, conserve, and utilize in the public interest the water resources of the region.
- (c) Exhibit 2 is a description of studies conducted or to be conducted with respect to the proposed project, including field studies. Exhibit 2 must supply the following information:
- (1) General requirement. For any proposed project, a study plan containing a description of:
- (i) Any studies, investigations, tests, or surveys that are proposed to be carried out, and any that have already taken place, for the purposes of determining the technical, economic, and financial feasibility of the proposed project, taking into consideration its environmental impacts, and of preparing an application for a license for the project; and
- (ii) The approximate locations and nature of any new roads that would be built for the purpose of conducting the studies; and
- (2) Work plan for new dam construction. For any development within the project that would entail new dam construction, a work plan and schedule containing:
- (i) A description, including the approximate location, of any field study, test, or other activity that may alter or disturb lands or waters in the vicinity of the proposed project, including floodplains and wetlands; measures that would be taken to minimize any such disturbance; and measures that would be taken to restore the altered or disturbed areas; and
- (ii) A proposed schedule (a chart or graph may be used), the total duration of which does not exceed the proposed term of the permit, showing the intervals at which the studies, investigations, tests, and surveys, identified under this paragraph are proposed to be completed.

- (iii) For purposes of this paragraph, new dam construction means any dam construction the studies for which would require test pits, borings, or other foundation exploration in the field.
- (3) Waiver. The Commission may waive the requirements of paragraph (c)(2) pursuant to §385.207 of this chapter, upon a showing by the applicant that the field studies, tests, and other activities to be conducted under the permit would not adversely affect cultural resources or endangered species and would cause only minor alterations or disturbances of lands and waters, and that any land altered or disturbed would be adequately restored.
- (4) Exhibit 2 must contain a statement of costs and financing, specifying and including, to the extent possible:
- (i) The estimated costs of carrying out or preparing the studies, investigations, tests, surveys, maps, plans or specifications identified under paragraph (c) of this section;
- (ii) The expected sources and extent of financing available to the applicant to carry out or prepare the studies, investigations, tests, surveys, maps, plans, or specifications identified under paragraph (c) of this section: and
- (d) Exhibit 3 must include a map or series of maps, to be prepared on United States Geological Survey topographic quadrangle sheets or similar topographic maps of a State agency, if available. The maps need not conform to the precise specifications of §4.39 (a) and (b). If the scale of any base map is not sufficient to show clearly and legibly all of the information required by this paragraph, the maps submitted must be enlarged to a scale that is adequate for that purpose. (If Exhibit 3 comprises a series of maps, it must also include an index sheet showing, by outline, the parts of the entire project covered by each map of the series.) The maps must show:
- (1) The location of the project as a whole with reference to the affected stream or other body of water and, if possible, to a nearby town or any permanent monuments or objects that can be noted on the maps and recognized in the field:
- (2) The relative locations and physical interrelationships of the principal

project features identified under paragraph (b) of this section;

- (3) A proposed boundary for the project, enclosing:
- (i) All principal project features identified under paragraph (b) of this section, including but not limited to any dam, reservoir, water conveyance facilities, powerplant, transmission lines, and other appurtenances; if the project is located at an existing Federal dam, the Federal dam and impoundment must be shown, but may not be included within the project boundary;
- (ii) Any non-Federal lands and any public lands or reservations of the United States [see 16 U.S.C. 796 (1) and (2)] necessary for the purposes of the project. To the extent that those public lands or reservations are covered by a public land survey, the project boundary must enclose each of and only the smallest legal subdivisions (quarter-quarter section, lots, or other subdivisions, identified on the map by subdivision) that may be occupied in whole or in part by the project.
- (4) Areas within or in the vicinity of the proposed project boundary which are included in or have been designated for study for inclusion in the National Wild and Scenic Rivers System; and
- (5) Areas within the project boundary that, under the provisions of the Wilderness Act, have been:
 - (i) Designated as wilderness area;
- (ii) Recommended for designation as wilderness area; or
- (iii) Designated as wilderness study area.

(Federal Power Act, as amended, 16 U.S.C. 792–828c (1976); Department of Energy Organization Act, 42 U.S.C. 7101–7352 (Supp. IV 1980); E.O. 12009, 3 CFR part 142 (1978); 5 U.S.C. 553 (Supp. IV 1980))

[Order 54, 44 FR 61336, Oct. 25, 1979, as amended by Order 123, 46 FR 9029, Jan. 28, 1981; 46 FR 11811, Feb. 11, 1981; Order 225, 47 FR 19056, May 3, 1982; Order 413, 50 FR 11685, Mar. 25, 1985; Order 2002, 68 FR 51120, Aug. 25, 2003; Order 655, 70 FR 33828, June 10, 2005; Order 699, 72 FR 45324, Aug. 14, 2007]

§ 4.82 Amendments.

(a) Any permittee may file an application for amendment of its permit, including any extension of the term of the permit that would not cause the total term to exceed three years.

(Transfer of a permit is prohibited by section 5 of the Federal Power Act.) Each application for amendment of a permit must conform to any relevant requirements of §4.81 (b), (c), (d), and (e).

- (b) If an application for amendment of a preliminary permit requests any material change in the proposed project, public notice of the application will be issued as required in §4.32(d)(2)(i).
- (c) If an application to extend the term of a permit is submitted not less than 30 days prior to the termination of the permit, the permit term will be automatically extended (not to exceed a total term for the permit of three years) until the Commission acts on the application for an extension. The Commission will not accept extension requests that are filed less than 30 days prior to the termination of the permit.

[Order 413, 50 FR 11685, Mar. 25, 1985, as amended by Order 499, 53 FR 27002, July 18, 1988]

§ 4.83 Cancellation and loss of priority.

- (a) The Commission may cancel a preliminary permit after notice and opportunity for hearing if the permittee fails to comply with the specific terms and conditions of the permit. The Commission may also cancel a permit for other good cause shown after notice and opportunity for hearing. Cancellation of a permit will result in loss of the permittee's priority of application for a license for the proposed project.
- (b) Failure of a permittee to file an acceptable application for a license before the permit expires will result in loss of the permittee's priority of application for a license for the proposed project.

[Order 413, 50 FR 11686, Mar. 25, 1985]

§4.84 Surrender of permit.

A permittee must submit a petition to the Commission before the permittee may voluntarily surrender its permit. Unless the Commission issues an order to the contrary, the permit will remain in effect through the thirtieth day after the Commission issues a public notice of receipt of the petition.

 $[{\rm Order}\ 413,\, 50\ {\rm FR}\ 11686,\, {\rm Mar.}\ 25,\, 1985]$

Subpart J—Exemption of Small Conduit Hydroelectric Facilities

§4.90 Applicability and purpose.

This subpart implements section 30 of the Federal Power Act and provides procedures for obtaining an exemption for constructed or unconstructed small conduit hydroelectric facilities, as defined in §4.30(b)(28), from all or part of the requirements of Part I of the Federal Power Act, including licensing, and the regulations issued under Part I.

[Order 76, 45 FR 28090, Apr. 28, 1980, as amended by Order 413, 50 FR 11686, Mar. 25, 1985; Order 2002, 68 FR 51121, Aug. 25, 2003]

§ 4.91 [Reserved]

§ 4.92 Contents of exemption application.

- (a) An application for exemption for this subpart must include:
- (1) An introductory statement, including a declaration that the facility for which application is made meets the requirements of §4.30(b)(28), the facility qualifies but for the discharge requirement of §4.30(b)(28)(v), the introductory statement must identify that fact and state that the application is accompanied by a petition for waiver of §4.30(b)(28)(v), filed pursuant to §385.207 of this chapter);
 - (2) Exhibits A, E, F, and G.
- (3) An appendix containing documentary evidence showing that the applicant has the real property interests required under §4.31(b); and
- (4) Identification of all Indian tribes that may be affected by the project.
- (b) *Introductory Statement*. The introductory statement must be set forth in the following format:

BEFORE THE FEDERAL ENERGY REGULATORY COMMISSION

Application for Exemption for Small Conduit Hydroelectric Facility

[Name of applicant] applies to the Federal Energy Regulatory Commission for an exemption for the [name of facility], a small conduit hydroelectric facility that meets the requirements of [insert the following language, as appropriate: "\$4.30(b)(28) of this subpart" or "\$4.30(b)(28) of this subpart, except paragraph (b)(28)(v)"], from certain provisions of Part I of the Federal Power Act.

The location of the facility is:
State or Territory:
County:
Fownship or nearby town:
The exact name and business address of eac applicant is:

The exact name and business address of each person authorized to act as agent for the applicant in this application is:

[Name of applicant] is [a citizen of the United States, an association of citizens of the United States, a municipality, State, or a corporation incorporated under the laws of (specify the United States or the state of incorporation, as appropriate), as appropriate].

The provisions of Part I of the Federal Power Act for which exemption is requested are:

[List here all sections or subsections for which exemption is requested.]

[If the facility does not meet the requirement of §4.30(b)(28)(v), add the following sentence: "This application is accompanied by a petition for waiver of §4.30(b)(28)(v), submitted pursuant to 18 CFR 385.207."]

- (c) Exhibit A. Exhibit A must describe the small conduit hydroelectric facility and proposed mode of operation with appropriate references to Exhibits F and G. To the extent feasible the information in this exhibit may be submitted in tabular form. The following information must be included:
- (1) A brief description of any conduits and associated consumptive water supply facilities, intake facilities, powerhouses, and any other structures associated with the facility.
- (2) The proximate natural sources of water that supply the related conduit.
- (3) The purposes for which the conduit is used.
- (4) The number of generating units, including auxiliary units, the capacity of each unit, and provisions, if any, for future units.
- (5) The type of each hydraulic tur-
- (6) A description of how the plant is to be operated, manually or automatically, and whether the plant is to be used for peaking.
- (7) Estimations of:
- (i) The average annual generation in kilowatt hours;
 - (ii) The average head of the plant;

- (iii) The hydraulic capacity of the plant (flow through the plant) in cubic feet per second;
- (iv) The average flow of the conduit at the plant or point of diversion (using best available data and explaining the sources of the data and the method of calculation); and
- (v) The average amount of the flow described in paragraph (c)(7)(iv) of this section available for power generation.
- (8) The planned date for beginning construction of the facility.
- (9) If the hydroelectric facility discharges directly into a natural body of water and a petition for waiver of §4.30(b)(28)(v) has not been submitted, evidence that a quantity of water equal to or greater than the quantity discharged from the hydroelectric facility is withdrawn from that water body downstream into a conduit that is part of the same water supply system as the conduit on which the hydroelectric facility is located.
- (10) If the hydroelectric facility discharges directly to a point of agricultural, municipal, or industrial consumption, a description of the nature and location of that point of consumption
- (11) A description of the nature and extent of any construction of a dam that would occur in association with construction of the proposed small conduit hydroelectric facility, including a statement of the normal maximum surface area and normal maximum surface elevation of any existing impoundment before and after that construction; and any evidence that the construction would occur for agricultural, municipal, or industrial consumptive purposes even if hydroelectric generating facilities were not installed.
- (d) Exhibit G. Exhibit G is a map of the project and boundary and must conform to the specifications of §4.41(h) of this chapter.
- (e) Exhibit E. This exhibit is an Environmental Report. It must be prepared pursuant to §4.38 and must include the following information, commensurate with the scope and environmental impact of the facility's construction and operation:
- (1) A description of the environmental setting in the vicinity of the facility, including vegetative cover, fish

- and wildlife resources, water quality and quantity, land and water uses, recreational use, socio-economic conditions, historical and archeological resources, and visual resources. The report must give special attention to endangered or threatened plant and animal species, critical habitats, and sites eligible for or included on the National Register of Historic Places. The applicant may obtain assistance in the preparation of this information from State natural resources agencies, the State historic preservation officer, and from local offices of Federal natural resources agencies.
- (2) A description of the expected environmental impacts resulting from the continued operation of an existing small conduit hydroelectric facility, or from the construction and operation of a proposed small conduit hydroelectric facility, including a discussion of the specific measures proposed by the applicant and others to protect and enhance environmental resources and to mitigate adverse impacts of the facility on them.
- (3) A description of alternative means of obtaining an amount of power equivalent to that provided by the proposed or existing facility.
- (4) Any additional information the applicant considers important.
- (f) Exhibit F. Exhibit F is a set of drawings showing the structures and equipment of the small conduit hydroelectric facility and must conform to the specifications of §4.41(g) of this chapter.

[Order 76, 45 FR 28090, Apr. 28, 1980, as amended by Order 413, 50 FR 11686, Mar. 25, 1985; Order 533, 56 FR 23153, May 20, 1991; Order 2002, 68 FR 51121, Aug. 25, 2003; Order 699, 72 FR 45324, Aug. 14, 2007]

§4.93 Action on exemption applica-

(a) An application for exemption that does not meet the eligibility requirements of \$4.30(b)(28)(v) may be accepted, provided the application has been accompanied by a request for waiver under \$4.92(a)(1) and the waiver request has not been denied. Acceptance of an application that has been accompanied by a request for waiver under \$4.92(a)(1) does not constitute a ruling on the

waiver request, unless expressly stated in the acceptance.

- (b) The Commission will circulate a notice of application for exemption to interested agencies and Indian tribes at the time the applicant is notified that the application is accepted for filing.
- (c) In granting an exemption the Commission may prescribe terms or conditions in addition to those set forth in §4.94, in order to:
- (1) Protect the quality or quantity of the related water supply for agricultural, municipal, or industrial consumption;
- (2) Otherwise protect life, health, or property:
- (3) Avoid or mitigate adverse environmental impact; or
- (4) Conserve, develop, or utilize in the public interest the water power resources of the region.
- (d) Conversion to license application. (1) If an application for exemption under this subpart is denied by the Commission, the applicant may convert the exemption application into an application for license for the hydroelectric project.
- (2) The applicant must provide the Commission with written notification, within 30 days after the date of issuance of the order denying exemption, that it intends to convert the exemption application into a license application. The applicant must submit to the Commission, no later than 90 days after the date of issuance of the order denying exemption, additional information that is necessary to conform the exemption application to the relevant regulations for a license application.
- (3) If all the information timely submitted is found sufficient, together with the application for exemption, to conform to the relevant regulations for a license application, the converted application will be considered accepted for filing as of the date that the exemption application was accepted for filing.

[Order 76, 45 FR 28090, Apr. 28, 1980, as amended by Order 413, 50 FR 11687, Mar. 25, 1985; Order 533, 56 FR 23153, May 20, 1991; Order 2002, 68 FR 51121, Aug. 25, 2003]

§4.94 Standard terms and conditions of exemption.

Any exemption granted under §4.93 for a small conduit hydroelectric facility is subject to the following standard terms and conditions:

- (a) Article 1. The Commission reserves the right to conduct investigations under sections 4(g), 306, 307, and 311 of the Federal Power Act with respect to any acts, complaints, facts, conditions, practices, or other matters related to the construction, operation, or maintenance of the exempt facility. If any term or condition of the exemption is violated, the Commission may revoke the exemption, issue a suitable order under section 4(g) of the Federal Power Act, or take appropriate action for enforcement, forfeiture, or penalties under Part III of the Federal Power Act.
- (b) Article 2. The construction, operation, and maintenance of the exempt project must comply with any terms and conditions that the United States Fish and Wildlife Service, the National Marine Fisheries Service, and any state fish and wildlife agencies have determined are appropriate to prevent loss of, or damage to, fish or wildlife resources or otherwise to carry out the purposes of the Fish and Wildlife Coordination Act, as specified in exhibit E of the application for exemption from licensing or in the comments submitted in response to the notice of exemption application.
- (c) Article 3. The Commission may revoke this exemption if actual construction of any proposed generating facilities has not begun within two years or has not been completed within four years from the effective date of this exemption. If an exemption is revoked under this article, the Commission will not accept from the prior exemption holder a subsequent application for exemption from licensing or a notice of exemption from licensing for the same project within two years of the revocation.
- (d) Article 4. In order to best develop, conserve, and utilize in the public interest the water resources of the region, the Commission may require that the exempt facilities be modified in structure or operation or may revoke this exemption.

Federal Energy Regulatory Commission

(e) Article 5. The Commission may revoke this exemption if, in the application process, material discrepancies, inaccuracies, or falsehoods were made by or on behalf of the applicant.

(f) Article 6. Before transferring any property interests in the exempt project, the exemption holder must inform the transferee of the terms and conditions of the exemption. Within 30 days of transferring the property interests, the exemption holder must inform the Commission of the identity and address of the transferee.

[Order 76, 45 FR 28090, Apr. 28, 1980, as amended by Order 413, 50 FR 11687, Mar. 25, 1985; Order 413—A, 56 FR 31331, July 10, 1991]

§ 4.95 Surrender of exemption.

(a) To voluntarily surrender its exemption, a holder of an exemption for a small conduit hydroelectric facility must file a petition with the Commission

(b)(1) If construction has begun, prior to filing a petition with the Commission, the exemption holder must consult with the fish and wildlife agencies in accordance with §4.38, substituting for the information required under §4.38(b)(1) information appropriate to the disposition and restoration of the project works and lands. The petition must set forth the exemption holder's plans with respect to disposition and restoration of the project works and lands.

(2) If construction has begun, public notice of the petition will be given, and, at least 30 days thereafter, the Commission will act upon the petition.

(c) If no construction has begun, unless the Commission issues an order to the contrary, the exemption will remain in effect through the thirtieth day after the Commission issues a public notice of receipt of the petition. New applications involving the site of the surrendered exemption may be filed on the next business day.

(d) Exemptions may be surrendered only upon fulfillment by the exemption holder of such obligations under the exemption as the Commission may prescribe and, if construction has begun, upon such conditions with respect to the disposition of such project works and restoration of project lands as may be determined by the Commission and

the Federal and state fish and wildlife agencies.

[Order 413, 50 FR 11687, Mar. 25, 1985]

§ 4.96 Amendment of exemption.

(a) An exemption holder must construct and operate its project as described in the exemption application approved by the Commission or its delegate.

(b) If an exemption holder desires to change the design, location, method of construction or operation of its project, it must first notify the appropriate Federal and state fish and wildlife agencies and inform them in writing of the changes it intends to implement. If these agencies determine that the changes would not cause the project to violate the terms and conditions imposed by the agencies, and if the changes would not materially alter the design, location, method of construction or operation of the project, the exemption holder may implement the changes. If any of these agencies determines that the changes would cause the project to violate the terms and conditions imposed by the agencies, or if the changes would materially alter the design, location, method of construction or the operation of the project works, the exemption holder may not implement the changes without first acquiring authorization from the Commission to amend its exemption, or acquiring a license that authorizes the project, as changed.

(c) An application to amend an exemption may be filed only by the holder of the exemption. An application to amend an exemption will be governed by the Commission's regulations governing applications for exemption. The Commission will not accept applications in competition with an application to amend an exemption, unless the Director of the Office of Energy Projects determines that it is in the public interest to do so.

[Order 413, 50 FR 11687, Mar. 25, 1985, as amended by Order 699, 72 FR 45324, Aug. 14, 2007]

Subpart K—Exemption of Small Hydroelectric Power Projects of 5 Megawatts or Less

§4.101 Applicability.

This subpart provides procedures for exemption on a case-specific basis from all or part of Part I of the Federal Power Act (Act), including licensing, for small hydroelectric power projects as defined in §4.30(b)(29).

(Energy Security Act of 1980, Pub. L. 96–294, 94 Stat. 611; Federal Power Act, as amended (16 U.S.C. 792–828c); Public Utility Regulatory Policies Act of 1978 (16 U.S.C. 2601–2645); and the Department of Energy Organization Act (42 U.S.C. 7101–7352); E.O. 12009, 3 CFR 142 (1978))

[Order 202, 47 FR 4243, Jan. 29, 1982, as amended by Order 413, 50 FR 11687, Mar. 25, 1985; Order 482, 52 FR 39630, Oct. 23, 1987; Order 2002, 68 FR 51121, Aug. 25, 2003]

§ 4.102 Surrender of exemption.

(a) To voluntarily surrender its exemption, a holder of an exemption for a small hydroelectric power project must file a petition with the Commission.

(b)(1) If construction has begun, prior to filing a petition with the Commission, the exemption holder must consult with the fish and wildlife agencies in accordance with §4.38, substituting for the information required under §4.38(b)(1) information appropriate to the disposition and restoration of the project works and lands. The petition must set forth the exemption holder's plans with respect to disposition and restoration of the project works and lands.

(2) If construction has begun, public notice of the petition will be given, and, at least 30 days thereafter, the Commission will act upon the petition. New applications involving the site may be filed on the next business day.

(c) If no construction had begun, unless the Commission issues an order to the contrary, the surrender will take effect at the close of the thirtieth day after the Commission issues a public notice of receipt of the petition. New applications involving the site may be filed on the next business day.

(d) Exemptions may be surrendered only upon fulfillment by the exemption holder of such obligations under the exemption as the Commission may prescribe and, if construction has begun, upon such conditions with respect to the disposition of such project works and restoration of project lands as may be determined by the Commission and the Federal and state fish and wildlife agencies.

(e) Where occupancy of United States lands or reservations has been permitted by a Federal agency having supervision over such lands, the exemption holder must concurrently notify that agency of the petition to surrender and of the steps that will be taken to restore the affected U.S. lands or reservations.

[Order 413, 50 FR 11688, Mar. 25, 1985]

§ 4.103 General provisions for case-specific exemption.

(a) Exemptible projects. Subject to the provisions in paragraph (b) of this section, §4.31(c), and §§4.105 and 4.106, the Commission may exempt on a case-specific basis any small hydroelectric power project from all or part of Part I of the Act, including licensing requirements. Any applications for exemption for a project shall conform to the requirements of §§4.107 or 4.108, as applicable.

(b) Limitation for licensed water power project. The Commission will not accept for filing an application for exemption from licensing for any project that is only part of a licensed water power project.

(c) Waiver. In applying for case-specific exemption from licensing, a qualified exemption applicant may petition under §385.207 of this chapter for waiver of any specific provision of §§4.102 through 4.107. The Commission will grant a waiver only if consistent with section 408 of the Energy Security Act of 1980.

[Order 413, 50 FR 11688, Mar. 25, 1985, as amended by Order 503, 53 FR 36568, Sept. 21, 1988]

§4.104 Amendment of exemption.

(a) An exemption holder must construct and operate its project as described in the exemption application approved by the Commission or its delegate.

(b) If an exemption holder desires to change the design, location, method of

of its construction or operation project, it must first notify the appropriate Federal and state fish and wildlife agencies and inform them in writing of the changes it intends to implement. If these agencies determine that the changes would not cause the project to violate the terms and conditions imposed by the agencies, and if the changes would not materially alter the design, location, method of construction or operation of the project, the exemption holder may implement the changes. If any of these agencies determines that the changes would cause the project to violate the terms and conditions imposed by that agency, or if the changes would materially alter the design, location, method of construction or the operation of the project works, the exemption holder may not implement the changes without first acquiring authorization from the Commission to amend its exemption or acquiring a license for the project works that authorizes the project, as changed.

(c) An application to amend an exemption may be filed only by the holder of an exemption. An application to amend an exemption will be governed by the Commission's regulations governing applications for exemption. The Commission will not accept applications in competition with an application to amend an exemption, unless the Director of the Office of Energy Projects determines that it is in the public interest to do so.

[Order 413, 50 FR 11688, Mar. 25, 1985, as amended by Order 699, 72 FR 45324, Aug. 14, 2007]

§ 4.105 Action on exemption applications.

(a) Exemption from provisions other than licensing. An application for exemption of a small hydroelectric power project from provisions of Part I of the Act other than the licensing requirement will be processed and considered as part of the related application for license or amendment of license.

(b)(1) Consultation. The Commission will circulate a notice of application for exemption from licensing to interested agencies and Indian tribes at the time the applicant is notified that the application is accepted for filing.

- (2) Non-standard terms and conditions. In approving any application for exemption from licensing, the Commission may prescribe terms or conditions in addition to those set forth in §4.106 in order to:
- (i) Protect the quality or quantity of the related water supply;
- (ii) Otherwise protect life, health, or property;
- (iii) Avoid or mitigate adverse environmental impact; or
- (iv) Better conserve, develop, or utilize in the public interest the water resources of the region.

(Energy Security Act of 1980, Pub. L. 96–294, 94 Stat. 611; Federal Power Act, as amended (16 U.S.C. 792–828c); Public Utility Regulatory Policies Act of 1978 (16 U.S.C. 2601–2645); and the Department of Energy Organization Act (42 U.S.C. 7101–7352); E.O. 12009, 3 CFR 142 (1978))

[Order 106, 45 FR 76123, Nov. 18, 1980, as amended by Order 202, 47 FR 4246, Jan. 29, 1982; Order 413, 50 FR 11688, Mar. 25, 1985; Order 533, 56 FR 23154, May 20, 1991]

§ 4.106 Standard terms and conditions of case-specific exemption from licensing.

Any case-specific exemption from licensing granted for a small hydroelectric power project is subject to the following standard terms and conditions:

(a) Article 1. The Commission reserves the right to conduct investigations under sections 4(g), 306, 307, and 311 of the Federal Power Act with respect to any acts, complaints, facts, conditions, practices, or other matters related to the construction, operation, or maintenance of the exempt project. If any term or condition of the exemption is violated, the Commission may revoke the exemption, issue a suitable order under section 4(g) of the Federal Power Act, or take appropriate action for enforcement, forfeiture, or penalties under Part III of the Federal Power Act.

(b) Article 2. The construction, operation, and maintenance of the exempt project must comply with any terms and conditions that the United States Fish and Wildlife Service, the National Marine Fisheries Service, and any state fish and wildlife agencies have determined are appropriate to prevent loss of, or damage to, fish or wildlife

resources or otherwise to carry out the purposes of the Fish and Wildlife Coordination Act, as specified in exhibit E of the application for exemption from licensing or in the comments submitted in response to the notice of exemption application.

- (c) Article 3. The Commission may revoke this exemption if actual construction of any proposed generating facilities has not begun within two years or has not been completed within four years from the date on which this exemption was granted. If an exemption is revoked under this article, the Commission will not accept from the prior exemption holder a subsequent application for exemption from licensing for the same project within two years of the revocation.
- (d) Article 4. This exemption is subject to the navigation servitude of the United States if the project is located on navigable waters of the United States.
- (e) Article 5. This exemption does not confer any right to use or occupy any Federal lands that may be necessary for the development or operation of the project. Any right to use or occupy any Federal lands for those purposes must be obtained from the administering Federal land agencies. The Commission may accept a license application submitted by any qualified license applicant and revoke this exemption, if any necessary right to use or occupy Federal lands for those purposes has not been obtained within one year from the date on which this exemption was granted.
- (f) Article 6. In order to best develop, conserve, and utilize in the public interest the water resources of the region, the Commission may require that the exempt facilities be modified in structure or operation or may revoke this exemption.
- (g) Article 7. The Commission may revoke this exemption if, in the application process, material discrepancies, inaccuracies, or falsehoods were made by or on behalf of the applicant.
- (h) Article 8. Any exempted small hydroelectric power project that utilizes a dam that is more than 33 feet in height above streambed, as defined in 18 CFR 12.31(c) of this chapter, impounds more than 2,000 acre-feet of

water, or has a significant or high hazard potential, as defined in 33 CFR part 222, is subject to the following provisions of 18 CFR part 12, as it may be amended:

- (1) Section 12.4(b)(1) (i) and (ii), (b)(2) (i) and (iii), (b)(iv), and (b)(v);
 - (2) Section 12.4(c);
 - (3) Section 12.5;
 - (4) Subpart C; and
 - (5) Subpart D.

For the purposes of applying these provisions of 18 CFR part 12, the exempted project is deemed to be a licensed project development and the owner of the exempted project is deemed to be a licensee.

(i) Before transferring any property interests in the exempt project, the exemption holder must inform the transferee of the terms and conditions of the exemption. Within 30 days of transferring the property interests, the exemption holder must inform the Commission of the identity and address of the transferee.

[Order 106, 45 FR 76123, Nov. 18, 1980; 45 FR 77420, Nov. 24, 1980, as amended by Order 202, 47 FR 4246, Jan. 29, 1982; Order 413, 50 FR 11688, Mar. 25, 1985; Order 482, 52 FR 39630, Oct. 23, 1987; Order 413-A, 56 FR 31331, July 10, 1991]

§ 4.107 Contents of application for exemption from licensing.

- (a) General requirements. An application for exemption from licensing submitted under this subpart must contain the introductory statement, the exhibits described in this section, and, if the project structures would use or occupy any lands other than Federal lands, an appendix containing documentary evidence showing that applicant has the real property interests required under §4.31(c)(2)(ii). The applicant must identify in its application all Indian tribes that may be affected by the project.
- (b) *Introductory statement*. The application must include an introductory statement that conforms to the following format:

BEFORE THE FEDERAL ENERGY REGULATORY COMMISSION

Application for Exemption of Small Hydroelectric Power Project From Licensing

(1) [Name of applicant] applies to the Federal Energy Regulatory Commission for an exemption for [name of project], a small hydroelectric power project that is proposed to have an installed capacity of 5 megawatts or less, from licensing under the Federal Power Act. [If applicable: The project is currently licensed as FERC Project No. ______.]

(2) The location of the project is:
[State or territory]

each applicant are:

[County]
[Township or nearby town]

[Stream or body of water]

(3) The exact name and business address of

(4) The exact name and business address of each person authorized to act as agent for

the applicant in this application are:

(5) [Name of applicant] is [specify, as appropriate: a citizen of the United States or other identified nation; an association of citizens of the United States or other identified nation; a municipality; a state; or a corporation incorporated under the laws of

(specify the United States or the state or na-

tion of incorporation, as appropriate).]

- (c) Exhibit A. Exhibit A must describe the small hydroelectric power project and its proposed mode of operation. To the extent feasible, the information in this exhibit may be submitted in tabular form. The applicant must submit the following information:
- (1) A brief description of any existing dam and impoundment proposed to be utilized by the small hydroelectric power project and any other existing or proposed project works and appurtenant facilities, including intake facilities, diversion structures, powerhouses, primary transmission lines, penstocks, pipelines, spillways, and other structures, and the sizes, capacities, and construction materials of those structures.
- (2) The number of existing and proposed generating units at the project, including auxiliary units, the capacity of each unit, any provisions for future

units, and a brief description of any plans for retirement or rehabilitation of existing generating units.

- (3) The type of each hydraulic turbine of the small hydroelectric power project.
- (4) A description of how the power plant is to be operated, that is, run-of-river or peaking.
- (5) A graph showing a flow duration curve for the project. Identify stream gauge(s) and period of record used. If a synthetic record is utilized, provide details concerning its derivation. Furnish justification for selection of installed capacity if the hydraulic capacity of proposed generating unit(s) plus the minimum flow requirements, if not usable for power production, is less than the stream flow that is exceeded 25 percent of the time.
 - (6) Estimations of:
- (i) The average annual generation in kilowatt-hours;
- (ii) The average and design head of the power plant;
- (iii) The hydraulic capacity of each turbine of the power plant (flow through the plant) in cubic feet per second:
- (iv) The number of surface acres of the man-made or natural impoundment used, if any, at its normal maximum surface elevation and its net and gross storage capacities in acre-feet.
- (7) The planned date for beginning and completing the proposed construction or development of generating facilities.
- (8) A description of the nature and extent of any repair, reconstruction, or other modification of a dam that would occur in association with construction or development of the proposed small hydroelectric power project, including a statement of the normal maximum surface area and normal maximum surface elevation of any existing impoundment before and after construction.
- (d) Exhibit G. Exhibit G is a map of the project and boundary and must conform to the specifications of §4.41(h) of this chapter.
- (e) Exhibit E. This exhibit is an environmental report that must include the following information, commensurate with the scope and environmental impact of the construction and operation of the small hydroelectric power

project. See §4.38 for consultation requirements.

- (1) A description of the environmental setting of the project, including vegetative cover, fish and wildlife resources, water quality and quantity, land and water uses, recreational uses, historical and archeological resources, and scenic and aesthetic resources. The report must list any endangered or threatened plant and animal species. any critical habitats, and any sites eligible for or included on the National Register of Historic Places. The applicant may obtain assistance in the preparation of this information from state natural resources agencies, the state historic preservation officer, and from local offices of Federal natural resources agencies.
- (2) A description of the expected environmental impacts from the proposed construction or development and the proposed operation of the small hydroelectric power project, including any impacts from any proposed changes in the capacity and mode of operation of the project if it is already generating electric power, and an explanation of the specific measures proposed by the applicant, the agencies consulted, and others to protect and enhance environmental resources and values and to mitigate adverse impacts of the project on such resources.
- (3) Any additional information the applicant considers important.
- (f) Exhibit F. Exhibit F is a set of drawings showing the structures and equipment of the small hydroelectric facility and must conform to the specifications of §4.41(g) of this chapter.

[Order 106, 45 FR 76123, Nov. 18, 1980, as amended by Order 225, 47 FR 19056, May 3, 1982; Order 413, 50 FR 11689, Mar. 25, 1985; Order 494, 53 FR 15381, Apr. 29, 1988; Order 533, 56 FR 23154, May 20, 1991; Order 2002, 68 FR 51121, Aug. 25, 2003; Order 699, 72 FR 45324, Aug. 14, 2007]

§ 4.108 Contents of application for exemption from provisions other than licensing.

An application for exemption of a small hydroelectric power project from provisions of Part I of the Act other than the licensing requirement need not be prepared according to any specific format, but must be included as an identified appendix to the related

application for license or amendment of license. The application for exemption must list all sections or subsections of Part I of the Act for which exemption is requested.

[Order 106, 45 FR 76123, Nov. 18, 1980]

Subpart L—Application for Amendment of License

§ 4.200 Applicability.

This part applies to any application for amendment of a license, if the applicant seeks to:

- (a) Make a change in the physical features of the project or its boundary, or make an addition, betterment, abandonment, or conversion, of such character as to constitute an alteration of the license:
- (b) Make a change in the plans for the project under license; or
- (c) Extend the time fixed in the license for commencement or completion of project works.

[Order 184, 46 FR 55943, Nov. 13, 1981, as amended by Order 2002, 68 FR 51121, Aug. 25, 2003]

§ 4.201 Contents of application.

An application for amendment of a license for a water power project must contain the following information in the form specified.

(a) Initial statement.

BEFORE THE FEDERAL ENERGY REGULATORY COMMISSION

Application for Amendment of License

- (1) [Name of applicant] applies to the Federal Energy Regulatory Commission for an amendment of license for the [name of project] water power project.
- (2) The exact name, business address, and telephone number of the applicant are:

(3) The applicant is a [citizen of the United
States, association of citizens of the United
States, domestic corporation, municipality,
or state, as appropriate, see 16 U.S.C. 796], li-
censee for the water power project, des-
ignated as Project No in the records
of the Federal Energy Regulatory Commis-
sion, issued on the day of
10

(4) The amendments of license proposed and the reason(s) why the proposed changes

are necessary, are: [Give a statement or description]

- (5)(i) The statutory or regulatory requirements of the state(s) in which the project would be located that affect the project as proposed with respect to bed and banks and to the appropriation, diversion, and use of water for power purposes are: [provide citation and brief identification of the nature of each requirement.]
- (ii) The steps which the applicant has taken or plans to take to comply with each of the laws cited above are: [provide brief description for each law.]
- (b) Required exhibits for capacity related amendments. Any application to amend a license for a hydropower project that involves additional capacity not previously authorized, and that would increase the actual or proposed total installed capacity of the project, would result in an increase in the maximum hydraulic capacity of the project of 15 percent or more, and would result in an increase in the installed nameplate capacity of 2 megawatts or more, must contain the following exhibits, or revisions or additions to any exhibits on file, commensurate with the scope of the licensed project:
- (1) For amendment of a license for a water power project that, at the time the application is filed, is not constructed and is proposed to have a total installed generating capacity of more than 5 MW—Exhibits A, B, C, D, E, F, and G under §4.41 of this chapter;
- (2) For amendment of a license for a water power project that, at the time the application is filed, is not constructed and is proposed to have a total installed generating capacity of 1.5 MW or less—Exhibits E, F, and G under § 4.61 of this chapter;
- (3) For amendment of a license for a water power project that, at the time the application is filed, is not constructed and is proposed to have a total installed generating capacity of 5 MW or less, but more than 1.5 MW—Exhibits F and G under §4.61 of this chapter, and Exhibit E under §4.41 of this chapter:
- (4) For amendment of a license for a water power project that, at the time the application for amendment is filed, has been constructed, and is proposed to have a total installed generating capacity of 5 MW or less—Exhibit E, F and G under §4.61 of this chapter;

- (5) For amendment of a license for a water power project that, at the time the application is filed, has been constructed and is proposed to have a total installed generating capacity of more than 5 MW—Exhibits A, B, C, D, E, F, and G under §4.51 of this chapter.
- (c) Required exhibits for non-capacity related amendments. Any application to amend a license for a water power project that would not be a capacity related amendment as described in paragraph (b) of this section must contain those exhibits that require revision in light of the nature of the proposed amendments.
- (d) Consultation and waiver. (1) If an applicant for license under this subpart believes that any exhibit required under paragraph (b) of this section is inappropriate with respect to the particular amendment of license sought by the applicant, a petition for waiver of the requirement to submit such exhibit may be submitted to the Commission under §385.207(c)(4) of this chapter, after consultation with the Commission's Division by Hydropower Licensing.
- (2) A licensee wishing to file an application for amendment of license under this section may seek advice from the Commission staff regarding which exhibits(s) must be submitted and whether the proposed amendment is consistent with the scope of the existing licensed project.

[Order 184, 46 FR 55943, Nov. 13, 1981, as amended by Order 225, 47 FR 19056, May 3, 1982; 48 FR 4459, Feb. 1, 1983; 48 FR 16653, Apr. 19, 1983; Order 413, 50 FR 11689, Mar. 25, 1985; Order 533, 56 FR 23154, May 20, 1991]

§ 4.202 Alteration and extension of license.

- (a) If it is determined that approval of the application for amendment of license would constitute a significant alteration of license pursuant to section 6 of the Act, 16 U.S.C. 799, public notice of such application shall be given at least 30 days prior to action upon the application.
- (b) Any application for extension of time fixed in the license for commencement or completion of construction of project works must be filed with the Commission not less than three

months prior to the date or dates so fixed.

[Order 184, 46 FR 55943, Nov. 13, 1981]

Subpart M—Fees Under Section 30(e) of the Act

SOURCE: Order 487, 52 FR 48404, Dec. 22, 1987, unless otherwise noted.

§4.300 Purpose, definitions, and applicability.

- (a) Purpose. This subpart implements the amendments of section 30 of the Federal Power Act enacted by section 7(c) of the Electric Consumers Protection Act of 1986 (ECPA). It establishes procedures for reimbursing fish and wildlife agencies for costs incurred in connection with applications for an exemption from licensing and applications for licenses seeking benefits under section 210 of the Public Utility Regulatory Policies Act of 1978, as amended, for a project that would impound or divert the water of a natural watercourse by means of a new dam or diversion.
- (b) *Definitions*. For the purposes of this subpart—
- (1) *Cost* means an expenditure made by a fish and wildlife agency:
- (i) On or after the effective date of this regulation for an application filed on or after the effective date of this regulation; and
- (ii) Directly related to setting mandatory terms and conditions for a proposed project pursuant to section 30(c) of the Federal Power Act.
- (2) Cost statement means a statement of the total costs for which a fish and wildlife agency requests reimbursement including an itemized schedule of costs including, but not limited to, costs of fieldwork and testing, contract costs, travel costs, personnel costs, and administrative and overhead costs.
- (3) Mandatory terms and conditions means terms and conditions of a license or exemption that a fish and wildlife agency determines are appropriate to prevent loss of, or damage to, fish and wildlife resources pursuant to section 30(c) of the Federal Power Act.
- (4) New dam or diversion license applicant means an applicant for a license for a project that would impound or di-

vert the water of a natural watercourse by means of a new dam or diversion, as defined in section 210(k) of the Public Utility Regulatory Policies Act of 1978, as amended.

- (5) PURPA benefits means benefits under section 210 of the Public Utility Regulatory Policies Act of 1978, as amended.
- (6) Section 30(c) application means an application for an exemption from licensing or a new dam or diversion license application seeking PURPA benefits.
- (c) *Applicability*. Except as provided in paragraph (d) of this section, this subpart applies to:
- (1) Any application for exemption filed on or after the effective date of these regulations for costs incurred by fish and wildlife agencies after the effective date of these regulations;
- (2) Any new dam or diversion license application seeking PURPA benefits filed on or after April 16, 1988;
- (3) Any new dam or diversion license application seeking PURPA benefits filed after the effective date of this regulation, but before April 16, 1988, if the applicant fails to demonstrate in a monetary resources petition filed with the Commission pursuant to §292.208 of this chapter that, before October 16, 1986, it had committed substantial monetary resources directly related to the development of the proposed project and to the diligent and timely completion of all requirements of the Commission for filing an acceptable application; and
- (4) Any new dam or diversion license application seeking PURPA benefits filed after the effective date of this regulation, if the application is not accepted for filing before October 16, 1989.
- (d) Exceptions. (1) This subpart does not apply to any new dam or diversion license application seeking PURPA benefits if the moratorium described in section 8(e) of ECPA is in effect. The moratorium will end at the expiration of the first full session of Congress following the session during which the Commission reports to Congress on the results of the study required under section 8(d) of ECPA.
- (2) This subpart does not apply to any new dam or diversion license application seeking PURPA benefits for a

project located at a Government dam, as defined in section 3(10) of the Federal Power Act, at which non-Federal hydroelectric development is permissible.

§ 4.301 Notice to fish and wildlife agencies and estimation of fees prior to filing.

- (a) Notice to agencies—(1) New dam or diversion license applicants. During the initial stage or pre-filing agency consultation under §4.38(b)(1), a prospective new dam or diversion license applicant must inform each fish and wildlife agency consulted in writing with a copy to the Commission whether it will seek PURPA benefits.
- (2) Exemption applicants. During the initial stage of pre-filing agency consultation under §4.38(b)(1), a prospective exemption applicant must notify each fish and wildlife agency consulted that it will seek an exemption from licensing.
- (b) Estimate of fees. Within the comment period provided in §4.38(b)(2)(iv), a fish and wildlife agency must provide a prospective section 30(c) applicant with a reasonable estimate of the total costs the agency anticipates it will incur to set mandatory terms and conditions for the proposed project. An agency may provide an applicant with an updated estimate as it deems necessary. If an agency believes that its most recent estimate will be exceeded by more than 25 percent, it must supply the prospective applicant or applicant with a new estimate and submit a copy to the Commission.

§ 4.302 Fees at filing.

- (a) Filing requirement. A section 30(c) application must be accompanied by a fee or a bond, together with copies of the most recent cost estimates provided by fish and wildlife agencies pursuant to §4.301(b).
- (b) Amount. The fee required under paragraph (a) of this section must be in an amount equal to 50 percent of the most recent cost estimates provided by fish and wildlife agencies pursuant to §4.301(b). In lieu of this amount, an applicant may provide an unlimited term surety bond from a company on the Department of Treasury's list of companies certified to write surety bonds.

Applicants bonded by a company whose certification by the Department of the Treasury lapses must provide evidence of purchase of another bond from a certified company. A bond must be for an amount no less than 100 percent of the agencies' most recent cost estimates pursuant to §4.301(b).

(c) Failure to file. The Commission will reject a section 30(c) application if the applicant fails to comply with the provisions of paragraphs (a) and (b) of this section.

§ 4.303 Post-filing procedures.

- (a) Submission of cost statement—1) Accepted applications. Within 60 days after the last date for filing mandatory terms and conditions pursuant to §4.32(c)(4) for a new dam or diversion license application seeking PURPA benefits, §4.93(b) for an application for exemption of a small conduit hydroelectric facility, or §4.105(b)(1) for an application for case-specific exemption of a small hydroelectric power project, a fish and wildlife agency must file with the Commission a cost statement of the reasonable costs the agency incurred in setting mandatory terms and conditions for the proposed project. An agency may request, in writing, along with any supporting documentation an extension of this 60-day period.
- (2) Rejected, withdrawn or dismissed applications. The Director of the Office of Energy Projects (Director) will, by letter, notify each fish and wildlife agency if a section 30(c) application is rejected, withdrawn or dismissed. Within 60 days from the date of notification, a fish and wildlife agency must file with the Commission a cost statement of the reasonable costs the agency incurred prior to the date the application was rejected, withdrawn, or dismissed. An agency may submit a written request for an extension of this 60day period along with any supporting documentation.
- (b) If an agency has not submitted a cost statement or extension request within the time provided in paragraph (a)(2) of this section, it waives its right to receive fees for that project pursuant to this subpart.
- (c) Billing. After the Commission receives a cost statement from all fish and wildlife agencies as required by

§ 4.303

paragraph (a) of this section, the Commission will bill the section 30(c) applicant. The bill will show:

- (1) The cost statement submitted to the Commission by each fish and wildlife agency:
- (2) Any amounts already paid by the applicant pursuant to §4.302; and
- (3)(i) The amount due, if the amount already paid by the applicant pursuant to §4.302 is less than the total of all the cost statements; or
- (ii) The amount to be refunded to the applicant, if the amount already paid by the applicant pursuant to §4.302 is more than the total of all the cost statements.
- (d) Within 45 days from the date of a bill issued under paragraph (b) of this section, a section 30(c) applicant must pay in full to the Commission any remaining amounts due on the cost statements regardless of whether any of these amounts are in dispute.
- (e) Dispute procedures—(1) When to dispute. Any dispute regarding the reasonableness of any fish and wildlife agency cost statement must be made within 45 days from the date of a bill issued under paragraph (b) of this section.
- (2) Assessment of disputed cost statements The burden of showing that an agency's cost statement is unreasonable is on the applicant. However, a fish and wildlife agency must supply the disputing applicant and the Commission with the documentation necessary to support its cost statement. The Director of the Office of Energy Projects will determine the reasonableness of a disputed fish and wildlife agency cost statement. The Director's decision will be in writing. The Director will notify the disputing applicant and the fish and wildlife agency of the decision by letter. Any decision of the Director may be appealed by either party pursuant to 18 CFR 385.1902. In deciding whether or not a disputed cost statement is reasonable, the Director will review the application, the disputed cost statement and any other documentation relating to the particular environmental problems associated with the disputing applicant's proposed project. The Director will consider such factors as:

- (i) The time the fish and wildlife agency spent reviewing the application:
- (ii) The proportion of the cost statement to the time the fish and wildlife agency spent reviewing the application;
- (iii) Whether the fish and wildlife agency's expenditures conform to Federal expenditure guidelines for such items as travel, per diem, personnel, and contracting; and
- (iv) Whether the studies conducted by the agency, if any, are duplicative, limited to the proposed project area, unnecessary to determine the impacts to or mitigation measures for the particular fish and wildlife resources affected by the proposed project, or otherwise unnecessary to set terms and conditions for the proposed project.
- (3) Unreasonable cost statements. If the Director determines that a disputed fish and wildlife agency cost statement is unreasonable, the disputing applicant and the fish and wildlife agency will be afforded 45 days from the date of notification to attempt to reach an agreement regarding the reimbursable costs of the agency. If the disputing applicant and the fish and wildlife agency fail to reach an agreement on the disputed cost statement within 45 days from the date of notification, the Director will determine the costs that the agency should reasonably have incurred.
- (f) Refunds. (1) If the amount paid by a section 30(c) applicant under §4.302 exceeds the total amount of the cost statements submitted by fish and wildlife agencies under paragraph (a) of this section, the Commission will notify the Treasury to refund the difference to the applicant within 45 days from the date of the bill issued to the applicant under paragraph (b) of this section.
- (2) If the amount paid by a section 30(c) applicant exceeds the amount determined to be reasonable by the Director pursuant to paragraph (d)(2) of this section, the Commission will notify the Treasury to refund the difference to the applicant within 45 days of the resolution of all dispute proceedings.

[Order 487, 52 FR 48404, Dec. 22, 1987, as amended by Order 647, 69 FR 32438, June 10,

§4.304 Payment.

- (a) A payment required under this subpart must be made by check payable to the United States Treasury. The check must indicate that the payment is for *ECPA Fees*.
- (b) If a payment required under this subpart is not made within the time period prescribed for making such payment, interest and penalty charges will be assessed. Interest and penalty charges will be computed in accordance with 31 U.S.C. 3717 and 4 CFR part 102.
- (c) The Commission will not issue a license or exemption, unless the applicant has made full payments of any fees due under §4.303(c).

§ 4.305 Enforcement.

- (a) The Commission may take any appropriate action permitted by law if a section 30(c) applicant does not make a payment required under this subpart. The Commission will not be liable to any fish and wildlife agency for failure to collect any amounts under this subpart.
- (b) If the Commission is unable to collect the full amount due by a section 30(c) applicant on behalf of more than one agency, the amount the Commission does collect will be distributed to the agencies on a *pro-rata* basis except if an agency's cost statement is greater than its most recent estimate to the applicant under §4.301(b), then the difference between the estimate and the cost statement will not be reimbursed until any amounts owed to other agencies have been paid.

PART 5—INTEGRATED LICENSE APPLICATION PROCESS

Sec.

- 5.1 Applicability, definitions, and requirement to consult.
- 5.2 Document availability
- 5.3 Process selection.
- 5.4 Acceleration of a license expiration date.
- 5.5 Notification of intent.
- 5.6 Pre-application document.
- 5.7 Tribal consultation.
- 5.8 Notice of commencement of proceeding and scoping document, or of approval to use traditional licensing process or alternative procedures.
- 5.9 Comments and information or study requests.

- 5.10 Scoping document 2.
- 5.11 Potential Applicant's proposed study plan and study plan meetings.
- 5.12 Comments on proposed study plan.
- 5.13 Revised study plan and study plan determination.
- 5.14 Formal study dispute resolution process.
- 5.15 Conduct of studies.
- 5.16 Preliminary licensing proposal.
- 5.17 Filing of application.
- 5.18 Application content.
- 5.19 Tendering notice and schedule.
- 5.20 Deficient applications.
- 5.21 Additional information.
- 5.22 Notice of acceptance and ready for environmental analysis.
- 5.23 Response to notice.
- 5.24 Applications not requiring a draft NEPA document.
- 5.25 Applications requiring a draft NEPA document.
- 5.26 Section 10(j) process.
- 5.27 Amendment of application.
- 5.28 Competing applications.
- 5.29 Other provisions.
- 5.30 Critical Energy Infrastructure Information.
- 5.31 Transition provision.

AUTHORITY: 16 U.S.C. 791a-825r, 2601–2645; 42 U.S.C. 7101–7352.

Source: Order 2002, $68\ \mathrm{FR}\ 51121,\ \mathrm{Aug}.\ 25,\ 2003,\ \mathrm{unless}\ \mathrm{otherwise}\ \mathrm{noted}.$

§5.1 Applicability, definitions, and requirement to consult.

- (a) This part applies to the filing and processing of an application for an:
- (1) Original license;
- (2) New license for an existing project subject to Sections 14 and 15 of the Federal Power Act; or
 - (3) Subsequent license.
- (b) Definitions. The definitions in $\S4.30(b)$ of this chapter and $\S16.2$ of this chapter apply to this chapter.
- (c) Who may file. Any citizen, association of citizens, domestic corporation, municipality, or state may develop and file a license application under this part.
- (d) Requirement to consult. (1) Before it files any application for an original, new, or subsequent license under this part, a potential applicant must consult with the relevant Federal, state, and interstate resource agencies, including as appropriate the National Marine Fisheries Service, the United States Fish and Wildlife Service, Bureau of Indian Affairs, the National

§ 5.2

Park Service, the United States Environmental Protection Agency, the Federal agency administering any United States lands utilized or occupied by the project, the appropriate state fish and wildlife agencies, the appropriate state water resource management agencies, the certifying agency or Indian tribe under Section 401(a)(1) of the Federal Water Pollution Control Act (Clean Water Act), 33 U.S.C. 1341(c)(1)), the agency that administers the Coastal Zone Management Act, 16 U.S.C. §1451-1465, any Indian tribe that may be affected by the project, and members of the public. A potential license applicant must file a notification of intent to file a license application pursuant to $\S 5.5$ and a pre-application document pursuant to the provisions of §5.6.

- (2) The Director of the Office of Energy Projects will, upon request, provide a list of known appropriate Federal, state, and interstate resource agencies, Indian tribes, and local, regional, or national non-governmental organizations likely to be interested in any license application proceeding.
- (e) Purpose. The purpose of the integrated licensing process provided for in this part is to provide an efficient and timely licensing process that continues to ensure appropriate resource protections through better coordination of the Commission's processes with those of Federal and state agencies and Indian tribes that have authority to condition Commission licenses.
- (f) Default process. Each potential original, new, or subsequent license applicant must use the license application process provided for in this part unless the potential applicant applies for and receives authorization from the Commission under this part to use the licensing process provided for in:
- (1) 18 CFR part 4, Subparts D-H and, as applicable, part 16 (*i.e.*, traditional process), pursuant to paragraph (c) of this section: or
- (2) Section 4.34(i) of this chapter, *Alternative procedures*.

[Order 2002, 68 FR 51121, Aug. 25, 2003; 68 FR 61742, Oct. 30, 2003; 68 FR 69957, Dec. 16, 2003]

§ 5.2 Document availability.

(a) Pre-application document. (1) From the date a potential license applicant files a notification of intent to seek a license pursuant to §5.5 until any related license application proceeding is terminated by the Commission, the potential license applicant must make reasonably available to the public for inspection at its principal place of business or another location that is more accessible to the public, the preapplication document and any materials referenced therein. These materials must be available for inspection during regular business hours in a form that is readily accessible, reviewable, and reproducible.

- (2) The materials specified in paragraph (a)(1) of this section must be made available to the requester at the location specified in paragraph (a)(1) of this section or through the mail, or otherwise. Except as provided in paragraph (a)(3) of this section, copies of the pre-application document and any materials referenced therein must be made available at their reasonable cost of reproduction plus, if applicable, postage.
- (3) A potential licensee must make requested copies of the materials specified in paragraph (a)(1) of this section available to the United States Fish and Wildlife Service, the National Marine Fisheries Service, the state agency responsible for fish and wildlife resources, any affected Federal land managing agencies, and Indian tribes without charge for the costs of reproduction or postage.
- (b) License application. (1) From the date on which a license application is filed under this part until the licensing proceeding for the project is terminated by the Commission, the license applicant must make reasonably available to the public for inspection at its principal place of business or another location that is more accessible to the public, a copy of the complete application for license, together with all exhibits, appendices, and any amendments, pleadings, supplementary or additional information, or correspondence filed by the applicant with the Commission in connection with the application. These materials must be available for inspection during regular business hours in a form that is readily

accessible, reviewable, and reproducible at the same time as the information is filed with the Commission or required by regulation to be made available.

- (2) The applicant must provide a copy of the complete application (as amended) to a public library or other convenient public office located in each county in which the proposed project is located.
- (3) The materials specified in paragraph (b)(1) of this section must be made available to the requester at the location specified in paragraph (b)(1) of this section or through the mail. Except as provided in paragraph (b)(4) of this section, copies of the license application and any materials referenced therein must be made available at their reasonable cost of reproduction plus, if applicable, postage.
- (4) A licensee applicant must make requested copies of the materials specified in paragraph (b)(1) of this section available to the United States Fish and Wildlife Service, the National Marine Fisheries Service, and the state agency responsible for fish and wildlife resources, any affected Federal land managing agencies, and Indian tribes without charge for the costs of reproduction or postage.
- (c) Confidentiality of cultural information. A potential applicant must delete from any information made available to the public under paragraphs (a) and (b) of this section, specific site or property locations the disclosure of which would create a risk of harm, theft, or destruction of archeological or native American cultural resources or of the site at which the sources are located, or would violate any Federal law, include the Archeological Resources Protection Act of 1979, 16 U.S.C. 470w-3, and the National Historic Preservation Act of 1966, 16 U.S.C. 470hh.
- (d) Access. Anyone may file a petition with the Commission requesting access to the information specified in paragraphs (a) or (b) of this section if it believes that the potential applicant or applicant is not making the information reasonably available for public inspection or reproduction. The petition must describe in detail the basis for the petitioner's belief.

§5.3 Process selection.

- (a)(1) Notwithstanding any other provision of this part or of parts 4 and 16 of this chapter, a potential applicant for a new, subsequent, or original license may until July 23, 2005 elect to use the licensing procedures of this part or the licensing procedures of parts 4 and 16.
- (2) Any potential license applicant that files its notification of intent pursuant to §5.5 and pre-application document pursuant to §5.6 after July 23, 2005 must request authorization to use the licensing procedures of parts 4 and 16, as provided for in paragraphs (b)–(f) of this section.
- (b) A potential license applicant may file with the Commission a request to use the traditional licensing process or alternative procedures pursuant to this Section with its notification of intent pursuant to §5.5.
- (c)(1)(i) An application for authorization to use the traditional process must include justification for the request and any existing written comments on the potential applicant's proposal and a response thereto.
- (ii) A potential applicant requesting authorization to use the traditional process should address the following considerations:
- (A) Likelihood of timely license issuance:
 - (B) Complexity of the resource issues;
 - (C) Level of anticipated controversy;
- (D) Relative cost of the traditional process compared to the integrated process;
- (E) The amount of available information and potential for significant disputes over studies; and
- (F) Other factors believed by the applicant to be pertinent
- (2) A potential applicant requesting the use of §4.34(i) alternative procedures of this chapter must:
- (i) Demonstrate that a reasonable effort has been made to contact all agencies, Indian tribes, and others affected by the applicant's request, and that a consensus exists that the use of alternative procedures is appropriate under the circumstances;
- (ii) Submit a communications protocol, supported by interested entities, governing how the applicant and other

participants in the pre-filing consultation process, including the Commission staff, may communicate with each other regarding the merits of the potential applicant's proposal and proposals and recommendations of interested entities; and

- (iii) Provide a copy of the request to all affected resource agencies and Indian tribes and to all entities contacted by the applicant that have expressed an interest in the alternative pre-filing consultation process.
- (d)(1) The potential applicant must provide a copy of the request to use the traditional process or alternative procedures to all affected resource agencies, Indian tribes, and members of the public likely to be interested in the proceeding. The request must state that comments on the request to use the traditional process or alternative procedures, as applicable, must be filed with the Commission within 30 days of the filing date of the request and, if there is no project number, that responses must reference the potential applicant's name and address.
- (2) The potential applicant must also publish notice of the filing of its notification of intent, of the pre-application document, and of any request to use the traditional process or alternative procedures no later than the filing date of the notification of intent in a daily or weekly newspaper of general circulation in each county in which the project is located. The notice must:
- (i) Disclose the filing date of the request to use the traditional process or alternative procedures, and the notification of intent and pre-application document:
- (ii) Briefly summarize these documents and the basis for the request to use the traditional process or alternative procedures;
- (iii) Include the potential applicant's name and address, and telephone number, the type of facility proposed to be applied for, its proposed location, the places where the pre-application document is available for inspection and reproduction;
- (iv) Include a statement that comments on the request to use the traditional process or alternative procedures are due to the Commission and the potential applicant no later than 30

days following the filing date of that document and, if there is no project number, that responses must reference the potential applicant's name and address:

- (v) State that comments on any request to use the traditional process should address, as appropriate to the circumstances of the request, the:
- (A) Likelihood of timely license issuance:
 - (B) Complexity of the resource issues;
 - (C) Level of anticipated controversy;
- (D) Relative cost of the traditional process compared to the integrated process; and
- (E) The amount of available information and potential for significant disputes over studies; and
- (F) Other factors believed by the commenter to be pertinent; and
- (vi) State that respondents must submit comments to the Secretary of the Commission in accordance with filing procedures posted on the Commission's Web site at http://www.ferc.gov.

[Order 2002, 68 FR 51121, Aug. 25, 2003; 68 FR 61742, Oct. 30, 2003, as amended by Order 737, 75 FR 43402, July 26, 2010]

§ 5.4 Acceleration of a license expiration date.

- (a) Request for acceleration. (1) No later than five and one-half years prior to expiration of an existing license, a licensee may file with the Commission, in accordance with the formal filing requirements in subpart T of part 385 of this chapter, a written request for acceleration of the expiration date of its existing license, containing the statements and information specified in §16.6(b) of this chapter and a detailed explanation of the basis for the acceleration request.
- (2) If the Commission grants the request for acceleration pursuant to paragraph (c) of this section, the Commission will deem the request for acceleration to be a notice of intent under §16.6 of this chapter and, unless the Commission directs otherwise, the licensee must make available the Pre-Application Document provided for in §5.6 no later than 90 days from the date that the Commission grants the request for acceleration.

- (b) Notice of request for acceleration. (1) Upon receipt of a request for acceleration, the Commission will give notice of the licensee's request and provide a 45-day period for comments by interested persons by:
- (i) Publishing notice in the FEDERAL REGISTER:
- (ii) Publishing notice once in a daily or weekly newspaper published in the county or counties in which the project or any part thereof or the lands affected thereby are situated; and
- (iii) Notifying appropriate Federal, state, and interstate resource agencies and Indian tribes, and non-governmental organizations likely to be interested, by electronic means if practical, otherwise by mail.
- (2) The notice issued pursuant to paragraphs (b)(1)(A) and (B) and the written notice given pursuant to paragraph (b)(1)(C) will be considered as fulfilling the notice provisions of §16.6(d) of this chapter should the Commission grant the acceleration request and will include an explanation of the basis for the licensee's acceleration request.
- (c) Commission order. If the Commission determines it is in the public interest, the Commission will issue an order accelerating the expiration date of the license to not less than five years and 90 days from the date of the Commission order.

[Order 2002, 68 FR 51121, Aug. 25, 2003, as amended by Order 653, 70 FR 8724, Feb. 23, 2005]

§ 5.5 Notification of intent.

- (a) Notification of intent. A potential applicant for an original, new, or subsequent license, must file a notification of its intent to do so in the manner provided for in paragraphs (b) and (c) of this section.
- (b) Requirement to notify. In order for a non-licensee to notify the Commission that it intends to file an application for an original, new, or subsequent license, or for an existing licensee to notify the Commission whether or not it intends to file an application for a new or subsequent license, a potential license applicant must file with the Secretary of the Commission in accordance with filing procedures posted on the Commission's Web site at http://

www.ferc.gov, a letter that contains the following information:

- (1) The potential applicant or existing licensee's name and address.
- (2) The project number, if any.
- (3) The license expiration date, if any.
- (4) An unequivocal statement of the potential applicant's intention to file an application for an original license, or, in the case of an existing licensee, to file or not to file an application for a new or subsequent license.
- (5) The type of principal project works licensed, if any, such as dam and reservoir, powerhouse, or transmission lines.
- (6) The location of the project by state, county, and stream, and, when appropriate, by city or nearby city.
- (7) The installed plant capacity, if anv.
- (8) The names and mailing addresses of:
- (i) Every county in which any part of the project is located, and in which any Federal facility that is used or to be used by the project is located;
- (ii) Every city, town, or similar political subdivision;
- (A) In which any part of the project is or is to be located and any Federal facility that is or is to be used by the project is located, or
- (B) That has a population of 5,000 or more people and is located within 15 miles of the existing or proposed project dam;
- (iii) Every irrigation district, drainage district, or similar special purpose political subdivision:
- (A) In which any part of the project is or is proposed to be located and any Federal facility that is or is proposed to be used by the project is located; or
- (B) That owns, operates, maintains, or uses any project facility or any Federal facility that is or is proposed to be used by the project;
- (iv) Every other political subdivision in the general area of the project or proposed project that there is reason to believe would be likely to be interested in, or affected by, the notification; and
 - (v) Affected Indian tribes.
- (c) Requirement to distribute. Before it files any application for an original, new, or subsequent license, a potential

§ 5.6

license applicant proposing to file a license application pursuant to this part or to request to file a license application pursuant to part 4 of this chapter and, as appropriate, part 16 of this chapter (i.e., the "traditional process"), including an application pursuant to §4.34(i) alternative procedures of this chapter must distribute to appropriate Federal, state, and interstate resource agencies, Indian tribes, local governments, and members of the public likely to be interested in the proceeding the notification of intent provided for in paragraph (a) of this section.

- (d) When to notify. An existing licensee or non-licensee potential applicant must notify the Commission as required in paragraph (b) of this section at least five years, but not more than five and one-half years, before the existing license expires.
- (e) Non-Federal representatives. A potential license applicant may at the same time it files its notification of intent and distributes its pre-application document, request to be designated as the Commission's non-Federal representative for purposes of consultation under section 7 of the Endangered Species Act and the joint agency regulations thereunder at 50 CFR part 402, Section 305(b) of the Magnuson-Stevens Fishery Conservation and Management Act and the implementing regulations at 50 CFR 600.920. A potential license applicant may at the same time request authorization to initiate consultation under section 106 of the National Historic Preservation Act and the implementing regulations at 36 CFR 800.2(c)(4).
- (f) Procedural matters. The provisions of subpart F of part 16 of this chapter apply to projects to which this part applies.
- (g) Construction of regulations. The provisions of this part and parts 4 and 16 shall be construed in a manner that best implements the purposes of each part and gives full effect to applicable provisions of the Federal Power Act.

[Order 2002, 68 FR 51121, Aug. 25, 2003; 68 FR 69957, Dec. 16, 2003, as amended by Order 737, 75 FR 43402, July 26, 2010]

§ 5.6 Pre-application document.

- (a) Pre-application document. (1) Simultaneously with the filing of its notification of intent to seek a license as provided for in §5.5, and before it files any application for an original, new, or subsequent license, a potential applicant for a license to be filed pursuant to this part or part 4 of this chapter and, as appropriate, part 16 of this chapter, must file with the Secretary of the Commission in accordance with filing procedures posted on the Commission's Web site a.t. www.ferc.gov and distribute to the appropriate Federal, state, and interstate resource agencies, Indian tribes, local governments, and members of the public likely to be interested in the proceeding, the pre-application document provided for in this section.
- (2) The agencies referred to in paragraph (a)(1) of this section include: Any state agency with responsibility for fish, wildlife, and botanical resources, water quality, coastal zone management plan consistency certification, shoreline management, and water resources; the U.S. Fish and Wildlife Service; the National Marine Fisheries Service: Environmental Protection State Historic Preservation Agency; Tribal Historic Preservation Officer: Officer; National Park Service; local, state, and regional recreation agencies and planning commissions; local and state zoning agencies; and any other state or Federal agency or Indian tribe with managerial authority over any part of project lands and waters.
- (b) Purpose of pre-application document. (1) The pre-application document provides the Commission and the entities identified in paragraph (a) of this section with existing information relevant to the project proposal that is in the potential applicant's possession or that the potential applicant can obtain with the exercise of due diligence. This existing, relevant, and reasonably available information is distributed to these entities to enable them to identify issues and related information needs, develop study requests and study plans, and prepare documents analyzing any license application that may be filed. It is also a precursor to the environmental analysis section of the Preliminary Licensing Proposal or

draft license application provided for in §5.16, Exhibit E of the final license application, and the Commission's scoping document(s) and environmental impact statement or environmental assessment under the National Environmental Policy Act (NEPA).

- (2) A potential applicant is not required to conduct studies in order to generate information for inclusion in the pre-application document. Rather, a potential applicant must exercise due diligence in determining what information exists that is relevant to describing the existing environment and potential impacts of the project proposal (including cumulative impacts), obtaining that information if the potential applicant does not already possess it, and describing or summarizing it as provided for in paragraph (d) of this section. Due diligence includes, but is not limited to, contacting appropriate agencies and Indian tribes that may have relevant information and review of Federal and state comprehensive plans filed with the Commission and listed on the Commission's Web site at http://www.ferc.gov.
- (c) Form and distribution protocol—(1) General requirements. As specifically provided for in the content requirements of paragraph (d) of this section, the pre-application document must describe the existing and proposed (if any) project facilities and operations, provide information on the existing environment, and existing data or studies relevant to the existing environment, and any known and potential impacts of the proposed project on the specified resources.
- (2) Availability of source information and studies. The sources of information on the existing environment and known or potential resource impacts included in the descriptions and summaries must be referenced in the relevant section of the document, and in an appendix to the document. The information must be provided upon request to recipients of the pre-application document. A potential applicant must provide the requested information within 20 days from receipt of the request. Potential applicants and requesters are strongly encouraged to use electronic means or compacts disks to distribute studies and other forms of

information, but a potential applicant must, upon request, provide the information in hard copy form. The potential applicant is also strongly encouraged to include with the pre-application document any written protocol for distribution consistent with this paragraph to which it has agreed with agencies, Indian tribes, or other entities.

- (d) Content requirements—(1) Process plan and schedule. The pre-application document must include a plan and schedule for all pre-application activity that incorporates the time frames for pre-filing consultation, information gathering, and studies set forth in this part. The plan and schedule must include a proposed location and date for the scoping meeting and site visit required by §5.8(b)(3)(viii).
- (2) Project location, facilities, and operations. The potential applicant must include in the pre-application document:
- (i) The exact name and business address, and telephone number of each person authorized to act as agent for the applicant;
- (ii) Detailed maps showing lands and waters within the project boundary by township, range, and section, as well as by state, county, river, river mile, and closest town, and also showing the specific location of any Federal and tribal lands, and the location of proposed project facilities, including roads, transmission lines, and any other appurtenant facilities;
- (iii) A detailed description of all existing and proposed project facilities and components, including:
- (A) The physical composition, dimensions, and general configuration of any dams, spillways, penstocks, canals, powerhouses, tailraces, and other structures proposed to be included as part of the project or connected directly to it;
- (B) The normal maximum water surface area and normal maximum water surface elevation (mean sea level), gross storage capacity of any impoundments:
- (C) The number, type, and minimum and maximum hydraulic capacity and installed (rated) capacity of any proposed turbines or generators to be included as part of the project;

- (D) The number, length, voltage, and interconnections of any primary transmission lines proposed to be included as part of the project, including a single-line diagram showing the transfer of electricity from the project to the transmission grid or point of use; and
- (E) An estimate of the dependable capacity, average annual, and average monthly energy production in kilowatt hours (or mechanical equivalent);
- (iv) A description of the current (if applicable) and proposed operation of the project, including any daily or seasonal ramping rates, flushing flows, reservoir operations, and flood control operations.
- (v) In the case of an existing licensed project;
- (A) A complete description of the current license requirements; *i.e.*, the requirements of the original license as amended during the license term;
- (B) A summary of project generation and outflow records for the five years preceding filing of the pre-application document;
 - (C) Current net investment; and
- (D) A summary of the compliance history of the project, if applicable, including a description of any recurring situations of non-compliance.
- (vi) A description of any new facilities or components to be constructed, plans for future development or rehabilitation of the project, and changes in project operation.
- (3) Description of existing environment and resource impacts—(i) General requirements. A potential applicant must, based on the existing, relevant, and reasonably available information, include a discussion with respect to each resource that includes:
- (A) A description of the existing environment as required by paragraphs (d)(3)(ii)-(xiii) of this section;
- (B) Summaries (with references to sources of information or studies) of existing data or studies regarding the resource;
- (C) A description of any known or potential adverse impacts and issues associated with the construction, operation or maintenance of the proposed project, including continuing and cumulative impacts; and
- (D) A description of any existing or proposed project facilities or oper-

- ations, and management activities undertaken for the purpose of protecting, mitigating impacts to, or enhancing resources affected by the project, including a statement of whether such measures are required by the project license, or were undertaken for other reasons. The type and amount of the information included in the discussion must be commensurate with the scope and level of resource impacts caused or potentially caused by the proposed project. Potential license applicants are encouraged to provide photographs or other visual aids, as appropriate, to supplement text, charts, and graphs included in the discussion.
- (ii) Geology and soils. Descriptions and maps showing the existing geology, topography, and soils of the proposed project and surrounding area. Components of the description must include:
- (A) A description of geological features, including bedrock lithology, stratigraphy, structural features, glacial features, unconsolidated deposits, and mineral resources at the project site:
- (B) A description of the soils, including the types, occurrence, physical and chemical characteristics, erodability and potential for mass soil movement;
- (C) A description of reservoir shorelines and streambanks, including:
- (1) Steepness, composition (bedrock and unconsolidated deposits), and vegetative cover; and
- (2) Existing erosion, mass soil movement, slumping, or other forms of instability, including identification of project facilities or operations that are known to or may cause these conditions
- (iii) Water resources. A description of the water resources of the proposed project and surrounding area. This must address the quantity and quality (chemical/physical parameters) of all waters affected by the project, including but not limited to the project reservoir(s) and tributaries thereto, bypassed reach, and tailrace. Components of the description must include:
 - (A) Drainage area;
- (B) The monthly minimum, mean, and maximum recorded flows in cubic feet per second of the stream or other body of water at the powerplant intake or point of diversion, specifying any

adjustments made for evaporation, leakage, minimum flow releases, or other reductions in available flow;

- (C) A monthly flow duration curve indicating the period of record and the location of gauging station(s), including identification number(s), used in deriving the curve; and a specification of the critical streamflow used to determine the project's dependable capacity:
- (D) Existing and proposed uses of project waters for irrigation, domestic water supply, industrial and other purposes, including any upstream or downstream requirements or constraints to accommodate those purposes:
- (E) Existing instream flow uses of streams in the project area that would be affected by project construction and operation; information on existing water rights and water rights applications potentially affecting or affected by the project:
- (F) Any federally-approved water quality standards applicable to project waters:
- (G) Seasonal variation of existing water quality data for any stream, lake, or reservoir that would be affected by the proposed project, including information on:
- (1) Water temperature and dissolved oxygen, including seasonal vertical profiles in the reservoir;
- (2) Other physical and chemical parameters to include, as appropriate for the project; total dissolved gas, pH, total hardness, specific conductance, cholorphyll a, suspended sediment concentrations, total nitrogen (mg/L as N), total phosphorus (mg/L as P), and fecal coliform (E. Coli) concentrations;
- (H) The following data with respect to any existing or proposed lake or reservoir associated with the proposed project; surface area, volume, maximum depth, mean depth, flushing rate, shoreline length, substrate composition; and
- (I) Gradient for downstream reaches directly affected by the proposed project.
- (iv) Fish and aquatic resources. A description of the fish and other aquatic resources, including invasive species, in the project vicinity. This section must discuss the existing fish and macroinvertebrate communities, in-

- cluding the presence or absence of anadromous, catadromous, or migratory fish, and any known or potential upstream or downstream impacts of the project on the aquatic community. Components of the description must include:
- (A) Identification of existing fish and aquatic communities;
- (B) Identification of any essential fish habitat as defined under the Magnuson-Stevens Fishery Conservation and Management Act and established by the National Marine Fisheries Service; and
- (C) Temporal and spacial distribution of fish and aquatic communities and any associated trends with respect to:
- (1) Species and life stage composition;
- (2) Standing crop;
- (3) Age and growth data;
- (4) Spawning run timing; and
- (5) The extent and location of spawning, rearing, feeding, and wintering habitat.
- (v) Wildlife and botanical resources. A description of the wildlife and botanical resources, including invasive species, in the project vicinity. Components of this description must include:
- (A) Upland habitat(s) in the project vicinity, including the project's transmission line corridor or right-of-way and a listing of plant and animal species that use the habitat(s); and
- (B) Temporal or spacial distribution of species considered important because of their commercial, recreational, or cultural value.
- (vi) Wetlands, riparian, and littoral habitat. A description of the floodplain, wetlands, riparian habitats, and littoral in the project vicinity. Components of this description must include:
- (A) A list of plant and animal species, including invasive species, that use the wetland, littoral, and riparian habitat;
- (B) A map delineating the wetlands, riparian, and littoral habitat; and
- (C) Estimates of acreage for each type of wetland, riparian, or littoral habitat, including variability in such availability as a function of storage at a project that is not operated in run-of-river mode.
- (vii) Rare, threatened and endangered species. A description of any listed rare, threatened and endangered, candidate,

or special status species that may be present in the project vicinity. Components of this description must include:

- (A) A list of Federal- and state-listed, or proposed to be listed, threatened and endangered species known to be present in the project vicinity;
- (B) Identification of habitat requirements:
- (C) References to any known biological opinion, status reports, or recovery plan pertaining to a listed species;
- (D) Extent and location of any federally-designated critical habitat, or other habitat for listed species in the project area; and
- (E) Temporal and spatial distribution of the listed species within the project vicinity.
- (viii) Recreation and land use. A description of the existing recreational and land uses and opportunities within the project boundary. The components of this description include:
- (A) Text description illustrated by maps of existing recreational facilities, type of activity supported, location, capacity, ownership and management;
- (B) Current recreational use of project lands and waters compared to facility or resource capacity;
- (C) Existing shoreline buffer zones within the project boundary;
- (D) Current and future recreation needs identified in current State Comprehensive Outdoor Recreation Plans, other applicable plans on file with the Commission, or other relevant local, state, or regional conservation and recreation plans;
- (E) If the potential applicant is an existing licensee, its current shoreline management plan or policy, if any, with regard to permitting development of piers, boat docks and landings, bulkheads, and other shoreline facilities on project lands and waters;
- (F) A discussion of whether the project is located within or adjacent to
- (1) River segment that is designated as part of, or under study for inclusion in, the National Wild and Scenic River System; or
 - (2) State-protected river segment;
- (G) Whether any project lands are under study for inclusion in the National Trails System or designated as,

or under study for inclusion as, a Wilderness Area.

- (H) Any regionally or nationally important recreation areas in the project vicinity:
- (I) Non-recreational land use and management within the project boundary; and
- (J) Recreational and non-recreational land use and management adjacent to the project boundary.
- (ix) Aesthetic resources. A description of the visual characteristics of the lands and waters affected by the project. Components of this description include a description of the dam, natural water features, and other scenic attractions of the project and surrounding vicinity. Potential applicants are encouraged to supplement the text description with visual aids.
- (x) Cultural resources. A description of the known cultural or historical resources of the proposed project and surrounding area. Components of this description include:
- (A) Identification of any historic or archaeological site in the proposed project vicinity, with particular emphasis on sites or properties either listed in, or recommended by the State Historic Preservation Officer or Tribal Historic Preservation Officer for inclusion in, the National Register of Historic Places:
- (B) Existing discovery measures, such as surveys, inventories, and limited subsurface testing work, for the purpose of locating, identifying, and assessing the significance of historic and archaeological resources that have been undertaken within or adjacent to the project boundary; and
- (C) Identification of Indian tribes that may attach religious and cultural significance to historic properties within the project boundary or in the project vicinity; as well as available information on Indian traditional cultural and religious properties, whether on or off of any federally-recognized Indian reservation (A potential applicant must delete from any information made available under this section specific site or property locations, the disclosure of which would create a risk of harm, theft, or destruction of archaeological or Native American cultural

resources or to the site at which the resources are located, or would violate any Federal law, including the Archaeological Resources Protection Act of 1979, 16 U.S.C. 470w-3, and the National Historic Preservation Act of 1966, 16 U.S.C. 470hh).

- (xi) Socio-economic resources. A general description of socio-economic conditions in the vicinity of the project. Components of this description include general land use patterns (e.g., urban, agricultural, forested), population patterns, and sources of employment in the project vicinity.
- (xii) *Tribal resources*. A description of Indian tribes, tribal lands, and interests that may be affected by the project Components of this description include:
- (A) Identification of information on resources specified in paragraphs (d)(2)(ii)-(xi) of this section to the extent that existing project construction and operation affecting those resources may impact tribal cultural or economic interests, *e.g.*, impacts of project-induced soil erosion on tribal cultural sites; and
- (B) Identification of impacts on Indian tribes of existing project construction and operation that may affect tribal interests not necessarily associated with resources specified in paragraphs (d)(3)(ii)–(xi) of this Section, e.g., tribal fishing practices or agreements between the Indian tribe and other entities other than the potential applicant that have a connection to project construction and operation.
- (xiii) *River basin description*. A general description of the river basin or subbasin, as appropriate, in which the proposed project is located, including information on:
- (A) The area of the river basin or subbasin and length of stream reaches therein:
- (B) Major land and water uses in the project area;
- (C) All dams and diversion structures in the basin or sub-basin, regardless of function; and
- (D) Tributary rivers and streams, the resources of which are or may be affected by project operations;
- (4) Preliminary issues and studies list. Based on the resource description and impacts discussion required by para-

- graph (d)(3) of this section; the pre-application document must include with respect to each resource area identified above, a list of:
- (i) Issues pertaining to the identified resources:
- (ii) Potential studies or information gathering requirements associated with the identified issues;
- (iii) Relevant qualifying Federal and state or tribal comprehensive waterway plans; and
- (iv) Relevant resource management plans.
- (5) Summary of contacts. An appendix summarizing contacts with Federal, state, and interstate resource agencies, Indian tribes, non-governmental organizations, or other members of the public made in connection with preparing the pre-application document sufficient to enable the Commission to determine if due diligence has been exercised in obtaining relevant information.
- (e) If applicable, the applicant must also provide a statement of whether or not it will seek benefits under section 210 of the Public Utility Regulatory Policies Act of 1978 (PURPA) by satisfying the requirements for qualifying hydroelectric small power production facilities in §292.203 of this chapter. If benefits under section 210 of PURPA are sought, a statement of whether or not the applicant believes the project is located at a new dam or diversion (as that term is defined in §292.202(p) of this chapter), and a request for the agencies' view on that belief, if any.

[Order 2002, 68 FR 51121, Aug. 25, 2003; 68 FR 69957, Dec. 16, 2003, as amended by Order 737, 75 FR 43402, July 26, 2010]

§ 5.7 Tribal consultation.

A meeting shall be held no later than 30 days following filing of the notification of intent required by §5.5 between each Indian tribe likely to be affected by the potential license application and the Commission staff if the affected Indian tribe agrees to such meeting.

[Order 2002, 68 FR 51121, Aug. 25, 2003; 68 FR 61742, Oct. 30, 2003]

- § 5.8 Notice of commencement of proceeding and scoping document, or of approval to use traditional licensing process or alternative procedures.
- (a) Notice. Within 60 days of the notification of intent required under §5.5, filing of the pre-application document pursuant to §5.6, and filing of any request to use the traditional licensing process or alternative procedures, the Commission will issue a notice of commencement of proceeding and scoping document or of approval of a request to use the traditional licensing process or alternative procedures.
- (b) Notice contents. The notice shall include:
- (1) The decision of the Director of the Office of Energy Projects on any request to use the traditional licensing process or alternative procedures.
- (2) If appropriate, a request by the Commission to initiate informal consultation under section 7 of the Endangered Species Act and the joint agency regulations thereunder at 50 CFR part 402, section 305(b) of the Magnuson-Stevens Fishery Conservation and Management Act and implementing regulations at 50 CFR 600.920, or section 106 of the National Historic Preservation Act and implementing regulations at 36 CFR 800.2, and, if applicable, designation of the potential applicant as the Commission's non-federal representative.
- (3) If the potential license application is to be developed and filed pursuant to this part, notice of:
- (i) The applicant's intent to file a license application;
- (ii) The filing of the pre-application document:
- (iii) Commencement of the proceeding;
- (iv) A request for comments on the pre-application document (including the proposed process plan and schedule);
- (v) A statement that all communications to or from the Commission staff related to the merits of the potential application must be filed with the Commission;
- (vi) The request for other Federal or state agencies or Indian tribes to be cooperating agencies for purposes of developing an environmental document;

- (vii) The Commission's intent with respect to preparation of an environmental impact statement; and
- (viii) A public scoping meeting and site visit to be held within 30 days of the notice.
- (c) Scoping Document 1. At the same time the Commission issues the notice provided for in paragraph (a) of this Section, the Commission staff will issue Scoping Document 1. Scoping Document 1 will include:
- (1) An introductory section describing the purpose of the scoping document, the date and time of the scoping meeting, procedures for submitting written comments, and a request for information or study requests from state and Federal resource agencies, Indian tribes, non-governmental organizations, and individuals;
- (2) Identification of the proposed action, including a description of the project's location, facilities, and operation, and any proposed protection and enhancement measures, and other alternatives to the proposed action, including alternatives considered but eliminated from further study, and the no action alternative:
- (3) Identification of resource issues to be analyzed in the environmental document, including those that would be cumulatively affected along with a description of the geographic and temporal scope of the cumulatively affected resources;
- (4) A list of qualifying Federal and state comprehensive waterway plans;
- (5) A list of qualifying tribal comprehensive waterway plans;
- (6) A process plan and schedule and a draft outline of the environmental document; and
 - (7) A list of recipients.
- (d) Scoping meeting and site visit. The purpose of the public meeting and site visit is to:
- (1) Initiate issues scoping pursuant to the National Environmental Policy Act;
- (2) Review and discuss existing conditions and resource management objectives:
- (3) Review and discuss existing information and make preliminary identification of information and study needs:

- (4) Review, discuss, and finalize the process plan and schedule for pre-filing activity that incorporates the time periods provided for in this part and, to the extent reasonably possible, maximizes coordination of Federal, state, and tribal permitting and certification processes, including consultation under section 7 of the Endangered Species Act and water quality certification or waiver thereof under section 401 of the Clean Water Act; and
- (5) Discuss the appropriateness of any Federal or state agency or Indian tribe acting as a cooperating agency for development of an environmental document pursuant to the National Environmental Policy Act.
- (e) *Method of notice*. The public notice provided for in this section will be given by:
- (1) Publishing notice in the FEDERAL REGISTER:
- (2) Publishing notice in a daily or weekly newspaper published in the county or counties in which the project or any part thereof or the lands affected thereby are situated, and, as appropriate, tribal newspapers;
- (3) Notifying appropriate Federal, state, and interstate resource agencies, state water quality and coastal zone management plan consistency certification agencies, Indian tribes, and nongovernmental organizations, by electronic means if practical, otherwise by

[Order 2002, 68 FR 51121, Aug. 25, 2003, as amended by Order 653, 70 FR 8724, Feb. 23, 2005]

§ 5.9 Comments and information or study requests.

(a) Comments and study requests. Comments on the pre-application document and the Commission staff's Scoping Document 1 must be filed with the Commission within 60 days following the Commission's notice of consultation procedures issued pursuant to §5.8. Comments, including those by Commission staff, must be accompanied by any information gathering and study requests, and should include information and studies needed for consultation under section 7 of the Endangered Species Act and water quality certification under Section 401 of the Clean Water Act.

- (b) Content of study request. Any information or study request must:
- (1) Describe the goals and objectives of each study proposal and the information to be obtained;
- (2) If applicable, explain the relevant resource management goals of the agencies or Indian tribes with jurisdiction over the resource to be studied;
- (3) If the requester is not a resource agency, explain any relevant public interest considerations in regard to the proposed study;
- (4) Describe existing information concerning the subject of the study proposal, and the need for additional information:
- (5) Explain any nexus between project operations and effects (direct, indirect, and/or cumulative) on the resource to be studied, and how the study results would inform the development of license requirements;
- (6) Explain how any proposed study methodology (including any preferred data collection and analysis techniques, or objectively quantified information, and a schedule including appropriate field season(s) and the duration) is consistent with generally accepted practice in the scientific community or, as appropriate, considers relevant tribal values and knowledge; and
- (7) Describe considerations of level of effort and cost, as applicable, and why any proposed alternative studies would not be sufficient to meet the stated information needs.
- (c) Applicant seeking PURPA benefits; estimate of fees. If a potential applicant has stated that it intends to seek PURPA benefits, comments on the preapplication document by a fish and wildlife agency must provide the potential applicant with a reasonable estimate of the total costs the agency anticipates it will incur in order to set mandatory terms and conditions for the proposed project. An agency may provide a potential applicant with an updated estimate as it deems necessary. If any agency believes that its most recent estimate will be exceeded by more than 25 percent, it must supply the potential applicant with a new

estimate and submit a copy to the Commission.

[Order 2002, 68 FR 51121, Aug. 25, 2003; 68 FR 61742, Oct. 30, 2003; 68 FR 69957, Dec. 16, 2003, as amended by Order 699, 72 FR 45324, Aug. 14, 20071

§ 5.10 Scoping Document 2.

Within 45 days following the deadline for filing of comments on Scoping Document 1, the Commission staff shall, if necessary, issue Scoping Document 2.

§5.11 Potential Applicant's proposed study plan and study plan meetings.

- (a) Within 45 days following the deadline for filing of comments on the preapplication document, including information and study requests, the potential applicant must file with the Commission a proposed study plan.
- (b) The potential applicant's proposed study plan must include with respect to each proposed study:
- (1) A detailed description of the study and the methodology to be used;
- (2) A schedule for conducting the study;
- (3) Provisions for periodic progress reports, including the manner and extent to which information will be shared; and sufficient time for technical review of the analysis and results; and
- (4) If the potential applicant does not adopt a requested study, an explanation of why the request was not adopted, with reference to the criteria set forth in §5.9(b).
- (c) The potential applicant's proposed study plan must also include provisions for the initial and updated study reports and meetings provided for in §5.15.
- (d) The applicant's proposed study plan must:
- (1) Describe the goals and objectives of each study proposal and the information to be obtained;
- (2) Address any known resource management goals of the agencies or Indian tribes with jurisdiction over the resource to be studied;
- (3) Describe existing information concerning the subject of the study proposal, and the need for additional information:
- (4) Explain any nexus between project operations and effects (direct,

indirect, and/or cumulative) on the resource to be studied:

- (5) Explain how any proposed study methodology (including any preferred data collection and analysis techniques, or objectively quantified information, and a schedule including appropriate field season(s) and the duration) is consistent with generally accepted practice in the scientific community or, as appropriate, considers any known tribal interests;
- (6) Describe considerations of level of effort and cost, as applicable.
- (e) The potential applicant's proposed study plan must be accompanied by a proposal for conducting a study plan meeting or meetings during the 90-day period provided for in §5.12 for the purpose of clarifying the potential applicant's proposed study plan and any initial information gathering or study requests, and to resolve any outstanding issues with respect to the proposed study plan. The initial study plan meeting must be held no later than 30 days after the deadline date for posed study plan.

§ 5.12 Comments on proposed study plan.

Comments on the potential applicant's proposed study plan, including any revised information or study requests, must be filed within 90 days after the proposed study plan is filed. This filing must also include an explanation of any study plan concerns and any accommodations reached with the potential applicant regarding those concerns. Any proposed modifications to the potential applicant's proposed study plan must address the criteria in §5.9(b).

§ 5.13 Revised study plan and study plan determination.

(a) Within 30 days following the deadline for filing comments on the potential applicant's proposed study plan, as provided for in §5.12, the potential applicant must file a revised study plan for Commission approval. The revised study plan shall include the comments on the proposed study plan and a description of the efforts made to resolve differences over study requests. If the

potential applicant does not adopt a requested study, it must explain why the request was not adopted, with reference to the criteria set forth in \$5.9(b).

- (b) Within 15 days following filing of the potential applicant's revised study plan, participants may file comments thereon.
- (c) Within 30 days following the date the potential applicant files its revised study plan, the Director of Energy Projects will issue a Study Plan Determination with regard to the potential applicant's study plan, including any modifications determined to be necessary in light of the record.
- (d) If no notice of study dispute is filed pursuant to §5.14 within 20 days of the Study Plan Determination, the study plan as approved in the Study Plan Determination shall be deemed to be approved and the potential applicant shall proceed with the approved studies. If a potential applicant fails to obtain or conduct a study as required by Study Plan Determination, its license application may be considered deficient.

§ 5.14 Formal study dispute resolution process.

- (a) Within 20 days of the Study Plan Determination, any Federal agency with authority to provide mandatory conditions on a license pursuant to FPA Section 4(e), 16 U.S.C. 797(e), or to prescribe fishways pursuant to FPA Section 18, 16 U.S.C. 811, or any agency or Indian tribe with authority to issue a water quality certification for the project license under section 401 of the Clean Water Act, 42 U.S.C. 1341, may file a notice of study dispute with respect to studies pertaining directly to the exercise of their authorities under sections 4(e) and 18 of the Federal Power Act or section 401 of the Clean Water Act.
- (b) The notice of study dispute must explain how the disputing agency's or Indian tribe's study request satisfies the criteria set forth in §5.9(b), and shall identify and provide contact information for the panel member designated by the disputing agency or Indian tribe, as discussed in paragraph (d) of this section.

- (c) Studies and portions of study plans approved in the Study Plan Determination that are not the subject of a notice of dispute shall be deemed to be approved, and the potential applicant shall proceed with those studies or portions thereof.
- (d) Within 20 days of a notice of study dispute, the Commission will convene one or more three-person Dispute Resolution Panels, as appropriate to the circumstances of each proceeding. Each such panel will consist of:
- (1) A person from the Commission staff who is not otherwise involved in the proceeding, and who shall serve as the panel chair:
- (2) One person designated by the Federal or state agency or Indian tribe that filed the notice of dispute who is not otherwise involved in the proceeding; and
- (3) A third person selected by the other two panelists from a pre-established list of persons with expertise in the resource area. The two panelists shall make every reasonable effort to select the third panel member. If however no third panel member has been selected by the other two panelists within 15 days, an appropriate third panel member will be selected at random from the list of technical experts maintained by the Commission.
- (e) If more than one agency or Indian tribe files a notice of dispute with respect to the decision in the preliminary determination on any informationgathering or study request, the disputing agencies or Indian tribes must select one person to represent their interests on the panel.
- (f) The list of persons available to serve as a third panel member will be posted, as revised from time-to-time, on the hydroelectric page of the Commission's Web site. A person on the list who is requested and willing to serve with respect to a specific dispute will be required to file with the Commission at that time a current statement of their qualifications, a statement that they have had no prior involvement with the proceeding in which the dispute has arisen, or other financial or other conflict of interest.
- (g) All costs of the panel members representing the Commission staff and the agency or Indian tribe which filed

the notice of dispute will be borne by the Commission or the agency or Indian tribe, as applicable. The third panel member will serve without compensation, except for certain allowable travel expenses as defined in 31 CFR part 301.

- (h) To facilitate the delivery of information to the dispute resolution panel, the identity of the panel members and their addresses for personal service with respect to a specific dispute resolution will be posted on the hydroelectric page of the Commission's Web site.
- (i) No later than 25 days following the notice of study dispute, the potential applicant may file with the Commission and serve upon the panel members comments and information regarding the dispute.
- (j) Prior to engaging in deliberative meetings, the panel shall hold a technical conference for the purpose of clarifying the matters in dispute with reference to the study criteria. The technical conference shall be chaired by the Commission staff member of the panel. It shall be open to all participants, and the panel shall receive information from the participants as it deems appropriate.
- (k) No later than 50 days following the notice of study dispute, the panel shall make and deliver to the Director of the Office of Energy Projects a finding, with respect to each information or study request in dispute, concerning the extent to which each criteria set forth in §5.9(b) is met or not met, and why, and make recommendations regarding the disputed study request based on its findings. The panel's findings and recommendations must be based on the record in the proceeding. The panel shall file with its findings and recommendations all of the materials received by the panel. Any recommendation for the potential applicant to provide information or a study must include the technical specifications, including data acquisition techniques and methodologies.
- (1) No later than 70 days from the date of filing of the notice of study dispute, the Director of the Office of Energy Projects will review and consider the recommendations of the panel, and will issue a written determination. The

Director's determination will be made with reference to the study criteria set forth in §5.9(b) and any applicable law or Commission policies and practices, will take into account the technical expertise of the panel, and will explain why any panel recommendation was rejected, if applicable. The Director's determination shall constitute an amendment to the approved study plan.

§ 5.15 Conduct of studies.

- (a) Implementation. The potential applicant must gather information and conduct studies as provided for in the approved study plan and schedule.
- (b) Progress reports. The potential applicant must prepare and provide to the participants the progress reports provided for in §5.11(b)(3). Upon request of any participant, the potential applicant will provide documentation of study results.
- (c) Initial study report. (1) Pursuant to the Commission-approved study plan and schedule provided for in §5.13 or no later than one year after Commission approval of the study plan, whichever comes first, the potential applicant must prepare and file with the Commission an initial study report describing its overall progress in implementing the study plan and schedule and the data collected, including an explanation of any variance from the study plan and schedule. The report must also include any modifications to ongoing studies or new studies proposed by the potential applicant.
- (2) Within 15 days following the filing of the initial study report, the potential applicant shall hold a meeting with the participants and Commission staff to discuss the study results and the potential applicant's and or other participant's proposals, if any, to modify the study plan in light of the progress of the study plan and data collected.
- (3) Within 15 days following the meeting provided for in paragraph (c)(2) of this section, the potential applicant shall file a meeting summary, including any modifications to ongoing studies or new studies proposed by the potential applicant.

Federal Energy Regulatory Commission

- (4) Any participant or the Commission staff may file a disagreement concerning the applicant's meeting summary within 30 days, setting forth the basis for the disagreement. This filing must also include any modifications to ongoing studies or new studies proposed by the Commission staff or other participant.
- (5) Responses to any filings made pursuant to paragraph (c)(4) of this section must be filed within 30 days.
- (6) No later than 30 days following the due date for responses provided for in paragraph (c)(5) of this section, the Director will resolve the disagreement and amend the approved study plan as appropriate.
- (7) If no participant or the Commission staff files a disagreement concerning the potential applicant's meeting summary and request to amend the approved study plan within 30 days, any proposed amendment shall be deemed to be approved.
- (d) Criteria for modification of approved study. Any proposal to modify an ongoing study pursuant to paragraphs (c)(1)–(4) of this section must be accompanied by a showing of good cause why the proposal should be approved, and must include, as appropriate to the facts of the case, a demonstration that:
- (1) Approved studies were not conducted as provided for in the approved study plan; or
- (2) The study was conducted under anomalous environmental conditions or that environmental conditions have changed in a material way.
- (e) Criteria for new study. Any proposal for new information gathering or studies pursuant to paragraphs (c)(1)–(4) of this section must be accompanied by a showing of good cause why the proposal should be approved, and must include, as appropriate to the facts of the case, a statement explaining:
- (1) Any material changes in the law or regulations applicable to the information request;
- (2) Why the goals and objectives of any approved study could not be met with the approved study methodology;
- (3) Why the request was not made earlier;
- (4) Significant changes in the project proposal or that significant new infor-

- mation material to the study objectives has become available; and
- (5) Why the new study request satisfies the study criteria in §5.9(b).
- (f) Updated study report. Pursuant to the Commission-approved study plan and schedule provided for in §5.13, or no later than two years after Commission approval of the study plan and schedule, whichever comes first, the potential applicant shall prepare and file with the Commission an updated study report describing its overall progress in implementing the study plan and schedule and the data collected, including an explanation of any variance from the study plan and schedule. The report must also include any modifications to ongoing studies or new studies proposed by the potential applicant. The review, comment, and disagreement resolution provisions of paragraphs (c)(2)-(7) of this section shall apply to the updated study report. Any proposal to modify an ongoing study must be accompanied by a showing of good cause why the proposal should be approved as set forth in paragraph (d) of this section. Any proposal for new information gathering or studies is subject to paragraph (e) of this section except that the proponent must demonstrate extraordinary circumstances warranting approval. The applicant must promptly proceed to complete any remaining undisputed information-gathering or studies under its proposed amendments to the study plan, if any, and must proceed to complete any information-gathering or studies that are the subject of a disagreement upon the Director's resolution of the disagreement.

[Order 2002, 68 FR 51121, Aug. 25, 2003; 68 FR 61742, Oct. 30, 2003]

§5.16 Preliminary licensing proposal.

- (a) No later than 150 days prior to the deadline for filing a new or subsequent license application, if applicable, the potential applicant must file for comment a preliminary licensing proposal.
- (b) The preliminary licensing proposal must:
- (1) Clearly describe, as applicable, the existing and proposed project facilities, including project lands and waters;

- (2) Clearly describe, as applicable, the existing and proposed project operation and maintenance plan, to include measures for protection, mitigation, and enhancement measures with respect to each resource affected by the project proposal; and
- (3) Include the potential applicant's draft environmental analysis by resource area of the continuing and incremental impacts, if any, of its preliminary licensing proposal, including the results of its studies conducted under the approved study plan.
- (c) A potential applicant may elect to file a draft license application which includes the contents of a license application required by §5.18 instead of the Preliminary Licensing Proposal. A potential applicant that elects to file a draft license application must include notice of its intent to do so in the updated study report required by §5.15(f).
- (d) A potential applicant that has been designated as the Commission's non-Federal representative may include a draft Biological Assessment, draft Essential Fish Habitat Assessment, and draft Historic Properties Management Plan with its Preliminary Licensing Proposal or draft license application.
- (e) Within 90 days of the date the potential applicant files the Preliminary Licensing Proposal or draft license application, participants and the Commission staff may file comments on the Preliminary Licensing Proposal or draft application, which may include recommendations on whether the Commission should prepare an Environmental Assessment (with or without a draft Environmental Assessment) or an Environmental Impact Statement. Any participant whose comments request new information, studies, or other amendments to the approved study plan must include a demonstration of extraordinary circumstances, pursuant to the requirements of §5.15(f).
- (f) A waiver of the requirement to file the Preliminary Licensing Proposal or draft license application may be requested, based on a consensus of the participants in favor of such waiver.

§5.17 Filing of application.

(a) Deadline—new or subsequent license application. An application for a new or

- subsequent license must be filed no later than 24 months before the existing license expires.
- (b) Subsequent licenses. An applicant for a subsequent license must file its application under part I of the Federal Power Act. The provisions of section 7(a) of the Federal Power Act do not apply to licensing proceedings involving a subsequent license.
- (c) Rejection or dismissal of application. If the Commission rejects or dismisses an application for a new or subsequent license filed under this part pursuant to the provisions of §5.20, the application may not be refiled after the new or subsequent license application filing deadline specified in paragraph (a) of this section.
- (d)(1) Filing and service. Each applicant for a license under this part must submit the application to the Commission's Secretary for filing pursuant to the requirements of subpart T of part 385 of this chapter. The applicant must serve one copy of the application on the Director of the Commission's Regional Office for the appropriate region and on each resource agency, Indian tribe, or member of the public consulted pursuant to this part.
- (2) An applicant must publish notice twice of the filing of its application, no later than 14 days after the filing date in a daily or weekly newspaper of general circulation in each county in which the project is located. The notice must disclose the filing date of the application and briefly summarize it, including the applicant's name and address, the type of facility applied for, its proposed location, and the places where the information specified in §5.2(b) is available for inspection and reproduction. The applicant must promptly provide the Commission with proof of the publication of this notice.
- (e) PURPA benefits. (1) Every application for a license for a project with a capacity of 80 megawatts or less must include in its application copies of the statements made under §4.38(b)(1)(vi).
- (2) If an applicant reverses a statement of intent not to seek PURPA benefits:
- (i) Prior to the Commission issuing a license, the reversal of intent will be

treated as an amendment of the application under §4.35 of this chapter and the applicant must:

- (A) Repeat the pre-filing consultation process under this part; and
- (B) Satisfy all the requirements in §292.208 of this chapter; or
- (ii) After the Commission issues a license for the project, the applicant is prohibited from obtaining PURPA benefits.
- (f) Limitations on submitting applications. The provisions of §§4.33(b), (c), and (e) of this chapter apply to license applications filed under this Section.
- (g) Applicant notice. An applicant for a subsequent license that proposes to expand an existing project to encompass additional lands must include in its application a statement that the applicant has notified, by certified mail, property owners on the additional lands to be encompassed by the project and governmental agencies and subdivisions likely to be interested in or affected by the proposed expansion.

§ 5.18 Application content.

- (a) General content requirements. Each license application filed pursuant to this part must:
- (1) Identify every person, citizen, association of citizens, domestic corporation, municipality, or state that has or intends to obtain and will maintain any proprietary right necessary to construct, operate, or maintain the project;
- (2) Identify (providing names and mailing addresses):
- (i) Every county in which any part of the project, and any Federal facilities that would be used by the project, would be located;
- (ii) Every city, town, or similar local political subdivision:
- (A) In which any part of the project, and any Federal facilities that would be used by the project, would be located: or
- (B) That has a population of 5,000 or more people and is located within 15 miles of the project dam;
- (iii) Every irrigation district, drainage district, or similar special purpose political subdivision:
- (A) In which any part of the project, and any Federal facilities that would

be used by the project, would be located; or

- (B) That owns, operates, maintains, or uses any project facilities that would be used by the project:
- (iv) Every other political subdivision in the general area of the project that there is reason to believe would likely be interested in, or affected by, the application; and
- (v) All Indian tribes that may be affected by the project.
- (3)(i) For a license (other than a license under section 15 of the Federal Power Act) state that the applicant has made, either at the time of or before filing the application, a good faith effort to give notification by certified mail of the filing of the application to:
- (A) Every property owner of record of any interest in the property within the bounds of the project, or in the case of the project without a specific project boundary, each such owner of property which would underlie or be adjacent to any project works including any impoundments; and
- (B) The entities identified in paragraph (a)(2) of this section, as well as any other Federal, state, municipal or other local government agencies that there is reason to believe would likely be interested in or affected by such application.
- (ii) Such notification must contain the name, business address, and telephone number of the applicant and a copy of the Exhibit G contained in the application, and must state that a license application is being filed with the Commission.
- (4)(i) As to any facts alleged in the application or other materials filed, be subscribed and verified under oath in the form set forth in paragraph (a)(3)(B) of this Section by the person filing, an officer thereof, or other person having knowledge of the matters set forth. If the subscription and verification is by anyone other than the person filing or an officer thereof, it must include a statement of the reasons therefor.
- (ii) This application is executed in the:

State of				
County of				
By:				
Name)				

(Address) being duly sworn, depose(s) and say(s) that the contents of this application are true to the best of (his or her) knowledge or belief. The undersigned Applicant(s) has (have) signed the application this ___ day of _, 2_

(Applicant(s))

Bv:

Subscribed and sworn to before me, a [Notary Public, or title of other official authorized by the state to notarize documents, as appropriatel this dav 2

/SEAL [if anv]

(Notary Public, or other authorized official)

- (5) Contain the information and documents prescribed in the following Sections of this chapter, except as provided in paragraph (b) of this Section, according to the type of application:
- (i) License for a minor water power project and a major water power project 5 MW or less: §4.61 (General instructions, initial statement, and Exhibits A, F, and G);
- License major for unconstructed project and a major modified project: §4.41 of this chapter (General instructions, initial statement, Exhibits A, B, C, D, F, and G);
- (iii) License for a major project-existing dam: §4.51 of this chapter (General instructions, initial statement, Exhibits A, B, C, D, F, and G); or
- (iv) License for a project located at a new dam or diversion where the applicant seeks PURPA benefits: §292.208 of this chapter.
- (b) Exhibit E—Environmental Exhibit. The specifications for Exhibit E in §§ 4.41, 4.51, or 4.61 of this chapter shall not apply to applications filed under this part. The Exhibit E included in any license application filed under this part must address the resources listed in the Pre-Application Document provided for in §5.6; follow the Commission's "Preparing Environmental Assessments: Guidelines for Applicants, Contractors, and Staff," as they may be updated from time-to-time; and meet the following format and content requirements:
- (1) General description of the river basin. Describe the river system, including relevant tributaries; measurements of the area of the basin and length of stream; identify the

project's river mile designation or other reference point; describe the topography and climate; and discuss major land uses and economic activities.

- (2) Cumulative effects. List cumulatively affected resources based on the Commission's Scoping Document, consultation, and study results. Discuss the geographic and temporal scope of analysis for those resources. Describe how resources are cumulatively affected and explain the choice of the geographic scope of analysis. Include a brief discussion of past, present, and future actions, and their effects on resources based on the new license term (30-50 years). Highlight the effect on the cumulatively affected resources from reasonably foreseeable future actions. Discuss past actions' effects on the resource in the Affected Environment Section.
- (3) Applicable laws. Include a discussion of the status of compliance with or consultation under the following laws, if applicable:
- (i) Section 401 of the Clean Water Act. The applicant must file a request for a water quality certification (WQC), as required by Section 401 of the Clean Water Act no later than the deadline specified in §5.23(b). Potential applicants are encouraged to consult with the certifying agency or tribe concerning information requirements as early as possible.
- (ii) Endangered Species Act (ESA). Briefly describe the process used to address project effects on Federally listed or proposed species in the project vicinity. Summarize any anticipated environmental effects on these species and provide the status of the consultation process. If the applicant is the Commission's non-Federal designee for informal consultation under the ESA. the applicant's draft biological assessment must be included.
- (iii) Magnuson-Stevens Fishery Conservation and Management Act. Document from the National Marine Fisheries Service (NMFS) and/or the appropriate Regional Fishery Management Council any essential fish habitat (EFH) that may be affected by the project. Briefly discuss each managed species and life stage for which EFH

158

was designated. Include, as appropriate, the abundance, distribution, available habitat, and habitat use by the managed species. If the project may affect EFH, prepare a draft "EFH Assessment" of the impacts of the project. The draft EFH Assessment should contain the information outlined in 50 CFR 600.920(e).

(iv) Coastal Zone Management Act (CZMA). Section 307(c)(3) of the CZMA requires that all Federally licensed and permitted activities be consistent with approved state Coastal Zone Management Programs. If the project is located within a coastal zone boundary or if a project affects a resource located in the boundaries of the designated coastal zone, the applicant must certify that the project is consistent with the state Coastal Zone Management Program. If the project is within or affects a resource within the coastal zone, provide the date the applicant sent the consistency certification information to the state agency, the date the state agency received the certification, and the date and action taken by the state agency (for example, the agency will either agree or disagree with the consistency statement, waive it, or ask for additional information). Describe any conditions placed on the state agency's concurrence and assess the conditions in the appropriate section of the license application. If the project is not in or would not affect the coastal zone, state so and cite the coastal zone program office's concurrence.

(v) National Historic Preservation Act (NHPA). Section 106 of NHPA requires the Commission to take into account the effect of licensing a hydropower project on any historic properties, and allow the Advisory Council on Historic Preservation (Advisory Council) a reasonable opportunity to comment on the proposed action. "Historic Properties" are defined as any district, site, building, structure, or object that is included in or eligible for inclusion in the National Register of Historic Places (NRHP). If there would be an adverse effect on historic properties, the applicant may include a Historic Properties Management Plan (HPMP) to avoid or mitigate the effects. The applicant must include documentation of consultation with the Advisory Council, the State Historic Preservation Officer, Tribal Historic Preservation Officer, National Park Service, members of the public, and affected Indian tribes, where applicable.

(vi) Pacific Northwest Power Planning and Conservation Act (Act). If the project is not within the Columbia River Basin, this section shall not be included. The Columbia River Basin Fish and Wildlife Program (Program) developed under the Act directs agencies to consult with Federal and state fish and wildlife agencies, appropriate Indian tribes, and the Northwest Power Planning Council (Council) during the study, design, construction, and operation of any hydroelectric development in the basin. Section 12.1A of the Program outlines conditions that should be provided for in any original or new license. The program also designates certain river reaches as protected from development. The applicant must document consultation with the Council, describe how the act applies to the project, and how the proposal would or would not be consistent with the program.

(vii) Wild and Scenic Rivers and Wilderness Acts. Include a description of any areas within or in the vicinity of the proposed project boundary that are included in, or have been designated for study for inclusion in, the National Wild and Scenic Rivers System, or that have been designated as wilderness area, recommended for such designation, or designated as a wilderness study area under the Wilderness Act.

- (4) Project facilities and operation. Provide a description of the project to include:
- (i) Maps showing existing and proposed project facilities, lands, and waters within the project boundary;
- (ii) The configuration of any dams, spillways, penstocks, canals, powerhouses, tailraces, and other structures;
- (iii) The normal maximum water surface area and normal maximum water surface elevation (mean sea level), gross storage capacity of any impoundments:
- (iv) The number, type, and minimum and maximum hydraulic capacity and installed (rated) capacity of existing

and proposed turbines or generators to be included as part of the project;

- (v) An estimate of the dependable capacity, and average annual energy production in kilowatt hours (or mechanical equivalent);
- (vi) A description of the current (if applicable) and proposed operation of the project, including any daily or seasonal ramping rates, flushing flows, reservoir operations, and flood control operations.
- (5) Proposed action and action alternatives. (i) The environmental document must explain the effects of the applicant's proposal on resources. For each resource area addressed include:
- (A) A discussion of the affected environment;
- (B) A detailed analysis of the effects of the applicant's licensing proposal and, if reasonably possible, any preliminary terms and conditions filed with the Commission; and
 - (C) Any unavoidable adverse impacts.
 (ii) The environmental document
- must contain, with respect to the resources listed in the Pre-Application Document provided for in §5.6, and any other resources identified in the Commission's scoping document prepared pursuant to the National Environmental Policy Act and §5.8, the following information, commensurate with the scope of the project:
- (A) Affected environment. The applicant must provide a detailed description of the affected environment or area(s) to be affected by the proposed project by each resource area. This description must include the information on the affected environment filed in the Pre-Application Document provided for in §5.6, developed under the applicant's approved study plan, and otherwise developed or obtained by the applicant. This section must include a general description of socio-economic conditions in the vicinity of the project including general land use patterns (e.g., urban, agricultural, forested), population patterns, and sources of employment in the project vicinity.
- (B) Environmental analysis. The applicant must present the results of its studies conducted under the approved study plan by resource area and use the data generated by the studies to evaluate the beneficial and adverse environ-

- mental effects of its proposed project. This section must also include, if applicable, a description of any anticipated continuing environmental impacts of continued operation of the project, and the incremental impact of proposed new development of project works or changes in project operation. This analysis must be based on the information filed in the Pre-Application Document provided for in \$5.6, developed under the applicant's approved study plan, and other appropriate information, and otherwise developed or obtained by the Applicant.
- (C) Proposed environmental measures. The applicant must provide, by resource area, any proposed new environmental measures, including, but not limited to, changes in the project design or operations, to address the environmental effects identified above and its basis for proposing the measures. The applicant must describe how each proposed measure would protect or enhance the existing environment, including, where possible, a non-monetary quantification of the anticipated environmental benefits of the measure. This section must also include a statement of existing measures to be continued for the purpose of protecting and improving the environment and any proposed preliminary environmental measures received from the consulted resource agencies, Indian tribes, or the public. If an applicant does not adopt a preliminary environmental measure proposed by a resource agency, Indian tribe, or member of the public, it must include its reasons, based on projectspecific information.
- (D) Unavoidable adverse impacts. Based on the environmental analysis, discuss any adverse impacts that would occur despite the recommended environmental measures. Discuss whether any such impacts are short- or long-term, minor or major, cumulative or site-specific.
- (E) Economic analysis. The economic analysis must include annualized, current cost-based information. For a new or subsequent license, the applicant must include the cost of operating and maintaining the project under the existing license. For an original license, the applicant must estimate the cost of

constructing, operating, and maintaining the proposed project. For either type of license, the applicant should estimate the cost of each proposed resource protection, mitigation, or enhancement measure and any specific measure filed with the Commission by agencies, Indian tribes, or members of the public when the application is filed. For an existing license, the applicant's economic analysis must estimate the value of developmental resources associated with the project under the current license and the applicant's proposal. For an original license, the applicant must estimate the value of the developmental resources for the proposed project. As applicable, these developmental resources may include power generation, water supply, irrigation, navigation, and flood control. Where possible, the value of developmental resources must be based on market prices. If a protection, mitigation, or enhancement measure reduces the amount or value of the project's developmental resources, the applicant must estimate the reduction.

- (F) Consistency with comprehensive plans. Identify relevant comprehensive plans and explain how and why the proposed project would, would not, or should not comply with such plans and a description of any relevant resource agency or Indian tribe determination regarding the consistency of the project with any such comprehensive plan.
- (G) Consultation Documentation. Include a list containing the name, and address of every Federal, state, and interstate resource agency, Indian tribe, or member of the public with which the applicant consulted in preparation of the Environmental Document.
- (H) Literature cited. Cite all materials referenced including final study reports, journal articles, other books, agency plans, and local government plans.
- (2) The applicant must also provide in the Environmental Document:
- (A) Functional design drawings of any fish passage and collection facilities or any other facilities necessary for implementation of environmental measures, indicating whether the facilities depicted are existing or pro-

- posed (these drawings must conform to the specifications of §4.39 of this chapter regarding dimensions of full-sized prints, scale, and legibility);
- (B) A description of operation and maintenance procedures for any existing or proposed measures or facilities;
- (C) An implementation or construction schedule for any proposed measures or facilities, showing the intervals following issuance of a license when implementation of the measures or construction of the facilities would be commenced and completed;
- (D) An estimate of the costs of construction, operation, and maintenance, of any proposed facilities, and of implementation of any proposed environmental measures.
- (E) A map or drawing that conforms to the size, scale, and legibility requirements of §4.39 of this chapter showing by the use of shading, crosshatching, or other symbols the identity and location of any measures or facilities, and indicating whether each measure or facility is existing or proposed (the map or drawings in this exhibit may be consolidated).
- (c) Exhibit H. The information required to be provided by this paragraph (c) must be included in the application as a separate exhibit labeled "Exhibit H."
- (1) Information to be provided by an applicant for new license: Filing requirements—(i) Information to be supplied by all applicants. All Applicants for a new license under this part must file the following information with the Commission:
- (A) A discussion of the plans and ability of the applicant to operate and maintain the project in a manner most likely to provide efficient and reliable electric service, including efforts and plans to:
- (1) Increase capacity or generation at the project;
- (2) Coordinate the operation of the project with any upstream or downstream water resource projects; and
- (3) Coordinate the operation of the project with the applicant's or other electrical systems to minimize the cost of production.
- (B) A discussion of the need of the applicant over the short and long term

for the electricity generated by the project, including:

- (1) The reasonable costs and reasonable availability of alternative sources of power that would be needed by the applicant or its customers, including wholesale customers, if the applicant is not granted a license for the project;
- (2) A discussion of the increase in fuel, capital, and any other costs that would be incurred by the applicant or its customers to purchase or generate power necessary to replace the output of the licensed project, if the applicant is not granted a license for the project;
- (3) The effect of each alternative source of power on:
- (i) The applicant's customers, including wholesale customers;
- (ii) The applicant's operating and load characteristics; and
- (iii) The communities served or to be served, including any reallocation of costs associated with the transfer of a license from the existing licensee.
- (C) The following data showing need and the reasonable cost and availability of alternative sources of power:
- (1) The average annual cost of the power produced by the project, including the basis for that calculation;
- (2) The projected resources required by the applicant to meet the applicant's capacity and energy requirements over the short and long term including:
- (i) Energy and capacity resources, including the contributions from the applicant's generation, purchases, and load modification measures (such as conservation, if considered as a resource), as separate components of the total resources required;
- (ii) A resource analysis, including a statement of system reserve margins to be maintained for energy and capacity; and
- (iii) If load management measures are not viewed as resources, the effects of such measures on the projected capacity and energy requirements indicated separately;
- (iv) For alternative sources of power, including generation of additional power at existing facilities, restarting deactivated units, the purchase of power off-system, the construction or purchase and operation of a new power plant, and load management measures

- such as conservation: The total annual cost of each alternative source of power to replace project power; the basis for the determination of projected annual cost; and a discussion of the relative merits of each alternative, including the issues of the period of availability and dependability of purchased power, average life of alternatives, relative equivalent availability of generating alternatives, and relative impacts on the applicant's power system reliability and other system operating characteristics; and the effect on the direct providers (and their immediate customers) of alternate sources of power.
- (D) If an applicant uses power for its own industrial facility and related operations, the effect of obtaining or losing electricity from the project on the operation and efficiency of such facility or related operations, its workers, and the related community.
- (E) If an applicant is an Indian tribe applying for a license for a project located on the tribal reservation, a statement of the need of such Indian tribe for electricity generated by the project to foster the purposes of the reservation.
- (F) A comparison of the impact on the operations and planning of the applicant's transmission system of receiving or not receiving the project license, including:
- (1) An analysis of the effects of any resulting redistribution of power flows on line loading (with respect to applicable thermal, voltage, or stability limits), line losses, and necessary new construction of transmission facilities or upgrading of existing facilities, together with the cost impact of these effects:
- (2) An analysis of the advantages that the applicant's transmission system would provide in the distribution of the project's power; and
- (3) Detailed single-line diagrams, including existing system facilities identified by name and circuit number, that show system transmission elements in relation to the project and other principal interconnected system elements. Power flow and loss data that represent system operating conditions may be appended if applicants believe such data would be useful to show

that the operating impacts described would be beneficial.

- (G) If the applicant has plans to modify existing project facilities or operations, a statement of the need for, or usefulness of, the modifications, including at least a reconnaissance-level study of the effect and projected costs of the proposed plans and any alternate plans, which in conjunction with other developments in the area would conform with a comprehensive plan for improving or developing the waterway and for other beneficial public uses as defined in Section 10(a)(1) of the Federal Power Act.
- (H) If the applicant has no plans to modify existing project facilities or operations, at least a reconnaissance-level study to show that the project facilities or operations in conjunction with other developments in the area would conform with a comprehensive plan for improving or developing the waterway and for other beneficial public uses as defined in Section 10(a)(1) of the Federal Power Act.
- (I) A statement describing the applicant's financial and personnel resources to meet its obligations under a new license, including specific information to demonstrate that the applicant's personnel are adequate in number and training to operate and maintain the project in accordance with the provisions of the license.
- (J) If an applicant proposes to expand the project to encompass additional lands, a statement that the applicant has notified, by certified mail, property owners on the additional lands to be encompassed by the project and governmental agencies and subdivisions likely to be interested in or affected by the proposed expansion.
- (K) The applicant's electricity consumption efficiency improvement program, as defined under Section 10(a)(2)(C) of the Federal Power Act, including:
- (1) A statement of the applicant's record of encouraging or assisting its customers to conserve electricity and a description of its plans and capabilities for promoting electricity conservation by its customers; and
- (2) A statement describing the compliance of the applicant's energy con-

- servation programs with any applicable regulatory requirements.
- (L) The names and mailing addresses of every Indian tribe with land on which any part of the proposed project would be located or which the applicant reasonably believes would otherwise be affected by the proposed project.
- (ii) Information to be provided by an applicant licensee. An existing licensee that applies for a new license must provide:
- (A) The information specified in paragraph (c)(1) of this section.
- (B) A statement of measures taken or planned by the licensee to ensure safe management, operation, and maintenance of the project, including:
- (1) A description of existing and planned operation of the project during flood conditions:
- (2) A discussion of any warning devices used to ensure downstream public safety;
- (3) A discussion of any proposed changes to the operation of the project or downstream development that might affect the existing Emergency Action Plan, as described in subpart C of part 12 of this chapter, on file with the Commission;
- (4) A description of existing and planned monitoring devices to detect structural movement or stress, seepage, uplift, equipment failure, or water conduit failure, including a description of the maintenance and monitoring programs used or planned in conjunction with the devices; and
- (5) A discussion of the project's employee safety and public safety record, including the number of lost-time accidents involving employees and the record of injury or death to the public within the project boundary.
- (C) A description of the current operation of the project, including any constraints that might affect the manner in which the project is operated.
- (D) A discussion of the history of the project and record of programs to upgrade the operation and maintenance of the project.
- (E) A summary of any generation lost at the project over the last five years because of unscheduled outages, including the cause, duration, and corrective action taken.

- (F) A discussion of the licensee's record of compliance with the terms and conditions of the existing license, including a list of all incidents of noncompliance, their disposition, and any documentation relating to each incident.
- (G) A discussion of any actions taken by the existing licensee related to the project which affect the public.
- (H) A summary of the ownership and operating expenses that would be reduced if the project license were transferred from the existing licensee.
- (I) A statement of annual fees paid under part I of the Federal Power Act for the use of any Federal or Indian lands included within the project boundary.
- (iii) Information to be provided by an applicant who is not an existing licensee. An applicant that is not an existing licensee must provide:
- (A) The information specified in paragraph (c)(1) of this section.
- (B) A statement of the applicant's plans to manage, operate, and maintain the project safely, including:
- (1) A description of the differences between the operation and maintenance procedures planned by the applicant and the operation and maintenance procedures of the existing licensee:
- (2) A discussion of any measures proposed by the applicant to implement the existing licensee's Emergency Action Plan, as described in subpart C of part 12 of this chapter, and any proposed changes;
- (3) A description of the applicant's plans to continue safety monitoring of existing project instrumentation and any proposed changes; and
- (4) A statement indicating whether or not the applicant is requesting the licensee to provide transmission services under section 15(d) of the Federal Power Act.
- (d) Consistency with comprehensive plans. An application for license under this part must include an explanation of why the project would, would not, or should not, comprehensive plan as defined in §2.19 of this chapter and a description of any relevant resource agency or Indian tribe determination regarding the con-

sistency of the project with any such comprehensive plan.

- (e) Response to information requests. An application for license under this Section must respond to any requests for additional information-gathering or studies filed with comments on its preliminary licensing proposal or draft license application. If the license applicant agrees to do the information-gathering or study, it must provide the information or include a plan and schedule for doing so, along with a schedule for completing any remaining work under the previously approved study plan, as it may have been amended. If the applicant does not agree to any additional information-gathering study requests made in comments on the draft license application, it must explain the basis for declining to do so.
- (f) Maps and drawings. All required maps and drawings must conform to the specifications of §4.39 of this chapter.

[Order 2002, 68 FR 51121, Aug. 25, 2003; 68 FR 61742, Oct. 30, 2003; 68 FR 69957, Dec. 16, 2003; Order 699, 72 FR 45324, Aug. 14, 2007]

§5.19 Tendering notice and schedule.

- (a) Notice. Within 14 days of the filing date of any application for a license developed pursuant to this part, the Commission will issue public notice of the tendering for filing of the application. The tendering notice will include a preliminary schedule for expeditious processing of the application, including dates for:
- (1) Issuance of the acceptance for filing and ready for environmental analysis notice provided for in §5.22.
- (2) Filing of recommendations, preliminary terms and conditions, and fishway prescriptions;
- (3) Issuance of a draft environmental assessment or environmental impact statement, or an environmental assessment not preceded by a draft.
- (4) Filing of comments on the draft environmental assessment or environmental impact statement, as applicable:
- (5) Filing of modified recommendations, mandatory terms and conditions, and fishway prescriptions in response to a draft NEPA document or Environmental Analysis, if no draft NEPA document is issued:

- (6) Issuance of a final NEPA document, if any;
- (7) In the case of a new or subsequent license application, a deadline for submission of final amendments, if any, to the application; and
- (8) Readiness of the application for Commission decision.
- (b) Modifications to process plan and schedule. The tendering notice shall also include any known modifications to the schedules developed pursuant to §5.8 for completion of consultation under section 7 of the Endangered Species Act and water quality certification under section 401 of the Clean Water Act.
- (c) *Method of notice*. The public notice provided for in paragraphs (a) and (b) of this Section will be given by:
- (1) Publishing notice in the FEDERAL REGISTER: and
- (2) Notifying appropriate Federal, state, and interstate resource agencies, state water quality and coastal zone management plan consistency certification agencies, Indian tribes, and nongovernmental organizations, by electronic means if practical, otherwise by mail
- (d) Resolution of pending information requests. Within 30 days of the filing date of any application for a license developed pursuant to this part, the Director of the Office of Energy Projects will issue an order resolving any requests for additional information-gathering or studies made in comments on the preliminary licensing proposal or draft license application.

[Order 2002, 68 FR 51121, Aug. 25, 2003; 68 FR 61742, Oct. 30, 2003; 68 FR 69957, Dec. 16, 2003; Order 653, 70 FR 8724, Feb. 23, 2005]

§ 5.20 Deficient applications.

- (a) Deficient applications. (1) If an applicant believes that its application conforms adequately to the pre-filing consultation and filing requirements of this part without containing certain required materials or information, it must explain in detail why the material or information is not being submitted and what steps were taken by the applicant to provide the material or information.
- (2) Within 30 days of the filing date of any application for a license under this part, the Director of the Office of En-

- ergy Projects will notify the applicant if, in the Director's judgment, the application does not conform to the prefiling consultation and filing requirements of this part, and is therefore considered deficient. An applicant having a deficient application will be afforded additional time to correct the deficiencies, not to exceed 90 days from the date of notification. Notification will be by letter or, in the case of minor deficiencies, by telephone. Any notification will specify the deficiencies to be corrected. Deficiencies must be corrected by submitting an a filing pursuant to the requirements of subpart T of part 385 of this chapter within the time specified in the notification of deficiency.
- (3) If the revised application is found not to conform to the prefiling consultation and filing requirements of this part, or if the revisions are not timely submitted, the revised application will be rejected. Procedures for rejected applications are specified in paragraph (b)(3) of this section.
- (b) Patently deficient applications. (1) If, within 30 days of its filing date, the Director of the Office of Energy Projects determines that an application patently fails to substantially comply with the prefiling consultation and filing requirements of this part, or is for a project that is precluded by law, the application will be rejected as patently deficient with the specification of the deficiencies that render the application patently deficient.
- (2) If, after 30 days following its filing date, the Director of the Office of Energy Projects determines that an application patently fails to comply with the prefiling consultation and filing requirements of this part, or is for a project that is precluded by law:
- (i) The application will be rejected by order of the Commission, if the Commission determines that it is patently deficient; or
- (ii) The application will be considered deficient under paragraph (a)(2) of this Section, if the Commission determines that it is not patently deficient.
- (3) Any application for an original license that is rejected may be submitted if the deficiencies are corrected and if, in the case of a competing application, the resubmittal is timely. The

date the rejected application is resubmitted will be considered the new filing date for purposes of determining its timeliness under §4.36 of this chapter and the disposition of competing applications under §4.37 of this chapter.

[Order 2002, 68 FR 51121, Aug. 25, 2003; 68 FR 61743, Oct. 30, 2003]

§5.21 Additional information.

An applicant may be required to submit any additional information or documents that the Commission considers relevant for an informed decision on the application. The information or documents must take the form, and must be submitted within the time, that the Commission prescribes. An applicant may also be required to provide within a specified time additional copies of the complete application, or any of the additional information or documents that are filed, to the Commission or to any person, agency, Indian tribe or other entity that the Commission specifies. If an applicant fails to provide timely additional information, documents, or copies of submitted materials as required, the Commission may dismiss the application, hold it in abeyance, or take other appropriate action under this chapter or the Federal Power Act.

§ 5.22 Notice of acceptance and ready for environmental analysis.

- (a) When the Commission has determined that the application meets the Commission's requirements as specified in §§5.18 and 5.19, the approved studies have been completed, any deficiencies in the application have been cured, and no other additional information is needed, it will issue public notice as required in the Federal Power Act:
- (1) Accepting the application for filing and specifying the date upon which the application was accepted for filing (which will be the application filing date if the Secretary receives all of the information and documents necessary to conform to the requirements of §§5.1 through 5.21, as applicable, within the time frame prescribed in §5.20 or §5.21);
- (2) Finding that the application is ready for environmental analysis;
- (3) Requesting comments, protests, and interventions;

- (4) Requesting recommendations, preliminary terms and conditions, and preliminary fishway prescriptions, including all supporting documentation; and
- (5) Establishing the date for final amendments to applications for new or subsequent licenses; and
- (6) Updating the schedule issued with the tendering notice for processing the application.
- (b) If the project affects lands of the United States, the Commission will notify the appropriate Federal office of the application and the specific lands affected, pursuant to Section 24 of the Federal Power Act.
- (c) For an application for a license seeking benefits under Section 210 of the Public Utility Regulatory Polices Act of 1978, as amended, for a project that would be located at a new dam or diversion, the Applicant must serve the public notice issued under paragraph (a)(1) of this Section to interested agencies at the time the applicant is notified that the application is accepted for filling.

[Order 2002, 68 FR 51121, Aug. 25, 2003; 68 FR 61743, Oct. 30, 2003]

§ 5.23 Response to notice.

- (a) Comments and reply comments. Comments, protests, interventions, recommendations, and preliminary terms and conditions or preliminary fishway prescriptions must be filed no later than 60 days after the notice of acceptance and ready for environmental analysis. All reply comments must be filed within 105 days of that notice.
- (b) Water quality certification. (1) With regard to certification requirements for a license applicant under Section 401(a)(1) of the Federal Water Pollution Control Act (Clean Water Act), the license applicant must file no later than 60 days following the date of issuance of the notice of acceptance and ready for environmental analysis provide for in \$5.22:
- (i) A copy of the water quality certification;
- (ii) A copy of the request for certification, including proof of the date on which the certifying agency received the request; or

- (iii) Evidence of waiver of water quality certification as described in paragraph (b)(5)(2) of this Section.
- (2) A certifying agency is deemed to have waived the certification requirements of section 401(a)(1) of the Clean Water Act if the certifying agency has not denied or granted certification by one year after the date the certifying agency received a written request for certification. If a certifying agency denies certification, the applicant must file a copy of the denial within 30 days after the applicant received it.
- (3) Notwithstanding any other provision in 18 CFR part 4, subpart B, any application to amend an existing license, and any application to amend a pending application for a license, requires a new request for water quality certification pursuant to §4.34(b)(5) of this chapter if the amendment would have a material adverse impact on the water quality in the discharge from the project or proposed project.

§ 5.24 Applications not requiring a draft NEPA document.

- (a) If the Commission determines that a license application will be processed with an environmental assessment rather than an environmental impact statement and that a draft environmental assessment will not be required, the Commission will issue the environmental assessment for comment no later than 120 days from the date responses are due to the notice of acceptance and ready for environmental analysis.
- (b) Each environmental assessment issued pursuant to this paragraph must include draft license articles, a preliminary determination of consistency of each fish and wildlife agency recommendation made pursuant to Federal Power Act section 10(j) with the purposes and requirements of the Federal Power Act and other applicable law, as provided for in §5.26, and any preliminary mandatory terms and conditions and fishway prescriptions.
- (c) Comments on an environmental assessment issued pursuant to paragraph (a) of this section, including comments in response to the Commission's preliminary determination with respect to fish and wildlife agency recommendations and on preliminary

mandatory terms and conditions or fishway prescriptions, must be filed no later than 30 or 45 days after issuance of the environmental assessment, as specified in the notice accompanying issuance of the environmental assessment, as should any revisions to supporting documentation.

(d) Modified mandatory prescriptions or terms and conditions must be filed no later than 60 days following the date for filing of comments provided for in paragraph (c) of this section, as specified in the notice accompanying issuance of the environmental analysis.

[Order 2002, 68 FR 51121, Aug. 25, 2003; 68 FR 61743, Oct. 30, 2003]

§5.25 Applications requiring a draft NEPA document.

- (a) If the Commission determines that a license application will be processed with an environmental impact statement, or a draft and final environmental assessment, the Commission will issue the draft environmental impact statement or environmental assessment for comment no later than 180 days from the date responses are due to the notice of acceptance and ready for environmental analysis provided for in §5.22.
- (b) Each draft environmental document will include for comment draft license articles, a preliminary determination of the consistency of each fish and wildlife agency recommendation made pursuant to section 10(j) of the Federal Power Act with the purposes and requirements of the Federal Power Act and other applicable law, as provided for in §5.26, and any preliminary mandatory terms and conditions and fishways prescriptions.
- (c) Comments on a draft environmental document issued pursuant to paragraph (b) of this section, including comments in response to the Commission's preliminary determination with respect to fish and wildlife agency recommendations and on preliminary mandatory terms and conditions or prescriptions must be filed no later than 30 or 60 days after issuance of the draft environmental document, as specified in the notice accompanying issuance of the draft environmental document.

- (d) Modified mandatory prescriptions or terms and conditions must be filed no later than 60 days following the date for filing of comments provided for in paragraph (c) of this section.
- (e) The Commission will issue a final environmental document within 90 days following the date for filing of modified mandatory prescriptions or terms and conditions.

§ 5.26 Section 10(i) process.

- (a) In connection with its environmental review of an application for license, the Commission will analyze all terms and conditions timely recommended by fish and wildlife agencies pursuant to the Fish and Wildlife Coordination Act for the protection, mitigation of damages to, and enhancement of fish and wildlife (including related spawning grounds and habitat) affected by the development, operation, and management of the proposed project. Submission of such recommendations marks the beginning of the process under section 10(j) of the Federal Power Act.
- (b) The agency must specifically identify and explain the recommendations and the relevant resource goals and objectives and their evidentiary or legal basis. The Commission may seek clarification of any recommendation from the appropriate fish and wildlife agency. If the Commission's request for clarification is communicated in writing, copies of the request will be sent by the Commission to all parties, affected resource agencies, and Indian tribes, which may file a response to the request for clarification within the time period specified by the Commission. If the Commission believes any fish and wildlife recommendation may be inconsistent with the Federal Power Act or other applicable law, the Commission will make a preliminary determination of inconsistency in the draft environmental document or, if none, the environmental assessment. The preliminary determination, for any recommendations believed to be inconsistent, shall include an explanation why the Commission believes the recommendation is inconsistent with the Federal Power Act or other applicable law, including any supporting analysis and conclusions and an explanation of

how the measures recommended in the environmental document would adequately and equitably protect, mitigate damages to, and enhance, fish and wildlife (including related spawning grounds and habitat) affected by the development, operation, and management of the project.

- (c) Any party, affected resource agency, or Indian tribe may file comments in response to the preliminary determination of inconsistency, including any modified recommendations, within the time frame allotted for comments on the draft environmental document or, if none, the time frame for comments on the environmental assessment. In this filing, the fish and wildlife agency concerned may also request a meeting, telephone or video conference, or other additional procedure to attempt to resolve any preliminary determination of inconsistency.
- (d) The Commission shall attempt, with the agencies, to reach a mutually acceptable resolution of any such inconsistency, giving due weight to the recommendations, expertise, and statutory responsibilities of the fish and wildlife agency. If the Commission decides, or an affected resource agency requests, the Commission will conduct a meeting, telephone or video conference, or other procedures to address issues raised by its preliminary determination of inconsistency and comments thereon. The Commission will give at least 15 days' advance notice to each party, affected resource agency, or Indian tribe, which may participate in the meeting or conference. Any meeting, conference, or additional procedure to address these issues will be scheduled to take place within 90 days of the date the Commission issues a preliminary determination of inconsistency. The Commission will prepare a written summary of any meeting held under this paragraph to discuss section 10(j) issues, including any proposed resolutions and supporting analysis, and a copy of the summary will be sent to all parties, affected resource agencies, and Indian tribes.
- (e) The section 10(j) process ends when the Commission issues an order granting or denying the license application in question. If, after attempting to resolve inconsistencies between the

fish and wildlife recommendations of a fish and wildlife agency and the purposes and requirements of the Federal Power Act or other applicable law, the Commission does not adopt in whole or in part a fish and wildlife recommendation of a fish and wildlife agency, the Commission will publish the findings and statements required by section 10(j)(2) of the Federal Power Act.

§ 5.27 Amendment of application.

- (a) Procedures. If an Applicant files an amendment to its application that would materially change the project's proposed plans of development, as provided in §4.35 of this chapter, an agency, Indian tribe, or member of the public may modify the recommendations or terms and conditions or prescriptions it previously submitted to the Commission pursuant to §§5.20-5.26. modified recommendations, terms and conditions, or prescriptions must be filed no later than the due date specified by the Commission for comments on the amendment.
- (b) Date of acceptance. The date of acceptance of an amendment of application for an original license filed under this part is governed by the provisions of §4.35 of this chapter.
- (c) New and subsequent licenses. The requirements of §4.35 of this chapter do not apply to an application for a new or subsequent license, except that the Commission will reissue a public notice of the application in accordance with the provisions of §4.32(d)(2) of this chapter if a material amendment, as that term is used in §4.35(f) of this chapter, is filed.
- (d) Deadline. All amendments to an application for a new or subsequent license, including the final amendment, must be filed with the Commission and served on all competing applicants no later than the date specified in the notice issued under §5.22.

[Order 2002, 68 FR 51121, Aug. 25, 2003; 68 FR 61743, Oct. 30, 2003]

§ 5.28 Competing applications.

(a) Site access for a competing applicant. The provisions of §16.5 of this chapter shall govern site access for a potential license application to be filed in competition with an application for a new or subsequent license by an ex-

isting licensee pursuant to this part, except that references in §16.5 to the pre-filing consultation provisions in parts 4 and 16 of this chapter shall be construed in a manner compatible with the effective administration of this part.

- (b) Competing applications. The provisions of §4.36 of this chapter shall apply to competing applications for original, new, or subsequent licenses filed under this part.
- (c) New or subsequent license applications—final amendments; better adapted statement. Where two or more mutually exclusive competing applications for new or subsequent license have been filed for the same project, the final amendment date and deadlines for complying with provisions of §4.36(d)(2) (ii) and (iii) of this chapter established pursuant to the notice issued under §5.22 will be the same for all such applications.
- (d) Rules of preference among competing applicants. The Commission will select among competing applications according to the provisions of §4.37 of this chapter.

[Order 2002, 68 FR 51121, Aug. 25, 2003; 68 FR 61743, Oct. 30, 2003]

§5.29 Other provisions.

- (a) Filing requirement. Unless otherwise provided by statute, regulation or order, all filings in hydropower hearings, except those conducted by trial-type procedures, must conform to the requirements of 18 CFR part 385, subpart T of this chapter.
- (b) Waiver of compliance with consultation requirements. (1) If an agency, Indian tribe, or member of the public waives in writing compliance with any consultation requirement of this part, an applicant does not have to comply with the requirement as to that agency, Indian tribe, or member of the public.
- (2) If an agency, Indian tribe, member of the public fails to timely comply with a provision regarding a requirement of this section, an applicant may proceed to the next sequential requirement of this section without waiting for the agency, Indian tribe, or member of the public.

- (c) Requests for privileged treatment of pre-filing submission. If a potential Applicant requests privileged treatment of any information submitted to the Commission during pre-filing consultation (except for the information specified in §5.4), the Commission will treat the request in accordance with the provisions in §388.112 of this chapter until the date the application is filed with the Commission.
- (d) Conditional applications. Any application, the effectiveness of which is conditioned upon the future occurrence of any event or circumstance, will be rejected.
- (e) Trial-type hearing. The Commission may order a trial-type hearing on an application for a license under this part either upon its own motion or the motion of any interested party of record. Any trial-type hearing will be limited to the issues prescribed by order of the Commission. In all other cases, the hearings will be conducted by notice and comment procedures.
- (f) Notice and comment hearings. (1) All comments and reply comments and all other filings described in this part must be served on all persons on the service list prepared by the Commission, in accordance with the requirements of §385.2010 of this chapter. If a party submits any written material to the Commission relating to the merits of an issue that may affect the responsibility of particular resource agency, the party must also serve a copy of the submission on that resource agency.
- (2) The Director of Energy Projects may waive or modify any of the provisions of this part for good cause. A commenter or reply commenter may obtain an extension of time from the Commission only upon a showing of good cause or extraordinary circumstances in accordance with § 385.2008 of this chapter.
- (3) Late-filed recommendations by fish and wildlife agencies pursuant to the Fish and Wildlife Coordination Act and section 10(j) of the Federal Power Act for the protection, mitigation of damages to, and enhancement of fish and wildlife affected by the development, operation, and management of the proposed project and late-filed terms and conditions or prescriptions filed pursuant to sections 4(e) and 18 of

- the Federal Power Act, respectively, will be considered by Commission under section 10(a) of the Federal Power Act if such consideration would not delay or disrupt the proceeding.
- (g) Settlement negotiations. (1) The Commission will consider, on a case-by-case basis, requests for a short suspension of the procedural schedule for the purpose of participants conducting settlement negotiations, where it determines that the suspension will not adversely affect timely action on a license application. In acting on such requests, the Commission will consider, among other things:
- (i) Whether requests for suspension of the procedural schedule have previously been made or granted;
- (ii) Whether the request is supported by a consensus of participants in the proceeding and an explanation of objections to the request expressed by any participant;
- (iii) The likelihood that a settlement agreement will be filed within the requested suspension period; and
- (iv) Whether the requested suspension is likely to cause any new or subsequent license to be issued after the expiration of the existing license.
- (2) The Commission reserves the right to terminate any suspension of the procedural schedule if it concludes that insufficient progress is being made toward the filing of a settlement agreement.
- (h) License conditions and required findings. (1) All licenses shall be issued on the conditions specified in Section 10 of the Federal Power Act and such other conditions as the Commission determines are lawful and in the public interest.
- (2) Subject to paragraph (f)(3) of this section, fish and wildlife conditions shall be based on recommendations timely received from the fish and wildlife agencies pursuant to the Fish and Wildlife Coordination Act.
- (3) The Commission will consider the timely recommendations of resource agencies, other governmental units, and members of the public, and the timely recommendations (including fish and wildlife recommendations) of Indian tribes affected by the project.
- (4) Licenses for a project located within any Federal reservation shall be

issued only after the findings required by, and subject to any conditions that may be timely filed pursuant to section 4(e) of the Federal Power Act.

- (5) The Commission will require the construction, maintenance, and operation of such fishways as may be timely prescribed by the Secretary of Commerce or the Secretary of the Interior, as appropriate, pursuant to section 18 of the Federal Power Act.
- (i) Standards and factors for issuing a new license. (1) In determining whether a final proposal for a new license under section 15 of the Federal Power Act is best adapted to serve the public interest, the Commission will consider the factors enumerated in sections 15(a)(2) and (a)(3) of the Federal Power Act.
- (2) If there are only insignificant differences between the final applications of an existing licensee and a competing Applicant after consideration of the factors enumerated in section 15(a)(2) of the Federal Power Act, the Commission will determine which Applicant will receive the license after considering:
- (i) The existing licensee's record of compliance with the terms and conditions of the existing license; and
- (ii) The actions taken by the existing licensee related to the project which affect the public.
- (iii) An existing licensee that files an application for a new license in conjunction with an entity or entities that are not currently licensees of all or part of the project will not be considered an existing licensee for the purpose of the insignificant differences provision of section 15(a)(2) of the Federal Power Act.
- (j) Fees under section 30(e) of the Federal Power Act. The requirements of 18 CFR part 4, subpart M, of this chapter, fees under section 30(e) of the Federal Power Act, apply to license applications developed under this part.

§ 5.30 Critical energy infrastructure information.

If any action required by this part requires a potential Applicant or Applicant to reveal Critical Energy Infrastructure Information, as defined by §388.113(c) of this chapter, to the public, the Applicant must follow the pro-

cedures set out in §4.32(k) of this chapter.

§ 5.31 Transition provision.

This part shall apply to license applications for which the deadline for filing a notification of intent to seek a new or subsequent license, or for filing a notification of intent to file an original license application, as required by §5.5 of this part, is July 23, 2005 or later.

PART 6—SURRENDER OR TERMINATION OF LICENSE

Sec.

- 6.1 Application for surrender.
- 6.2 Surrender of license.
- 6.3 Termination of license.6.4 Termination by implied surrender.
- 6.5 Annual charges.

AUTHORITY: Secs. 6, 10(i), 13, 41 Stat. 1067, 1068, 1071, as amended, sec. 309, 49 Stat. 858; 16 U.S.C. 799, 803(i), 806, 825h; Pub. L. 96-511, 94 Stat. 2812 (44 U.S.C. 3501 et seq.), unless otherwise noted.

§ 6.1 Application for surrender.

Every application for surrender of a license shall state the reason therefor; and, except in the case of an application for surrender of a license for a minor project, or for a transmission line only, shall be executed by the licensee and filed in the same form and manner as the application for license, and shall be accompanied by the license and all amendments thereof. Public notice of such application shall be given at least 30 days prior to action upon the application.

(Secs. 308 and 309; 49 Stat. 858, 859 (16 U.S.C. 825g, 825h))

[Order No. 570, 42 FR 40191, Aug. 9, 1977]

CROSS REFERENCES: For application for license, general provisions, see §§ 4.30 to 4.33, inclusive, of this chapter. For application for license for proposed major project or minor part thereof, see §§ 4.40 to 4.41, inclusive, of this chapter. For application for license for constructed major project or minor part thereof, see §§ 4.50 and 4.51 of this chapter.

§6.2 Surrender of license.

Licenses may be surrendered only upon the fulfillment by the licensee of such obligations under the license as the Commission may prescribe, and, if

§ 6.3

the project works authorized under the license have been constructed in whole or in part, upon such conditions with respect to the disposition of such works as may be determined by the Commission. Where project works have been constructed on lands of the United States the licensee will be required to restore the lands to a condition satisfactory to the Department having supervision over such lands and annual charges will continue until such restoration has been satisfactorily completed.

[Order 175, 19 FR 5217, Aug. 18, 1954]

§ 6.3 Termination of license.

Licenses may be terminated by written order of the Commission not less than 90 days after notice thereof shall have been mailed to the licensee by certified mail to the last address whereof the Commission has been notified by the licensee, if there is failure to commence actual construction of the project works within the time prescribed in the license, or as extended by the Commission. Upon like notice, the authority granted under a license with respect to any separable part of the project works may be terminated if there is failure to begin construction of such separable part within the time prescribed or as extended by the Commission.

(Administrative Procedure Act, 5 U.S.C. 551–557 (1976); Federal Power Act, as amended, 16 U.S.C. 291–628 (1976 & Supp. V 1981), Dept. of Energy Organization Act 42 U.S.C. 7101–7352 (Supp. V 1981); E.O. 12009, 3 CFR 142 (1978))

[Order 141, 12 FR 8491, Dec. 19, 1947, as amended by Order 344, 48 FR 49010, Oct. 24, 1983]

§ 6.4 Termination by implied surrender.

If any licensee holding a license subject to the provisions of section 10(i) of the Act shall cause or suffer essential project property to be removed or destroyed, or become unfit for use, without replacement, or shall abandon, or shall discontinue good faith operation of the project for a period of three years, the Commission will deem it to be the intent of the licensee to surrender the license; and not less than 90

days after public notice may in its discretion terminate the license.

[Order 141, 12 FR 8491, Dec. 19, 1947]

§ 6.5 Annual charges.

Annual charges arising under a license surrendered or terminated shall continue until the effective date set forth in the Commission's order with respect to such surrender or termination.

[Order 175, 19 FR 5217, Aug. 18, 1954]

CROSS REFERENCE: For annual charges, see part 11 of this chapter.

PART 8—RECREATIONAL OPPORTU-NITIES AND DEVELOPMENT AT LI-CENSED PROJECTS

Sec.

- 8.1 Publication of license conditions relating to recreation.
- 8.2 Posting of project lands as to recreational use and availability of information.
- 8.3 Discrimination prohibited.
- 8.11 Information respecting use and development of public recreational opportunities.

AUTHORITY: 5 U.S.C. 551-557; 16 U.S.C. 791a-825r; 42 U.S.C. 7101-7352.

§8.1 Publication of license conditions relating to recreation.

Following the issuance or amendment of a license, the licensee shall make reasonable efforts to keep the public informed of the availability of project lands and waters for recreational purposes, and of the license conditions of interest to persons who may be interested in the recreational aspects of the project or who may wish to acquire lands in its vicinity. Such efforts shall include but not be limited to: the publication of notice in a local newspaper once each week for 4 weeks of the project's license conditions which relate to public access to and the use of the project waters and lands for recreational purposes, recreational plans, installation of recreation and fish and wildlife facilities, reservoir water surface elevations, minimum water releases or rates of change of water releases and such other conditions of general public interest as the

Commission may designate in the order issuing or amending the license.

[Order 299, 30 FR 7313, June 3, 1965]

§8.2 Posting of project lands as to recreational use and availability of information.

(a) Following the issuance or amendment of a license, the licensee shall post and shall maintain at all points of public access which are required by the license (or at such access points as are specifically designated for this purpose by the licensee) and at such other points as are subsequently prescribed by the Commission on its own motion or upon the recommendation of a public recreation agency operating in the area in which the project is located, a conspicuous sign giving the name of the project and the owner of the project, a statement that it is licensed by the Commission and the project number, directions to the areas of the project which are available for public recreation use, permissible times and activities, and other regulations regarding such use, and advising that further information may be obtained at local offices of the licensee in the vicinity of the project. In addition, the licensee shall post at such locations conspicuous notice that the recreation facilities are open to all members of the public without discrimination.

(b) The licensee shall make available for inspection at its local offices in the vicinity of the project the recreation plan approved by the Commission and the entire license instrument, properly indexed for easy reference to the license conditions designated for publications in §8.1.

[Order 299, 30 FR 7313, June 3, 1965, as amended by Order 341, 32 FR 6488, Apr. 27, 1967; 32 FR 11640, Aug. 11, 1967]

§8.3 Discrimination prohibited.

Every licensee maintaining recreation facilities for the use of the public at a licensed project, or employing or permitting any other person to maintain such facilities, shall permit, or require such other person to permit, equal and unobstructed use of such facilities to all members of the public

without regard to race, color, religious creed or national origin.

[Order 341, 32 FR 6488, Apr. 27, 1967]

§8.11 Information respecting use and development of public recreational opportunities.

- (a) Applicability. (1) Except as provided in paragraph (b) of this section, each licensee of a project under major or minor Commission license shall prepare with respect to each development within such project a FERC Form No. 80 and submit them to the Secretary of the Commission in accordance with filing procedures posted on the Commission's Web site at http://www.ferc.gov.
- (2) FERC Form No. 80 is due on April 1, 2009, for data compiled during the calendar year ending December 31, 2008. Thereafter, FERC Form No. 80 is due on April 1 of every sixth year for data compiled during the previous calendar year
- (3) A copy of the Form No. 80 should be retained by the respondent licensee in its file.
- (b) Initial Form No. 80 filings. Each licensee of an unconstructed project shall file an initial Form No. 80 after such project has been in operation for a full calendar year prior to the filing deadline. Each licensee of an existing (constructed) project shall file an initial Form No. 80 after such project has been licensed for a full calendar year prior to the filing deadline.
- (c) Exemptions. A licensee who has filed a Form No. 80 may request an exemption from any further filing of the form for any development that has no existing or potential recreational use or only a minor existing or potential recreational use (as indicated by fewer than 100 recreation days of use during the previous calendar year) by submitting a statement not later than 6 months prior to the due date for the next filing, stating that Form No. 80 has been filed previously for such development and setting out the basis for believing that the development has no existing or potential recreational use

Pt. 9

or a minor existing or potential recreational use.

(Approved by the Office of Management and Budget under control number 1902–0106)

[46 FR 50059, Oct. 9, 1981, as amended by 49 FR 5073, Feb. 10, 1984; Order 419, 50 FR 20096, May 14, 1985; Order 540, 57 FR 21737, May 22, 1992; Order 699, 72 FR 45324, Aug. 14, 2007; Order 737, 75 FR 43403, July 26, 2010]

PART 9—TRANSFER OF LICENSE OR LEASE OF PROJECT PROPERTY

APPLICATION FOR TRANSFER OF LICENSE

Sec.

9.1 Filing.

9.2 Contents of application.

9.3 Transfer.

APPLICATION FOR LEASE OF PROJECT PROPERTY

9.10 Filing.

AUTHORITY: Sec. 8, 41 Stat. 1068, sec. 309, 49 Stat. 858; 16 U.S.C. 801, 825h; Pub. L. 96–511, 94 Stat. 2812 (44 U.S.C. 3501 et seq.)

CROSS REFERENCE: For application for approval of transfer of license, see §131.20 of this chapter.

APPLICATION FOR TRANSFER OF LICENSE

§ 9.1 Filing.

Any licensee desiring to transfer a license or rights thereunder granted, and the person, association, corporation, State, or municipality desiring to acquire the same, shall jointly or severally file an application for approval of such transfer and acquisition. Such application shall be verified, shall conform to §131.20 of this chapter, and shall be filed in accordance with §4.32 of this chapter.

[Order 501, 39 FR 2267, Jan. 18, 1974, as amended by Order 2002, 68 FR 51139, Aug. 25, 2003]

§ 9.2 Contents of application.

Every application for approval of such transfer and acquisition by the proposed transferee shall set forth in appropriate detail the qualifications of the transferee to hold such license and to operate the property under license, which qualifications shall be the same as those required of applicants for license.

[Order 141, 12 FR 8491, Dec. 19, 1947]

CROSS REFERENCES: For administrative rules relating to applicants for license, see part 385 of this chapter. For regulations as to licenses and permits, see part 4 of this chapter.

§ 9.3 Transfer.

- (a) Approval by the Commission of transfer of a license is contingent upon the transfer of title to the properties under license, delivery of all license instruments, and a showing that such transfer is in the public interest. The transferee shall be subject to all the conditions of the license and to all the provisions and conditions of the act, as though such transferee were the original licensee and shall be responsible for the payment of annual charges which accrue prior to the date of transfer.
- (b) When the Commission shall have approved the transfer of the license, its order of approval shall be forwarded to the transferee for acknowledgment of acceptance. Unless application for rehearing is filed, or unless the order is stayed by the Commission, the order shall become final thirty (30) days from date of issuance and the acknowledgment of acceptance shall be filed in triplicate with the Commission within sixty (60) days from date of issuance accompanied by a certified copy of the deed of conveyance or other instrument evidencing transfer of the property under license, together with evidence of the recording thereof.

[Order 175, 19 FR 5217, Aug. 18, 1954]

APPLICATION FOR LEASE OF PROJECT PROPERTY

§9.10 Filing.

Any licensee desiring to lease the project property covered by a license or any part thereof, where the lessee is granted the exclusive occupancy, possession, or use of project works for purposes of generating, transmitting, or distributing power, and the person, association, or corporation, State, or municipality desiring to acquire the project property by lease, must file the proposed lease together with the application in accordance with §4.32(b)(1) of this chapter. The application and the

Commission's action on it will, in general, be subject to the provisions of §§ 9.1 through 9.3.

[Order 737, 75 FR 43403, July 26, 2010]

PART 11—ANNUAL CHARGES UNDER PART I OF THE FEDERAL POWER ACT

Subpart A—Charges for Costs of Administration, Use of Tribal Lands and Other Government Lands, and Use of Government Dams

Sec.

- 11.1 Costs of administration.
- 11.2 Use of government lands.
- 11.3 Use of government dams, excluding pumped storage projects.
- 11.4 Use of government dams for pumped storage projects, and use of tribal lands.
- 11.5 Exemption of minor projects.
- 11.6 Exemption of State and municipal licensees and exemptees.
- 11.7 Effective date.
- 11.8 Adjustment of annual charges.

Subpart B—Charges for Headwater Benefits

- 11.10 General provision; waiver and exemption; definitions.
- 11.11 Energy gains method of determining headwater benefits charges.
- 11.12 Determination of section 10(f) costs.
- 11.13 Energy gains calculations.
- 11.14 Procedures for establishing charges without an energy gains investigation.
- 11.15 Procedures for determining charges by energy gains investigation.
- 11.16 Filing requirements.
- 11.17 Procedures for payment of charges and costs.

Subpart C—General Procedures

- 11.20 Time for payment.
- 11.21 Penalties.

APPENDIX A TO PART 11—FEE SCHEDULE FOR FY 2010

AUTHORITY: 16 U.S.C. 791a–825r; 42 U.S.C. 7101–7352.

Subpart A—Charges for Costs of Administration, Use of Tribal Lands and Other Government Lands, and Use of Government Dams

§11.1 Costs of administration.

(a) Authority. Pursuant to section 10(e) of the Federal Power Act and sec-

tion 3401 of the Omnibus Budget Reconciliation Act of 1986, the Commission will assess reasonable annual charges against licensees and exemptees to reimburse the United States for the costs of administration of the Commission's hydropower regulatory program.

- (b) *Scope*. The annual charges under this section will be charged to and allocated among:
- (1) All licensees of projects of more than 1.5 megawatts of installed capacity; and
- (2) All holders of exemptions under either section 30 of the Federal Power Act or sections 405 and 408 of the Public Utility Regulatory Policies Act of 1978, as amended by section 408 of the Energy Security Act of 1980, but only if the exemption was issued subsequent to April 21, 1995 and is for a project of more than 1.5 megawatts of installed capacity.
- (3) If the exemption for a project of more than 1.5 megawatts of installed capacity was issued subsequent to April 21, 1995 but pursuant to an application filed prior to that date, the exemptee may credit against its annual charge any filing fee paid pursuant to §381.601 of this chapter, which was removed effective April 21, 1995, 18 CFR 381.601 (1994), until the total of all such credits equals the filing fee that was paid.
- (c) Licenses and exemptions other than State or municipal. For licensees and exemptees, other than State or municipal:
- (1) A determination shall be made for each fiscal year of the costs of administration of Part I of the Federal Power Act chargeable to such licensees or exemptees, from which shall be deducted any administrative costs that are stated in the license or exemption or fixed by the Commission in determining headwater benefit payments.
- (2) For each fiscal year the costs of administration determined under paragraph (c)(1) of this section will be assessed against such licenses or exemptee in the proportion that the annual charge factor for each such project bears to the total of the annual charge factors under all such outstanding licenses and exemptions.
- (3) The annual charge factor for each such project shall be found as follows:

§ 11.1

- (i) For a conventional project the factor is its authorized installed capacity plus 112.5 times its annual energy output in millions of kilowatt-hours.
- (ii) For a pure pumped storage project the factor is its authorized installed capacity.
- (iii) For a mixed conventional-pumped storage project the factor is its authorized installed capacity plus 112.5 times its gross annual energy output in millions of kilowatt-hours less 75 times the annual energy used for pumped storage pumping in million of kilowatt-hours.
- (iv) For purposes of determining their annual charges factor, projects that are operated pursuant to an exemption will be deemed to have an annual energy output of zero.
- (4) To enable the Commission to determine such charges annually, each licensee whose authorized installed capacity exceeds 1.5 megawatts must file with the Commission, on or before November 1 of each year, a statement under oath showing the gross amount of power generated (or produced by nonelectrical equipment) and amount of power used for pumped storage pumping by the project during the preceding fiscal year, expressed in kilowatt hours. If any licensee does not report the gross energy output of its project within the time specified above, the Commission's staff will estimate the energy output and this estimate may be used in lieu of the filings required by this section made by such licensee after November 1.
- (5) For unconstructed projects, the assessments start on the date of commencement of project construction. For constructed projects, the assessments start on the effective date of the license or exemption, except for any new capacity authorized therein. The assessments for new authorized capacity start on the date of commencement of construction of such new capacity. In the event that construction commences during a fiscal year, the charges will be prorated based on the date on which construction commenced.
- (d) State and municipal licensees and exemptees. For State or municipal licensees and exemptees:

- (1) A determination shall be made for each fiscal year of the cost of administration under Part I of the Federal Power Act chargeable to such licensees and exemptees, from which shall be deducted any administrative costs that are stated in the license or exemption or that are fixed by the Commission in determining headwater benefit payments
- (2) An exemption will be granted to a licensee or exemptee to the extent, if any, to which it may be entitled under section 10(e) of the Act provided the data is submitted as requested in paragraphs (d) (4) and (5) of this section.
- (3) For each fiscal year the total actual cost of administration as determined under paragraph (d)(1) of this section will be assessed against each such licensee or exemptee (except to the extent of the exemptions granted pursuant to paragraph (d)(2) of this section) in the proportion that the authorized installed capacity of each such project bears to the total such capacity under all such outstanding licenses or exemptions.
- (4) To enable the Commission to compute on the bill for annual charges the exemption to which State and municipal licensees and exemptees are entitled because of the use of power by the licensee or exemptee for State or municipal purposes, each such licensee or exemptee must file with the Commission, on or before November 1 of each year, a statement under oath showing the following information with respect to the power generated by the project and the disposition thereof during the preceding fiscal year, expressed in kilowatt-hours:
- (i) Gross amount of power generated by the project.
- (ii) Amount of power used for station purposes and lost in transmission, etc.
- (iii) Net amount of power available for sale or use by licensee or exemptee, classified as follows:
 - (A) Used by licensee or exemptee.
 - (B) Sold by licensee or exemptee.
- (5) When the power from a licensed or exempted project owned by a State or municipality enters into its electric system, making it impracticable to meet the requirements of this section with respect to the disposition of project power, such licensee or

exemptee may, in lieu thereof, furnish similar information with respect to the disposition of the available power of the entire electric system of the licensee or exemptee.

- (6) The assessments commence on the date of commencement of project operation. In the event that project operation commences during a fiscal year, the charges will be prorated based on the date on which operation commenced.
- (e) *Transmission lines*. For projects involving transmission lines only, the administrative charge will be stated in the license.
- (f) Maximum charge. No licensed or exempted project's annual charge may exceed a maximum charge established each year by the Commission to equal 2.0 percent of the adjusted Commission costs of administration of the hydropower regulatory program. For every project with an annual charge determined to be above the maximum charge, that project's annual charge will be set at the maximum charge, and any amount above the maximum charge will be reapportioned to the remaining projects. The reapportionment will be computed using the method outlined in paragraphs (c) and (d) of this section (but excluding any project whose annual charge is already set at the maximum amount). This procedure will be repeated until no project's annual charge exceeds the maximum
- (g) Commission's costs. (1) With respect to costs incurred by the Commission, the assessment of annual charges will be based on an estimate of the costs of administration of Part I of the Federal Power Act that will be incurred during the fiscal year in which the annual charges are assessed. After the end of the fiscal year, the assessment will be recalculated based on the costs of administration that were actually incurred during that fiscal year; the actual costs will be compared to the estimated costs; and the difference between the actual and estimated costs will be carried over as an adjustment to the assessment for the subsequent fiscal year.
- (2) The issuance of bills based on the administrative costs incurred by the Commission during the year in which

the bill is issued will commence in 1993. The annual charge for the administrative costs that were incurred in fiscal year 1992 will be billed in 1994. At the licensee's option, the charge may be paid in three equal annual installments in fiscal years 1994, 1995, and 1996, plus any accrued interest. If the licensee elects the three-year installment plan, the Commission will accrue interest (at the most recent yield of two-year Treasury securities) on the unpaid charges and add the accrued interest to the installments billed in fiscal years 1995 and 1996.

- (h) In making their annual reports to the Commission on their costs in administering Part I of the Federal Power Act, the United States Fish and Wildlife Service and the National Marine Fisheries Service are to deduct any amounts that were deposited into their Treasury accounts during that year as reimbursements for conducting studies and reviews pursuant to section 30(e) of the Federal Power Act.
- (i) Definition. As used in paragraphs (c) and (d) of this section, authorized installed capacity means the lesser of the ratings of the generator or turbine units. The rating of a generator is the product of the continuous-load capacity rating of the generator in kilovoltamperes (kVA) and the system power factor in kW/kVA. If the licensee or exemptee does not know its power factor, a factor of 1.0 kW/kVA will be used. The rating of a turbine is the product of the turbine's capacity in horsepower (hp) at best gate (maximum efficiency point) opening under the manufacturer's rated head times a conversion factor of 0.75 kW/hp. If the generator or turbine installed has a rating different from that authorized in the license or exemption, or the installed generator is rewound or otherwise modified to change its rating, or the turbine is modified to change its rating, the licensee or exemptee must apply to the Commission to amend its authorized installed capacity to reflect the change.
- (j) Transition. For a license having the capacity of the project for annual charge purposes stated in horsepower, that capacity shall be deemed to be the capacity stated in kilowatts elsewhere

§ 11.2

in the license, including any amendments thereto.

[60 FR 15047, Mar. 22, 1995, as amended by Order 584, 60 FR 57925, Nov. 24, 1995]

§11.2 Use of government lands.

- (a) Reasonable annual charges for recompensing the United States for the use, occupancy, and enjoyment of its lands (other than lands adjoining or pertaining to Government dams or other structures owned by the United States Government) or its other property, will be fixed by the Commission. In fixing such charges the Commission may take into consideration such factors as commercial value, the most profitable use for which the lands or other property may be suited, the beneficial purpose for which said lands or other property have been or may be used, and such other factors as the Commission may deem pertinent.
- (b) Pending further order of the Commission and subject to adjustments as conditions may warrant. annual charges for the use of government lands will be payable in advance, and will be set on the basis of the schedule of rental fees for linear rights-of-way as set out in Appendix A of this part. Annual charges for transmission line rights-of-way will be equal to the peracre charges established by the above schedule. Annual charges for other project lands will be equal to twice the charges established by the schedule. The Commission, by its designee the Executive Director, will update its fees schedule to reflect changes in land values established by the Forest Service. The Executive Director will publish the updated fee schedule in the FEDERAL REGISTER.
- (c)(1) The annual land use charge payable for the nine month transition year of the implementation of this rule (1987) will be payable in three equal installments, with an installment included in the land use charges bills for 1988, 1989, and 1990.
- (2) The charge for one year will equal an amount as computed under the procedures outlined in this section, or twice the previous full normal year's bill (not including the installments described in paragraph (c)(1) of this section), whichever is less.

- (d) The minimum annual charge for use of Government lands under any license will be \$25.
- (e) No licensee under a license issued prior to August 26, 1935, shall be required to pay annual charges in an amount greater than that prescribed in such license, except as may be otherwise provided in the license.

[Order 560, 42 FR 1229, Jan. 6, 1977; 42 FR 6366, Feb. 2, 1977. Redesignated at 51 FR 24318, July 3, 1986; Order No. 469, 52 FR 18209, May 14, 1987; 53 FR 44859, Nov. 7, 1988]

§11.3 Use of government dams, excluding pumped storage projects.

- (a) General rule. (1) Any licensee whose non-Federal project uses a Government dam or other structure for electric power generation and whose annual charges are not already specified in final form in the license must pay the United States an annual charge for the use of that dam or other structure as determined in accordance with this section. Payment of such annual charge is in addition to any reimbursement paid by a licensee for costs incurred by the United States as a direct result of the licensee's project development at such Government dam.
- (2) Any licensee that is obligated under the terms of a license issued on or before September 16, 1986 to pay specified annual charges for the use of a Government dam must continue to pay the annual charges prescribed in the project license pending any readjustment of the annual charge for the project made pursuant to section 10(e) of the Federal Power Act.
- (b) Graduated flat rates. Annual charges for the use of Government dams or other structures owned by the United States are 1 mill per kilowatthour for the first 40 gigawatt-hours of energy a project produces, 1½ mills per kilowatt-hour for over 40 up to and including 80 gigawatt-hours, and 2 mills per kilowatt-hour for any energy the project produces over 80 gigawatt-hours.
- (c) Information reporting. (1) Except as provided in paragraph (c)(2) of this section, each licensee must file with the Commission, on or before November 1 of each year, a sworn statement showing the gross amount of energy generated during the preceding fiscal year

and the amount of energy provided free of charge to the Government. The determination of the annual charge will be based on the gross energy production less the energy provided free of charge to the Government.

(2) A licensee who has filed these data under another section of part 11 or who has submitted identical data with FERC or the Energy Information Administration for the same fiscal year is not required to file the information described in paragraph (c)(1) of this section. Referenced filings should be identified by company name, date filed, docket or project number, and form, number

(d) Credits. A licensee may file a request with the Director of the Office of Energy Projects for a credit for contractual payments made for construction, operation, and maintenance of a Government dam at any time before 30 days after receiving a billing for annual charges determined under this section. The Director, or his designee, will grant such a credit only when the licensee demonstrates that a credit is reasonably justified. The Director, or his designee, shall consider, among other factors, the contractual arrangements between the licensee and the Federal agency which owns the dam and whether these arrangements reveal clearly that substantial payments are being made for power purposes, relevant legislation, and other equitable factors.

[Order 379, 49 FR 22778, June 1, 1984, as amended by Order 379-A, 49 FR 33862, Aug. 27, 1984. Redesignated at 51 FR 24318, July 3, 1986; Order No. 469, 52 FR 18209, May 14, 1987; 52 FR 33802, Sept. 8, 1987; 53 FR 44859, Nov. 7, 1988; Order 647, 69 FR 32438, June 10, 2004]

§ 11.4 Use of government dams for pumped storage projects, and use of tribal lands.

(a) General Rule. The Commission will determine on a case-by-case basis under section 10(e) of the Federal Power Act the annual charges for any pumped storage project using a Government dam or other structure and for any project using tribal lands within Indian reservations.

(b) Information reporting. (1) Except as provided in paragraph (b)(2) of this section a Licensee whose project includes

pumped storage facilities must file with the Commission, on or before November 1 of each year, a sworn statement showing the gross amount of energy generated during the preceding fiscal year, and the amount of energy provided free of charge to the Government, and the amount of energy used for pumped storage pumping.

(2) A licensee who has filed these data under another section of part 11 or who has submitted identical data with FERC or the Energy Information Administration for the same fiscal year is not required to file the information required in paragraph (b)(1) of this section. Referenced filings should be identified by company name, date filed, docket or project number, and form number

(c) Commencing in 1993, the annual charges for any project using tribal land within Indian reservations will be billed during the fiscal year in which the land is used, for the use of that land during that year.

[Order 379, 49 FR 22778, June 1, 1984. Redesignated at 51 FR 24318, July 3, 1986; Order 469, 52 FR 18209, May 14, 1987; 52 FR 33802, Sept. 8, 1987; Order 551, 58 FR 15770, Mar. 24, 1993]

§11.5 Exemption of minor projects.

No exemption will be made from payment of annual charges for the use of Government dams or tribal lands within Indian reservations but licenses may be issued without charges other than for such use for the development, transmission, or distribution of power for domestic, mining, or other beneficial use in minor projects.

[Order 141, 12 FR 8492, Dec. 19, 1947. Redesignated by Order 379, 49 FR 22778, June 1, 1984. Redesignated at 51 FR 24318, July 3, 1986]

§ 11.6 Exemption of State and municipal licensees and exemptees.

- (a) Bases for exemption. A State or municipal licensee or exemptee may claim total or partial exemption from the assessment of annual charges upon one or more of the following grounds:
- (1) The project was primarily designed to provide or improve navigation:
- (2) To the extent that power generated, transmitted, or distributed by

§ 11.6

the project was sold directly or indirectly to the public (ultimate consumer) without profit;

- (3) To the extent that power generated, transmitted, or distributed by the project was used by the licensee for State or municipal purposes.
- (b) Projects primarily for navigation. No State or municipal licensee shall be entitled to exemption from the payment of annual charges on the ground that the project was primarily designed to provide or improve navigation unless the licensee establishes that fact from the actual conditions under which the project was constructed and was operated during the calendar year for which the charge is made.
- (c) State or municipal use. A State or municipal licensee shall be entitled to exemption from the payment of annual charges for the project to the extent that power generated, transmitted, or distributed by the project is used by the licensee itself for State or municipal purposes, such as lighting streets, highways, parks, public buildings, etc., for operating licensee's water or sewerage system, or in performing other public functions of the licensee.
- (d) Sales to public. No State or municipal licensee shall be entitled to exemption from the payment of annual charges on the ground that power generated, transmitted, or distributed by the project is sold to the public without profit, unless such licensee shall show:
- (1) That it maintains an accounting system which segregates the operations of the licensed project and reflects with reasonable accuracy the revenues and expenses of the project;
- (2) That an income statement, prepared in accordance with the Commission's Uniform System of Accounts, shows that the revenues from the sale of project power do not exceed the total amount of operating expenses, maintenance, depreciation, amortization, taxes, and interest on indebtedness, applicable to the project property. Periodic accruals or payments for redemption of the principal of bonds or other indebtedness may not be deducted in determining the net profit of the project.
- (e) Sales for resale. Notwithstanding compliance by a State or municipal li-

- censee with the requirements of paragraph (d) of this section, it shall be subject to the payment of annual charges to the extent that electric power generated, transmitted, or distributed by the project is sold to another State, municipality, person, or corporation for resale, unless the licensee shall show that the power was sold to the ultimate consumer without profit. The matter of whether or not a profit was made is a question of fact to be established by the licensee.
- (f) Interchange of power. Notwithstanding compliance by a State or municipal licensee with the requirements of paragraph (d) of this section, it shall be subject to the payment of annual charges to the extent that power generated, transmitted, or distributed by the project was supplied under an interchange agreement to a State, municipality, person, or corporation for sale at a profit (which power was not offset by an equivalent amount of power received under such interchange agreement) unless the licensee shall show that the power was sold to ultimate consumers without profit.
- (g) Construction period. During the period when the licensed project is under construction and is not generating power, it will be considered as operating without profit within the meaning of this section, and licensee will be entitled to total exemption from the payment of annual charges, except as to those charges relating to the use of a Government dam or tribal lands within Indian reservations.
- (h) Optional showing. When the power from the licensed project enters into the electric power system of the State or municipal licensee, making it impracticable to meet the requirements set forth in this section with respect to the operations of the project only, such licensee may, in lieu thereof, furnish the same information with respect to the operations of said electric power system as a whole
- (i) Application for exemption. Applications for exemption from payment of annual charges shall be signed by an authorized executive officer or chief accounting officer of the licensee or exemptee and verified under oath. The

application must be filed with the Secretary of the Commission in accordance with filing procedures posted on the Commission's Web site at http:// www.ferc.gov within the time allowed (by §11.20) for the payment of the annual charges. If the licensee exemptee, within the time allowed for the payment of the annual charges, files notice that it intends to file an application for exemption, an additional period of 30 days is allowed within which to complete and file the application for exemption. The filing of an application for exemption does not by itself alleviate the requirement to pay the annual charges, nor does it exonerate the licensee or exemptee from the assessment of penalties under §11.21. If a bill for annual charges becomes payable after an application for an exemption has been filed and while the application is still pending for decision, the bill may be paid under protest and subject to refund.

[Order 143, 13 FR 6681, Nov. 13, 1948. Redesignated and amended by Order 379, 49 FR 22778, June 1, 1984. Redesignated at 51 FR 24318, July 3, 1986; 60 FR 15048, Mar. 22, 1995; Order 737, 75 FR 43403, July 26, 2010]

§11.7 Effective date.

All annual charges imposed under this subpart will be computed beginning on the effective date of the license unless some other date is fixed in the license.

[51 FR 24318, July 3, 1986]

§11.8 Adjustment of annual charges.

All annual charges imposed under this subpart continue in effect as fixed unless changed as authorized by law.

 $[51~{\rm FR}~24318,~{\rm July}~3,~1986]$

Subpart B—Charges for Headwater Benefits

SOURCE: Order 453, 51 FR 24318, July 3, 1986, unless otherwise noted.

§11.10 General provision; waiver and exemptions; definitions.

(a) Headwater benefits charges. (1) The Commission will assess or approve charges under this subpart for direct benefits derived from headwater projects constructed by the United

States, a licensee, or a pre-1920 permittee. Charges under this subpart will amount to an equitable part of the annual costs of interest, maintenance, and depreciation expenses of such headwater projects and the costs to the Commission of determining headwater benefits charges. Except as provided in paragraph (b) of this section, the owner of any non-Federal downstream project that receives headwater benefits must pay charges determined under this subpart.

- (2) Headwater benefits are the additional electric generation at a downstream project that results from regulation of the flow of the river by the headwater, or upstream, project, usually by increasing or decreasing the release of water from a storage reservoir.
- (b) Waiver and exemptions. The owner of a downstream project with installed generating capacity of 1.5 MW (2000 horsepower) or less or for which the Commission has granted an exemption from section 10(f) is not required to pay headwater benefits charges.
- (c) *Definitions*. For purposes of this subpart:
- (1) Energy gains means the difference between the number of kilowatt-hours of energy produced at a downstream project with the headwater project and that which would be produced without the headwater project.
- (2) Generation means gross generation of electricity at a hydroelectric project, including generation needed for station use or the equivalent for direct drive units, measured in kilowatthours. It does not include energy used for or derived from pumping in a pumped storage facility.
- (3) Headwater project costs means the total costs of an upstream project constructed by the United States, a licensee, or pre-1920 permittee.
- (4) Separable cost means the difference between the cost of a multiple-function headwater project with and without any particular function.
- (5) Remaining benefits means the difference between the separable cost of a specific function in a multiple-function project and the lesser of:
- (i) The benefits of that function in the project, as determined by the responsible Federal agency at the time

§ 11.11

the project or function was authorized;

- (ii) The cost of the most likely alternative single-function project providing the same benefits.
- (6) Joint-use cost means the difference between the total project cost and the total separable costs. Joint-use costs are allocated among the project functions according to each function's percentage of the total remaining benefits.
- (7) Specific power cost means that portion of the headwater project costs that is directly attributable to the function of power generation at the headwater project, including, but not limited to, the cost of the electric generators, turbines, penstocks, and substation.
- (8) *Joint-use power cost* means the portion of the joint-use cost allocated to the power function of the project.
- (9) Section 10(f) costs means the annual interest, depreciation, and maintenance expense portion of the jointuse power cost, including costs of nonpower functions required by statute to be paid by revenues from the power function.
 - (10) Party means:
- (i) The owner of a non-Federal downstream hydroelectric project which is directly benefited by a headwater project constructed by the United States, a licensee, or a pre-1920 permittee;
- (ii) The owner of a headwater project constructed by the United States, a licensee, or a pre-1920 permittee;
- (iii) An operating agency of, or an agency marketing power from, a headwater project constructed by the United States; or
- (iv) Any party, as defined in §385.102(c) of this chapter.
- (11) Final charge means a charge assessed on an annual basis to recover section 10(f) costs and which represents the final determination of the charge for the period for which headwater benefits are assessed. Final charges may be established retroactively, to finalize an interim charge, or prospectively.
- (12) Interim charge means a charge assessed to recover section 10(f) costs for a specified period of headwater benefits pending determination of a final charge for that period.

- (13) Investment cost means the sum of: (i) Project construction costs, including cost of land, labor and materials, cost of pre- and post-authorization investigations, and cost of engineering, supervision, and administration during
- construction of the project; and (ii) Interest during construction.

[Order 453, 51 FR 24318, July 3, 1986, as amended by Order 699, 72 FR 45324, Aug. 14, 2007]

§11.11 Energy gains method of determining headwater benefits charges.

- (a) Applicability. This section applies to any determination of headwater benefits charges, unless:
- (1) The Commission has approved headwater benefits charges pursuant to an existing coordination agreement among the parties;
- (2) The parties reach, and the Commission approves, a settlement with respect to headwater benefits charges, pursuant to §11.14(a) of this subpart; or
- (3) Charges may be assessed under \$11.14(b).
- (b) General rule—(1) Summary. Except as provided in paragraph (b)(3) of this section, a headwater benefits charge for a downstream project is determined under this subpart by apportioning the section 10(f) costs of the headwater project among the headwater project and all downstream projects that are not exempt from or waived from headwater benefits charges under §11.10(b) of this chapter, according to each project's share of the total energy benefits to those projects resulting from the headwater project.
- (2) Calculation; headwater benefits formula. The annual headwater benefits charge for a downstream project is derived by multiplying the section 10(f) cost by the ratio of the energy gains received by the downstream project to the sum of total energy gains received by all downstream projects (except those projects specified in §11.10(b) of this chapter) plus the energy generated at the headwater project that is assigned to the joint-use power cost, as follows:

$$P = C_p \times \frac{E_n}{E_i + E_d}$$

In which:

P=annual payment to be made for headwater benefits received by a downstream project, C_p =annual section 10(f) cost of the headwater project.

 E_n =annual energy gains received at a downstream project, or group of projects if owned by one entity,

 $E_d{=}annual$ energy gains received at all downstream projects (except those specified in \$11.10(b) of this chapter), and

 E_j =portion of the annual energy generated at the headwater project assigned to the joint-use power cost.

- (3) If power generation is not a function of the headwater project, section 10(f) costs will be apportioned only among the downstream projects.
- (4) If the headwater project is constructed after the downstream project, liability for headwater benefits charges will accrue beginning on the day on which any energy losses at the downstream project due to filling the headwater reservoir have been offset by subsequent energy gains. If the headwater project is constructed prior to the downstream project, liability for headwater benefits charges will accrue beginning on the day on which benefits are first realized by the downstream project.
- (5) No final charge assessed by the Commission under this subpart may exceed 85 percent of the value of the energy gains. If a party demonstrates, within the time specified in §11.17(b)(3) for response to a preliminary assessment, that any final charge assessed under this subpart, not including the cost of the investigation assessed under §11.17(c), exceeds 85 percent of the value of the energy gains provided to the downstream project for the period for which the charge is assessed, the Commission will reduce the charge to not more than 85 percent of the value. For purposes of this paragraph, the value of the energy gains is the cost of obtaining an equivalent amount of electricity from the most likely alternative source during the period for which the charge is assessed.

§ 11.12 Determination of section 10(f) costs.

(a) for non-Federal headwater projects. If the headwater project was constructed by a licensee or pre-1920 permittee and a party requests the Commission to determine charges, the

Commission will determine on a caseby-case basis what portion of the annual interest, maintenance, and depreciation costs of the headwater project constitutes the section 10(f) costs, for purposes of this subpart.

- (b) For Federal headwater projects. (1) If the headwater project was constructed or is operated by the United States, and the Commission has not approved a settlement between the downstream project owner and the headwater project owner, the section 10(f) cost will be determined by deriving, from information provided by the headwater project owner pursuant to §11.16 of this subpart, the joint-use power cost and the portion of the annual joint-use power cost that represents the interest, maintenance, and depreciation costs of the project.
- (2) If power is not an authorized function of the headwater project, the section 10(f) cost is the annual interest, maintenance, and depreciation portion of the headwater project costs designated as the joint-use power cost, derived by deeming a power function at the project. The value of the benefits assigned to the deemed power function, for purposes of determining the value of remaining benefits of the joint-use power cost, is the total value of downstream energy gains included in the headwater benefits formula.
- (3) For purposes of this paragraph, total value of downstream energy gains means the lesser of:
- (i) The cost of generating an equivalent amount of electricity at the most likely alternative facility at the time the headwater project became operational; or
- (ii) The incremental cost of installing electrical generation at the headwater project at the time the project became operational.

§11.13 Energy gains calculations.

(a) Energy gains at a downstream project. (1) Energy gains at a downstream project are determined by simulating operation of the downstream project with and without the effects of the headwater project. Except for determinations which are not complex or in which headwater benefits are expected to be small, calculations will be made by application of the Headwater

§ 11.14

Benefits Energy Gains Model, as presented in *The Headwater Benefits Energy Gains (HWBEG) Model Description and Users Manual*, which is available for the National Technical Information Service, U.S. Department of Commerce, 5285 Port Royal Road, Springfield, VA 22161.

- (2) If more than one headwater project provide energy gains to a downstream project, the energy gains at the downstream project are attributed to the headwater projects according to the time sequence of commencement of operation in which each headwater project provided energy gains at the downstream project, by:
- (i) Crediting the headwater project that is first in time with the amount of energy gains that it provided to the downstream project prior to operation of the headwater project that is next in time; and
- (ii) Crediting any subsequent headwater project with the additional increment of energy gains provided by it to the downstream project.
- (3) Annual energy losses at a downstream project, or group of projects owned by the same entity, that are attributable to the headwater project will be subtracted from energy gains for the same annual period at the downstream project or group of projects. A net loss in one calendar year will be subtracted from net gains in subsequent years until no net loss remains.
- (b) Energy generated at the headwater project. (1) Except as provided in paragraphs (b)(2) and (b)(3) of this section, the portion of the total annual energy generation at the headwater project that is to be attributed to the joint-use power cost is derived by multiplying the total annual generation at the headwater project and the ratio of the project investment cost assigned to the joint-use power cost to the sum of the investment cost assigned to both the specific power cost and the joint-use power cost of the headwater project, as follows:

$$E_j = E \times \frac{C_j}{C_s + C_i}$$

In which:

 E_j =annual energy generated at the headwater project to be attributed to the jointuse power cost.

E=total annual generation at the headwater project,

C_j=project investment costs assigned to the joint-use power cost, and

 C_s =project investment costs assigned to specific power costs.

- (2) If the headwater project contains a pumped storage facility, calculation of the portion of the total annual energy generation at the headwater project that is attributable to the joint-use power cost will be determined on a case-by-case basis.
- (3) If no power is generated at the headwater project, the amount of energy attributable to the joint-use power cost under this section is the total of all downstream energy gains included in the headwater benefits formula.

§ 11.14 Procedures for establishing charges without an energy gains investigation.

- (a) Settlements. (1) Owners of downstream and headwater projects subject to this subpart may negotiate a settlement for headwater benefits charges. Settlements must be filed with the Commission for its approval, according to the provisions of §385.602.
- (2) If the headwater project is a Federal project, any settlement under this section must result in headwater benefits payments that approximate those that would result under the energy gains method.
- (b) Continuation of previous headwater benefits determinations. (1) For any downstream project being assessed headwater benefit charges on or before September 16, 1986, the Commission will continue to assess charges to that project on the same basis until changes occur in the river basin, including hydrology or project development, that affect headwater benefits.
- (2) Any procedures that apply to §11.17(b)(5) of this subpart will apply to any prospectively fixed charges that are continued under this paragraph.

§ 11.15 Procedures for determining charges by energy gains investigation.

(a) Purpose of investigations; limitation. Except as permitted under §11.14, the

Commission will conduct an investigation to obtain information for establishing headwater benefits charges under this subpart. The Commission will investigate and determine charges for a project downstream from a non-Federal headwater project only if the parties are unable to agree to a settlement and one of the parties requests the Commission to determine charges.

- (b) Notification. The Commission will notify each downstream project owner and each headwater project owner when it initiates an investigation under this section, and the period of project operations to be studied will be specified. An investigation will continue until a final charge has been established for all years studied in the investigation.
- (c) Jurisdictional objections. If any project owner wishes to object to the assessment of a headwater benefits charge on jurisdictional grounds, such objection must:
- (1) Be raised within 30 days after the notice of the investigation is issued; and
- (2) State in detail the grounds for its objection.
- (d) *Investigations*. (1) For any downstream project for which a final charge pursuant to an investigation has never been established, the Commission will conduct an initial investigation to determine a final charge.
- (2) The Commission may, for good cause shown by a party or on its own motion, initiate a new investigation of a river basin to determine whether, because of any change in the hydrology, project development, or other characteristics of the river basin that effects headwater benefits, it should:
- (i) Establish a new final charge to replace a final charge previously established under §11.17(b)(5); or
- (ii) Revise any variable of the headwater benefits formula that has become a constant in calculating a final charge.
- (3) Scope of investigations. (i) The Commission will establish a final charge pursuant to an investigation based on information available to the Commission through the annual data submission requirements of §11.16, if such information is adequate to establish a reasonably accurate final charge.

(ii) If the information available to the Commission is not sufficient to provide a reasonably accurate calculation of the final charge, the Commission will request additional data and conduct any studies, including studies of the hydrology of the river basin and project operations, that it determines necessary to establish the charge.

§11.16 Filing requirements.

- (a) Applicability. (1) Any party subject to a headwater benefits determination under this subpart must supply project-specific data, in accordance with this section, by February 1 of each year for data from the preceding calendar year.
- (2) Within 30 days of notice of initiation of an investigation under §11.15, a party must supply project-specific data, in accordance with this section, for the years specified in the notice.
- (b) Data required from owner of the headwater project. The owner of any headwater project constructed by the United States, a licensee, or a pre-1920 permittee that is upstream from a non-Federal hydroelectric project must submit the following:
- (1) Name and location of the headwater project, including the name of the stream on which it is located.
- (2) The total nameplate rating of installed generating capacity of the project, expressed in kilowatts, with the portion of total capacity that represents pumped storage generating capacity separately designated.
- (3) A description of the total storage capacity of the reservoir and allocation of storage capacity to each of its functions, such as dead storage, power storage, irrigation storage, and flood control storage. Identification, by reservoir elevation, of the portion of the reservoir assigned to each of its respective storage functions.
- (4) An elevation-capacity curve, or a tabulation of reservoir pool elevations with corresponding reservoir storage capacities.
- (5) A copy of rule curves, coordination contracts, agreements, or other relevant data governing the release of water from the reservoir, including a separate statement of their effective dates.

§11.17

- (6) A curve or tabulation showing actual reservoir pool elevations throughout the immediately preceding calendar year and for each year included in an investigation.
- (7) The total annual gross generation of the hydroelectric plant in kilowatthours, not including energy from pumped storage operation.
- (8) The total number of kilowatthours of energy produced from pumped storage operation.
- (9) The investigation costs attributed to the power generation function of the project as of the close of the calendar year or at a specified date during the year, categorized according to that portion that is attributed to the specific power costs, and that portion that is attributed to the joint-use power costs
- (10) The portion of the joint-use power cost, and other costs required by law to be allocated to joint-use power cost, each item shown separately, that are attributable to the annual costs of interest, maintenance, and depreciation, identifying the annual interest rate and the method used to compute the depreciation charge, or the interest rate and period used to compute amortization if used in lieu of depreciation, including any differing interest rates used for major replacements or rehabilitation.
- (c) Data required from owners of downstream projects. The owner of any hydroelectric project which is downstream from a headwater project constructed by the United States, a licensee, or pre-1920 permittee must submit the following:
- (1) Name and location of the downstream project, including the name of the stream on which it is located.
- (2) Total nameplate rating of the installed generating capacity of the plant, expressed in kilowatts, with the portion of total capacity that represents pumped storage generating capacity separately designated.
- (3) Record of daily gross generation, not including energy used for pumped storage, and any unit outage which may have occurred.
- (4) The total number of kilowatthours of energy produced from pumped storage operation.

- (d) Abbreviated data submissions. (1) For those items in paragraphs (b) and (c) of this section in which data for the current period are the same as data furnished for a prior period, the data need not be resubmitted if the owner identifies the last period for which the data were reported.
- (2) The Commission will notify the project owner that certain data items in paragraphs (b) and (c) are no longer required to be submitted annually if:
- (i) A variable in the headwater benefits formula has become a constant; or
- (ii) A prospective final charge, as described in §11.17(b)(5), has been established.
- (e) Additional data. Owners of headwater projects or downstream projects must furnish any additional data required by the Commission staff under paragraph (a) of this section and may provide other data which they consider relevant.

§ 11.17 Procedures for payment of charges and costs.

- (a) Payment for benefits from a non-Federal headwater project. Any billing procedures and payments determined between a non-Federal headwater project owner and a downstream project owner will occur according to the agreement of those parties.
- (b) Charges and payment for benefits from a Federal headwater project—(1) Interim charges. (i) If the Commission has not established a final charge and an investigation is pending, the Commission will issue a downstream project owner a bill for the interim charge and costs and a staff report explaining the calculation of the interim charge.
- (ii) An interim charge will be a percentage of the estimate by the Commission staff of what the final charge will be, as follows:
- (A) 100 percent of the estimated final charge if the Commission previously has completed an investigation of the project for which it is assessed; or
- (B) 80 percent of the estimated final charge if the Commission has not completed an investigation of the project for which it is assessed.
- (iii) When a final charge is established for a period for which an interim charge was paid, the Commission will

apply the amount paid to the final charge.

- (2) Preliminary assessment of a final charge. Unless the project owner was assessed a final charge in the previous year, the Commission will issue to the downstream project owner a preliminary assessment of any final charge when it is determined. A staff technical report explaining the basis of the assessment will be enclosed with the preliminary assessment. Copies of the preliminary assessment will be mailed to all parties.
- (3) Opportunity to respond. After issuance of a preliminary assessment of a final charge, parties may respond in writing within 60 days after the preliminary assessment.
- (4) Order and bill. (i) After the opportunity for written response by the parties to the preliminary assessment of a final charge, the Commission will issue to the downstream project owner an order establishing the final charge. Copies of the order will be mailed to all parties. A bill will be issued for the amount of the final charge and costs.
- (ii) If a final charge is not established prospectively under paragraph (b)(5) of this section, the Commission will issue an order and a bill for the final charge and costs each year until prospective final charges are established. After the Commission issues an order establishing a prospective final charge, a bill will be issued annually for the amount of the final charge and costs.
- (5) Prospective final charges. When the Commission determines that historical data, including the hydrology, development, and other characteristics of the river basin, demonstrate sufficient stability to project average energy gains and section 10(f) costs, the Commission will issue to the downstream project owner an order establishing the final charge from future years. Copies of the order will be mailed to all parties. The prospective final charge will remain in effect until a new investigation is initiated under §11.15(d)(2).
- (6) Payment under protest. Any payment of a final charge required by this section may be made under protest if a party is also appealing the final charge pursuant to §385.1902, or requesting rehearing. If payment is made under pro-

test, that party will avoid any penalty for failure to pay under §11.21.

- (7) Accounting for payments pending appeal or rehearing. The Commission will retain any payment received for final charges from bills issued pursuant to this section in a special account. No disbursements to the U.S. Treasury will be made from the account until 31 days after the bill is issued. If an appeal under §385.1902 or a request for rehearing is filed by any party, no disbursements to the U.S. Treasury will be made until final disposition of the appeal or request for rehearing.
- (c) Charges for costs of determinations of headwater benefits charges. (1) Any owner of a downstream project that benefits from a Federal headwater project must pay to the United States the cost of making any investigation, study, or determination relating to the assessment of the relevant headwater benefits charge under this subpart.
- (2) If any owner of a headwater or downstream project requests that the Commission determine headwater benefits charges for benefits provided by non-Federal headwater projects, the headwater project owners must pay a pro rata share of 50 percent of the cost of making the investigation and determination, in proportion to the benefits provided by their projects, and the downstream project owners must pay a pro rata share of the remaining 50 percent in proportion to the energy gains received by their projects.
- (3) Any charge assessed under this paragraph is separate from and will be added to, any final or interim charge under this subpart.

Subpart C—General Procedures

§ 11.20 Time for payment.

Annual charges must be paid no later than 45 days after rendition of a bill by the Commission. If the licensee or exemptee believes that the bill is incorrect, no later than 45 days after its rendition the licensee or exemptee may file an appeal of the bill with the Chief Financial Officer. No later than 30 days after the date of issuance of the Chief Financial Officer's decision on the appeal, the licensee or exemptee may file a request for rehearing of that decision pursuant to §385.713 of this chapter. In

County

§11.21

the event that a timely appeal to the Chief Financial Officer or a timely request to the Commission for rehearing is filed, the payment of the bill may be made under protest, and subject to refund pending the outcome of the appeal or rehearing.

[60 FR 15048, Mar. 22, 1995]

§11.21 Penalties.

If any person fails to pay annual charges within the periods specified in §11.20, a penalty of 5 percent of the total delinquent amount will be assessed and added to the total charges for the first month or part of month in which payment is delinquent. An additional penalty of 3 percent for each full month thereafter will be assessed until the charges and penalties are satisfied in accordance with law. The Commission may, by order, waive any penalty imposed by this subsection, for good cause shown.

[51 FR 24318, July 3, 1986]

APPENDIX A TO PART 11—FEE SCHEDULE FOR FY 2010

State	County	(Fee/acre/ YR)
Alabama	Autauga	\$62.78
Alabama	Baldwin	94.17
Alabama	Barbour	31.39
Alabama	Bibb	47.08
Alabama	Blount	94.17
Alabama	Bullock	47.08
Alabama	Butler	47.08
Alabama	Calhoun	94.17
Alabama	Chambers	31.39
Alabama	Cherokee	47.08
Alabama	Chilton	47.08
Alabama	Choctaw	47.08
Alabama	Clarke	47.08
Alabama	Clay	47.08
Alabama	Cleburne	62.78
Alabama	Coffee	31.39
Alabama	Colbert	47.08
Alabama	Conecuh	31.39
Alabama	Coosa	47.08
Alabama	Covington	47.08
Alabama	Crenshaw	47.08
Alabama	Cullman	94.17
Alabama	Dale	47.08
Alabama	Dallas	31.39
Alabama	DeKalb	62.78
Alabama	Elmore	62.78
Alabama	Escambia	47.08
Alabama	Etowah	94.17
Alabama	Fayette	31.39
Alabama	Franklin	47.08
Alabama	Geneva	47.08
Alabama	Greene	31.39
Alabama	Hale	31.39
Alabama	Henry	31.39
Alabama	Houston	47.08
Alabama	Jackson	62.78

Alabama	y re-			YH)
Agrice Alabama Lamar Alabama Lauderdale 47.08	ring	Alahama	Jefferson	94 17
O Pe				
Alabama	-			47.08
Alabama	_	Alabama	Lawrence	47.08
Alabama	pear		Lee	62.78
Alabama				62.78
Alabama				
Alabama				
Marion				
Muschall 94.17 9				
Sed in	nual			
the Alabama Montoe 47.08				94.17
Alabama		Alabama	Monroe	47.08
Arges Alabama Perry 31.39			Montgomery	62.78
Ch in Alabama				94.17
Alabama	$_{ m rges}$			
Tull	th in			
Full	addi-			
Section Sect	f1111			
Sfied Alabama St. Clair 62.78 mis-Alabama Sumter 31.39 talty Alabama Talladega 94.17 talty Alabama Talladopas 47.08 good Alabama Maloama Walker 47.08 Alabama Walker 47.08 47.08 Alabama Wilcox 31.39 Viscore/ Alaska Aleutian Islands 7.85 Area*** Alaska Jeurian Islands 7.85 Area*** Alaska Juneau Area*** 1.569.43 Arizona Arizona Apache 7.85 Arizona Apache 7.85 Arizona				
Mished Alabama Sumter 31.39				
Mail	_			31.39
Alabama		Alabama	Talladega	94.17
Alabama	ıalty			47.08
Alabama	good			62.78
Alabama				
Alabama				
DULE Alaska Aleutian Islands 7.85 Alaska Anchorage Area *** 62.78 Alaska Anchorage Area *** 1,569.43 Alaska Juneau Area *** 1,569.43 \$62.78 Arizona Apache 7.85 \$4.70 Arizona Cochise 31.39 \$4.70 Arizona Goconino 7.85 \$1.39 Arizona Gila 7.85 \$47.08 Arizona Graham 15.69 \$47.08 Arizona Graham 15.69 \$47.08 Arizona Greenlee 47.08 \$47.08 Arizona Maricopa 94.17 \$47.08 Arizona Maricopa 94.17 \$47.08 Arizona Navajo 7.85 \$47.08 Arizona Navajo 7.85 \$47.08 Arizona Pima 7.85 \$47.08 Arizona Pima 7.85 \$47.08 Arizona Pinal 31.39				
Area **				
Alaska	DULE			
Alaska		Alaska	Anchorage Area **	62.78
Alaska			Fairbanks Area **	31.39
\$62.78 Arizona Apache 7.85 94.17 Arizona Cochise 31.39 47.08 Arizona Gila 7.85 47.08 Arizona Gila 7.85 47.08 Arizona Graham 15.69 47.08 Arizona La Paz 31.39 47.08 Arizona Maricopa 94.17 47.08 Arizona Maricopa 94.17 31.39 Arizona Mohave 15.69 47.08 Arizona Navajo 7.85 47.08 Arizona Pima 7.85 47.08 Arizona Pima 7.85 47.08 Arizona Pinal 31.39 47.08 Arizona Yavapai 15.69 47.08 Arizona Yavapai 15.69 47.08 Arizona Yuma 156.94 47.08 Arkansas Arkansas 47.08 47.08 Arkansas Arkansas 47.08				1,569.43
\$62.78	H)			
94.17 Arizona Coconino 7.85 31.39 Arizona Gila 7.85 47.08 Arizona Graham 15.69 94.17 Arizona Greenlee 47.08 47.08 Arizona Maricopa 94.17 47.08 Arizona Mohave 15.69 94.17 Arizona Navajo 7.85 47.08 Arizona Pima 7.85 47.08 Arizona Pima 7.85 47.08 Arizona Pinal 31.39 47.08 Arizona Yavapai 15.69 47.08 Arizona Yavapai 156.99 47.08 Arizona Yuma 156.94 47.08 Arizona Yuma 156.94 47.08 Arizona Yuma 156.99 47.08 Arizona Yuma 156.99 47.08 Arizona Yuma 156.99 47.08 Arkansas Arkansas 47.08 <td>\$62.78</td> <td></td> <td></td> <td></td>	\$62.78			
31.39 Arizona Gila 7.85 47.08 Arizona Graham 15.69 94.17 Arizona Greenlee 47.08 47.08 Arizona La Paz 31.39 94.17 Arizona Maricopa 94.17 31.39 Arizona Mohave 15.69 47.08 Arizona Pima 7.85 47.08 Arizona Pima 7.85 47.08 Arizona Pinal 31.39 47.08 Arizona Yavapai 15.69 47.08 Arizona Yavapai 15.69 47.08 Arizona Yuma 156.94 47.08 Arizona Yuma 156.94 47.08 Arkansas Arkansas 47.08 47.08 Arkansas Arkansas 47.08 47.08 Arkansas Ashley 47.08 47.08 Arkansas Benton 94.17 47.08 Arkansas Bradley 62.78 <td></td> <td></td> <td></td> <td></td>				
47.08 Arizona Graham 15.69 94.17 Arizona Greenlee 47.08 47.08 Arizona Marizopa 94.17 94.17 Arizona Maricopa 94.17 31.39 Arizona Mohave 15.69 47.08 Arizona Pima 7.85 47.08 Arizona Pinal 31.39 47.08 Arizona Pinal 31.39 47.08 Arizona Yavapai 15.69 47.08 Arizona Yuma 155.94 47.08 Arizona Yuma 155.94 47.08 Arkansas Arkansas 47.08 47.08 Arkansas Arkansas 47.08 47.08 Arkansas Baxter 47.08 47.08 Arkansas Baxter 47.08 47.08 Arkansas Bradley 62.78 47.08 Arkansas Bradley 62.78 47.08 Arkansas Bradley <td< td=""><td>31.39</td><td></td><td></td><td></td></td<>	31.39			
47.08 Arizona La Paz 31.39 47.08 Arizona Maricopa 94.17 31.39 Arizona Mohave 15.69 47.08 Arizona Navajo 7.85 47.08 Arizona Pima 7.85 47.08 Arizona Pinal 31.39 47.08 Arizona Santa Cruz 47.08 47.08 Arizona Yuma 15.69 62.78 Arizona Yuma 15.69 47.08 Arkansas Arkansas 47.08 47.08 Arkansas Arkansas 47.08 47.08 Arkansas Baxter 47.08 47.08 Arkansas Benton 94.17 47.08 Arkansas Bradley 62.78 47.08 Arkansas Bradley 62.78 47.08 Arkansas Bradley 62.78 47.08 Arkansas Calhoun 47.08 47.08 Arkansas Claiv				15.69
47.08 Arizona Maricopa 94.17 94.17 Arizona Mohave 15.69 47.08 Arizona Navajo 7.85 47.08 Arizona Pima 7.85 47.08 Arizona Pinal 31.39 47.08 Arizona Santa Cruz 47.08 47.08 Arizona Yavapai 15.69 62.78 Arizona Yuma 156.94 31.39 Arkansas Arkansas 47.08 47.08 Arkansas Ashley 47.08 47.08 Arkansas Baxter 47.08 47.08 Arkansas Baxter 47.08 47.08 Arkansas Benton 94.17 47.08 Arkansas Boone 47.08 47.08 Arkansas Bradley 62.78 94.17 Arkansas Calhoun 47.08 47.08 Arkansas Calhoun 47.08 31.39 Arkansas Clay		Arizona	Greenlee	47.08
94.17 Arizona Maricopa 94.17 31.39 Arizona Mohave 15.69 47.08 Arizona Pima 7.85 47.08 Arizona Pima 31.39 47.08 Arizona Yavapai 15.69 47.08 Arizona Yuma 156.94 47.08 Arizona Yuma 156.99 47.08 Arizona Yuma 156.99 47.08 Arkansas Arkansas 47.08 47.08 Arkansas Ashley 47.08 47.08 Arkansas Baxter 47.08 47.08 Arkansas Benton 94.17 47.08 Arkansas Boone 47.08 47.08 Arkansas Bradley 62.78 47.08 Arkansas Calhoun 47.08 47.08 Arkansas Calhoun 47.08 47.08 Arkansas Calhoun 47.08 47.08 Arkansas Clark 47.0				31.39
31.39 Arizona Moriave 15.09 47.08 Arizona Navajo 7.85 47.08 Arizona Pinal 31.39 47.08 Arizona Santa Cruz 47.08 47.08 Arizona Yavapai 15.69 47.08 Arizona Yuma 155.94 47.08 Arkansas Arkansas 47.08 47.08 Arkansas Arkansas 47.08 47.08 Arkansas Baxter 47.08 47.08 Arkansas Benton 94.17 47.08 Arkansas Bradley 62.78 47.08 Arkansas Bradley 62.78 47.08 Arkansas Calhoun 47.08 47.08 Arkansas Calhoun 47.08 47.08 Arkansas Calroll 47.08 47.08 Arkansas Claiv 47.08 47.08 Arkansas Claiv 47.08 47.08 Arkansas Claiv				
47.08 Arizona Pima 7.85 47.08 Arizona Pinal 31.39 47.08 Arizona Santa Cruz 47.08 47.08 Arizona Yavapai 15.69 62.78 Arizona Yuma 155.94 31.39 Arkansas Arkansas 47.08 47.08 Arkansas Baxter 47.08 47.08 Arkansas Benton 94.17 47.08 Arkansas Bradley 62.78 47.08 Arkansas Carloun 47.08 47.08 Arkansas Carloun 47.08 47.08 Arkansas Carroll 47.08 47.08 Arkansas Carroll 47.08 47.08 Arkansas Cloicot 31.39 47.08 Arkansas Clark 47.08 47.08 Arkansas Cloum 47.08 47.08 Arkansas Cloum 47.08 47.08 Arkansas Clebure				
47.08 Arizona Pinal 31.39 47.08 Arizona Santa Cruz 47.08 47.08 Arizona Yavapai 15.69 62.78 Arizona Yuma 156.94 31.39 Arkansas Arkansas 47.08 47.08 Arkansas Ashley 47.08 47.08 Arkansas Baxter 47.08 47.08 Arkansas Benton 94.17 47.08 Arkansas Bradley 62.78 94.17 Arkansas Calhoun 47.08 47.08 Arkansas Calhoun 47.08 47.08 Arkansas Ciroll 44.08 31.39 Arkansas Claroll 47.08 62.78 Arkansas Clar 47.08 62.78 Arkansas Clay 47.08 47.08 Arkansas Cleburne 47.08 47.08 Arkansas Cleveland 62.78 47.08 Arkansas Columbia				
47.08 Arizona Santa Cruz 47.08 47.08 Arizona Yavapaja 15.69 62.78 Arizona Yuma 156.94 31.39 Arkansas Arkansas 47.08 47.08 Arkansas Ashley 47.08 31.39 Arkansas Baxter 47.08 47.08 Arkansas Benton 94.17 47.08 Arkansas Boone 47.08 47.08 Arkansas Bradley 62.78 47.08 Arkansas Calhoun 47.08 47.08 Arkansas Calhoun 47.08 47.08 Arkansas Chicot 31.39 47.08 Arkansas Clark 47.08 62.78 Arkansas Clay 47.08 47.08 Arkansas Cleburne 47.08 47.08 Arkansas Cleveland 62.78 31.39 Arkansas Columbia 47.08 47.08 Arkansas Columbia<				
47.08 Arizona Yavapai 15.69 62.78 Arizona Yuma 156.94 31.39 Arkansas 47.08 47.08 Arkansas Ashley 47.08 47.08 Arkansas Baxter 47.08 47.08 Arkansas Benton 94.17 47.08 Arkansas Bradley 62.78 47.08 Arkansas Calhoun 47.08 47.08 Arkansas Calhoun 47.08 47.08 Arkansas Ciartoll 47.08 47.08 Arkansas Clicot 31.39 62.78 Arkansas Clark 47.08 62.78 Arkansas Clebure 47.08 47.08 Arkansas Clebure 47.08 47.08 Arkansas Cloumbia 47.08 47.08 Arkansas Columbia 47.08 47.08 Arkansas Craighead 47.08 47.08 Arkansas Crawford 47.08<				
62.78 Arizona Yuma 156.94 431.39 Arkansas 47.08 47.08 Arkansas 47.08 47.08 Arkansas Baxter 47.08 47.08 Arkansas Benton 94.17 47.08 Arkansas Bradley 62.78 47.08 Arkansas Calhoun 47.08 94.17 Arkansas Calhoun 47.08 47.08 Arkansas Carroll 47.08 62.78 Arkansas Clark 47.08 62.78 Arkansas Clay 47.08 47.08 Arkansas Clebure 47.08 47.08 Arkansas Cleburne 47.08 47.08 Arkansas Columbia 47.08 47.08 Arkansas Columbia 47.08 47.08 Arkansas Craighead 47.08 47.08 Arkansas Craighead 47.08 31.39 Arkansas Crawford 47.08 <tr< td=""><td></td><td></td><td></td><td>15.69</td></tr<>				15.69
31.39 Arkansas 47.08 47.08 Arkansas Ashley 47.08 47.08 Arkansas Baxter 47.08 47.08 Arkansas Benton 94.17 47.08 Arkansas Boone 47.08 47.08 Arkansas Bradley 62.78 94.17 Arkansas Calhoun 47.08 47.08 Arkansas Carroll 47.08 31.39 Arkansas Claic 31.39 62.78 Arkansas Clay 47.08 47.08 Arkansas Clay 47.08 47.08 Arkansas Cleburne 47.08 47.08 Arkansas Cleveland 62.78 31.39 Arkansas Columbia 47.08 47.08 Arkansas Conway 47.08 47.08 Arkansas Craighead 47.08 31.39 Arkansas Crawford 47.08 31.39 Arkansas Crittenden 47.08<		Arizona	Yuma	156.94
47.08 Arkansas Ashley 47.08 31.39 Arkansas Baxter 47.08 47.08 Arkansas Benton 94.17 47.08 Arkansas Boone 47.08 47.08 Arkansas Bradley 62.78 47.08 Arkansas Calhoun 47.08 47.08 Arkansas Chicot 31.39 62.78 Arkansas Clarv 47.08 62.78 Arkansas Clay 47.08 47.08 Arkansas Cleburne 47.08 47.08 Arkansas Cleveland 62.78 31.39 Arkansas Columbia 47.08 47.08 Arkansas Columbia 47.08 47.08 Arkansas Craighead 47.08 47.08 Arkansas Craighead 47.08 31.39 Arkansas Craighead 47.08 31.39 Arkansas Crawford 47.08 31.39 Arkansas C				47.08
47.08 Arkansas Benton 94.17 47.08 Arkansas Boone 47.08 47.08 Arkansas Bradley 62.78 94.17 Arkansas Calhoun 47.08 47.08 Arkansas Carroll 47.08 31.39 Arkansas Chicot 31.39 62.78 Arkansas Clark 47.08 62.78 Arkansas Cley 47.08 47.08 Arkansas Cleburne 47.08 94.17 Arkansas Colevalend 62.78 31.39 Arkansas Columbia 47.08 47.08 Arkansas Conway 47.08 47.08 Arkansas Craighead 47.08 31.39 Arkansas Crawford 47.08 31.39 Arkansas Crittenden 47.08 31.39 Arkansas Dallas 47.08				
47.08 Arkansas Boone 47.08 47.08 Arkansas Bradley 62.78 94.17 Arkansas Calhoun 47.08 94.708 Arkansas Carroll 47.08 13.39 Arkansas Chicot 31.39 62.78 Arkansas Clark 47.08 62.78 Arkansas Clay 47.08 47.08 Arkansas Cleburne 47.08 94.17 Arkansas Cleveland 62.78 31.39 Arkansas Columbia 47.08 47.08 Arkansas Conway 47.08 47.08 Arkansas Craighead 47.08 31.39 Arkansas Crawford 47.08 31.39 Arkansas Crawford 47.08 31.39 Arkansas Crittenden 47.08 31.39 Arkansas Crowford 47.08 47.08 Arkansas Crowford 47.08 47.08 Arkansas <				
47.08 Arkansas Bradley 62.78 94.17 Arkansas Calhoun 47.08 47.08 Arkansas Carroll 47.08 31.39 Arkansas Chicot 31.39 62.78 Arkansas Clark 47.08 62.78 Arkansas Clay 47.08 47.08 Arkansas Cleburne 47.08 94.17 Arkansas Cleveland 62.78 31.39 Arkansas Columbia 47.08 47.08 Arkansas Conway 47.08 47.08 Arkansas Craighead 47.08 31.39 Arkansas Crawford 47.08 31.39 Arkansas Crittenden 47.08 31.39 Arkansas Crittenden 47.08 47.08 Arkansas Dallas 47.08				
94.17 Arkansas Calhoun 47.08 47.08 Arkansas Carroll 47.08 31.39 Arkansas Chicot 31.39 62.78 Arkansas Clark 47.08 62.78 Arkansas Clay 47.08 47.08 Arkansas Cleburne 47.08 94.17 Arkansas Cleveland 62.78 31.39 Arkansas Columbia 47.08 47.08 Arkansas Conway 47.08 47.08 Arkansas Craighead 47.08 31.39 Arkansas Crawford 47.08 31.39 Arkansas Crittenden 47.08 31.39 Arkansas Cross 47.08 47.08 Arkansas Dallas 47.08				
47.08 Arkansas Carroll 47.08 31.39 Arkansas Chicot 31.39 62.78 Arkansas Clark 47.08 62.78 Arkansas Clay 47.08 47.08 Arkansas Cleburne 47.08 94.17 Arkansas Cleveland 62.78 31.39 Arkansas Columbia 47.08 47.08 Arkansas Conway 47.08 47.08 Arkansas Craighead 47.08 31.39 Arkansas Crawford 47.08 31.39 Arkansas Crittenden 47.08 31.39 Arkansas Cross 47.08 47.08 Arkansas Dallas 47.08				47.08
62.78 Arkansas Clark 47.08 62.78 Arkansas Clay 47.08 47.08 Arkansas Cleburne 47.08 94.17 Arkansas Cleveland 62.78 31.39 Arkansas Columbia 47.08 47.08 Arkansas Conway 47.08 47.08 Arkansas Craighead 47.08 31.39 Arkansas Crawford 47.08 31.39 Arkansas Crittenden 47.08 47.08 Arkansas Cross 47.08 47.08 Arkansas Dallas 47.08				47.08
62.78 Arkansas Clay 47.08 47.08 Arkansas Cleburne 47.08 94.17 Arkansas Cleveland 62.78 31.39 Arkansas Columbia 47.08 47.08 Arkansas Conway 47.08 47.08 Arkansas Craighead 47.08 31.39 Arkansas Crawford 47.08 31.39 Arkansas Crittenden 47.08 31.39 Arkansas Cross 47.08 47.08 Arkansas Dallas 47.08	31.39			31.39
47.08 Arkansas Cleburne 47.08 94.17 Arkansas Cleveland 62.78 31.39 Arkansas Columbia 47.08 47.08 Arkansas Conway 47.08 47.08 Arkansas Craighead 47.08 31.39 Arkansas Crawford 47.08 31.39 Arkansas Crittenden 47.08 47.08 Arkansas Dallas 47.08				47.08
94.17 Arkansas Cleveland 62.78 31.39 Arkansas Columbia 47.08 47.08 Arkansas Conway 47.08 47.08 Arkansas Craighead 47.08 31.39 Arkansas Crawford 47.08 31.39 Arkansas Crittenden 47.08 47.08 Arkansas Cross 47.08 47.08 Arkansas Dallas 47.08		Arkansas	Claburna	
31.39 Arkansas Columbia 47.08 47.08 Arkansas Conway 47.08 47.08 Arkansas Craighead 47.08 31.39 Arkansas Crawford 47.08 31.39 Arkansas Crittenden 47.08 31.39 Arkansas Cross 47.08 47.08 Arkansas Dallas 47.08				
47.08 Arkansas Conway 47.08 47.08 Arkansas Craighead 47.08 31.39 Arkansas Crawford 47.08 31.39 Arkansas Crittenden 47.08 31.39 Arkansas Cross 47.08 47.08 Arkansas Dallas 47.08				
47.08 Arkansas Craighead 47.08 31.39 Arkansas Crawford 47.08 31.39 Arkansas Crittenden 47.08 Arkansas Cross 47.08 Arkansas Dallas 47.08				
31.39 Arkansas Crawford 47.08 31.39 Arkansas Crittenden 47.08 31.39 Arkansas Cross 47.08 47.08 Arkansas Dallas 47.08				47.08
31.39 Arkansas Cross 47.08 47.08 Arkansas Dallas 47.08		Arkansas	Crawford	47.08
47.08 Arkansas Dallas 47.08				47.08
				47.08
02.70 AIRAIISAS Desila 31.39				
	02.70	AIRAIISAS	Desila	31.39

State	County	(Fee/acre/ YR)	State	County	(Fee/acre/ YR)
Arkansas	Drew	47.08	California	Madera	94.1
Arkansas	Faulkner	47.08	California	Marin	94.1
Arkansas	Franklin	47.08	California	Mariposa	31.3
rkansas	Fulton	31.39	California	Mendocino	62.7
Arkansas	Garland	62.78	California	Merced	156.9
Arkansas	Grant	47.08	California	Modoc	31.3
Arkansas	Greene	47.08	California	Mono	47.0
Arkansas	Hempstead	47.08	California	Monterey	94.1
Arkansas	Hot Spring	47.08	California	Napa	627.7
Arkansas	Howard	47.08	California	Nevada	94.1
Arkansas	Independence	31.39	California	Orange	313.8
Arkansas	Izard	31.39	California	Placer	156.9
Arkansas	Jackson	31.39	California	Plumas	31.3
rkansas	Jefferson	31.39	California	Riverside	156.9
rkansas	Johnson	62.78	California	Sacramento	156.9
rkansas	Lafayette	31.39	California	San Benito	62.7
rkansas	Lawrence	47.08	California	San Bernardino	62.7
rkansas	Lee	31.39	California	San Diego	313.8
rkansas	Lincoln	31.39	California	San Francisco	941.6
rkansas	Little River	31.39	California	San Joaquin	313.8
rkansas	Logan	47.08	California	San Luis Obispo	94.1
rkansas	Lonoke	47.08	California	San Mateo	156.9
rkansas	Madison	47.08	California	Santa Barbara	94.1
rkansas	Marion	47.08	California	Santa Clara	94.1
rkansas	Miller	31.39	California	Santa Cruz	313.8
rkansas	Mississippi	47.08	California	Shasta	47.0
rkansas	Monroe	31.39	California	Sierra	47.0
rkansas	Montgomery	47.08	California	Siskiyou	47.0
rkansas	Nevada	31.39	California	Solano	156.9
rkansas	Newton	47.08	California	Sonoma	313.8
rkansas	Ouachita	47.08	California	Stanislaus	156.9
rkansas	Perry	47.08	California	Sutter	156.9
rkansas	Phillips	31.39	California	Tehama	47.0
rkansas	Pike	47.08	California	Trinity	31.3
rkansas	Poinsett	47.08	California	Tulare	156.9
rkansas	Polk	47.08	California	Tuolumne	47.0
rkansas	Pope	62.78	California	Ventura	313.8
rkansas	Prairie	31.39	California	Yolo	94.1
rkansas	Pulaski	47.08	California	Yuba	94.1
rkansas	Randolph	47.08	Colorado	Adams	31.3
rkansas	Saline	62.78	Colorado	Alamosa	31.3
rkansas	Scott	47.08	Colorado	Arapahoe	31.3
rkansas	Searcy	31.39	Colorado	Archuleta	47.0
rkansas	Sebastian	62.78	Colorado	Baca	7.8
rkansas	Sevier	47.08	Colorado	Bent	15.6
rkansas	Sharp	31.39	Colorado	Boulder	313.8
rkansas	St. Francis	31.39	Colorado	Broomfield*	31.0
rkansas	Stone	31.39	Colorado	Chaffee	62.7
rkansas	Union	62.78	Colorado	Cheyenne	15.6
rkansas	Van Buren	47.08	Colorado	Clear Creek	47.0
rkansas	Washington	94.17	Colorado	Conejos	31.0
rkansas	White	47.08	Colorado	Costilla	15.6
rkansas	Woodruff	31.39	Colorado	Crowley	7.8
rkansas	Yell	47.08	Colorado	Custer	47.0
alifornia	Alameda	94.17	Colorado	Delta	62.
alifornia	Alpine	62.78	Colorado	Denver*	31.0
alifornia	Amador	62.78	Colorado	Dolores	31.0
alifornia	Butte	156.94	Colorado	Douglas	94.
alifornia	Calaveras	47.08	Colorado	Eagle	47.0
alifornia	Colusa	94.17	Colorado	El Paso	31.
alifornia	Contra Costa	313.89	Colorado	Elbert	31.0
alifornia	Del Norte	156.94	Colorado	Fremont	31.0
alifornia	El Dorado	94.17	Colorado	Garfield	47.0
alifornia	Fresno	94.17	Colorado	Gilpin	94.
alifornia	Glenn	62.78	Colorado	Grand	31.0
alifornia	Humboldt	31.39	Colorado	Gunnison	47.0
alifornia	Imperial	94.17	Colorado	Hinsdale	94.
alifornia	Inyo	31.39	Colorado	Huerfano	15.0
alifornia	Kern	47.08	Colorado	Jackson	15.6
alifornia	Kings	94.17	Colorado	Jefferson	156.9
alifornia	Lake	156.94	Colorado	Kiowa	7.8
alifornia	Lassen	31.39	Colorado	Kit Carson	15.
alifornia	Los Angeles	627.77	Colorado	La Plata	31.

State	County	(Fee/acre/ YR)	State	County	(Fee/acre/ YR)
Colorado	Lake	47.08	Florida	Jefferson	47.08
Colorado	Larimer	62.78	Florida	Lafayette	47.08
Colorado	Las Animas	7.85	Florida	Lake	156.94
Colorado	Lincoln	7.85	Florida	Lee	94.17
Colorado	Logan	15.69	Florida	Leon	62.78
Colorado	Mesa	47.08	Florida	Levy	62.78
Colorado	Mineral	47.08	Florida	Liberty	47.08
Colorado	Moffat	15.69	Florida	Madison	47.08
Colorado	Montezuma	15.69 31.39	Florida Florida	Manatee Marion	94.17 156.94
Colorado	Montrose Morgan	31.39	Florida	Martin	94.17
Colorado	Otero	15.69	Florida	Monroe	627.77
Colorado	Ouray	47.08	Florida	Nassau	156.94
Colorado	Park	31.39	Florida	Okaloosa	94.17
Colorado	Phillips	31.39	Florida	Okeechobee	62.78
Colorado	Pitkin	156.94	Florida	Orange	156.94
Colorado	Prowers	15.69	Florida	Osceola	47.08
Colorado	Pueblo	15.69	Florida	Palm Beach	94.17
Colorado	Rio Blanco	31.39	Florida	Pasco	156.94
Colorado	Rio Grande	47.08	Florida	Pinellas	941.66
Colorado	Routt	62.78	Florida	Polk	94.17
Colorado	Saguache	31.39	Florida	Putnam	62.78
Colorado	San Juan *	31.39	Florida	Santa Rosa	94.17
Colorado	San Miguel	31.39	Florida	Sarasota	94.17
Colorado	Sedgwick	31.39	Florida	Seminole	156.94
Colorado	Summit	47.08	Florida	St. Johns	156.94
Colorado	Teller	47.08	Florida	St. Lucie	94.17
Colorado	Washington	15.69	Florida	Sumter	62.78
Colorado	Weld Yuma	47.08	Florida Florida	Suwannee	94.17
Colorado Connecticut	Fairfield	15.69 941.66		Taylor	47.08 47.08
Connecticut	Hartford	627.77	Florida Florida	Union Volusia	156.94
Connecticut	Litchfield	313.89	Florida	Wakulla	94.17
Connecticut	Middlesex	313.89	Florida	Walton	62.78
Connecticut	New Haven	627.77	Florida	Washington	62.78
Connecticut	New London	313.89	Georgia	Appling	47.08
Connecticut	Tolland	156.94	Georgia	Atkinson	47.08
Connecticut	Windham	313.89	Georgia	Bacon	62.78
Delaware	Kent	94.17	Georgia	Baker	47.08
Delaware	New Castle	156.94	Georgia	Baldwin	62.78
Delaware	Sussex	156.94	Georgia	Banks	156.94
Florida	Alachua	94.17	Georgia	Barrow	156.94
Florida	Baker	156.94	Georgia	Bartow	94.17
Florida	Bay	94.17	Georgia	Ben Hill	47.08
Florida	Bradford	62.78	Georgia	Berrien	47.08
Florida	Brevard	62.78	Georgia	Bibb	62.78
Florida	Broward	627.77	Georgia	Bleckley	47.08
Florida	Calhoun	47.08	Georgia	Brantley	47.08
Florida	Charlotte	47.08	Georgia	Brooks	47.08
Florida	Citrus	62.78	Georgia	Bryan	47.08
Florida	Clay	62.78	Georgia	Bulloch	47.08
Florida	Collier	94.17 47.08	Georgia	Butts	47.08 62.78
Florida	Dade	313.89	GeorgiaGeorgia	Calhoun	47.08
Florida	DeSoto	62.78	Georgia	Camden	47.08
Florida	Dixie	47.08	Georgia	Candler	47.08
Florida	Duval	156.94	Georgia	Carroll	156.94
Florida	Escambia	62.78	Georgia	Catoosa	156.94
Florida	Flagler	47.08	Georgia	Charlton	62.78
Florida	Franklin	31.39	Georgia	Chatham	62.78
Florida	Gadsden	62.78	Georgia	Chattahoochee	47.08
Florida	Gilchrist	62.78	Georgia	Chattooga	47.08
Florida	Glades	47.08	Georgia	Cherokee	313.89
Florida	Gulf	62.78	Georgia	Clarke	156.94
Florida	Hamilton	47.08	Georgia	Clay	31.39
Florida	Hardee	62.78	Georgia	Clayton	156.94
Florida	Hendry	156.94	Georgia	Clinch	47.08
Florida	Hernando	156.94	Georgia	Cobb	313.89
Florida	Highlands	62.78	Georgia	Coffee	47.08
Florida	Hillsborough	156.94	Georgia	Colquitt	47.08
Florida	Holmes	47.08	Georgia	Columbia	156.94
					47.08 156.94
Florida	Indian River	94.17 47.08	Georgia	Cook	

Federal Energy Regulatory Commission

Pt. 11, App. A

State	County	(Fee/acre/ YR)	State	County	(Fee/acre/ YR)
Georgia	Crawford	62.78	Georgia	Pickens	156.9
Georgia		47.08	Georgia	Pierce	47.0
eorgia	Dade	62.78	Georgia	Pike	94.1
Georgia	Dawson	156.94	Georgia	Polk	62.7
Georgia	Decatur	47.08	Georgia	Pulaski	47.0
Georgia	DeKalb	313.89	Georgia	Putnam	94.1
Georgia	Dodge	31.39	Georgia	Quitman	47.0
Georgia	Dooly	47.08	Georgia	Rabun	156.9
Georgia	Dougherty	47.08	Georgia	Randolph	31.3
Georgia		156.94	Georgia	Richmond	94.1
Georgia		47.08	Georgia	Rockdale	156.9
Georgia		47.08	Georgia	Schley	47.0
Georgia		47.08	Georgia	Screven	47.0
Georgia		62.78	Georgia	Seminole	47.0
Georgia		31.39	Georgia	Spalding	156.9
Georgia		47.08	Georgia	Stephens	156.9
Georgia		94.17	Georgia	Stewart	47.0
Georgia		156.94	Georgia	Sumter	47.0
Georgia		94.17	Georgia	Talbot	47.0
Georgia		313.89	Georgia	Taliaferro	47.0
Georgia		156.94	Georgia	Tattnall	62.7
Georgia		156.94	Georgia	Taylor	47.0
Georgia		156.94 47.08	Georgia	Telfair	47.0
Georgia		47.08 47.08	Georgia	Terrell	47.0
Georgia			Georgia	Thomas	47.0
Georgia		156.94 47.08	Georgia	Tift	62.7
Reorgia			Georgia		47.0
Georgia		94.17	Georgia	Towns	156.9 47.0
Georgia		313.89 156.94	Georgia		
ieorgia		156.94	Georgia	Troup	47.0 47.0
Georgia			Georgia	Turner	
ieorgia		31.39 94.17	Georgia	Twiggs	47.0
ieorgia		62.78	Georgia	Union	156.9 62.7
Reorgia		94.17	GeorgiaGeorgia	Upson Walker	94.1
Georgia		62.78	Georgia	Walton	313.8
Georgia		156.94	_ ~ .	Ware	47.0
Georgia	1	62.78	Georgia	Warren	47.0
Seorgia Seorgia		47.08	Georgia Georgia	Washington	47.0
Georgia		156.94	Georgia	Wayne	47.0
Georgia		62.78		Webster	47.0
Georgia		47.08	Georgia Georgia	Wheeler	31.3
eorgia		47.08	Georgia	White	156.9
eorgia		47.08	Georgia	Whitfield	62.7
Georgia		47.08	Georgia	Wilcox	47.0
eorgia		62.78	Georgia	Wilkes	47.0
eorgia	l .	62.78	Georgia	Wilkinson	47.0
Georgia		31.39	Georgia	Worth	47.0
eorgia		47.08	Hawaii	Hawaii	94.1
Georgia		47.08	Hawaii	Honolulu	313.8
Georgia		62.78	Hawaii	Kauai	156.9
eorgia		94.17	Hawaii	Maui	156.9
ieorgia		47.08	Idaho	Ada	94.1
eorgia		62.78	Idaho	Adams	15.6
eorgia		156.94	Idaho	Bannock	31.3
ieorgia	laa '	47.08	Idaho	Bear Lake	31.3
ieorgia		156.94	Idaho	Benewah	31.3
eorgia		47.08	Idaho	Bingham	31.0
ieorgia		62.78	Idaho	Blaine	47.0
ieorgia		47.08	Idaho	Boise	31.3
eorgia		62.78	Idaho	Bonner	94.
eorgia		47.08	Idaho	Bonneville	94. 47.0
eorgia		47.08 47.08	Idaho	Boundary	62.7
eorgia		62.78	Idaho	Butte	31.0
eorgia		47.08	Idaho	Camas	31.3
ieorgiaieorgia		94.17	Idaho	Canyon	156.9
		94.17	Idaho	Caribou	31.3
eorgia		94.17	Idaho	Cassia	
ieorgiaieorgia		156.94	Idaho	Clark	31.3 31.3
		156.94	Idaho	Clark	47.0
ieorgiaieorgia		94.17	Idaho		47.0
ieorgiaieorgia		94.17 313.89	IdahoIdaho	Custer	47.0 31.3
	F	313.69	IUdIIU	. LIHOIE	.51 :

	State	County	(Fee/acre/ YR)	State	County	(Fee/acre/ YR)
Idaho		Fremont	31.39	Illinois	Lawrence	47.08
		Gem	31.39	Illinois	Lee	94.17
		Gooding	94.17	Illinois	Livingston	94.17
		Idaho	31.39	Illinois	Logan	94.17
		Jefferson	47.08	Illinois	Macon	94.17
		Jerome	62.78	Illinois	Macoupin	62.78
		Kootenai Latah	62.78 47.08	Illinois	Madison Marion	62.78 47.08
		Lemhi	31.39	Illinois	Marshall	94.17
		Lewis	31.39	Illinois	Mason	62.78
		Lincoln	31.39	Illinois	Massac	31.39
		Madison	62.78	Illinois	McDonough	62.78
Idaho		Minidoka	62.78	Illinois	McHenry	156.94
Idaho		Nez Perce	31.39	Illinois	McLean	94.17
		Oneida	31.39	Illinois	Menard	62.78
		Owyhee	31.39	Illinois	Mercer	62.78
		Payette	47.08	Illinois	Monroe	94.17
		Power	31.39	Illinois	Montgomery	62.78
		Shoshone	94.17	Illinois	Morgan	62.78
		Teton Twin Falls	62.78 62.78	Illinois	Moultrie Ogle	94.17 94.17
		Valley	47.08	Illinois	Peoria	94.17
		Washington	31.39	Illinois	Perry	47.08
		Adams	62.78	Illinois	Piatt	94.17
		Alexander	47.08	Illinois	Pike	47.08
		Bond	62.78	Illinois	Pope	31.39
Illinois .		Boone	94.17	Illinois	Pulaski	47.08
Illinois .		Brown	47.08	Illinois	Putnam	94.17
Illinois .		Bureau	94.17	Illinois	Randolph	62.78
		Calhoun	47.08	Illinois	Richland	47.08
		Carroll	62.78	Illinois	Rock Island	94.17
		Cass	62.78	Illinois	Saline	47.08
		Champaign	94.17	Illinois	Sangamon	94.17
		Christian	94.17	Illinois	Schuyler	47.08
		Clark	62.78	Illinois	Scott	62.78
		Clay	47.08 62.78	Illinois	Shelby	62.78 94.17
		Clinton	94.17	Illinois	St. Clair	94.17
		Cook	313.89	Illinois	Stephenson	62.78
		Crawford	47.08	Illinois	Tazewell	94.17
		Cumberland	62.78	Illinois	Union	62.78
		De Witt	94.17	Illinois	Vermilion	62.78
		DeKalb	156.94	Illinois	Wabash	47.08
Illinois .		Douglas	94.17	Illinois	Warren	94.17
Illinois .		DuPage	156.94	Illinois	Washington	62.78
Illinois .		Edgar	62.78	Illinois	Wayne	31.39
		Edwards	47.08	Illinois	White	47.08
		Effingham	62.78	Illinois	Whiteside	94.17
		Fayette	47.08	Illinois	Will	156.94
		Ford	94.17	Illinois	Williamson	62.78
		Franklin	47.08 62.78	Illinois	Woodford	94.17
		Gallatin	47.08	IllinoisIndiana	Woodford	94.17 94.17
		Greene	47.08	Indiana	Allen	94.17
		Grundy	94.17	Indiana	Bartholomew	94.17
		Hamilton	47.08	Indiana	Benton	62.78
		Hancock	94.17	Indiana	Blackford	62.78
		Hardin	47.08	Indiana	Boone	94.17
Illinois .		Henderson	62.78	Indiana	Brown	94.17
Illinois .		Henry	62.78	Indiana	Carroll	94.17
Illinois .		Iroquois	62.78	Indiana	Cass	62.78
		Jackson	47.08	Indiana	Clark	94.17
		Jasper	62.78	Indiana	Clay	62.78
		Jefferson	47.08	Indiana	Clinton	94.17
		Jersey	62.78	Indiana	Crawford	47.08
		Jo Daviess	62.78	Indiana	Daviess	62.78
		Johnson	47.08	Indiana	Dearborn	94.17
		Kane	156.94	Indiana	Decatur	94.17
		Kankakee	94.17	Indiana	DeKalb	62.78
		Kendall	156.94 62.78	IndianaIndiana	Delaware	94.17 62.78
		Knox La Salle	94.17	Indiana	Dubois	156.94

Federal Energy Regulatory Commission

Pt. 11, App. A

	State	County	(Fee/acre/ YR)	State	County	(Fee/acre/ YR)
Indiana		Floyd	94.17	lowa	Allamakee	47.0
ndiana		Fountain	62.78	lowa	Appanoose	31.3
ndiana		Franklin	62.78	lowa	Audubon	47.0
		Fulton	62.78	lowa	Benton	62.7
		Gibson	62.78	lowa	Black Hawk	94.1
		Grant	94.17	lowa	Boone	62.7
		Greene	62.78	lowa	Bremer	94.1
		Hamilton	156.94 94.17	lowa	Buchanan	62.78 62.78
		Hancock	94.17	lowalowa	Buena Vista Butler	62.7
		Hendricks	94.17	lowa	Calhoun	62.7
		Henry	94.17	lowa	Carroll	62.7
		Howard	94.17	lowa	Cass	47.0
		Huntington	62.78	lowa	Cedar	62.7
ndiana		Jackson	62.78	lowa	Cerro Gordo	62.7
ndiana		Jasper	62.78	lowa	Cherokee	62.7
		Jay	94.17	lowa	Chickasaw	62.7
		Jefferson	62.78	lowa	Clarke	31.3
		Jennings	62.78	lowa	Clay	62.7
		Johnson	156.94	lowa	Clayton	62.7
		Knox	62.78	lowa	Clinton	62.7
		Kosciusko	94.17	lowa	Crawford	62.7
		LaGrange	94.17 94.17	lowa	Dallas Davis	94.1 31.3
		Lake LaPorte	94.17	lowalowa	Decatur	31.3
		Lawrence	47.08	lowa	Delaware	62.7
		Madison	94.17	lowa	Des Moines	62.7
		Marion	156.94	lowa	Dickinson	62.7
		Marshall	62.78	lowa	Dubuque	62.7
		Martin	62.78	lowa	Emmet	62.7
		Miami	62.78	lowa	Fayette	62.7
		Monroe	62.78	lowa	Floyd	62.7
		Montgomery	62.78	lowa	Franklin	62.7
ndiana		Morgan	94.17	lowa	Fremont	47.0
ndiana		Newton	62.78	lowa	Greene	62.7
ndiana		Noble	94.17	lowa	Grundy	94.1
		Ohio	94.17	lowa	Guthrie	47.0
		Orange	62.78	lowa	Hamilton	62.7
		Owen	62.78	lowa	Hancock	62.7
		Parke	62.78	lowa	Hardin	62.7
		Perry	47.08	lowa	Harrison	47.0
		Pike	62.78 94.17	lowalowa	Henry	62.7 62.7
		Posey	62.78	lowa	Humboldt	62.7
		Pulaski	62.78	lowa	Ida	62.7
		Putnam	62.78	lowa	lowa	47.0
		Randolph	62.78	lowa	Jackson	47.0
		Ripley	94.17	lowa	Jasper	62.7
		Rush	94.17	lowa	Jefferson	47.0
ndiana		Scott	62.78	lowa	Johnson	62.7
ndiana		Shelby	94.17	lowa	Jones	62.7
		Spencer	62.78	lowa	Keokuk	47.0
		St. Joseph	94.17	lowa	Kossuth	62.7
		Starke	62.78	lowa	Lee	47.0
		Steuben	62.78	lowa	Linn	94.1
		Sullivan	62.78	lowa	Louisa	62.7
		Switzerland	62.78	lowa	Lucas	31.3
		Tippecanoe	94.17 94.17	lowa	Lyon	62.7 47.0
		Union	62.78	lowalowa	Madison Mahaska	47.0
		Vanderburgh	94.17	lowa	Marion	47.0
		Vermillion	62.78	lowa	Marshall	62.7
		Vigo	62.78	lowa	Mills	47.0
		Wabash	94.17	lowa	Mitchell	62.7
		Warren	62.78	lowa	Monona	47.0
		Warrick	62.78	lowa	Monroe	31.3
		Washington	62.78	lowa	Montgomery	47.0
		Wayne	62.78	lowa	Muscatine	62.7
		Wells	62.78	lowa	O'Brien	94.1
		White	94.17	lowa	Osceola	62.7
		Whitley	94.17	lowa	Page	47.0
		Adair	47.08	lowa	Palo Alto	62.7
		Adams	47.08	lowa	Plymouth	62.7

State	County	(Fee/acre/ YR)	State	County	(Fee/acre/ YR)
lowa	Pocahontas	62.78	Kansas	Labette	31.39
lowa	Polk	62.78	Kansas	Lane	15.69
lowa	Pottawattamie	62.78	Kansas	Leavenworth	47.08
lowa	Poweshiek	47.08	Kansas	Lincoln	15.69
lowa	Ringgold	31.39	Kansas	Linn	31.39
lowa	Sac	62.78	Kansas	Logan	15.69
lowalowa	Scott	94.17 62.78	Kansas Kansas	Lyon Marion	31.39 31.39
lowa	Sioux	94.17	Kansas	Marshall	31.39
lowa	Story	62.78	Kansas	McPherson	31.39
lowa	Tama	62.78	Kansas	Meade	15.69
owa	Taylor	31.39	Kansas	Miami	47.08
lowa	Union	47.08	Kansas	Mitchell	31.39
lowa	Van Buren	31.39	Kansas	Montgomery	31.39
lowa	Wapello	47.08	Kansas	Morris	31.39
lowa	Warren	47.08	Kansas	Morton	15.69
lowa	Washington	62.78	Kansas	Nemaha	31.39
lowa	Wayne	31.39	Kansas	Neosho	31.39
lowa	Webster	62.78	Kansas	Ness	15.69
lowa	Winnebago	62.78 47.08	Kansas	Norton	15.69
lowalowa	Winneshiek Woodbury	47.08	Kansas Kansas	Osage Osborne	31.39 15.69
lowa	Worth	62.78	Kansas	Ottawa	15.69
lowa	Wright	62.78	Kansas	Pawnee	15.69
Kansas	Allen	31.39	Kansas	Phillips	15.69
Kansas	Anderson	31.39	Kansas	Pottawatomie	31.39
Kansas	Atchison	31.39	Kansas	Pratt	31.39
Kansas	Barber	15.69	Kansas	Rawlins	15.69
Kansas	Barton	15.69	Kansas	Reno	31.39
Kansas	Bourbon	31.39	Kansas	Republic	31.39
Kansas	Brown	31.39	Kansas	Rice	31.39
Kansas	Butler	31.39	Kansas	Riley	31.39
Kansas	Chase	15.69	Kansas	Rooks	15.69
Kansas	Chautauqua	15.69	Kansas	Rush	15.69
Kansas	Cherokee	31.39	Kansas	Russell	15.69
Kansas Kansas	Cheyenne	15.69 15.69	Kansas	Saline	31.39 15.69
Kansas	Clark	31.39	Kansas Kansas	ScottSedgwick	31.39
Kansas	Cloud	15.69	Kansas	Seward	31.39
Kansas	Coffey	31.39	Kansas	Shawnee	47.08
Kansas	Comanche	15.69	Kansas	Sheridan	15.69
Kansas	Cowley	31.39	Kansas	Sherman	15.69
Kansas	Crawford	31.39	Kansas	Smith	31.39
Kansas	Decatur	15.69	Kansas	Stafford	31.39
Kansas	Dickinson	31.39	Kansas	Stanton	15.69
Kansas	Doniphan	47.08	Kansas	Stevens	31.39
Kansas	Douglas	62.78	Kansas	Sumner	31.39
Kansas	Edwards	15.69	Kansas	Thomas	15.69
Kansas	Elk	15.69	Kansas	Trego	15.69
Kansas	Ellis	15.69	Kansas	Wabaunsee	31.39
Kansas	Ellsworth	15.69	Kansas	Wallace	15.69
Kansas	Finney	15.69	Kansas	Washington	31.39
Kansas Kansas	Ford	15.69 31.39	Kansas Kansas	Wichita Wilson	15.69 31.39
Kansas	Geary	31.39	Kansas	Woodson	15.69
Kansas	Gove	15.69	Kansas	Wyandotte	156.94
Kansas	Graham	15.69	Kentucky	Adair	47.08
Kansas	Grant	31.39	Kentucky	Allen	47.08
Kansas	Gray	31.39	Kentucky	Anderson	62.78
Kansas	Greeley	15.69	Kentucky	Ballard	47.08
Kansas	Greenwood	15.69	Kentucky	Barren	47.08
Kansas	Hamilton	15.69	Kentucky	Bath	47.08
Kansas	Harper	15.69	Kentucky	Bell	47.08
Kansas	Harvey	31.39	Kentucky	Boone	94.17
Kansas	Haskell	31.39	Kentucky	Bourbon	94.17
Kansas	Hodgeman	15.69	Kentucky	Boyd	47.08
Kansas	Jackson	31.39	Kentucky	Boyle	62.78
Kansas	Jefferson	31.39	Kentucky	Bracken	47.08
Kansas	Jewell	31.39	Kentucky	Breathitt	31.39
	Johnson	62.78	Kentucky	Breckinridge	47.08
Kansas		4= 00		D. III	~
Kansas Kansas	Kearny	15.69 31.39	KentuckyKentucky	Bullitt Butler	94.17 47.08

State	County	(Fee/acre/ YR)	State	County	(Fee/acre/ YR)
Kentucky	Calloway	47.08	Kentucky	Nicholas	47.0
Kentucky	Campbell	156.94	Kentucky	Ohio	47.0
Kentucky	Carlisle	47.08	Kentucky	Oldham	156.9
Kentucky	Carroll	62.78	Kentucky	Owen	47.0
Kentucky	Carter	47.08	Kentucky	Owsley	47.0
Kentucky	Casey	31.39	Kentucky	Pendleton	47.0
Kentucky Kentucky	Christian	47.08 62.78	Kentucky Kentucky	Perry	31.3 31.3
Kentucky	Clay	31.39	Kentucky	Powell	47.0
Kentucky	Clinton	47.08	Kentucky	Pulaski	47.0
Kentucky	Crittenden	31.39	Kentucky	Robertson	31.3
Kentucký	Cumberland	31.39	Kentucky	Rockcastle	47.0
Kentucky	Daviess	62.78	Kentucky	Rowan	47.0
Kentucky	Edmonson	31.39	Kentucky	Russell	62.7
Kentucky	Elliott	31.39	Kentucky	Scott	94.1
Kentucky	Estill	31.39	Kentucky	Shelby	94.1
Kentucky	Fayette	156.94	Kentucky	Simpson	62.7
Kentucky	Fleming	47.08	Kentucky	Spencer	94.1
Kentucky	Floyd	47.08 62.78	Kentucky	Taylor	47.08 47.08
Kentucky Kentucky	Franklin Fulton	47.08	Kentucky Kentucky	Todd	47.0
Kentucky	Gallatin	62.78	Kentucky	Trimble	47.0
Kentucky	Garrard	47.08	Kentucky	Union	47.0
Kentucky	Grant	94.17	Kentucky	Warren	62.7
Kentucky	Graves	47.08	Kentucky	Washington	47.0
Kentucký	Grayson	47.08	Kentucky	Wayne	62.7
Kentucky	Green	47.08	Kentucky	Webster	47.0
Kentucky	Greenup	31.39	Kentucky	Whitley	47.0
Kentucky	Hancock	47.08	Kentucky	Wolfe	31.3
Kentucky	Hardin	62.78	Kentucky	Woodford	156.9
Kentucky	Harlan	62.78	Louisiana	Acadia	47.0
Kentucky	Harrison	47.08	Louisiana	Allen	31.3
Kentucky	Hart	47.08	Louisiana	Ascension	94.1
Kentucky	Henderson	62.78	Louisiana Louisiana	Assumption	47.0
Kentucky Kentucky	Henry Hickman	62.78 47.08	Louisiana	Avoyelles Beauregard	47.08 47.08
Kentucky	Hopkins	47.08	Louisiana	Bienville	47.0
Kentucky	Jackson	31.39	Louisiana	Bossier	47.0
Kentucky	Jefferson	156.94	Louisiana	Caddo	47.0
Kentucky	Jessamine	94.17	Louisiana	Calcasieu	47.0
Kentucký	Johnson	47.08	Louisiana	Caldwell	47.0
Kentucky	Kenton	156.94	Louisiana	Cameron	47.0
Kentucky	Knott	47.08	Louisiana	Catahoula	31.3
Kentucky	Knox	47.08	Louisiana	Claiborne	47.0
Kentucky	Larue	62.78	Louisiana	Concordia	31.3
Kentucky	Laurel	62.78	Louisiana	De Soto	47.0
Kentucky	Lawrence	31.39	Louisiana	East Baton Rouge	94.1
Kentucky	Lee	31.39	Louisiana	East Carroll	31.3
Kentucky	Leslie	31.39	Louisiana	East Feliciana	62.7
Kentucky Kentucky	Letcher	31.39 31.39	Louisiana Louisiana	Evangeline Franklin	47.0 31.3
Kentucky	Lincoln	47.08	Louisiana	Grant	47.0
Kentucky	Livingston	31.39	Louisiana	Iberia	62.7
Kentucky	Logan	47.08	Louisiana	Iberville	47.0
Kentucky	Lyon	31.39	Louisiana	Jackson	94.1
Centucky	Madison	62.78	Louisiana	Jefferson	62.7
Kentucký	Magoffin	31.39	Louisiana	Jefferson Davis	31.3
Kentucky	Marion	47.08	Louisiana	La Salle	47.0
Kentucky	Marshall	47.08	Louisiana	Lafayette	94.1
Centucky	Martin	15.69	Louisiana	Lafourche	47.0
Centucky	Mason	62.78	Louisiana	Lincoln	62.7
Centucky	McCracken	47.08	Louisiana	Livingston	94.1
Centucky	McCreary	62.78	Louisiana	Madison	31.3
Centucky	McLean	47.08	Louisiana	Morehouse	31.3
KentuckyKentucky	Meade Menifee	62.78 62.78	Louisiana Louisiana	Natchitoches	47.0
Centucky	Mercer	94.17	Louisiana	Ouachita	1,569.4 47.0
Kentucky	Metcalfe	94.17 47.08	Louisiana	Plaguemines	94.1
Kentucky	Monroe	47.08 47.08	Louisiana	Pointe Coupee	94.1 47.0
Kentucky	Montgomery	62.78	Louisiana	Rapides	47.0
Kentucky	Morgan	31.39	Louisiana	Red River	31.3
Kentucky	Muhlenberg	47.08	Louisiana	Richland	31.3
	Nelson	62.78	Louisiana		62.7

State	County	(Fee/acre/ YR)	State	County	(Fee/acre/ YR)
Louisiana	St. Bernard	156.94	Massachusetts	Worcester	313.89
Louisiana	St. Charles	156.94	Michigan	Alcona	62.78
Louisiana	St. Helena	62.78	Michigan	Alger	47.08
Louisiana	St. James	47.08	Michigan	Allegan	94.17
Louisiana	St. John the Baptist	94.17	Michigan	Alpena	62.78
Louisiana	St. Landry	47.08	Michigan	Antrim	94.17
Louisiana	St. Martin	47.08	Michigan	Arenac	62.78
Louisiana	St. Mary	47.08	Michigan	Baraga	31.39
Louisiana	St. Tammany	156.94	Michigan	Barry	94.17
Louisiana	Tangipahoa	94.17	Michigan	Bay	94.17
Louisiana	Tensas	31.39	Michigan	Benzie	94.17
Louisiana	Terrebonne	47.08	Michigan	Berrien	156.94
Louisiana	Union	62.78	Michigan	Branch	62.78
Louisiana	Vermilion	47.08	Michigan	Calhoun	62.78
Louisiana	Vernon	47.08	Michigan	Cass	62.78
Louisiana	Washington	62.78	Michigan	Charlevoix	94.17
Louisiana	Webster	94.17	Michigan	Cheboygan	62.78
Louisiana	West Baton Rouge	62.78	Michigan	Chippewa	47.08
Louisiana	West Carroll	47.08	Michigan	Clare	62.78
Louisiana	West Feliciana	47.08	Michigan	Clinton	62.78
Louisiana	Winn	47.08	Michigan	Crawford	94.17
Maine	Androscoggin	62.78	Michigan	Delta	47.08
Maine	Aroostook	31.39	Michigan	Dickinson	47.08
Maine	Cumberland	156.94	Michigan	Eaton	94.17
Maine	Franklin	47.08	Michigan	Emmet	94.17
Maine	Hancock	62.78	Michigan	Genesee	156.94
Maine	Kennebec	62.78	Michigan	Gladwin	62.78
Maine	Knox	94.17	Michigan	Gogebic	47.08
Maine	Lincoln	94.17	Michigan	Grand Traverse	156.94
Maine	Oxford	62.78	Michigan	Gratiot	62.78
Maine	Penobscot	47.08	Michigan	Hillsdale	62.78
Maine	Piscataquis	31.39	Michigan	Houghton	47.08
Maine	Sagadahoc	94.17	Michigan	Huron	62.78
Maine	Somerset	47.08	Michigan	Ingham	94.17
Maine	Waldo	47.08	Michigan	Ionia	94.17
Maine	Washington	31.39	Michigan	losco	62.78
Maine	York	156.94	Michigan	Iron	47.08
Maryland	Allegany	62.78	Michigan	Isabella	62.78
Maryland	Anne Arundel	313.89	Michigan	Jackson	94.17
Maryland	Baltimore	313.89	Michigan	Kalamazoo	94.17
Maryland	Calvert	156.94	Michigan	Kalkaska	62.78
Maryland	Caroline	94.17	Michigan	Kent	156.94
Maryland	Carroll	156.94	Michigan	Keweenaw	62.78
Maryland	Cecil	156.94	Michigan	Lake	62.78
Maryland	Charles	94.17	Michigan	Lapeer	156.94
Maryland	Dorchester	94.17	Michigan	Leelanau	156.94
Maryland	Frederick	156.94	Michigan	Lenawee	94.17
Maryland	Garrett	62.78	Michigan	Livingston	156.94
Maryland	Harford	156.94	Michigan	Luce	47.08
Maryland	Howard	156.94	Michigan	Mackinac	47.08
Maryland	Kent	94.17	Michigan	Macomb	156.94
Maryland	Montgomery	156.94	Michigan	Manistee	62.78
Maryland	Prince George's	313.89	Michigan	Marguette	47.08
Maryland	Queen Anne's	94.17	Michigan	Mason	62.78
Maryland	Somerset	94.17	Michigan	Mecosta	62.78
	St. Mary's	94.17		Menominee	47.08
Maryland	Talbot	156.94	Michigan	Midland	94.17
	Washington	156.94	Michigan	Missaukee	62.78
Maryland	Wicomico	94.17	Michigan		94.17
Maryland		62.78	Michigan	Monroe	62.78
Maryland	Worcester Barnstable		Michigan	Montcalm	
Massachusetts		627.77	Michigan	Montmorency	62.78 94.17
Massachusetts	Berkshire Bristol	156.94 627.77	Michigan Michigan	Muskegon	94.17
	Dukes			Newaygo	
Massachusetts		313.89	Michigan	Oakland	313.89
Massachusetts	Essex	627.77	Michigan	Oceana	94.17
Massachusetts	Franklin	156.94	Michigan	Ogemaw	62.78
Massachusetts	Hampden	313.89	Michigan	Ontonagon	31.39
Massachusetts	Hampshire	313.89	Michigan	Osceola	62.78
Massachusetts	Middlesex	627.77	Michigan	Oscoda	62.78
Massachusetts	Nantucket	1,569.43	Michigan	Otsego	62.78
Massachusetts	Norfolk	627.77	Michigan	Ottawa	156.94
		697 77	Muchigan	L Proceduo Iclo	60 70
Massachusetts	Plymouth Suffolk	627.77 1,569.43	Michigan Michigan	Presque Isle	62.78 94.17

Federal Energy Regulatory Commission

Pt. 11, App. A

State	County	(Fee/acre/ YR)	State	County	(Fee/acre YR)
Michigan	Saginaw	62.78	Minnesota	Red Lake	31.
Michigan	Sanilac	62.78	Minnesota	Redwood	47.
/lichigan	Schoolcraft	47.08	Minnesota	Renville	62.
/lichigan	Shiawassee	62.78	Minnesota	Rice	94.
lichigan	St. Clair	156.94	Minnesota	Rock	47.
lichigan	St. Joseph	62.78	Minnesota	Roseau	15.
lichigan	Tuscola	62.78	Minnesota	Scott	94.
lichigan	Van Buren	94.17	Minnesota	Sherburne	94.
lichigan	Washtenaw	156.94	Minnesota	Sibley	62.
lichigan	Wayne	313.89	Minnesota	St. Louis	47.
lichigan	Wexford	94.17	Minnesota	Stearns	47.
finnesota	Aitkin	31.39	Minnesota	Steele	62.
finnesota	Anoka	156.94	Minnesota	Stevens	47.
linnesota	Becker	31.39	Minnesota	Swift	31.
linnesota	Beltrami	31.39	Minnesota	Todd	31.
linnesota	Benton	62.78	Minnesota	Traverse	31.
innesota	Big Stone	31.39	Minnesota	Wabasha	47.
innesota	Blue Earth	62.78	Minnesota	Wadena	31.
linnesota	Brown	62.78	Minnesota	Waseca	62.
linnesota	Carlton	31.39	Minnesota	Washington	156.
innesota	Carver	94.17	Minnesota	Watonwan	47.
innesota	Cass	31.39	Minnesota	Wilkin	31.
linnesota	Chippewa	47.08	Minnesota	Winona	62.
innesota	Chisago	94.17	Minnesota	Wright	94
innesota	Clay	31.39	Minnesota	Yellow Medicine	47
innesota	Clearwater	31.39	Mississippi	Adams	31
innesota	Cook	47.08	Mississippi	Alcorn	47
innesota	Cottonwood	47.08	Mississippi	Amite	47
innesota	Crow Wing	31.39	Mississippi	Attala	47
nnesota	Dakota	94.17	Mississippi	Benton	31
nnesota	Dodge	62.78	Mississippi	Bolivar	31
nnesota	Douglas	47.08	Mississippi	Calhoun	31
innesota	Faribault	62.78	Mississippi	Carroll	31
innesota	Fillmore	47.08	Mississippi	Chickasaw	31
innesota	Freeborn	62.78	Mississippi	Choctaw	31
innesota	Goodhue	62.78	Mississippi	Claiborne	31
innesota	Grant	47.08	Mississippi	Clarke	47
innesota	Hennepin	156.94	Mississippi	Clay	31
innesota	Houston	47.08	Mississippi	Coahoma	31
innesota	Hubbard	31.39	Mississippi	Copiah	47
innesota	Isanti	62.78	Mississippi	Covington	47
innesota	Itasca	31.39	Mississippi	DeSoto	62
nnesota	Jackson	47.08	Mississippi	Forrest	94
nnesota	Kanabec	47.08	Mississippi	Franklin	47
innesota	Kandiyohi	47.08	Mississippi	George	94
innesota	Kittson	15.69	Mississippi	Greene	47
innesota	Koochiching	31.39	Mississippi	Grenada	31
innesota	Lac qui Parle	31.39	Mississippi	Hancock	62
innesota	Lake	47.08	Mississippi	Harrison	156
innesota	Lake of the Woods	15.69	Mississippi	Hinds	47
innesota	Le Sueur	62.78	Mississippi	Holmes	31
nnesota	Lincoln	31.39	Mississippi	Humphreys	31
nnesota	Lyon	47.08	Mississippi	Issaquena	31
nnesota	Mahnomen	31.39	Mississippi	Itawamba	31
nnesota	Marshall	15.69	Mississippi	Jackson	156
nnesota	Martin	62.78	Mississippi	Jasper	47
innesota	McLeod	62.78	Mississippi	Jefferson	47
nnesota	Meeker	47.08	Mississippi	Jefferson Davis	47
nnesota	Mille Lacs	47.08	Mississippi	Jones	62
nnesota	Morrison	47.08	Mississippi	Kemper	31
nnesota	Mower	62.78	Mississippi	Lafayette	47
nnesota	Murray	47.08	Mississippi	Lamar	62
nnesota	Nicollet	62.78	Mississippi	Lauderdale	47
nnesota	Nobles	47.08	Mississippi	Lawrence	47
nnesota	Norman	31.39	Mississippi	Leake	47
innesota	Olmsted	62.78	Mississippi	Lee	47
nnesota	Otter Tail	31.39	Mississippi	Leflore	31
innesota	Pennington	15.69	Mississippi	Lincoln	62
innesota	Pine	47.08	Mississippi	Lowndes	31
innesota	Pipestone	47.08	Mississippi	Madison	47
innesota	Polk	31.39	Mississippi	Marion	47
innesota	Pope	31.39	Mississippi	Marshall	47

State	County	(Fee/acre/ YR)	State	County	(Fee/acre/ YR)
Mississippi	. Montgomery	31.39	Missouri	Grundy	31.39
Mississippi		62.78	Missouri	Harrison	31.39
Mississippi	Newton	94.17	Missouri	Henry	31.39
Mississippi		31.39	Missouri	Hickory	31.39
Mississippi		47.08	Missouri	Holt	47.08
Mississippi		31.39	Missouri	Howard	47.08
Mississippi Mississippi		94.17 62.78	Missouri Missouri	Iron	47.08 47.08
Mississippi		62.78	Missouri	Jackson	94.17
Mississippi		31.39	Missouri	Jasper	47.08
Mississippi		31.39	Missouri	Jefferson	94.17
Mississippi		31.39	Missouri	Johnson	47.08
Mississippi		47.08	Missouri	Knox	47.08
Mississippi		47.08	Missouri	Laclede	47.08
Mississippi		31.39 62.78	Missouri Missouri	Lafayette	47.08 47.08
Mississippi Mississippi		62.78	Missouri	Lawrence	31.39
Mississippi		47.08	Missouri	Lincoln	62.78
Mississippi		31.39	Missouri	Linn	31.39
Mississippi		31.39	Missouri	Livingston	47.08
Mississippi		47.08	Missouri	Macon	31.39
Mississippi	. Tippah	31.39	Missouri	Madison	31.39
Mississippi		47.08	Missouri	Maries	31.39
Mississippi		31.39	Missouri	Marion	31.39
Mississippi		47.08	Missouri	McDonald	62.78
Mississippi		94.17 31.39	Missouri Missouri	Mercer Miller	156.94 47.08
Mississippi Mississippi		47.08	Missouri	Mississippi	47.08
Mississippi		47.08	Missouri	Moniteau	47.08
Mississippi		31.39	Missouri	Monroe	31.39
Mississippi		47.08	Missouri	Montgomery	47.08
Mississippi	. Winston	47.08	Missouri	Morgan	47.08
Mississippi		31.39	Missouri	New Madrid	47.08
Mississippi		31.39	Missouri	Newton	47.08
Missouri		31.39	Missouri	Nodaway	31.39
Missouri		47.08 47.08	Missouri	Oregon	31.39 47.08
Missouri		47.08	Missouri	Osage	47.08
Missouri		47.08	Missouri	Pemiscot	47.08
Missouri		31.39	Missouri	Perry	47.08
Missouri		31.39	Missouri	Pettis	47.08
Missouri	Benton	31.39	Missouri	Phelps	47.08
Missouri		47.08	Missouri	Pike	47.08
Missouri		94.17	Missouri	Platte	62.78
Missouri		47.08 47.08	Missouri Missouri	Polk Pulaski	47.08 47.08
Missouri		47.08 47.08	Missouri	Putnam	31.39
Missouri		47.08	Missouri	Ralls	47.08
Missouri		47.08	Missouri	Randolph	31.39
Missouri		62.78	Missouri	Ray	47.08
Missouri		47.08	Missouri	Reynolds	31.39
Missouri		31.39	Missouri	Ripley	31.39
Missouri		47.08	Missouri	Saline	47.08
Missouri		31.39	Missouri	Schuyler	31.39
Missouri Missouri		47.08 62.78	Missouri Missouri	Scotland	31.39 47.08
Missouri		31.39	Missouri	Shannon	31.39
Missouri		94.17	Missouri	Shelby	31.39
Missouri		47.08	Missouri	St Louis	94.17
Missouri	. Cole	62.78	Missouri	St. Charles	156.94
Missouri	Cooper	47.08	Missouri	St. Clair	31.39
Missouri	. Crawford	31.39	Missouri	St. Francois	62.78
Missouri		47.08	Missouri	Ste. Genevieve	47.08
Missouri		47.08	Missouri	Stoddard	62.78
Missouri		31.39	Missouri Missouri	Stone	62.78
Missouri		31.39 31.39	Missouri	Sullivan	31.39 47.08
Missouri		31.39	Missouri	Texas	31.39
Missouri		62.78	Missouri	Vernon	31.39
Missouri		62.78	Missouri	Warren	62.78
Missouri		47.08	Missouri	Washington	47.08
Missouri		31.39	Missouri	Wayne	31.39
				Webster	

State	County	(Fee/acre/ YR)	State	County	(Fee/acr YR)
Missouri	Worth	31.39	Nebraska	Cherry	7
Missouri	Wright	47.08	Nebraska	Cheyenne	15
Montana	Beaverhead	15.69	Nebraska	Clay	47
Montana	Big Horn	7.85	Nebraska	Colfax	47
Iontana	Blaine	7.85	Nebraska	Cuming	47
fontana	Broadwater	15.69	Nebraska	Custer	15
Iontana	Carbon	31.39	Nebraska	Dakota	47
Iontana	Carter	7.85	Nebraska	Dawes	15
Iontana	Cascade	15.69	Nebraska	Dawson	31
lontana	Chouteau	15.69	Nebraska	Deuel	15
lontana	Custer	7.85	Nebraska	Dixon	31
lontana	Daniels	7.85	Nebraska	Dodge	62
ontana	Dawson	7.85	Nebraska	Douglas	156
ontana	Deer Lodge	31.39	Nebraska	Dundy	15
ontana	Fallon	7.85	Nebraska	Fillmore	47
ontana	Fergus	15.69	Nebraska	Franklin	31
ontana	Flathead	62.78	Nebraska	Frontier	15
ontana	Gallatin	31.39	Nebraska	Furnas	15
ontana	Garfield	7.85	Nebraska	Gage	31
ontana	Glacier	15.69	Nebraska	Garden	7
ontana	Golden Valley	7.85	Nebraska	Garfield	15
ontana	Granite	31.39	Nebraska	Gosper	31
ontana	Hill	15.69	Nebraska	Grant	7
ontana	Jefferson	15.69	Nebraska	Greeley	31
ontana	Judith Basin	15.69	Nebraska	Hall	47
ontana	Lake	31.39	Nebraska	Hamilton	47
ontana	Lewis and Clark	15.69	Nebraska	Harlan	31
ontana	Liberty	15.69	Nebraska	Hayes	15
ontana	Lincoln	94.17	Nebraska	Hitchcock	15
ontana	Madison	31.39	Nebraska	Holt	15
ontana	McCone	7.85	Nebraska	Hooker	7
ontana	Meagher	15.69	Nebraska	Howard	31
ontana	Mineral	62.78	Nebraska	Jefferson	31
ontana	Missoula	47.08	Nebraska	Johnson	31
ontana	Musselshell	7.85	Nebraska	Kearney	47
ontana	Park	31.39	Nebraska	Keith	15
ontana	Petroleum	7.85	Nebraska	Keya Paha	15
ontana	Phillips	7.85	Nebraska	Kimball	7
ontana	Pondera	15.69	Nebraska	Knox	31
ontana	Powder River	7.85	Nebraska	Lancaster	62
ontana	Powell	15.69	Nebraska	Lincoln	15
ontana	Prairie	7.85	Nebraska	Logan	7
ontana	Ravalli	94.17	Nebraska	Loup	7
ontana	Richland	7.85	Nebraska	Madison	47
ontana	Roosevelt	7.85	Nebraska	McPherson	7
ontana	Rosebud	7.85	Nebraska	Merrick	47
ontana	Sanders	31.39	Nebraska	Morrill	15
ontana	Sheridan	15.69	Nebraska	Nance	31
ontana	Silver Bow	31.39	Nebraska	Nemaha	47
ontana	Stillwater	15.69	Nebraska	Nuckolls	31
ontana	Sweet Grass	15.69	Nebraska	Otoe	47
ontana	Teton	15.69	Nebraska	Pawnee	31
ontana	Toole	15.69	Nebraska	Perkins	31
ontana	Treasure	7.85	Nebraska	Phelps	47
ontana	Valley	7.85	Nebraska	Pierce	31
ontana	Wheatland	7.85	Nebraska	Platte	47
ontana	Wibaux	7.85	Nebraska	Polk	47
ontana	Yellowstone	15.69	Nebraska	Red Willow	15
braska	Adams	47.08	Nebraska	Richardson	31
braska	Antelope	31.39	Nebraska	Rock	15
braska	Arthur	7.85	Nebraska	Saline	47
braska	Banner	7.85	Nebraska	Sarpy	94
ebraska	Blaine	7.85	Nebraska	Saunders	62
ebraska	Boone	31.39	Nebraska	Scotts Bluff	31
ebraska	Box Butte	15.69	Nebraska	Seward	47
ebraska	Boyd	15.69	Nebraska	Sheridan	7
ebraska	Brown	15.69	Nebraska	Sherman	15
ebraska	Buffalo	47.08	Nebraska	Sioux	7
ebraska	Burt	47.08	Nebraska	Stanton	47
ebraska	Butler	62.78	Nebraska	Thayer	47
ebraska	Cass	62.78	Nebraska	Thomas	7
ebraska	Cedar	31.39	Nebraska	Thurston	47

State	County	(Fee/acre/ YR)	State	County	(Fee/acre/ YR)
Nebraska	Washington	62.78	New Mexico	Quay	7.85
Nebraska	Wayne	47.08	New Mexico	Rio Arriba	15.69
Nebraska	Webster	31.39	New Mexico	Roosevelt	7.85
Nebraska	Wheeler	15.69	New Mexico	San Juan	15.69
Nebraska	York	62.78	New Mexico	San Miguel	7.85
Nevada	Carson City	94.17	New Mexico	Sandoval	7.85
Nevada	Churchill	47.08	New Mexico	Santa Fe	15.69
Nevada	Clark	94.17	New Mexico	Sierra	7.85
Nevada Nevada	Douglas Elko	31.39 7.85	New Mexico New Mexico	Socorro	7.85 15.69
Nevada	Esmeralda	31.39	New Mexico	Torrance	7.85
Nevada	Eureka	7.85	New Mexico	Union	7.85
Nevada	Humboldt	15.69	New Mexico	Valencia	31.39
Nevada	Lander	7.85	New York	Albany	94.17
Nevada	Lincoln	31.39	New York	Allegany	31.39
Nevada	Lyon	47.08	New York	Bronx *	47.08
Nevada	Mineral	7.85	New York	Broome	94.17
Nevada	Nye	31.39	New York	Cattaraugus	47.08
Nevada	Pershing	31.39	New York	Cayuga	47.08
Nevada	Storey	941.66	New York	Chautauqua	47.08
Nevada	Washoe	15.69	New York	Chemung	47.08
Nevada	White Pine	15.69	New York	Chenango	31.39
New Hampshire	Belknap	94.17	New York	Clinton	31.39
New Hampshire	Carroll	94.17	New York	Columbia	94.17
New Hampshire	Cheshire	94.17	New York	Cortland	31.39
New Hampshire	Coos	31.39	New York	Delaware	47.08
New Hampshire	Grafton	62.78	New York	Dutchess	313.89
New Hampshire	Hillsborough	156.94	New York	Erie	47.08
New Hampshire	Merrimack	94.17	New York	Essex	47.08
New Hampshire	Rockingham	313.89 94.17	New York New York	Franklin	31.39
New Hampshire	Strafford Sullivan	94.17	New York	Fulton	47.08 47.08
	Atlantic	156.94	New York	Greene	62.78
New Jersey New Jersey	Bergen	1,569.43	New York	Hamilton *	47.08
New Jersey	Burlington	313.89	New York	Herkimer	31.39
New Jersey	Camden	313.89	New York	Jefferson	31.39
New Jersey	Cape May	313.89	New York	Kings*	47.08
New Jersey	Cumberland	156.94	New York	Lewis	31.39
New Jersey	Essex	1,569.43	New York	Livingston	47.08
New Jersey	Gloucester	313.89	New York	Madison	47.08
New Jersey	Hudson *	313.89	New York	Monroe	62.78
New Jersey	Hunterdon	313.89	New York	Montgomery	47.08
New Jersey	Mercer	627.77	New York	Nassau	941.66
New Jersey	Middlesex	627.77	New York	New York	313.89
New Jersey	Monmouth	627.77	New York	Niagara	47.08
New Jersey	Morris	941.66	New York	Oneida	31.39
New Jersey	Ocean	627.77	New York	Onondaga	47.08
New Jersey	Passaic	941.66	New York	Ontario	47.08
New Jersey	Salem	156.94	New York	Orange	156.94
New Jersey New Jersey	Somerset	627.77 313.89	New York New York	Orleans Oswego	31.39 62.78
New Jersey	Union	3,138.86	New York	Otsego	47.08
New Jersey	Warren	313.89	New York	Putnam	313.89
New Mexico	Bernalillo	15.69	New York	Queens	47.08
New Mexico	Catron	7.85	New York	Rensselaer	94.17
New Mexico	Chaves	7.85	New York	Richmond	3,138.86
New Mexico	Cibola	7.85	New York	Rockland	941.66
New Mexico	Colfax	7.85	New York	Saratoga	94.17
New Mexico	Curry	15.69	New York	Schenectady	62.78
New Mexico	De Baca	7.85	New York	Schoharie	47.08
New Mexico	Dona Ana	47.08	New York	Schuyler	47.08
New Mexico	Eddy	7.85	New York	Seneca	47.08
New Mexico	Grant	7.85	New York	St. Lawrence	31.39
New Mexico	Guadalupe	7.85	New York	Steuben	31.39
New Mexico	Harding*	7.85	New York	Suffolk	627.77
New Mexico	Hidalgo	7.85	New York	Sullivan	94.17
New Mexico	Lea	7.85	New York	Tioga	47.08
New Mexico	Lincoln	7.85	New York	Tompkins	47.08
New Mexico	Los Alamos*	7.85	New York	Ulster	94.17
NIOW Marvioo	Luna	7.85	New York	Warren	94.17
New Mexico				144 11 1	
New Mexico New Mexico	McKinley Mora	7.85 7.85	New York New York	Washington Wayne	47.08 62.78

Federal Energy Regulatory Commission

State	County	(Fee/acre/ YR)	State	County	(Fee/acre/ YR)
New York	Wyoming	47.08	North Carolina	Perquimans	62.7
New York	Yates	47.08	North Carolina	Person	62.7
North Carolina	Alamance	156.94	North Carolina	Pitt	62.7
North Carolina	Alexander	156.94	North Carolina	Polk	156.9
North Carolina	Alleghany	94.17	North Carolina	Randolph	156.9
North Carolina	Anson	94.17	North Carolina	Richmond	62.7
North Carolina	Ashe	156.94	North Carolina	Robeson	62.7
North Carolina	Avery	156.94	North Carolina	Rockingham	94.1
North Carolina	Beaufort	62.78	North Carolina	Rowan	94.1
North Carolina	Bertie	62.78	North Carolina	Rutherford	94.1
North Carolina	Bladen	94.17	North Carolina	Sampson	94.1
North Carolina	Brunswick	94.17	North Carolina	Scotland	62.7
North Carolina	Buncombe	156.94	North Carolina	Stanly	94.1
North Carolina	Burke	156.94	North Carolina	Stokes	94.1
North Carolina	Cabarrus	156.94	North Carolina	Surry	94.1
North Carolina	Caldwell	156.94	North Carolina	Swain	156.9
North Carolina	Camden	62.78	North Carolina	Transylvania	313.8
North Carolina	Carteret	62.78	North Carolina	Tyrrell	47.0
North Carolina	Caswell	94.17	North Carolina	Union	94.1
North Carolina	Catawba	94.17	North Carolina	Vance	62.7
North Carolina	Chatham	94.17	North Carolina	Wake	313.8
North Carolina	Cherokee	156.94	North Carolina	Warren	62.7
North Carolina	Chowan	62.78	North Carolina	Washington	62.7
North Carolina	Clay	156.94	North Carolina	Watauga	156.9
North Carolina	Cleveland	94.17	North Carolina	Wayne	94.1
North Carolina	Columbus	62.78	North Carolina	Wilkes	94.1
North Carolina	Craven	62.78	North Carolina	Wilson	62.7
North Carolina	Cumberland	94.17	North Carolina	Yadkin	94.1
North Carolina	Currituck	94.17	North Carolina	Yancey	156.9
lorth Carolina	Dare	47.08	North Dakota	Adams	7.8
lorth Carolina	Davidson	156.94	North Dakota	Barnes	15.6
lorth Carolina	Davie	156.94	North Dakota	Benson	15.6
North Carolina	Duplin	94.17	North Dakota	Billings	7.8
North Carolina	Durham	156.94	North Dakota	Bottineau	15.6
North Carolina		62.78	North Dakota	Bowman	7.8
North Carolina	Edgecombe	156.94	North Dakota		7.8
	Forsyth			Burke	
North Carolina	Franklin	94.17	North Dakota	Burleigh	15.6
North Carolina	Gaston	156.94	North Dakota	Cass	31.3
North Carolina	Gates	47.08	North Dakota	Cavalier	15.6
North Carolina	Graham	94.17	North Dakota	Dickey	15.6
North Carolina	Granville	94.17	North Dakota	Divide	7.8
North Carolina	Greene	94.17	North Dakota	Dunn	7.8
North Carolina	Guilford	156.94	North Dakota	Eddy	15.6
North Carolina	Halifax	47.08	North Dakota	Emmons	7.8
North Carolina	Harnett	94.17	North Dakota	Foster	15.6
North Carolina	Haywood	156.94	North Dakota	Golden Valley	7.8
North Carolina	Henderson	156.94	North Dakota	Grand Forks	31.3
lorth Carolina	Hertford	62.78	North Dakota	Grant	7.8
North Carolina	Hoke	94.17	North Dakota	Griggs	15.6
lorth Carolina	Hyde	47.08	North Dakota	Hettinger	15.6
Iorth Carolina	Iredell	156.94	North Dakota	Kidder	7.8
lorth Carolina	Jackson	156.94	North Dakota	LaMoure	15.6
lorth Carolina	Johnston	94.17	North Dakota	Logan	7.8
lorth Carolina	Jones	62.78	North Dakota	McHenry	15.6
Iorth Carolina	Lee	94.17	North Dakota	McIntosh	7.8
lorth Carolina	Lenoir	94.17	North Dakota	McKenzie	7.8
lorth Carolina	Lincoln	156.94	North Dakota	McLean	15.6
lorth Carolina	Macon	156.94	North Dakota	Mercer	7.8
lorth Carolina	Madison	156.94	North Dakota	Morton	7.8
lorth Carolina	Martin	62.78	North Dakota	Mountrail	7.8
lorth Carolina	McDowell	94.17	North Dakota	Nelson	15.6
lorth Carolina	Mecklenburg	313.89	North Dakota	Oliver	7.8
lorth Carolina	Mitchell	156.94	North Dakota	Pembina	31.0
lorth Carolina	Montgomery	94.17	North Dakota	Pierce	15.6
lorth Carolina	Moore	94.17	North Dakota	Ramsey	15.6
lorth Carolina	Nash	94.17	North Dakota	Ransom	15.0
Iorth Carolina	New Hanover	313.89	North Dakota	Renville	15.0
lorth Carolina	Northampton	62.78	North Dakota	Richland	31.
lorth Carolina	Onslow	94.17	North Dakota	Rolette	15.0
lorth Carolina	Orange	156.94	North Dakota	Sargent	15.0
North Carolina	Pamlico	62.78	North Dakota	Sheridan	7.8
North Carolina	Pasquotank	62.78	North Dakota	Sioux	7.8
North Carolina	Pender	94.17	North Dakota	Slope	7.

North Dakida			(Fee/acre/			(Fee/acre/
North Dakota	е	e County	YR)	State	County	YR)
North Dakota	aSta	aStark	15.69	Ohio	Pickaway	94.17
North Dakota	aSte	a Steele	15.69	Ohio		47.08
North Dakota						156.94
North Dakota Walsh						94.17
North Dakota Ward						62.78
North Dakota Wells						94.17
North Dakota						62.78 62.78
Ohio Adams 62.78 Ohio Seneca Ohio Allen 94.17 Ohio Stark Ohio Ashland 94.17 Ohio Stark Ohio Ashland 94.17 Ohio Turmin Ohio Auglaze 94.17 Ohio Turmin Ohio Belmont 47.08 Ohio Union Ohio Belmont 47.08 Ohio Union Ohio Brown 62.78 Ohio Union Ohio Brown 62.78 Ohio Union Ohio Brown 62.78 Ohio Union Ohio Carroll 62.78 Ohio Warren Ohio Carroll 62.78 Ohio Warren Ohio Carroll 94.17 Ohio Warren Ohio Carroll 94.17 Ohio Wayne Ohio Carroll 94.17 Ohio Wayne Ohio						47.08
Ohio Allen 94.17 Ohio Shelby Ohio Ashtabula 62.78 Ohio Summit Ohio Athens 47.08 Ohio Turmit Ohio Auglaze 94.17 Ohio Tuscarawas Ohio Belmont 47.08 Ohio Union Ohio Butler 156.94 Ohio Van Wert Ohio Clar Ohio Warner Ohio Clar Ohio Warner Ohio Clar Ohio Warner Ohio Clark 94.17 Ohio Warner Ohio Clark 94.17 Ohio Warner Ohio Clark 94.17 Ohio Wayner Ohio Clark 94.17 Ohio Wayner Ohio Clark 94.17 Ohio Wayner Ohio Columbiana 94.17 Ohio Wayned Ohio Colario 62.78 Oklahoma<						62.78
Ohio Ashland 94.17 Ohio Stark Ohio Ashlabula 62.78 Ohio Tummil Ohio Athens 47.08 Ohio Trumbull Ohio Auglaize 94.17 Ohio Union Ohio Belmont 47.08 Ohio Union Ohio Brown 62.78 Ohio Van Wert Ohio Carroll 62.78 Ohio Van Wert Ohio Carroll 62.78 Ohio Warren Ohio Champaign 94.17 Ohio Washington Ohio Clarak 94.17 Ohio Wayne Ohio Clarmot 94.17 Ohio Wayne Ohio Cliromot 94.17 Ohio Wayne Ohio Columbiana 94.17 Ohio Wood Ohio Columbiana 94.17 Ohio Wayne Ohio Columbiana 94.17 Ohio Wayne						94.17
Ohio Athens 47.08 Ohio Trumbull Ohio Belmont 47.08 Ohio Union Ohio Brown 62.78 Ohio Union Ohio Brown 62.78 Ohio Van Wert Ohio Bulter 156.94 Ohio Vinton Ohio Clarroll 62.78 Ohio Warren Ohio Champaign 94.17 Ohio Washington Ohio Clarmont 94.17 Ohio Washington Ohio Cliermont 94.17 Ohio Waliams Ohio Cliermont 94.17 Ohio Wood Ohio Cliermont 94.17 Ohio Wood Ohio Costocton 62.78 Oklahoma Adair Ohio Carwford 62.78 Oklahoma Adaira Ohio Darke 94.17 Oklahoma Beckhar Ohio Delaware 156.94 Oklahoma Beckhar </td <td></td> <th></th> <td></td> <td></td> <td></td> <td>156.94</td>						156.94
Ohio Auglaize 94.17 Ohio Tuscarawas Ohio Belmont 47.08 Ohio Union Ohio Brown 62.78 Ohio Van Wert Ohio Butler 156.94 Ohio Vinton Ohio Carroll 62.78 Ohio Waren Ohio Charmagin 94.17 Ohio Washington Ohio Clark 94.17 Ohio Washington Ohio Clinton 94.17 Ohio Wayne Ohio Clinton 94.17 Ohio Wood Ohio Cosbocton 62.78 Oklahoma Adair Ohio Cravford 62.78 Oklahoma Adair Ohio Cravford 62.78 Oklahoma Adair Ohio Darke 94.17 Oklahoma Beaver Ohio Delaware 156.94 Oklahoma Beaver Ohio Delaware 156.94 Oklahoma Beavin	Asl	Ashtabula	62.78	Ohio		156.94
Ohio Belmont 47.08 Ohio Union Ohio Brown 62.78 Ohio Van Wert Ohio Buller 156.94 Ohio Vinton Ohio Caroll 62.78 Ohio Warren Ohio Champaign 94.17 Ohio Washington Ohio Clark 94.17 Ohio Washington Ohio Clermont 94.17 Ohio Williams Ohio Clinton 94.17 Ohio Wyandot Ohio Columbiana 94.17 Ohio Wyandot Ohio Cadrofd 62.78 Oklahoma Affaifa Ohio Cavarlord 62.78 Oklahoma Affaifa Ohio Darke 94.17 Oklahoma Becaver Ohio Deliance 62.78 Oklahoma Becaver Ohio Erie 94.17 Oklahoma Belaine Ohio Fairield 94.17 Oklahoma <t< td=""><td></td><th></th><td></td><td></td><td></td><td>94.17</td></t<>						94.17
Dhio						94.17
Ohio Butler 156.94 Ohio Winton Ohio Carroll 62.78 Ohio Waren Ohio Champaign 94.17 Ohio Waren Ohio Clark 94.17 Ohio Wayne Ohio Clermont 94.17 Ohio Williams Ohio Clumbiana 94.17 Ohio Williams Ohio Coshocton 62.78 Oklahoma Adair Ohio Crawford 62.78 Oklahoma Alfalfa Ohio Darke 94.17 Oklahoma Alcoka Ohio Darke 94.17 Oklahoma Beaver Ohio Delaware 156.94 Oklahoma Beckham Ohio Erie 94.17 Oklahoma Blaine Ohio Fairliel 94.17 Oklahoma Carder Ohio Fairliel 94.17 Oklahoma Carder Ohio Faratklin 156.94 Oklahoma						94.17
Ohio Carroll 62.78 bit Ohio Waren (Mayne) Ohio Clark 94.17 Ohio Wayne Ohio Clermont 94.17 Ohio Wayne Ohio Clermont 94.17 Ohio Williams Ohio Columbiana 94.17 Ohio Wood Ohio Columbiana 94.17 Ohio Wyandot Ohio Columbiana 94.17 Ohio Wyandot Ohio Crawford 62.78 Oklahoma Affalfa Ohio Crawford 62.78 Oklahoma Affalfa Ohio Darke 94.17 Oklahoma Beaver Ohio Darke 94.17 Oklahoma Beaver Ohio Defance 62.78 Oklahoma Blaine Ohio Firaffield 94.17 Oklahoma Caddo Ohio Farifield 94.17 Oklahoma Caddo Ohio Faryete 62.78 Oklahoma Carler Ohio Faryete 62.78 Oklahoma Carler Ohio Gallia 47.08 Oklaho		1 -				94.17
Ohio Champaign 94.17 Ohio Wayne Ohio Clark 94.17 Ohio Wayne Ohio Clermont 94.17 Ohio Williams Ohio Columbiana 94.17 Ohio Williams Ohio Coshocton 62.78 Oklahoma Adair Ohio Crawford 62.78 Oklahoma Alfalfa Ohio Darke 94.17 Oklahoma Altoka Ohio Darke 94.17 Oklahoma Beaver Ohio Deflance 62.78 Oklahoma Beckham Ohio Deflance 62.78 Oklahoma Beckham Ohio Eiriell 94.17 Oklahoma Beckham Ohio Fairliell 94.17 Oklahoma Bryan Ohio Fairliell 94.17 Oklahoma Caradian Ohio Faryette 62.78 Oklahoma Carec Ohio Galilla 47.08 Oklahoma						62.78 156.94
Ohio Clark 94.17 Ohio Wayne Ohio Cliermont 94.17 Ohio Williams Ohio Columbiana 94.17 Ohio Wood Ohio Coshocton 62.78 Oklahoma Adair Ohio Crawford 62.78 Oklahoma Allalfa Ohio Cuydhoga 62.77 Oklahoma Allalfa Ohio Darke 94.17 Oklahoma Beaver Ohio Deflance 62.78 Oklahoma Beaver Ohio Deflance 62.78 Oklahoma Beaver Ohio Deflance 62.78 Oklahoma Blaine Ohio Fairfield 94.17 Oklahoma Cardo Ohio Fairfield 94.17 Oklahoma Cardo Ohio Franklin 156.94 Oklahoma Carter Ohio Faryette 62.78 Oklahoma Cherokee Ohio Gailia 47.08 Oklah						62.78
Ohio Clemont 94.17 Ohio Williams Ohio Columbiana 94.17 Ohio Wood Ohio Columbiana 94.17 Ohio Wyandot Ohio Coshocton 62.78 Oklahoma Addat Ohio Cuyahoga 92.77 Oklahoma Altafa Ohio Defiance 62.78 Oklahoma Beaver Ohio Defiance 62.78 Oklahoma Beever Ohio Delaware 156.94 Oklahoma Beckham Ohio Erie 94.17 Oklahoma Bryan Ohio Fairlield 94.17 Oklahoma Carddo Ohio Faryette 62.78 Oklahoma Carddo Ohio Fayette 62.78 Oklahoma Cardera Ohio Gallia 47.08 Oklahoma Cherokee Ohio Gallia 47.08 Oklahoma Cherokee Ohio Guernsey 62.78 Ok						156.94
Ohio Clinton 94.17 Ohio Wood Ohio Columbiana 94.17 Ohio Wyandot Ohio Coshocton 62.78 Oklahoma Adair Ohio Crawford 62.78 Oklahoma Adair Ohio Darke 94.17 Oklahoma Aloka Ohio Darke 94.17 Oklahoma Beaver Ohio Delaware 156.94 Oklahoma Blaine Ohio Erie 94.17 Oklahoma Caddo Ohio Fairfield 94.17 Oklahoma Canddian Ohio Fayette 62.78 Oklahoma Canddian Ohio Franklin 156.94 Oklahoma Canddian Ohio Franklin 156.94 Oklahoma Cherokee Ohio Geauga 156.94 Oklahoma Cherokee Ohio Geauga 156.94 Oklahoma Cherokee Ohio Geauga 156.94 O						62.78
Ohio Coshocton 62.78 Oklahoma Afair Ohio Crawford 62.77 Oklahoma Affaira Ohio Darke 94.17 Oklahoma Atoka Ohio Deliance 62.77 Oklahoma Beckham Ohio Deliance 62.78 Oklahoma Blaine Ohio Erie 94.17 Oklahoma Blaine Ohio Fairfield 94.17 Oklahoma Caddo Ohio Fairfield 94.17 Oklahoma Canadian Ohio Fayette 62.78 Oklahoma Carter Ohio Faraklin 156.94 Oklahoma Cherokee Ohio Galia 47.08 Oklahoma Cherokee Ohio Geauga 156.94 Oklahoma Choctaw Ohio Geauga 156.94 Oklahoma Cloveland Ohio Geauga 156.94 Oklahoma Colotawa Ohio Haracko 62.78	I					94.17
Ohio Crawford 62.78 Oklahoma Alfalfa Ohio Darke 94.17 Oklahoma Beaver Ohio Deffance 62.78 Oklahoma Beckham Ohio Delaware 156.94 Oklahoma Blaine Ohio Erie 94.17 Oklahoma Bryan Ohio Farifield 94.17 Oklahoma Caddo Ohio Farifield 94.17 Oklahoma Canadian Ohio Faryette 62.78 Oklahoma Caradian Ohio Franklin 156.94 Oklahoma Cherokee Ohio Gallia 47.08 Oklahoma Cherokee Ohio Gaeauga 156.94 Oklahoma Chrockee Ohio Greene 94.17 Oklahoma Cleveland Ohio Greene 94.17 Oklahoma Cola Ohio Hamiton 156.94 Oklahoma Contan Ohio Hardin 62.78	Co	Columbiana	94.17	Ohio	Wyandot	94.17
Ohio Cuyahoga 627.77 Oklahoma Atoka Ohio Darke 94.17 Oklahoma Beaver Ohio Deflance 62.78 Oklahoma Blaine Ohio Delaware 156.94 Oklahoma Blaine Ohio Fairfield 94.17 Oklahoma Cardo Ohio Fairfield 94.17 Oklahoma Cardo Ohio Fayette 62.78 Oklahoma Carde Ohio Franklin 156.94 Oklahoma Cherokee Ohio Fulton 94.17 Oklahoma Cherokee Ohio Gaalii 47.08 Oklahoma Cimarron Ohio Geauga 156.94 Oklahoma Cimarron Ohio Gerene 94.17 Oklahoma Cola Ohio Hamitton 156.94 Oklahoma Conanche Ohio Hamiton 156.94 Oklahoma Corai Ohio Hardino 31.39	Co	Coshocton	62.78	Oklahoma	Adair	31.39
Ohio Darke 94.17 Oklahoma Beaver Ohio Defiance 62.78 Oklahoma Beckham Ohio Delaware 156.94 Oklahoma Blaine Ohio Erie 94.17 Oklahoma Cardo Ohio Fayette 62.78 Oklahoma Canadian Ohio Franklin 156.94 Oklahoma Carder Ohio Franklin 156.94 Oklahoma Cherokee Ohio Geauga 156.94 Oklahoma Choctaw Ohio Geauga 156.94 Oklahoma Choctaw Ohio Greene 94.17 Oklahoma Cleveland Ohio Guernsey 62.78 Oklahoma Coal Ohio Hamilton 156.94 Oklahoma Coal Ohio Hamilton 156.94 Oklahoma Coranche Ohio Harrison 31.39 Oklahoma Creek Ohio Harrison 31.39						31.39
Ohio Defiance 62.78 Oklahoma Beckharm Ohio Delaware 156.94 Oklahoma Blaine Ohio Erie 94.17 Oklahoma Bryan Ohio Fairfield 94.17 Oklahoma Caddo Ohio Fayette 62.78 Oklahoma Caradian Ohio Franklin 156.94 Oklahoma Caradian Ohio Gallia 47.08 Oklahoma Cherokee Ohio Gallia 47.08 Oklahoma Cherokee Ohio Gaeuga 156.94 Oklahoma Cherokee Ohio Geauga 156.94 Oklahoma Cleveland Ohio Gerene 94.17 Oklahoma Colar Ohio Hamilton 156.94 Oklahoma Colar Ohio Harock 62.78 Oklahoma Colar Ohio Harrison 31.39 Oklahoma Crese Ohio Harrison 31.39						31.39
Ohio Delaware 156.94 Oklahoma Blaine Ohio Erie 94.17 Oklahoma Bryan Ohio Falifield 94.17 Oklahoma Caddo Ohio Fayette 62.78 Oklahoma Canadian Ohio Franklin 156.94 Oklahoma Carater Ohio Gallia 47.08 Oklahoma Cherokee Ohio Gallia 47.08 Oklahoma Cherokee Ohio Geauga 156.94 Oklahoma Cherokee Ohio Greene 94.17 Oklahoma Clowaloma Ohio Hamilton 156.94 Oklahoma Cleveland Ohio Hamilton 156.94 Oklahoma Coral Ohio Hardin 62.78 Oklahoma Corton Ohio Hardin 62.78 Oklahoma Create Ohio Harrison 31.39 Oklahoma Create Ohio Highland 62.78						15.69
Ohio Frie 94.17 Oklahoma Bryan Ohio Fairfield 94.17 Oklahoma Caddo Ohio Fayette 62.78 Oklahoma Canadian Ohio Franklin 156.94 Oklahoma Cherokee Ohio Gallia 47.08 Oklahoma Choctaw Ohio Geauga 156.94 Oklahoma Choctaw Ohio Greene 94.17 Oklahoma Choctaw Ohio Guernsey 62.78 Oklahoma Cola Ohio Hamilton 156.94 Oklahoma Cola Ohio Hamilton 156.94 Oklahoma Cola Ohio Harmilton 156.94 Oklahoma Cola Ohio Harcock 62.78 Oklahoma Cotton Ohio Harrison 31.39 Oklahoma Cotton Ohio Henry 94.17 Oklahoma Custer Ohio Hocking 94.17 Okla						15.69
Ohio Fairfield 94.17 Oklahoma Caddo Ohio Fayette 62.78 Oklahoma Canadian Ohio Franklin 156.94 Oklahoma Carter Ohio Fulton 94.17 Oklahoma Cherokee Ohio Gallia 47.08 Oklahoma Chorckee Ohio Geauga 156.94 Oklahoma Cimarron Ohio Greene 94.17 Oklahoma Cleveland Ohio Hamilton 156.94 Oklahoma Coal Ohio Hamilton 156.94 Oklahoma Coal Ohio Hardin 62.78 Oklahoma Comanche Ohio Hardin 62.78 Oklahoma Creek Ohio Harrison 31.39 Oklahoma Creek Ohio Harrison 31.39 Oklahoma Creek Ohio Henry 94.17 Oklahoma Dewey Ohio Hoking 94.17 O						15.69 31.39
Ohio Fayette 62.78 Oklahoma Canadian Ohio Franklin 156.94 Oklahoma Carter Ohio Fulton 94.17 Oklahoma Cherokee Ohio Gallia 47.08 Oklahoma Chectaw Ohio Geauga 156.94 Oklahoma Cimarron Ohio Greene 94.17 Oklahoma Colevaland Ohio Guernsey 62.78 Oklahoma Coal Ohio Hamilton 156.94 Oklahoma Comanche Ohio Hardin 62.78 Oklahoma Cotton Ohio Hardin 62.78 Oklahoma Creek Ohio Harrison 31.39 Oklahoma Custer Ohio Henry 94.17 Oklahoma Custer Ohio Highland 62.78 Oklahoma Delaware Ohio Hokking 94.17 Oklahoma Ellis Ohio Holmes 94.17 <						15.69
Ohio Franklin 156.94 Oklahoma Carter Ohio Gallia 47.08 Oklahoma Cherokee Ohio Gallia 47.08 Oklahoma Choctaw Ohio Geauga 156.94 Oklahoma Cimarron Ohio Greene 94.17 Oklahoma Cleveland Ohio Hamilton 156.94 Oklahoma Coal Ohio Hamilton 156.94 Oklahoma Coal Ohio Harison 31.93 Oklahoma Cotton Ohio Hardin 62.78 Oklahoma Creek Ohio Harrison 31.39 Oklahoma Creek Ohio Henry 94.17 Oklahoma Creek Ohio Highland 62.78 Oklahoma Delaware Ohio Hocking 94.17 Oklahoma Delaware Ohio Holmes 94.17 Oklahoma Gariel Ohio Jefferson 47.08 <td< td=""><td></td><th></th><td></td><td></td><td></td><td>31.39</td></td<>						31.39
Ohio Fulton 94.17 Oklahoma Cherokee Ohio Gallia 47.08 Oklahoma Choctaw Ohio Geauga 156.94 Oklahoma Cleveland Ohio Greene 94.17 Oklahoma Cleveland Ohio Hamilton 156.94 Oklahoma Coal Ohio Hamilton 156.94 Oklahoma Cotton Ohio Harmilton 62.78 Oklahoma Cotton Ohio Hardin 62.78 Oklahoma Cotton Ohio Harrison 31.39 Oklahoma Creek Ohio Henry 94.17 Oklahoma Creek Ohio Henry 94.17 Oklahoma Delware Ohio Hoking 94.17 Oklahoma Delware Ohio Hoking 94.17 Oklahoma Garfield Ohio Hoking 94.17 Oklahoma Garvin Ohio Jefferson 47.08						31.39
Ohio Gallia 47.08 Oklahoma Choctaw Ohio Geauga 156.94 Oklahoma Cimarron Ohio Greene 94.17 Oklahoma Cleveland Ohio Hamilton 156.94 Oklahoma Coal Ohio Hamilton 156.94 Oklahoma Comanche Ohio Hardin 62.78 Oklahoma Cotton Ohio Harrison 31.39 Oklahoma Creek Ohio Henry 94.17 Oklahoma Custer Ohio Highland 62.78 Oklahoma Delaware Ohio Hocking 94.17 Oklahoma Delewey Ohio Holmes 94.17 Oklahoma Ellis Ohio Holmes 94.17 Oklahoma Garfield Ohio Holmes 94.17 Oklahoma Garfield Ohio Jefferson 47.08 Oklahoma Grady Ohio Lake 313.89 <td< td=""><td>I</td><th></th><td></td><td></td><td></td><td>31.39</td></td<>	I					31.39
Ohio Greene 94.17 Oklahoma Cleveland Ohio Guernsey 62.78 Oklahoma Coal Ohio Hamilton 156.94 Oklahoma Comanche Ohio Hancock 62.78 Oklahoma Crotton Ohio Hardin 62.78 Oklahoma Craig Ohio Harrison 31.39 Oklahoma Creek Ohio Henry 94.17 Oklahoma Delaware Ohio Hocking 94.17 Oklahoma Delaware Ohio Holmes 94.17 Oklahoma Dewey Ohio Holmes 94.17 Oklahoma Ellis Ohio Huron 94.17 Oklahoma Garrield Ohio Jefferson 47.08 Oklahoma Grady Ohio Jefferson 47.08 Oklahoma Grady Ohio Lake 313.89 Oklahoma Greer Ohio Lake 313.89 Oklahoma						15.69
Ohio Guernsey 62.78 Oklahoma Coal Ohio Hamilton 156.94 Oklahoma Comanche Ohio Hancock 62.78 Oklahoma Cotton Ohio Hardin 62.78 Oklahoma Craig Ohio Harrison 31.39 Oklahoma Creek Ohio Henry 94.17 Oklahoma Delaware Ohio Helmy 94.17 Oklahoma Delaware Ohio Hocking 94.17 Oklahoma Dewey Ohio Holmes 94.17 Oklahoma Dewey Ohio Holmes 94.17 Oklahoma Garfield Ohio Huron 94.17 Oklahoma Garvin Ohio Jackson 47.08 Oklahoma Garvin Ohio Knox 94.17 Oklahoma Grady Ohio Knox 94.17 Oklahoma Grant Ohio Lake 313.89 Oklahoma	Ge	Geauga	156.94	Oklahoma	Cimarron	7.85
Ohio Hamilton 156,94 Oklahoma Comanche Ohio Hancock 62.78 Oklahoma Cotton Ohio Hardin 62.78 Oklahoma Craig Ohio Harrison 31.39 Oklahoma Creek Ohio Henry 94.17 Oklahoma Deuser Ohio Highland 62.78 Oklahoma Delwer Ohio Hoking 94.17 Oklahoma Dewey Ohio Holmes 94.17 Oklahoma Dewey Ohio Holmes 94.17 Oklahoma Garfield Ohio Jackson 47.08 Oklahoma Garfield Ohio Jefferson 47.08 Oklahoma Grardy Ohio Lake 313.89 Oklahoma Grear Ohio Lake 313.89 Oklahoma Harmon Ohio Logan 62.78 Oklahoma Harper Ohio Logan 62.78 Oklahoma				Oklahoma	Cleveland	47.08
Ohio Hancock 62.78 Oklahoma Cotton Ohio Hardin 62.78 Oklahoma Craig Ohio Harrison 31.39 Oklahoma Creek Ohio Henry 94.17 Oklahoma Delaware Ohio Highland 62.78 Oklahoma Delaware Ohio Hocking 94.17 Oklahoma Delaware Ohio Holmes 94.17 Oklahoma Delaware Ohio Holmes 94.17 Oklahoma Delaware Ohio Holmes 94.17 Oklahoma Ellis Ohio Huron 94.17 Oklahoma Garfield Ohio Jackson 47.08 Oklahoma Garvin Ohio Jefferson 47.08 Oklahoma Greaty Ohio Lake 313.89 Oklahoma Grear Ohio Lake 313.89 Oklahoma Harmon Ohio Logan 62.78 Oklahoma<						31.39
Ohio Hardin 62.78 Oklahoma Craig Ohio Harrison 31.39 Oklahoma Creek Ohio Henry 94.17 Oklahoma Dewey Ohio Highland 62.78 Oklahoma Delaware Ohio Hocking 94.17 Oklahoma Dewey Ohio Holmes 94.17 Oklahoma Ellis Ohio Huron 94.17 Oklahoma Garfield Ohio Jackson 47.08 Oklahoma Garvin Ohio Jefferson 47.08 Oklahoma Grady Ohio Knox 94.17 Oklahoma Grady Ohio Lake 313.89 Oklahoma Grant Ohio Lake 313.89 Oklahoma Harmon Ohio Lawrence 47.08 Oklahoma Harmon Ohio Logan 62.78 Oklahoma Harskell Ohio Logan 62.78 Oklahoma	I					31.39
Ohio Harrison 31.39 Oklahoma Creek Ohio Henry 94.17 Oklahoma Custer Ohio Highland 62.78 Oklahoma Delware Ohio Hocking 94.17 Oklahoma Dewey Ohio Holmes 94.17 Oklahoma Ellis Ohio Huron 94.17 Oklahoma Garfield Ohio Jackson 47.08 Oklahoma Garvin Ohio Jefferson 47.08 Oklahoma Grady Ohio Lake 313.89 Oklahoma Greer Ohio Lake 313.89 Oklahoma Harmon Ohio Lawrence 47.08 Oklahoma Harmon Ohio Logan 62.78 Oklahoma Harper Ohio Logan 62.78 Oklahoma Hasper Ohio Lucas 94.17 Oklahoma Hughes Ohio Madison 94.17 Oklahoma						15.69
Ohio Henry 94.17 Oklahoma Custer Ohio Highland 62.78 Oklahoma Delaware Ohio Hocking 94.17 Oklahoma Dewey Ohio Holmes 94.17 Oklahoma Ellis Ohio Huron 94.17 Oklahoma Garfield Ohio Jackson 47.08 Oklahoma Garvin Ohio Jefferson 47.08 Oklahoma Grady Ohio Knox 94.17 Oklahoma Grady Ohio Knox 94.17 Oklahoma Greer Ohio Lake 313.89 Oklahoma Harmon Ohio Lawrence 47.08 Oklahoma Harmon Ohio Licking 94.17 Oklahoma Harmon Ohio Logan 62.78 Oklahoma Harsel Ohio Logan 62.78 Oklahoma Jackson Ohio Madoison 94.17 Oklahoma						31.39
Ohio Highľand 62.78 Oklahoma Delaware Ohio Hocking 94.17 Oklahoma Dewey Ohio Holmes 94.17 Oklahoma Ellis Ohio Huron 94.17 Oklahoma Garfield Ohio Jackson 47.08 Oklahoma Garvin Ohio Jefferson 47.08 Oklahoma Grady Ohio Knox 94.17 Oklahoma Grady Ohio Lake 313.89 Oklahoma Greer Ohio Lake 313.89 Oklahoma Harmon Ohio Licking 94.17 Oklahoma Harmon Ohio Licking 94.17 Oklahoma Harmon Ohio Logan 62.78 Oklahoma Haskell Ohio Lorain 94.17 Oklahoma Haskell Ohio Lucas 94.17 Oklahoma Jackson Ohio Madison 94.17 Oklahoma						31.39 15.69
Ohio Hocking 94.17 Oklahoma Dewey Ohio Holmes 94.17 Oklahoma Ellis Ohio Huron 94.17 Oklahoma Garfield Ohio Jackson 47.08 Oklahoma Garvin Ohio Jefferson 47.08 Oklahoma Grady Ohio Knox 94.17 Oklahoma Greart Ohio Lake 313.89 Oklahoma Greer Ohio Lawrence 47.08 Oklahoma Harmon Ohio Licking 94.17 Oklahoma Harper Ohio Logan 62.78 Oklahoma Hasper Ohio Lorain 94.17 Oklahoma Hughes Ohio Lucas 94.17 Oklahoma Jefferson Ohio Madison 94.17 Oklahoma Jefferson Ohio Madison 94.17 Oklahoma Jefferson Ohio Marion 62.78 Oklahoma <td></td> <th></th> <td></td> <td></td> <td></td> <td>47.08</td>						47.08
Ohio Holmes 94.17 Oklahoma Ellis Ohio Huron 94.17 Oklahoma Garfield Ohio Jackson 47.08 Oklahoma Garvin Ohio Jefferson 47.08 Oklahoma Grady Ohio Knox 94.17 Oklahoma Graed Ohio Lake 313.89 Oklahoma Greer Ohio Lawrence 47.08 Oklahoma Harmon Ohio Licking 94.17 Oklahoma Harron Ohio Logan 62.78 Oklahoma Harsell Ohio Lorain 94.17 Oklahoma Haskell Ohio Lorain 94.17 Oklahoma Haskell Ohio Lorain 94.17 Oklahoma Jackson Ohio Madison 94.17 Oklahoma Jackson Ohio Madison 94.17 Oklahoma Jackson Ohio Marion 62.78 Oklahoma						15.69
Ohio Huron 94.17 Oklahoma Garfield Ohio Jackson 47.08 Oklahoma Garvin Ohio Jefferson 47.08 Oklahoma Grady Ohio Knox 94.17 Oklahoma Grant Ohio Lake 313.89 Oklahoma Harmon Ohio Lawence 47.08 Oklahoma Harmon Ohio Licking 94.17 Oklahoma Harmon Ohio Logan 62.78 Oklahoma Haskell Ohio Lorain 94.17 Oklahoma Hughes Ohio Lucas 94.17 Oklahoma Hughes Ohio Madison 94.17 Oklahoma Jefferson Ohio Madison 94.17 Oklahoma Jefferson Ohio Marion 62.78 Oklahoma Kay Ohio Marion 62.78 Oklahoma Kingfisher Ohio Medina 156.94 Oklahoma <td>I</td> <th>1 0</th> <td></td> <td></td> <td></td> <td>15.69</td>	I	1 0				15.69
Ohio Jefferson 47.08 Oklahoma Grady Ohio Knox 94.17 Oklahoma Grant Ohio Lake 313.89 Oklahoma Greer Ohio Lawrence 47.08 Oklahoma Harmon Ohio Licking 94.17 Oklahoma Harper Ohio Logan 62.78 Oklahoma Haskell Ohio Lorain 94.17 Oklahoma Hughes Ohio Lucas 94.17 Oklahoma Jackson Ohio Madison 94.17 Oklahoma Jefferson Ohio Mahoning 94.17 Oklahoma Jefferson Ohio Mahoning 94.17 Oklahoma Kay Ohio Marion 62.78 Oklahoma Kay Ohio Medina 156.94 Oklahoma Kingfisher Ohio Mercer 94.17 Oklahoma Kiowa Ohio Mercer 94.17 Oklahoma						31.39
Ohio Knox 94.17 Oklahoma Grant Ohio Lake 313.89 Oklahoma Greer Ohio Lawrence 47.08 Oklahoma Harmon Ohio Licking 94.17 Oklahoma Harmon Ohio Logan 62.78 Oklahoma Haskell Ohio Lorain 94.17 Oklahoma Hughes Ohio Lucas 94.17 Oklahoma Jackson Ohio Madison 94.17 Oklahoma Jefferson Ohio Mahoning 94.17 Oklahoma Jefferson Ohio Marion 62.78 Oklahoma Kay Ohio Medina 156.94 Oklahoma Kingfisher Ohio Mercer 94.17 Oklahoma Kingfisher Ohio Miami 94.17 Oklahoma Latiner Ohio Mercer 94.17 Oklahoma Le Flore Ohio Miami 94.17 Oklahoma	Jac	Jackson	47.08	Oklahoma	Garvin	31.39
Ohio Lake 313.89 Oklahoma Greer Ohio Lawrence 47.08 Oklahoma Harmon Ohio Licking 94.17 Oklahoma Harper Ohio Logan 62.78 Oklahoma Haskell Ohio Lorain 94.17 Oklahoma Hughes Ohio Lucas 94.17 Oklahoma Jackson Ohio Madison 94.17 Oklahoma Jefferson Ohio Marion 62.78 Oklahoma Kay Ohio Medina 156.94 Oklahoma Kingfisher Ohio Meigs 47.08 Oklahoma Kiowa Ohio Miami 94.17 Oklahoma Le Flore Ohio Miami 94.17 Oklahoma Lincoln Ohio Morgan 47.08 Oklahoma Le Flore Ohio Morgan 47.08 Oklahoma Lincoln Ohio Morgan 47.08 Oklahoma				Oklahoma	Grady	31.39
Ohio Lawrence 47.08 Oklahoma Harmon Ohio Licking 94.17 Oklahoma Harper Ohio Logan 62.78 Oklahoma Haskell Ohio Lorain 94.17 Oklahoma Hughes Ohio Lucas 94.17 Oklahoma Jackson Ohio Madison 94.17 Oklahoma Jefferson Ohio Mahoning 94.17 Oklahoma Kay Ohio Marion 62.78 Oklahoma Kay Ohio Medina 156.94 Oklahoma Kingfisher Ohio Meigs 47.08 Oklahoma Latimer Ohio Mizami 94.17 Oklahoma Le Flore Ohio Monroe 47.08 Oklahoma Le Flore Ohio Monroe 47.08 Oklahoma Lincoln Ohio Morgan 47.08 Oklahoma Love Ohio Muskingum 62.78 Oklahoma						15.69
Ohio Licking 94.17 Oklahoma Harper Ohio Logan 62.78 Oklahoma Haskell Ohio Lorain 94.17 Oklahoma Hughes Ohio Lucas 94.17 Oklahoma Jackson Ohio Madison 94.17 Oklahoma Jefferson Ohio Mahoning 94.17 Oklahoma Johnston Ohio Marion 62.78 Oklahoma Kay Ohio Medina 156.94 Oklahoma Kingfisher Ohio Mercer 94.17 Oklahoma Kingfisher Ohio Miami 94.17 Oklahoma Latimer Ohio Miami 94.17 Oklahoma Le Flore Ohio Miami 94.17 Oklahoma Le Flore Ohio Monroe 47.08 Oklahoma Lincoln Ohio Morgan 47.08 Oklahoma Love Ohio Morgan 47.08 Oklaho	I					15.69
Ohio Logan 62.78 Oklahoma Haskell Ohio Lorain 94.17 Oklahoma Hughes Ohio Lucas 94.17 Oklahoma Jackson Ohio Madison 94.17 Oklahoma Jefferson Ohio Mahoning 94.17 Oklahoma Johnston Ohio Marion 62.78 Oklahoma Kay Ohio Medina 156.94 Oklahoma Kingfisher Ohio Meigs 47.08 Oklahoma Kiowa Ohio Miami 94.17 Oklahoma Le Flore Ohio Miami 94.17 Oklahoma Le Flore Ohio Montgomery 47.08 Oklahoma Lincoln Ohio Montgomery 156.94 Oklahoma Lincoln Ohio Morgan 47.08 Oklahoma Love Ohio Morrow 62.78 Oklahoma Maryas						15.69
Ohio Lorain 94.17 Oklahoma Hughes Ohio Lucas 94.17 Oklahoma Jackson Ohio Madison 94.17 Oklahoma Jefferson Ohio Mahoning 94.17 Oklahoma Johnston Ohio Marion 62.78 Oklahoma Kay Ohio Medina 156.94 Oklahoma Kingfisher Ohio Meigs 47.08 Oklahoma Kiowa Ohio Mercer 94.17 Oklahoma Latimer Ohio Miami 94.17 Oklahoma Le Flore Ohio Monroe 47.08 Oklahoma Lincoln Ohio Montgomery 156.94 Oklahoma Logan Ohio Morgan 47.08 Oklahoma Love Ohio Muskingum 62.78 Oklahoma Major Ohio Noble 47.08 Oklahoma Mayes						15.69 31.39
Ohio Lucas 94.17 Oklahoma Jackson Ohio Madison 94.17 Oklahoma Jefferson Ohio Mahoning 94.17 Oklahoma Johnston Ohio Marion 62.78 Oklahoma Kay Ohio Medina 156.94 Oklahoma Kingfisher Ohio Meigs 47.08 Oklahoma Kiowa Ohio Mercer 94.17 Oklahoma Latimer Ohio Miami 94.17 Oklahoma Le Flore Ohio Monroe 47.08 Oklahoma Lincoln Ohio Morgan 47.08 Oklahoma Love Ohio Morgan 47.08 Oklahoma Love Ohio Muskingum 62.78 Oklahoma Marshall Ohio Noble 47.08 Oklahoma Mayes						15.69
Ohio Madison 94.17 Oklahoma Jefferson Ohio Mahoning 94.17 Oklahoma Johnston Ohio Marion 62.78 Oklahoma Kay Ohio Medina 156.94 Oklahoma Kingfisher Ohio Meigs 47.08 Oklahoma Kiowa Ohio Mercer 94.17 Oklahoma Latiner Ohio Miami 94.17 Oklahoma Le Flore Ohio Monroe 47.08 Oklahoma Lincoln Ohio Mortgomery 156.94 Oklahoma Loyan Ohio Morgan 47.08 Oklahoma Love Ohio Morrow 62.78 Oklahoma Major Ohio Muskingum 62.78 Oklahoma Marshall Ohio Noble 47.08 Oklahoma Mayes						15.69
Ohio Mahoning 94.17 Oklahoma Johnston Ohio Marion 62.78 Oklahoma Kay Ohio Medina 156.94 Oklahoma Kingfisher Ohio Meigs 47.08 Oklahoma Kiowa Ohio Mercer 94.17 Oklahoma Latimer Ohio Miami 94.17 Oklahoma Le Flore Ohio Monroe 47.08 Oklahoma Lincoln Ohio Montgomery 156.94 Oklahoma Logan Ohio Morgan 47.08 Oklahoma Love Ohio Morrow 62.78 Oklahoma Major Ohio Muskingum 62.78 Oklahoma Mayes						15.69
Ohio Marion 62.78 Oklahoma Kay Ohio Medina 156.94 Oklahoma Kingfisher Ohio Meigs 47.08 Oklahoma Kinowa Ohio Mercer 94.17 Oklahoma Latimer Ohio Miami 94.17 Oklahoma Le Flore Ohio Monroe 47.08 Oklahoma Lincoln Ohio Morgan 47.08 Oklahoma Loyan Ohio Morrow 62.78 Oklahoma Love Ohio Muskingum 62.78 Oklahoma Marshall Ohio Noble 47.08 Oklahoma Mayes	I					31.39
Ohio Medina 156.94 Oklahoma Kingfisher Ohio Meigs 47.08 Oklahoma Kiowa Ohio Mercer 94.17 Oklahoma Latimer Ohio Miami 94.17 Oklahoma Le Flore Ohio Monroe 47.08 Oklahoma Lincoln Ohio Mortgomery 156.94 Oklahoma Logan Ohio Morgan 47.08 Oklahoma Love Ohio Morrow 62.78 Oklahoma Major Ohio Muskingum 62.78 Oklahoma Marshall Ohio Noble 47.08 Oklahoma Mayes						31.39
Ohio Meigs 47.08 Oklahoma Kiowa Ohio Mercer 94.17 Oklahoma Latimer Ohio Miami 94.17 Oklahoma Le Flore Ohio Monroe 47.08 Oklahoma Lincoln Ohio Montgomery 156.94 Oklahoma Logan Ohio Morgan 47.08 Oklahoma Love Ohio Morrow 62.78 Oklahoma Major Ohio Muskingum 62.78 Oklahoma Marshall Ohio Noble 47.08 Oklahoma Mayes						31.39
Ohio Miami 94.17 Oklahoma Le Flore Ohio Monroe 47.08 Oklahoma Lincoln Ohio Montgomery 156.94 Oklahoma Logan Ohio Morgan 47.08 Oklahoma Love Ohio Morrow 62.78 Oklahoma Major Ohio Muskingum 62.78 Oklahoma Marshall Ohio Noble 47.08 Oklahoma Mayes	Me	Meigs	47.08	Oklahoma	Kiowa	15.69
Ohio Monroe 47.08 Oklahoma Lincoln Ohio Montgomery 156.94 Oklahoma Logan Ohio Morgan 47.08 Oklahoma Love Ohio Morrow 62.78 Oklahoma Major Ohio Muskingum 62.78 Oklahoma Marshall Ohio Noble 47.08 Oklahoma Mayes						31.39
Ohio Montgomery 156.94 Oklahoma Logan Ohio Morgan 47.08 Oklahoma Love Ohio Morrow 62.78 Oklahoma Major Ohio Muskingum 62.78 Oklahoma Marshall Ohio Noble 47.08 Oklahoma Mayes	I					31.39
Ohio Morgan 47.08 Oklahoma Love Ohio Morrow 62.78 Oklahoma Major Ohio Muskingum 62.78 Oklahoma Marshall Ohio Noble 47.08 Oklahoma Mayes						31.39
Ohio Morrow 62.78 Oklahoma Major Ohio Muskingum 62.78 Oklahoma Marshall Ohio Noble 47.08 Oklahoma Mayes	I	1 0 1				31.39
Ohio Muskingum 62.78 Oklahoma Marshall Ohio Noble 47.08 Oklahoma Mayes	I	0				31.39
Ohio						15.69
						31.39 31.39
Ohio						31.39
Ohio Paulding 62.78 Oklahoma McCurtain McCurtain						31.39
Ohio Perry 62.78 Oklahoma McIntosh McIntosh						31.39

Oklahoma Musk Oklahoma Noble Oklahoma Noble Oklahoma Oklahoma Oklahoma Oklahoma Oklahoma Oklahoma Oklahoma Ottaw Oklahoma Patm Oklahoma Pawn Oklahoma Pittsb Oklahoma Ponte Oklahoma Potta Oklahoma Potta Oklahoma Potta Oklahoma Roge Oklahoma Sequ Oklahoma Sequ Oklahoma Sequ Oklahoma Tillm Oklahoma Wash Oklahoma Woo Oklahoma Woo	ay kogee	31.39 31.39 31.39 31.39 62.78 31.39 15.69 47.08 31.39 31.39 31.39 31.39 47.08 31.39 47.08 31.39 47.08	Pennsylvania	Chester	156 94 62 156 94 313 47 47 94 94 156 627 94 62
Oklahoma Musk Oklahoma Noble Oklahoma Noble Oklahoma Oklahoma Oklahoma Oklahoma Oklahoma Oklahoma Oklahoma Oklahoma Oklahoma Osag Oklahoma Osag Oklahoma Osag Oklahoma Pawn Oklahoma Pawn Oklahoma Pawn Oklahoma Potta Oklahoma Roge Oklahoma Sequ Oklahoma Sequ Oklahoma Steph Oklahoma Steph Oklahoma Tillma Oklahoma Tillma Oklahoma Tillma Oklahoma Wash Oklahoma Wash Oklahoma Oklahoma Oklahoma Oklahoma Oklahoma Tolahoma Oklahoma Oklahoma Tolahoma Oklahoma Oklahoma Oklahoma Oklahoma Oklahoma Wash Oklahoma Wash Oklahoma Wooc Oklahoma Wooc Oregon Colats Oregon Colats Oregon Colats Oregon Colats Oregon Doug Oregon Doug Oregon Doug Oregon Doug Oregon Doug Oregon Harne Oregon Jacks Oregon Lane Oregon Lane Oregon Mario Oregon Tillar	kogee e e	31.39 31.39 31.39 62.78 31.39 15.69 47.08 31.39 31.39 31.39 47.08 31.39 47.08 31.39 47.08	Pennsylvania	Cambria Cameron Carbon Centre Chester Clarion Clearfield Clinton Columbia Crawford Cumberland Dauphin Delaware Elk Erie Fayette Forest	62 156 94 313 47 47 94 47 156 627 94 62 47 62
Oklahoma Noble Oklahoma Nowa Oklahoma Oklahoma Oklahoma Oklahoma Oklahoma Oklam Oklahoma Osag Oklahoma Ottaw Oklahoma Paym Oklahoma Pittsb Oklahoma Potta Oklahoma Potta Oklahoma Push Oklahoma Push Oklahoma Semi Oklahoma Semi Oklahoma Semi Oklahoma Semi Oklahoma Semi Oklahoma Texa Oklahoma Wash Oklahoma Wash Oklahoma Wash Oklahoma Wooc Oklahoma Wooc <t< td=""><td>e a ata ata ata ata ata ata ata ata ata</td><td>31.39 31.39 62.78 31.39 15.69 47.08 15.69 31.39 31.39 15.69 47.08 31.39 47.08 31.39 15.69</td><td>Pennsylvania Pennsylvania Pennsylvania</td><td>Carbon Centre Chester Clarion Clearfield Clinton Columbia Crawford Cumberland Dauphin Delaware Elk Erie Fayette Forest</td><td>156 94 313 47 47 94 94 47 156 627 94 62 47</td></t<>	e a ata ata ata ata ata ata ata ata ata	31.39 31.39 62.78 31.39 15.69 47.08 15.69 31.39 31.39 15.69 47.08 31.39 47.08 31.39 15.69	Pennsylvania	Carbon Centre Chester Clarion Clearfield Clinton Columbia Crawford Cumberland Dauphin Delaware Elk Erie Fayette Forest	156 94 313 47 47 94 94 47 156 627 94 62 47
bklahoma Nowa bklahoma Okfula bklahoma Okfula bklahoma Oklah bklahoma Osag bklahoma Ottaw bklahoma Paym bklahoma Pittsb bklahoma Potta bklahoma Potta bklahoma Roge bklahoma Roge bklahoma Sequ bklahoma Steph bklahoma Tillma bklahoma Wash bklahoma Wash bklahoma Wash bklahoma Wash bklahoma Wash bklahoma Wash bklahoma Wooc	ata skee homa uligee ge ge wa nee hoe burg otoc awatomie mataha er Mills ers inole Joyah hens sis aa ooner hington hita ds	31.39 31.39 62.78 31.39 15.69 47.08 15.69 31.39 31.39 15.69 47.08 31.39 47.08 31.39 15.69	Pennsylvania	Carbon Centre Chester Clarion Clearfield Clinton Columbia Crawford Cumberland Dauphin Delaware Elk Erie Fayette Forest	94 313 47 47 94 47 156 156 627 94 62 47 62
Iklahoma	Iskee homa	31.39 62.78 31.39 15.69 47.08 15.69 31.39 31.39 15.69 47.08 31.39 47.08 31.39 15.69 62.78	Pennsylvania	Centre Chester Clarion Clearfield Clinton Columbia Crawford Cumberland Dauphin Delaware Elk Erie Fayette Forest	94 313 47 47 94 47 156 156 627 94 62 47 62
Ikilahoma Oklar	homa uligee ge wa nee he burg otoc mataha er Mills ers inole uoyah hens as an a ooner hington hita ds	62.78 31.39 15.69 47.08 15.69 31.39 31.39 31.39 15.69 47.08 31.39 15.69 15.69 62.78	Pennsylvania	Chester Clarion Clearfield Clinton Columbia Crawford Cumberland Dauphin Delaware Elk Erie Fayette Forest	313 47 47 94 94 156 156 627 94 62 47 62
kilahoma Okmu kilahoma Osag kilahoma Osag Osag Osag Osag Osag Osag Osag Osa	uulgee	31.39 15.69 47.08 15.69 31.39 31.39 31.39 15.69 47.08 31.39 47.08 31.39 15.69 62.78	Pennsylvania	Clarion Clearfield Clinton Columbia Crawford Cumberland Dauphin Delaware Elk Erie Fayette Forest	47 47 94 94 47 156 156 627 94 62 47
kilahoma Osag kilahoma Ottaw kilahoma Pawn kilahoma Pawn kilahoma Pawn kilahoma Pawn kilahoma Pawn kilahoma Ponto Potta Potta Kilahoma Ponto Potta Kilahoma Potta Kilahoma Potta Kilahoma Roge kilahoma Roge kilahoma Semin Kilahoma Semin Kilahoma Semin Kilahoma Steph kilahoma Texas Kilahoma Tillma Kilahoma Tillma Kilahoma Wags Kilahoma Wags Kilahoma Wags Kilahoma Wags Kilahoma Wooch Kilahoma Kilahoma Kilahoma Kilahoma Wooch Kilahoma Wooch Kilahoma Wooch Kilahoma Kilahoma Kilahoma Kilahoma Kilahoma Wooch Kilahoma Kilahoma Kilahoma Kilahoma Wooch Kilahoma Wooch Kilahoma Wooch Kilahoma Kilahoma Wooch Kilahoma Kilahoma Wooch Kilahoma Wooch Kilahoma Wooch Kilahoma Wooch Kilahoma Wooch Kilahoma Kilahoma Wooch Kilahoma Kilahoma Wooch Kilahom	ge wa nee mee mee mee mee mee mee mee mee mee	15.69 47.08 15.69 31.39 31.39 31.39 15.69 47.08 31.39 47.08 31.39 15.69 62.78	Pennsylvania	Clearfield Clinton Columbia Crawford Cumberland Dauphin Delaware Elk Erie Fayette Forest	47 94 94 47 156 156 627 94 62 47
iklahoma Ottawiklahoma Pawmiklahoma Pawmiklahoma Paymiklahoma Paymiklahoma Paymiklahoma Ponta Paymiklahoma Potta Ponta P	wa mee mee mee mee mee mee mee mee mee me	47.08 15.69 31.39 31.39 31.39 15.69 47.08 31.39 47.08 31.39 15.69 62.78	Pennsylvania	Clinton Columbia Crawford Cumberland Dauphin Delaware Elk Erie Fayette Forest	94 94 47 156 156 627 94 62 47
	nee he burg burg otoc awatomie mataha er Mills ers inole Joyah hens ss sa a oner hington hita ds	15.69 31.39 31.39 31.39 31.39 15.69 47.08 31.39 47.08 31.39 15.69 62.78	Pennsylvania	Columbia Crawford Cumberland Dauphin Delaware Elk Erie Fayette Forest	94 47 156 156 627 94 62 47 62
Iklahoma	ne burg	31.39 31.39 31.39 15.69 47.08 31.39 47.08 31.39 15.69 62.78	Pennsylvania	Crawford Cumberland Dauphin Delaware Elk Erie Fayette Forest	47 156 156 627 94 62 47 62
klahoma Pittsb klahoma Pontta klahoma Potta klahoma Push klahoma Roge klahoma Seqiu klahoma Seqiu klahoma Stepri klahoma Tillma klahoma Tillma klahoma Wash klahoma Wash klahoma Wooc clade regon </td <td>burg otoc wavatomie mataha er Mills ers inole uoyah hens as aa ooner hington hita dos</td> <td>31.39 31.39 31.39 15.69 47.08 31.39 47.08 31.39 15.69 15.69</td> <td>Pennsylvania Pennsylvania Pennsylvania Pennsylvania Pennsylvania Pennsylvania Pennsylvania Pennsylvania Pennsylvania Pennsylvania Pennsylvania</td> <td>Cumberland Dauphin Delaware Elk Erie Fayette Forest</td> <td>156 156 627 94 62 47 62</td>	burg otoc wavatomie mataha er Mills ers inole uoyah hens as aa ooner hington hita dos	31.39 31.39 31.39 15.69 47.08 31.39 47.08 31.39 15.69 15.69	Pennsylvania	Cumberland Dauphin Delaware Elk Erie Fayette Forest	156 156 627 94 62 47 62
klahoma Pontta klahoma Potta klahoma Potta klahoma Roge klahoma Semi klahoma Sequ klahoma Steph klahoma Tillma klahoma Tillma klahoma Tillma klahoma Wash klahoma Wash klahoma Wooc regon Bakei regon Clats regon Clats regon Colur regon Colur regon Colur regon Colur regon Grant regon Grant regon Hame regon Hame	otoc watomie mmataha er Mills ers inole Joyah hens as a oner hington hita	31.39 31.39 15.69 15.69 47.08 31.39 47.08 31.39 15.69 15.69 62.78	Pennsylvania	Dauphin Delaware Elk Erie Fayette Forest	156 627 94 62 47 62
klahoma Potta klahoma Pushi klahoma Roge klahoma Roge klahoma Semin klahoma Steph klahoma Texas klahoma Tillma klahoma Tulsa klahoma Wags klahoma Wash klahoma Wooc klahoma Wooc klahoma Wooc regon Bakei regon Clack regon Colur regon Colur regon Colur regon Colur regon Colur regon Doug regon Doug regon Doug regon Grant regon Hane regon Jacks regon Jacks regon Jacks regon Lake regon Lake	awatomie mataha mataha er Mills ers inole uoyah hens ss aa oner hington hita ds	31.39 15.69 15.69 47.08 31.39 47.08 31.39 15.69 15.69	Pennsylvania	Delaware Elk Erie Fayette Forest	627 94 62 47 62
klahoma Pushi klahoma Roge klahoma Roge klahoma Seqii klahoma Steph klahoma Tillma klahoma Tillma klahoma Wash klahoma Wash klahoma Wooc regon Bakei regon Clack regon Clats regon Colur regon Crost regon Crost regon Crost regon Doug regon Grant regon Harne regon Jacks regon Jacks regon Lake regon Lake regon Lake	ımataha er Mills ers inole ıoyah hens as a oner hington hita ds	15.69 15.69 47.08 31.39 47.08 31.39 15.69 15.69 62.78	Pennsylvania	Elk	94 62 47 62
klahoma Pushi klahoma Roge klahoma Roge klahoma Seqii klahoma Steph klahoma Tillma klahoma Tillma klahoma Wash klahoma Wash klahoma Wooc regon Bakei regon Clack regon Clats regon Colur regon Crost regon Crost regon Crost regon Doug regon Grant regon Harne regon Jacks regon Jacks regon Lake regon Lake regon Lake	ımataha er Mills ers inole ıoyah hens as a oner hington hita ds	15.69 15.69 47.08 31.39 47.08 31.39 15.69 15.69 62.78	Pennsylvania	Elk	94 62 47 62
klahoma Roge klahoma Roge klahoma Sequiklahoma klahoma Sequiklahoma klahoma Tillma klahoma Tillma klahoma Tulsa klahoma Wash klahoma Wash klahoma Wash klahoma Wooc klapon Clatis kegon Clatis kegon Colur regon Coos regon Coos regon Garant regon Harm regon Jase regon Jase <td>er Mills ers inole loyah hens ss an a oner hington hita ds</td> <td>15.69 47.08 31.39 47.08 31.39 15.69 15.69 62.78</td> <td>Pennsylvania</td> <td>Erie</td> <td>62 47 62</td>	er Mills ers inole loyah hens ss an a oner hington hita ds	15.69 47.08 31.39 47.08 31.39 15.69 15.69 62.78	Pennsylvania	Erie	62 47 62
klahoma Roge klahoma Semin klahoma Steph klahoma Steph klahoma Texas klahoma Tillma klahoma Tulsa klahoma Wags klahoma Wash klahoma Wooc klapon Clats kegon Clats kegon Colur kegon Hane kegon Jacks kergon Lake	ers inole	47.08 31.39 47.08 31.39 15.69 15.69 62.78	Pennsylvania Pennsylvania Pennsylvania Pennsylvania Pennsylvania	Fayette	47 62
klahoma Semii klahoma Sequi klahoma Steph klahoma Tillma klahoma Tillma klahoma Wago klahoma Wash klahoma Wood regon Bakei regon Clack regon Clack regon Colur regon Colur regon Doug regon Doug regon Doug regon Grant regon Harne regon Jacke regon Jacke regon Lake regon Lake regon Lake	inole Joyah Hens IS	31.39 47.08 31.39 15.69 15.69 62.78	Pennsylvania Pennsylvania Pennsylvania Pennsylvania	Forest	62
klahoma Sequ klahoma Steph klahoma Tillma klahoma Tillma klahoma Tillma klahoma Wash klahoma Wash klahoma Wash klahoma Wood klapon Bente kegon Bente kegon Clats kegon Colur kegon Coos kegon Coos kegon Doug kegon Jacks kegon Jacks kegon Jacks kegon Lake kegon Lake kegon Lake kegon Lake	Joyah	47.08 31.39 15.69 15.69 62.78	Pennsylvania Pennsylvania Pennsylvania		
klahoma Steph klahoma Texas klahoma Tillma klahoma Tulsa klahoma Wags klahoma Wash klahoma Wooc klahoma Wooc klahoma Wooc regon Bakei regon Clack regon Clack regon Cos regon Cos regon Coro regon Doug regon Grant regon Grant regon Hann regon Jacke regon Jacke regon Jacke regon Jacke regon Jacke regon Lake regon Lino regon Lino regon Mario regon Mario regon Mario regon Mario rego	hens	31.39 15.69 15.69 62.78	Pennsylvania Pennsylvania	Franklin	
klahoma Texas klahoma Tillma klahoma Tillma klahoma Wago klahoma Wash klahoma Wooc klahoma Wooc klahoma Wooc klahoma Wooc klahoma Wooc klahoma Wooc regon Bakei regon Clack regon Clack regon Colur regon Cross regon Cross regon Doug regon Doug regon Grant regon Harne regon Jacks regon Jacks regon Lake regon Lane regon Lane regon Mario regon Mario regon Mario regon Mario regon Mario <	as	15.69 15.69 62.78	Pennsylvania		156
klahoma Tillma klahoma Tulsa klahoma Wagk klahoma Wash klahoma Wash klahoma Wood klagon Clatsk klagon Colur klagon Coos klagon Coos klagon Coos klagon Coos klagon Garant klagon Harm klagon Harm klagon Harm klagon Lake klagon Lake klagon Lake klagon Mario klagon Mario klagon Mario klagon Mario klagon Mario	an	15.69 62.78		Fulton	62
klahoma Tulsa klahoma Wags klahoma Wash klahoma Wash klahoma Wood klahoma Wood eegon Baket regon Bentt regon Clats regon Clats regon Coos regon Coos regon Croot regon Doug regon Doug regon Grant regon Hand regon Jacket regon Jacket regon Jacket regon Laket regon Lanet regon Linco regon Mariot regon Mariot regon Mariot regon Mariot regon Mariot regon Mariot regon Morrot regon Polk <	ahingtonhitads	62.78		Greene	31
dahoma Tulsa dahoma Wagc dahoma Wash dahoma Wash dahoma Wood dahoma Wooc egon Bakei egon Bente egon Clack egon Clack egon Coos egon Coro egon Coro egon Doug egon Gallila egon Grant egon Hood egon Jacke egon Jacke egon Jacke egon Lake egon Lane egon Lino egon Mario egon Mario egon Mario egon Mario egon More egon More egon Polk egon Sherr egon Tillar <td>ahingtonhitads</td> <td>62.78</td> <td>Pennsylvania</td> <td>Huntingdon</td> <td>62</td>	ahingtonhitads	62.78	Pennsylvania	Huntingdon	62
dahoma Wago dahoma Wash dahoma Wash dahoma Wood dahoma Wood dahoma Wood dahoma Wood degon Bake eegon Bentc eegon Clack eegon Colur eegon Coos eegon Curry eegon Doug eegon Doug eegon Gallia eegon Harne eegon Jacks eegon Jacks eegon Jacks eegon Lake eegon Lane eegon Linco eegon Malin eegon Mario eegon Morc eegon Morc eegon Polk eegon Sherr eegon Tillar	onerhingtonhitads		Pennsylvania	Indiana	62
dahoma Wash dahoma Wash dahoma Wash dahoma Wooc dahoma Wooc egon Bakei egon Bento egon Clack egon Colur egon Coolur egon Crool egon Crool egon Doug egon Doug egon Grant egon Harne egon Jacks egon Lake egon Lake egon Lane egon Morro egon Morro egon Morro egon Morro egon Polk egon Sherr </td <td>hingtonhitads</td> <td></td> <td>Pennsylvania</td> <td>Jefferson</td> <td>47</td>	hingtonhitads		Pennsylvania	Jefferson	47
klahoma Wash klahoma Wood klahoma Wood klahoma Wood klahoma Wood kegon Baket kegon Clack kegon Clack kegon Colut kegon Coor kegon Crool kegon Deug kegon Deug kegon Grant kegon Jacke kegon Jacke kegon Lake kegon Lane kegon Lane kegon Mario kegon Mario kegon Morra kegon Polk kegon Sherr kegon Sherr kegon Tillar kegon Hord kegon Lane kegon Lane kegon Morra kegon Lane kegon	hitads	31.39	Pennsylvania	Juniata	94
klahoma Wood klahoma Wood klahoma Wood kegon Bakkel egon Bentc egon Clack egon Colur egon Coolur egon Coolur egon Coolur egon Curry egon Desc egon Doug egon Galliar egon Harne egon Jacks egon Jacks egon Jacks egon Lake egon Lake egon Linco egon Maine egon Mario egon Mario egon Morr egon Polk egon Sherr egon Tillar	ds	15.69			
klahoma Wood egon Bakei egon Bente egon Clack egon Clack egon Colur egon Coos egon Crool egon Crool egon Descl egon Doug egon Grant egon Harm egon Jacks egon Jacks egon Jeffer egon Jose egon Lake egon Lane egon Linc egon Mario egon Morr egon Morr egon Polk egon Sherr egon Tillar			Pennsylvania	Lackawanna	94
egon Bakei egon Bentic egon Clats egon Clats egon Colur egon Colur egon Crook egon Crook egon Descl egon Descl egon Gilliar egon Gilliar egon Hame egon Jacks egon Josep egon Josep egon Klam egon Lake egon Lane egon Linn egon Mario egon Mario egon Miltn egon Polk egon Sherr egon Tillar		15.69	Pennsylvania	Lancaster	313
egon Bente egon Clack egon Clats egon Colur egon Coolur egon Crool egon Crool egon Curry egon Doug egon Gillia egon Grant egon Harne egon Jacks egon Jaffer egon Jase egon Lake egon Lane egon Linn egon Mario egon Mario egon Morra egon Multn egon Polk egon Tillar egon Tillar	dward	15.69	Pennsylvania	Lawrence	62
egon Clack egon Clats egon Colur egon Coos egon Croos egon Croos egon Croos egon Desci egon Gillian egon Gillian egon Harne egon Jacks egon Jacks egon Jacks egon Jacks egon Jacks egon Lake egon Linc egon Linc egon Mario egon Morra egon Morra egon Polk egon Sherr egon Tillar	er	15.69	Pennsylvania	Lebanon	156
egon Clatse egon Colur egon Crook egon Crook egon Crook egon Desc egon Doug egon Gillia egon Harm egon Hann egon Jacks egon Jeffer egon Jose egon Jaks egon Lake egon Lino egon Lino egon Mario egon Mario egon Multn egon Polk egon Tillar egon Tillar	on	156.94	Pennsylvania	Lehigh	156
egon Clatse egon Colur egon Crook egon Crook egon Curry egon Desc egon Doug egon Gillia egon Harm egon Hann egon Jacks egon Jeffer egon Jaks egon Jaks egon Lake egon Linc egon Linc egon Mario egon Mario egon Multn egon Polk egon Tillar egon Tillar	kamas	313.89	Pennsylvania	Luzerne	94
egon Colur egon Coos egon Crook egon Curry egon Desc egon Doug egon Gillian egon Grant egon Harm egon Jacks egon Jacks egon Jeffer egon Jose egon Lake egon Linc egon Linc egon Mario egon Morra egon Multn egon Polk egon Sherr egon Tillar egon Tillar	sop	94.17	Pennsylvania	Lycoming	62
egon Coos egon Crook egon Crook egon Curry egon Doug egon Gillian egon Grant egon Harne egon Hoe egon Jose egon Jose egon Jose egon Lake egon Linc egon Linc egon Mario egon Morra egon Morra egon Morra egon Polk egon Sherr egon Tillar egon Umat	mbia	156.94	Pennsylvania	McKean	31
egon Crool egon egon Curry gon egon Descleegon egon Gallilar gon egon Galiliar gon egon Harne gon egon Hood gon egon Jefter gon egon Josep gon egon Lake gon egon Lane gon egon Linco gon egon Malne gon egon Mario gon egon Multn egon Polk gon egon Tillar gon egon Tillar gon egon Tillar gon					
egon Curry egon Desci egon Doug egon Gilliar egon Grant egon Harne egon Hood egon Jeffer egon Jeffer egon Jeffer egon Lake egon Lane egon Linco egon Main egon Morre egon Multn egon Polk egon Sherr egon Tillar egon Umat	S	94.17	Pennsylvania	Mercer	62
egon Desciper egon Doug egon Gilliar egon Grant egon Harne egon Harne egon Jacks egon Jeffer egon Jacks egon Jacks egon Lifam egon Lake egon Linco egon Linn egon Mario egon Morra egon Multn egon Sherr egon Tillar egon Umat	ok	15.69	Pennsylvania	Mifflin	94
egon Doug egon Gillian egon Grant egon Harne egon Hood egon Jacks egon Jeffer egon Josep egon Lake egon Lane egon Linn egon Malne egon Morro egon Morro egon Polk egon Sherr egon Tillar egon Umat	y	62.78	Pennsylvania	Monroe	156
regon Gillian regon Grant regon Harm regon Jose regon Jose regon Jose regon Jose regon Lake regon Lane regon Lino regon Malh regon Morro regon Morro regon Polk regon Sherr regon Tillan regon Umat	chutes	156.94	Pennsylvania	Montgomery	627
egon Gilliar egon Grant egon Harm egon Haod egon Jacks egon Josep egon Klam egon Lake egon Lane egon Linco egon Mario egon Mario egon Morr egon Morr egon Polk egon Polk egon Him egon Harm egon Morr egon Mith egon Mith egon Holk egon Hillar	glas	62.78	Pennsylvania	Montour	94
regon Grant regon Harne regon Hood regon Jacks regon Jeffer regon Josep regon Klam regon Lake regon Linco regon Linco regon Malh regon Morro regon Multn regon Polk regon Sherr regon Tillar regon Umat	im	7.85	Pennsylvania	Northampton	156
regon Harne regon Hood regon Jacks regon Jeffer regon Josep regon Klam regon Lake regon Linco regon Linco regon Malhe regon Morro regon Multn regon Polk regon Sherr regon Tillam regon Umat	nt	7.85	Pennsylvania	Northumberland	94
regon Hood regon Jacks regon Jeffer regon Josep regon Klam regon Lake regon Linco regon Linn regon Malin regon Morro regon Morro regon Polk regon Sherr regon Tillar regon Umat	ney	7.85	Pennsylvania	Perry	94
egon Jacks egon Jeffer egon Josep egon Klam egon Lake egon Lane egon Linco egon Malh egon Morra egon Morra egon Polk egon Sherr egon Tillar egon Umat	d River	313.89			941
regon Jeffer egon Josep egon Klam egon Lake egon Lane egon Linco egon Malne egon Mario egon Morro egon Multn egon Polk egon Sherr egon Tillam egon Umat			Pennsylvania	Philadelphia	
egon Josep egon Klam egon Lake egon Lane egon Linco egon Malhe egon Mario egon Morro egon Multn egon Polk egon Sherr egon Tillar egon Umat	son	94.17	Pennsylvania	Pike	94
egon Klami egon Lake egon Lane egon Linco egon Malh egon Mario egon Morr egon Multn egon Polk egon Sherr egon Tillar egon Umat	erson	15.69	Pennsylvania	Potter	47
regon Lake egon Lane egon Linco egon Linn egon Malh egon Morre egon Multn egon Polk egon Sherr egon Tillar egon Umat	phine	156.94	Pennsylvania	Schuylkill	94
egon Lane egon Linco egon Linn egon Malhe egon Morro egon Motro egon Polk egon Sherr egon Tillar egon Umat	nath	31.39	Pennsylvania	Snyder	94
egon Lane egon Linco egon Linn egon Malhe egon Morro egon Motro egon Polk egon Sherr egon Tillar egon Umat		15.69	Pennsylvania	Somerset	62
egon Linco egon Linn egon Malh egon Mario egon Morre egon Multn egon Polk egon Sherr egon Tillar egon Umat		156.94	Pennsylvania		62
regon Linn egon Malh regon Mario egon Morro regon Multn egon Polk regon Sherr regon Tillar egon Umat	oln	94.17	Pennsylvania	Susquehanna	62
egon Malhe egon Mario egon Morre egon Multn egon Polk egon Sherr egon Tillam egon Umat		94.17	Pennsylvania	Tioga	62
egon Mario egon Morro egon Multn egon Polk egon Sherr egon Tillar egon Umat		15.69	Pennsylvania		156
egon Morro egon Multh egon Polk egon Sherr egon Tillar egon Umat	eur			Union	
egon Multn egon Polk egon Sherr egon Tillan egon Umat	on	156.94	Pennsylvania	Venango	47
egon	ow	15.69	Pennsylvania	Warren	47
egon	nomah	313.89	Pennsylvania	Washington	62
egon Tillam egon Umat		156.94	Pennsylvania	Wayne	62
egon Tillam egon Umat	man	15.69	Pennsylvania	Westmoreland	94
egon Umat	mook	156.94	Pennsylvania	Wyoming	62
	tilla	31.39	Pennsylvania	York	156
	n	31.39	Puerto Rico	All Areas	156
		15.69	Rhode Island	Bristol	
	owa				627
	co	15.69	Rhode Island	Kent	313
	hington	313.89	Rhode Island	Newport	627
		7.85	Rhode Island	Providence	313
	eler	313.89	Rhode Island	Washington	313
	hill	156.94	South Carolina	Abbeville	62
		156.94	South Carolina	Aiken	62
	hill ns	62.78	South Carolina	Allendale	47
	hill ns gheny	94.17	South Carolina	Anderson	94
	hill ns gheny strong			Bamberg	
	hillghenygheny	62.78	South Carolina		47
	hill ms gheny strong ver	156.94	South Carolina		47
	hill		South Carolina		62
nnsylvania Bradf	hill	94.17 47.08	South Carolina	Berkeley	94

State	9	County	(Fee/acre/ YR)	State	County	(Fee/acre/ YR)
South Carolii	na	Charleston	156.94	South Dakota	Jones	7.85
South Carolin		Cherokee	62.78	South Dakota	Kingsbury	31.39
South Carolin		Chester	62.78	South Dakota	Lake	31.39
South Carolin		Chesterfield	47.08	South Dakota	Lawrence	31.39
South Carolin		Clarendon	47.08	South Dakota	Lincoln	47.08
South Carolin		Colleton	47.08	South Dakota	Lyman	15.69
South Carolin		Darlington	31.39	South Dakota	Marshall	15.69
South Carolin		Dillon	47.08	South Dakota	McCook	31.39
South Carolii		Dorchester	62.78	South Dakota	McPherson	15.69
South Carolin		Edgefield	62.78	South Dakota	Meade	7.85
South Carolin		Fairfield	47.08	South Dakota	Mellette	7.85
South Carolii		Florence	47.08	South Dakota	Miner	31.39
South Carolin		Georgetown	62.78	South Dakota	Minnehaha	47.08
South Carolii		Greenville	94.17	South Dakota	Moody	31.39
South Carolii		Greenwood	47.08	South Dakota	Pennington	15.69
South Carolii		Hampton	47.08	South Dakota	Perkins	7.85
South Carolii		Horry	62.78	South Dakota	Potter	15.69
South Carolii		Jasper	47.08	South Dakota	Roberts	31.39
South Carolin		Kershaw	62.78	South Dakota	Sanborn	15.69
South Carolin		Lancaster	62.78	South Dakota	Shannon	7.85
South Carolin		Laurens	62.78	South Dakota	Spink	15.69
South Carolii		Lee	47.08	South Dakota	Stanley	7.85
South Carolin		Lexington	94.17	South Dakota	Sully	15.69
South Carolii		Marion	47.08	South Dakota	Todd	7.85
South Carolii		Marlboro	31.39	South Dakota	Tripp	15.69
South Carolii		McCormick	94.17	South Dakota	Turner	47.08
South Carolin		Newberry	62.78	South Dakota	Union	62.78
South Carolii		Oconee	156.94	South Dakota	Walworth	15.69
South Carolin		Orangeburg	47.08	South Dakota	Yankton	31.39
South Carolin		Pickens	156.94	South Dakota	Ziebach	7.85
South Carolin		Richland	94.17	Tennessee	Anderson	156.94
South Carolin		Saluda	62.78	Tennessee	Bedford	62.78
South Carolin		Spartanburg	156.94	Tennessee	Benton	47.08
South Carolin		Sumter	62.78	Tennessee	Bledsoe	62.78
South Carolin		Union	47.08	Tennessee	Blount	156.94
South Carolii		Williamsburg	47.08	Tennessee	Bradley	156.94
South Carolin		York	156.94	Tennessee	Campbell	62.78
South Dakota		Aurora	15.69	Tennessee	Cannon	94.17
South Dakota		Beadle	15.69	Tennessee	Carroll	47.08
South Dakota		Bennett	7.85	Tennessee	Carter	94.17
South Dakota		Bon Homme	31.39	Tennessee	Cheatham	94.17
South Dakota		Brookings	31.39	Tennessee	Chester	47.08
South Dakota		Brown	31.39	Tennessee	Claiborne	47.08
South Dakota		Brule	15.69	Tennessee	Clay	47.08
South Dakota		Buffalo	7.85	Tennessee	Cocke	94.17
South Dakota		Butte	7.85	Tennessee	Coffee	94.17
South Dakota		Campbell	15.69	Tennessee	Crockett	62.78
South Dakota		Charles Mix	15.69	Tennessee	Cumberland	94.17
South Dakota		Clark	31.39	Tennessee	Davidson	313.89
South Dakota		Clay	47.08	Tennessee	Decatur	47.08
South Dakota		Codington	31.39	Tennessee	DeKalb	94.17
South Dakota		Corson	7.85	Tennessee	Dickson	94.17
South Dakota		Custer	15.69	Tennessee	Dver	62.78
South Dakota		Davison	31.39	Tennessee	Fayette	62.78
South Dakota		Day	15.69	Tennessee	Fentress	62.78
South Dakota		Deuel	31.39	Tennessee	Franklin	94.17
South Dakota		Dewey	7.85	Tennessee	Gibson	47.08
South Dakota			31.39	Tennessee	Giles	62.78
		Douglas				62.78
South Dakota South Dakota		Fall River	15.69 7.85	Tennessee Tennessee	Grainger	94.17
South Dakota		Faulk	15.69	Tennessee	Grundy	62.78
South Dakota		Grant	31.39	Tennessee Tennessee	Hamblen	156.94
South Dakota		Gregory	15.69			94.17
South Dakota			7.85	Tennessee	Hancock	62.78
South Dakota		Hamlin	31.39	Tennessee		31.39
South Dakota		Hand	15.69	Tennessee	Hardin	47.08
South Dakota		Hanson	31.39	Tennessee	Hawkins	94.17
South Dakota		Harding	7.85	Tennessee	Haywood	47.08
South Dakota		Hughes	15.69	Tennessee	Henderson	47.08
South Dakota		Hutchinson	31.39	Tennessee	Henry	47.08
South Dakota		Hyde	7.85	Tennessee	Hickman	47.08
South Dakots	a	Jackson	7.85	Tennessee	Houston	47.08
South Dakota		Jerauld	15.69	Tennessee	Humphreys	47.08

State	County	(Fee/acre/ YR)	State	County	(Fee/acr
ennessee	Jackson	47.08	Texas	Brewster	7
ennessee	Jefferson	156.94	Texas	Briscoe	7
ennessee	Johnson	94.17	Texas	Brooks	15
ennessee	Knox	156.94	Texas	Brown	31
ennessee	Lake	47.08	Texas	Burleson	47
ennessee	Lauderdale	47.08	Texas	Burnet	47
ennessee	Lawrence	47.08	Texas	Caldwell	47
ennessee	Lewis	62.78	Texas	Calhoun	31
ennessee	Lincoln	62.78	Texas	Callahan	15
ennessee	Loudon	156.94	Texas	Cameron	47
ennessee	Macon	94.17	Texas	Camp	62
ennessee	Madison	94.17	Texas	Carson	15
ennessee	Marion	62.78	Texas	Cass	47
ennessee	Marshall	62.78	Texas	Castro	31
ennessee	Maury	94.17	Texas	Chambers	31
ennessee	McMinn	94.17	Texas	Cherokee	47
ennessee	McNairy	31.39	Texas	Childress	15
ennessee	Meigs	94.17	Texas	Clay	31
ennessee	Monroe	94.17	Texas	Cochran	15
ennessee	Montgomery	62.78	Texas	Coke	15
ennessee	Moore	62.78	Texas	Coleman	15
ennessee	Morgan	62.78	Texas	Collin	94
ennessee	Obion	47.08	Texas	Collingsworth	15
ennessee	Overton	62.78	Texas	Colorado	47
ennessee	Perry	47.08	Texas	Comal	62
ennessee	Pickett	62.78	Texas	Comanche	31
ennessee	Polk	156.94	Texas	Concho	15
ennessee	Putnam	94.17	Texas	Cooke	47
ennessee	Rhea	94.17	Texas	Coryell	31
ennessee	Roane	94.17	Texas	Cottle	7
ennessee	Robertson	94.17	Texas Texas	Crane	7
ennessee	Rutherford	94.17		Crockett	7
ennessee	Scott	62.78	Texas	Crosby	15
ennessee	Sequatchie	62.78	Texas	Culberson	7
ennessee	Sevier	156.94	Texas	Dallam	15
ennessee	Shelby	156.94	Texas	Dallas	94
ennessee	Smith	62.78	Texas	Dawson	15
ennessee	Stewart	62.78	Texas	Deaf Smith	15
ennessee	Sullivan	94.17	Texas	Delta	31
ennessee	Sumner	94.17	Texas	Denton	94
ennessee	Tipton	62.78	Texas	DeWitt	31
ennessee	Trousdale	94.17	Texas	Dickens	7
ennessee	Unicoi	313.89	Texas	Dimmit	15
ennessee	Union	94.17	Texas	Donley	15
ennessee	Van Buren	62.78	Texas	Duval	31
ennessee	Warren	62.78	Texas	Eastland	31
ennessee	Washington	156.94	Texas	Ector	7
ennessee	Wayne	47.08	Texas	Edwards	15
ennessee	Weakley	47.08	Texas	El Paso	62
ennessee	White	94.17	Texas	Ellis	47
ennessee	Williamson	156.94	Texas	Erath	47
ennessee	Wilson	94.17	Texas	Falls	31
exas	Anderson	31.39	Texas	Fannin	31
exas	Andrews	7.85	Texas	Fayette	62
exas	Angelina	62.78	Texas	Fisher	15
exas	Aransas	31.39	Texas	Floyd	15
exas	Archer	15.69	Texas	Foard	15
exas	Armstrong	15.69	Texas	Fort Bend	62
exas	Atascosa	31.39	Texas	Franklin	31
exas	Austin	62.78	Texas	Freestone	31
exas	Bailey	15.69	Texas	Frio	31
exas	Bandera	47.08	Texas	Gaines	15
exas	Bastrop	47.08	Texas	Galveston	47
exas	Baylor	15.69	Texas	Garza	7
exas	Bee	31.39	Texas	Gillespie	62
exas	Bell	47.08	Texas	Glasscock	15
exas	Bexar	62.78	Texas	Goliad	31
exas	Blanco	62.78	Texas	Gonzales	31
exas	Borden	15.69	Texas	Gray	15
exas	Bosque	47.08	Texas	Grayson	62
	Bowie	47.08	Texas	Gregg	47
exasexas	Brazoria	47.08	Texas	Grimes	47

	State	County	(Fee/acre/ YR)	State	County	(Fee/acre/ YR)
Texas		Hale	15.69	Texas	Mitchell	15.69
		Hall	7.85	Texas	Montague	47.08
Texas		Hamilton	31.39	Texas	Montgomery	94.17
		Hansford	15.69	Texas	Moore	15.69
		Hardeman	15.69	Texas	Morris	31.39
_		Hardin	47.08	Texas	Motley	7.85
		Harris	94.17 31.39	Texas	Nacogdoches	47.08 31.39
		Harrison	15.69	Texas Texas	Navarro	31.39
		Hartley Haskell	15.69	Texas	Nolan	15.69
		Hays	94.17	Texas	Nueces	31.39
		Hemphill	7.85	Texas	Ochiltree	15.69
		Henderson	47.08	Texas	Oldham	7.85
		Hidalgo	62.78	Texas	Orange	47.08
Texas		Hill	31.39	Texas	Palo Pinto	31.39
Texas		Hockley	15.69	Texas	Panola	31.39
		Hood	62.78	Texas	Parker	62.78
_		Hopkins	47.08	Texas	Parmer	15.69
		Houston	31.39	Texas	Pecos	7.85
_		Howard	15.69	Texas	Polk	47.08
		Hudspeth	7.85	Texas	Potter	15.69
		Hunt	47.08	Texas	Presidio	15.69
		Hutchinson	7.85 7.85	Texas Texas	Rains	47.08 15.69
		Jack	31.39	Texas	Reagan	7.85
		Jackson	31.39	Texas	Real	15.69
		Jasper	47.08	Texas	Red River	31.39
		Jeff Davis	7.85	Texas	Reeves	7.85
		Jefferson	31.39	Texas	Refugio	15.69
Texas		Jim Hogg	15.69	Texas	Roberts	7.85
Texas		Jim Wells	15.69	Texas	Robertson	31.39
Texas		Johnson	62.78	Texas	Rockwall	94.17
		Jones	15.69	Texas	Runnels	15.69
		Karnes	31.39	Texas	Rusk	47.08
		Kaufman	47.08	Texas	Sabine	62.78
		Kendall	62.78	Texas	San Augustine	47.08
		Kenedy	15.69	Texas	San Jacinto	62.78
_		Kent	7.85 31.39	Texas	San Patricio	31.39 31.39
		Kerr	31.39	Texas Texas	San Saba Schleicher	15.69
		King	7.85	Texas	Scurry	15.69
_		Kinney	15.69	Texas	Shackelford	15.69
_		Kleberg	15.69	Texas	Shelby	47.08
		Knox	7.85	Texas	Sherman	15.69
_		La Salle	15.69	Texas	Smith	47.08
		Lamar	31.39	Texas	Somervell	47.08
Texas		Lamb	15.69	Texas	Starr	31.39
Texas		Lampasas	31.39	Texas	Stephens	15.69
Texas		Lavaca	47.08	Texas	Sterling	7.85
Texas		Lee	47.08	Texas	Stonewall	7.85
		Leon	31.39	Texas	Sutton	15.69
		Liberty	47.08	Texas	Swisher	15.69
		Limestone	31.39	Texas	Tarrant	94.17
		Lipscomb	15.69	Texas	Taylor	31.39
		Live Oak	31.39 47.08	Texas	Terrell	7.85
		Llano	7.85	Texas Texas	Terry	15.69 15.69
		Lubbock	31.39	Texas	Throckmorton	47.08
		Lynn	15.69	Texas	Tom Green	31.39
		Madison	31.39	Texas	Travis	47.08
		Marion	31.39	Texas	Trinity	31.39
		Martin	15.69	Texas	Tyler	62.78
		Mason	31.39	Texas	Upshur	47.08
		Matagorda	31.39	Texas	Upton	7.85
		Maverick	7.85	Texas	Uvalde	31.39
		McCulloch	31.39	Texas	Val Verde	7.85
Texas		McLennan	31.39	Texas	Van Zandt	47.08
		McMullen	31.39	Texas	Victoria	31.39
		Medina	31.39	Texas	Walker	62.78
		Menard	15.69	Texas	Waller	94.17
		Midland	15.69	<u>T</u> exas	Ward	7.85
		Milam	31.39 31.39	Texas Texas	Washington	62.78
						15.69

State	County	(Fee/acre/ YR)	State	County	(Fee/acre/ YR)
exas	Wharton	31.39	Virginia	Caroline	62.7
exas	Wheeler	15.69	Virginia	Carroll	94.1
exas	Wichita	31.39	Virginia	Charles City	94.1
exas	Wilbarger	15.69	Virginia	Charlotte	47.0
exas	Willacy	31.39	Virginia	Chesapeake City	94.1
exas	Williamson	62.78	Virginia	Chesterfield	156.9
exas	Wilson	47.08	Virginia	Clarke	156.9
exas	Winkler Wise	7.85 62.78	Virginia Virginia	Craig	62.7 156.9
exas	Wood	47.08	Virginia	Cumberland	62.7
exas	Yoakum	15.69	Virginia	Dickenson	47.0
exas	Young	15.69	Virginia	Dinwiddie	47.0
exas	Zapata	31.39	Virginia	Essex	62.7
exas	Zavala	31.39	Virginia	Fairfax	313.8
Itah	Beaver	62.78	Virginia	Fauquier	156.9
Itah	Box Elder	15.69	Virginia	Floyd	62.7
tah	Cache	62.78	Virginia	Fluvanna	62.7
tah	Carbon	15.69	Virginia	Franklin	62.7
tah	Daggett	31.39	Virginia	Frederick	94.1
tah	Davis	156.94	Virginia	Giles	62.7
tah	Duchesne	15.69	Virginia	Gloucester	94.1
tah	Emery	31.39	Virginia	Goochland	94.1
tah	Garfield	47.08	Virginia	Grayson	94.1
tahtah	Iron	31.39 31.39	Virginia	Greene	156.9 47.0
tah	Juab	15.69	Virginia Virginia	Halifax	47.0
tah	Kane	15.69	Virginia	Hanover	156.9
tah	Millard	31.39	Virginia	Henrico	156.9
tah	Morgan	31.39	Virginia	Henry	47.0
tah	Piute	47.08	Virginia	Highland	62.7
tah	Rich	15.69	Virginia	Isle of Wight	62.
tah	Salt Lake	156.94	Virginia	James City	156.9
tah	San Juan	7.85	Virginia	King and Queen	62.
tah	Sanpete	31.39	Virginia	King George	94.1
tah	Sevier	47.08	Virginia	King William	62.7
tah	Summit	31.39	Virginia	Lancaster	62.7
tah	Tooele	15.69	Virginia	Lee	47.0
tah	Uintah	7.85	Virginia	Loudoun	313.8
tah	Utah	94.17	Virginia	Louisa	62.7
tah	Wasatch	94.17	Virginia	Lunenburg	47.0
tah	Washington	47.08	Virginia	Madison	94.
tah	Wayne	47.08	Virginia	Mathews	94.
tah	Weber	156.94	Virginia	Mecklenburg	47.0
ermont	Addison	47.08	Virginia	Middlesex	94.
ermont	Bennington	47.08	Virginia	Montgomery	94.
ermont	Caledonia	62.78	Virginia	Nelson	62.7
ermont	Chittenden	62.78 47.08	Virginia	New Kent	94. 62.
ermont	Franklin	47.08	Virginia Virginia	Northampton	62.7
ermont	Grand Isle	94.17	Virginia	Nottoway	62.7
ermont	Lamoille	62.78	Virginia	Orange	94.
ermont	Orange	47.08	Virginia	Page	156.9
ermont	Orleans	47.08	Virginia	Patrick	47.0
ermont	Rutland	94.17	Virginia	Pittsylvania	47.0
ermont	Washington	62.78	Virginia	Powhatan	94.
ermont	Windham	62.78	Virginia	Prince Edward	47.0
ermont	Windsor	94.17	Virginia	Prince George	62.7
rginia	Accomack	62.78	Virginia	Prince William	313.
rginia	Albemarle	156.94	Virginia	Pulaski	62.
rginia	Alleghany	62.78	Virginia	Rappahannock	94.
rginia	Amelia	62.78	Virginia	Richmond	47.
rginia	Amherst	62.78	Virginia	Roanoke	94.
rginia	Appomattox	47.08	Virginia	Rockbridge	94.
rginia	Arlington	94.17	Virginia	Rockingham	156.
rginia	Augusta	94.17	Virginia	Russell	47.
rginia	Bath	62.78	Virginia	Scott	47.
rginia	Bedford	94.17	Virginia	Shenandoah	94.
rginia	Bland	47.08	Virginia	Smyth	47.
rginia	Botetourt	94.17	Virginia	Southampton	62.
rginia	Brunswick	47.08	Virginia	Spotsylvania	156.
rginia	Buchanan	94.17	Virginia	Stafford	156.
irginia	Buckingham	62.78	Virginia	Suffolk	62.

State	County	(Fee/acre/ YR)	State	County	(Fee/acre/ YR)
Virginia	Sussex	47.08	West Virginia	Mason	47.08
Virginia	Tazewell	47.08	West Virginia	McDowell	31.39
Virginia	Virginia Beach City	94.17	West Virginia	Mercer	47.08
Virginia	Warren	156.94	West Virginia	Mineral	47.08
Virginia	Washington	62.78	West Virginia	Mingo	31.39
Virginia	Westmoreland	62.78	West Virginia	Monongalia	47.08
Virginia	Wise	62.78 62.78	West Virginia West Virginia	Monroe	47.08 62.78
Virginia Virginia	Wythe	1,569.43	West Virginia	Morgan	47.08
Washington	Adams	31.39	West Virginia	Ohio	31.39
Washington	Asotin	15.69	West Virginia	Pendleton	31.39
Washington	Benton	47.08	West Virginia	Pleasants	31.39
Washington	Chelan	313.89	West Virginia	Pocahontas	31.39
Washington	Clallam	313.89	West Virginia	Preston	47.08
Washington	Clark	313.89	West Virginia	Putnam	47.08
Washington	Columbia	31.39	West Virginia	Raleigh	47.08
Washington	Cowlitz	156.94	West Virginia	Randolph	31.39
Washington	Douglas	31.39	West Virginia	Ritchie	31.39
Washington	Ferry	15.69	West Virginia	Roane	31.39
Washington	Franklin Garfield	47.08 15.69	West Virginia West Virginia	Summers	31.39 47.08
Washington Washington	Grant	62.78	West Virginia	Taylor	31.39
Washington	Grays Harbor	62.78	West Virginia	Tyler	31.39
Washington	Island	313.89	West Virginia	Upshur	31.39
Washington	Jefferson	156.94	West Virginia	Wayne	31.39
Washington	King	627.77	West Virginia	Webster	31.39
Washington	Kitsap	627.77	West Virginia	Wetzel	31.39
Washington	Kittitas	94.17	West Virginia	Wirt	31.39
Washington	Klickitat	31.39	West Virginia	Wood	47.08
Washington	Lewis	94.17	West Virginia	Wyoming	31.39
Washington	Lincoln	15.69	Wisconsin	Adams	62.78
Washington	Mason	156.94	Wisconsin	Ashland	31.39
Washington	Okanogan	31.39	Wisconsin	Barron	47.08
Washington	Pacific	62.78	Wisconsin	Bayfield	31.39
Washington	Pend Oreille	47.08	Wisconsin	Brown	94.17
Washington	Pierce	313.89 313.89	Wisconsin	Buffalo	47.08 47.08
Washington Washington	San Juan	156.94	Wisconsin Wisconsin	Burnett	94.17
Washington	Skamania	156.94	Wisconsin	Chippewa	47.08
Washington	Snohomish	313.89	Wisconsin	Clark	47.08
Washington	Spokane	62.78	Wisconsin	Columbia	94.17
Washington	Stevens	31.39	Wisconsin	Crawford	47.08
Washington	Thurston	313.89	Wisconsin	Dane	94.17
Washington	Wahkiakum	94.17	Wisconsin	Dodge	62.78
Washington	Walla Walla	47.08	Wisconsin	Door	62.78
Washington	Whatcom	156.94	Wisconsin	Douglas	31.39
Washington	Whitman	31.39	Wisconsin	Dunn	47.08
Washington	Yakima	47.08	Wisconsin	Eau Claire	47.08
West Virginia	Barbour	31.39	Wisconsin	Florence	47.08
West Virginia	Berkeley	94.17	Wisconsin	Fond du Lac	62.78
West Virginia	Boone	31.39	Wisconsin	Forest	47.08
West Virginia	Braxton	31.39 31.39	Wisconsin Wisconsin	Grant	62.78 62.78
West Virginia West Virginia	Cabell	47.08	Wisconsin	Green Lake	62.78
West Virginia	Calhoun	31.39	Wisconsin	lowa	62.78
West Virginia	Clay	31.39	Wisconsin	Iron	31.39
West Virginia	Doddridge	31.39	Wisconsin	Jackson	47.08
West Virginia	Fayette	47.08	Wisconsin	Jefferson	94.17
West Virginia	Gilmer	31.39	Wisconsin	Juneau	47.08
West Virginia	Grant	47.08	Wisconsin	Kenosha	156.94
West Virginia	Greenbrier	47.08	Wisconsin	Kewaunee	94.17
West Virginia	Hampshire	47.08	Wisconsin	La Crosse	62.78
West Virginia	Hancock	62.78	Wisconsin	Lafayette	62.78
West Virginia	Hardy	47.08	Wisconsin	Langlade	47.08
West Virginia	Harrison	31.39	Wisconsin	Lincoln	47.08
West Virginia	Jackson	47.08	Wisconsin	Manitowoc	94.17
West Virginia	Jefferson Kanawha	94.17	Wisconsin	Marathon Marinette	47.08
West Virginia West Virginia		47.08 31.39	Wisconsin		47.08 62.78
West Virginia West Virginia	Lincoln	31.39	Wisconsin	Marquette Menominee	62.78 31.39
West Virginia	Logan	62.78	Wisconsin	Milwaukee	313.89
West Virginia	Marion	62.78 47.08	Wisconsin	Monroe	62.78

State	County	(Fee/acre/ YR)
Wisconsin	Oneida	62.78
Wisconsin	Outagamie	94.17
Wisconsin	Ozaukee	156.94
Wisconsin	Pepin	47.08
Wisconsin	Pierce	62.78
Wisconsin	Polk	62.78
Wisconsin	Portage	94.17
Wisconsin	Price	47.08
Wisconsin	Racine	156.94
Wisconsin	Richland	62.78
Wisconsin	Rock	94.17
Wisconsin	Rusk	62.78
Wisconsin	Sauk	94.17
Wisconsin	Sawyer	62.78
Wisconsin	Shawano	94.17
Wisconsin	Sheboygan	94.17
Wisconsin	St. Croix	94.17
Wisconsin	Taylor	47.08
Wisconsin	Trempealeau	47.08
Wisconsin	Vernon	47.08
Wisconsin	Vilas	94.17
Wisconsin	Walworth	156.94
Wisconsin	Washburn	47.08
Wisconsin	Washington	156.94
Wisconsin	Waukesha	156.94
Wisconsin	Waupaca	62.78
Wisconsin	Waushara	94.17
Wisconsin	Winnebago	94.17
Wisconsin	Wood	47.08
Wyoming	Albany	7.85
Wyoming	Big Horn	31.39
Wyoming	Campbell	7.85
Wyoming	Carbon	7.85
Wyoming	Converse	7.85
Wyoming	Crook	15.69
Wyoming	Fremont	7.85
Wyoming	Goshen	15.69
Wyoming	Hot Springs	7.85
Wyoming	Johnson	7.85
Wyoming	Laramie	7.85
Wyoming	Lincoln	31.39
Wyoming	Natrona	7.85
Wyoming	Niobrara	7.85
Wyoming	Park	31.39
Wyoming	Platte	15.69
Wyoming	Sheridan	15.69
Wyoming	Sublette	31.39
Wyoming	Sweetwater	7.85
Wyoming	Teton	94.17
Wyoming	Uinta	15.69
Wyoming	Washakie	15.69
,y	Weston	7.85

^{*}State-average Land and Building value used when no county-specific is available.
**Land areas to be determined.

[75 FR 44094, July 28, 2010; 75 FR 48553, Aug. 11, 2010]

PART 12—SAFETY OF **WATER** POWER PROJECTS AND PROJECT WORKS

Subpart A—General Provisions

Sec.

- 12.1 Applicability.
- 12.2Rules of construction.
- 12.3 Definitions.

- 12.4 Staff administrative responsibility and supervisory authority.
- 12.5 Responsibilities of licensee or applicant.

Subpart B—Reports and Records

- 12.10 Reporting safety-related incidents.
- 12.11 Reporting modifications of the project or project works.
- 12.12 Maintenance of records.
- 12.13 Verification form.

Subpart C—Emergency Action Plans

- General requirements.
- Exemptions.
- Contents of emergency action plan.
- 12.23 Time for filing emergency action plan.
- 12.24 Review and updating of plans.
- 12.25 Posting and readiness.

Subpart D—Inspection by Independent Consultant

- 12.30 Applicability.
- 12.31 Definitions.
- 12.32 General inspection requirement.
- 12.33 Exemption.
- 12.34 Approval of independent consultant.
- 12.35 Specific inspection requirements. 12.36 Emergency corrective measures.
- 12.37 Report of the independent consultant.
- 12.38 Time for inspections and reports.
- 12.39 Taking corrective measures after the

Subpart E—Other Responsibilities of **Applicant or Licensee**

- 12.40 Quality control programs.
- 12.41 Monitoring instruments.
- 12.42 Warning and safety devices.
- 12.43 Power and communication lines and gas pipelines.
- 12.44 Testing spillway gates.

AUTHORITY: 16 U.S.C. 792-828c; 42 U.S.C. 7101–7352; E.O. 12009, 3 CFR 142 (1978).

Source: Order 122, 46 FR 9036, Jan. 28, 1981, unless otherwise noted.

EDITORIAL NOTE: Nomenclature changes to part 12 appear at 69 FR 32438, June 10, 2004.

Subpart A—General Provisions

§ 12.1 Applicability.

- (a) Except as otherwise provided in this part or ordered by the Commission or its authorized representative, the provisions of this part apply to:
- (1) Any project licensed under Part I of the Federal Power Act;
- Any unlicensed constructed project for which the Commission has

§ 12.2

determined that an application for license must be filed under Part I of the Act: and

- (3) Any project exempted from licensing under Part I of the Federal Power Act, pursuant to subparts J or K of part 4 of this chapter, to the extent that the Commission has conditioned the exemption on compliance with any particular provisions of this part.
- (b) The provisions of this part apply to a project that uses a Government dam only with respect to those project works, lands, and waters specifically licensed by the Commission.

§12.2 Rules of construction.

- (a) If any term, condition, article, or other provision in a project license is similar to any provision of this part, the licensee must comply with the relevant provision of this part, unless the Commission or the Director of the Office of Energy Projects Licensing determines that compliance with the relevant provision of the license will better protect life, health, or property.
- (b) A licensee may request from the Director of the Office of Energy Projects Licensing a ruling on the applicability to its actions of any provision of its license that is similar to a provision of this part. A ruling by the Director may be appealed under § 385.207 of this chapter.

[Order 122, 46 FR 9036, Jan. 28, 1981, as amended by Order 225, 47 FR 19056, May 3, 1982; 49 FR 29370, July 20, 1984]

§ 12.3 Definitions.

- (a) General rule. For purposes of this part, terms defined in section 3 of the Federal Power Act, 16 U.S.C. 796, have the same meaning as they have under the Act.
- (b) *Definitions*. The following definitions apply for the purposes of this part:
- (1) Applicant means any person, state, or municipality that has applied for a license for an unlicensed, constructed project and any owner of an unlicensed, constructed project for which the Commission has determined that an application for license must be filed.
- (2) Owner means any person, state, or municipality, or combination thereof, that has a real property interests in a

water power project sufficient to operate and maintain the project works.

- (3) Authorized Commission representative means the Director of the Office of Energy Projects Licensing, the Director of the Division of Inspections, the Regional Engineer, or any other member of the Commission staff whom the Commission may specifically designate.
- (4) Condition affecting the safety of a project or project works means any condition, event, or action at the project which might compromise the safety, stability, or integrity of any project work or the ability of any project work to function safely for its intended purposes, including navigation, water power development, or other beneficial public uses; or which might otherwise adversely affect life, health, or property. Conditions affecting the safety of a project or project works include, but are not limited to:
- (i) Unscheduled rapid draw-down of impounded water;
- (ii) Failure of any facility that controls the release or storage of impounded water, such as a gate or a valve:
- (iii) Failure or unusual movement, subsidence, or settlement of any part of a project work;
- (iv) Unusual concrete deterioration or cracking, including development of new cracks or the lengthening or widening of existing cracks;
- (v) Piping, slides, or settlements of materials in any dam, abutment, dike, or embankment:
- (vi) Significant slides or settlements of materials in areas adjacent to reservoirs;
- (vii) Significant damage to slope protection;
- (viii) Unusual instrumentation readings;
- (ix) New seepage or leakage or significant gradual increase in pre-existing seepage or leakage;
 - (x) Sinkholes;
- (xi) Significant instances of vandalism or sabotage;
- (xii) Natural disasters, such as floods, earthquakes, or volcanic activity;
- (xiii) Any other signs of instability of any project work.
- (5) Constructed project means any project with an existing dam.

- (6) Dam means any structure for impounding or diverting water.
- (7) Development means that part of a project comprising an impoundment and its associated dams, forebays, water conveyance facilities, power plants, and other appurtenant facilities. A project may comprise one or more developments.
- (8) Modification means any activity, including repair or reconstruction, that in any way changes the physical features of the project from the state reflected in the plans or drawings or other documents filed with the Commission.
- (9) Project emergency means an impending or actual sudden release of water at the project caused by natural disaster, accident, or failure of project works.
- (10) Regional Engineer means the person in charge of the Commission's regional office for the region (Atlanta, Chicago, Portland, New York, or San Francisco) where a particular project is located.
- (11) Act means the Federal Power Act.

[Order 122, 46 FR 9036, Jan. 28, 1981, as amended at 49 FR 29370, July 20, 1984; Order 647, 69 FR 32438, June 10, 2004]

§ 12.4 Staff administrative responsibility and supervisory authority.

- (a) Administrative responsibility. The Director of the Office of Energy Projects Licensing is responsible for administering the Commission's project safety program and reports directly to the Chairman of the Federal Energy Regulatory Commission.
- (b) Supervisory authority of the Regional Engineer or other authorized representative. (1) Any water power project and the construction, operation, maintenance, use, repair, or modification of any project works are subject to the inspection and the supervision of the Regional Engineer or any other authorized Commission representative for the purpose of:
- (i) Achieving or protecting the safety, stability, and integrity of the project works or the ability of any project work to function safely for its intended purposes, including navigation, water power development, or other beneficial public uses; or

- (ii) Otherwise protecting life, health, or property.
- (2) For the purposes set forth in paragraph (b)(1) of this section, a Regional Engineer or other authorized Commission representative may:
- (i) Test or inspect any water power project or project works or require that the applicant or licensee perform such tests or inspections or install monitoring instruments;
- (ii) Require an applicant or a licensee to submit reports or information, regarding:
- (A) The design, construction, operation, maintenance, use, repair, or modification of a water power project or project works; and
- (B) Any condition affecting the safety of a project or project works or any death or injury that occurs at, or might be attributable to, the water power project;
- (iii) Require an applicant or a licensee to modify:
- (A) Any emergency action plan filed under subpart C of this part: or
- (B) Any plan of corrective measures, including related schedules, submitted after the report of an independent consultant pursuant to §12.37 or any other inspection report;
- (iv) Require an applicant or licensee to take any other action with respect to the design, construction, operation, maintenance, repair, use, or modification of the project or its works that is, in the judgment of the Regional Engineer or other authorized Commission representative, necessary or desirable.
- (v) Establish the time for an applicant or licensee to provide a schedule for or to perform any actions specified in this paragraph.
- (c) Appeal, stay, rescission, or amendment of order or directive. (1) Any order or directive issued under this section or under the provisions of subparts B through E of this part by a Regional Engineer or other authorized Commission representative may be appealed to the Commission under §385.207 of this chapter.
- (2) Any order or directive issued under this section by a Regional Engineer or other authorized Commission representative is immediately effective and remains in effect until:

§ 12.5

- (i) The Regional Engineer or other authorized Commission representative who issued the order or directive rescinds or amends that order or directive or stays its effect; or
- (ii) The Commission stays the effect of the order or directive, or amends or rescinds the order or directive on appeal.
- (3) An appeal or motion for rescission, amendment, or stay of any order or directive issued under this section must contain a full explanation of why granting the appeal or the request for rescission or amendment of the order or directive, or for stay for the period requested, will not endanger life, health, or property.

[Order 122, 46 FR 9036, Jan. 28, 1981, as amended by Order 225, 47 FR 19056, May 3, 1982; 49 FR 29370, July 20, 1984]

§ 12.5 Responsibilities of licensee or applicant.

A licensee or applicant must use sound and prudent engineering practices in any action relating to the design, construction, operation, maintenance, use, repair, or modification of a water power project or project works.

Subpart B—Reports and Records

§ 12.10 Reporting safety-related incidents.

- (a) Conditions affecting the safety of a project or its works—1) Oral reports. An applicant or licensee must report by telephone to the Regional Engineer any condition affecting the safety of a project or projects works, as defined in §12.3(b)(4). The initial oral report must be made as soon as practicable after that condition is discovered, without unduly interfering with any necessary or appropriate emergency repair, alarm, or other emergency action procedure.
- (2) Written reports. Following the initial oral report required in paragraph (a)(1), the applicant or licensee must submit to the Regional Engineer a written report on the condition affecting the safety of the project or project works verified in accordance with §12.13. The written report must be submitted within the time specified by the Regional Engineer and must contain

any information the Regional Engineer directs, including:

- (i) The causes of the condition;
- (ii) A description of any unusual occurrences or operating circumstances preceding the condition;
- (iii) An account of any measure taken to prevent worsening of the condition;
- (iv) A detailed description of any damage to project works and the status of any repair:
- (v) A detailed description of any personal injuries;
- (vi) A detailed description of the nature and extent of any private property damages: and
- (vii) Any other relevant information requested by the Regional Engineer.
- (3) The level of detail required in any written report must be commensurate with the severity and complexity of the condition.
- (b) Deaths or serious injuries. (1) Promptly after becoming aware of any drowning or other accident resulting in death or serious injury that occurs at the project, the applicant or licensee must report that drowning or other accident to the Regional Engineer in writing, including a description of the cause and location of the accident.
- (2) The written report of any death or serious injury considered or alleged to be project related must also describe any remedial actions taken or proposed to avoid or reduce the chance of similar occurrences in the future and be verified in accordance with §12.13.
- (3) Accidents that are not project-related may be reported by providing a copy of a clipping from a newspaper article, if available.
- (4) For the purposes of this paragraph, *project-related* includes any deaths or serious injuries involving a dam, spillway, intake, or power line, or which take place at or immediately above or below a dam.

§ 12.11 Reporting modifications of the project or project works.

(a) Reporting requirement. Regardless of whether a particular modification is permitted without specific prior Commission approval, an applicant or licensee must report any modification of the project or project works to the Regional Engineer in writing, verified in

accordance with §12.13, at the time specified in paragraph (b) of this section.

- (b) *Time of reporting*. (1) Any modification that is an emergency measure taken in response to a condition affecting the safety of the project or project works must be submitted with the report of that condition required by §12.10(a)(2).
- (2) In all other instances, the modification must be reported at least 60 days before work on the modification begins.

§12.12 Maintenance of records.

- (a) Kinds of records—1) General rule. Except as provided in paragraph (a)(2) of this section, the applicant or licensee must maintain as permanent project records in addition to those required in part 125 of this chapter, the following information:
- (i) Engineering and geological data relating to design, construction, maintenance, repair, or modification of the project, including design memoranda and drawings, laboratory and other testing reports, geologic data (such as maps, sections, or logs of exploratory borings or trenches, foundation treatment, and excavation), plans and specifications, inspection and quality control reports, as built construction drawings, designers' operating criteria, photographs, and any other data necessary to demonstrate that construction, maintenance, repair, or modification of the project has been performed in accordance with plans and specifications;
- (ii) Instrumentation observations and data collected during construction, operation, or maintenance of the project, including continuously maintained tabular records and graphs illustrating the data collected pursuant to §12.41; and
- (iii) The operational and maintenance history of the project, including:
- (A) The dates, times, nature, and causes of any complete or partial unscheduled shut-down, suspension of project operations, or reservoir filling restrictions related to the safety of the project or project works; and
- (B) Any reports of project modifications, conditions affecting the safety of the project or project works, or deaths or serious injuries at the project.

- (2) Exception. The applicant or licensee is not required to maintain as permanent project records any information specified in paragraph (a)(1) of this section that was or reasonably would have been prepared before the applicant or licensee acquired control of the project and that the applicant or the licensee never acquired or reasonably could have acquired.
- (b) Location of records—1) Original records. The applicant or licensee must maintain the originals of all permanent project records at a central location, such as the project site or the main business office of the applicant or licensee, secure from damage from any conceivable failure of the project works and convenient for inspection. The applicant or licensee must keep the Regional Engineer advised of the location of the permanent project records.
- (2) Record copies. If the originals of the permanent project records are maintained at a central location other than the project site, the applicant or licensee must maintain at the project site copies of at least the project Exhibit G or L (design drawings), instrumentation data, and operational history that are necessary to the safe and efficient operation of the project.
- (3) In accordance with the provisions of part 125 of this chapter, the applicant or licensee may maintain original records, or record copies at the project site, in microform, if appropriate equipment is readily available to view the records.
- (c) Transfer of records. If the project is taken over by the United States at the end of a license term or the Commission issues a new license to a different licensee, the prior licensee must transfer the originals of all permanent project records to the custody of the administering Federal agency or department or to the new licensee.

§12.13 Verification form.

If a document submitted in accordance with the provisions of this part must be verified, the form of verification attached to the document must be the following:

State of [], County of [], ss:

§ 12.20

The undersigned, being first duly sworn, states that [he, she] has read the above document and knows the contents of it, and that all of the statements contained in that document are true and correct, to the best of [his, her] knowledge and belief.

[Name of person signing]

Sworn to and subscribed before me this [day] of [month], [year].
[Seal]

[Signature of notary public or other state or local official authorized by law to notarize documents.]

Subpart C—Emergency Action Plans

§12.20 General requirements.

- (a) Unless provided with a written exemption pursuant to §12.21, every applicant or licensee must develop and file with the Regional Engineer three copies of an emergency action plan and appendices, verified in accordance with §12.13.
- (b) The emergency action plan must be:
- (1) Developed in consultation and cooperation with appropriate Federal, state, and local agencies responsible for public health and safety; and
- (2) Designed to provide early warning to upstream and downstream inhabitants, property owners, operators of water-related facilities, recreational users, and other persons in the vicinity who might be affected by a project emergency as defined in §12.3(b)(9).

§12.21 Exemptions.

- (a) Grant of exemption. Except as provided in paragraph (b), if an applicant or licensee satisfactorily demonstrates that no reasonably foreseeable project emergency would endanger life, health, or property, the Regional Engineer may exempt the applicant or licensee from filing an emergency action plan.
- (b) No exemption. A licensee or applicant may not be exempted from the requirements of §12.22(c) for a radiological response plan.
- (c) Conditions of exemptions. (1) An applicant or licensee who receives an exemption from filing an emergency action plan has the continuing responsi-

bility to review circumstances upstream and downstream from the project to determine if, as a result of changed circumstances, a project emergency might endanger life, health, or property.

- (2) Promptly after the applicant or licensee learns that, as a result of any change in circumstances, a project emergency might endanger life, health, or property, the applicant or licensee must inform the Regional Engineer of that changed condition without unduly delaying the preparation and implementation of the emergency action plan.
- (3) Comprehensive review of the necessity for an emergency action plan must be conducted at least once each year
- (d) Revocation of exemption. (1) The Regional Engineer may revoke an exemption granted under this section if it is determined that, as a result of any change in circumstances, a project emergency might endanger life, health, or property.
- (2) If an exemption is revoked, the applicant or licensee must file an emergency action plan within the time specified by the Regional Engineer.

§ 12.22 Contents of emergency action plan.

- (a) Contents—(1) The plan itself. An emergency action plan must conform with the guidelines established, and from time to time revised, by the Director of the Office of Energy Projects Licensing (available from the division of Inspections or the Regional Engineer) to provide:
- (i) Instructions to project operators and attendants and other responsible personnel about the actions they are to take during a project emergency;
- (ii) Detailed plans for notifying potentially affected persons, appropriate Federal, state, and local agencies, including public safety and law enforcement bodies, and medical units; and
- (iii) Procedures for controlling the flow of water, including actions to reduce in-flows to reservoirs, such as limiting outflows from upstream dams or control structures, and actions to reduce downstream flows, such as increasing or decreasing outflows from

downstream dams or control structures, on the waterway on which the project is located or its tributaries.

- (2) Appendix to the plan. Each copy of the emergency action plan submitted to the Regional Engineer must be accompanied by an appendix conforming with the guidelines established by the Director of the Office of Energy Projects Licensing that contains:
- (i) Plans for training project operators, attendants, and other responsible personnel to respond properly during a project emergency, including instructions on the procedures to be followed throughout a project emergency and the manner in which the licensee will periodically review the knowledge and understanding that these personnel have of those procedures:
- (ii) A summary of the study used for determining the upstream and downstream areas that may be affected by sudden release of water, including a summary of all criteria and assumptions used in the study and, if required by the Regional Engineer, inundation maps; and
- (iii) Documentation of consultations with Federal, state, and local agencies, including public safety and law enforcement bodies, and medical units.
- (b) Special factors. The applicant or licensee must take into account in its emergency action plan the time of day, particularly hours of darkness, in establishing the proper actions and procedures for use during a project emergency.
- (c) Additional requirements for projects near nuclear power plants-1) Radiological response plan. If the personnel operating any powerhouse or any spillway control facilities, such as gates or valves, of a project would be located within ten miles of a nuclear power plant reactor, the applicant or licensee must file, separately or as a supplement to any required emergency action plan, a radiological response plan that provides for emergency procedures to be taken if an accident or other incident results in the release of radioactive materials from the nuclear power plant reactor.
- (2) A radiological response plan must: (i) To the maximum extent practicable, include sufficient procedural safeguards to ensure that, during or

- following an accident or other incident involving the nearby nuclear power plant reactor, the project may be safely operated and, if evacuation is necessary, the project may be left unattended without danger to the safety of any project dam or to life, health, or safety upstream or downstream from the project; and
- (ii) Explain the provisions, developed after consultation with the direct purchasers of project power, for cessation, curtailment, or continuation of generation of electric power at the project during or following an accident or other incident involving the nearby nuclear power plant reactor.
- (3) Time of filing radiological response plan. (i) For a constructed project with an otherwise acceptable emergency action plan on file, any radiological response plan required must be filed:
- (A) If an operating license for the nuclear power plant has been issued on or before March 1, 1981, not later than three months from March 1, 1981; or
- (B) In all other instances, not later than three months after the date an operating license for the nuclear power plant is issued.
- (ii) For any project not described in §12.22(c)(3)(i), any radiological response plan required must be filed contemporaneously with the emergency action plan or, if the project has been exempted from filing an emergency action plan, at the time the emergency action plan would otherwise have been required to be filed pursuant to §12.23.

[Order 122, 46 FR 9036, Jan. 28, 1981, as amended at 49 FR 29370, July 20, 1984]

§ 12.23 Time for filing emergency action plan.

- (a) Unconstructed project. (1) Except as set forth in paragraph (a)(2), the emergency action plan for an unconstructed project must be filed no later than 60 days before the initial filling of the project reservoir begins.
- (2) Temporary impoundment during construction. (i) For any unconstructed project, if a temporary impoundment would be created during construction, such as through construction of temporary or permanent cofferdams or large sediment control structures, and an accident to or failure of the impounding structures might endanger

§ 12.24

construction workers or otherwise endanger public health or safety, a temporary construction emergency action plan must be filed no later than 60 days before construction begins.

- (ii) No later than 60 days before the initial filling of a project reservoir begins at a project for which a temporary emergency action plan has been filed the applicant or licensee must file modifications to that plan or a new plan, taking into account the differences in circumstances between the construction and post-construction periods.
- (b) Unlicensed constructed project. (1) If the Commission has determined on or before March 1, 1981 that a license is required for an unlicensed constructed project, the emergency action plan for that project must be filed no later than:
 - (i) Six months after March 1, 1981; or (ii) Any earlier date specified by the
- (11) Any earner date specified by the Commission or its authorized representative.
- (2) Except as set forth in paragraph (b)(1) of this section, the emergency action plan for an unlicensed constructed project must be filed no later than the earliest of:
- (i) Six months after the date that a license application is filed;
- (ii) Six months after the date that the Commission issues an order determining that licensing is required; or
- (iii) A date specified by the Commission or its authorized representative.
- (c) Licensed constructed project. If a licensed constructed project does not have an acceptable emergency action plan on file on March 1, 1981 the emergency action plan must be filed no later than:
 - (1) Six months after March 1, 1981; or
- (2) Any earlier date specified by the Commission or its authorized representative.
- (d) For good cause shown, the Regional Engineer may grant an extension of time for filing all or any part of an emergency action plan.

§12.24 Review and updating of plans.

(a) The emergency action plan must be continually updated to reflect any changes in the names or titles of project operators and attendants and other personnel with specified respon-

- sibilities for actions in an emergency and any changes in names of persons to call, telephone numbers, radio call signals, or other information critical to providing notification to affected persons, Federal, state, and local agencies, and medical units.
- (b) An applicant or licensee has continuing responsibility to review the adequacy of the emergency action plan in light of any significant changes in upstream or downstream circumstances which might affect water flows or the location or extent of the areas, persons, or property that might be harmed in a project emergency.
- (c) Promptly after an applicant or licensee learns of any change in circumstances described in paragraph (b) of this section, the applicant or licensee must:
- (1) Inform the Regional Engineer of that change in circumstances;
- (2) Consult and cooperate with appropriate Federal, state, and local agencies responsible for public health and safety to determine any advisable revisions to the emergency action plan; and
- (3) File with the Regional Engineer three copies of any revisions to the appropriate studies, maps, plans, procedures, or other information in the emergency action plan itself or its appendices that have changed as a result of that consultation.
- (d) An applicant or licensee must conduct a comprehensive review of the adequacy of the emergency action plan at least once each year.

§12.25 Posting and readiness.

- (a) A copy of the current emergency action plan itself must be posted in a prominent location readily accessible to the licensee's or applicant's operating personnel who are responsible for controlling water flows and for notifying public health and safety agencies and affected persons.
- (b) Each licensee or applicant must annually test the state of training and readiness of key licensee or applicant personnel responsible for responding properly during a project emergency to ensure that they know and understand the procedures to be followed throughout a project emergency.

Subpart D—Inspection by Independent Consultant

§ 12.30 Applicability.

This subpart applies to any licensed project development that has a dam:

- (a) That is more than 32.8 feet (10 meters) in height above streambed, as defined in §12.31(c);
- (b) That impounds an impoundment with a gross storage capacity of more than 2,000 acre-feet (2.5 million cubic meters); or
- (c) That has a high hazard potential and is determined by the Regional Engineer or other authorized Commission representative to require inspection by an independent consultant under this subpart.

§12.31 Definitions.

For purposes of this subpart:

- (a) Independent consultant means any person who:
- (1) Is a licensed professional engineer:
- (2) Has at least 10 years experience and expertise in dam design and construction and in the investigation of the safety of existing dams; and
- (3) Is not, and has not been within two years before being retained to perform an inspection under this subpart, an employee of the licensee or its affiliates or an agent acting on behalf of the licensee or its affiliates.
- (b) Dam that has a high hazard potential means any dam whose failure, in the judgment of the Commission or its authorized representative, might endanger human life or cause significant property damage, or which meets the criteria for high hazard potential as defined by the Corps of Engineers in 33 CFR part 222.
 - (c) Height above streambed means:
- (1) For a dam with a spillway, the vertical distance from the lowest elevation of the natural streambed at the downstream toe of the dam to the maximum water storage elevation possible without any discharge from the spillway. The maximum water storage elevation is:
- (i) For gated spillways, the elevation of the tops of the gates;
- (ii) For ungated spillways, the elevation of the spillway crest or the top of any flashboards, whichever is higher;

- (2) For a dam without a spillway, the vertical distance from the lowest elevation of the natural streambed at the downstream tow of the dam to the lowest point on the crest of the dam.
- (d) Gross storage capacity means the maximum possible volume of water impounded by a dam with zero spill, that is, without the discharge of water over the dam or a spillway.
- (e) The Director of the Office of Energy Projects Licensing may, for good cause shown, grant a waiver of the 10 year requirement in paragraph (a)(2) of this section. Any petition for waiver under this paragraph must be filed in accordance with §1.7(b) of this chapter.

[Order 122, 46 FR 9036, Jan. 28, 1981, as amended at 49 FR 29370, July 20, 1984]

§ 12.32 General inspection requirement.

In accordance with the procedures in §12.35, the project works of each development to which this subpart applies, excluding transmission and transformation facilities and generating equipment, must be periodically inspected and evaluated by or under the responsibility and direction of at least one independent consultant, who may be a member of a consulting firm, to identify any actual or potential deficiencies, whether in the condition of those project works or in the quality or adequacy of project maintenance, surveillance, or methods of operation, that might endanger public safety.

§ 12.33 Exemption.

- (a) Upon written request from the licensee, the Director of the Office of Energy Projects Licensing may grant an exemption from the requirements of this subpart in extraordinary circumstances that clearly establish good cause for exemption.
- (b) Good cause for exemption may include the finding that the development in question has no dam except dams that meet the criteria for low hazard potential as defined by the Corps of Engineers in 33 CFR part 222.

 $[{\rm Order~122,~46~FR~9036,~Jan.~28,~1981,~as~amend-ed~at~49~FR~29370,~July~20,~1984}]$

§ 12.34

§ 12.34 Approval of independent consultant.

At least 60 days before the initiation of an inspection under this subpart, the licensee must submit to the Director of the Office of Energy Projects Licensing for approval, with a copy to the Regional Engineer, a detailed resume that (a) describes the experience of the independent consultant; and, (b) shows that the consultant is an independent consultant as defined in §12.31(a).

[Order 122, 46 FR 9036, Jan. 28, 1981, as amended at 49 FR 29370, July 20, 1984]

§ 12.35 Specific inspection requirements.

- (a) Scope of inspection. The inspection by the independent consultant shall include:
- (1) Due consideration of all relevant reports on the safety of the development made by or written under the direction of Federal or state agencies, submitted under Commission regulations, or made by other consultants;
- (2) Physical field inspection of the project works and review and assessment of all relevant data concerning:
 - (i) Settlement:
 - (ii) Movement;
 - (iii) Erosion;
 - (iv) Seepage;
 - (v) Leakage;
 - (vi) Cracking;
 - (vii) Deterioration;
 - (viii) Seismicity;
- (ix) Internal stress and hydrostatic pressures in project structures or their foundations or abutments;
- (x) The functioning of foundation drains and relief wells;
- (xi) The stability of critical slopes adjacent to a reservoir or project works; and
- (xii) Regional and site geological conditions; and
 - (3) Specific evaluation of:
 - (i) The adequacy of spillways;
- (ii) The effects of overtopping of nonoverflow structures:
- (iii) The structural adequacy and stability of structures under all credible loading conditions;
- (iv) The relevant hydrological data accumulated since the project was constructed or last inspected under this subpart;

- (v) The history of the performance of the project works through analysis of data from monitoring instruments; and
- (vi) The quality and adequacy of maintenance, surveillance, and methods of project operations for the protection of public safety.
- (b) Evaluation of spillway adequacy. The adequacy of any spillway must be evaluated by considering hazard potential which would result from failure of the project works during flood flows.
- (1) If structural failure would present a hazard to human life or cause significant property damage, the independent consultant must evaluate the ability of project works to withstand the loading or overtopping which may occur from a flood up to the probable maximum flood or the capacity of spillways to prevent the reservoir from rising to an elevation that would endanger the project works.
- (2) If structural failure would not present a hazard to human life of cause significant property damage, spillway adequacy may be evaluated by means of a design flood of lesser magnitude than the probable maximum flood, if the report of the independent consultant pursuant to \$12.37 provides a detailed explanation of the bases for the finding that structural failure would not present a hazard to human life or cause significant property damage.

§ 12.36 Emergency corrective measures.

If, in the course of an inspection, an independent consultant discovers any condition for which emergency corrective measures are advisable, the independent consultant must immediately notify the licensee and the licensee must report that condition to the Regional Engineer pursuant to §12.10(a) of this part.

§ 12.37 Report of the independent consultant.

(a) General requirement. Following inspection of a project development as required under this subpart, the independent consultant must prepare a report and the licensee must file three copies of that report with the Regional Engineer. The report must conform to the provisions of this section and be

satisfactory to the authorized Commission representative.

- (b) General information in the initial report. (1) The initial report filed under this subpart for any project development must contain:
- (i) A description of the project development;
- (ii) A map of the region indicating the location of the project development:
- (iii) Plans, elevations, and sections of the principal project works;
- (iv) A summary of the design assumptions, design analyses, spillway design flood, and the factors of safety used to evaluate the structural adequacy and stability of the project works; and
- (v) A summary of the geological conditions that may affect the safety of the project works.
- (2) To the extent that the information and analyses required in paragraph (b)(1) of this section, are contained in a report of an independent consultant prepared and filed in compliance with Commission regulations in effect before March 1, 1981 the information and analyses may be incorporated by specific reference into the first report prepared and filed under this subpart.
- (c) Information required for all reports. Any report of an independent consultant filed under this subpart must contain the information specified in this paragraph.
- (1) Monitoring information. The report must contain monitoring information that includes time-versus-reading graphs depicting data compiled from any existing critical or representative monitoring instruments that measure the behavior, movement, deflection, or loading of project works or from which the stability, performance, or functioning of the structures may be determined
- (i) Any monitoring data plotted on graphs must be presented in a manner that will facilitate identification and analysis of trends. The data may be summarized to facilitate graphical representation.
- (ii) Plan and sectional drawings of project structures sufficient to show the location of all critical or representative existing monitoring instruments must be included. If these drawings

have been included in a previous report prepared and filed by an independent consultant, they may be incorporated by specific reference to that earlier report.

- (2) Analyses. The report must:
- (i) Analyze the safety of the project works and the maintenance and methods of operation of the development fully in light of the independent consultant's reviews, field inspections, assessments, and evaluations described in §12.35;
- (ii) Identify any changes in the information and analyses required by paragraph (b) of this section that have occurred since the last report by an independent consultant under this subpart and analyze the implications of those changes; and
- (iii) Analyze the adequacy of existing monitoring instruments, periodic observation programs, and other methods of monitoring project works and conditions effecting the safety of the project or project works with respect to the development.
- (3) Incorporation by reference. To the extent that conditions, assumptions, and available information have not changed since the last previous report by an independent consultant under this subpart, the analyses required under paragraphs (c)(2)(i) and (ii) of this section may be incorporated by specific reference to the last previous report.
- (4) Recommendations. Based on the independent consultant's field observations and evaluations of the project works and the maintenance, surveillance, and methods of operation of the development, the report must contain the independent consultant's recommendations on:
- (i) Any corrective measures necessary for the structures or for the maintance or surveillance procedures or methods of operation of the project works:
- (ii) A reasonable time to carry out each corrective measure; and
- (iii) Any new or additional monitoring instruments, periodic observations, or other methods of monitoring project works or conditions that may be required.

§ 12.38

- (5) Dissenting views. If the inspection and report were conducted and prepared by more than one independent consultant, the report must clearly indicate any dissenting views concerning the analyses or recommendations of the report that might be held by any individual consultant.
- (6) List of participants. The report must identify all professional personnel who have participated in the inspection of the project or in preparation of the report and the independent consultant who directed those activities.
- (7) Statement of independence. The independent consultant must declare that all conclusions and recommendations in the report are made independently of the licensee, its employees, and its representatives.
- (8) Signature. The report must be signed by each independent consultant responsible for the report.

§ 12.38 Time for inspections and reports.

- (a) General rule. After the initial inspection and report under this subpart for a project development, a new inspection under this subpart must be completed and the report on it filed not later than five years from the date the last report on an inspection was to be filed under this subpart.
- (b) Initial inspection and report. (1) For any development that has a dam that is more than 32.8 feet (10 meters) in height above streambed or impounds an impoundment with a gross storage capacity of more than 2,000 acre feet (2.5 million cubic meters), which development was constructed before the date of issuance of the order licensing or amending a license to include that development, the initial inspection under this subpart must be completed and the report on it filed not later than two years after the date of issuance of the order licensing the development or amending the license to include the development.
- (2) For any development that was constructed after the date of issuance of the order licensing or amending a license to include the development, the initial inspection under this subpart must be completed and the report on it filed not later than five years from the

- date of first commercial operation, or the date on which the impoundment first reaches its normal maximum surface elevation, whichever occurs first.
- (3) For any development not set forth in either subparagraph (b)(1) or (b)(2), the initial inspection under this subpart must be completed and the report on it filed by a date specified by the Regional Engineer. The filing date must not be more than two years after the date of notification that an inspection and report under this subpart are required.
- (4) The last independent consultant's inspection and report made for a development before March 1, 1981 in compliance with the Commission's rules then in effect is deemed to fulfill the requirements for an initial inspection and report under this subpart for that development, except that the first report filed under this subpart for that development after March 1, 1981 must contain the information and analyses required by §12.37(b).
- (c) Extension of time. For good cause shown, the Regional Engineer may extend the time for filing an independent consultant's report under this subpart.

§ 12.39 Taking corrective measures after the report.

- (a) Corrective plan and schedule. (1) Not later than 60 days after the report of the independent consultant is filed with the Regional Engineer, the licensee must submit to the Regional Engineer three copies of a plan and schedule for designing and carrying out any corrective measures that the licensee proposes.
- (2) The plan and schedule may include any proposal, including taking no action, that the licensee considers a preferable alternative to any corrective measure recommended in the report of the independent consultant. Any proposed alternative must be accompanied by the licensee's complete justification and detailed analysis and evaluation in support of that alternative.
- (b) Carrying out the plan. The licensee must complete all corrective measures in accordance with the plan and schedule submitted to, and approved or modified by, the Regional Engineer.

(c) Extension of time. For good cause shown, the Regional Engineer may extend the time for filing the plan and schedule required by this section.

Subpart E—Other Responsibilities of Applicant or Licensee

§ 12.40 Quality control programs.

- (a) General rule. During any construction, repair, or modification of project works, including any corrective measures taken pursuant to §12.39 of this part, the applicant or licensee must maintain any quality control program that may be required by the Regional Engineer, commensurate with the scope of the work and meeting any requirements or standards set by the Regional Engineer. If a quality control program is required, the construction, repair, or modification may not begin until the Regional Engineer has approved the program.
- (b) If the construction, repair, or modification work is performed by a construction contractor, quality control inspection must be performed by the licensee, the design engineer, or an independent firm, other than the construction contractor, directly accountable to the licensee. This paragraph is not intended to prohibit additional quality control inspections by the construction contractor, or a firm accountable to the construction contractor, for the construction contractor's purposes.
- (c) If the construction, repair, or modification of project works is performed by the applicant's or licensee's own personnel, the applicant or licensee must provide for separation of authority within its organization to make certain that the personnel responsible for quality control inspection are, to the satisfaction of the Regional Engineer or other authorized Commission representative, independent from the personnel who are responsible for the construction, repair or modification.

§12.41 Monitoring instruments.

(a) In designing a project, a licensee must make adequate provision for installing and maintaining appropriate monitoring instrumentation whenever any physical condition that might affect the stability of a project structure has been discovered or is anticipated. The instrumentation must be satisfactory to the Regional Engineer and may include, for example, instruments to monitor movement of joints, foundation or embankment deformation, seismic effects, hydrostatic pore pressures, structural cracking, or internal stresses on the structure.

(b) If an applicant or licensee discovers any condition affecting the safety of the project or project works during the course of construction or operation, the applicant or licensee must install and maintain any monitoring devices and instruments that may be required by the Regional Engineer or other authorized Commission representative to monitor that condition.

§ 12.42 Warning and safety devices.

To the satisfaction of, and within a time specified by, the Regional Engineer, an applicant or licensee must install, operate, and maintain any signs, lights, sirens, barriers, or other safety devices that may reasonably be necessary or desirable to warn the public of fluctuations in flow from the project or otherwise to protect the public in the use of project lands and waters.

§ 12.43 Power and communication lines and gas pipelines.

- (a) A licensee must take all reasonable precautions, and comply with all reasonable specifications that may be provided by the Regional Engineer, to ensure that any power or communication line or gas pipeline that is located over, under, or in project waters does not obstruct navigation for recreational or commercial purposes or otherwise endanger public safety.
- (b) Clearances between any power or communication line constructed after March 1, 1981 and any vessels using project waters must be at least sufficient to conform to any applicable requirements of the National Electrical Safety Code in effect at the time the power or communication line is constructed.
- (c) The Regional Engineer may require a licensee or applicant to provide signs at or near power or communication lines to advise the public of the

§ 12.44

clearances for any power or communication lines located over, under, or in project waters.

§12.44 Testing spillway gates.

- (a) General requirement. An applicant or licensee must make adequate provision, to the satisfaction of the Regional Engineer or other authorized Commission representative, to ensure that all spillway gates are operable at all times, particularly during adverse weather conditions.
- (b) Annual test. (1) At least once each year, each spillway gate at a project must be operated to spill water, either during regular project operation or on a test basis.
- (2) If an applicant or licensee does not operate each spillway gate on a test basis during the periodic inspection by the Commission staff, the applicant or licensee must submit to the Regional Engineer at least once each year a written statement, verified in accordance with §12.13, that each spillway has been operated at least once during the twelve months preceding the inspection.
- (c) Load-test of standby power. (1) An applicant or licensee must load-test the standby emergency power for spill-way gate operation at regular intervals, but not less than once during each year, and submit to the Regional Engineer, at least once each year, a written statement, verified in accordance with \$12.13, describing the intervals at which the standby emergency power was load-tested during the year preceding the inspection.
- (2) The Commission staff may direct that a spillway gate be operated using standby emergency power during the periodic inspection.

PART 16—PROCEDURES RELATING TO TAKEOVER AND RELICENSING OF LICENSED PROJECTS

Subpart A—General Provisions

Sec.

16.1 Applicability.

16.2 Definitions.

- 16.3 Public notice of projects under expiring licenses.
- 16.4 Acceleration of a license expiration date.

16.5 Site access for a competing applicant.

Subpart B—Applications for Projects Subject to Sections 14 and 15 of the Federal Power Act

- 16.6 Notification procedures under section 15 of the Federal Power Act.
- 16.7 Information to be made available to the public at the time of notification of intent under section 15(b) of the Federal Power Act.
- 16.8 Consultation requirements.
- 16.9 Applications for new licenses and nonpower licenses for projects subject to sections 14 and 15 of the Federal Power Act.
- 16.10 Information to be provided by an applicant for new license: Filing requirements.
- 16.11 Nonpower licenses.
- 16.12 Application for exemption from licensing by a licensee whose license is subject to sections 14 and 15 of the Federal Power Act
- 16.13 Standards and factors for issuing a new license.

Subpart C—Takeover Provisions for Projects Subject to Sections 14 and 15 of the Federal Power Act

- 16.14 Departmental recommendation for takeover.
- 16.15 Commission recommendation to Congress.
- 16.16 Motion for stay by Federal department or agency.
- 16.17 Procedures upon Congressional authorization of takeover

Subpart D—Annual Licenses for Projects Subject to Sections 14 and 15 of the Federal Power Act

16.18 Annual licenses for projects subject to sections 14 and 15 of the Federal Power Act.

Subpart E—Projects With Minor and Minor Part Licenses Not Subject to Sections 14 and 15 of the Federal Power Act

- 16.19 Procedures for an existing licensee of a minor hydroelectric power project or of a minor part of a hydroelectric power project with a license not subject to sections 14 and 15 of the Federal Power Act.
- 16.20 Applications for subsequent license for a project with an expiring license not subject to sections 14 and 15 of the Federal Power Act.
- 16.21 Operation of projects with a minor or minor part license not subject to sections 14 and 15 of the Federal Power Act after expiration of a license.

16.22 Application for an exemption by a licensee with a minor or minor part license for a project not subject to sections 14 and 15 of the Federal Power Act.

Subpart F—Procedural Matters

- 16.23 Failure to file timely notices of intent.
- 16.24 Prohibitions against filing applications for new license, nonpower license, exemption, or subsequent license.
- 16.25 Disposition of a project for which no timely application is filed following a notice of intent to file.
- 16.26 Disposition of a project for which no timely application is filed following a notice of intent not to file.

AUTHORITY: 16 U.S.C. 791a-825r; 42 U.S.C. 7101-7352.

Source: Order 513, 54 FR 23806, June 2, 1989, unless otherwise noted

Subpart A—General Provisions

§ 16.1 Applicability.

This part applies to the filing and processing of an application for:

- (a) A new license, a nonpower license, or an exemption from licensing for a hydroelectric project with an existing license subject to the provisions of sections 14 and 15 of the Federal Power Act.
- (b) A subsequent license or an exemption from licensing for a hydroelectric project with an existing minor license or minor part license not subject to the provisions of sections 14 and 15 of the Federal Power Act because those sections were waived pursuant to section 10(i) of the Federal Power Act.
- (c) Any potential applicant for a new or subsequent license for which the deadline for the notice of intent required by §16.6 falls on or after July 23, 2005 and which wishes to develop and file its application pursuant to this part, must seek Commission authorization to do so pursuant to the provisions of part 5 of this chapter.

[Order 513, 54 FR 23806, June 2, 1989, as amended by Order 2002, 68 FR 51139, Aug. 25, 2003]

§ 16.2 Definitions.

For purposes of this part:

(a) New license means a license, except an annual license, for a water power project that is issued under sec-

tion 15(a) of the Federal Power Act after an original license expires.

- (b) New license application filing deadline, as provided in section 15(c)(1) of the Federal Power Act, is the date 24 months before the expiration of an existing license.
- (c) Nonpower license means a license for a nonpower project issued under section 15(b) of the Federal Power Act.
- (d) Subsequent license means a license for a water power project issued under Part I of the Federal Power Act after a minor or minor part license that is not subject to sections 14 and 15 of the Federal Power Act expires.

[Order 513, 54 FR 23806, June 2, 1989, as amended by Order 513-A, 55 FR 15, Jan. 2, 1990; Order 533, 56 FR 23154, May 20, 1991]

§ 16.3 Public notice of projects under expiring licenses.

In addition to the notice of a licensee's intent to file or not to file an application for a new license provided in §16.6(d), the Commission will publish, in its annual report and annually in the FEDERAL REGISTER, a table showing the projects whose licenses will expire during the succeeding six years. The table will:

- (a) List the licenses according to their expiration dates; and
- (b) Contain the following information: license expiration date; licensee's name; project number; type of principal project works licensed, e.g., dam and reservoir, powerhouse, transmission lines; location by state, county, and stream; location by city or nearby city when appropriate; whether the existing license is subject to sections 14 and 15 of the Federal Power Act; and plant installed capacity.

§ 16.4 Acceleration of a license expiration date.

- (a) Request for acceleration. (1) A licensee may file with the Commission, in accordance with the formal filing requirements in subpart T of part 385 of this chapter, a written request for acceleration of the expiration date of its existing license, containing the statements and information specified in §16.6(b) and a detailed explanation of the basis for the acceleration request.
- (2) If the Commission grants the request for acceleration pursuant to

paragraph (c), the Commission will deem the request for acceleration to be a notice of intent under §16.6 and, unless the Commission directs otherwise, the licensee shall make available the information specified in §16.7 no later than 90 days from the date that the Commission grants the request for acceleration.

- (b) Notice of request for acceleration. (1) Upon receipt of a request for acceleration, the Commission will give notice of the licensee's request and provide a 45-day period for comments by interested persons by:
- (i) Publishing notice in the FEDERAL REGISTER:
- (ii) Publishing notice once in a daily or weekly newspaper published in the county or counties in which the project or any part thereof or the lands affected thereby are situated; and
- (iii) Notifying appropriate Federal, state, and interstate resource agencies and Indian tribes by mail.
- (2) The notice issued pursuant to paragraphs (1) (i) and (ii) and the written notice given pursuant to paragraph (1)(iii) will be considered as fulfilling the notice provisions of §16.6(d) should the Commission grant the acceleration request and will include an explanation of the basis for the licensee's acceleration request.
- (c) Commission order. If the Commission determines it is in the public interest, the Commission will issue an order accelerating the expiration date of the license to not less than five years and 90 days from the date of the Commission order.

§ 16.5 Site access for a competing applicant.

(a) Access. If a potential applicant for a new license, subsequent license, or nonpower license for a project has complied with the first stage consultation provisions of §16.8(b)(1) and has notified the existing licensee in writing of the need for and extent of the access required, the existing licensee must allow the potential applicant to enter upon or into designated land, buildings, or other property in the project area at a reasonable time and under reasonable conditions, including, but not limited to, reasonable liability conditions, conditions for compensation to the exist-

ing licensee for all reasonable costs incurred in providing access, including energy generation lost as a result of modification of project operations that may be necessary to provide access, and in a manner that will not adversely affect the environment, for the purposes of:

- (1) Conducting a study or gathering information required by a resource agency under §16.8 or by the Commission pursuant to §4.32 of this chapter;
- (2) Conducting a study or gathering information not covered by paragraph (a)(1) but necessary to prepare an application for new license, subsequent license, or nonpower license; or
- (3) Holding a site visit for a resource agency under §16.8.
- (b)(1) Disputes. Except as specified by paragraph (b)(2), disputes regarding the timing and conditions of access for the purposes specified in paragraphs (a) (1), (2), or (3) of this section and the need for the studies or information specified in paragraph (a)(2) may be referred to the Director of the Office of Energy Projects for resolution in the manner specified in §16.8(b)(5) prior to the providing of access.
- (2) Disputes regarding the amount of compensation to be paid the existing licensee for access may be referred to the Director of the Office of Energy Projects for resolution in the manner specified in §16.8(b)(5) after the access has been provided.

Subpart B—Applications for Projects Subject to Sections 14 and 15 of the Federal Power Act

§ 16.6 Notification procedures under section 15 of the Federal Power Act.

- (a) Applicability. This section applies to a licensee of an existing project subject to sections 14 and 15 of the Federal Power Act.
- (b) Requirement to notify. In order to notify the Commission under section 15 of the Federal Power Act whether a licensee intends to file or not to file an application for new license, the licensee must file with the Secretary of the Commission in accordance with filing procedures posted on the Commission's Web site at http://www.ferc.gov a

Federal Energy Regulatory Commission

letter, that contains the following information:

- (1) The licensee's name and address.
- (2) The project number.
- (3) The license expiration date.
- (4) An unequivocal statement of the licensee's intention to file or not to file an application for a new license.
- (5) The type of principal project works licensed, such as dam and reservoir, powerhouse, or transmission lines.
- (6) Whether the application is for a power or nonpower license.
- (7) The location of the project by state, county and stream, and, when appropriate, by city or nearby city.
 - (8) The installed plant capacity.
- (9) The location or locations of all the sites where the information required under §16.7 is available to the public.
- (10) The names and mailing addresses of:
- (i) Every county in which any part of the project is located, and in which any Federal facility that is used by the project is located;
- (ii) Every city, town, or similar local political subdivision:
- (A) In which any part of the project is located and any Federal facility that is used by the project is located, or
- (B) That has a population of 5,000 or more people and is located within 15 miles of the project dam,
- (iii) Every irrigation district, drainage district, or similar special purpose political subdivision:
- (A) In which any part of the project is located and any Federal facility that is used by the project is located, or
- (B) That owns, operates, maintains, or uses any project facility or any Federal facility that is used by the project;
- (iv) Every other political subdivision in the general area of the project that there is reason to believe would be likely to be interested in, or affected by, the notification;
 - (v) Affected Indian tribes.
- (c) When to notify. (1) Except as provided in paragraph (c)(2) of this section, if a license expires on or after October 17, 1992, the licensee must notify the Commission as required in paragraph (b) of this section at least five years, but no more than five and one-

half years, before the existing license expires.

- (2) The requirement in paragraph (c)(1) of this section does not apply if a licensee filed notice more than five and one-half years before its existing license expired and before the effective date of this rule.
- (d) Commission notice. Upon receipt of the notification required under paragraph (c) of this Section, the Commission will provide notice of the licensee's intent to file or not to file an application for a new license by:
- (1) If the notification is filed prior to July 23, 2005;
- (i) Publishing notice in the FEDERAL REGISTER:
- (ii) Publishing notice once in a daily or weekly newspaper published in the county or counties in which the project or any part thereof or the lands affected thereby are situated; and
- (iii) Notifying the appropriate Federal and state resource agencies, state water quality and coastal zone management consistency certifying agencies, and Indian tribes, by electronic means if practical, otherwise by mail.
- (2) If the notification is filed on or after July 23, 2005, pursuant to the provisions of §5.8 of this chapter.

[Order 496, 53 FR 15810, May 4, 1988. Redesignated and amended by Order 513, 54 FR 23807, June 2, 1989; Order 2002, 68 FR 51139, Aug. 25, 2003; Order 653, 70 FR 8724, Feb. 23, 2005; Order 737, 75 FR 43403, July 26, 2010]

§ 16.7 Information to be made available to the public at the time of notification of intent under section 15(b) of the Federal Power Act.

- (a) Applicability. This section applies to a licensee of an existing project subject to sections 14 and 15 of the Federal Power Act.
- (b) Requirement to make information available. A licensee must make the information specified in paragraph (d) of this section reasonably available to the public for inspection and reproduction, from the date on which the licensee notifies the Commission pursuant to §16.6(b) of this part until the date any relicensing proceeding for the project is terminated.
- (c) Requirement to supplement information. A licensee must supplement the information it is required to make

available under the provisions of paragraph (d) with any additional information developed after the filing of a notice of intent.

- (d) Information to be made available. (1) A licensee for which the deadline for filing a notification of intent to seek a new or subsequent license is on or after July 23, 2005 must, at the time it files a notification of intent to seek a license pursuant to §5.5 of this chapter, provide a copy of the pre-application document required by §5.6 of this chapter to the entities specified in that paragraph.
- (2) A licensee for which the deadline for filing a notification of intent to seek a new or subsequent license is prior to July 23, 2005, and which elects to seek a license pursuant to this part must make the following information regarding its existing project reasonably available to the public as provided in paragraph (b) of this section:
- (i) The following construction and operation information:
- (A) The original license application and the order issuing the license and any subsequent license application and subsequent order issuing a license for the existing project, including
- (1) Approved Exhibit drawings, including as-built exhibits,
- (2) Any order issuing amendments or approving exhibits,
- (3) Any order issuing annual licenses for the existing project;
- (B) All data relevant to whether the project is and has been operated in accordance with the requirements of each license article, including minimum flow requirements, ramping rates, reservoir elevation limitations, and environmental monitoring data;
- (C) A compilation of project generation and respective outflow with time increments not to exceed one hour, unless use of another time increment can be justified, for the period beginning five years before the filing of a notice of intent;
- (D) Any public correspondence related to the existing project;
- (E) Any report on the total actual annual generation and annual operation and maintenance costs for the period beginning five years before the filing of a notice of intent;

- (F) Any reports on original project costs, current net investment, and available funds in the amortization reserve account:
- (G) A current and complete electrical single-line diagram of the project showing the transfer of electricity from the project to the area utility system or point of use; and
- (H) Any bill issued to the existing licensee for annual charges under Section 10(e) of the Federal Power Act.
- (ii) The following safety and structural adequacy information:
- (A) The most recent emergency action plan for the project or a letter exempting the project from the emergency action plan requirement;
- (B) Any independent consultant's reports required by part 12 of this chapter and filed on or after January 1, 1981;
- (C) Any report on operation or maintenance problems, other than routine maintenance, occurring within the five years preceding the filing of a notice of intent or within the most recent five-year period for which data exists, and associated costs of such problems under the Commission's Uniform System of Accounts;
- (D) Any construction report for the existing project; and
- (E) Any public correspondence relating to the safety and structural adequacy of the existing project.
- (iii) The following fish and wildlife resources information:
- (A) Any report on the impact of the project's construction and operation on fish and wildlife resources;
- (B) Any existing report on any threatened or endangered species or critical habitat located in the project area, or affected by the existing project outside the project area;
- (C) Any fish and wildlife management plan related to the project area prepared by the existing licensee or any resource agency; and
- (D) Any public correspondence relating to the fish and wildlife resources within the project area.
- (iv) The following recreation and land use resources information:
- (A) Any report on past and current recreational uses of the project area;
- (B) Any map showing recreational facilities and areas reserved for future

development in the project area, designated or proposed wilderness areas in the project area; Land and Water Conservation Fund lands in the project area, and designated or proposed Federal or state wild and scenic river corridors in the project area.

- (C) Any documentation listing the entity responsible for operating and maintaining any existing recreational facilities in the project area; and
- (D) Any public correspondence relating to recreation and land use resources within the project area.
- (v) The following cultural resources information:
- (A) Except as provided in paragraph (d)(2)(v)(B) of this section, a licensee must make available:
- (1) Any report concerning documented archeological resources identified in the project area;
- (2) Any report on past or present use of the project area and surrounding areas by Native Americans; and
- (3) Any public correspondence relating to cultural resources within the project area.
- (B) A licensee must delete from any information made available under paragraph (d)(2)(v)(A) of this section, specific site or property locations the disclosure of which would create a risk of harm, theft, or destruction of archeological or Native American cultural resources or to the site at which the resources are located, or would violate any Federal law, including the Archeological Resources Protection Act of 1979, 16 U.S.C. 470w-3, and the National Historic Preservation Act of 1966, 16 U.S.C. 470hh.
- (vi) The following energy conservation information under section 10(a)(2)(C) of the Federal Power Act related to the licensee's efforts to conserve electricity or to encourage conservation by its customers including:
 - (A) Any plan of the licensee;
 - (B) Any public correspondence; and
- (C) Any other pertinent information relating to a conservation plan.
 - (3)-(6) [Reserved]
- (7)(i) If paragraph (d) of this section requires an applicant to reveal Critical Energy Infrastructure Information (CEII), as defined in §388.113(c) of this chapter, to any person, the applicant shall omit the CEII from the informa-

tion made available and insert the following in its place:

- (A) A statement that CEII is being withheld:
- (B) A brief description of the omitted information that does not reveal any CEII; and
- (C) This statement: "Procedures for obtaining access to Critical Energy Infrastructure Information (CEII) may be found at 18 CFR 388.113. Requests for access to CEII should be made to the Commission's CEII Coordinator."
- (ii) The applicant, in determining whether information constitutes CEII, shall treat the information in a manner consistent with any filings that applicant has made with the Commission and shall to the extent practicable adhere to any previous determinations by the Commission or the CEII Coordinator involving the same or like information.
- (iii) The procedures contained in §§388.112 and 388.113 of this chapter regarding designation of, and access to, CEII, shall apply in the event of a challenge to a CEII designation or a request for access to CEII. If it is determined that information is not CEII or that a requester should be granted access to CEII, the applicant will be directed to make the information available to the requester.
- (iv) Nothing in this section shall be construed to prohibit any persons from voluntarily reaching arrangements or agreements calling for the disclosure of CEII.
- (e) Form, place, and hours of availability, and cost of reproduction. (1) A licensee must make the information specified in paragraph (d) of this section, or the pre-application document, as applicable, available to the public for inspection:
- (i) At its principal place of business or at any other location or locations that are more accessible to the public, provided that all of the information is available in at least one location;
- (ii) During regular business hours;
- (iii) In a form that is readily accessible, reviewable, and reproducible.
- (2) Except as provided in paragraph (d)(3) of this section, a licensee must

make requested copies of the information specified in paragraph (c) of this section available either:

- (i) At its principal place of business or at any other location or locations that are more accessible to the public, after obtaining reimbursement for reasonable costs of reproduction; or
- (ii) Through the mail, after obtaining reimbursement for postage fees and reasonable costs of reproduction.
- (3) A licensee must make requested copies of the information specified in paragraph (d) of this section available to the United States Fish and Wildlife Service, the National Marine Fisheries Service, Indian tribes, and the state agency responsible for fish and wildlife resources without charge for the costs of reproduction or postage.
- (f) Unavailability of required information. Anyone may file a petition with the Commission requesting access to the information specified in paragraph (d) of this section if it believes that a licensee is not making the information reasonably available for public inspection or reproduction. The petition must describe in detail the basis for the petitioner's belief.
- (g) Public correspondence. A licensee may compile and make available in one file all the public correspondence required to be made available for inspection and reproduction by \$16.7(d)(1)(iv), (d)(2)(v), (d)(3)(iv), (d)(4)(iv), and (d)(6)(ii).

[Order 496, 53 FR 15810, May 4, 1988. Redesignated by Order 513, 54 FR 23807, June 2, 1989; Order 513-C, 55 FR 10768, Mar. 23, 1990; Order 2002, 68 FR 51139, Aug. 25, 2003; Order 643, 68 FR 52095. Sept. 2, 2003

§ 16.8 Consultation requirements.

(a) Requirement to consult. (1) Before it files any application for a new license, a nonpower license, an exemption from licensing, or, pursuant to §16.25 or §16.26 of this part, a surrender of a project, a potential applicant must consult with the relevant Federal, State, and interstate resource agencies, including the National Marine Fisheries Service, the United States Fish and Wildlife Service, the National Park Service, the United States Environmental Protection Agency, the Federal agency administering any United States lands or facilities utilized or oc-

cupied by the project, the appropriate state fish and wildlife agencies, the appropriate State water resource management agencies, the certifying agency under section 401(a)(1) of the Federal Water Pollution Control Act (Clean Water Act), 33 U.S.C. 1341(c)(1), and any Indian tribe that may be affected by the project.

- (2) Each requirement in this section to contact or consult with resource agencies or Indian tribes shall require as well that the potential Applicant contact or consult with members of the public.
- (3) If the potential applicant for a new or subsequent license commences first stages pre-filing consultation under this part on or after July 23, 2005, it must file a notification of intent to \\$5.5 of this chapter and a pre-application document pursuant to the provisions of \\$5.6 of this chapter.
- (4) The Director of the Office of Energy Projects will, upon request, provide a list of known appropriate Federal, state, and interstate resource agencies, and Indian tribes, and local, regional, or national non-governmental organizations likely to be interested in any license application proceeding.
- (5)(i) Before it files an amendment that would be considered as material under §4.35 of this part, to any application subject to this section, an applicant must consult with the resource agencies and Indian tribes listed in paragraph (a)(1) of this section and allow such agencies and tribes at least 60 days to comment on a draft of the proposed amendment and to submit recommendations and conditions to the applicant. The amendment as filed with the Commission must summarize the consultation with the resource agencies and Indian tribes on the proposed amendment and respond to any obligations, recommendations or conditions submitted by the agencies or Indian tribes.
- (ii) If an applicant has any doubt as to whether a particular amendment would be subject to the pre-filing consultation requirements of this section, the applicant may file a written request for clarification with the Director, Office of Energy Projects.

- (b) First stage of consultation. (1) A potential Applicant for a new or subsequent license must, at the time it files its notification of intent to seek a license pursuant to §5.5 of this chapter, provide a copy of the pre-application document required by §5.6 of this chapter to the entities specified in §5.6(a) of this chapter.
- (2) A potential applicant for a nonpower license or exemption or a potential applicant which elects to use the licensing procedures of Parts 4 or 16 of this chapter prior to July 23, 2005, must promptly contact each of the appropriate resource agencies, Indian tribes, and members of the public listed in paragraph (a)(1) of this section, and the Commission with the following information:
- (i) Detailed maps showing existing project boundaries, if any, proper land descriptions of the entire project area by township, range, and section, as well as by state, county, river, river mile, and closest town, and also showing the specific location of all existing and proposed project facilities, including roads, transmission lines, and any other appurtenant facilities;
- (ii) A general engineering design of the existing project and any proposed changes, with a description of any existing or proposed diversion of a stream through a canal or penstock;
- (iii) A summary of the existing operational mode of the project and any proposed changes:
- (iv) Identification of the environment affected or to be affected, the significant resources present and the applicant's existing and proposed environmental protection, mitigation, and enhancement plans, to the extent known at that time:
- (v) Streamflow and water regime information, including drainage area, natural flow periodicity, monthly flow rates and durations, mean flow figures illustrating the mean daily streamflow curve for each month of the year at the point of diversion or impoundment, with location of the stream gauging station, the method used to generate the streamflow data provided, and copies of all records used to derive the flow data used in the applicant's engineering calculations;

- (vi) Detailed descriptions of any proposed studies and the proposed methodologies to be employed; and
- (vii) Any statement required by §4.301(a) of this chapter.
- (3)(i) A potential applicant for an exemption, a new or subsequent license for which the deadline for filing a notification of intent to seek a license is prior to July 23, 2005 and which elects to commence pre-filing consultation under this part, or a new or subsequent license for which the deadline for filing a notification of intent to seek a license is on or after July 23, 2005 and which receives Commission approval to use the license application procedures of this part must:
- (A) Hold a joint meeting, including an opportunity for a site visit, with all pertinent agencies, Indian tribes and members of the public to review the information and to discuss the data and studies to be provided by the potential applicant as part of the consultation process; and
- (B) Consult with the resource agencies, Indian tribes and members of the public on the scheduling of the joint meeting; and provide each resource agency, Indian tribe, member of the public, and the Commission with written notice of the time and place of the joint meeting and a written agenda of the issues to be discussed at the meeting at least 15 days in advance.
- (ii) The joint meeting must be held no earlier than 30 days, and no later than 60 days from, as applicable:
- (A) The date of the potential applicant's letter transmitting the information required by paragraph (b)(2) of this section, in the case of a potential exemption applicant or a potential license applicant that commences prefiling consultation under this part prior to July 23, 2005; or
- (B) The date of the Commission's approval of the potential license applicant's request to use the license application procedures of this part pursuant to the provisions of part 5, in the case of a potential license applicant for which the deadline for filing a notification of intent to seek a license is on or after July 23, 2005.
- (4) Members of the public are invited to attend the joint meeting held pursuant to paragraph (b)(3) of this section.

Members of the public attending the meeting are entitled to participate fully in the meeting and to express their views regarding resource issues that should be addressed in any application for a new license that may be filed by the potential applicant. Attendance of the public at any site visit held pursuant to paragraph (b)(3) of this section shall be at the discretion of the potential applicant. The potential applicant must make either audio recordings or written transcripts of the joint meeting, and must upon request promptly provide copies of these recordings or transcripts to the Commission and any resource agency and Indian tribe.

- (5) Unless otherwise extended by the Director of Office of Energy Projects pursuant to paragraph (b)(6) of this section, not later than 60 days after the joint meeting held under paragraph (b)(3) of this section each interested resource agency, and Indian tribe, and member of the public must provide a potential applicant with written comments:
- (i) Identifying its determination of necessary studies to be performed or information to be provided by the potential applicant:
- (ii) Identifying the basis for its determination:
- (iii) Discussing its understanding of the resource issues and its goals objectives for these resources:
- (iv) Explaining why each study methodology recommended by it is more appropriate than any other available methodology alternatives, including those identified by the potential applicant pursuant to paragraph (b)(2)(vi) of this section;
- (v) Documenting that the use of each study methodology recommended by it is a generally accepted practice; and
- (vi) Explaining how the studies and information requested will be useful to the agency, Indian tribe, or member of the public in furthering its resource goals and objectives.
- (6)(i) If a potential applicant and a resource agency, Indian tribe, or member of the public disagree as to any matter arising during the first stage of consultation or as to the need to conduct a study or gather information referenced in paragraph (c)(2) of this sec-

tion, the potential applicant or resource agency, or Indian tribe, or member of the public may refer the dispute in writing to the Director of the Office of Energy Projects (Director) for resolution.

- (ii) The entity referring the dispute must serve a copy of its written request for resolution on the disagreeing party at the time the request is submitted to the Director. The disagreeing party may submit to the Director a written response to the referral within 15 days of the referral's submittal to the Director.
- (iii) Written referrals to the Director and written responses thereto pursuant to paragraphs (b)(6)(i) or (b)(6)(ii) of this section must be filed with the Secretary of the Commission in accordance with the Commission's Rules of Practice and Procedure, and must indicate that they are for the attention of the Director of the Office of Energy Projects pursuant to §16.8(b)(6).
- (iv) The Director will resolve disputes by an order directing the potential applicant to gather such information or conduct such study or studies as, in the Director's view, is reasonable and necessary.
- (v) If a resource agency, Indian tribe, or member of the public fails to refer a dispute regarding a request for a potential applicant to obtain information or conduct studies (other than a dispute regarding the information specified in paragraph (b)(1) or (b)(2) of this section, as applicable), the Commission will not entertain the dispute following the filing of the license application.
- (vi) If a potential applicant fails to obtain information or conduct a study as required by the Director pursuant to paragraph (b)(6)(iv) of this section, its application will be considered deficient.
- (7) Unless otherwise extended by the Director pursuant to paragraph (b)(6) of this section, the first stage of consultation ends when all participating agencies, Indian tribes, and members of the public provide the written comments required under paragraph (b)(5) of this section or 60 days after the joint meeting held under paragraph (b)(3) of this section, whichever occurs first.

- (c) Second stage of consultation. (1) Unless determined otherwise by the Director of the Office of Energy Projects pursuant to paragraph (b)(6) of this section, a potential applicant must complete all reasonable and necessary studies and obtain all reasonable and necessary information requested by resource agencies and Indian tribes under paragraph (b):
- (i) Prior to filing the application, if the results:
- (A) Would influence the financial (e.g., instream flow study) or technical feasibility of the project (e.g., study of potential mass soil movement); or
- (B) Are needed to determine the design or location of project features, reasonable alternatives to the project, the impact of the project on important natural or cultural resources (e.g., resource surveys), suitable mitigation or enhancement measures, or to minimize impact on significant resources (e.g., wild and scenic river, anadromous fish, endangered species, caribou migration routes):
- (ii) After filing the application but before license issuance, if the applicant complied with the provisions of paragraph (b)(1) or (b)(2) of this section, as applicable, no later than four years prior to the expiration date of the existing license and the results:
- (A) Would be those described in paragraphs (c)(1)(i) (A) or (B) of this section; and
- (B) Would take longer to conduct and evaluate than the time between the conclusion of the first stage of consultation and the new license application filing deadline.
- (iii) After a new license is issued, if the studies can be conducted or the information obtained only after construction or operation of proposed facilities, would determine the success of protection, mitigation, or enhancement measures (e.g., post-construction monitoring studies), or would be used to refine project operation or modify project facilities.
- (2) If, after the end of the first stage of consultation as defined in paragraph (b)(7) of this section, a resource agency, Indian tribe, or member of the public requests that the potential applicant conduct a study or gather information not previously identified and specifies

- the basis for its request, under paragraphs (b)(5)(i)–(vi) of this section, the potential applicant will promptly initiate the study or gather the information, unless the Director of the Office of Energy Projects determines under paragraph (b)(5) of this section either that the study or information is unreasonable or unnecessary or that use of the methodology requested by a resource agency or Indian tribe for conducting the study is not a generally accepted practice.
- (3) (i) The results of studies and information gathering referenced in paragraphs (c)(1)(ii) and (c)(2) of this section will be treated as additional information; and
- (ii) Filing and acceptance of an application will not be delayed and an application will not be considered deficient or patently deficient pursuant to §4.32 (e)(1) or (e)(2) of this chapter merely because the study or information gathering is not complete before the application is filed.
- (4) A potential applicant must provide each resource agency and Indian tribe with:
- (i) A copy of its draft application that:
- (A) Indicates the type of application the potential applicant expects to file with the Commission; and
- (B) Responds to any comments and recommendations made by any resource agency or Indian tribe either during the first stage of consultation or under paragraph (c)(2) of this section;
- (ii) The results of all studies and information gathering either requested by that resource agency or Indian tribe in the first stage of consultation (or under paragraph (c)(2) of this section if available) or which pertains to resources of interest to that resource agency or Indian tribe and which were identified by the potential applicant pursuant to paragraph (b)(2)(vi) of this section, including a discussion of the results and any proposed protection, mitigation, or enhancement measure; and
- (iii) A written request for review and comment.
- (5) A resource agency or Indian tribe will have 90 days from the date of the

potential applicant's letter transmitting the paragraph (c)(4) of this section information to it to provide written comments on the information submitted by a potential applicant under paragraph (c)(4) of this section.

- (6) If the written comments provided under paragraph (c)(5) of this section indicate that a resource agency or Indian tribe has a substantive disagreement with a potential applicant's conclusions regarding resource impacts or its proposed protection, mitigation, or enhancement measures, the potential applicant will:
- (i) Hold at least one joint meeting with the disagreeing resource agency or Indian tribe and other agencies with similar or related areas of interest, expertise, or responsibility not later than 60 days from the date of the disagreeing agency's or Indian tribe's written comments to discuss and to attempt to reach agreement on its plan for environmental protection, mitigation, or enhancement measures; and
- (ii) Consult with the disagreeing agency or Indian tribe and other agencies with similar or related areas of interest, expertise, or responsibility on the scheduling of the joint meeting and provide the disagreeing resource agency or Indian tribe, other agencies with similar or related areas of interest, expertise, or responsibility, and the Commission with written notice of the time and place of each meeting and a written agenda of the issues to be discussed at the meeting at least 15 days in advance.
- (7) The potential applicant and any disagreeing resource agency or Indian tribe may conclude a joint meeting with a document embodying any agreement among them regarding environmental protection, mitigation, or enhancement measures and any issues that are unresolved.
- (8) The potential applicant must describe all disagreements with a resource agency or Indian tribe on technical or environmental protection, mitigation, or enhancement measures in its application, including an explanation of the basis for the applicant's disagreement with the resource agency or Indian tribe, and must include in its application any document developed

pursuant to paragraph (c)(7) of this section.

- (9) A potential applicant may file an application with the Commission if:
- (i) It has complied with paragraph (c)(4) of this section and no resource agency or Indian tribe has responded with substantive disagreements by the deadline specified in paragraph (c)(5) of this section; or
- (ii) It has complied with paragraph (c)(6) of this section if any resource agency or Indian tribe has responded with substantive disagreements.
- (10) The second stage of consultation ends:
- (i) Ninety days after the submittal of information pursuant to paragraph (c)(4) of this section in cases where no resource agency or Indian tribe has responded with substantive disagreements; or
- (ii) At the conclusion of the last joint meeting held pursuant to paragraph (c)(6) of this section in cases where a resource agency or Indian tribe has responded with substantive disagreements.
- (d) Third stage of consultation. (1) The third stage of consultation is initiated by the filing of an application for a new license, nonpower license, exemption from licensing, or surrender of license, accompanied by a transmittal letter certifying that at the same time copies of the application are being distributed to the resource agencies, Indian tribes, and other government offices specified in paragraph (d)(2) of this section and §16.10(f) of this part, if applicable.
- (2) As soon as an applicant files such application documents with the Commission, or promptly after receipt in the case of documents described in paragraph (d)(2)(iii) of this section, as the Commission may direct, the applicant must serve on every resource agency and Indian tribe consulted, on other government offices, and, in the case of applications for surrender or nonpower license, any state, municipal, interstate, or Federal agency which is authorized to assume regulatory supervision over the land, waterways, and facilities covered by the application for surrender or nonpower license, copies

- (i) Its application for a new license, a nonpower license, an exemption from licensing, or a surrender of the project;
- (ii) Any deficiency correction, revision, supplement, response to additional information request, or amendment to the application; and
- (iii) Any written correspondence from the Commission requesting the correction of deficiencies or the submittal of additional information.
- (e) Resource agency or Indian tribe waiver of compliance with consultation requirement. (1) If a resource agency or Indian tribe waives in writing compliance with any requirement of this section, a potential applicant does not have to comply with that requirement as to that agency or Indian tribe.
- (2) If a resource agency or Indian tribe fails to timely comply with a provision regarding a requirement of this section, a potential applicant may proceed to the next sequential requirement of this section without waiting for the resource agency or Indian tribe to comply.
- (3) The failure of a resource agency or Indian tribe to timely comply with a provision regarding a requirement of this section does not preclude its participation in subsequent stages of the consultation process.
- (4) Following July 23, 2003 a potential license applicant engaged in pre-filing consultation under this part may during first stage consultation request to incorporate into pre-filing consultation any element of the integrated license application process provided for in part 5 of this chapter. Any such request must be accompanied by a:
- (i) Specific description of how the element of the part 5 license application would fit into the pre-filing consultation process under this part; and
- (ii) Demonstration that the potential license applicant has made every reasonable effort to contact all resource agencies, Indian tribes, non-governmental organizations, and others affected by the potential applicant's proposal, and that a consensus exists in favor of incorporating the specific element of the part 5 process into the prefiling consultation under this part.
- (f) Application requirements documenting consultation and any disagreements with resource agencies or Indian

- tribes. An applicant must show in Exhibit E of its application that it has met the requirements of paragraphs (b) through (d) of this section, and §16.8(i), and must include:
- (1) Any resource agency's or Indian tribe's letters containing comments, recommendations, and proposed terms and conditions;
- (2) Any letters from the public containing comments and recommenda-
- (3) Notice of any remaining disagreement with a resource agency or Indian tribe on:
- (i) The need for a study or the manner in which a study should be conducted and the applicant's reasons for disagreement, and
- (ii) Information on any environmental protection, mitigation, or enhancement measure, including the basis for the applicant's disagreement with the resource agency or Indian tribe.
- (4) Evidence of any waivers under paragraph (e) of this section;
- (5) Evidence of all attempts to consult with a resource agency or Indian tribe, copies of related documents showing the attempts, and documents showing the conclusion of the second stage of consultation:
- (6) An explanation of how and why the project would, would not, or should not, comply with any relevant comprehensive plan as defined in §2.19 of this chapter and a description of any relevant resource agency or Indian tribe determination regarding the consistency of the project with any such comprehensive plan;
- (7) A description of how the applicant's proposal addresses the significant resource issues raised by members of the public during the joint meeting held pursuant to paragraph (b)(2) of this section.
- (g) Requests for privileged treatment of pre-filing submission. If a potential applicant requests privileged treatment of any information submitted to the Commission during pre-filing consultation (except for the information specified in paragraph (b)(1) of this section), the Commission will treat the request in accordance with the provisions in §388.112 of this chapter until the date

the application is filed with the Commission.

- (h) Other meetings. Prior to holding a meeting with a resource agency or Indian tribe, other than a joint meeting pursuant to paragraph (b)(3)(i) or (c)(6)(i) of this section, a potential applicant must provide the Commission and each resource agency or Indian tribe (with an area of interest, expertise, or responsibility similar or related to that of the resource agency or Indian tribe with which the potential applicant is to meet) with written notice of the time and place of each meeting and a written agenda of the issues to be discussed at the meeting at least 15 days in advance.
- (i) Public participation. (1) At least 14 days in advance of the joint meeting held pursuant to paragraph (b)(3), the potential applicant must publish notice, at least once, of the purpose, location, and timing of the joint meeting, in a daily or weekly newspaper published in the county or counties in which the existing project or any part thereof or the lands affected thereby are situated. The notice shall include a copy of the written agenda of the issues to be discussed at the joint meeting prepared pursuant to paragraph (b)(3)(ii) of this section.
- (2)(i) A potential applicant must make available to the public for inspection and reproduction the information specified in paragraph (b)(1) of this section from the date on which the notice required by paragraph (i)(1) of this section is first published until a final order is issued on the license application.
- (ii) The provisions of \$16.7(e) shall govern the form and manner in which the information is to be made available for public inspection and reproduction.
- (iii) A potential applicant must make available to the public for inspection at the joint meeting required by paragraph (b)(3) of this section the information specified in paragraph (b)(2) of this section.
- (j) Critical Energy Infrastructure Information. If this section requires an applicant to reveal Critical Energy Infrastructure Information (CEII), as defined by §388.113(c) of this chapter, to

any person, the applicant shall follow the procedures set out in §16.7(d)(7).

[Order 513, 54 FR 23806, June 2, 1989, as amended by Order 513-A, 55 FR 16, Jan. 2, 1990; Order 533, 56 FR 23154, May 20, 1991; 56 FR 61156, Dec. 2, 1991; Order 2002, 68 FR 51140, Aug. 25, 2003; Order 643, 68 FR 52095, Sept. 2, 2003; 68 FR 61743, Oct. 30, 2003]

§ 16.9 Applications for new licenses and nonpower licenses for projects subject to sections 14 and 15 of the Federal Power Act.

- (a) Applicability. This section applies to an applicant for a new license or nonpower license for a project subject to sections 14 and 15 of the Federal Power Act.
- (b) Filing requirement. (1) An applicant for a license under this section must file its application at least 24 months before the existing license expires.
- (2) An application for a license under this section must meet the requirements of §4.32 (except that the Director of the Office of Energy Projects may provide more than 90 days in which to correct deficiencies in applications) and, as appropriate, §§4.41, 4.51, or 4.61 of this chapter.
- (3) The requirements of §4.35 of this chapter do not apply to an application under this section, except that the Commission will reissue a public notice of the application in accordance with the provisions of §16.9(d)(1) if an amendment described in §4.35(f) of this chapter is filed.
- (4) If the Commission rejects or dismisses an application pursuant to the provisions of §4.32 of this chapter, the application may not be refiled after the new license application filing deadline specified in §16.9(b)(1).
- (c) Final amendments. All amendments to an application, including the final amendment, must be filed with the Commission and served on all competing applicants no later than the date specified in the notice issued under paragraph (d)(2).
- (d) Commission notice. (1) Upon acceptance of an application for a new license or a nonpower license, the Commission will give notice of the application and of the dates for comment, intervention, and protests by:
- (i) Publishing notice in the FEDERAL REGISTER;

- (ii) Publishing notice once every week for four weeks in a daily or weekly newspaper published in the county or counties in which the project or any part thereof or the lands affected thereby are situated; and
- (iii) Notifying appropriate Federal, state, and interstate resource agencies, Indian tribes, and non-governmental organizations, by electronic means if practical, otherwise by mail.
- (2) Within 60 days after the new license application filing deadline, the Commission will issue a notice on the processing deadlines established under §4.32 of this chapter, estimated dates for further processing deadlines under §4.32 of this chapter, deadlines for complying with the provisions of §4.36(d)(2) (ii) and (iii) of this chapter in cases where competing applications are filed, and the date for final amendments and will:
- (i) Publish the notice in the FEDERAL REGISTER:
- (ii) Provide the notice to appropriate Federal, state, and interstate resource agencies and Indian tribes, by electronic means if practical, otherwise by mail: and
- (iii) Serve the notice on all parties to the proceedings pursuant to §385.2010 of this chapter.
- (3) Where two or more mutually exclusive competing applications have been filed for the same project, the final amendment date and deadlines for complying with the provisions of §4.36(d)(2) (ii) and (iii) of this chapter established pursuant to the notice issued under paragraph (d)(2) of this section will be the same for all such applications.
- (4) The provisions of §4.36(d)(2)(i) of this chapter will not be applicable to applications filed pursuant to this section.

[Order 513, 54 FR 23806, June 2, 1989, as amended by Order 2002, 68 FR 51142, Aug. 25, 2003; Order 653, 70 FR 8724, Feb. 23, 2005]

§ 16.10 Information to be provided by an applicant for new license: Filing requirements.

(a) Information to be supplied by all applicants. All applicants for a new license under this part must file the following information with the Commission:

- (1) A discussion of the plans and ability of the applicant to operate and maintain the project in a manner most likely to provide efficient and reliable electric service, including efforts and plans to:
- (i) Increase capacity or generation at the project;
- (ii) Coordinate the operation of the project with any upstream or downstream water resource projects; and
- (iii) Coordinate the operation of the project with the applicant's or other electrical systems to minimize the cost of production.
- (2) A discussion of the need of the applicant over the short and long term for the electricity generated by the project, including:
- (i) The reasonable costs and reasonable availability of alternative sources of power that would be needed by the applicant or its customers, including wholesale customers, if the applicant is not granted a license for the project;
- (ii) A discussion of the increase in fuel, capital, and any other costs that would be incurred by the applicant or its customers to purchase or generate power necessary to replace the output of the licensed project, if the applicant is not granted a license for the project;
- (iii) The effect of each alternative source of power on:
- (A) The applicant's customers, including wholesale customers;
- (B) The applicant's operating and load characteristics; and
- (C) The communities served or to be served, including any reallocation of costs associated with the transfer of a license from the existing licensee.
- (3) The following data showing need and the reasonable cost and availability of alternative sources of power:
- (i) The average annual cost of the power produced by the project, including the basis for that calculation;
- (ii) The projected resources required by the applicant to meet the applicant's capacity and energy requirements over the short and long term including:
- (A) Energy and capacity resources, including the contributions from the applicant's generation, purchases, and load modification measures (such as

conservation, if considered as a resource), as separate components of the total resources required;

- (B) A resource analysis, including a statement of system reserve margins to be maintained for energy and capacity; and
- (C) If load management measures are not viewed as resources, the effects of such measures on the projected capacity and energy requirements indicated separately:
- (iii) For alternative sources of power, including generation of additional power at existing facilities, restarting deactivated units, the purchase of power off-system, the construction or purchase and operation of a new power plant, and load management measures such as conservation:
- (A) The total annual cost of each alternative source of power to replace project power;
- (B) The basis for the determination of projected annual cost; and
- (C) A discussion of the relative merits of each alternative, including the issues of the period of availability and dependability of purchased power, average life of alternatives, relative equivalent availability of generating alternatives, and relative impacts on the applicant's power system reliability and other system operating characteristics; and
- (iv) The effect on the direct providers (and their immediate customers) of alternate sources of power.
- (4) If an applicant uses power for its own industrial facility and related operations, the effect of obtaining or losing electricity from the project on the operation and efficiency of such facility or related operations, its workers, and the related community.
- (5) If an applicant is an Indian tribe applying for a license for a project located on the tribal reservation, a statement of the need of such tribe for electricity generated by the project to foster the purposes of the reservation.
- (6) A comparison of the impact on the operations and planning of the applicant's transmission system of receiving or not receiving the project license, including:
- (i) An analysis of the effects of any resulting redistribution of power flows on line loading (with respect to appli-

cable thermal, voltage, or stability limits), line losses, and necessary new construction of transmission facilities or upgrading of existing facilities, together with the cost impact of these effects:

- (ii) An analysis of the advantages that the applicant's transmission system would provide in the distribution of the project's power; and
- (iii) Detailed single-line diagrams, including existing system facilities identified by name and circuit number, that show system transmission elements in relation to the project and other principal interconnected system elements. Power flow and loss data that represent system operating conditions may be appended if applicants believe such data would be useful to show that the operating impacts described would be beneficial.
- (7) If the applicant has plans to modify existing project facilities or operations, a statement of the need for, or usefulness of, the modifications, including at least a reconnaissance-level study of the effect and projected costs of the proposed plans and any alternate plans, which in conjunction with other developments in the area would conform with a comprehensive plan for improving or developing the waterway and for other beneficial public uses as defined in section 10(a)(1) of the Federal Power Act.
- (8) If the applicant has no plans to modify existing project facilities or operations, at least a reconnaissance-level study to show that the project facilities or operations in conjunction with other developments in the area would conform with a comprehensive plan for improving or developing the waterway and for other beneficial public uses as defined in section 10(a)(1) of the Federal Power Act.
- (9) A statement describing the applicant's financial and personnel resources to meet its obligations under a new license, including specific information to demonstrate that the applicant's personnel are adequate in number and training to operate and maintain the project in accordance with the provisions of the license.

- (10) If an applicant proposes to expand the project to encompass additional lands, a statement that the applicant has notified, by certified mail, property owners on the additional lands to be encompassed by the project and governmental agencies and subdivisions likely to be interested in or affected by the proposed expansion.
- (11) The applicant's electricity consumption efficiency improvement program, as defined under section 10(a)(2)(C) of the Federal Power Act, including:
- (i) A statement of the applicant's record of encouraging or assisting its customers to conserve electricity and a description of its plans and capabilities for promoting electricity conservation by its customers; and
- (ii) A statement describing the compliance of the applicant's energy conservation programs with any applicable regulatory requirements.
- (12) The names and mailing addresses of every Indian tribe with land on which any part of the proposed project would be located or which the applicant reasonably believes would otherwise be affected by the proposed project.
- (b) Information to be provided by an applicant who is an existing licensee. An existing licensee that applies for a new license must provide:
- (1) The information specified in paragraph (a).
- (2) A statement of measures taken or planned by the licensee to ensure safe management, operation, and maintenance of the project, including:
- (i) A description of existing and planned operation of the project during flood conditions;
- (ii) A discussion of any warning devices used to ensure downstream public safety:
- (iii) A discussion of any proposed changes to the operation of the project or downstream development that might affect the existing Emergency Action Plan, as described in subpart C of part 12 of this chapter, on file with the Commission;
- (iv) A description of existing and planned monitoring devices to detect structural movement or stress, seepage, uplift, equipment failure, or water conduit failure, including a description

- of the maintenance and monitoring programs used or planned in conjunction with the devices; and
- (v) A discussion of the project's employee safety and public safety record, including the number of lost-time accidents involving employees and the record of injury or death to the public within the project boundary.
- (3) A description of the current operation of the project, including any constraints that might affect the manner in which the project is operated.
- (4) A discussion of the history of the project and record of programs to upgrade the operation and maintenance of the project.
- (5) A summary of any generation lost at the project over the last five years because of unscheduled outages, including the cause, duration, and corrective action taken.
- (6) A discussion of the licensee's record of compliance with the terms and conditions of the existing license, including a list of all incidents of noncompliance, their disposition, and any documentation relating to each incident.
- (7) A discussion of any actions taken by the existing licensee related to the project which affect the public.
- (8) A summary of the ownership and operating expenses that would be reduced if the project license were transferred from the existing licensee.
- (9) A statement of annual fees paid under Part I of the Federal Power Act for the use of any Federal or Indian lands included within the project boundary.
- (c) Information to be provided by an applicant who is not an existing licensee. An applicant that is not an existing licensee must provide:
- (1) The information specified in paragraph (a).
- (2) A statement of the applicant's plans to manage, operate, and maintain the project safely, including:
- (i) A description of the differences between the operation and maintenance procedures planned by the applicant and the operation and maintenance procedures of the existing licensee;
- (ii) A discussion of any measures proposed by the applicant to implement the existing licensee's Emergency Action Plan, as described in subpart C of

part 12 of this chapter, and any proposed changes;

- (iii) A description of the applicant's plans to continue safety monitoring of existing project instrumentation and any proposed changes; and
- (iv) A statement indicating whether or not the applicant is requesting the licensee to provide transmission services under section 15(d) of the Federal Power Act.
- (d) *Inclusion in application*. The information required to be provided by this section must be included in the application as a separate exhibit labeled "Exhibit H."

[Order 513, 54 FR 23806, June 2, 1989, as amended by Order 533, 56 FR 23154, May 20, 1991; 56 FR 61156, Dec. 2, 1991; Order 2002, 68 FR 51142, Aug. 25, 2003]

§16.11 Nonpower licenses.

- (a) Information to be provided by all applicants for nonpower licenses. (1) An applicant for a nonpower license must provide the following information in its application:
- (i) The information required by §§ 4.51 or 4.61 of this chapter, as appropriate;
- (ii) A description of the nonpower purpose for which the project is to be used:
- (iii) A showing of how the nonpower use conforms with a comprehensive plan for improving or developing the waterway and for other beneficial public uses as defined in section 10(a)(1) of the Federal Power Act:
- (iv) A statement of any impact that converting the project to nonpower use may have on the power supply of the system served by the project, including the additional cost of power if an alternative generating source is used to offset the loss of the project's generation;
- (v) A statement identifying the state, municipal, interstate, or Federal agency, which is authorized and willing to assume regulatory supervision over the land, waterways, and facilities to be included within the nonpower project;
- (vi) Copies of written communication and documentation of oral communication that the applicant may have had with any jurisdictional agency or governmental unit authorized and willing to assume regulatory control over the project and the point of time at which

the agency or unit would assume regulatory control;

- (vii) A statement that demonstrates that the applicant has complied with the requirements of §16.8(d)(2);
- (viii) A proposal that shows the manner in which the applicant plans to remove or otherwise dispose of the project's power facilities;
- (ix) Any proposal to repair or rehabilitate any nonpower facilities;
- (x) A statement of the costs associated with removing the project's power facilities and with any necessary restoration and rehabilitation work; and
- (xi) A statement that demonstrates that the applicant has resources to ensure the integrity and safety of the remaining project facilities and to maintain the nonpower functions of the project until the governmental unit or agency assumes regulatory control over the project.
 - (2) [Reserved]
- (b) Termination of a proceeding for a nonpower license. The Commission may deny an application for a nonpower license and turn the project over to any agency that has jurisdiction over the land or reservations if:
- (1) An existing project is located on public lands or reservations of the United States;
- (2) Neither the existing licensee nor any other entity has filed an application for a new license for the project;
- (3) No one has filed a recommendation to take over the project pursuant to \$16.14; and
- (4) The agency that has jurisdiction over the land or reservations demonstrates that it is able and willing to:
- (i) Accept immediate responsibility for the nonpower use of the project; and.
- (ii) Pay the existing licensee for its net investment in the project and any severance damages specified in section 14(a) of the Federal Power Act.
- (c) Termination of nonpower license. A nonpower license will be terminated by Commission order when the Commission determines that a state, municipal, interstate, or Federal agency has jurisdiction over, and is willing to assume regulatory responsibility for, the

land, waterways, and facilities included within the nonpower license.

[Order 513, 54 FR 23806, June 2, 1989, as amended by Order 2002, 68 FR 51142, Aug. 25, 2003]

§16.12 Application for exemption from licensing by a licensee whose license is subject to sections 14 and 15 of the Federal Power Act.

- (a) An existing licensee whose license is subject to sections 14 and 15 of the Federal Power Act may apply for an exemption for the project.
- (b) An applicant for an exemption under paragraph (a) must meet the requirements of subpart K or subpart J of part 4 of this chapter, and §§ 16.5, 16.6, 16.7, 16.8, 16.9(b) (1), (2) (except the requirement to comply with §§ 4.41, 4.51, or 4.61 of this chapter), 16.9(c), 16.10(a), 16.10(b), and 16.10(d).
- (c) The Commission will process an application by an existing licensee for an exemption for the project in accordance with §§16.9(b)(3), 16.9(b)(4), and 16.9(d).
- (d) If a license application is filed in competition with an application for exemption filed by the existing licensee, the Commission will decide among the competing applications in accordance with the standards of §16.13 and not in accordance with the provisions of §4.37(d)(2) of this chapter.

[Order 513, 54 FR 23806, June 2, 1989, as amended by Order 699, 72 FR 45324, Aug. 14, 2007]

§ 16.13 Standards and factors for issuing a new license.

- (a) In determining whether a final proposal for a new license under section 15 of the Federal Power Act is best adapted to serve the public interest, the Commission will consider the factors enumerated in sections 15(a)(2) and (a)(3) of the Federal Power Act.
- (b) If there are only insignificant differences between the final applications of an existing licensee and a competing applicant after consideration of the factors enumerated in section 15(a)(2) of the Federal Power Act, the Commission will determine which applicant will receive the license after considering:

- (1) The existing licensee's record of compliance with the terms and conditions of the existing license; and
- (2) The actions taken by the existing licensee related to the project which affect the public.
- (c) An existing licensee that files an application for a new license in conjunction with an entity or entities that are not currently licensees of all or part of the project will not be considered an existing licensee for the purpose of the insignificant differences provision of section 15(a)(2) of the Federal Power Act.

Subpart C—Takeover Provisions for Projects Subject to Sections 14 and 15 of the Federal Power Act

§16.14 Departmental recommendation for takeover.

- (a) A Federal department or agency may file a recommendation that the United States exercise its right to take over a hydroelectric power project with a license that is subject to sections 14 and 15 of the Federal Power Act. The recommendation must:
- (1) Be filed no earlier than five years before the license expires and no later than the end of the comment period specified by the Commission in:
- (i) A notice of application for a new license, a nonpower license, or an exemption for the project; or
- (ii) A notice of an amendment to an application for a new license, a nonpower license, or an exemption;
- (2) Be filed in accordance with the formal requirements for filings in subpart T of part 385 of the Commission's regulations and be served on each relevant Federal and state resource agency, all applicants for new license, nonpower license or exemption, and any other party to the proceeding;
- (3) Specify the project works that would be taken over by the United States;
- (4) Describe the proposed Federal operation of the project, including any plans for its redevelopment, and discuss the manner in which takeover would serve the public interest as fully as non-Federal development and operation:

- (5) State whether the agency intends to undertake the operation of the project; and
- (6) Include the information required by §§ 4.41, 4.51, or 4.61 of this chapter, as appropriate.
- (b) A department or agency that files a takeover recommendation becomes a party to the proceeding.
- (c) An applicant or potential applicant for a new license, a nonpower license, or an exemption that involves a takeover recommendation may file a reply to the recommendation, within 120 days from the date the takeover recommendation is filed with the Commission. The reply must be filed with the Commission in accordance with part 385 of the Commission's regulations and a copy of such a reply must be served on the agency recommending the takeover and on any other party to the proceeding.

§ 16.15 Commission recommendation to Congress.

Upon receipt of a recommendation from any Federal department or agency, a proposal of any party, or on the Commission's own motion, and after notice and opportunity for hearing, the Commission may determine that a project may be taken over by the United States, issue an order on its findings and recommendations, and forward a copy to Congress.

§ 16.16 Motion for stay by Federal department or agency.

- (a) Within 30 days of the date on which an order granting a new license or exemption is issued, a Federal department or agency that has filed a takeover recommendation under §16.14 may file a motion under §385.212 of this chapter to request a stay of the effective date of the license or exemption order.
- (b)(1) If a Federal department or agency files a motion under paragraph (a), the Commission will stay the effective date of the order issuing the license or exemption for two years.
- (2) The stay issued under paragraph (b)(1) of this section may be terminated either:
- (i) Upon motion of the department or agency that requested the stay; or
 - (ii) By action of Congress.

- (c) The Commission will notify Congress if:
- (1) An order granting a stay under paragraph (b)(1) of this section is issued:
- (2) Any license or exemption order becomes effective by reason of the termination of a stay; or
- (3) Any license or exemption order becomes effective by reason of the expiration of a stay.
- (d) The Commission's order granting the license or exemption will automatically become effective:
- (1) Thirty days after issuance, if no request for stay is filed, provided that no appeal or rehearing is filed;
- (2) When the period of the stay expires; or
- (3) When the stay is terminated under paragraph (b)(2) of this section.

[Order 513, 54 FR 23806, June 2, 1989, as amended by Order 699, 72 FR 45324, Aug. 14, 2007]

§ 16.17 Procedures upon Congressional authorization of takeover.

- If Congress authorizes the takeover of a hydroelectric power project as provided under section 14 of the Federal Power Act:
- (a) The Commission or the Director of the Office of Energy Projects will notify the existing licensee in writing of the authorization at least two years before the takeover occurs; and
- (b) The licensee must present any claim for compensation to the Commission:
- (1) Within six months of issuance of the notice of takeover; and
- (2) As provided in section 14 of the Federal Power Act.

Subpart D—Annual Licenses for Projects Subject to Sections 14 and 15 of the Federal Power Act

§ 16.18 Annual licenses for projects subject to sections 14 and 15 of the Federal Power Act.

- (a) This section applies to projects with licenses subject to sections 14 and 15 of the Federal Power Act.
- (b) The Commission will issue an annual license to an existing licensee under the terms and conditions of the

existing license upon expiration of its existing license to allow:

- (1) The licensee to continue to operate the project while the Commission reviews any applications for a new license, a nonpower license, an exemption, or a surrender;
- (2) The orderly removal of a project, if the United States does not take over a project and no new power or nonpower license or exemption will be issued; or
- (3) The orderly transfer of a project to:
- (i) The United States, if takeover is elected: or
- (ii) A new licensee, if a new power or nonpower license is issued to that licensee.
- (c) An annual license issued under this section will be considered renewed automatically without further order of the Commission, unless the Commission orders otherwise.
- (d) In issuing an annual license, the Commission may incorporate additional or revised interim conditions if necessary and practical to limit adverse impacts on the environment.

[Order 513, 54 FR 23806, June 2, 1989, as amended by Order 513-A, 55 FR 18, Jan. 2, 1990; Order 540, 57 FR 21738, May 22, 1992]

Subpart E—Projects With Minor and Minor Part Licenses Not Subject to Sections 14 and 15 of the Federal Power Act

- § 16.19 Procedures for an existing licensee of a minor hydroelectric power project or of a minor part of a hydroelectric power project with a license not subject to sections 14 and 15 of the Federal Power Act.
- (a) Applicability. This section applies to an existing licensee of a minor hydroelectric power project or of a minor part of a hydroelectric power project that is not subject to sections 14 and 15 of the Federal Power Act.
- (b) Notification procedures. (1) An existing licensee with a minor license or a license for a minor part of a hydroelectric project must file a notice of intent pursuant to §16.6(b).
- (2) If the license of an existing licensee expires on or after October 17, 1994, the licensee must notify the Commission as required under §16.6(b) at

least five years before the expiration of the existing license.

- (3) The Commission will give notice of a licensee's intent to file or not to file an application for a subsequent license in accordance with §16.6(d).
- (c) Requirement to make information available. (1) Except as provided in paragraph (c)(2) of this section, a licensee must make the information described in §16.7 available to the public for inspection and reproduction when it gives notice to the Commission under paragraph (b).
- (2) The requirement of paragraph (c)(1) of this section does not apply if an applicant filed an application for a subsequent license on or before July 3, 1989

[Order 513, 54 FR 23806, June 2, 1989, as amended by Order 2002, 68 FR 51142, Aug. 25, 2003; Order 699, 72 FR 45324, Aug. 14, 2007]

§ 16.20 Applications for subsequent license for a project with an expiring license not subject to sections 14 and 15 of the Federal Power Act.

- (a) Applicability. This section applies to an application for subsequent license for a project with an expiring license that is not subject to sections 14 and 15 of the Federal Power Act.
- (b) Licensing proceeding. (1) An applicant for a license for a project with an expiring license not subject to sections 14 and 15 of the Federal Power Act must file its application under Part I of the Federal Power Act.
- (2) The provisions of section 7(a) of the Federal Power Act do not apply to licensing proceedings involving an application described in paragraph (b)(1).
- (c) Requirement to file. An applicant must file an application for subsequent license at least 24 months before the expiration of the existing license.
- (d) Requirements for and processing of applications. An application for subsequent license must meet the requirements of, and will be processed in accordance with, §§16.5, 16.8, 16.9(b)(2), 16.9(b)(3), 16.9(b)(4), 16.9(c), and 16.9(d).
- (e) Applicant notice. An applicant for subsequent license or exemption that proposes to expand an existing project to encompass additional lands must include in its application a statement

that the applicant has notified, by certified mail, property owners on the additional lands to be encompassed by the project and governmental agencies and subdivisions likely to be interested in or affected by the proposed expansion

[Order 513, 54 FR 23806, June 2, 1989, as amended by Order 2002, 68 FR 51142, Aug. 25, 2003]

§16.21 Operation of projects with a minor or minor part license not subject to sections 14 and 15 of the Federal Power Act after expiration of a license.

(a) A licensee of a minor or minor part project not subject to sections 14 and 15 of the Federal Power Act that has filed an application for a subsequent license or exemption may continue to operate the project in accordance with the terms and conditions of the license after the minor or minor part license expires until the Commission acts on its application.

(b) If the licensee of a minor or minor part project not subject to sections 14 and 15 of the Federal Power Act has not filed an application for a subsequent license or exemption, the Commission may issue an order requiring the licensee to continue to operate its project in accordance with the terms and conditions of the license until the Commission either acts on any applications for subsequent license timely filed by another entity or takes action pursuant to §§ 16.25 or 16.26.

§ 16.22 Application for an exemption by a licensee with a minor or minor part license for a project not subject to sections 14 and 15 of the Federal Power Act.

(a) Applicability. This section applies to an existing licensee with a license for a project not subject to sections 14 and 15 of the Federal Power Act.

(b) Information requirements. An applicant for an exemption must meet the requirements of, and will be processed in accordance with, subpart K or subpart J of part 4 of this chapter, and §§16.5, 16.8, 16.9(b)(2) (except the requirement to comply with §§4.41, 4.51, or 4.61 of this chapter), §§16.9(b)(3), 16.9(b)(4), 16.9(c), and 16.9(d).

(c) Standard of comparison. If an application for subsequent license is filed in

competition with an application for exemption by an existing licensee, the Commission will decide among competing applications in accordance with the standards of §16.13 and not in accordance with the provisions of §4.37(d)(2) of this chapter.

[Order 513, 54 FR 23806, June 2, 1989, as amended by Order 699, 72 FR 45324, Aug. 14, 2007]

Subpart F—Procedural Matters

§ 16.23 Failure to file timely notices of intent.

(a) An existing licensee of a water power project with a license subject to sections 14 and 15 of the Federal Power Act that fails to file a notice of intent pursuant to §16.6(b) by the deadlines specified in §16.6(c) shall be deemed to have filed a notice of intent indicating that it does not intend to file an application for new license, nonpower license, or exemption.

(b) An existing licensee of a water power project with a license not subject to sections 14 and 15 of the Federal Power Act that fails to file a notice of intent pursuant to §16.6(b) by the deadlines specified in §16.20(c) shall be deemed to have filed a notice of intent indicating that it does not intend to file an application for subsequent license or exemption.

§ 16.24 Prohibitions against filing applications for new license, nonpower license, exemption, or subsequent license.

(a) Licenses subject to sections 14 and 15 of the Federal Power Act. (1) An existing licensee with a license subject to sections 14 and 15 of the Federal Power Act that informs the Commission that it does not intend to file an application for new license, nonpower license, or exemption for a project, as required by §16.6, may not file an application for new license, nonpower license, or exemption for the project, either individually or in conjunction with an entity or entities that are not currently licensees of the project.

(2) An existing licensee with a license subject to sections 14 and 15 of the Federal Power Act that fails to file an application for new license, nonpower license, or exemption for a project at

least 24 months before the expiration of the existing license for the project may not file an application for new license, nonpower license, or exemption for the project, either individually or in conjunction with an entity or entities that are not currently licensees of the project.

- (b) Licenses not subject to sections 14 and 15 of the Federal Power Act. (1) An existing licensee with a license not subject to sections 14 and 15 of the Federal Power Act that informs the Commission that it does not intend to file an application for subsequent license or exemption for a project, as required by §16.6, may not file an application for subsequent license or exemption for the project, either individually or in conjunction with an entity or entities that are not currently licensees of the project.
- (2) An existing licensee with a license not subject to sections 14 and 15 of the Federal Power Act that fails to file an application for subsequent license or exemption for a project by the deadlines specified in §16.20(c) may not file an application for subsequent license or exemption for the project, either individually or in conjunction with an entity or entities that are not currently licensees of the project.

§ 16.25 Disposition of a project for which no timely application is filed following a notice of intent to file.

(a) If an existing licensee that indicates in the notice filed pursuant to §16.6 that it will file an application for new license, nonpower license, subsequent license, or an exemption does not file its application individually or in conjunction with an entity or entities that are not currently licensees of the project at least 24 months before its existing license expires in the case of licenses subject to sections 14 and 15 of the Federal Power Act, or by the deadlines specified in §16.20(c) in the case of licenses not subject to sections 14 and 15 of the Federal Power Act, and no other applicant files an application within the appropriate time or all pending applications filed before the applicable filing deadline are subsequently rejected or dismissed pursuant to §4.32 of this chapter, the Commission will publish in the FEDERAL REG-

ISTER and once in a daily or weekly newspaper published in the county or counties in which the project or any part thereof or the lands affected thereby are situated, notice soliciting applications from potential applicants other than the existing licensee.

- (b) A potential applicant that files a notice of intent within 90 days from the date of the public notice issued pursuant to paragraph (a):
- (1) May apply for a license under Part I of the Federal Power Act and part 4 of this chapter (except §4.38) within 18 months of the date on which it files its notice; and
- (2) Must comply with the requirements of §16.8 and, if the project would have a total installed capacity of over 2,000 horsepower, §16.10.
- (c) The existing licensee must file a schedule for the filing of a surrender application for the project, for the approval of the Director of the Office of Energy Projects, 90 days:
- (1) After the due date established for any notice of intent issued under paragraph (a), if no notices of intent were received: or
- (2) After the due date for any application filed under paragraph (b)(1), if no application has been filed.
- (d) Any application for surrender must be filed according to the approved schedule, must comply with the requirements of §16.8 and part 6 of this chapter, and must provide for disposition of any project facility.

§ 16.26 Disposition of a project for which no timely application is filed following a notice of intent not to file.

(a) If an existing licensee indicates in the notice filed pursuant to §16.6 that it will not file an application for new license, nonpower license, subsequent license, or exemption and no other applicant files an application at least 24 months before the existing license expires in the case of licenses subject to sections 14 and 15 of the Federal Power Act, or by the deadlines specified in §16.20(c) in the case of licenses not subject to sections 14 and 15 of the Federal Power Act, the Director of the Office of Energy Projects will provide the existing licensee with written notice that

Pt. 20

no timely applications for the project have been filed.

- (b) The existing licensee, within 90 days from the date of the written notice provided in paragraph (a), must file a schedule for the filing of a surrender application for the project for the approval of the Director of the Office of Energy Projects.
- (c) Any application for surrender must be filed according to the approved schedule, must comply with the requirements of §16.8 and part 6 of this chapter, and must provide for disposition of any project facility.

PART 20—AUTHORIZATION OF THE ISSUANCE OF SECURITIES BY LI-CENSEES AND COMPANIES SUB-JECT TO SECTIONS 19 AND 20 OF THE FEDERAL POWER ACT

Sec.

20.1 Applicability.

20.2 Regulation of issuance of securities.

AUTHORITY: Secs. 3(16), 19, 20, 41 Stat. 1063, 1073; secs. 201, 309, 49 Stat. 838, 858; 16 U.S.C. 796 (16), 812, 813, 825k.

SOURCE: Order 170, 19 FR 2013, Apr. 8, 1954, unless otherwise noted.

§ 20.1 Applicability.

- (a) Without special proceeding for regulation. Every security issue within the scope of the jurisdiction conferred upon the Commission by sections 19 and 20 of the Federal Power Act shall be subject to the provisions of §20.2, except a security issue by a person organized and operating in a State under the laws of which its security issues are regulated by a State commission, or by any one described in subsection 201(f) of the act. No other security issue within the scope of sections 19 and 20 shall be subject to §20.2 except as provided in paragraph (b) of this section.
- (b) Reservation of possibility of regulation in other cases. Not later than 10 days prior to any proposed security issuance which is within the scope of section 19 or section 20 of the act, but excepted by paragraph (a) of this section, any person or state entitled to do so under section 19 or section 20, may file a complaint or request in accordance with the applicable rules of the Commission, or the Commission upon

its own motion may by order initiate a proceeding, raising the question whether issuance of such security should be subjected by Commission order to the provisions of §20.2. After notice of such filing or order, and until such request or complaint is denied or dismissed or the proceeding initiated by such order is terminated without subjecting the issuance of the security to the provisions of §20.2, the security in question shall not be issued except it be issued subject to and in compliance with §20.2.

$\S 20.2$ Regulation of issuance of securities.

The licensee or other person issuing or proposing to issue any security subjected to this section by or pursuant to §20.1, shall be subject to and shall comply with the same requirements as the Commission would administer to it if it were a public utility issuing the security within the meaning and subject to the requirements of section 204 of the Act and part 34 of this subchapter.

CROSS REFERENCE: For applications for authorization of the issuance of securities or the assumption of liabilities, see part 34 of this chapter.

PART 24—DECLARATION OF INTENTION

AUTHORITY: 16 U.S.C. 791a–825r; 44 U.S.C. 3501 $et\ seq.$; 42 U.S.C. 7101–7352.

§24.1 Filing.

A declaration of intention under the provisions of section 23(b) of the Act shall be filed with the Secretary of the Commission in accordance with filing procedures posted on the Commission's Web site at http://www.ferc.gov. The declaration shall give the name and post office address of the person to whom correspondence in regard to it shall be addressed, and shall be accompanied by:

(a) A brief description of the proposed project and its purposes, including such data as maximum height of the dams, a storage capacity curve of the reservoir or reservoirs showing the maximum average, and minimum operating pool levels, the initial and ultimate installed capacity of the project, the

rated horsepower and head on the turbines, and a curve of turbine discharge versus output at average and minimum operating heads.

- (b)(1) A general map (one tracing and three prints) of any convenient size and scale, showing the stream or streams to be utilized and the approximate location and the general plan of the project.
- (2) Also a detailed map of the proposed project area showing all Federal lands, and lands owned by States, if any, occupied by the project.
- (3) A profile of the river within the vicinity of the project showing the location of the proposed project and any existing improvements in the river.
- (4) A duration curve and hydrograph for the natural and proposed regulated flows at the dam site. Furnish references to the published stream flow records used and submit copies of any unpublished records used in preparation of these curves.
- (c) (1) A definite statement of the proposed method of utilizing storage or pondage seasonally, weekly and daily, during periods of low and normal flows after the plant is in operation and the system load has grown to the extent that the capacity of the plant is required to meet the load. For example, furnish:
- (i) Hydrographs covering a 10-day low water period showing the natural flow of the stream and the effect thereon caused by operations of the proposed power plant:
- (ii) Similar hydrographs covering a 10-day period during which the discharge of the stream approximates average recorded yearly flow, and
- (iii) Similar hydrographs covering a low water year using average monthly flows.
- (2) A system load curve, both daily and monthly, and the position on the load curve that the proposed project would have occupied had it been in operation.
- (3) A proposed annual rule of operation for the storage reservoir or reservoirs.

[Order 175, 19 FR 5217, Aug. 18, 1954, as amended by Order 260, 28 FR 315, Jan. 11, 1963; Order 540, 57 FR 21738, May 22, 1992; Order 737, 75 FR 43403, July 26, 2010]

PART 25—APPLICATION FOR VA-CATION OF WITHDRAWAL AND FOR DETERMINATION PERMITTING RESTORATION TO ENTRY

Sec

25.1 Contents of application.

25.2 Hearings.

§25.1 Contents of application.

Any application for vacation of a reservation effected by the filing of an application for preliminary permit or license, or for a determination under the provisions of section 24 of the Act permitting restoration for location, entry, or selection under the public lands laws, or such lands reserved or classified as power sites shall, unless the subject lands are National Forest Lands, be filed with the Bureau of Land Management, Department of the Interior, at the Bureau's office in Washington, DC or at the appropriate regional or field office of the Bureau. If the lands included in such application are National Forest Lands, the application shall be filed with the U.S. Forest Service, Department of Agriculture at the Forest Service's office in Washington, DC, or at the appropriate regional office of the U.S. Forest Service. Such application shall contain the following data: (a) Full name of applicant; (b) post-office address; (c) description of land by legal subdivisions, including section, township, range, meridian, county, State, and river basin (both main and tributary) in which the land is located; (d) public land act under which entry is intended to be made if land is restored to entry; (e) the use to which it is proposed to put the land, and a statement as to its suitability for the intended use.

(Secs. 24, 309, 41 Stat. 1075, as amended; 49 Stat. 858; 16 USC. 818, 825h)

[Order 175, 19 FR 5218, Aug. 18, 1954, as amended by Order 346, 32 FR 7495, May 20, 1967]

CROSS REFERENCE: For entries subject to section 24 of the Federal Power Act, see also 43 CFR subpart 2320.

$\S 25.2$ Hearings.

A hearing upon such an application may be ordered by the Commission in

§ 25.2

its discretion and shall be in accordance with the provisions of subpart E of part 385 of this chapter.

NOTE 1: On April 17, 1922, the Commission made the following general determination:

(a) That where lands of the United States have heretofore been, or hereafter may be, reserved or classified as power sites, such reservation or classification being made solely because such lands are either occupied by power transmission lines or their occupancy and use for such purposes has been applied for or authorized under appropriate laws of the United States, and such lands have otherwise no value for power purposes, and are not occupied in trespass, the Commission determines that the value of such lands so reserved or classified, or so applied for or authorized, will not be injured or destroyed for the purposes of power development by location, entry, or selection under the public land laws, subject to the reservation of section 24 of the Federal Water Power Act (41 Stat. 1075; 16 U.S.C. 818).

(b) That when notice is given to the Secretary of the Interior of reservations made under the provisions of section 24 of the Federal Water Power Act, such notice shall indicate what lands so reserved, if any, may, in accordance with the determination of the preceding paragraph, be declared open to location, entry, or selection, subject to the reservation of said section 24. Second Annual Report, page 128.

Note 2: On February 16, 1937, the Commission took the following action:

Consent to Establishment of Grazing Districts, Issuance of Grazing Permits, and Leasing for Grazing Purposes Under the Act of June 28, 1934, as Amended, Government Lands Reserved for Power Purposes

Upon request under date of November 2, 1936, by the acting director, Division of Grazing, Department of the Interior, for consent of the Commission, pursuant to the act of June 28, 1934 (48 Stat. 1269), to the establishment of grazing districts and the issuance of grazing permits on lands of the United States withdrawn, classified, or otherwise reserved for power purposes, except in those instances where grazing will interfere with such purposes; and

Upon request under date of December 7, 1936, by the Acting Secretary of the Interior for consent of the Commission, pursuant to the Act of June 28, 1934 (48 Stat. 1269), as amended by the Act of June 26, 1936 (49 Stat. 1976), to the leasing under section 15 of said Act as amended, of isolated tracts of lands of the United States, withdrawn for power purposes:

The Commission upon consideration of the matter finds and determines: That the establishment of grazing districts, the issuance of grazing permits, and the leasing for grazing purposes, under said Act as amended, of lands of the United States theretofore or thereafter withdrawn, classified or otherwise reserved for power purposes, but not including lands embraced within the project area of any power project theretofore licensed by the Commission or otherwise authorized by the United States, will not injure or destroy the value of such lands for the purposes of power development nor otherwise abridge the jurisdiction of the Commission; Provided, That such grazing districts shall be established and such permits and leases for grazing permits issued subject to the following conditions:

- (1) That the establishment of the grazing district or the issuance of the grazing permit or lease for grazing purposes shall in no wise diminish or affect the jurisdiction of the Commission at any time to issue permits or licenses pursuant to the provisions of the Federal Power Act (49 Stat. 838; 16 U.S.C., Sup., 791–819); and that the issuance by the Commission of a license shall immediately and automatically terminate such grazing district, permit, or lease for grazing purposes as to all lands within the project area described in such license;
- (2) That the establishment of the grazing district or the issuance of the grazing permit or lease for grazing purposes involving lands withdrawn for power purposes shall in no wise diminish or affect the jurisdiction of the Commission at any time to make further determinations that the value of any such lands for the purposes of power development will not be injured or destroyed by location entry or selection, as provided by section 24 of the

Federal Energy Regulatory Commission

Act and none of such lands shall be declared open, otherwise than as hereinbefore provided, to location, entry or selection except upon such further determination by the Commission; and any such further determination shall immediately and automatically terminate such grazing district, permit, or lease for grazing purposes as to any lands involved in such further determination.

Now, therefore, the Commission consents to the establishment of such grazing districts and the issuance of grazing permits and leases for grazing purposes of lands of the United States reserved for power purposes subject to the conditions hereinabove set out;

Provided, however, That this determination and consent shall be effective for lands embraced within grazing districts, as of the date of the establishment of such districts, and for isolated tracts of lands leased for grazing purposes, it shall be in effect when such leases are issued, provided that notice thereof is received by this Commission from the Bureau of Land Management, Department of the Interior, within 30 days thereafter, such notice to include full legal description of the lands, withdrawn for power purposes which are involved.

(Secs. 24, 308, 39, 41 Stat. 1075, as amended, 40 Stat. 858; 16 U.S.C. 818, 825g, 825h)

[Order 141, 12 FR 8493, Dec. 19, 1947, as amended by Order 225, 47 FR 19056, May 3, 1997]

CROSS REFERENCE: For regulations of the Bureau of Land Management, relating to grazing, see the Index to title 43 Chapter II.

PART 32—INTERCONNECTION OF FACILITIES

APPLICATION FOR AN ORDER DIRECTING THE ESTABLISHMENT OF PHYSICAL CONNECTION OF FACILITIES

Sec

- 32.1 Contents of the application.
- 32.2 Required exhibits.
- 32.3 Other information.
- 32.4 Filing procedure.

AUTHORITY: 42 U.S.C. 7101-7352; E.O. No. 12,009, 3 CFR 1978 Comp., p. 142; 31 U.S.C. 9701; 16 U.S.C. 791a-825r; 16 U.S.C. 2601-2645 (1988).

SOURCE: Order 141, 12 FR 8494, Dec. 19, 1947, unless otherwise noted.

APPLICATION FOR AN ORDER DIRECTING THE ESTABLISHMENT OF PHYSICAL CONNECTION OF FACILITIES

§ 32.1 Contents of the application.

Every application under section 202(b) of the Act shall set forth the following information:

- (a) The exact legal name of the applicant and of all persons named as parties in the application.
- (b) The name, title, and post office address of the person to whom correspondence in regard to the application shall be addressed.
- (c) The person named in the application who is a public utility subject to the act.
- (d) The State or States in which each electric utility named in the application operates, together with a brief description of the business of and territory, by counties and States, served by such utility.
- (e) Description of the proposed interconnection, showing proposed location, capacity and type of construction.
- (f) Reasons why the proposed connection, of facilities will be in the public interest.
- (g) What steps, if any, have been taken to secure voluntary interconnection under the provisions of section 202(a) of the Act.

[Order 141, 12 FR 8494, Dec. 19, 1947, as amended by Order 427, 36 FR 5596, Mar. 25, 1971; Order 435, 50 FR 40357, Oct. 3, 1985; Order 737, 75 FR 43403, July 26, 2010]

§ 32.2 Required exhibits.

There shall be filed with the application and as a part thereof the following exhibits:

Exhibit A. Statement of the estimated capital cost of all facilities required to establish the connection, and the estimated annual cost of operating such facilities.

Exhibit B. A general or key map on a scale not greater than 20 miles to the inch showing, in separate colors, the territory served by each utility, and the location of the facilities used for the generation and transmission of electric energy, indicating on said map the points between which connection may be established most economically.

§ 32.3 Other information.

The Commission may require additional information when it appears to be pertinent in a particular case.

§ 32.4

§32.4 Filing procedure.

All applications under Part 32 must be filed with the Secretary of the Commission in accordance with filing procedures posted on the Commission's Web site at http://www.ferc.gov.

[Order 737, 75 FR 43403, July 26, 2010]

PART 33—APPLICATIONS UNDER FEDERAL POWER ACT SECTION 203

Sec.

- 33.1 Applicability, definitions, and blanket authorizations.
- 33.2 Contents of application—general information requirements.
- 33.3 Additional information requirements for applications involving horizontal competitive impacts.
- 33.4 Additional information requirements for applications involving vertical competitive impacts.
- 33.5 Proposed accounting entries.
- 33.7 Verification.
- 33.8 Number of copies.
- 33.9 Protective order.
- 33.10 Additional information.
- 33.11 Commission procedures for the consideration of applications under section 203 of the FPA.

AUTHORITY: 16 U.S.C. 791a-825r, 2601-2645; 31 U.S.C. 9701; 42 U.S.C. 7101-7352.

SOURCE: Order 642, 65 FR 71014, Nov. 28, 2000, unless otherwise noted.

§ 33.1 Applicability, definitions, and blanket authorizations.

- (a) Applicability. (1) The requirements of this part will apply to any public utility seeking authorization under section 203 of the Federal Power Act to:
- (i) Sell, lease, or otherwise dispose of the whole of its facilities subject to the jurisdiction of the Commission, or any part thereof of a value in excess of \$10 million;
- (ii) Merge or consolidate, directly or indirectly, such facilities or any part thereof with those of any other person, by any means whatsoever;
- (iii) Purchase, acquire, or take any security with a value in excess of \$10 million of any other public utility; or
- (iv) Purchase, lease, or otherwise acquire an existing generation facility:
- (A) That has a value in excess of \$10 million; and

- (B) That is used in whole or in part for wholesale sales in interstate commerce by a public utility.
- (2) The requirements of this part shall also apply to any holding company in a holding company system that includes a transmitting utility or an electric utility if such holding company seeks to purchase, acquire, or take any security with a value in excess of \$10 million of, or, by any means whatsoever, directly or indirectly, merge or consolidate with, a transmitting utility, an electric utility company, or a holding company in a holding company system that includes a transmitting utility, or an electric utility company, with a value in excess of \$10 million.
- (b) *Definitions*. For the purposes of this part, as used in section 203 of the Federal Power Act (16 U.S.C. 824b).
- (1) Existing generation facility means a generation facility that is operational at or before the time the section 203 transaction is consummated. "The time the transaction is consummated" means the point in time when the transaction actually closes and control of the facility changes hands. "Operational" means a generation facility for which construction is complete (i.e., it is capable of producing power). The Commission will rebuttably presume that section 203(a) applies to the transfer of any existing generation facility unless the utility can demonstrate with substantial evidence that the generator is used exclusively for retail sales.
- (2) Non-utility associate company means any associate company in a holding company system other than a public utility or electric utility company that has wholesale or retail customers served under cost-based regulation.
 - (3) Value when applied to:
- (i) Transmission facilities, generation facilities, transmitting utilities, electric utility companies, and holding companies, means the market value of the facilities or companies for transactions between non-affiliated companies; the Commission will rebuttably presume that the market value is the transaction price. For transactions between affiliated companies, value means original cost undepreciated, as

defined in the Commission's Uniform System of Accounts prescribed for public utilities and licensees in part 101 of this chapter, or original book cost, as applicable:

- (ii) Wholesale contracts, means the market value for transactions between non-affiliated companies; the Commission will rebuttably presume that the market value is the transaction price. For transactions between affiliated companies, value means total expected nominal contract revenues over the remaining life of the contract; and
- (iii) Securities, means market value for transactions between non-affiliated companies; the Commission will rebuttably presume that the market value is the agreed-upon transaction price. For transactions between affiliated companies, value means market value if the securities are widely traded, in which case the Commission will rebuttably presume that market value is the market price at which the securities are being traded at the time the transaction occurs; if the securities are not widely traded, market value is determined by:
- (A) Determining the value of the company that is the issuer of the equity securities based on the total undepreciated book value of the company's assets;
- (B) Determining the fraction of the securities at issue by dividing the number of equity securities involved in the transaction by the total number of outstanding equity securities for the company; and
- (C) Multiplying the value determined in paragraph (b)(3)(iii)(A) of this section by the value determined in paragraph (b)(3)(iii)(B) of this section (i.e., the value of the company multiplied by the fraction of the equity securities at issue).
- (4) The terms associate company, electric utility company, foreign utility company, holding company, and holding company system have the meaning given those terms in the Public Utility Holding Company Act of 2005. The term holding company does not include: A State, any political subdivision of a State, or any agency, authority or instrumentality of a State or political subdivision of a State; or an electric power cooperative.

- (5) For purposes of this part, the term captive customers means any wholesale or retail electric energy customers served by a franchised public utility under cost-based regulation.
- (c) Blanket Authorizations. (1) Any holding company in a holding company system that includes a transmitting utility or an electric utility is granted a blanket authorization under section 203(a)(2) of the Federal Power Act to purchase, acquire, or take any security of:
- (i) A transmitting utility or company that owns, operates, or controls only facilities used solely for transmission in intrastate commerce and/or sales of electric energy in intrastate commerce, provided that if any public utility within the holding company system has captive customers, or owns or provides transmission service over jurisdictional transmission facilities, the holding company must report the acquisition to the Commission, including any state actions or conditions related to the transaction, and shall provide an explanation of why the transaction does not result in cross-subsidization:
- (ii) A transmitting utility or company that owns, operates, or controls only facilities used solely for local distribution and/or sales of electric energy at retail regulated by a state commission, provided that if any public utility within the holding company system has captive customers, or owns or provides transmission service over jurisdictional transmission facilities, the holding company must report the acquisition to the Commission, including any state actions or conditions related to the transaction, and shall provide an explanation of why the transaction does not result in cross-subsidization;
- (iii) An electric utility company that owns generating facilities that total 100 MW or less and are fundamentally used for its own individual load or for sales to affiliated end-users.
- (2) Any holding company in a holding company system that includes a transmitting utility or an electric utility is granted a blanket authorization under section 203(a)(2) of the Federal Power Act to purchase, acquire, or take:
- (i) Any non-voting security (that does not convey sufficient veto rights

§ 33.1

over management actions so as to convey control) in a transmitting utility, an electric utility company, or a holding company in a holding company system that includes a transmitting utility or an electric utility company; or

- (ii) Any voting security in a transmitting utility, an electric utility company, or a holding company in a holding company system that includes a transmitting utility or an electric utility company if, after the acquisition, the holding company will own less than 10 percent of the outstanding voting securities; or
- (iii) Any security of a subsidiary company within the holding company system.
- (3) The blanket authorizations granted under paragraph (c)(2) of this section are subject to the conditions that the holding company shall not:
- (i) Borrow from any electric utility company subsidiary in connection with such acquisition; or
- (ii) Pledge or encumber the assets of any electric utility company subsidiary in connection with such acquisition.
- (4) A holding company granted blanket authorizations in paragraph (c)(2) of this section shall provide the Commission copies of any Schedule 13D, Schedule 13G and Form 13F, at the same time and on the same basis, as filed with the Securities and Exchange Commission in connection with any securities purchased, acquired or taken pursuant to this section.
- (5) Any holding company in a holding company system that includes a transmitting utility or an electric utility is granted a blanket authorization under section 203(a)(2) of the Federal Power Act to acquire a foreign utility company. However, if such holding company or any of its affiliates, its subsidiaries, or associate companies within the holding company system has captive customers in the United States, or owns or provides transmission service over jurisdictional transmission facilities in the United States, the authorization is conditioned on the holding company, consistent with 18 CFR 385.2005(b), verifying by a duly authorized corporate official of the holding company that the proposed transaction:

- (i) Will not have any adverse effect on competition, rates, or regulation; and
- (ii) Will not result in, at the time of the transaction or in the future:
- (A) Any transfer of facilities between a traditional public utility associate company that has captive customers or that owns or provides transmission service over jurisdictional transmission facilities, and an associate company;
- (B) Any new issuance of securities by a traditional public utility associate company that has captive customers or that owns or provides transmission service over jurisdictional transmission facilities, for the benefit of an associate company:
- (C) Any new pledge or encumbrance of assets of a traditional public utility associate company that has captive customers or that owns or provides transmission service over jurisdictional transmission facilities, for the benefit of an associate company; or
- (D) Any new affiliate contracts between a non-utility associate company and a traditional public utility associate company that has captive customers or that owns or provides transmission service over jurisdictional transmission facilities, other than nonpower goods and services agreements subject to review under sections 205 and 206 of the Federal Power Act.
- (iii) A transaction by a holding company subject to the conditions in paragraphs (c)(5)(i) and (ii) of this section will be deemed approved only upon filing the information required in paragraphs (c)(5)(i) and (ii) of this section.
- (6) Any public utility or any holding company in a holding company system that includes a transmitting utility or an electric utility is granted a blanket authorization under sections 203(a)(1) or 203(a)(2) of the Federal Power Act, as relevant, for internal corporate reorganizations that do not result in the reorganization of a traditional public utility that has captive customers or that owns or provides transmission service over jurisdictional transmission facilities, and that do not present cross-subsidization issues.
- (7) Any public utility in a holding company system that includes a transmitting utility or an electric utility is

granted a blanket authorization under section 203(a)(1) of the Federal Power Act to purchase, acquire, or take any security of a public utility in connection with an intra-system cash management program, subject to safeguards to prevent cross-subsidization or pledges or encumbrances of utility assets.

- (8) A person that is a holding company solely with respect to one or more exempt wholesale generators (EWGs), foreign utility companies (FUCOs), or qualifying facilities (QFs) is granted a blanket authorization under section 203(a)(2) of the Federal Power Act to acquire the securities of additional EWGs, FUCOs, or QFs.
- (9) A holding company, or a subsidiary of that company, that is regulated by the Board of Governors of the Federal Reserve Bank or by the Office of the Comptroller of the Currency, under the Bank Holding Company Act of 1956 as amended by the Gramm-Leach-Bliley Act of 1999, is granted a blanket authorization under section 203(a)(2) of the Federal Power Act to acquire and hold an unlimited amount of the securities of holding companies that include a transmitting utility or an electric utility company if such acquisitions and holdings are in the normal course of its business and the securities are held:
 - (i) As a fiduciary;
- (ii) As principal for derivatives hedging purposes incidental to the business of banking and it commits not to vote such securities to the extent they exceed 10 percent of the outstanding shares;
 - (iii) As collateral for a loan; or
- (iv) Solely for purposes of liquidation and in connection with a loan previously contracted for and owned beneficially for a period of not more than two years, with the following conditions and reporting requirement: The holding does not confer a right to control, positively or negatively, through debt covenants or any other means, the operation or management of the public utility or public utility holding company, except as to customary creditors' rights or as provided under the United States Bankruptcy Code; and the parent holding company files with the Commission on a public basis and with-

in 45 days of the close of each calendar quarter, both its total holdings and its holdings as principal, each by class, unless the holdings within a class are less than one percent of outstanding shares, irrespective of the capacity in which they were held.

- (10) Any holding company, or a subsidiary of that company, is granted a blanket authorization under section 203(a)(2) of the Federal Power Act to acquire any security of a public utility or a holding company that includes a public utility:
- (i) For purposes of conducting underwriting activities, subject to the condition that holdings that the holding company or its subsidiary are unable to sell or otherwise dispose of within 45 days are to be treated as holdings as principal and thus subject to a limitation of 10 percent of the stock of any class unless the holding company or its subsidiary has within that period filed an application under section 203 of the Federal Power Act to retain the securities and has undertaken not to vote the securities during the pendency of such application; and the parent holding company files with the Commission on a public basis and within 45 days of the close of each calendar quarter, both its total holdings and its holdings as principal, each by class, unless the holdings within a class are less than one percent of outstanding shares, irrespective of the capacity in which they were held;
- (ii) For purposes of engaging in hedging transactions, subject to the condition that if such holdings are 10 percent or more of the voting securities of a given class, the holding company or its subsidiary shall not vote such holdings to the extent that they are 10 percent or more.
- (11) Any public utility is granted a blanket authorization under section 203(a)(1) of the Federal Power Act to transfer a wholesale market-based rate contract to any other public utility affiliate that has the same ultimate upstream ownership, provided that neither affiliate is affiliated with a traditional public utility with captive customers.
- (12) A public utility is granted a blanket authorization under section 203(a)(1) of the Federal Power Act to

§ 33.2

transfer its outstanding voting securities to:

- (i) Any holding company granted blanket authorizations in paragraph (c)(2)(ii) of this section if, after the transfer, the holding company and any of its associate or affiliate companies in aggregate will own less than 10 percent of the outstanding voting interests of such public utility; or
- (ii) Any person other than a holding company if, after the transfer, such person and any of its associate or affiliate companies in aggregate will own less than 10 percent of the outstanding voting interests of such public utility, and within 30 days after the end of the calendar quarter in which such transfer has occurred the public utility notifies the Commission in accordance with paragraph (c)(17) of this section.
- (13) A public utility is granted a blanket authorization under section 203(a)(1) of the Federal Power Act to transfer its outstanding voting securities to any holding company granted blanket authorization in paragraph (c)(8) of this section if, after the transfer, the holding company and any of its associate or affiliate companies in aggregate will own less than 10 percent of the outstanding voting interests of such public utility.
- (14) A public utility is granted a blanket authorization under section 203(a)(1) of the Federal Power Act to transfer its outstanding voting securities to any holding company granted blanket authorization in paragraph (c)(9) of this section.
- (15) A public utility is granted a blanket authorization under section 203(a)(1) of the Federal Power Act to transfer its outstanding voting securities to any holding company granted blanket authorization in paragraph (c)(10) of this section.
- (16) A public utility is granted a blanket authorization under section 203(a)(1) of the Federal Power Act for the acquisition or disposition of a jurisdictional contract where neither the acquirer nor transferor has captive customers or owns or provides transmission service over jurisdictional transmission facilities, the contract does not convey control over the operation of a generation or transmission

facility, and the acquirer is a public utility.

- (17) A public utility granted blanket authorization under paragraph (c)(12)(ii) of this section to transfer its outstanding voting securities shall, within 30 days after the end of the calendar quarter in which such transfer has occurred, file with the Commission a report containing the following information:
- (i) The names of all parties to the transaction;
- (ii) Identification of the pre- and post-transaction voting security holdings (and percentage ownership) in the public utility held by the acquirer and its associate or affilate companies;
- (iii) The date the transaction was consummated:
- (iv) Identification of any public utility or holding company affiliates of the parties to the transaction; and
- (v) A statement indicating that the proposed transaction will not result in, at the time of the transaction or in the future, cross-subsidization of a non-utility associate company or pledge or encumbrance of utility assets for the benefit of an associate company as required in §33.2(j)(1).

[Order 669-A, 71 FR 28443, May 16, 2006, as amended by Order 708, 73 FR 11013, Feb. 29, 2008; Order 708-A, 73 FR 43072, July 24, 2008; Order 708-B, 74 FR 25413, May 28, 2009]

§ 33.2 Contents of application—general information requirements.

Each applicant must include in its application, in the manner and form and in the order indicated, the following general information with respect to the applicant and each entity whose jurisdictional facilities or securities are involved:

- (a) The exact name of the applicant and its principal business address.
- (b) The name and address of the person authorized to receive notices and communications regarding the application, including phone and fax numbers, and E-mail addresses.
- (c) A description of the applicant, including:
- (1) All business activities of the applicant, including authorizations by charter or regulatory approval (to be identified as Exhibit A to the application):

- (2) A list of all energy subsidiaries and energy affiliates, percentage ownership interest in such subsidiaries and affiliates, and a description of the primary business in which each energy subsidiary and affiliate is engaged (to be identified as Exhibit B to the application):
- (3) Organizational charts depicting the applicant's current and proposed post-transaction corporate structures (including any pending authorized but not implemented changes) indicating all parent companies, energy subsidiaries and energy affiliates unless the applicant demonstrates that the proposed transaction does not affect the transaction (to be identified as Exhibit C to the application);
- (4) A description of all joint ventures, strategic alliances, tolling arrangements or other business arrangements, including transfers of operational control of transmission facilities to Commission approved Regional Transmission Organizations, both current, and planned to occur within a year from the date of filing, to which the applicant or its parent companies, energy subsidiaries, and energy affiliates is a party, unless the applicant demonstrates that the proposed transaction does not affect any of its business interests (to be identified as Exhibit D to the application);
- (5) The identity of common officers or directors of parties to the proposed transaction (to be identified as Exhibit E to the application); and
- (6) A description and location of wholesale power sales customers and unbundled transmission services customers served by the applicant or its parent companies, subsidiaries, affiliates and associate companies (to be identified as Exhibit F to the application).
- (d) A description of jurisdictional facilities owned, operated, or controlled by the applicant or its parent companies, subsidiaries, affiliates, and associate companies (to be identified as Exhibit G to the application).
- (e) A narrative description of the proposed transaction for which Commission authorization is requested, including:

- (1) The identity of all parties involved in the transaction;
- (2) All jurisdictional facilities and securities associated with or affected by the transaction (to be identified as Exhibit H to the application);
- (3) The consideration for the transaction; and
- (4) The effect of the transaction on such jurisdictional facilities and securities
- (f) All contracts related to the proposed transaction together with copies of all other written instruments entered into or proposed to be entered into by the parties to the transaction (to be identified as Exhibit I to the application).
- (g) A statement explaining the facts relied upon to demonstrate that the proposed transaction is consistent with the public interest. The applicant must include a general explanation of the effect of the transaction on competition, rates and regulation of the applicant by the Commission and state commissions with jurisdiction over any party to the transaction. The applicant should also file any other information it believes relevant to the Commission's consideration of the transaction. The applicant must supplement its application promptly to reflect in its analysis material changes that occur after the date a filing is made with the Commission, but before final Commission action. Such changes must be described and their effect on the analysis explained (to be identified as Exhibit J to the application).
- (h) If the proposed transaction involves physical property of any party, the applicant must provide a general or key map showing in different colors the properties of each party to the transaction (to be identified as Exhibit K to the application).
- (i) If the applicant is required to obtain licenses, orders, or other approvals from other regulatory bodies in connection with the proposed transaction, the applicant must identify the regulatory bodies and indicate the status of other regulatory actions, and provide a copy of each order of those regulatory bodies that relates to the proposed transaction (to be identified as Exhibit L to the application). If the regulatory bodies issue orders pertaining to the

§ 33.3

proposed transaction after the date of filing with the Commission, and before the date of final Commission action, the applicant must supplement its Commission application promptly with a copy of these orders.

- (j) An explanation, with appropriate evidentiary support for such explanation (to be identified as Exhibit M to this application):
- (1) Of how applicants are providing assurance, based on facts and circumstances known to them or that are reasonably foreseeable, that the proposed transaction will not result in, at the time of the transaction or in the future, cross-subsidization of a non-utility associate company or pledge or encumbrance of utility assets for the benefit of an associate company, including:
- (i) Disclosure of existing pledges and/ or encumbrances of utility assets; and (ii) A detailed showing that the
- transaction will not result in:
- (A) Any transfer of facilities between a traditional public utility associate company that has captive customers or that owns or provides transmission service over jurisdictional transmission facilities, and an associate company;
- (B) Any new issuance of securities by a traditional public utility associate company that has captive customers or that owns or provides transmission service over jurisdictional transmission facilities, for the benefit of an associate company;
- (C) Any new pledge or encumbrance of assets of a traditional public utility associate company that has captive customers or that owns or provides transmission service over jurisdictional transmission facilities, for the benefit of an associate company; or
- (D) Any new affiliate contract between a non-utility associate company and a traditional public utility associate company that has captive customers or that owns or provides transmission service over jurisdictional transmission facilities, other than nonpower goods and services agreements subject to review under sections 205 and 206 of the Federal Power Act; or
- (2) If no such assurance can be provided, an explanation of how such cross-subsidization, pledge, or encum-

brance will be consistent with the public interest.

[Order 642, 65 FR 71014, Nov. 28, 2000, as amended by Order 669–A, 71 FR 28446, May 16, 2006; Order 669–B, 71 FR 42586, July 27, 2006; Order 659–B, 71 FR 45736, Aug. 10, 2006]

§ 33.3 Additional information requirements for applications involving horizontal competitive impacts.

- (a)(1) The applicant must file the horizontal Competitive Analysis Screen described in paragraphs (b) through (f) of this section if, as a result of the proposed transaction, a single corporate entity obtains ownership or control over the generating facilities of previously unaffiliated merging entities (for purposes of this section, merging entities means any party to the proposed transaction or its parent companies, energy subsidiaries or energy affiliates).
- (2) A horizontal Competitive Analysis Screen need not be filed if the appli-
- (i) Affirmatively demonstrates that the merging entities do not currently conduct business in the same geographic markets or that the extent of the business transactions in the same geographic markets is *de minimis*; and
- (ii) No intervenor has alleged that one of the merging entities is a perceived potential competitor in the same geographic market as the other.
- (b) All data, assumptions, techniques and conclusions in the horizontal Competitive Analysis Screen must be accompanied by appropriate documentation and support.
- (1) If the applicant is unable to provide any specific data required in this section, it must identify and explain how the data requirement was satisfied and the suitability of the substitute data.
- (2) The applicant may provide other analyses for defining relevant markets (e.g. the Hypothetical Monopolist Test with or without the assumption of price discrimination) in addition to the delivered price test under the horizontal Competitive Analysis Screen.
- (3) The applicant may use a computer model to complete one or more steps in the horizontal Competitive Analysis Screen. The applicant must fully explain, justify and document any model

used and provide descriptions of model formulation, mathematical specifications, solution algorithms, as well as the annotated model code in executable form, and specify the software needed to execute the model. The applicant must explain and document how inputs were developed, the assumptions underlying such inputs and any adjustments made to published data that are used as inputs. The applicant must also explain how it tested the predictive value of the model, for example, using historical data.

- (c) The horizontal Competitive Analysis Screen must be completed using the following steps:
- (1) Define relevant products. Identify and define all wholesale electricity products sold by the merging entities during the two years prior to the date of the application, including, but not limited to, non-firm energy, short-term capacity (or firm energy), long-term capacity (a contractual commitment of more than one year), and ancillary services (specifically spinning reserves, non-spinning reserves, and imbalance energy, identified and defined separately). Because demand and supply conditions for a product can vary substantially over the year, periods corresponding to those distinct conditions must be identified by load level, and analyzed as separate products.
- (2) Identify destination markets. Identify each wholesale power sales customer or set of customers (destination market) affected by the proposed transaction. Affected customers are, at a minimum, those entities directly interconnected to any of the merging entities and entities that have purchased electricity at wholesale from any of the merging entities during the two years prior to the date of the application. If the applicant does not identify an entity to whom the merging entities have sold electricity during the last two years as an affected customer, the applicant must provide a full explanation for each exclusion.
- (3) Identify potential suppliers. The applicant must identify potential suppliers to each destination market using the delivered price test described in paragraph (c)(4) of this section. A seller may be included in a geographic market to the extent that it can economi-

cally and physically deliver generation services to the destination market.

- (4) Perform delivered price test. For each destination market, the applicant must calculate the amount of relevant product a potential supplier could deliver to the destination market from owned or controlled capacity at a price, including applicable transmission prices, loss factors and ancillary services costs, that is no more than five (5) percent above the pretransaction market clearing price in the destination market.
- (i) Supplier's presence. The applicant must measure each potential supplier's presence in the destination market in terms of generating capacity, using economic capacity and available economic capacity measures. Additional adjustments to supplier presence may be presented; applicants must support any such adjustment.
- (A) Economic capacity means the amount of generating capacity owned or controlled by a potential supplier with variable costs low enough that energy from such capacity could be economically delivered to the destination market. Prior to applying the delivered price test, the generating capacity meeting this definition must be adjusted by subtracting capacity committed under long-term firm sales contracts and adding capacity acquired under long-term firm purchase contracts (i.e., contracts with a remaining commitment of more than one year). The capacity associated with any such adjustments must be attributed to the party that has authority to decide when generating resources are available for operation. Other generating capacity may also be attributed to another supplier based on operational control criteria as deemed necessary, but the applicant must explain the reasons for doing so.
- (B) Available economic capacity means the amount of generating capacity meeting the definition of economic capacity less the amount of generating capacity needed to serve the potential supplier's native load commitments, as described in paragraph (d)(4)(i) of this section.
- (C) Available transmission capacity. Each potential supplier's economic capacity and available economic capacity

§ 33.3

(and any other measure used to determine the amount of relevant product that could be delivered to a destination market) must be adjusted to reflect available transmission capability to deliver each relevant product. The allocation to a potential supplier of limited capability of constrained transmission paths internal to the merging entities' systems or interconnecting the systems with other control areas must recognize both the transmission capability not subject to firm reservations by others and any firm transmission rights held by the potential supplier that are not committed to long-term transactions. For each such instance where limited transmission capability must be allocated among potential suppliers, the applicant must explain the method used and show the results of such allocation.

- (D) Internal interface. If the proposed transaction would cause an interface that interconnects the transmission systems of the merging entities to become transmission facilities for which the merging entities would have a "native load" priority under their open access transmission tariff (i.e., where the merging entities may reserve existing transmission capacity needed for native load growth and network transmission customer load growth reasonable forecasted within the utility's current planning horizon), all of the unreserved capability of the interface must be allocated to the merging entities for purposes of the horizontal Competitive Analysis Screen, unless the applicant demonstrates one of the following:
- (1) The merging entities would not have adequate economic capacity to fully use such unreserved transmission capability;
- (2) The merging entities have committed a portion of the interface capability to third parties; or
- (3) Suppliers other than the merging entities have purchased a portion of the interface capability.
 - (ii) [Reserved]
- (5) Calculate market concentration. The applicant must calculate the market share, both pre- and post-merger, for each potential supplier, the Herfindahl-Hirschman Index (HHI) statistic for the market, and the change in the HHI statistic. (The HHI statistic is a measure

of market concentration and is a function of the number of firms in a market and their respective market shares. The HHI statistic is calculated by summing the squares of the individual market shares, expressed as percentages, of all potential suppliers to the destination market.) To make these calculations, the applicant must use the amounts of generating capacity (i.e., economic capacity and available economic capacity, and any other relevant measure) determined in paragraph (c)(4)(i) of this section, for each product in each destination market.

- (6) Provide historical transaction data. The applicant must provide historical trade data and historical transmission data to corroborate the results of the horizontal Competitive Analysis Screen. The data must cover the twoyear period preceding the filing of the application. The applicant may adjust the results of the horizontal Competitive Analysis Screen, if supported by historical trade data or historical transmission service data. Any adjusted results must be shown separately, along with an explanation of all adjustments to the results of the horizontal Competitive Analysis Screen. The applicant must also provide an explanation of any significant differences between results obtained by the horizontal Competitive Analysis Screen and trade patterns in the last two
- (d) In support of the delivered price test required by paragraph (c)(4) of this section, the applicant must provide the following data and information used in calculating the economic capacity and available economic capacity that a potential supplier could deliver to a destination market. The transmission data required by paragraphs (d)(7) through (d)(9) of this section must be supplied for the merging entities' systems. The transmission data must also be supplied for other relevant systems, to the extent data are publicly available.
- (1) Generation capacity. For each generating plant or unit owned or controlled by each potential supplier, the applicant must provide:
 - (i) Supplier name;
 - (ii) Name of the plant or unit;

- (iii) Primary and secondary fueltypes:
 - (iv) Nameplate capacity;
- (v) Summer and winter total capacity; and
- (vi) Summer and winter capacity adjusted to reflect planned and forced outages and other factors, such as fuel supply and environmental restrictions.
- (2) Variable cost. For each generating plant or unit owned or controlled by each potential supplier, the applicant must also provide variable cost components.
- (i) These cost components must include at a minimum:
- (A) Variable operation and maintenance, including both fuel and non-fuel operation and maintenance; and
 - (B) Environmental compliance
- (ii) To the extent costs described in paragraph (d)(2)(i) of this section are allocated among units at the same plant, allocation methods must be fully described.
- (3) Long-term purchase and sales data. For each sale and purchase of capacity, the applicant must provide the following information:
 - (i) Purchasing entity name;
 - (ii) Selling entity name;
 - (iii) Duration of the contract;
- (iv) Remaining contract term and any evergreen provisions;
- (v) Provisions regarding renewal of the contract;
- (vi) Priority or degree of interruptibility;
- (vii) FERC rate schedule number, if applicable;
- (viii) Quantity and price of capacity and/or energy purchased or sold under the contract; and
- (ix) Information on provisions of contracts which confer operational control over generation resources to the purchaser.
- (4) Native load commitments. (i) Native load commitments are commitments to serve wholesale and retail power customers on whose behalf the potential supplier, by statute, franchise, regulatory requirement, or contract, has undertaken an obligation to construct and operate its system to meet their reliable electricity needs.
- (ii) The applicant must provide supplier name and hourly native load commitments for the most recent two

- years. In addition, the applicant must provide this information for each load level, if load-differentiated relevant products are analyzed.
- (iii) If data on native load commitments are not available, the applicant must fully explain and justify any estimates of these commitments.
- (5) Transmission and ancillary service prices, and loss factors. (i) The applicant must use in the horizontal Competitive Analysis Screen the maximum rates stated in the transmission providers' tariffs. If necessary, those rates should be converted to a dollars-per-megawatt hour basis and the conversion method explained.
- (ii) If a regional transmission pricing regime is in effect that departs from system-specific transmission rates, the horizontal Competitive Analysis Screen must reflect the regional pricing regime.
- (iii) The following data must be provided for each transmission system that would be used to deliver energy from each potential supplier to a destination market:
 - (A) Supplier name;
 - (B) Name of transmission system;
 - (C) Firm point-to-point rate;
- (D) Non-firm point-to-point rate;
- (E) Scheduling, system control and dispatch rate;
- (F) Reactive power/voltage control rate;
- (G) Transmission loss factor; and
- (H) Estimated cost of supplying energy losses.
- (iv) The applicant may present additional alternative analysis using discount prices if the applicant can support it with evidence that discounting is and will be available.
- (6) Destination market price. The applicant must provide, for each relevant product and destination market, market prices for the most recent two years. The applicant may provide suitable proxies for market prices if actual market prices are unavailable. Estimated prices or price ranges must be supported and the data and approach used to estimate the prices must be included with the application. If the applicant relies on price ranges in the analysis, such ranges must be reconciled with any actual market prices that are supplied in the application.

§ 33.3

Applicants must demonstrate that the results of the analysis do not vary significantly in response to small variations in actual and/or estimated prices.

- (7) Transmission capability. (i) The applicant must provide simultaneous transfer capability data, if available, for each of the transmission paths, interfaces, or other facilities used by suppliers to deliver to the destination markets on an hourly basis for the most recent two years.
- (ii) Transmission capability data must include the following information:
- (A) Transmission path, interface, or facility name;
- (B) Total transfer capability (TTC); and
- (C) Firm available transmission capability (ATC).
- (iii) Any estimated transmission capability must be supported and the data and approach used to make the estimates must be included with the application.
- (8) Transmission constraints. (i) For each existing transmission facility that affects supplies to the destination markets and that has been constrained during the most recent two years or is expected to be constrained within the planning horizon, the applicant must provide the following information:
- (A) Name of all paths, interfaces, or facilities affected by the constraint;
- (B) Locations of the constraint and all paths, interfaces, or facilities affected by the constraint;
- (C) Hours of the year when the transmission constraint is binding; and
- (D) The system conditions under which the constraint is binding.
- (ii) The applicant must include information regarding expected changes in loadings on transmission facilities due to the proposed transaction and the consequent effect on transfer capability.
- (iii) To the extent possible, the applicant must provide system maps showing the location of transmission facilities where binding constraints have been known or are expected to occur.
- (9) Firm transmission rights (Physical and Financial). For each potential supplier to a destination market that holds firm transmission rights nec-

essary to directly or indirectly deliver energy to that market, or that holds transmission congestion contracts, the applicant must provide the following information:

- (i) Supplier name;
- (ii) Name of transmission path interface, or facility;
- (iii) The FERC rate schedule number, if applicable, under which transmission service is provided; and
- (iv) A description of the firm transmission rights held (including, at a minimum, quantity and remaining time the rights will be held, and any relevant time restrictions on transmission use, such as peak or off-peak rights).
- (10) Summary table of potential suppliers' presence. (i) The applicant must provide a summary table with the following information for each potential supplier for each destination market:
 - (A) Potential supplier name;
- (B) The potential supplier's total amount of economic capacity (not subject to transmission constraints); and
- (C) The potential supplier's amount of economic capacity from which energy can be delivered to the destination market (after adjusting for transmission availability).
- (ii) A similar table must be provided for available economic capacity, and for any other generating capacity measure used by the applicant.
- (11) Historical trade data. (i) The applicant must provide data identifying all of the merging entities' wholesale sales and purchases of electric energy for the most recent two years.
- (ii) The applicant must include the following information for each transition:
- (A) Type of transaction (such as non-firm, short-term firm, long-term firm, peak, off-peak, etc.);
 - (B) Name of purchaser;
 - (C) Name of seller;
- (D) Date, duration and time period of the transaction;
- (E) Quantity of energy purchased or sold;
- (F) Energy charge per unit;
- (G) Megawatt hours purchased or sold:
 - (H) Price; and
- (I) The delivery points used to effect the sale or purchase.

- (12) Historical transmission data. The applicant must provide information concerning any transmission service denials, interruptions and curtailments on the merging entities' systems, for the most recent two years, to the extent the information is available from OASIS data, including the following information:
- (i) Name of the customer denied, interrupted or curtailed;
- (ii) Type, quantity and duration of service at issue;
- (iii) The date and period of time involved:
- (iv) Reason given for the denial, interruption or curtailment;
 - (v) The transmission path; and
- (vi) The reservations or other use anticipated on the affected transmission path at the time of the service denial, curtailment or interruption.
- (e) Mitigation. Any mitigation measures proposed by the applicant (including, for example, divestiture or participation in a regional transmission organization) which are intended to mitigate the adverse effect of the proposed transaction must, to the extent possible, be factored into the horizontal Competitive Analysis Screen as an additional post-transaction analysis. Any mitigation commitments that involve facilities (e.g., in connection with divestiture of generation) must identify the facilities affected by the commitment, along with a timetable for implementing the commitments.
- (f) Additional factors. If the applicant does not propose mitigation, the applicant must address:
- (1) The potential adverse competitive effects of the transaction.
- (2) The potential for entry in the market and the role that entry could play in mitigating adverse competitive effects of the transaction;
- (3) The efficiency gains that reasonably could not be achieved by other means; and
- (4) Whether, but for the transaction, one or more of the merging entities would be likely to fail, causing its assets to exit the market.

[65 FR 71014, Nov. 28, 2000; 65 FR 76005, Dec. 5, 2000]

§ 33.4 Additional information requirements for applications involving vertical competitive impacts.

- (a)(1) The applicant must file the vertical Competitive Analysis described in paragraphs (b) through (e) of this section if, as a result of the proposed transaction, a single corporate entity has ownership or control over one or more merging entities that provides inputs to electricity products and one or more merging entities that provides electric generation products (for purposes of this section, merging entities means any party to the proposed transaction or its parent companies, energy subsidiaries or energy affiliates).
- (2) A vertical Competitive Analysis need not be filed if the applicant can affirmatively demonstrate that:
- (i) The merging entities currently do not provide inputs to electricity products (i.e., upstream relevant products) and electricity products (i.e., downstream relevant products) in the same geographic markets or that the extent of the business transactions in the same geographic market is de minimis; and no intervenor has alleged that one of the merging entities is a perceived potential competitor in the same geographic market as the other.
- (ii) The extent of the upstream relevant products currently provided by the merging entities is used to produce a *de minimis* amount of the relevant downstream products in the relevant destination markets, as defined in paragraph (c)(2) of §33.3.
- (b) All data, assumptions, techniques and conclusions in the vertical Competitive Analysis must be accompanied by appropriate documentation and support.
- (c) The vertical Competitive Analysis must be completed using the following steps:
- (1) Define relevant products—(i) Downstream relevant products. The applicant must identify and define as downstream relevant products all products sold by merging entities in relevant downstream geographic markets, as outlined in paragraph (c)(1) of §33.3.
- (ii) Upstream relevant products. The applicant must identify and define as upstream relevant products all inputs

§ 33.5

to electricity products provided by upstream merging entities in the most recent two years.

- (2) Define geographic markets—(i) Downstream geographic markets. The applicant must identify all geographic markets in which it or any merging entities sell the downstream relevant products, as outlined in paragraphs (c)(2) and (c)(3) of §33.3.
- (ii) Upstream geographic markets The applicant must identify all geographic markets in which it or any merging entities provide the upstream relevant products.
- (3) Analyze competitive conditions—(i) Downstream geographic market. (A) The applicant must compute market share for each supplier in each relevant downstream geographic market and the HHI statistic for the downstream market. The applicant must provide a summary table with the following information for each relevant downstream geographic market:
- (1) The economic capacity of each downstream supplier (specify the amount of such capacity served by each upstream supplier);
- (2) The total amount of economic capacity in the downstream market served by each upstream supplier;
- (3) The market share of economic capacity served by each upstream supplier; and
- (4) The HHI statistic for the down-stream market.
- (B) A similar table must be provided for available economic capacity and for any other measure used by the applicant.
- (ii) Upstream geographic market. The applicant must provide a summary table with the following information for each upstream relevant product in each relevant upstream geographic market:
- (A) The amount of relevant product provided by each upstream supplier;
- (B) The total amount of relevant product in the market;
- (C) The market share of each upstream supplier; and
- (D) The HHI statistic for the upstream market.
- (d) Mitigation. Any mitigation measures proposed by the applicant (including, for example, divestiture or participation in an Regional Transmission Or-

ganization) which are intended to mitigate the adverse effect of the proposed transaction must, to the extent possible, be factored into the vertical competitive analysis as an additional post-transaction analysis. Any mitigation measures that involve facilities must identify the facilities affected by the commitment.

- (e) Additional factors. (1) If the applicant does not propose mitigation measures, the applicant must address:
- (i) The potential adverse competitive effects of the transaction.
- (ii) The potential for entry in the market and the role that entry could play in mitigating adverse competitive effects of the transaction;
- (iii) The efficiency gains that reasonably could not be achieved by other means; and
- (iv) Whether, but for the proposed transaction, one or more of the parties to the transaction would be likely to fail, causing its assets to exit the market.
- (2) The applicant must address each of the additional factors in the context of whether the proposed transaction is likely to present concerns about raising rivals' costs or anticompetitive coordination.

§ 33.5 Proposed accounting entries.

If the applicant is required to maintain its books of account in accordance with the Commission's Uniform System of Accounts in part 101 of this chapter, the applicant must present proposed accounting entries showing the effect of the transaction with sufficient detail to indicate the effects on account balances (including amounts transferred on an interim basis), the effect on the income statement, and the effects on other relevant financial statements. The applicant must also explain how the amount of each entry was determined.

§33.7 Verification.

The original application must be signed by a person or persons having authority with respect thereto and having knowledge of the matters therein set forth, and must be verified under oath.

§ 33.8 Number of copies.

The applicant must submit the application or petition to the Secretary of the Commission in accordance with filing procedures posted on the Commission's Web site at http://www.ferc.gov. If the applicant submits a public and a non-public version (containing information filed under a request for privileged treatment), the original and at least three of the eight copies must be of the non-public version of the filing, pursuant to §388.112(b)(ii). If the applicant must submit information specified in paragraphs (b), (c), (d), (e) and (f) of §33.3 or paragraphs (b), (c), (d) and (e) of §33.4, the applicant must submit all such information in electronic format (e.g., on computer diskette or on CD) along with a printed description and summary. The electronic version must be submitted in accordance with §385.2011 of the Commission's regulations. The printed portion of the applicant's submission must include documentation for the electronic submission, including all file names and a summary of the data contained in each file. Each column (or data item) in each separate data table or chart must be clearly labeled in accordance with the requirements of §33.3 and §33.4. Any units of measurement associated with numeric entries must also be included.

[Order 642, 65 FR 71014, Nov. 28, 2000, as amended by Order 737, 75 FR 43403, July 26, 2010]

§33.9 Protective order.

If the applicant seeks to protect any portion of the application, or any attachment thereto, from public disclosure pursuant to §388.112 of this chapter, the applicant must include with its request for privileged treatment a proposed protective order under which the parties to the proceeding will be able to review any of the data, information, analysis or other documentation relied upon by the applicant for which privileged treatment is sought.

§33.10 Additional information.

The Director of the Office of Energy Market Regulation, or his designee, may, by letter, require the applicant to submit additional information as is needed for analysis of an application filed under this part.

[Order 642, 65 FR 71014, Nov. 28, 2000, as amended by Order 699, 72 FR 45324, Aug. 14, 2007; Order 701, 72 FR 61053, Oct. 29, 2007]

§ 33.11 Commission procedures for the consideration of applications under section 203 of the FPA.

- (a) The Commission will act on a completed application for approval of a transaction (i.e., one that is consistent with the requirements of this part) not later than 180 days after the completed application is filed. If the Commission does not act within 180 days, such application shall be deemed granted unless the Commission finds, based on good cause, that further consideration is required to determine whether the proposed transaction meets the standards of section 203(a)(4) of the FPA and issues, by the 180th day, an order tolling the time for acting on the application for not more than 180 days, at the end of which additional period the Commission shall grant or deny the application.
- (b) The Commission will provide for the expeditious consideration of completed applications for the approval of transactions that are not contested, do not involve mergers, and are consistent with Commission precedent.
- (c) Transactions, provided that they are not contested, do not involve mergers and are consistent with Commission precedent, that will generally be subject to expedited review include:
- (1) A disposition of only transmission facilities, including, but not limited to, those that both before and after the transaction remain under the functional control of a Commission-approved regional transmission organization or independent system operator; and
- (2) Transactions that do not require an Appendix A analysis; $^{\rm 1}$ and
- (3) Internal corporate reorganizations that result in the reorganization of a

¹Inquiry Concerning the Commission's Merger Policy Under the Federal Power Act; Policy Statement, Order No. 592, 61 FR 68,595 (Dec. 30, 1996), FERC Stats. & Regs. ¶31,044 (1996), reconsideration denied, Order No. 592–A, 62 FR 33,340 (June 19, 1977), 79 FERC ¶61,321 (1997) (Merger Policy Statement).

Pt. 34

traditional public utility that has captive customers or owns or provides transmission service over jurisdictional transmission facilities, but do not present cross-subsidization issues.

[Order 669-A, 71 FR 28446, May 16, 2006]

PART 34—APPLICATION FOR AUTHORIZATION OF THE ISSUANCE OF SECURITIES OR THE ASSUMPTION OF LIABILITIES

Sec.

- 34.1 Applicability; definitions; exemptions in case of certain State regulation, certain short-term issuances and certain qualifying facilities.
- 34.2 Placement of securities.
- 34.3 Contents of application for issuance of securities.
- 34.4 Required exhibits.
- 34.5 Additional information.
- 34.6 Form and style.
- 34.7 Filing requirements.
- 34.8 Verification.
- 34.9 Reports.

AUTHORITY: 16 U.S.C. 791a–825r, 2601–2645; 31 U.S.C. 9701; 42 U.S.C. 7101–7352.

Source: Order 182, 46 FR 50514, Oct. 14, 1981, unless otherwise noted.

CROSS REFERENCES: For rules of practice and procedure, see part 385 of this chapter. For Approved Forms, Federal Power Act, see part 131 of this chapter.

OMB REFERENCE: "FERC Filing No. 523" is the identification number used by the Commission and the Office of Management and Budget to reference the filing requirements in part 34.

§ 34.1 Applicability; definitions; exemptions in case of certain State regulation, certain short-term issuances and certain qualifying facilities.

- (a) Applicability. This part applies to applications for authorization from the Commission to issue securities or assume an obligation or liability which are filed by:
- (1) Licensees and other entities pursuant to sections 19 and 20 of the Federal Power Act (41 Stat. 1073, 16 U.S.C. 812, 813) and part 20 of the Commission's regulations; and
- (2) Public utilities pursuant to section 204 of the Federal Power Act (49 Stat. 850, 16 U.S.C. 824e).
- (b) *Definitions*. For the purpose of this part:

- (1) The term *utility* means a licensee, public utility or other entity seeking authorization under sections 19, 20 or 204 of the Federal Power Act:
- (2) The term *securities* includes any note, stock, treasury stock, bond, or debenture or other evidence of interest in or indebtedness of a utility;
- (3) The term issuance or placement of securities means issuance or placement of securities, or assumption of obligation or liability; and
- (4) The term *State* means a State admitted to the Union, the District of Columbia, and any organized Territory of the United States.
- (c) Exemptions. (1) If an agency of the State in which the utility is organized and operating approves or authorizes, in writing, the issuance of securities prior to their issuance, the utility is exempt from the provisions of sections 19, 20 and 204 of the Federal Power Act and the regulations under this part, with respect to such securities.
- (2) This part does not apply to the issue or renewal of, or assumption of liability on, a note or draft maturing one year or less after the date of such issue, renewal, or assumption of liability, if the aggregate of such note or draft and all other then-outstanding notes and drafts of a maturity of one year or less on which the utility is primarily or secondarily liable, is not more than 5 percent of the par value of the other then-outstanding securities of the utility as of the date of issue or renewal of, or assumption of liability on, the note or draft. In the case of securities having no par value, the par value for the purpose of this part is the fair market value, as of the date of issue or renewal of, or assumption of liability on, the note or draft.
- (3) For certain qualifying facilities. Any cogeneration or small power production facility which is exempt from sections 19, 20 and 204 of the Federal Power Act pursuant to §292.601 of this chapter shall be exempt from the provisions of this part.

[Order 182, 46 FR 50514, Oct. 14, 1981, as amended at 48 FR 9851, Mar. 9, 1983; Order 575, 60 FR 4852, Jan. 25, 1995]

§34.2 Placement of securities.

- (a) Method of issuance. Upon obtaining authorization from the Commission, utilities may issue securities by either a competitive bid or negotiated placement, provided that:
- (1) Competitive bids are obtained from at least two prospective dealers, purchasers or underwriters; or
- (2) Negotiated offers are obtained from at least three prospective dealers, purchasers or underwriters; and
 - (3) The utility:
- (i) Accepts the bid or offer that provides the utility with the lowest cost of money for securities with fixed or variable interest or dividend rates, or
- (ii) Accepts the bid or offer that provides the utility with the greatest net proceeds for securities with no specified interest or dividend rates, or
- (iii) The utility has filed for and obtained authorization from the Commission to accept bids or offers other than those specified in paragraphs (a)(3)(i) or (a)(3)(ii) of this section.
- (b) Exemptions. The provisions of paragraph (a) of this section do not apply where:
- (1) The securities are to be issued to existing holders of securities on a prorata basis:
- (2) The utility receives an unsolicited offer to purchase the securities;
- (3) The securities have a maturity of one year or less; or
- (4) The securities are to be issued in support of or to guarantee securities issued by governmental or quasi-governmental bodies for the benefit of the utility.
- (c) *Prohibitions*. No securities will be placed with any person who:
- (1) Has performed any service or accepted any fee or compensation with respect to the proposed issuance of securities prior to submission of bids or entry into negotiations for placement of such securities; or
- (2) Would be in violation of section 305(a) of the Federal Power Act with respect to the issuance.

[Order 575, 60 FR 4853, Jan. 25, 1995]

§ 34.3 Contents of application for issuance of securities.

Each application to the Commission for authority to issue securities shall

- contain the information specified in this section. In lieu of filing the information required in paragraphs (e), (i) and (j) of this section, a specific reference may be made to the portion of the registration statement filed under §34.4(f), which includes the information required in these paragraphs.
- (a) The official name of the applicant and address of its principal business office
- (b) The State in which the utility is incorporated, the date of incorporation, and each State in which it operates.
- (c) The name, address and telephone number of a person within the utility authorized to receive notices and communications with respect to the application.
- (d) The date by which Commission action is requested.
- (e) A full description of the securities proposed to be issued, including:
 - (1) Type and nature of securities;
- (2) Amount of securities (par or stated value and number of units);
 - (3) Interest or dividend rate, if any;
 - (4) Dates of issuance and maturity;
- (5) Institutional rating of the securities—or if the securities are not rated, an explanation as to why they are not rated, and if the securities will be rated, an estimate of the rating; and
- (6) Any stock exchange on which the securities will be listed.
- (f) The purpose for which the securities for which application is made are to be issued:
- (1) If the purpose of such issuance is the construction, completion, extension, or improvement of facilities, describe in reasonable detail the construction program for which the funds were or are to be used.
- (2) If the purpose for such issuance is for the refunding of obligations, describe in detail the obligations to be refunded, including the character, principal amounts, applicable discount or premium, dates of issuance and maturity, and all other material facts concerning such obligations.
- (3) If the purpose for such issuance is for other than construction or refunding, explain such other purpose(s) in detail.
- (g) A statement as to whether or not any application with respect to the

§ 34.4

transaction or any part thereof is required to be filed with any State regulatory body.

- (h) A detailed statement of the facts relied upon by the applicant to show that the issuance:
- (1) Is for some lawful object, within the corporate purposes of the applicant and compatible with the public interest, is necessary or appropriate for or consistent with the proper performances by the applicant of service as a public utility and will not impair its ability to perform that service, and
- (2) Is reasonably necessary or appropriate for such purposes.
- (i) A detailed statement of the bond indenture(s) or other limitations on interest and dividend coverage, and the effects of such limitations on the issuance of additional debt or equity securities.
- (j) A brief summary of any rate changes which were made effective during the period for which financial statements are submitted or which became or will become effective after the period for which statements are submitted

[Order 182, 46 FR 50514, Oct. 14, 1981, as amended by Order 390, 49 FR 32505, Aug. 14, 1984; Order 575, 60 FR 4853, Jan. 25, 1995; Order 593, 62 FR 1283, Jan. 9, 1997; Order 647, 69 FR 32438, June 10, 2004; Order 737, 75 FR 43403, July 26, 2010]

§34.4 Required exhibits.

- (a) Exhibit A. The applicant must file the statement of corporate purposes from its articles of incorporation.
- (b) Exhibit B. A copy of all resolutions of the applicant's directors authorizing the issuance of securities for which the application is made; and copies of the resolution of the stockholders approving such issuance if approval of the stockholders has been obtained.
- (c) Exhibit C. The Balance Sheet and attached notes for the most recent 12-month period for which financial statements have been published, provided that the 12-month period ended no more than 4 months prior to the date of the filing of the application, on both an actual basis and a pro forma basis in the form prescribed for the "Comparative Balance Sheet" of FERC Form No. 1, "Annual Report for major electric

utilities, licensees and others." Each adjustment made in determining the pro forma basis must be clearly identified.

- (d) Exhibit D. The Income Statement and attached notes for the most recent 12-month period for which financial statements have been published, provided that the 12-month period ended no more than 4 months prior to the date of the filing of the application, on both an actual basis and a pro forma basis in the form prescribed for the "Statement of Income for the Year" of FERC Form No. 1, "Annual Report for major electric utilities, licensees and others." Each adjustment made in determining the pro forma basis must be clearly identified.
- (e) Exhibit E. A Statement of Cash Flows and Computation of Interest Coverage on an actual basis and a pro forma basis for the most recent 12month period for which financial statements have been published, provided that the 12-month period ended no more than 4 months prior to the date of the filing of the application. The Statement of Cash Flows must be in the form prescribed for the "Statement of Cash Flows" of the FERC Form No. 1, Annual Report for major electric utilities, licensees and others," followed by a computation of interest coverage, in the form of the following worksheet:

Federal Energy Regulatory Commission worksheet for computation of interest coverage	Actual for the year ended mm-dd-yy	OMB control No. 1902– 0043, pro forma for the year ended mm-dd-yy
Net income Add: Interest on Long-Term Debt, Interest on Short-Term Debt, Other Interest Expense, Total Interest Expense Federal and State Income Taxes Income Before Interest and Income Taxes		
Computation of Interest Coverage		
Income Before Interest and Income Taxes + Total Interest Expense = Interest Coverage		

(f) Exhibit F. A copy of registration statement and exhibits which are filed

with the Securities and Exchange Commission for the proposed security issuance.

[Order 182, 46 FR 50514, Oct. 14, 1981, as amended by Order 390, 49 FR 32505, Aug. 14, 1984; Order 575, 60 FR 4853, Jan. 25, 1995; 60 FR 27882, May 26, 1995]

§34.5 Additional information.

The Commission may, in its discretion, require the filing of additional information which appears necessary to reach a determination on any particular application.

§34.6 Form and style.

Each application pursuant to this part 34 shall conform to the requirements of subpart T of part 385 of this chapter.

[Order 182, 46 FR 50514, Oct. 14, 1981, as amended by Order 225, 47 FR 19056, May 3, 1982]

§34.7 Filing requirements.

Applications must be filed with the Secretary of the Commission in accordance with filing procedures posted on the Commission's Web site at http://www.ferc.gov.

[Order 737, 75 FR 43403, July 26, 2010]

§34.8 Verification.

The original application shall be signed by an authorized representative of the applicant, who has knowledge of the matters set forth therein, and it shall be verified under oath.

EFFECTIVE DATE NOTE: At 70 FR 35375, June 20, 2005, §34.8 was revised, effective at the time of the next e-filing release during the Commission's next fiscal year. For the convenience of the user, the revised text follows:

§ 34.8 Verification.

An application verification shall be signed under oath by an authorized representative of the applicant, who has knowledge of the matters set forth therein and as provided in \$385.2005 of this chapter, and retained at the applicant's business location until the relevant proceeding has been concluded.

§34.9 Reports.

The applicant must file reports under §131.43 and §131.50 of this chapter no later than 30 days after the sale or placement of long-term debt or equity securities or the entry into guarantees

or assumptions of liabilities pursuant to authority granted under this part.

[Order 575, 60 FR 4853, Jan. 25, 1995. Redesignated by Order 737, 75 FR 43403, July 26, 2010]

PART 35—FILING OF RATE SCHEDULES AND TARIFFS

Subpart A—Application

Sec.

- 35.1 Application; obligation to file rate schedules, tariffs and certain service agreements.
- 35.2 Definitions.
- 35.3 Notice requirements.
- 35.4 Permission to become effective is not approval.
- 35.5 Rejection of material submitted for filing.
- 35.6 Submission for staff suggestions.
- 35.7 Electronic filing requirements.
- 35.8 Protests and interventions by interested parties.
- 35.9 Requirements for filing rate schedules, tariffs or service agreements.
- tariffs or service agreements. 35.10 Form and style of rate schedules, tar-
- iffs and service agreements. 35.10a Forms of service agreements.
- 35.10b Electric Quarterly Reports.
- 35.11 Waiver of notice requirement.

Subpart B—Documents To Be Submitted With a Filing

- 35.12 Filing of initial rate schedules and tariffs.
- 35.13 Filing of changes in rate schedules, tariffs or service agreements.

Subpart C—Other Filing Requirements

- 35.14 Fuel cost and purchased economic power adjustment clauses.
- 35.15 Notices of cancellation or termination.
- 35.16 Notice of succession.
- 35.17 Withdrawals and amendments of rate schedule, tariff or service agreement filings.
- 35.18 Asset retirement obligations.
- 35.19 Submission of information by reference.
- 35.19a Refund requirements under suspension orders.
- 35.21 Applicability to licensees and others subject to section 19 or 20 of the Federal Power Act.
- 35.22 Limits for percentage adders in rates for transmission services; revision of rate schedules, tariffs or service agreements.
- 35.23 General provisions.
- 35.24 Tax normalization for public utilities.
- 35.25 Construction work in progress.
- 35.26 Recovery of stranded costs by public utilities and transmitting utilities.

- 35.27 Authority of State commissions.
- 35.28 Non-discriminatory open access transmission tariff.
- 35.29 Treatment of special assessments levied under the Atomic Energy Act of 1954, as amended by Title XI of the Energy Policy Act of 1992.

Subpart D—Procedures and Requirements for Public Utility Sales of Power to Bonneville Power Administration Under **Northwest Power Act**

- 35.30 General provisions.
- 35.31 Commission review.

Subpart E—Regulations Governing Nuclear Plant Decommissioning Trust Funds

- 35.32 General provisions.
- 35.33 Specific provisions.

Subpart F—Procedures and Requirements Regarding Regional Transmission Organizations

35.34 Regional Transmission Organizations.

Subpart G—Transmission Infrastructure **Investment Procedures**

35.35 Transmission infrasturcture invest-

Subpart H—Wholesale Sales of Electric Energy, Capacity and Ancillary Services at Market-Based Rates

- 35.36 Generally.
- 35.37 Market power analysis required.
- 35.38 Mitigation.
- Affiliate restrictions. 35.39
- 35.40 Ancillary services.
- 35.41 Market behavior rules.
- 35.42 Change in status reporting requirement.
- APPENDIX A TO SUBPART H STANDARD SCREEN FORMAT
- APPENDIX B TO SUBPART H CORPORATE ENTI-TIES AND ASSETS

Subpart I—Cross-Subsidization Restrictions on Affiliate Transactions

- 35.43 Generally.
- 35.44 Protections against affiliate cross-subsidization

Subpart J—Credit Practices In Organized **Wholesale Electric Markets**

- 35.45 Applicability. 35.46 Definitions.
- 35.47 Tariff provisions governing credit practices in organized wholesale electric markets.

AUTHORITY: 16 U.S.C. 791a-825r, 2601-2645; 31 U.S.C. 9701; 42 U.S.C. 7101-7352.

Source: Order 271, 28 FR 10573, Oct. 2, 1963, unless otherwise noted.

Subpart A—Application

§35.1 Application; obligation to file rate schedules, tariffs and certain service agreements.

(a) Every public utility shall file with the Commission and post, in conformity with the requirements of this part, full and complete rate schedules and tariffs and those service agreements not meeting the requirements of §35.1(g), clearly and specifically setting forth all rates and charges for any transmission or sale of electric energy subject to the jurisdiction of this Commission, the classifications, practices, rules and regulations affecting such rates, charges, classifications, services, rules, regulations or practices, as required by section 205(c) of the Federal Power Act (49 Stat. 851; 16 U.S.C. 824d(c)). Where two or more public utilities are parties to the same rate schedule or tariff, each public utility transmitting or selling electric energy subject to the jurisdiction of this Commission shall post and file such rate schedule, or the rate schedule may be filed by one such public utility and all other parties having an obligation to file may post and file a certificate of concurrence on the form indicated in §131.52 of this chapter: Provided, however. In cases where two or more public utilities are required to file rate schedules or certificates of concurrence such public utilities may authorize a designated representative to file upon behalf of all parties if upon written request such parties have been granted Commission authorization therefor.

- (b) A rate schedule, tariff, or service agreement applicable to a transmission or sale of electric energy, other than that which proposes to supersede, cancel or otherwise change the provisions of a rate schedule, tariff, or service agreement required to be on file with this Commission, shall be filed as an initial rate in accordance with §35.12.
- (c) A rate schedule, tariff, or service agreement applicable to a transmission or sale of electric energy which proposes to supersede, cancel or otherwise

change any of the provisions of a rate schedule, tariff, or service agreement required to be on file with this Commission (such as providing for other or additional rates, charges, classifications or services, or rules, regulations, practices or contracts for a particular customer or customers) shall be filed as a change in rate in accordance with §35.13, except cancellation or termination which shall be filed as a change in accordance with §35.15.

- (d)(1) The provisions of this paragraph (d) shall apply to rate schedules, tariffs or service agreements tendered for filing on or after August 1, 1976, which are applicable to the transmission or sale of firm power for resale to an all-requirements customer, whether tendered pursuant to §35.12 as an initial rate schedule or tendered pursuant to §35.13 as a change in an existing rate schedule whose term has expired or whose term is to be extended.
- (2) Rate schedules covered by the terms of paragraph (d)(1) of this section shall contain the following provision when it is the intent of the contracting parties to give the party furnishing service the unrestricted right to file unilateral rate changes under section 205 of the Federal Power Act:

Nothing contained herein shall be construed as affecting in any way the right of the party furnishing service under this rate schedule to unilaterally make application to the Federal Energy Regulatory Commission for a change in rates under section 205 of the Federal Power Act and pursuant to the Commission's Rules and Regulations promulgated thereunder.

(3) Rate schedules covered by the terms of paragraph (d)(1) of this section shall contain the following provision when it is the intent of the contracting parties to withhold from the party furnishing service the right to file any unilateral rate changes under section 205 of the Federal Power Act:

The rates for service specified herein shall remain in effect for the term of _____ or until _____, and shall not be subject to change through application to the Federal Energy Regulatory Commission pursuant to the provisions of Section 205 of the Federal Power Act absent the agreement of all parties thereto.

- (4) Rate schedules covered by the terms of paragraph (d)(1) of this section, but which are not covered by paragraphs (d)(2) or (d)(3) of this section, are not required to contain either of the boilerplate provisions set forth in paragraph (d)(2) or (d)(3) of this section
- (e) No public utility shall, directly or indirectly, demand, charge, collect or receive any rate, charge or compensation for or in connection with electric service subject to the jurisdiction of the Commission, or impose any classification, practice, rule, regulation or contract with respect thereto, which is different from that provided in a rate schedule required to be on file with this Commission unless otherwise specifically provided by order of the Commission for good cause shown.
- (f) A rate schedule applicable to the sale of electric power by a public utility to the Bonneville Power Administration under section 5(c) of the Pacific Northwest Electric Power Planning and Conservation Act (Pub. L. No. 96–501 (1980)) shall be filed in accordance with subpart D of this part.
- (g) For the purposes of paragraph (a) of this section, any service agreement that conforms to the form of service agreement that is part of the public utility's approved tariff pursuant to §35.10a of this chapter and any marketbased rate agreement pursuant to a tariff shall not be filed with the Commission. All agreements must, however, be retained and be made available for public inspection and copying at the public utility's business office during regular business hours and provided to the Commission or members of the public upon request. Any individually executed service agreement for transmission, cost-based power sales, or other generally applicable services that deviates in any material respect from the applicable form of service agreement contained in the public utility's tariff and all unexecuted agreements under which service will commence at the request of the customer,

are subject to the filing requirements of this part.

[Order 271, 28 FR 10573, Oct. 2, 1963, as amended by Order 541, 40 FR 56425, Dec. 3, 1975; Order 541—A, 41 FR 27831, July 7, 1976; 46 FR 50520, Oct. 14, 1981; Order 337, 48 FR 46976, Oct. 17, 1983; Order 541, 57 FR 21734, May 22, 1992; Order 2001, 67 FR 31069, May 8, 2002; Order 714, 73 FR 57530, 57533, Oct. 3, 2008; 74 FR 55770, Oct. 29, 2009]

§ 35.2 Definitions.

- (a) Electric service. The term electric service as used herein shall mean the transmission of electric energy in interstate commerce or the sale of electric energy at wholesale for resale in interstate commerce, and may be comprised of various classes of capacity and energy sales and/or transmission services. Electric service shall include the utilization of facilities owned or operated by any public utility to effect any of the foregoing sales or services whether by leasing or other arrangements. As defined herein, electric service is without regard to the form of payment or compensation for the sales or services rendered whether by purchase and sale, interchange, exchange, wheeling charge, facilities charge, rental or otherwise.
- (b) Rate schedule. The term rate schedule as used herein shall mean a statement of (1) electric service as defined in paragraph (a) of this section, (2) rates and charges for or in connection with that service, and (3) all classifications, practices, rules, or regulations which in any manner affect or relate to the aforementioned service, rates, and charges. This statement shall be in writing and may take the physical form of a contract, purchase or sale or other agreement, lease of facilities, or other writing. Any oral agreement or understanding forming a part of such statement shall be reduced to writing and made a part thereof. A rate schedule is designated with a Rate Schedule
- (c)(1) Tariff. The term tariff as used herein shall mean a statement of (1) electric service as defined in paragraph (a) of this section offered on a generally applicable basis, (2) rates and charges for or in connection with that service, and (3) all classifications, practices, rules, or regulations which in

- any manner affect or relate to the aforementioned service, rates, and charges. This statement shall be in writing. Any oral agreement or understanding forming a part of such statement shall be reduced to writing and made a part thereof. A tariff is designated with a Tariff Volume number.
- (2) Service agreement. The term service agreement as used herein shall mean an agreement that authorizes a customer to take electric service under the terms of a tariff. A service agreement shall be in writing. Any oral agreement or understanding forming a part of such statement shall be reduced to writing and made a part thereof. A service agreement is designated with a Service Agreement number.
- (d) Filing date. The term filing date as used herein shall mean the date on which a rate schedule, tariff or service greement filing is completed by the receipt in the office of the Secretary of all supporting cost and other data required to be filed in compliance with the requirements of this part, unless such rate schedule is rejected as provided in §35.5. If the material submitted is found to be incomplete, the Director of the Office of Energy Market Regulation will so notify the filing utility within 60 days of the receipt of the submittal.
- (e) *Posting* (1) The term posting as used in this part shall mean:
- (i) Keeping a copy of every rate schedule, service agreement, or tariff of a public utility as currently on file, or as tendered for filing, with the Commission open and available during regular business hours for public inspection in a convenient form and place at the public utility's principal and district or division offices in the territory served, and/or accessible in electronic format, and
- (ii) Serving each purchaser under a rate schedule, service agreement, or tariff either electronically or by mail in accordance with the service regulations in Part 385 of this chapter with a copy of the rate schedule, service agreement, or tariff. Posting shall include, in the event of the filing of increased rates or charges, serving either electronically or by mail in accordance with the service regulations in Part 385 of this chapter each purchaser under a

rate schedule, service agreement or tariff proposed to be changed and to each State Commission within whose jurisdiction such purchaser or purchasers distribute and sell electric energy at retail, a copy of the rate schedule, service agreement or tariff showing such increased rates or charges, comparative billing data as required under this part, and, if requested by a purchaser or State Commission, a copy of the supporting data required to be submitted to this Commission under this part. Upon direction of the Secretary, the public utility shall serve copies of rate schedules, service agreements, or tariffs, and supplementary data, upon designated parties other than those specified herein.

(2) Unless it seeks a waiver of electronic service, each customer, State Commission, or other party entitled to service under this paragraph (e) must notify the public utility of the e-mail address to which service should be directed. A customer, State Commission, or other party may seek a waiver of electronic service by filing a waiver request under Part 390 of this chapter providing good cause for its inability to accept electronic service.

(f) Effective date. As used herein the effective date of a rate schedule, tariff or service agreement shall mean the date on which a rate schedule filed and posted pursuant to the requirements of this part is permitted by the Commission to become effective as a filed rate schedule. The effective date shall be 60 days after the filing date, or such other date as may be specified by the Commission.

(16 U.S.C. 284(d), 792 et seq.; Pub. L. 95-617; Pub. L. 95-91; E.O. 12009, 42 FR 46267)

[Order 271, 28 FR 10573, Oct. 2, 1963, as amended at 28 FR 11404, Oct. 24, 1963; 43 FR 36437, Aug. 17, 1978; 44 FR 16372, Mar. 19, 1979; 44 FR 20077, Apr. 4, 1979; Order 39, 44 FR 46454, Aug. 8, 1979; Order 699, 72 FR 45325, Aug. 14, 2007; Order 701, 72 FR 61054, Oct. 29, 2007; Order 714, 73 FR 57530, Oct. 3, 2008]

§ 35.3 Notice requirements.

(a)(1) Rate schedules or tariffs. All rate schedules or tariffs or any part thereof shall be tendered for filing with the Commission and posted not less than sixty days nor more than one hundred-twenty days prior to the date on which

the electric service is to commence and become effective under an initial rate schedule or tariff or the date on which the filing party proposes to make any change in electric service and/or rate. charge, classification, practice, rule, regulation, or contract effective as a change in rate schedule or tariff, except as provided in paragraph (b) of this section, or unless a different period of time is permitted by the Commission. Nothing herein shall be construed as in any way precluding a public utility from entering into agreements which, under this section, may not be filed at the time of execution thereof by reason of the aforementioned sixty to one hundred-twenty day prior filing requirements. The proposed effective date of any rate schedule or tariff filing having a filing date in accordance with §35.2(d) may be deferred by the public utility making a filing requesting deferral prior to the rate schedule or tariff's acceptance by the Commission.

(2) Service agreements. Service agreements that are required to be filed and posted authorizing a customer to take electric service under the terms of a tariff, or any part thereof, shall be tendered for filing with the Commission and posted not more than 30 days after electric service has commenced or such other date as may be specified by the Commission.

(b) Construction of facilities. Rate schedules, tariffs or serviceagreements predicated on the construction of facilities may be tendered for filing and posted no more than one hundred-twenty days prior to the date set by the parties for the contract to go into effect. The Commission, upon request, may permit a rate schedule or service agreement or part thereof to be tendered for filing and posted more than one hundred-twenty days before it is to become effective.

(16 U.S.C. 284(d); Pub. L. 95–617; Pub. L. 95–91; E.O. 12009, 42 FR 46267)

 $[44~{\rm FR}~16372,~{\rm Mar.}~19,~1979;~44~{\rm FR}~20077,~{\rm Apr.}~4,~1979;~{\rm as~amended~by~Order}~714,~73~{\rm FR}~57531,~{\rm Oct.}~3,~2008]$

§ 35.4 Permission to become effective is not approval.

The fact that the Commission permits a rate schedule, tariff or service

agreement or any part thereof or any notice of cancellation to become effective shall not constitute approval by the Commission of such rate schedule or tariff or part thereof or notice of cancellation.

[Order 271, 28 FR 10573, Oct. 2, 1963, as amended by Order 714, 73 FR 57531, 57533, Oct. 3, 2008]

§ 35.5 Rejection of material submitted for filing.

- (a) The Secretary, pursuant to the Commission's rules of practice and procedure and delegation of Commission authority, shall reject any material submitted for filing with the Commission which patently fails to substantially comply with the applicable requirements set forth in this part, or the Commission's rules of practice and procedure.
- (b) A rate filing that fails to comply with this Part may be rejected by the Director of the Office of Energy Market Regulation pursuant to the authority delegated to the Director in § 375.307(a)(1)(ii) of this chapter.

[Order 271, 28 FR 10573, Oct. 2, 1963, as amended by Order 614, 65 FR 18227, Apr. 7, 2000; Order 699, 72 FR 45325, Aug. 14, 2007; Order 701, 72 FR 61054, Oct. 29, 2007]

§35.6 Submission for staff suggestions.

Any public utility may submit a rate schedule, tariff or service agreement or any part thereof or any material relating thereto for the purpose of receiving staff suggestions and comments thereon prior to filing with the Commission.

[Order 271, 28 FR 10573, Oct. 2, 1963, as amended by Order 714, 73 FR 57531, Oct. 3, 2008]

§35.7 Electronic filing requirements.

- (a) General rule. All filings made in proceedings initiated under this part must be made electronically, including tariffs, rate schedules and service agreements, or parts thereof, and material that relates to or bears upon such documents, such as cancellations, amendments, withdrawals, termination, or adoption of tariffs.
- (b) Requirement for signature. All filings must be signed in compliance with the following:
- (1) The signature on a filing constitutes a certification that: the con-

tents are true and correct to the best knowledge and belief of the signer; and that the signer possesses full power and authority to sign the filing.

- (2) A filing must be signed by one of the following:
- (i) The person on behalf of whom the filing is made;
- (ii) An officer, agent, or employee of the company, governmental authority, agency, or instrumentality on behalf of which the filing is made; or,
- (iii) A representative qualified to practice before the Commission under §385.2101 of this chapter who possesses authority to sign.
- (3) All signatures on the filing or any document included in the filing must comply, where applicable, with the requirements in Part 385 of this chapter with respect to sworn declarations or statements and electronic signatures.
- (c) Format requirements for electronic filing. The requirements and formats for electronic filing are listed in instructions for electronic filing and for each form. These formats are available on the Internet at http://www.ferc.gov and can be obtained at the Federal Energy Regulatory Commission, Public Reference Room, 888 First Street, NE., Washington, DC 20426.

[Order 714, 73 FR 57531, Oct. 3, 2008]

§ 35.8 Protests and interventions by interested parties.

Unless the notice issued by the Commission provides otherwise, any protest or intervention to a rate filing made pursuant to this part must be filed in accordance with §§ 385.211 and 385.214 of this chapter, on or before 21 days after the subject rate filing. A protest must state the basis for the objection. A protest will be considered by the Commission in determining the appropriate action to be taken, but will not serve to make the protestant a party to the proceeding. A person wishing to become a party to the proceeding must file a motion to intervene.

[Order 612, 64 FR 72537, Dec. 28, 1999; 65 FR 18229, Apr. 7, 2000, as amended by Order 647, 69 FR 32438, June 10, 2004; Order 714, 73 FR 57531, Oct. 3, 2008]

§ 35.9 Requirements for filing rate schedules, tariffs or service agreements.

(a) Rate schedules, tariffs, and service agreements may be filed either by dividing the rate schedule, tariff, or service agreements into individual sheets or sections, or as an entire document except as provided in paragraphs (b) and (c) of this section.

(b) Open Access Transmission Tariffs (OATT) filed by utilities that are not Independent System Operators or Regional Transmission Organizations must be filed either as individual sheets or sections. If filed as sections, the sections must be no larger than the 1.0 level, although each schedule or attachment may be a single section. Individual service agreements that are entered into pursuant to the OATT may be filed as entire documents.

(c) OATT and other open access documents filed by Independent System Operators or Regional Transmission Organizations must be filed either as individual sheets or sections. If filed as sections, the sections must be no larger than the 1.1 level, including schedules or attachments. Individual service agreements that are part entered into pursuant to the OATT may be filed as entire documents.

[Order 714, 73 FR 57531, Oct. 3, 2008]

§35.10 Form and style of rate schedules, tariffs and service agreements.

(a) Every rate schedule, tariff or service agreement offered for filing with the Commission under this part, shall show on a title page, which shall be otherwise blank, (1) the name of the filing public utility, (2) the names of other utilities rendering or receiving service under the rate schedule, tariff or service agreement; and (3) a brief description of the service to be provided under the rate schedule, tariff or service agreement.

(b) At the time a public utility files with the Commission and posts under this part to supersede or change the provisions of a rate schedule, tariff, or service agreement previously filed with the Commission under this part, in addition to the other requirements of this part, it must list in the transmittal letter the sheets or sections revised, and file a marked version of the rate

schedule, tariff or service agreement sheets or sections showing additions and deletions. New language must be marked by either highlighting, background shading, bold text, or underlined text. Deleted language must be marked by strike-through.

(c) In any filing to supersede or change the provisions of a rate schedule, tariff, or service agreement previously filed with the Commission under this part, only those revisions appropriately designated and marked under paragraph (b) of this section constitute the filing. Revisions to unmarked portions of the rate schedule, tariff or service agreement are not considered part of the filing nor will any acceptance of the filing by the Commission constitute acceptance of such unmarked changes.

[Order 271, 28 FR 10573, Oct. 2, 1963, as amended by Order 568, 59 FR 40240, Aug. 8, 1994; Order 714, 73 FR 57532, Oct. 3, 2008]

§35.10a Forms of service agreements.

(a) To the extent a public utility adopts a standard form of service agreement for a service other than market-based power sales, the public utility shall include as part of its applicable tariff(s) an unexecuted standard service agreement approved by the Commission for each category of generally applicable service offered by the public utility under its tariff(s). The standard format for each generally applicable service must reference the service to be rendered and where it is located in its tariff(s). The standard format must provide spaces for insertion of the name of the customer, effective date, expiration date, and term. Spaces may be provided for the insertion of receipt and delivery points, contract quantity, and other specifics of each transaction, as appropriate.

(b) Forms of service agreement submitted under this section shall be filed electronically as prescribed in §35.7 for the filing of rate schedules.

[Order 2001, 67 FR 31069, May 8, 2002; as amended by Order 714, 73 FR 57532, Oct. 3, 2008]

§35.10b Electric Quarterly Reports.

Each public utility shall file an updated Electric Quarterly Report with the Commission covering all services it

provides pursuant to this part, for each of the four calendar quarters of each year, in accordance with the following schedule: for the period from January 1 through March 31, file by April 30; for the period from April 1 through June 30, file by July 31; for the period July 1 through September 30, file by October 31; and for the period October 1 through December 31, file by January 31. Electric Quarterly Reports must be prepared in conformance with the Commission's software and guidance posted and available for downloading from the FERC Web site (http://www.ferc.gov).

[Order 2001, 67 FR 31069, May 8, 2002]

§35.11 Waiver of notice requirement.

Upon application and for good cause shown, the Commission may, by order, provide that a rate schedule, tariff, or service agreement, or part thereof, shall be effective as of a date prior to the date of filing or prior to the date the rate schedule or tariff would become effective in accordance with these rules. Application for waiver of the prior notice requirement shall show (a) how and the extent to which the filing public utility and purchaser(s) under such rate schedule or tariff, or part thereof, would be affected if the notice requirement is not waived, and (b) the effects of the waiver, if granted, upon purchasers under other rate schedules. The filing public utility requesting such waiver of notice shall serve copies of its request therefor upon all purchasers.

[Order 271, 28 FR 10573, Oct. 2, 1963, as amended by Order 714, 73 FR 57532, 57533, Oct. 3, 2008]

Subpart B—Documents To Be Submitted With a Filing

§35.12 Filing of initial rate schedules and tariffs.

(a) The letter of a public utility transmitting to the Commission for filing an initial rate schedule or tariff shall list the documents submitted with the filing; give the date on which the service under that rate schedule or tariff is expected to commence; state the names and addresses of those to whom the rate schedule or tariff has been mailed; contain a brief descrip-

tion of the kinds of services to be furnished at the rates specified therein; and summarize the circumstances which show that all requisite agreement to the rate schedule or tariff or the filing thereof, including any contract embodied therein, has in fact been obtained. In the case of coordination and interchange arrangements in the nature of power pooling transactions, all supporting data required to be submitted in support of a rate schedule or tariff filing shall also be submitted by parties filing certificates of concurrence, or a representative to file supporting data on behalf of all parties may be designated as provided in §35.1.

- (b) In addition, the following material shall be submitted:
- (1) Estimates of the transactions and revenues under an initial rate schedule. This shall include estimates, by months and for the year, of the quantities of services to be rendered and of the revenues to be derived therefrom during the 12 months immediately following the month in which those services will commence. Such estimates should be subdivided by classes of service, customers, and delivery points and shall show all billing determinants, e.g., kw, kwh, fuel adjustment, power factor adjustment. These estimates will not be required where they cannot be made with relative accuracy as, for example, in cases of interconnection arrangements containing schedules of rates for emergency energy, spinning reserve or economy energy or in cases of coordination and integration of hydroelectric generating resources whose output cannot be predicted quantitatively due to water conditions.

(2)(i) Basis of the rate or charge proposed in an initial rate schedule or tariff and an explanation of how the proposed rate or charge was derived. For example, is it a standard rate of the filing public utility; is it a special rate arrived at through negotiations and, if so, were unusual customer requirements or competitive factors involved; and is it designed to produce a return substantially equal to the filing public utility's overall rate of return or is it essentially an increment cost plus a share of the savings rate? Were special

cost of service studies prepared in connection with the derivation of the rate?

- (ii) A summary statement of all cost (whether fully distributed, incremental or other) computations involved in arriving at the derivation of the level of the rate, in sufficient detail to justify the rate, shall be submitted with the filing, except that if the filing includes nothing more than service to one or more added customers under an established rate of the utility for a particular class of service, such summary statement of cost computations is not required. In all cases, the Secretary is authorized to require the submission of the complete cost studies as part of the filing and each filing public utility shall submit the same upon request by the Secretary in such form as he or she shall direct.
- (3) A comparison of the proposed initial rate with other rates of the filing public utility for similar wholesale for resale and transmission services.
- (4) If any facilities are installed or modified in order to supply the service to be furnished under the proposed rate schedule or tariff, the filing public utility shall show on an appropriate available map (or sketch) and single line diagram the additions or changes to be made.
- (5) In support of the design of the proposed rate, the filing public utility shall submit the same material required to be furnished pursuant to §35.13(h)(37) Statement BL. In addition to the summary cost analysis required by Statement BL, the public utility shall also submit a complete explanation as to the method used in arriving at the cost of service allocated to the sales and service for which the rate or charge is proposed, and showing the principal determinants used for allocation purposes. In connection therewith, the following data should be submitted:
- (i) In the event the filing public utility considers certain special facilities as being devoted entirely to the service involved, it shall show the cost of service related to such special facilities.
- (ii) Computations showing the energy responsibility of the service, based upon considerations of energy sales under the proposed rate schedule or tariff and the kWh delivered from the filing public utility's supply system.

(iii) Computations showing the demand responsibility of the service, and explaining the considerations upon which such responsibility was determined (e.g., coincident or non-coincident peak demands, etc.).

(Federal Power Act, 16 U.S.C. 792-828c; Department of Energy Organization Act, 42 U.S.C. 7101–7352; E.O. 12009, 42 FR 46267; Pub. L. 96-511, 94 Stat. 2812 (44 U.S.C. 3501 et seq.))

[Order 271, 28 FR 10573, Oct. 2, 1963, as amended at 28 FR 11404, Oct. 24, 1963; Order 537, 40 FR 48674, Oct. 17, 1975; Order 91, 45 FR 46363, July 10, 1980; Order 714, 73 FR 57532, Oct. 3,

§ 35.13 Filing of changes in rate schedules, tariffs or service agreements.

CONTENTS

- (a) General rule.
- (1) Filing for any rate schedule change not otherwise excepted.
- (2) Abbreviated filing requirements.
- (3) Cost of service data required by letter.
- (b) General information.
- (c) Information relating to the effect of the rate schedule change.
- (d) Cost of service information.
 - (1) Filing of Period I data.
 - (2) Filing of Period II data.
 - (3) Definitions.
 - (4) Test period.
 - (5) Work papers.
 - (6) Additional information.
 - (7) Attestation.
- (e) Testimony and exhibits.
- (1) Filing requirements.
- (2) Case in chief.
- (3) Burden of proof.
- (f) Filing by parties concurring in coordination and interchange arrangements.
- (g) Commission precedents and policy.
- (h) Cost of service statements.
 - AA—Balance sheets.
- (2) AB—Income statements.
- (3) AC—Retained earnings statements.
- (4) AD-Cost of plant.
- (5) AE-Accumulated depreciation and amortization.
- (6) AF—Specified deferred credits.
- (7) AG-Specified plant accounts (other than plant in service) and deferred debits
- (8) AH-Operation and maintenance expenses.
- (9) AI—Wages and salaries.
- (10) AJ-Depreciation and amortization expenses.
- (11) AK—Taxes other than income taxes.
- (12) AL—Working capital.
- (13) AM—Construction work in progress.
- (14) AN—Notes payable.
- (15) AO—Rate for allowance for funds used during construction.

- (16) AP—Federal income tax deductions interest.
- (17) AQ—Federal income tax deductions—other than interest.
- (18) AR—Federal tax adjustments.
- (19) AS—Additional state income tax deductions.
- (20) AT-State tax adjustments.
- (21) AU-Revenue credits.
- (22) AV—Rate of return.
- (23) AW—Cost of short-term debt.
- (24) AX—Other recent and pending rate changes.
- (25) AY—Income and revenue tax rate data. (26) BA—Wholesale customer rate groups.
- (27) BB—Allocation demand and capability
- (28) BC-Reliability data.
- (29) BD—Allocation energy and supporting data.
- (30) BE-Specific assignment data.
- (31) BF—Exclusive-use commitments of major power supply facilities.
- (32) BG—Revenue data to reflect changed rates.
- (33) BH—Revenue data to reflect present rates.
- (34) BI—Fuel cost adjustment factors.
- (35) BJ-Summary data tables.
- (36) BK—Electric utility department cost of service, total and as allocated.
- (37) BL—Rate design information.
- (38) Statement BM—Construction program statement
- (a) General rule. Every public utility shall file the information required by this section, as applicable, at the time it files with the Commission under §35.1 all or part of a rate schedule, tariff or service agreement to supersede or otherwise change the provisions of a rate schedule, tariff or service agreement filed with the Commission under §35.1. Any petition filed under §385.207 of this chapter for waiver of any provision of this section shall specifically identify the requirement that the applicant wishes the Commission to waive.
- (1) Filing for any rate schedule change or tariff not otherwise excepted. Except as provided in paragraph (a)(2) of this section, any utility that files a rate schedule, tariff, or service agreement change shall submit with its filing the information specified in paragraphs (b), (c), (d), (e), and (h) of this section, in accordance with paragraph (g) of this section.
- (2) Abbreviated filing requirements—(i) For certain small rate increases. Any utility that files a rate increase for power or transmission services not cov-

- ered by paragraph (a)(2)(ii) of this section may elect to file under this paragraph instead of paragraph (a)(1) of this section, if the proposed increase for the Test Period, as defined in paragraph (a)(2)(i)(A) of this section, is equal to or less than \$200,000, regardless of customer consent, or equal to or less than \$1 million if all wholesale customers that belong to the affected rate class consent.
- (A) Definition: The Test Period, for purposes of paragraph (a)(2)(i) of this section, means the most recent calendar year for which actual data are available, the last day of which is no more than fifteen months before the date of tender for filing under §35.1 of the notice of rate schedule.
- (B) Any utility that elects to file under this subparagraph must file the following information, conforming its submission to any rule of general applicability and to any Commission order specifically applicable to such utility:
- (1) A complete cost of service analysis for the Test Period, consistent with the requirements of paragraph (h)(36), Statement BK, of this section.
- (2) A complete derivation and explanation of all allocation factors and special assignments, consistent with the information required in §35.12(b)(5).
- (3) A complete calculation of revenues for the Test Period and for the first 12 months after the proposed effective date, consistent with the requirements of paragraph (c)(1) of this section.
- (4) If the proposed rates contain a fuel cost or purchased economic power adjustment clause, as defined in §35.14, the company must provide the derivation of its base cost of fuel (Fb) and its monthly fuel factors (Fm) for the Test Period and the resulting fuel adjustment clause revenues. If any pro forma adjustments affect the fuel clause in any way, the company must show the impact on Fm, kWh sales in the base period (Sm), Fb and kWh sales in the current period (Sb), as well as on fuel adjustment clause revenues.
- (5) Rate design calculations and narrative consistent with the information required in paragraph (h)(37) of this section and in §35.12(b)(5).

- (6) The information required in paragraphs (b), (c)(2) and (c)(3) of this section and in $\S 35.12(b)(2)$.
- (C) Data shall be reconciled with the utility's most recent FERC Form 1. If the utility has not yet submitted Form 1 for the Test Period, the utility shall submit the relevant Form 1 pages in draft form.
- (D) The utility may make pro forma adjustments for post-Test Period changes that occur before the proposed effective date and that are known and measurable at the time of filing. The utility shall provide a narrative statement explaining all pro forma adjustments.
- (E) If the utility models its filing in whole or in part on retail rate decisions or settlements, the utility must provide detailed calculations and a narrative statement showing how all retail rate treatments are factored into the cost of service.
- (F) If the Commission sets the filing for hearing, the Commission will allow the company a specific time period in which to file testimony, exhibits, and supplemental workpapers to complete its case-in-chief. While not required under this subpart, a utility may elect to submit Statements AA through BM for the Test Period in accord with the requirements of paragraphs (d), (g) and (h) of this section.
- (ii) Rate increases for service of short duration or for interchange or coordination service. Any utility that files a rate increase for any service of short duration and of a type for which the need and usage cannot be reasonably forecasted (such as emergency or shortterm power), or for service that is an integral part of a coordination and interchange arrangement, may submit with its filing only the information required in paragraphs (b), (c) and (h)(37) of this section and in §35.12(b)(2) and (b)(5), conforming its submission to any rule of general applicability and to any Commission order specifically applicable to such utility.
- (iii) For rate schedule, tariff, or service agreement changes other than rate increases. Any utility that files a rate change that does not provide for a rate increase or that provides for a rate increase that is based solely on a change in delivery points, a change in delivery

- voltage, or a similar change in service, must submit with its filing only the information required in paragraphs (b) and (c) of this section.
- (iv) Computing rate increases. For purposes of this subparagraph and paragraph (d)(2)(ii) of this section, the amount of any rate increase shall be the difference between the total revenues to be recovered under the rate change and the total revenues recovered or recoverable under the rate to be superseded or supplemented and shall be determined by:
- (A) applying the components of the rate to be superseded or supplemented to the billing determinants for the twelve months of Period I;
- (B) Applying the components of the rate change to the billing determinants for the twelve months of Period I; and
- (C) Subtracting the total revenues under subclause (A) from the total revenues under subclause (B).
- (3) Cost of service data required by letter. The Director of the Office of Energy Market Regulation may, by letter, require a utility that is not required under paragraph (a)(1) of this section to submit cost of service data as are needed for Commission analysis of the rate schedule change.
- (b) General information. Any utility subject to paragraph (a) of this section shall file the following general information:
- (1) A list of documents submitted with the rate change;
- (2) The date on which the utility proposes to make the rate change effective:
- (3) The names and addresses of persons to whom a copy of the rate change has been posted;
- (4) A brief description of the rate change;
- (5) A statement of the reasons for the rate change;
- (6) A showing that all requisite agreement to the rate change, or to the filing of the rate change, including any agreement required by contract, has in fact been obtained:
- (7) A statement showing any expenses or costs included in the cost of service statements for Period I or Period II, as defined in paragraph (d)(3) of this section, that have been alleged or

judged in any administrative or judicial proceeding to be illegal, duplicative, or unnecessary costs that are demonstrably the product of discriminatory employment practices; and

- (c) Information relating to the effect of the rate change. Any utility subject to paragraph (a) of this section shall also file the following information or materials:
- (1) A table or statement comparing sales and services and revenues from sales and services under the rate schedule, tariff, or service agreement to be superseded and under the rate change, by applying the components of each such rate schedule or tariff to the billing determinants for each class of service, for each customer, and for each delivery point or set of delivery points that constitutes a billing unit:
- (i) Except as provided in clause (ii), for each of the twelve months immediately before and each of the twelve months immediately after the proposed effective date of the rate change, and the total for each of the two twelve month periods; or
 - (ii) At the election of the utility:
- (A) If the utility files Statements BG and BH under paragraph (h) for Period I, for each of the twelve months of Period I instead of for the twelve months immediately before the proposed effective date of the rate change; and
- (B) If Period II is the test period, for each of the twelve months of Period II instead of for the twelve months immediately after the proposed effective date of the rate change;
- (2) A comparison of the rate change and the utility's other rates for similar wholesale for resale and transmission services; and
- (3) If any specifically assignable facilities have been or will be installed or modified in order to supply service under the ratechange, an appropriate map or sketch and single line diagram showing the additions or changes to be made.
- (d) Cost of service information—(1) Filing of Period I data. Any utility that is required under paragraph (a)(1) of this section to submit cost of service information, or that is subject to the exceptions in paragraphs (a)(2)(i) and (a)(2)(ii) of this section but elects to file such information, shall submit

Statements AA through BM under paragraph (h) of this section using:

- (i) Unadjusted Period I data; or
- (ii) Period I data adjusted to reflect changes that affect revenues and costs prior to the proposed effective date of the rate change and that are known and measurable with reasonable accuracy at the time the rate schedule change is filed, if such utility:
- (A) Is not required to and does not file Period II data;
- (B) Adjusts all Period I data to reflect such changes; and
- (C) Fully supports the adjustments in the appropriate cost of service statements.
- (2) Filing of Period II data. (i) Except as provided in clause (ii) of this subparagraph, any utility that is required under paragraph (a)(1) of this section to submit cost of service information shall submit Statements AA through BM described in paragraph (h) using estimated costs and revenues for Period II;
- (ii) A utility may elect not to file Period II data if:
- (A) The utility files a rate increase that is less than one million dollars for Period I; or
- (B) All wholesale customers that belong to the affected rate class have consented to the rate increase.
- (3) *Definitions*. For purposes of this section:
- (i) Period I means the most recent twelve consecutive months, or the most recent calendar year, for which actual data are available, the last day of which is no more than fifteen months before the date of tender for filing under §35.1 of the notice of rate change;
- (ii) Period II means any period of twelve consecutive months after the end of Period I that begins:
- (A) No earlier than nine months before the date on which the rate change is proposed to become effective; and
- (B) No later than three months after the date on which the rate change is proposed to become effective.
- (4) Test period. If Period II data are not submitted for Statements AA through BM, Period I shall be the test period. If Period II data are submitted for Statements AA through BM, Period II shall be the test period.

- (5) Work papers. A utility that files adjusted Period I data or that files Period II data shall submit all work papers relating to such data. The utility shall provide a comprehensive explanation of the bases for the adjustments or estimates and, if such adjustments or estimates are based on a regularly prepared corporate budget, shall include relevant excerpts from such budget. Work papers and documents containing additional explanatory material shall be provided in electronic format, shall be legible, shall be assigned page numbers, and shall be marked, organized and indexed according to:
 - (A) Subject matter;
- (B) The cost of service statements to which they apply; and
 - (C) Witness.
- (6) Attestation. A utility shall include in its filing an attestation by its chief accounting officer or another of its officers that, to the best of that officer's knowledge, information, and belief, the cost of service statements and supporting data submitted under this paragraph are true, accurate, and current representations of the utility's books, budgets, or other corporate documents.
- (e) Testimony and exhibits—(1) Filing requirements. (i) A utility subject to paragraph (a)(1) of this section shall file Statements AA through BM under paragraph (h) as exhibits with its rate change and may file any other exhibits in support of its rate schedule change.
- (ii) A utility subject to paragraph (a)(1) of this section shall file prepared testimony. Such testimony shall include an explanation of all exhibits, inlcuding Statements AA through BM, and shall include support for all adjustments to book or budgeted data relied on in preparing the exhibits.
- (iii) To the extent that testimony and exhibits other than Statements AA through BM duplicate information required to be submitted in such statements, the testimony and exhibits may incorporate such information by referencing the specific statement containing such material.
- (2) Case in chief. In order to avoid delay in processing rate filings, such cost of service statements, testimony, and other exhibits described in para-

- graph (e)(1) of this section shall be the utility's case in chief in the event the matter is set for hearing.
- (3) Burden of proof. Any utility that files a rate increase shall be prepared to go forward at a hearing on reasonable notice on the data submitted under this section, to sustain the burden of proof under the Federal Power Act of establishing that the rate increase is just and reasonable and not unduly discriminatory or preferential or otherwise unlawful within the meaning of the Act.
- (f) Filing by parties concurring in coordination and interchange arrangements. For coordination and interchange arrangements in the nature of power pooling transactions, all information required to be submitted in support of a rate change under paragraphs (a)(1), (2), and (3) of this section shall be submitted by each party filing a certificate of concurrence under §35.1. If a representative is designated and authorized in accordance with §35.1 to file supporting information on behalf of all parties to a rate change, such filing shall fulfill the requirement in this paragraph for individual submittals by each party.
- (g) Commission precedents and policy. If a utility submits cost of service data under paragraph (d) of this section, it shall conform all such submissions to any rule of general applicability and to any Commission order specifically applicable to such utility.
- (h) Cost of service statements. Any utility subject to paragraph (a)(1) of this section shall submit the following Statements AA through BM in accordance with the requirements of paragraphs (d) and (g) of this section.
- (1) Statement AA—Balance sheets. Statement AA consists of balance sheets as of the beginning and the end of both Period I and Period II, and the most recently available balance sheet, including any applicable notes, and an explanation of any significant accounting changes since the most recent filing by the utility under this section that involves the same wholesale customer rate class. Balance sheets shall be constructed in accordance with the annual report form for electric utilities specified in part 141.

- (2) Statement AB—Income statements. Statement AB consists of income statements for both Period I and Period II, and the most recently available income statement, including any applicable notes, and an explanation of any significant accounting changes since the most recent filing by the utility under this section that involves the same wholesale customer rate class. Income statements shall be prepared in accordance with the annual report form for electric utilities specified in part 141.
- (3) Statement AC—Retained earnings statements. Statement AC consists of retained earnings statements for both Period I and Period II, and the most recently available retained earnings statement, including any notes applicable thereto. Retained earnings statements shall be prepared in accordance with the annual report form for electric utilities specified in part 141.
- (4) Statement AD—Cost of plant. Statement AD is a statement of the original cost of total electric plant in service according to functional classification for Period I and Period II. If the plant functions and subfunctions for Period I and Period II are different, the utility shall explain and justify the differences.
- (i) For each separately identified function and subfunction of production plant or transmission plant, the utility shall state the original cost as of the beginning of the first month and the end of each month of both Period I and Period II, with an average of the thirteen balances for each period. If any of the Period I or Period II thirteen monthly balances is not available or is unrepresentative of the current plan of the utility for plant in service, the utility shall provide an explanation of the relevant circumstances.
- (ii) For each separately identified function and subfunction of plant other than production or transmission, the utility shall state the original cost as of the beginning and the end of both Period I and Period II, with an average of the beginning and end balances for each period. If any of the Period I or Period II balances is not available or is unrepresentative of the current plan of the utility for plant in service, the

utility shall provide an explanation of the relevant circumstances.

- (iii) The utility shall show the electric plant in service in accordance with each of the following five major functional classifications:
 - (A) Production;
 - (B) Transmission;
 - (C) Distribution;
 - (D) General and Intangible; and
 - (E) Common and Other.
- (iv) To the extent feasible, the utility shall show completed construction not classified in accordance with clause (iii) in accordance with tentative classification to major functional accounts. If this is not feasible, the utility shall describe such facilities as other plant under clause (iii)(E).
- (v) If a utility designs its rate change so that subdivision of the major functional classifications is necessary to support the changed rate, the utility shall supply the original cost information for any of the five major functional plant classifications in clause (iii) that are divided into subfunctional categories. If subfunctional original cost information is provided, the utility shall explain the importance of providing such information to support the changed rate. The utility shall describe each subfunctional category in substantive terms, such as steam electric production or high voltage transmission.
- (vi) The utility shall select any subfunctional categories, as appropriate, under the following criteria:
- (A) Production plant categories shall be established as necessary to segregate costs for production services with special characteristics, such as base, intermediate or peaking load. The utility shall provide a description of each such service and shall list a brief descriptive title for each corresponding subfunctional category.
- (B) Transmission plant categories shall be chosen to reflect the extent to which the facilities are proposed to be allocated on a common basis among all or specific segments of utility services. For descriptive purposes, plant may also be categorized according to accounting or engineering terminology, such as high voltage overhead lines. The utility shall provide brief descriptive

transmission category titles and explain the basis for the titles. If a utility allocates all transmission plant among utility services on the basis of a single set of allocation data, the utility may show original cost in total without subfunctionalization.

- (C) Distribution plant categories shall be selected according to engineering or use characteristics meaningful for allocations or assignments to wholesale services such as substations, overhead lines, meters, or non-wholesale. The utility shall provide brief descriptive distribution category titles and shall explain the basis for the titles.
- (D) If the utility divides any general, intangible, common, and other plant functional classifications into subfunctional categories, the subfunctional categories shall be chosen to group together facilities that share a common basis for allocation between wholesale and other electric services. The utility shall provide a brief descriptive title for each general and intangible subfunctional category, and for each common and other subfunctional category, with an explanation of the basis of each category selection. A utility need not divide the functional classifications of plant into subfunctional categories if these functions of plant are allocated in Statement BK on the basis of utility labor expenses.
- (E) A separate category shall be provided for each specific assignment of plant reported in Statement BE. Such assignments are applicable principally but not necessarily exclusively to distribution facilities. The utility shall provide brief descriptive titles consistent with Statement BE.
- (F) A separate category shall be provided for each exclusive-use commitment of major power supply facilities as required to be reported at Statement BF. The utility shall provide brief descriptive titles consistent with Statement BF.
- (5) Statement AE—Accumulated depreciation and amortization. Statement AE is a statement of the accumulated provision for depreciation and amortization of electric plant for Period I and Period II, provided according to major functional classifications selected by the utility in Statement AD under

- paragraph (h)(4) and divided into the subfunctional categories selected by the utility in Statement AD, to the extent that subfunctionalized data are available.
- (i) For each function and subfunction of electric production and transmission plant in service identified in Statement AD, the utility shall set forth the accumulated depreciation and amortization as of the beginning of the first month and the end of each month of both Period I and Period II. The utility shall state an average for each period computed as the average of the thirteen balances.
- (ii) For each function and subfunction of electric plant in service other than production or transmission, identified in Statement AD, the utility shall state the accumulated depreciation and amortization as of the beginning and the end of Period I and Period II, with an average of the beginning and end balances for each period.
- (iii) If any of the Period I or Period II balances is not available or is unrepresentative of the current plan of the utility for depreciation reserves, the utility shall provide an explanation of the relevant circumstances.
- (iv) If accumulated depreciation and amortization data are not available for any subfunction selected in Statement AD, the utility shall:
- (A) Provide a comparison of the current depreciation rate of the major functional classification and the depreciation rate estimated to be appropriate to the subfunctional category; and
- (B) State and explain the estimation techniques which the utility proposes to utilize in the absence of subfunctional data, such as the proration of accumulated depreciation and amortization data among the subfunctional categories according to the data for electric plant in service in Statement AD. If any of the proposed estimation techniques require data that are not provided elsewhere in the cost of service statements in paragraph (h) of this section, the utility shall supply the necessary data in Statement AE.
- (6) Statement AF—Specified deferred credits. Statement AF consists of balances of specified accounts and items

which are to be considered in the determination of the net original cost rate base. All required balances are to be stated as of the beginning and end of both Period I and Period II, with an average of the beginning and end balances for each period. If any of the Period I and Period II balances is not available or is unrepresentative of the current operating plan of the utility, the utility shall include an explanation of the relevant circumstances. If subaccounts are maintained to reflect differences in ratemaking treatment among regulatory authorities that have jurisdiction, balances shall be provided in accordance with such subaccounts, with detailed explanations of the bases upon which the subaccounts were established and are maintained. The balances of deferred credits required to be filed in this statement are described in paragraph (h)(6) through (v) of this section. All references to numbered accounts refer to the Commission's Uniform System of Accounts, 18 CFR part 101.

(i) The utility shall state total electric balances for accumulated deferred investment tax credits Account 255, and shall separate the credits into balances applicable to pre-1971 and post-1970 qualifying property additions. If the utility maintains records to show Account 255 component balances according to the major functional classifications identified in Statement AD under paragraph (h)(4), the utility shall provide the component balances by function. If the data are not available by function, the utility shall describe the procedure by which the utility believes it can reasonably estimate the portion of the total electric balances for each major functional classification. The utility may show by function the component balances obtained by applying the procedure. If such estimation requires data that are not provided elsewhere in the cost of service statements in this paragraph, the utility shall supply in Statement AF the necessary data, such as historical functional patterns of plant additions eligible for the tax credits. The utility shall state whether the Internal Revenue Code General Rule, §46(f)(1), is applicable with respect to restrictions on credit treatment for ratemaking purposes.

If the General Rule is not applicable, the utility shall state which election it has made with respect to special rules for ratable or immediate flow-through for ratemaking purposes.

(ii) The utility shall state the total electric component balances for accumulated deferred income tax Account 281 pertaining to accelerated amortization property. The utility shall show separate components for defense, pollution control, and other facilities. The utility shall show balances for each component and totaled for the electric utility department. If the utility maintains records to show Account 281 component balances according to the major functional classifications identified in Statement AD under paragraph (h)(4), the utility shall provide such component balances. If data are not available by function, the utility shall describe the procedure by which the utility believes it can reasonably estimate the portion of the total electric balances for each major functional classification. The utility may show by function the component balances obtained by applying the procedure. If such estimation requires data that are not provided elsewhere in the cost of service statements in this paragraph, the utility shall supply in Statement AF the necessary data.

(iii) The utility shall state the total electric component balances for accumulated deferred income tax Account 282 pertaining to electric utility property other than accelerated amortization property. The utility shall itemize the balances in Account 282, to the extent data are available, in detail sufficient to identify the specific major properties involved and shall list the balances according to the accounting entries, such as liberalized depreciation, for which interperiod tax allocation was used and included in this account. Component balances shall be shown individually and in total for the electric utility department. If the utility maintains records to show account 282 component balances according to the major functional classifications identified in Statement AD under paragraph (h)(4), the utility shall provide such component balances by function.

If the data are not available by function, the utility shall describe the procedure by which the utility believes it can reasonably estimate the portion of the total electric balances for each major functional classification. The utility may show by function the component balances obtained by applying the procedure. If such estimation requires data that are not provided elsewhere in the cost of service statements in this paragraph, the utility shall supply in Statement AF the necessary data, such as historical functional patterns of plant additions.

(iv) The utility shall state the total electric component balances for accumulated deferred income tax Account 283 pertaining to interperiod income tax allocations not related to property. The utility shall itemize in detail balances in Account 283, to the extent data are available, and shall list the balances according to the accounting entries for which interperiod tax allocation was used and included in this account. Component balances shall be shown individually and in total for the electric utility department. If the utility maintains records to show Account 283 component balances according to the major functional classifications identified in Statement AD under paragraph (h)(4), the utility shall provide such component balances by function. If the data are not available by function, the utility shall describe the procedure by which the utility believes it can reasonably estimate the portion of the total electric balances for each major functional classification. The utility may show by function the component balances obtained by applying the procedure. If such estimation requires data that are not provided elsewhere in the cost of service statements in this paragraph, the filing shall supply in Statement AF the necessary data.

- (v) The utility shall show electric utility balances for every other item that the utility believes should be included in Statement AF. The utility shall explain the reasons for inclusion of each item.
- (7) Statement AG—Specified plant accounts (other than plant in service) and deferred debits. Statement AG is a statement of balances of specified ac-

counts and items that are to be considered in the determination of the net original cost rate base. Except as prescribed in clause (ii), the utility shall state all required balances as of the beginning and the end of Period I and Period II, with an average of the beginning and end balances for each period. If any of the Period I or Period II balances is not available or is unrepresentative of the current operating plan of the utility, the utility shall provide a full explanation of the relevant circumstances. If subaccounts are maintained to reflect differences in ratemaking treatment among regulatory authorities having jurisdiction, the utility shall provide balances in accordance with such subaccounts, with detailed explanations of the bases upon which the subaccounts were established and are maintained. The balances required to be submitted under Statement AG are described in clauses (7)(i) through (vi).

- (i) For each separately identified major functional classification selected by the utility in Statement AD, the utility shall state the electric utility land and land rights balances for electric plant held for future use in account 105. If itemized in detail, balances shall be totaled for each major functional classification.
- (ii) The utility shall state the electric utility component balances in Accounts 107 and 120.1, individually and in total, for each item of construction work in progress for pollution control facilities, fuel conversion facilities, or any other facilities that qualify for inclusion in rate base under §35.26. The utility shall state such balances for each major functional and subfunctional classification under Statement AD as of the beginning of the first month and the end of each month of Period I and Period II with an average of the 13 balances for each period.
- (iii) For each major functional classification under Statement AD and with respect to property otherwise includable in plant in service, the utility shall state the balances for extraordinary property losses Account 182. If

itemized in detail, balances shall be totaled for each major functional classification. The utility shall provide information about Commission authorization for any loss included in Account 182 and shall state when the loss was claimed for tax purposes.

- (iv) The utility shall state the total electric component balances for accumulated deferred income taxes Account 190. The component balances in Account 190 shall be itemized in detail and listed according to the accounting entries for which interperiod tax allocation was used. Component balances shall be shown individually and in total for the electric utility department. If the utility maintains records to show Account 190 component balances according to the major functional classifications identified in Statement AD under paragraph (h)(4), the utility shall provide such component balances by function. If the data are not available by function, the filing utility shall describe the procedure by which the utility believes it can reasonably estimate the portion of the total electric balances for each major functional classification. The utility may show by function the component balances obtained by applying the procedure. If such estimation requires data that are not provided elsewhere in the cost of service statements in this paragraph, the utility shall supply in Statement AG the necessary data.
- (v) Balances shall be shown for every other item that the utility believes should be included in Statement AG. The utility shall provide support for inclusion of each item, and a brief descriptive title for each such item.
- (8) Statement AH—Operation and maintenance expenses. Statement AH is a statement of electric utility operation and maintenance expenses for Period I and Period II provided according to the accounts prescribed by the Commission's Uniform System of Accounts, 18 CFR part 101.
- (i) For Period I and Period II, the utility shall itemize and subtotal all operation and maintenance expenses according to the major functional classifications of Statement AD in paragraph (h)(4) and the subfunctional categories of those classifications. The utility shall further divide the oper-

ation and maintenance expenses itemized under the production classification and each of its subfunctional categories to reflect expenses relating to the energy component (list each item by account number and compute fuel costs on an as-burned basis), the demand component, and any other production expenses.

- (ii) For Period I and Period II, the utility shall report production operation and maintenance expenses according to appropriate account numbers. The utility shall apply the following principles in developing Period I and Period II production operation and maintenance data for this statement:
- (A) Total production operation and maintenance expenses shall be segregated into energy, demand, and other components. The utility shall specifically state and support its criteria for classifications between energy and demand, and for use of the production other classification, such as specific assignments related to sales from particular generating units.
- (B) Fuel expense for cost of service purposes shall be the total as-burned expense incurred. If the utility defers a portion of such expense for accounting purposes, the deferral amount shall be separately stated and accompanied by material that shows computational detail in support of such amount. If claimed nuclear fuel expense reflects a change in the estimated net salvage value of nuclear fuel, the utility shall show the amounts involved and explain the relevant circumstances.
- (C) If the amount of production fuel expense is significantly affected by abnormal Period I water availability for hydroelectric generation, the utility shall explain how water availability was taken into account in developing projected Period II production fuel expenses.
- (iii) For Period I and Period II, the utility shall report operation and maintenance expenses attributable to the transmission and distribution functions according to appropriate account numbers. If Period II transmission and distribution plant data are not provided by subfunctional category in Statement AD, the utility need only provide for Period II total operation

and maintenance expenses for each function.

- (iv) For Period I and Period II, the utility shall report in total for each period, operation and maintenance expenses incurred under each of the categories of customer accounting, customer service and information, and sales.
- (v) For Period I and Period II, the utility shall itemize administrative and general expenses by groups that are directly assignable, such as regulatory Commission expenses, or that are related to selected plant or expense items for which an allocation to wholesale services is independently determinable, such as items related to labor expense or to a category of production plant in service. Administrative and general expenses shall include a detailed itemization of the general advertising Account 930.1 and the miscellaneous general expenses Account 930.2. If Account 930 data are not projected on a detailed basis for Period II, the utility shall provide its best estimate of the Account 930.1 expense items and a descriptive list of expense items anticipated as miscellaneous general expenses in Account 930.2. Where applicable, separate items shall be shown for general plant maintenance, and for common and other plant maintenance.
- (vi) In addition to annual production data for Period I and Period II, the utility shall provide monthly expense data by accounts for fuel in Accounts 501, 518, and 547 and purchased power in Account 555. For each type of transaction, such as firm power or economy interchange power, monthly purchased power expense data shall be subtotaled separately for interchange receipts and deliveries. For monthly fuel Accounts 501, 518, and 547, and for each type of purchased power transaction, monthly data shall identify components to be claimed under the fuel adjustment clause of the utility.
- (9) Statement AI—Wages and salaries. Statement AI consists of statements of the electric utility wages and salaries, for Period I and Period II, that are included in operation and maintenance expenses reported in Statement AH.
- (i) For Period I and Period II, the utility shall show the distribution of wages and salaries by function accord-

- ing to the form prescribed for operation and maintenance expenses by the Commission's Uniform System of Accounts, 18 CFR part 101. The statement shall also include by function additional wages and salaries attributable to common and other plant classifications identified in Statement AD in paragraph (h)(4).
- (ii) For Period I and Period II, the utility shall show total production wages and salaries, itemized and subtotaled into energy and demand related components in accordance with classifications of Statement AH operation and maintenance production expenses of which production wages and salaries are a part.
- (10) Statement AJ—Depreciation and amortization expenses. Statement AJ consists of statements of depreciation and amortization expenses for Period I and Period II.
- (i) For Period I and Period II, the utility shall show the depreciation and amortization expenses and the depreciable plant balances of the filing utility, in accordance with major functional classifications selected by the utility in Statement AD under paragraph (h)(4).
- (ii) The utility shall divide the major functional classifications of depreciation and amortization expenses shown in clause (i) into the subfunctional categories selected by the utility for electric plant in service in Statement AD, to the extent such data are available.
- (iii) If depreciation and amortization expense data are not available for any subfunctional category selected in Statement AD, the utility shall:
- (A) Provide a comparison of the current depreciation rate of the major functional classification and the depreciation rate estimated to be appropriate to the subfunctional category; and
- (B) State and explain the estimation techniques that the utility utilized in developing each estimated subfunctional depreciation rate. If utilization of such estimation techniques requires data that are not provided elsewhere in the cost of service statements in this paragraph, the utility shall supply such data in Statement AJ.

- (iv) For Period I and Period II, the utility shall show the annual depreciation rate applicable to each function and subfunction for which depreciation expense is reported. The utility shall indicate the bases upon which the depreciation rates were established. If the depreciation rates used for Period I or Period II data differ from those employed to support the utility's prior approved jurisdictional electric rate, the utility shall include in or append to Statement AJ detailed studies in support of such changes. These detailed studies shall include:
- (A) Copies of any reports or analyses prepared by any independent consultant or utility personnel to support the proposed depreciation rates; and
- (B) A detailed capital recovery study showing by primary account the depreciation base, accumulated provision for depreciation, cost of removal, net salvage, estimated service life, attained age of survivors, accrual rate, and annual depreciation expense.
- (11) Statement AK—Taxes other than income taxes. Statement AK consists of statements of taxes other than income taxes for Period I and Period II.
- (i) For Period I and Period II, the utility shall itemize and total any taxes other than income taxes according to clauses (i) (A) through (D).
- (A) Revenue taxes. The utility shall show total revenue taxes levied by each taxing authority and identify the revenue taxes, under both the present and changed rate, applicable to wholesale services for which a rate change is filed. The utility shall identify revenue taxes associated with each revenue credit item reported in Statement AU under paragraph (h)(21).
- (B) Real estate and property taxes. The utility shall itemize and total all real estate and property taxes. If the utility maintains records to show tax component balances according to the major functional classifications identified in Statement AD under paragraph (h)(4), the utility shall supply the component balances by function. If the data are not available by function, the utility shall describe the procedure by which the utility believes it can reasonably estimate the portion of the total electric balances for each major functional classification. The utility may show by

- function the component balances obtained by applying the procedure. If such estimation requires data that are not provided elsewhere in the cost of service statements in this paragraph, the utility shall supply the necessary data in Statement AK.
- (C) Payroll taxes. The utility shall itemize and total all payroll taxes. If the utility maintains records to show tax component balances according to the major functional classifications identified in Statement AD in paragraph (h)(4), the utility shall provide the component balances by function. If the data are not available by function, the utility shall describe the procedure by which the utility believes it can reasonably estimate the portion of the total electric balances for each major functional classification. The utility may show by function the component balances obtained by applying the procedure. If such estimation requires data that are not provided elsewhere in the cost of service statements in this paragraph, the utility shall provide the necessary data in Statement AK.
- (D) Miscellaneous taxes. The utility shall itemize and total all miscellaneous taxes which are directly assignable or which are related to any selected plant or expense item for which an allocation to wholesale services is independently determinable, such as items related to transmission plant in service or to net distribution plant.
- (ii) If any of the taxes itemized under clause (11)(i) are levied by a taxing authority that is a customer, or is related to a customer, whose services would be affected by the changed rate schedule, the utility shall show amounts of such taxes according to the taxing authority, identify the related customer, and provide an explanation of the relevant circumstances.
- (12) Statement AL—Working capital. Statement AL consists of statements for Period I and Period II designed to establish the need for working capital to maintain adequate levels of operating supplies, to meet required prepayments, and to meet ongoing cash disbursements that must be made at a time different than related revenue receipts for utility services rendered.
- (i) Supplies and prepayments. The utility shall supply statements to show

monthly balances of operating supplies and prepayments itemized under clauses (i) (A) through (C). The utility shall state all required balances as of the beginning of the first month and the end of each month of both Period I and Period II, with an average of the thirteen balances for each period. If any of the Period I or Period II balances is not available or is unrepresentative of the current operating plan of the utility for supplies or prepayments, the utility shall include an explanation of the relevant cumstances. Operating supply and prepayment balances shall be itemized under the following categories:

(A) Fuel supplies. The utility shall state the fuel supply balances for each type of electric utility production plant, except hydraulic. The utility shall describe its overall fossil fuel supply objectives for Period I and Period II, in terms of projected average days of burn for major fossil fuel generating stations, if feasible. The utility shall explain substantial differences, if any, between actual Period I inventories and the target objectives, or between Period II objectives and Period I objectives. Nuclear fuel balances shall include fuel in stock, fuel in the reactor and spent fuel in the process of cooling in Accounts 120.2, 120.3, 120.4, less accumulated provisions for amortization of nuclear fuel assemblies in Account 120.5.

(B) Plant materials and operating supplies. The utility shall state materials and operating supply balances for each of the major electric utility operating functions of production, transmission, and distribution, and for each significant type of miscellaneous operating supplies. Miscellaneous supplies shall be grouped to facilitate suitable allocations or assignments among utility services

(C) Prepayments. The utility shall indicate prepayment balances for each major prepayment item, with a brief description of the item. Balances shall be grouped and subtotaled to facilitate suitable allocations or assignments among utility services.

(ii) Cash working capital. The utility shall indicate average monthly working cash requirements that reflect the extent to which day-to-day operational

utility service revenues are received later or earlier than cash disbursements necessary to provide the services, with an explanation of how such requirements are derived.

(13) Statement AM—Construction work in progress. Statement AM is a statement of the amount of construction work in progress described according to functional classification for Period I and Period II. For production plant and transmission plant, the utility shall state the balances as of the beginning of the first month and the end of each month of both Period I and Period II, with an average of the 13 balances for each period. For each function of plant identified in Statement AD other than production or transmission, the utility shall state the balances as of the beginning and the end of both Period I and Period II, with an average of the beginning and end balances for each period. If any Period I or Period II balance is not available, the utility shall include monthly estimates and an explanation of the relevant circumstances. Pollution control facilities, fuel conversion facilities. orotherconstruction amounts reported in Statement AG shall be excluded from amounts reported in this Statement.

(14) Statement AN—Notes payable. Statement AN is a statement of the electric utility portion of average notes payable for Period I and Period II. The utility shall indicate balances as of the beginning of the first month and the end of each month of both Period I and Period II, with an average of the thirteen balances for each period. If any of the Period I or Period II balances is not available or is unrepresentative of the current financing plan of the utility, the utility shall provide an explanation of the relevant circumstances. If a utility has operations other than electric, the utility shall also show allocations between electric and other utility departments on an appropriate basis, such as the average amount of construction work in progress and net plant.

(15) Statement AO—Rate for allowance for funds used during construction. Statement AO is a statement of the

basis of the rate for computing the allowance for funds used during construction (AFUDC) for Period I and Period II.

- (i) The utility shall show the computations of the maximum rates for the construction allowances computed in accordance with plant instructions of the Commission's Uniform System of Accounts, 18 CFR part 101. The utility shall show the rates computed annually, and shall provide the rates for each annual period that includes any part of Period I or Period II. If the utility proposes to use a net-of-tax rate, the utility shall show the derivation for both the gross-of-tax and net-of-tax rates.
- (ii) If the book allowance amounts of AFUDC do not reflect the maximum rates for allowances for funds computed in accordance with clause (i), the utility shall show the derivation for the actual rates utilized in computing AFUDC, including derivation of any net-of-tax rate utilized by the utility.
- (16) Statement AP—Federal income tax deductions—interest. Statement AP is a statement of electric utility interest charges for Period I and Period II. For each period, the utility shall state the total electric utility interest in terms of three or more component items described in clauses (i) through (iv).
- (i) The utility shall state the allowance for borrowed funds used for electric utility construction Account 432 as a separate component. The utility shall show supporting detail, including computation of the amounts on the basis of AFUDC rates claimed in Statement AO.
- (ii) The utility shall state interest for borrowed funds used for electric utility construction Account 431 as a separate component. If applicable, the utility shall also show all elements of Account 431 related to purposes other than electric utility construction, with detailed supporting material, such as a computation of allocations between electric and other utility departments with explanatory material to support the bases of such allocations.
- (iii) The utility shall state the interest on long-term debt required for electric rate base investment as a separate component. The interest amount shall be consistent with that shown and uti-

lized in Statement BK under paragraph (h)(36) of this section.

- (iv) The utility shall show other interest items appropriate in the determination of net taxable income allocable to the wholesale services at issue. The utility shall describe and support each item and shall accompany each item with a statement of the basis on which the item is allocable to the wholesale services. The utility shall also list a short descriptive title for each item.
- (17) Statement AQ—Federal income tax deductions—other than interest. Statement AQ is a statement of other deductions from net operating income before Federal income taxes, for Period I and Period II, which deductions are appropriate in determining the net taxable income allocable to the wholesale services subject to the changed rate. The utility shall show unallowable deductions as negative entries in this statement. The utility shall itemize deductions in accordance with clause (i) through (iii) and individually identify each by a brief descriptive title.
- (i) The utility shall report, as a separate component of this statement, the difference between tax and book depreciation, in total, or in individual amounts based on the Internal Revenue Code provisions that permit the utility to use various methods of computing depreciation for tax purposes, such as liberalized depreciation or the asset depreciation range. If the utility reports the differences in total only, it shall list the specific Internal Revenue Code provisions that result in the difference.
- (ii) The utility shall state taxes and pensions capitalized as a separate component.
- (iii) The utility shall describe and support other deduction items appropriate in the determination of net taxable income allocable to the wholesale services. Each item shall be accompanied by a brief explanation of the basis on which the item is allocable to the wholesale services.
- (18) Statement AR—Federal tax adjustments. Statement AR is a statement of adjustments to Federal income taxes for Period I and Period II. If subaccounts are maintained to reflect differences in ratemaking treatment among regulatory authorities having

jurisdiction, the utility shall provide adjustment amounts in accordance with such subaccounts. The utility shall report detailed explanations of the bases upon which the subaccounts were established and are maintained.

(i) For each major function of plant identified in Statement AD under paragraph (h)(4), the utility shall state the electric utility component adjustment for the Federal portions of the provision for deferred income tax Account 410.1. If the data are not available by function, the utility shall state the amounts for the total electric utility and shall describe the procedure by which the utility believes it can reasonably estimate the portion of the total electric balances for each major functional classification. The utility may show by function the component balances obtained by applying the procedure. If such estimation requires data that are not provided elsewhere in the cost of service statements in this paragraph, the utility shall supply in Statement AR the necessary data. The utility shall provide the adjustment amounts for total electric and, to the extent available for each such major functional component, accompanied by summary totals segregated in accordance with related balance sheet Accounts 281, 282, 283, and 190 [see Statements AF and AG]. Account 190 items require a negative sign for entries in Statement AR. The utility shall identify the summarized items by account number.

(ii) The utility shall provide for the Federal portions of the provision for deferred income tax-credit Account 411.1 the data required by clause (i) for Account 410.1.

(iii) For each major functional classification of plant identified in Statement AD under paragraph (h)(4), the utility shall provide the electric utility component for investment tax credits generated for Period I and Period II, credits utilized for each period, and the allocations to current income for each period. If the data are not available by function, the utility shall state the amounts for total electric utility and shall describe the procedure by which the utility believes it can reasonably estimate the portion of the total electric balances for each major functional

classification. The utility may show by function the component balance obtained by applying the procedure. If such estimation requires data that are not provided elsewhere in the cost of service statements in this paragraph, the utility shall supply in Statement AR the necessary data. If itemized in detail, balances shall be subtotaled for each major function, and totaled for the electric utility department. Detailed data shall be consistent with that provided in Statement AF under paragraph (h)(6) of this section.

(iv) The utility shall list and designate as other adjustment items any additional Federal income tax adjustments and shall provide a brief descriptive title for each item. The utility shall explain the reasons for inclusion of each item, and shall indicate the basis on which each will be assigned or allocated to the wholesale services subject to the changed rate and to the other electric utility services.

(19) Statement AS-Additional state income tax deductions. Statement AS is a listing of state income tax deductions for Period I and Period II, in addition to those listed at Statements AP and AQ for Federal tax purposes. The utility shall explain the reasons for inclusion of each item. The utility shall indicate the basis on which each item is to be assigned or allocated to the wholesale services at issue and to the other electric utility services. If applicable, the utility shall show unallowable deductions as negative entries in this statement. The utility shall provide the percentage of Federal income tax payable which is deductible for state income tax purposes, if applicable. [See also Statement AY, dealing with tax rate data.]

(20) Statement AT—State tax adjustments. Statement AT is a statement of adjustments to state income taxes for Period I and Period II. The utility shall prepare and present the data in statement AT as prescribed for Federal tax adjustments in Statement AR. The utility shall annotate Statement At data as necessary to identify state tax adjustments that are not properly deductible for Federal tax purposes.

(21) Statement AU—Revenue credits. Statement AU is, for Period I and Period II, a statement of the operating

revenue balances in Accounts 450 through 456, and other revenue items, such as short-term sales in Account 447, that are appropriately credited to the cost of service for determinations of costs allocable to the wholesale services subject to the changed rate. The utility shall include revenue credits proposed for exclusive-use commitment of major power supply facilities according to instructions for preparation of Statement BF under paragraph (h)(31) of this section. When applicable, the utility shall state revenue taxes for each revenue credit item. The utility shall explain the reasons for inclusion of each item, and shall indicate the basis for assigning or allocating each item to the wholesale services subject to the changed rate and to the other electric utility services.

- (22) Statement AV—Rate of return. Statement AV is a statement and explanation of the percentage rate of return requested by the utility. The utility shall provide the complete capital structure, including ratios, component costs and weighted component costs claimed by the utility. The utility shall submit additional data where any component of the capital of the utility is not primarily obtained through its own financing, but is primarily obtained from a company by which the utility is controlled, as defined in the Commission's Uniform System of Accounts, 18 CFR part 101. The utility shall submit the additional data, if required with respect to the debt capital, preferred stock capital and common stock capital of such controlling company or any intermediate company through which such funds have been secured.
- (i) General. The utility shall show, based on the capitalization of the utility, the cost of debt capital and preferred stock capital, the claimed rate of return on the common equity of the utility and the resulting overall rate of return requested.
- (A) For Period I and, if applicable, Period II, the utility shall show in tabular form the following:
- Cost of each capital element, including claimed rate of return on equity capital;
- (2) Capitalization amounts and ratios;

- (3) Weighted cost of each capital element; and
- (4) Overall claimed rate of return.
- (B) When a Period II filing is submitted the utility shall provide:
- (I) A full explanation of, and supporting work papers for, the *pro forma* adjustments to the actual capitalization data to arrive at the Period II capitalization; and
- (2) The *pro forma* adjustment to Period I data to arrive at the Period II amount for unappropriated undistributed subsidiary earnings in Account 216.1.
- (C) If not included elsewhere in the filing, the utility shall submit the amount for Account 216.1 for Period I as part of this statement.
- (ii) Debt capital. (A) The utility shall show the weighted cost for all issues of long-term debt capital as of the end of Period I, as expected on the date the changed rate is filed, and, if applicable, as estimated for the end of Period II. The weighted cost is calculated by: (1) Multiplying the cost of money for each issue under clause (B)(6) below by the principal amount outstanding for each issue, which yields the annualized cost for each issue: and (2) adding the annual cost of each issue to obtain the total for all issues, which is divided by the total principal amount outstanding for all issues to obtain the weighted cost for all issues.
- (B) The utility shall show the following for each class and series of long-term debt outstanding as of the end of Period I, as expected on the date the changed rate is filed, and, if applicable, as estimated to be outstanding as of the end of Period II.
 - (1) Title;
- (2) Date of offering and date of maturity;
 - (3) Interest rate:
 - (4) Principal amount of issue;
 - (5) Net proceeds to the utility;
- (6) Cost of money, which is the yield to maturity at issuance based on the interest rate and net proceeds to the utility determined by reference to any generally accepted table of bond yields;
 - (7) Principal amount outstanding;
- (δ) Name and relationship of issuer and if the debt issue was issued by an affiliate; and

- (9) If the utility has acquired at a discount or premium some part of the outstanding debt which could be used in meeting sinking fund requirements, or for some other reason, the annual amortization of the discount or premium for each issue of debt from the date of the reacquisition over the remaining life of the debt being retired. The utility shall show separately the total discount and premium to be amortized, and the amortized amount applicable to Period I and, if applicable, Period II.
- (C) The utility shall show the beforetax interest coverage, for the twelve months of Period I based on the indenture requirements. The utility shall provide a copy of the work papers used to make the calculations, with explanations appropriate to understand the calculations.
- (iii) Preferred stock and preference stock capital. (A) This statement shall show the weighted cost for all issues of preferred and preference stock capital as of the end of Period I, as expected on the date the changed rate is filed, and, if applicable, as estimated for the end of Period II. The weighted cost is calculated by: (1) Multiplying the cost of money for each issue under clause (B)(9) by the par amount outstanding for each issue, which yields the annualized cost for each issue; and (2) adding the annual cost of each issue to obtain the total for all issues, which is divided by the total par amount outstanding for all issues to obtain the weighted cost for all issues.
- (B) The statement shall show for each class and issue of preferred and preference stock outstanding as of the end of Period I, as expected on the date the changed rate is filed, and, if applicable, as estimated to be outstanding as of the end of Period II:
 - (1) Title;
 - (2) Date of offering;
 - (3) If callable, call price;
- (4) If convertible, terms of conversion;
- (5) Dividend rate;
- (6) Par or stated amount of issue;
- (7) Net proceeds to the filing utility;
- (8) Ratio of net proceeds to gross proceeds received by the filing utility;

- (9) Cost of money (dividend rate divided by the ratio of net proceeds to gross proceeds for each issue);
- (10) Par or stated amount outstanding: and
- (11) If issue is owned by an affiliate, name and relationship of owner.
- (iv) Common stock capital. This statement shall show the following information for each sale of common stock during the five-year period preceding the date of the balance sheet for the end of Period I and for each sale of common stock between the end of Period I and the date that the changed rate is filed:
 - (A) Number of shares offered;
 - (B) Date of offering;
 - (C) Gross proceeds at offering price;
 - (D) Underwriters' commissions;
 - (E) Dividends per share;
 - (F) Net proceeds to company;
 - (G) Issuance expenses; and
- (H) Whether issue was offered to stockholders through subscription rights or to the public and whether common stock was issued for property or for capital stock of others.
- (v) Supplementary financial data. The utility shall submit a statement indicating the sources and uses of funds for Period I and as estimated for Period II and a copy of the utility's most recent annual report to the stockholders. The utility shall also supply a prospectus for its most recent issue of securities and a copy of the latest prospectus issued by any subsidiary of the filing utility or by any holding company of which the filing utility is a subsidiary.
- (23) Statement AW—Cost of short-term debt. In Statement AW, the utility shall provide a statement of the cost of capital rate for short-term debt of the utility as of the end of Period I, as expected on the date the proposed rate is filed, and, if applicable, as estimated for the end of Period II, with details supporting each stated cost. The short-term debt rate shown in Statement AW shall include only the short-term debt that appears on the income statement as interest expense and shall not include nominal forms of financing, such as trust agreements.
- (24) Statement AX—Other recent and pending rate changes. Statement AX is a statement describing the extent to which operating revenues are subject

to refund for Period I and, if applicable, Period II, for each rate change filed with any Federal, state, or other regulatory body that has jurisidiction. The utility shall list and submit any orders in which applications for a rate increase have been acted on by any regulatory body during Period I, Period II, or the interval between Period I and Period II, and a copy of each transmittal letter or equivalent written document by which a utility summarized and submitted any pending applications that have not been acted on. Statement AX shall reflect information available at the time of submittal under this paragraph. Notwithstanding any other provision of this section, Statement AX is required to be filed only if the proposed rate design tracks retail rates.

- (25) Statement AY—Income and revenue tax rate data. (i) Statement AY is a statement of tax rate data for Period I and Period II arranged as follows:
 - (A) Nominal Federal income tax rate:
 - (B) Nominal state income tax rate;
- (C) Proportion of Federal income taxes payable which is deductible for state income tax purposes. If an allowable deduction is stated in other terms, the utility shall provide an estimate of the effective deduction as a percentage of Federal tax payable; and
- (D) Revenue tax rate. If the revenue tax rate is scaled, the utility shall show approximate weighted average rates for relevant revenue levels and full supporting data.
- (ii) If the utility serves in more than one jurisdiction for revenue or state income tax purposes, the utility shall state the appropriate tax rates for each wholesale customer group at issue and for all other customers as a composite group. [See, Statement BA under paragraph (h)(26) for wholesale customer grouping criteria.] If there are any changes in tax rates that occur in Period I or that may occur in Period II, the utility shall describe such changes and the effective date of the changes.
- (26) Statement BA—Wholesale customer rate groups. (i) Statement BA is a list of wholesale customers by group for the purpose of:
- (A) Allocating the allowable costs of the utility to such customer groups on

the basis of electric utility services rendered; and

- (B) Comparing proposed revenues from each customer group with the cost of service as allocated to that group.
- (ii) The utility shall limit the number of wholesale customer groups listed to the minimum required under the following criteria:
- (A) At least one customer group shall be specified for each separate wholesale rate subject to the changed rate filing.
- (B) In general, all customers proposed to be served on the same rate shall be included in a common group. If the utility believes that there are significant differences in services provided under the same rate, the utility shall subdivide the common group served by the same rate into separate customer groups characterized by the type of service provided each group and shall demonstrate whether the common rate is cost-based by means of cost-justification for each service group. Certain customer groupings, such as cooperatives or municipals, may also be utilized to facilitate purchaser evaluations of the changed rate.
- (C) In all cases, the utility shall select customer groupings on a basis consistent with rate design information provided in Statement BL under paragraph (h)(37) of this section.
- (iii) The utility shall enumerate all wholesale customer rate groups, together with a brief descriptive title for each group. For example:

Group 1. Full Requirements Tariff FR-1.

 $Group\ 2.$ Partial Requirements Tariff PR-1.

- (27) Statement BB—Allocation demand and capability data. Statement BB is a statement of electric utility demand and capability data for Period I and Period II to be considered as a basis for allocating related costs to the wholesale services subject to the changed rate.
- (i) For each month of Period I and Period II, with an average for each period, the utility shall show the maximum peak firm kilowatt demand on the power supply system of the utility, and the kilowatt demands of the wholesale services that coincide with the

system monthly maximum power supply demand, including for Period I the date and hour for such coincidental peak demands. The utility shall state these kilowatt demands in terms of 60minute intervals or other intervals adjusted to the equivalent of 60 minutes. The utility shall not include in the data the demands associated with interruptible power supply services, firm or nonfirm transmission wheeling services, or demands associated with other services the revenues from which are shown as revenue credits in Statement AU under paragraph (h)(21). The utility shall provide wholesale service demand data as follows:

(A) The wholesale service data for each individual customer delivery point or set of delivery points that constitutes an individual wholesale customer billing unit shall include demands at delivery. The individual customer wholesale service data shall be summarized and subtotaled in accordance with Statement BA customer groupings.

(B) The data supplied for each wholesale customer group under clause (A) shall be adjusted for losses to reflect demand at the power supply level. The data shall be totaled to show total customer group demand at power supply level for each month of Period I and Period II.

(ii) To the extent such data are available, the utility shall state Period I and Period II monthly maximum demand data for interruptible power supply services, firm wheeling services, and nonfirm wheeling services. The utility shall also provide, to the extent data are available, firm wheeling demand data for any of the 60-minute periods that coincide with the times of power supply peak demands shown under clause (i). The utility shall indicate the basis of all demands, such as metered demands or contract demands. reported under this clause. For interruptible services, the utility shall provide a description of the conditions under which service may be interrupted or curtailed. The utility shall include available information on actual interruptions or curtailments during a three-year period that includes Period I. If any of the wholesale rates at issue are for interruptible

curtailable service, the utility shall provide any demand data specifically relevant to such service.

(iii) If a utility establishes plant categories in Statement AD under paragraph (h)(4) of this section for the purpose of supporting wholesale rates for firm power supply services with special characteristics, such as base load, intermediate, or peaking, the utility shall provide in Statement BB the demand data required by clause (i) in total and in separate corresponding demand values consistent with the service characteristics. Corresponding values shall be stated for the system demand of the utility, and for each applicable wholesale service group.

(iv) If a utility establishes plant categories in Statement AD under paragraph (h)(4) of this section for the purpose of supporting wholesale rates for nonfirm power supply services, such as capacity sales, the utility shall include in Statement BB for each month of Period I and Period II the monthly capability data relied on by the utility in developing costs allocable to such rates, with an explanation of the underlying cost allocation rationale.

(v) If a utility establishes production plant categories in Statement AD under paragraph (h)(4) of this section for the purpose of supporting wholesale rates based on specialized ratemaking theories such as marginal cost pricing, time-of-day pricing, or base, intermediate, and peaking characteristics, the utility shall include in Statement BB all demand and capability data relied on by the utility in developing support on a cost of service basis, with appropriate explanatory material.

(vi) For each month of Period I and Period II, the utility shall provide any additional demand data that the utility believes to be relevant to the allocation of electric utility costs to the wholesale services at issue. The utility shall fully support all such data and shall explain the rationale and the specific application proposed.

(vii) Based upon information reported in Statements BB and BC, the utility shall list selected months that are normally the months of greatest significance in determining the need of the utility for power supply capability throughout the year. All twelve

months may be selected, if appropriate. In its selection, the utility shall take into account any effects of local weather seasons and, particularly, the extent to which peak demands may tend to be similar in magnitude in two or more months of a weather season. The utility shall explain the reasons for the selections and describe the significance for the selections of seasonal variations in the weather.

(28) Statement BC—Reliability data. Statement BC is a statement relating to reference standards of the filing utility for electric power supply reliability, and to information designed to reflect monthly availability of generating capacity reserves.

- (i) For Period II, Period I, and each of the three calendar years preceding Period I, the utility shall state and briefly explain its objective reference standard of production power supply reliability and the rationale underlying its choice of a reliability standard, including whether it participates with other electric utilities in the selection of a common standard on an area or pool basis. The utility shall identify any such participating utilities, and provide a general explanation of the basis upon which the reliability standard was jointly developed.
- (ii) The utility shall describe how its objective standard for production power supply reliability affects its electric generating facility construction planning and purchased power planning.
- (iii) For the peak day of each month of Period II, Period I, and, to the extent data are available, for the peak day of each month of the three calendar years preceding Period I, the utility shall include tabular schedules designed to show the following:
- (A) Net peak load in megawatts, itemized to show:
- (1) Gross peak firm load, including all firm sales assured available by the reserve capacity of the utility:
- (2) All firm purchases assured available by the reserve capacity of the supplier; and
- (3) Net peak load, computed as gross peak load under clause (1) minus all firm purchases under clause (2).
- (B) Net available dependable capacity, that is, the load-carrying ability of

the electric production facilities determined for the purpose of scheduling capacity in day-to-day operations, provided in megawatts and itemized to show:

- (1) The owned dependable capacity of the utility for each production plant category selected in Statement AD under paragraph (h)(4);
- (2) Scheduled maintenance of owned dependable capacity of the utility;
- (3) Purchased dependable capacity of the utility;
- (4) Scheduled maintenance of purchased dependable capacity of the utility; and
- (5) Net available dependable capacity, computed as the owned dependable capacity under clause (1), minus scheduled maintenance of owned capacity under clause (2), plus purchased dependable capacity under clause (3), minus scheduled maintenance of purchased capacity under clause (4).
- (C) Available reserves in megawatts, which is the net available dependable capacity under clause (iii)(B) minus net peak load under clause (iii)(A).
- (D) Available reserves as a percent of peak load, which is the available reserves under clause (iii)(C) divided by net peak load under clause (iii)(A).
- (29) Statement BD—Allocation energy and supporting data. Statement BD is a statement of electric utility energy data for Period I and Period II to be considered as bases for allocating related costs to the wholesale services subject to the changed rate.
- (i) For each month of Period I and Period II, and as totaled for the twelve months of each period, the utility shall show the megawatt-hours of firm power supply energy required by the system of the utility and the magawatt-hour energy requirements of the wholesale customer groups whose services will be subject to the changed rate. The wholesale service data for each individual customer point or set of delivery points that constitutes an individual wholesale customer billing unit shall include megawatt-hours at delivery. The utility shall summarize and subtotal these individual customer data in accordance Statement BAcustomer with groupings under paragraph (h)(26). The utility shall show a loss adjustment for

each wholesale customer group to reflect energy at the power supply level. The utility shall total the data to show total customer group energy requirements at power supply level for each month of Period I and Period II.

(ii) Data provided under clause (i) shall not include energy associated with interruptible or curtailable services, or energy associated with other services, the revenues from which are shown as revenue credits in Statement AU under paragraph (h)(21) of this section. The utility shall separately state Period I and Period II monthly and total energy data for any such services provided by the utility. If any of the proposed wholesale rates at issue are for interruptible or curtailable service, the utility shall provide descriptive material and energy data specifically relevant to such services.

(iii) If a utility selects subfunctional categories in Statement AD under paragraph (h)(4) of this section for the purpose of supporting any changed wholesale rate for firm power supply services with special characteristics, such as base load, intermediate, and peaking services, the utility shall separate the energy data required by clause (i) into corresponding energy values consistent with the service characteristics and consistent with energy-related expense categories utilized in Statement AH under paragraph (h)(8) of this section. The utility shall state the corresponding values for the utility's system energy and for each applicable wholesale service group.

(iv) If a utility establishes plant categories in Statement AD under paragraph (h)(4) of this section for the purpose of supporting any changed wholesale rate for nonfirm production services, or the changed wholesale rate based on specialized ratemaking theories [see paragraph (h)(27)(v) of this section], the utility shall include in Statement BD all energy data relied on by the utility in developing the support on a cost of service basis and relevant explanatory material. Energy data provided under this clause shall be consistent with related expense categories utilized in Statement AH under paragraph (h)(8) of this section.

(v) For each month of Period I and Period II, and as totaled for the twelve

months of each period, the utility shall show the megawatt-hours generated, itemized in accordance with Statement AD production subfunctional categories, and the megawatt-hours purchased or interchanged, itemized to show each type of transaction, such as firm energy or economy interchanged energy. The utility shall quantitatively reconcile such data with the system allocation energy reported in this statement, and with energy data underlying the fuel and purchased power expense reported in Statement AH.

(30) Statement BE—Specific assignment data. (i) Statement BE is a statement of specific components of the electric costs of service of the utility for Period I and Period II. Statement BE costs of service are those apportioned among wholesale services subject to the rate change and other utility services, on a basis other than:

- (A) Demand, capability, or energy data provided in Statements BB and BD:
- (B) A proportional relationship based on a selected plant category or expense item for which an allocation to wholesale services is to be independently determined; or
- (C) Exclusive-use commitment in Statement BF under paragraph (h)(31) of this section.
- (ii) The utility shall include specific assignments considered appropriate by the utility. Typical cost of service components that could be specifically assigned are distribution plant [see examples listed in Statement AD under paragraph (h)(4) of this section], certain total electric wages and salaries provided in Statement AI under paragraph (h)(9) of this section, such as wages and salaries for customer accounting and for customer service and information, and certain administrative and general expense items. [See examples listed in Statement AH under paragraph (h)(8) of this section.]
- (iii) The utility shall limit specific assignments to the minimum required to adequately provide for costs not otherwise appropriately allocable.
- (iv) For each specific assignment, the utility shall include at least the following information:

- (A) Brief descriptive component title, such as distribution substations or rate case expenses:
 - (B) Total electric amount in dollars;
- (C) Wholesale customer group dollar amounts stated individually for each wholesale customer rate group identified in Statement BA under paragraph (h)(26), and stated in total for all such groups; and
- (D) Explanation of the basis on which assignments were made, accompanied by supporting detailed computations.
- (31) Statement BF—Exclusive-use commitments of major power supply facilities. Statement BF is a statement describing and justifying the commitment to exclusive-use for particular services of all or a stated portion of electric utility generation units or plants, or major transmission facilities.
- (i) For Period I and Period II, the utility shall list each transaction in which all or a stated portion of the output of a specified filing utility-owned generating unit or group of units was committed exclusively to a particular customer or group of customers, or to a power pool or similar power supply entity. For each such transaction, the utility shall provide the following information:
- (A) Brief descriptive title for each commitment;
- (B) Name of plant and unit designation;
- (C) Name of the purchaser or power pool or other similar power supply entity;
 - (D) Duration of the transaction;
- (E) Basis of rates or charges, stated in terms of whether a transaction reflects marginal, incremental, or fully distributed costs, the specific overall and common equity rates of return included in costs, provided on both a claimed and earned basis to the extent such information is available, the approximate date of the cost analysis on which the rates and charges were based, and any other considerations significant to the transaction;
- (F) Revenue received for each month of Period I and Period II or, if applicable, monthly quantities of power and energy received or available from power pools as consideration for commitment to a pool; and

- (G) Proposed treatment in the cost of service determinations for the whole-sale services at issue. For example, a credit of revenue to the total electric cost of service, in Statement AU under paragraph (h)(21), could be proposed to account for unit capacity sales based upon incremental capital costs. The utility shall include explanatory material and support for the proposed procedures.
- (ii) For Period I and Period II, the utility shall list each transaction in which all, or a portion, of a major transmission facility owned by the filing utility was committed exclusively to a particular customer or group of customers. For each such transaction, the utility shall provide information similar to that required by clause (i).
- (32) Statement BG—Revenue data to reflect changed rates. Statement BG is a statement of revenues for Period I and Period II, including those under the changed rate for the wholesale services at issue.
- (i) For each month of Period I and Period II, and in total for each of the two periods, the utility shall show all billing determinants and metered quantities for each delivery point or set of delivery points that constitutes an individual wholesale customer billing unit, and the result of applying each specific rate component to the billing determinants for each billing unit stated with the total of the computed monthly bill for the customer. If the rates include a fuel clause, the utility shall compute and total the revenues under the fuel clause to reflect fuel costs incurred during each month of Period I and Period II. That is, the fuel clause revenues for the first month of Period I shall reflect fuel costs incurred for that month, and so on for each month of Period I and Period II. In computing fuel clause revenues, the utility shall determine fuel cost according to §35.14 of this chapter.
- (ii) If the form of the proposed fuel clause would produce revenues different from those computed in accordance with clause (i), the utility shall separately compute and state such fuel clause revenues for each customer for each month of Period I and Period II.

(iii) The utility shall summarize separately revenue data computed in accordance with clauses (i) and (ii) above for each month and in total for Period I and Period II, in accordance with wholesale rate groups specified in Statement BA under paragraph (h)(26) of this section. The utility shall show total electric department revenues for each period to include revenues under the changed rate for all such wholesale customer rate groups.

(iv) For Period I and as estimated for Period II, the utility shall summarize all billing determinants and revenues from interruptible received curtailable services. Billing determinants and revenue data shall be consistent with interruptible demand and energy data in Statements BB and BD. The utility shall include an explanation of the extent to which interruptible or curtailable service revenues are or are not included in revenue credits in Statement AU under paragraph (h)(21) of this section.

(33) Statement BH—Revenue data to reflect present rates. Statement BH is a statement of revenues for Period I and Period II, including those under present rates for wholesale services at issue, and for total electric service to reflect such revenues for wholesale services. The utility shall prepare this statement to include data consistent with criteria specified for presentation of revenue under the changed rate in Statement BG under paragraph (h)(32) of this section.

(34) Statement BI—Fuel cost adjustment factors. Statement BI is a statement of monthly fuel cost adjustment factors under the changed rate and under the present rates, for Period I and Period II.

(i) If the changed rate schedule embodies a fuel cost adjustment clause, the utility shall show detailed derivations of fuel cost adjustment factors computed to reflect fuel cost incurred during each month of Period I and Period II. Fuel cost adjustment factors are those required for revenue determinations in accordance with paragraph (h)(32)(i) of Statement BG.

(ii) If additional proposed fuel clause revenue data are reported in accordance with paragraph (h)(32)(ii) of Statement BG, the utility shall show detailed derivation of applicable monthly fuel adjustment factors.

(iii) If the present rate includes a fuel cost adjustment change, the utility shall show detailed derivations of fuel cost adjustment factors for each month of Period I and Period II. The utility shall include in Statement BI derivations for all monthly factors required in the computation of present fuel clause revenues reported in Statement BH. The utility shall provide an explanation of the differences between the present and proposed fuel clauses.

(iv) All fuel cost adjustment factors shall be cost-based. The utility shall make a computational showing that shall develop adjustment factors in a manner consistent with the requirements of §35.14 of this chapter. The utility shall provide supporting detail on cost by type of fuel, and shall show separately the allowable fuel clause cost component of purchased or interchanged energy. All fuel cost data shall be consistent with that included in operation and maintenance expenses in Statement AH under paragraph (h)(8) of this section.

(35) Statement BJ—Summary data tables. Statement BJ is a tabular summary of portions of Period I and Period II data from specific cost of service statements in this paragraph. The utility shall summarize under descriptive titles the Period I and Period II data from the cost of service provisions listed in this subparagraph. The utility shall supply the data in the manner described for each cost of service statement and in this subparagraph.

(i) If a utility provides in Statement BK information that is substantially equivalent to the information required in this statement, the utility may fulfill the requirements of this statement by specifically referring to the location in Statement BK of the information required in this subparagraph.

(ii) The utility shall provide the information in the following statements as average total electric department monthly balances for each function and subfunction of plant:

- (A) Statement AD—(h)(4)(i) and (ii);
- (B) Statement AE—(h)(5)(i) and (ii);
- (C) Statement AF—(h)(6)(i) through

- (D) Statement AG—(h)(7)(i) through (vi):
 - (E) Statement AL—(h)(12)(i) and (ii);
 - (F) Statement AM—(h)(13); and
 - (G) Statement AN—(h)(14).
- (iii) The utility shall provide the information in the following statements as total electric department annual revenue and expense amounts:
- (A) Statement AH—(h)(8)(i), (iv) and (v):
- (B) Statement AI—(h)(9)(i) and (ii);
- (C) Statement AJ—(h)(10)(i);
- (D) Statement AK—(h)(11)(i);
- (E) Statement AP—(h)(16)(i) through (iv):
- (F) Statement AQ—(h)(17)(i) through (iii);
- (G) Statement AR—(h)(18)(i) through (iv);
 - (H) Statement AS—(h)(19);
 - (I) Statement AT—(h)(20); and
 - (J) Statement AU—(h)(21).
- (iv) The utility shall provide all cost of capital amounts in the following statements.
 - (A) Statement AV—(h)(22)(i)(A); and
 - (B) Statement AW—(h)(23);
- (v) The utility shall provide all tax rate data in Statement AY, paragraph (h)(25)(i) of this section.
- (vi) The utility shall provide the information in the following statements as appropriate, for total electric department values and individual customer group values:
- (A) Statement BB—(h)(27)(i) through
- (B) Statement BD—(h)(29)(i) through (iv);
- (C) Statement BE—(h)(30)(iv) (A), (B), and (C):
 - (D) Statement BG—(h)(32)(iii); and
 - (E) Statement BH—(h)(33).
- (36) Statement BK—Electric utility department cost of service, total and as allocated. Statement BK is a statement of the claimed fully allocated cost of service of the utility developed and shown for Period I and Period II. The utility shall include analytical support for each rate proposed to be differentiated on a time-of-use basis. The utility shall also provide any marginal or incremental cost information that is required to support the changed rate developed on a marginal or incremental cost basis. The utility shall show allocations of fully distributed

- costs to the wholesale services subject to the changed rate accompanied by a comparison of allocated costs with revenues under the changed rate. Nothing in this subparagraph shall preclude use by any utility of any cost of service technique it believes reasonable and that is consistent with the requirements of paragraph (g) of this section.
- (i) The utility shall base the fully distributed cost of service and the allocations thereof upon data provided in the accompanying detailed statements required under this section and additional data which the utility may submit and support in connection with this statement. The cost of service data of the utility shall conform to the following requirements:
- (A) The total electric rate base and cost of service shall be itemized and summarized by major functions and in a format designed to facilitate review and analysis.
- (B) Based on the total electric rate base and cost of service, and on allocated or assigned component elements, the cost of service for each Statement BA wholesale customer rate group under paragraph (h)(26) shall be itemized and summarized by major functions in a format consistent with that shown for total electric.
- (C) The costs of service data for total electric and for each of the wholesale customer groups shall include data that show the return and the income taxes by components and in total, based upon the rate of return claimed by the utility in Statement AV under paragraph (h)(22). Individual components of income taxes shall include income taxes payable, provision for deferred income tax—debits and deferred income tax—credits, investment tax credits, or other adjustments.
- (D) The fully distributed cost of service study of the utility shall disclose the principal determinants for allocation of total electric costs among the wholesale customer groups, including but not limited to the following:
- (1) Computations showing the energy responsibilities of the wholesale services, with supporting detail;
- (2) Computations showing the demand responsibilities of the wholesale services, with supporting detail; and

- (3) Computations showing the specific assignment responsibilities of the wholesale services, with supporting detail
- (ii) For the total electric service and for each wholesale customer rate group, the utility shall compare the fully distributed cost of service with the revenues under the changed rate. Based on the comparison, the utility shall show the revenue excess or deficiency and the earned rate of return computed for the total electric service and for each wholesale customer rate group.
- (iii) For any filing that contains Period II data, the utility shall supply any work papers and additional explanatory material necessary to support Statement BK, indexed, referenced and paginated as provided in paragraph (d)(5) of this section.
- (iv) The utility shall provide a tabular comparison of Period II total electric fully distributed cost items with those of Period I. The comparisons shall show item amounts for each of the two periods, and also shall show Period II item amounts as percentages of equivalent items for Period I. Comparisons shall include at least the following items, accompanied by explanatory notes with respect to significant variations among the comparative percentages:
 - (A) Rate base;
 - (B) Production expenses;
 - (C) Transmission expenses;
- (D) Customer accounting, customer service and information, and sales expenses;
 - (E) Depreciation expenses;
 - (F) Taxes except income and revenue;
 - (G) Income taxes;
 - (H) Revenue taxes; and
 - (I) Return claimed.
- (37) Statement BL—Rate design information. In support of the design of the changed rate, the utility shall submit the following material:
- (i) A narrative statement describing and justifying the objectives of the design of the changed rate. If the purpose of the rate design is to reflect costs, the utility shall state how that objective is achieved, and shall accompany it with a summary cost analysis that would justify the rate design, including any discounts or surcharges based on

- delivery voltage level or other specific considerations. Such summary cost analysis shall be consistent with, derived from, and cross-referenced to the data in cost of service Statement BK. If the rate design is not intended to reflect costs, whether fully distributed, marginal, incremental, or other, the utility shall provide a statement to justify the departure from cost-based rates.
- (ii) If the billing determinants, such as quantities of demand, energy, or delivery points, are on different bases than the cost allocation determinants supporting such charges, the utility shall submit an explanation setting forth the economic or other considerations that warrant such departure. The information shall include at least the following:
- (A) If the individual rate for the demand, energy and customer charges do not correspond to the comparable cost classifications supporting such charges, a detailed explanation stating the reasons for the differences.
- (B) If the changed rate contains more than one demand or energy block, a detailed explanation indicating the rationale for the blocking and the considerations upon which such blocking is based, including adequate cost support for the specified blocking.
- (38) Statement BM-Construction program statement. Statement BM is a summary of data and supporting assumptions relating to the economics of any construction program to replace or expand the utility's power supply that shall be filed if the utility is filing for construction work in progress in rate base under §35.26(c)(3) of this chapter. The filing utility shall describe generally its program for providing reliable and economic power for the period beginning with the date of the filing and ending with the tenth year after the test period. The statement shall include an assessment of the relative costs of adopting alternative strategies including an analysis of alternative production plant, e.g., cogeneration, small power production, heightened load management and conservation efforts, additions to transmission plant or increased purchases of power, and an

explanation of why the program adopted is prudent and consistent with a least-cost energy supply program.

(Federal Power Act, 16 U.S.C. 791–828c; Dept. of Energy Organization Act, 42 U.S.C. 7101–7352; E.O. 12009, 42 FR 46267, 3 CFR 142 (1978); Pub. L. 96–511, 94 Stat. 2812 (44 U.S.C. 3501 et seq.))

[Order 91, 45 FR 46363, July 10, 1980]

EDITORIAL NOTE: For FEDERAL REGISTER citations affecting §35.13, see the List of CFR Sections Affected, which appears in the Finding Aids section of the printed volume and at www.fdsys.gov.

Subpart C—Other Filing Requirements

§ 35.14 Fuel cost and purchased economic power adjustment clauses.

- (a) Fuel adjustment clauses (fuel clause) which are not in conformity with the principles set out below are not in the public interest. These regulations contemplate that the filing of proposed rate schedules, tariffs or service agreements which embody fuel clauses failing to conform to the following principles may result in suspension of those parts of such rate schedules, tariffs, or service agreements:
- (1) The fuel clause shall be of the form that provides for periodic adjustments per kWh of sales equal to the difference between the fuel and purchased economic power costs per kWh of sales in the base period and in the current period:

Adjustment Factor =Fm/Sm-Fb/Sb

- Where: *F* is the expense of fossil and nuclear fuel and purchased economic power in the base (*b*) and current (*m*) periods; and S is the kWh sales in the base and current periods, all as defined below.
- (2) Fuel and purchased economic power costs (F) shall be the cost of:
- (i) Fossil and nuclear fuel consumed in the utility's own plants, and the utility's share of fossil and nuclear fuel consumed in jointly owned or leased plants.
- (ii) The actual identifiable fossil and nuclear fuel costs associated with energy purchased for reasons other than identified in paragraph (a)(2)(iii) of this section.

- (iii) The total cost of the purchase of economic power, as defined in paragraph (a)(11) of this section, if the reserve capacity of the buyer is adequate independent of all other purchases where non-fuel charges are included in either F_b or F_m ;
- (iv) Energy charges for any purchase if the total amount of energy charges incurred for the purchase is less than the buyer's total avoided variable cost:
- (v) And less the cost of fossil and nuclear fuel recovered through all intersystem sales.
- (3) Sales (S) must be all kWh's sold, excluding inter-system sales. Where for any reason, billed system sales cannot be coordinated with fuel costs for the billing period, sales may be equated to the sum of: (i) Generation, (ii) purchases, (iii) exchange received, less (iv) energy associated with pumped storage operations, less (v) inter-system sales referred to in paragraph (a)(2)(iv) of this section, less (vi) total system losses.
- (4) The adjustment factor developed according to this procedure shall be modified to properly allow for losses (estimated if necessary) associated only with wholesale sales for resale.
- (5) The adjustment factor developed according to this procedure may be further modified to allow the recovery of gross receipts and other similar revenue based tax charges occasioned by the fuel adjustment revenues.
- (6) The cost of fossil fuel shall include no items other than those listed in Account 151 of the Commission's Uniform System of Accounts for Public Utilities and Licensees. The cost of nuclear fuel shall be that as shown in Account 518, except that if Account 518 also contains any expense for fossil fuel which has already been included in the cost of fossil fuel, it shall be deducted from this account. (Paragraph C of Account 518 includes the cost of other fuels used for ancillary steam facilities.)
- (7) Where the cost of fuel includes fuel from company-owned or controlled 1 sources, that fact shall be

¹As defined in the Commission's Uniform System of Accounts 18 CFR part 101, Definitions 5R

noted and described as part of any filing. Where the utility purchases fuel from a company-owned or controlled source, the price of which is subject to the jurisdiction of a regulatory body, and where the price of such fuel has been approved by that regulatory body, such costs shall be presumed, subject to rebuttal, to be reasonable and includable in the adjustment clause. If the current price, however, is in litigation and is being collected subject to refund, the utility shall so advise the Commission and shall keep a separate account of such amounts paid which are subject to refund, and shall advise the Commission of the final disposition of such matter by the regulatory body having jurisdiction. With respect to the price of fuel purchases from companyowned or controlled sources pursuant to contracts which are not subject to regulatory authority, the utility company shall file such contracts and amendments thereto with the Commission for its acceptance at the time it files its fuel clause or modification thereof. Any subsequent amendment to such contracts shall likewise be filed with the Commission as a rate schedule change and may be subject to suspension under section 205 of the Federal Power Act. Fuel charges by affiliated companies which do not appear to be reasonable may result in the suspension of the fuel adjustment clause or cause an investigation thereof to be made by the Commission on its own motion under section 206 of the Federal Power Act.

(8) All rate filings which contain a proposed new fuel clause or a change in an existing fuel clause shall conform such clauses with the regulations. Within one year of the effectiveness of this rulemaking, all public utilities with rate schedules that contain a fuel clause should conform such clauses with the regulations. Recognizing that individual public utilities may have special operating characteristics that warrant granting temporary delays in the implementation of the regulations, the Commission may, upon showing of good cause, waive the requirements of this section of the regulations for an additional one-year period so as to permit the public utilities

sufficient time to adjust to the requirements.

- (9) All rate filings containing a proposed new fuel clause or change in an existing fuel clause shall include:
- (i) A description of the fuel clause with detailed cost support for the base cost of fuel and purchased economic power or energy.
- (ii) Full cost of service data unless the utility has had the rate approved by the Commission within a year, provided that such cost of service may not be required when an existing fuel cost adjustment clause is being modified to conform to the Commission's regulations.
- (10) Whenever particular circumstances prevent the use of the standards provided for herein, or the use thereof would result in an undue burden, the Commission may, upon application under §385.207 of this chapter and for good cause shown, permit deviation from these regulations.
- (11) For the purpose of paragraph (a)(2)(iii) of this section, the following definitions apply:
- (i) *Economic power* is power or energy purchased over a period of twelve months or less where the total cost of the purchase is less than the buyer's total avoided variable cost.
- (ii) Total cost of the purchase is all charges incurred in buying economic power and having such power delivered to the buyer's system. The total cost includes, but is not limited to, capacity or reservation charges, energy charges, adders, and any transmission or wheeling charges associated with the purchase.
- (iii) Total avoided variable cost is all identified and documented variable costs that would have been incurred by the buyer had a particular purchase not been made. Such costs include, but are not limited to, those associated with fuel, start-up, shut-down or any purchases that would have been made in lieu of the purchase made.
- (12) For the purpose of paragraph (a)(2)(iii) of this section, the following procedures and instructions apply:
- (i) A utility proposing to include purchase charges other than those for fuel or energy in fuel and purchased economic power costs (*F*) under paragraph (a)(2)(iii) of this section shall amend its

fuel cost adjustment clause so that it is consistent with paragraphs (a)(1) and (a)(2)(iii) of this section. Such amendment shall state the system reserve capacity criteria by which the system operator decides whether a reliability purchase is required. Where the utility filing the statement is required by a State or local regulatory body (including a plant site licensing board) to file a capacity criteria statement with that body, the system reserve capacity criteria in the statement filed with the Commission shall be identical to those contained in the statement filed with the State or local regulatory body. Any utility that changes its reserve capacity criteria shall, within 45 days of such change, file an amended fuel cost and purchased economic power adjustment clause to incorporate the new criteria.

- (ii) Reserve capacity shall be deemed adequate if, at the time a purchase was initiated, the buyer's system reserve capacity criteria were projected to be satisfied for the duration of the purchase without the purchase at issue.
- (iii) The total cost of the purchase must be projected to be less than total avoided variable cost, at the time a purchase was initiated, before any nonfuel purchase charge may be included in F_m
- (iv) The purchasing utility shall make a credit to F_m after a purchase terminates if the total cost of the purchase exceeds the total avoided variable cost. The amount of the credit shall be the difference between the total cost of the purchase and the total avoided variable cost. This credit shall be made in the first adjustment period after the end of the purchase. If a utility fails to make the credit in the first adjustment period after the end of the purchase, it shall, when making the credit, also include in F_m interest on the amount of the credit. Interest shall be calculated at the rate required by §35.19a(a)(2)(iii) of this chapter, and shall accrue from the date the credit should have been made under this paragraph until the date the credit is made.
- (v) If a purchase is made of more capacity than is needed to satisfy the buyer's system reserve capacity criteria because the total costs of the extra capacity and associated energy

are less than the buyer's total avoided variable costs for the duration of the purchase, the charges associated with the non-reliability portion of the purchase may be included in F.

(Approved by the Office of Management and Budget under control number 1902–0096)

(Federal Power Act, 16 U.S.C. 824d, 824e and 825h (1976 & Supp. IV 1980); Department of Energy Organization Act, 42 U.S.C. 7171, 7172 and 7173(c) (Supp. IV 1980); E.O. 12009, 3 CFR part 142 (1978); 5 U.S.C. 553 (1976))

[Order 271, 28 FR 10573, Oct. 2, 1963, as amended by Order 421, 36 FR 3047, Feb. 17, 1971; 39 FR 40583, Nov. 19, 1974; Order 225, 47 FR 19056, May 3, 1982; Order 352, 48 FR 55436, Dec. 13, 1983; 49 FR 5073, Feb. 10, 1984; Order 529, 55 FR 47321, Nov. 13, 1990; Order 600, 63 FR 53809, Oct. 7, 1998; Order 714, 73 FR 57532, Oct. 3, 2008; 73 FR 63886, Oct. 28, 2008]

§ 35.15 Notices of cancellation or termination.

- (a) General rule. When a rate schedule, tariff or service agreement or part thereof required to be on file with the Commission is proposed to be cancelled or is to terminate by its own terms and no new rate schedule, tariff or service agreement or part thereof is to be filed in its place, a filing must be made to cancel such rate schedule, tariff or service agreement or part thereof at least sixty days but not more than one hundred-twenty days prior to the date such cancellation or termination is proposed to take effect. A copy of such notice to the Commission shall be duly posted. With such notice, each filing party shall submit a statement giving the reasons for the proposed cancellation or termination, and a list of the affected purchasers to whom the notice has been provided. For good cause shown, the Commission may by order provide that the notice of cancellation or termination shall be effective as of a date prior to the date of filing or prior to the date the filing would become effective in accordance with these rules.
- (b) Applicability. (1) The provisions of paragraph (a) of this section shall apply to all contracts for unbundled transmission service and all power sale contracts:
- (i) Executed prior to July 9, 1996; or
- (ii) If unexecuted, filed with the Commission prior to July 9, 1996.

- (2) Any power sales contract executed on or after July 9, 1996 that is to terminate by its own terms shall not be subject to the provisions of paragraph (a) of this section.
- (c) Notice. Any public utility providing jurisdictional services under a power sales contract that is not subject to the provisions of paragraph (a) of this section shall notify the Commission of the date of the termination of such contract within 30 days after such termination takes place.

[Order 888, 61 FR 21692, May 10, 1996, as amended by Order 714, 73 FR 57532, Oct. 3, 2008]

§35.16 Notice of succession.

Whenever the name of a public utility is changed, or its operating control is transferred to another public utility in whole or in part, or a receiver or trustee is appointed to operate any public utility, the exact name of the public utility, receiver, or trustee which will operate the property thereafter shall be filed within 30 days thereafter with the Commission with a tariff consistent with the electronic filing requirements in §35.7 of this part.

[Order 271, 28 FR 10573, Oct. 2, 1963, as amended by Order 714, 73 FR 57533, Oct. 3, 2008]

§ 35.17 Withdrawals and amendments of rate schedule, tariff or service agreement filings.

- (a) Withdrawals of rate schedule, tariff or service agreement filings prior to Commission action. (1) A public utility may withdraw in its entirety a rate schedule, tariff or service agreement filing that has not become effective and upon which no Commission or delegated order has been issued by filing a withdrawal motion with the Commission. Upon the filing of such motion, the proposed rate schedule, tariff or service agreement sections will not become effective under section 205(d) of the Federal Power Act in the absence of Commission action making the rate schedule, tariff or service agreement filing
- (2) The withdrawal motion will become effective, and the rate schedule, tariff or service agreement filing will be deemed withdrawn, at the end of 15 days from the date of filing of the withdrawal motion, if no answer in opposi-

tion to the withdrawal motion is filed within that period and if no order disallowing the withdrawal is issued within that period. If an answer in opposition is filed within the 15 day period, the withdrawal is not effective until an order accepting the withdrawal is issued.

- (b) Amendments or modifications to rate schedule, tariff or service agreement sections prior to Commission action on the filing. A public utility may file to amend or modify, and may file a settlement that would amend or modify, a rate schedule, tariff or service agreement section contained in a rate schedule, tariff or service agreement filing that has not become effective and upon which no Commission or delegated order has yet been issued. Such filing will toll the notice period in section 205(d) of the Federal Power Act for the original filing, and establish a new date on which the entire filing will become effective, in the absence of Commission action, no earlier than 61 days from the date of the filing of the amendment or modification.
- (c) Withdrawal of suspended rate schedules, tariffs, or service agreements, or parts thereof. Where a rate schedule, tariff, or service agreement, or part thereof has been suspended by the Commission, it may be withdrawn during the period of suspension only by special permission of the Commission granted upon application therefor and for good cause shown. If permitted to be withdrawn, any such rate schedule, tariff, or service agreement may be refiled with the Commission within a one-year period thereafter only with special permission of the Commission for good cause shown.
- (d) Changes in suspended rate schedules, tariffs, or service agreements, or parts thereof. A public utility may not, within the period of suspension, file any change in a rate schedule, tariff, or service agreement, or part thereof, which has been suspended by order of the Commission except by special permission of the Commission granted upon application therefor and for good cause shown.
- (e) Changes in rate schedules or tariffs or parts thereof continued in effect and which were proposed to be changed by the suspended filing. A public utility may

not, within the period of suspension, file any change in a rate schedule or tariff or part thereof continued in effect by operation of an order of suspension and which was proposed to be changed by the suspended filing, except by special permission of the Commission granted upon application therefor and for good cause shown.

[Order 271, 28 FR 10573, Oct. 2, 1963, as amended by Order 714, 73 FR 57533, Oct. 3, 2008; 74 FR 55770, Oct. 29, 2009]

§35.18 Asset retirement obligations.

- (a) A public utility that files a rate schedule, tariff or service agreement under §35.12 or §35.13 and has recorded an asset retirement obligation on its books must provide a schedule, as part of the supporting work papers, identifying all cost components related to the asset retirement obligations that are included in the book balances of all accounts reflected in the cost of service computation supporting the proposed rates. However, all cost components related to asset retirement obligations that would impact the calculation of rate base, such as electric plant and related accumulated depreciation and accumulated deferred income taxes, may not be reflected in rates and must be removed from the rate base calculation through a single adjustment.
- (b) A public utility seeking to recover nonrate base costs related to asset retirement costs in rates must provide, with its filing under §35.12 or §35.13, a detailed study supporting the amounts proposed to be collected in rates.
- (c) A public utility that has recorded asset retirement obligations on its books, but is not seeking recovery of the asset retirement costs in rates, must remove all asset-retirement-obligations-related cost components from the cost of service supporting its proposed rates.

[Order 631, 68 FR 19619, Apr. 21, 2003, as amended by Order 714, 73 FR 57533, Oct. 3, 2008]

§ 35.19 Submission of information by reference.

If all or any portion of the information called for in this part has already been submitted to the Commission, substantially in the form prescribed above, specific reference thereto may be made in lieu of re-submission in response to the requirements of this part.

§35.19a Refund requirements under suspension orders.

- (a) Refunds. (1) The public utility whose proposed increased rates or charges were suspended shall refund at such time in such amounts and in such manner as required by final order of the Commission the portion of any increased rates or charges found by the Commission in that suspension proceeding not to be justified, together with interest as required in paragraph (a)(2) of this section.
- (2) Interest shall be computed from the date of collection until the date refunds are made as follows:
- (i) At a rate of seven percent simple interest per annum on all excessive rates or charges held prior to October 10. 1974:
- (ii) At a rate of nine percent simple interest per annum on all excessive rates or charges held between October 10, 1974, and September 30, 1979; and
- (iii)(A) At an average prime rate for each calendar quarter on all excessive rates or charges held (including all interest applicable to such rates or charges) on or after October 1, 1979. The applicable average prime rate for each calendar quarter shall be the arithmetic mean, to the nearest one-hundredth of one percent, of the prime rate values published in the Federal Reserve Bulletin, or in the Federal Reserve Bulletin, or in the Federal Reserve's "Selected Interest Rates" (Statistical Release H. 15), for the fourth, third, and second months preceeding the first month of the calendar quarter
- (B) The interest required to be paid under clause (iii)(A) shall be compounded quarterly.
- (3) Any public utility required to make refunds pursuant to this section shall bear all costs of such refunding.
- (b) Reports. Any public utility whose proposed increased rates or charges were suspended and have gone into effect pending final order of the Commission pursuant to section 205(e) of the Federal Power Act shall keep accurate account of all amounts received under the increased rates or charges which

became effective after the suspension period, for each billing period, specifiying by whom and in whose behalf such amounts are paid.

[44 FR 53503, Sept. 14, 1979, as amended at 45 FR 3889, Jan. 21, 1980; Order 545, 57 FR 53990, Nov. 16, 1992; 74 FR 54463, Oct. 22, 2009]

§ 35.21 Applicability to licensees and others subject to section 19 or 20 of the Federal Power Act.

Upon further order of this Commission issued upon its own motion or upon complaint or request by any person or State within the meaning of sections 19 or 20 of the Federal Power Act, the provisions of §§35.1 through 35.19 shall be operative as to any licensee or others who are subject to this Commission's jurisdiction in respect to services and the rates and charges of payment therefor by reason of the requirements of sections 19 or 20 of the Federal Power Act. The requirement of this section for compliance with the provisions of §§35.1 through 35.19 shall be in addition to and independent of any obligation for compliance with those regulations by reason of the provisions of sections 205 and 206 of the Federal Power Act. For purposes of applying this section Electric Service as otherwise defined in §35.2(a) shall mean: Services to customers or consumers of power within the meaning of sections 19 or 20 of the Federal Power Act which may be comprised of various classes of capacity and energy and/or transmission services subject to the jurisdiction of this Commission. Electric Service shall include the utilization of facilities owned or operated by any licensee or others to effect any of the foregoing sales or services whether by leasing or other arrangements. As defined herein *Electric Service* is without regard to the form of payment or compensation for the sales or services rendered, whether by purchase and sale, exchange, interchange, wheeling charge, facilities charge, rental or otherwise. For purposes of applying this section, Rate Schedule as otherwise defined in §35.2(b) shall mean: A state-

- (1) Electric service as defined in this §35.21,
- (2) Rates and charges for or in connection with that service, and

(3) All classifications, practices, rules, regulations, or contracts which in any manner affect or relate to the aforementioned service, rates and charges. This statement shall be in writing and may take the physical form of a contractual document, purchase or sale agreement, lease of facilities, tariff⁵ or other writing. Any oral agreement or understanding forming a part of such statement shall be reduced to writing and made a part thereof.

[Order 271, 28 FR 10573, Oct. 2, 1963, as amended by Order 714, 73 FR 57533, Oct. 3, 2008]

§ 35.22 Limits for percentage adders in rates for transmission services; revision of rate schedules, tariffs or service agreements.

- (a) Applicability. This section applies to all electric rate schedules, tariffs or service agreements required to be filed under this part that are used for transactions in which the utility or system performs a transmission or purchase and resale function.
- (b) Definition. For purposes of this section, purchased power price means the amount paid by a utility or system that performs a transmission or purchase and resale function for electric power generated by another utility or system.
- (c) General rule. (1) If a utility or system uses a rate component that recovers revenues computed wholly or in part as a percentage of the purchased power price, the utility or system shall establish a limit on the revenues recovered by such rate component in any transaction, in accordance with paragraph (d) of this section.
- (2) The limit established under this paragraph shall be stated in mills per kilowatt-hour.
- (d) Cost support information. (1) A utility or system shall submit cost support information to justify any revenue limit established under paragraph (c) of this section, except as provided in paragraph (e) of this section.
- (2) The information submitted under this section shall consist of those costs, other than the purchased power price, incurred by a utility or system as a result of a transmission or purchase and resale transaction, which

⁵ See § 35.2.

costs are not recovered under any other rate component.

- (e) Exception. A utility or system need not submit the cost support information required under paragraph (d) of this section if the limit established under paragraph (c) of this section is not more than one mill per kilowatthour.
- (f) Revision of rate schedules, tariffs or service agreements. Every utility or system shall:
- (1) Amend any rate schedule, tariffs or service agreements to indicate any limit established pursuant to this section, not later than 60 days after the effective date of this rule; and
- (2) Hereafter conform any rate or rate change filed under this part to the requirements of this section.

(Federal Power Act, as amended, 16 U.S.C. 792–828c; Department of Energy Organization Act, 42 U.S.C. 7101–7352; E.O. 12009, 3 CFR 142 (1978))

[Order 84, 45 FR 31300, May 13, 1980. Redesignated by Order 545, 57 FR 53990, Nov. 16, 1992, as amended by Order 714, 73 FR 57533, Oct. 3, 20081

§35.23 General provisions.

- (a) Applicability. This subpart applies to any wholesale sale of electric energy in a coordination transaction by a public utility if that sale requires the use of an emissions allowance.
- (b) Implementation Procedures. (1) If a public utility has a coordination rate schedule on file that expressly provides for the recovery of all incremental or out-of-pocket costs, such utility may make an abbreviated rate filing detailing how it will recover emissions allowance costs. Such filing must include the following: the index or combination of indices to be used; the method by which the emission allowance amounts will be calculated; timing procedures; how inconsistencies, if any, with dispatch criteria will be reconciled; and how any other rate impacts will be addressed. In addition, a utility making an abbreviated filing must:
- (i) Clearly identify the filing as being limited to an amendment to a coordination rate to reflect the cost of emissions allowances, in the first paragraph of the letter of transmittal accompanying the filing;

- (ii) Submit the revisions in accordance with §35.7; and
- (iii) Identify each rate schedule to which the amendment applies.
- (2) The abbreviated filing must apply consistent treatment to all coordination rate schedules. If the filing does not apply consistent rate treatment, the public utility must explain why it does not do so.
- (3) If a public utility wants to charge incremental costs for emissions allowances, but its rate schedule on file with the Commission does not provide for the recovery of all incremental costs, the selling public utility may submit an abbreviated filing if all customers agree to the rate change. If customers do not agree, the selling public utility must tender its emissions allowance proposal in a separate section 205 rate filing, fully justifying its proposal.

[59 FR 65938, Dec. 22, 1994, as amended by Order 714, 73 FR 57533, Oct. 3, 2008]

§ 35.24 Tax normalization for public utilities.

- (a) Applicability. (1) Except as provided in subparagraph (2) of this paragraph, this section applies, with respect to rate schedules filed under §§ 35.12 and 35.13 of this part, to the ratemaking treatment of the tax effects of all transactions for which there are timing differences.
- (2) This section does not apply to the following timing differences:
- (i) Differences that result from the use of accelerated depreciation;
- (ii) Differences that result from the use of Class Life Asset Depreciation Range (ADR) provisions of the Internal Revenue Code;
- (iii) Differences that result from the use of accelerated amortization provisions on certified defense and pollution control facilities:
- (iv) Differences that arise from recognition of extraordinary property losses as a current expense for tax purposes but as a deferred and amortized expense for book purposes;
- (v) Differences that arise from recognition of research, development, and demonstration expenditures as a current expense for tax purposes but as a deferred and amortized expense for book purposes;

- (vi) Differences that result from different tax and book reporting of deferred gains or losses from disposition of utility plant;
- (vii) Differences that result from the use of the Asset Guideline Class "Repair Allowance" provision of the Internal Revenue Code;
- (viii) Differences that result from recognition of purchased gas costs as a current expense for tax purposes but as a deferred expense for book purposes.
- (See Order 13, issued October 18, 1978; Order 203, issued May 29, 1958; Order 204, issued May 29, 1958; Order 404, issued May 15, 1970; Order 408, issued August 26, 1970; Order 432, issued April 23, 1971; Order 504, issued February 11, 1974; Order 505, issued February 11, 1974; Order 566, issued June 3, 1977; Opinion 578, issued June 3, 1970; and Opinion 801, issued May 31, 1977.)
- (b) General rules—1) Tax normalization required. (i) A public utility must compute the income tax component of its cost of service by using tax normalization for all transactions to which this section applies.
- (ii) Except as provided in paragraph (c) of this section, application of tax normalization by a public utility under this section to compute the income tax component will not be subject to case-by-case adjudication.
- (2) Reduction of, and addition to, rate base. (i) The rate base of a public utility using tax normalization under this section must be reduced by the balances that are properly recordable in Account 281, "Accumulated deferred income taxes-accelerated amortization property;" Account 282, "Accumulated deferred income taxes—other property;" and Account 283, "Accumulated deferred income taxes—other." Balances that are properly recordable in Account 190, "Accumulated deferred income taxes," must be treated as an addition to rate base.
- (ii) Such rate base reductions or additions must be limited to deferred taxes related to rate base, construction or other jurisdictional activities.
- (iii) If a public utility uses an approved purchased gas adjustment clause or a research, development and demonstration tracking clause, the rate base reductions or additions required under this subparagraph must apply only to the extent that the bal-

- ances in Account 190 and Accounts 281 through 283 are not used, for purposes of calculating carrying charges, as an offset to balances properly recordable in Account 188, "Research development and demonstration expenditures," or Account 191, "Unrecovered purchased gas costs."
- (c) Special rules. (1) This paragraph applies:
- (i) If the public utility has not provided deferred taxes in the same amount that would have accrued had tax normalization been applied for the tax effects of timing difference transactions originating at any time prior to the test period; or
- (ii) If, as a result of changes in tax rates, the accumulated provision for deferred taxes becomes deficient in or in excess of amounts necessary to meet future tax liabilities as determined by application of the current tax rate to all timing difference transactions originating in the test period and prior to the test period.
- (2) The public utility must compute the income tax component in its cost of service by making provision for any excess or deficiency in deferred taxes described in subparagraphs (1)(i) or (1)(ii) of this paragraph.
- (3) The public utility must apply a Commission-approved ratemaking method made specifically applicable to the public utility for determining the cost of service provision described in subparagraph (2) of this paragraph. If no Commission-approved ratemaking method has been made specifically applicable to the public utility, then the public utility must use some ratemaking method for making such provision, and the appropriateness of this method will be subject to case-by-case determination.
- (d) *Definitions*. For purposes of this section, the term:
- (1) Tax normalization means computing the income tax component as if the amounts of timing difference transactions recognized in each period for ratemaking purposes were also recognized in the same amount in each such period for income tax purposes.
- (2) Timing differences means differences between amounts of expenses or revenues recognized for income tax purposes and amounts of expenses or

revenues recognized for ratemaking purposes, which differences arise in one time period and reverse in one or more other time periods so that the total amounts of expenses or revenues recognized for income tax purposes and for ratemaking purposes are equal.

- (3) Commission-approved ratemaking method means a ratemaking method approved by the Commission in a final decision including approval of a settlement agreement containing a ratemaking method only if such settlement agreement applies that method beyond the effective term of the settlement agreement.
- (4) Income tax purposes means for the purpose of computing income tax under the provisions of the Internal Revenue Code or the income tax provisions of the laws of a State or political subdivision of a State (including franchise taxes).
- (5) *Income tax component* means that part of the cost of service that covers income tax expenses allowable by the Commission.
- (6) Ratemaking purposes means for the purpose of fixing, modifying, approving, disapproving or rejecting rates under the Federal Power Act or the Natural Gas Act.
- (7) Tax effect means the tax reduction or addition associated with a specific expense or revenue transaction.
- (8) Transaction means an activity or event that gives rise to an accounting entry that is used in determining revenues or expenses.

[46 FR 26636, May 14, 1981. Redesignated and amended by Order 144–A, 47 FR 8342, Feb. 26, 1982; Redesignated by Order 545, 57 FR 53990, Nov. 16, 1992]

§ 35.25 Construction work in progress.

- (a) Applicability. This section applies to any rate schedule filed under this part by any public utility as defined in subsection 201(e) of the Federal Power Act.
- (b) Definitions. For purposes of this section:
- (1) Construction work in progress or CWIP means any expenditure for public utility plant in process of construction that is properly included in Accounts 107 (construction work in progress) and 120.1 (nuclear fuel in process of refinement, conversion, enrichment, and fab-

rication) of part 101 of this chapter, the Uniform System of Accounts Prescribed for Public Utilities and Licensees Subject to the Provisions of the Federal Power Act (Major and Nonmajor), that would otherwise be eligible for allowance for funds used during construction (AFUDC) treatment.

- (2) Double whammy means a situation which may arise when a wholesale electric rate customer embarks upon its own or participates in a construction program to supply itself with all or a portion of its future power needs, thereby reducing its future dependence on the CWIP of the rate applicant, but is simultaneously forced to pay to the CWIP public utility rate applicant the CWIP portion of the wholesale rates that reflects existing levels of service or a different anticipated service level.
- (3) Fuel conversion facility means any addition to public utility plant that enables a natural gas-burning plant to convert to the use of other fuels, or that enables an oil-burning plant to convert to the use of other fuels, other than natural gas. Such facilities include those that alter internal plant workings, such as oil or coal burners, soot blowers, bottom ash removal systems and concomitant air pollution control facilities, and any facility needed for receiving and storing the fuel to which the plant is being converted, which facility would not be necessary if the plant continued to burn gas or oil.
- (4) Pollution control facility means an identifiable structure or portions of a structure that is designed to reduce the amount of pollution produced by the power plant, but does not include any facility that reduces pollution by substituting a different method of generation or that generates the additional power necessitated by the operation of a pollution control facility.
- (c) General rule. For purposes of any initial rate schedule or any rate schedule change filed under §35.12 or §35.13 of this part, a public utility may include in its rate base any costs of construction work in progress (CWIP), including allowance for funds used during construction (AFUDC), as provided in this section.

- (1) Pollution control facilities—(i) General rule. Any CWIP for pollution control facilities allocable to electric power sales for resale may be included in the rate base of the public utility.
- (ii) Qualification as a pollution control facility. In determining whether a facility is a pollution control facility for purposes of this section, the Commission will consider:
- (A) Whether such facility is the type facility described in the Internal Revenue Service laws, 26 U.S.C. 169(d)(1), as follows:
- "A new identifiable treatment facility which is used * * * to abate or control water or atmospheric pollution or contamination by removing, altering, disposing, storing, or preventing the creation or emission of pollutants, contaminants, wastes or heat":
- (B) Whether such facility has been certified by a local, state, or federal agency as being in conformity with, or required by, a program of pollution control:
- (C) Other evidence showing that such facilities are for pollution control.
- (2) Fuel conversion facilities. Any CWIP for fuel conversion facilities allocable to electric power sales for resale may be included in the rate base of the public utility.
- (3) Non-pollution control of fuel conversion (non-PC/FC) CWIP. No more than 50 percent of any CWIP allocable to electric power sales for resale not otherwise included in rate base under paragraphs (c) (1) and (2) of this section, may be included in the rate base of the public utility.
- (4) Forward looking allocation ratios. Every test period CWIP project requested for wholesale rate base treatment pursuant to \$35.26(c)(1). (2), and (3) of this part will be allocated to the customer classes on the basis of forward looking allocation ratios reflecting the anticipated average annual use the wholesale customers will make of the system over the estimated service life of the project. Supporting documentation, as required by §§ 35.12 and 35.13 of this part, must be in sufficient detail to permit examination and verification of the forward looking allocation ratio's recognition of each wholesale customer's plans, if any, for future alternative or supplementary power supplies. For the purpose of pre-

- venting anticompetitive effects, including CWIP-induced price squeeze and double whammy, sufficient recognition of such plans may require the public utility applicant to provide for separate customer groups or provide for a rate design incorporating selected CWIP project credits.
- (d) Effective date. If a public utility proposes in its filed rates to include CWIP in rate base under this section, that portion of the rate related to CWIP is collectable at the time the general rates become effective pursuant to Commission order, whether or not subject to refund, except as provided in paragraph (g) of this section.
- (e) Discontinuance of AFUDC. On the date that any proposed rate that includes CWIP in rate base becomes effective, a public utility that has included CWIP in rate base must discontinue the capitalization of any AFUDC related to those amounts of CWIP is rate base.
- (f) Accounting procedures. When a public utility files to include CWIP in its rate base pursuant to this section, it must propose accounting procedures in that rate schedule filing that:
- (1) Ensure that wholesale customers will not be charged for both capitalized AFUDC and corresponding amounts of CWIP proposed to be included in rate base: and
- (2) Ensure that wholesale customers will not be charged for any corresponding AFUDC capitalized as a result of different accounting or ratemaking treatments accorded CWIP by state or local regulatory authorities.
- (g) Anticompetitive procedures—(1) Filing requirements. In order to facilitate Commission review of the anticompetitive effects of applications for CWIP pursuant to §35.26(c)(3), a public utility applying for rates based upon inclusion of such CWIP in rate base must include the following information in its filing:
- (i) The percentage of the proposed increase in the jurisdictional rate level attributable to non-pollution control/fuel conversion CWIP and the percentage of non-pollution control/fuel conversion CWIP supporting the proposed rate level:

- (ii) The percentage of non-pollution control/fuel conversion CWIP permitted by the state or local commission supporting the current retail rates of the public utility against which the relevant wholesale customers compete; and
- (iii) Individual earned rate of return analyses of each of the competing retail rates developed on a basis fully consistent with the wholesale cost of service for the same test period if the requested percentage of wholesale non-pollution control/fuel conversion CWIP exceeds that permitted by the relevant state or local authority to support the currently competing retail rates.
- (2) Preliminary relief. (i) If an intervenor in its initial pleading alleges that a price squeeze will occur as a direct result of the public utility's request for CWIP pursuant to §35.26(c)(3), makes a showing that it is likely to incur harm if such CWIP is allowed subject to refund, and makes a showing of how the harm to the intervenor would be mitigated or eliminated by the types of preliminary relief requested, the Commission will consider preliminary relief at the suspension stage of the case pursuant to paragraph (g)(4) of this section. In determining whether to grant preliminary relief, the Commission will balance the following public interest considerations:
- (A) The harm to the intervenor if it is not granted preliminary relief from the requested CWIP:
- (B) The harm to the public utility if, during the interim period of preliminary relief, the public utility is required to recover its financing charges later through AFUDC rather than immediately through CWIP; and
- (C) Mitigating bias against investment in new plants, ensuring accurate price signals, and fostering rate stability.
- (ii) Whether or not preliminary relief is granted at the suspension stage will not preclude consideration of further interim or final remedies later in the preceedings, if warranted.
- (3) If the Commission makes a final determination that a price squeeze due solely to allowance of a lower percentage of non-pollution control/fuel conversion CWIP in the public utility's retail rate base than allowed by this

- Commission, the Commission will consider an adjustment to non-pollution control/fuel conversion CWIP in order to eliminate or mitigate the price squeeze.
- (4) If an intervenor meets the requirements of paragraph (g)(2) of this section, the Commission, depending on the type of showing made including the likelihood, immediacy, and severity of any anticompetitive harm, may:
- (i) Suspend the entire rate increase or all or a portion of the non-pollution control/fuel conversion CWIP component for up to five months;
- (ii) Allow all or a portion of the nonpollution control/fuel conversion CWIP only prospectively from the issuance of the Commission's final order on rehearing on the matter; or
- (iii) Take such other action as is proper under the circumstances.

[Order 474, 52 FR 23965, June 26, 1987, as amended by Order 474-A, 52 FR 35702, Sept. 23, 1987; Order 474-B, 54 FR 32804, Aug. 10, 1989. Redesignated by Order 545, 57 FR 53990, Nov. 16, 1992, as amended by Order 626, 67 FR 36096, May 23, 2002]

§ 35.26 Recovery of stranded costs by public utilities and transmitting utilities.

- (a) *Purpose*. This section establishes the standards that a public utility or transmitting utility must satisfy in order to recover stranded costs.
- (b) Definitions—1) Wholesale stranded cost means any legitimate, prudent and verifiable cost incurred by a public utility or a transmitting utility to provide service to:
- (i) A wholesale requirements customer that subsequently becomes, in whole or in part, an unbundled wholesale transmission services customer of such public utility or transmitting utility; or
- (ii) A retail customer that subsequently becomes, either directly or through another wholesale transmission purchaser, an unbundled wholesale transmission services customer of such public utility or transmitting utility.
- (2) Wholesale requirements customer means a customer for whom a public utility or transmitting utility provides by contract any portion of its bundled wholesale power requirements.

- (3) Wholesale transmission services means the transmission of electric energy sold, or to be sold, at wholesale in interstate commerce or ordered pursuant to section 211 of the Federal Power Act (FPA).
- (4) Wholesale requirements contract means a contract under which a public utility or transmitting utility provides any portion of a customer's bundled wholesale power requirements.
- (5) Retail stranded cost means any legitimate, prudent and verifiable cost incurred by a public utility to provide service to a retail customer that subsequently becomes, in whole or in part, an unbundled retail transmission services customer of that public utility.
- (6) Retail transmission services means the transmission of electric energy sold, or to be sold, in interstate commerce directly to a retail customer.
- (7) New wholesale requirements contract means any wholesale requirements contract executed after July 11, 1994, or extended or renegotiated to be effective after July 11, 1994.
- (8) Existing wholesale requirements contract means any wholesale requirements contract executed on or before July 11, 1994
- (c) Recovery of wholesale stranded costs—1) General requirement. A public utility or transmitting utility will be allowed to seek recovery of wholesale stranded costs only as follows:
- (i) No public utility or transmitting utility may seek recovery of wholesale stranded costs if such recovery is explicitly prohibited by a contract or settlement agreement, or by any power sales or transmission rate schedule or tariff.
- (ii) No public utility or transmitting utility may seek recovery of stranded costs associated with a new wholesale requirements contract if such contract does not contain an exit fee or other explicit stranded cost provision.
- (iii) If wholesale stranded costs are associated with a new wholesale requirements contract containing an exit fee or other explicit stranded cost provision, and the seller under the contract is a public utility, the public utility may seek recovery of such costs, in accordance with the contract, through rates for electric energy under sections 205–206 of the FPA. The public utility

- may not seek recovery of such costs through any transmission rate for FPA section 205 or 211 transmission services.
- (iv) If wholesale stranded costs are associated with a new wholesale requirements contract, and the seller under the contract is a transmitting utility but not also a public utility, the transmitting utility may not seek an order from the Commission allowing recovery of such costs.
- (v) If wholesale stranded costs are associated with an existing wholesale requirements contract, if the seller under such contract is a public utility, and if the contract does not contain an exit fee or other explicit stranded cost provision, the public utility may seek recovery of stranded costs only as follows:
- (A) If either party to the contract seeks a stranded cost amendment pursuant to a section 205 or section 206 filing under the FPA made prior to the expiration of the contract, and the Commission accepts or approves an amendment permitting recovery of stranded costs, the public utility may seek recovery of such costs through FPA section 205–206 rates for electric energy.
- (B) If the contract is not amended to permit recovery of stranded costs as described in paragraph (c)(1)(v)(A) of this section, the public utility may file a proposal, prior to the expiration of the contract, to recover stranded costs through FPA section 205–206 or section 211–212 rates for wholesale transmission services to the customer.
- (vi) If wholesale stranded costs are associated with an existing wholesale requirements contract, if the seller under such contract is a transmitting utility but not also a public utility, and if the contract does not contain an exit fee or other explicit stranded cost provision, the transmitting utility may seek recovery of stranded costs through FPA section 211–212 transmission rates.
- (vii) If a retail customer becomes a legitimate wholesale transmission customer of a public utility or transmitting utility, e.g., through municipalization, and costs are stranded as a result of the retail-turned-wholesale customer's access to wholesale transmission, the utility may seek

recovery of such costs through FPA section 205–206 or section 211–212 rates for wholesale transmission services to that customer.

- (2) Evidentiary demonstration for wholesale stranded cost recovery. A public utility or transmitting utility seeking to recover wholesale stranded costs in accordance with paragraphs (c)(1) (v) through (vii) of this section must demonstrate that:
- (i) It incurred costs to provide service to a wholesale requirements customer or retail customer based on a reasonable expectation that the utility would continue to serve the customer;
- (ii) The stranded costs are not more than the customer would have contributed to the utility had the customer remained a wholesale requirements customer of the utility, or, in the case of a retail-turned-wholesale customer, had the customer remained a retail customer of the utility; and
- (iii) The stranded costs are derived using the following formula: Stranded Cost Obligation = (Revenue Stream Estimate—Competitive Market Value Estimate) × Length of Obligation (reasonable expectation period).
- (3) Rebuttable presumption. If a public utility or transmitting utility seeks recovery of wholesale stranded costs associated with an existing wholesale requirements contract, as permitted in paragraph (c)(1) of this section, and the existing wholesale requirements contract contains a notice provision, there will be a rebuttable presumption that the utility had no reasonable expectation of continuing to serve the customer beyond the term of the notice provision.
- (4) Procedure for customer to obtain stranded cost estimate. A customer under an existing wholesale requirements contract with a public utility seller may obtain from the seller an estimate of the customer's stranded cost obligation if it were to leave the public utility's generation supply system by filing with the public utility a request for an estimate at any time prior to the termination date specified in its contract.
- (i) The public utility must provide a response within 30 days of receiving the request. The response must include:

- (A) An estimate of the customer's stranded cost obligation based on the formula in paragraph (c)(2)(iii) of this section:
- (B) Supporting detail indicating how each element in the formula was derived:
- (C) A detailed rationale justifying the basis for the utility's reasonable expectation of continuing to serve the customer beyond the termination date in the contract;
- (D) An estimate of the amount of released capacity and associated energy that would result from the customer's departure; and
- (E) The utility's proposal for any contract amendment needed to implement the customer's payment of stranded costs.
- (ii) If the customer disagrees with the utility's response, it must respond to the utility within 30 days explaining why it disagrees. If the parties cannot work out a mutually agreeable resolution, they may exercise their rights to Commission resolution under the FPA.
- (5) A customer must be given the option to market or broker a portion or all of the capacity and energy associated with any stranded costs claimed by the public utility.
- (i) To exercise the option, the customer must so notify the utility in writing no later than 30 days after the public utility files its estimate of stranded costs for the customer with the Commission.
- (A) Before marketing or brokering can begin, the utility and customer must execute an agreement identifying, at a minimum, the amount and the price of capacity and associated energy the customer is entitled to schedule, and the duration of the customer's marketing or brokering of such capacity and energy.
- (ii) If agreement over marketing or brokering cannot be reached, and the parties seek Commission resolution of disputed issues, upon issuance of a Commission order resolving the disputed issues, the customer may reevaluate its decision in paragraph (c)(5)(i) of this section to exercise the marketing or brokering option. The customer must notify the utility in writing within 30 days of issuance of the Commission's order resolving the

Federal Energy Regulatory Commission

disputed issues whether the customer will market or broker a portion or all of the capacity and energy associated with stranded costs allowed by the Commission.

- (iii) If a customer undertakes the brokering option, and the customer's brokering efforts fail to produce a buyer within 60 days of the date of the brokering agreement entered into between the customer and the utility, the customer shall relinquish all rights to broker the released capacity and associated energy and will pay stranded costs as determined by the formula in paragraph (c)(2)(iii) of this section.
- (d) Recovery of retail stranded costs—1) General requirement. A public utility may seek to recover retail stranded costs through rates for retail transmission services only if the state regulatory authority does not have authority under state law to address stranded costs at the time the retail wheeling is required.
- (2) Evidentiary demonstration necessary for retail stranded cost recovery. A public utility seeking to recover retail stranded costs in accordance with paragraph (d)(1) of this section must demonstrate that:
- (i) It incurred costs to provide service to a retail customer that obtains retail wheeling based on a reasonable expectation that the utility would continue to serve the customer; and
- (ii) The stranded costs are not more than the customer would have contributed to the utility had the customer remained a retail customer of the utility.

[Order 888-A, 62 FR 12460, Mar. 14, 1997]

§ 35.27 Authority of State commissions.

Nothing in this part—

- (a) Shall be construed as preempting or affecting any jurisdiction a State commission or other State authority may have under applicable State and Federal law, or
- (b) Limits the authority of a State commission in accordance with State and Federal law to establish
- (1) Competitive procedures for the acquisition of electric energy, including demand-side management, purchased at wholesale, or
- (2) Non-discriminatory fees for the distribution of such electric energy to

retail consumers for purposes established in accordance with State law.

[Order 697, 72 FR 40038, July 20, 2007]

§ 35.28 Non-discriminatory open access transmission tariff.

- (a) Applicability. This section applies to any public utility that owns, controls or operates facilities used for the transmission of electric energy in interstate commerce and to any non-public utility that seeks voluntary compliance with jurisdictional transmission tariff reciprocity conditions.
- (b) Definitions—(1) Requirements service agreement means a contract or rate schedule under which a public utility provides any portion of a customer's bundled wholesale power requirements.
- (2) Economy energy coordination agreement means a contract, or service schedule thereunder, that provides for trading of electric energy on an "if, as and when available" basis, but does not require either the seller or the buyer to engage in a particular transaction.
- (3) Non-economy energy coordination agreement means any non-requirements service agreement, except an economy energy coordination agreement as defined in paragraph (b)(2) of this section.
- (4) *Demand response* means a reduction in the consumption of electric energy by customers from their expected consumption in response to an increase in the price of electric energy or to incentive payments designed to induce lower consumption of electric energy.
- (5) Demand response resource means a resource capable of providing demand response.
- (6) An operating reserve shortage means a period when the amount of available supply falls short ofdemand plus the operating reserve requirement.
- (7) Market Monitoring Unit means the person or entity responsible for carrying out the market monitoring functions that the Commission has ordered Commission-approved independent system operators and regional transmission organizations to perform.
- (8) Market Violation means a tariff violation, violation of a Commission-

approved order, rule or regulation,market manipulation, or inappropriate dispatch that creates substantial concerns regarding unnecessarymarket inefficiencies.

- (c) Non-discriminatory open access transmission tariffs. (1) Every public utility that owns, controls, or operates facilities used for the transmission of electric energy in interstate commerce must have on file with the Commission a tariff of general applicability for transmission services, including ancillary services, over such facilities. Such tariff must be the open access pro forma tariff contained in Order No. 888, FERC Stats. & Regs. ¶31,036 (Final Rule on Open Access and Stranded Costs), as revised by the open access pro forma tariff contained in Order No. 890, FERC Stats. & Regs. ¶31,241 (Final Rule on Open Access Reforms), or such other open access tariff as may be approved by the Commission consistent with Order No. 888, FERC Stats. & Regs ¶31,306 and Order No. 890, FERC Stats. & Regs. ¶31.241.
- (i) Subject to the exceptions in paragraphs (c)(1)(ii), (c)(1)(iii), (c)(1)(iv) and (c)(1)(v) of this section, the pro forma tariff contained in Order No. 888, FERC Stats. & Regs. ¶31,036, as revised by the open access pro forma tariff contained in Order No. 890, FERC Stats. & Regs. ¶31,241, and accompanying rates, must be filed no later than 60 days prior to the date on which a public utility would engage in a sale of electric energy at wholesale in interstate commerce or in the transmission of electric energy in interstate commerce.
- (ii) If a public utility owns, controls, or operates facilities used for the transmission of electric energy in interstate commerce as of May 14, 2007, it must file the revisions to the proforma tariff contained in Order No. 890, FERC Stats. & Regs. ¶31,241, pursuant to section 206 of the FPA and accompanying rates pursuant to section 205 of the FPA in accordance with the procedures set forth in Order No. 890, FERC Stats. & Regs ¶31,241.
- (iii) If a public utility owns, controls, or operates transmission facilities used for the transmission of electric energy in interstate commerce as of May 14, 2007, such facilities are jointly owned with a non-public utility, and the joint

ownership contract prohibits transmission service over the facilities to third parties, the public utility with respect to access over the public utility's share of the jointly owned facilities must file no later than May 14, 2007 the revisions to the pro forma tariff contained in Order No. 890, FERC Stats. & Regs. ¶31,241, pursuant to section 206 of the FPA and accompanying rates pursuant to section 205 of the FPA.

- (iv) Any public utility whose transmission facilities are under the independent control of a Commission-approved ISO or RTO may satisfy its obligation under paragraph (c)(1) of this section, with respect to such facilities, through the open access transmission tariff filed by the ISO or RTO.
- (v) If a public utility obtains a waiver of the tariff requirement pursuant to paragraph (d) of this section, it does not need to file the pro forma tariff required by this section.
- (vi) Any public utility that seeks a deviation from the pro forma tariff contained in Order No. 888, FERC Stats. & Regs. ¶31,036, as revised in Order No. 890, FERC Stats. & Regs. ¶31,241, must demonstrate that the deviation is consistent with the principles of Order No. 888, FERC Stats. & Regs. ¶31,036 and Order No. 890, FERC Stats. & Regs. ¶31,036 and Order No. 890, FERC Stats. & Regs. ¶31,241.
- (vii) Each public utility's open access transmission tariff must include the standards incorporated by reference in part 38 of this chapter.
- (2) Subject to the exceptions in paragraphs (c)(2)(i) and (c)(3)(iii) of this section, every public utility that owns, controls, or operates facilities used for the transmission of electric energy in interstate commerce, and that uses those facilities to engage in wholesale sales and/or purchases of electric energy, or unbundled retail sales of electric energy, must take transmission service for such sales and/or purchases under the open access transmission tariff filed pursuant to this section.
- (i) For sales of electric energy pursuant to a requirements service agreement executed on or before July 9, 1996, this requirement will not apply unless separately ordered by the Commission. For sales of electric energy pursuant to

a bilateral economy energy coordination agreement executed on or before July 9, 1996, this requirement is effective on December 31, 1996. For sales of electric energy pursuant to a bilateral non-economy energy coordination agreement executed on or before July 9, 1996, this requirement will not apply unless separately ordered by the Commission.

(ii) [Reserved]

- (3) Every public utility that owns, controls, or operates facilities used for the transmission of electric energy in interstate commerce, and that is a member of a power pool, public utility holding company, or other multi-lateral trading arrangement or agreement that contains transmission rates, terms or conditions, must have on file a joint pool-wide or system-wide open access transmission tariff, which tariff must be the pro forma tariff contained in Order No. 888, FERC Stats. & Regs. ¶31,036, as revised by the pro forma tariff contained in Order No. 890, FERC Stats. & Regs. ¶31,241, or such other open access tariff as may be approved by the Commission consistent with Order No. 888, FERC Stats. & Regs. ¶31,036 and Order No. 890, FERC Stats. & Regs. ¶31,241.
- (i) For any power pool, public utility holding company or other multi-lateral arrangement or agreement that contains transmission rates, terms or conditions and that is executed after May 14, 2007, this requirement is effective on the date that transactions begin under the arrangement or agreement.
- (ii) For any power pool, public utility holding company or other multi-lateral arrangement or agreement that contains transmission rates, terms or conditions and that is executed on or before May 14, 2007, a public utility member of such power pool, public utility holding company or other multi-lateral arrangement or agreement that owns. controls, or operates facilities used for the transmission of electric energy in interstate commerce must file the revisions to its joint pool-wide or systemwide contained in Order No. 890, FERC Stats. & Regs. ¶31,241, pursuant to section 206 of the FPA and accompanying rates pursuant to section 205 of the FPA in accordance with the procedures

set forth in Order No. 890, FERC Stats. & Regs ¶31,241.

- (iii) A public utility member of a power pool, public utility holding company or other multi-lateral arrangement or agreement that contains transmission rates, terms or conditions and that is executed on or before July 9, 1996 must take transmission service under a joint pool-wide or system-wide open access transmission tariff filed pursuant to this section for wholesale trades among the pool or system members.
- (4) Consistent with paragraph (c)(1) of this section, every Commission-approved ISO or RTO must have on file with the Commission a tariff of general applicability for transmission services, including ancillary services, over such facilities. Such tariff must be the pro forma tariff contained in Order No. 888, FERC Stats. & Regs. ¶31,036, as revised by the pro forma tariff contained in Order No. 890, FERC Stats. & Regs. ¶31,241, or such other open access tariff as may be approved by the Commission consistent with Order No. 888, FERC Stats. & Reg. ¶31,036 and Order No. 890, FERC Stats. & Regs. ¶31,241.
- (i) Subject to paragraph (c)(4)(ii) of this section, a Commission-approved ISO or RTO must file the revisions to the pro forma tariff contained in Order No. 890, FERC Stats. & Regs. ¶31,241, pursuant to section 206 of the FPA and accompanying rates pursuant to section 205 of the FPA in accordance with the procedures set forth in Order No. 890, FERC Stats. & Regs ¶31,241.
- (ii) If a Commission-approved ISO or RTO can demonstrate that its existing open access tariff is consistent with or superior to the revisions to the proforma tariff contained in Order No. 888, FERC Stats. & Regs. ¶31,036, as revised by the pro forma tariff in Order No. 890, FERC Stats. & Regs. ¶31,241, or any portions thereof, the Commission-approved ISO or RTO may instead set forth such demonstration in its filing pursuant to section 206 in accordance with the procedures set forth in Order No. 890, FERC Stats. & Regs. ¶31,241.
- (d) Waivers. A public utility subject to the requirements of this section and Order No. 889, FERC Stats. & Regs. ¶31,037 (Final Rule on Open Access Same-Time Information System and

Standards of Conduct) may file a request for waiver of all or part of the requirements of this section, or Part 37 (Open Access Same-Time Information System and Standards of Conduct for Public Utilities), for good cause shown. Except as provided in paragraph (f) of this section, an application for waiver must be filed either:

- (1) No later than May 14, 2007, or
- (2) No later than 60 days prior to the time the public utility would otherwise have to comply with the requirement.
- (e) Non-public utility procedures for tariff reciprocity compliance. (1) A non-public utility may submit a transmission tariff and a request for declaratory order that its voluntary transmission tariff meets the requirements of Order No. 888, FERC Stats. & Regs. ¶31,036 and Order No. 890, FERC Stats. & Regs. ¶31,241.
- (i) Any submittal and request for declaratory order submitted by a nonpublic utility will be provided an NJ (non-jurisdictional) docket designation.
- (ii) If the submittal is found to be an acceptable transmission tariff, an applicant in a Federal Power Act (FPA) section 211 or 211A proceeding against the non-public utility shall have the burden of proof to show why service under the open access tariff is not sufficient and why a section 211 or 211A order should be granted.
- (2) A non-public utility may file a request for waiver of all or part of the reciprocity conditions contained in a public utility open access tariff, for good cause shown. An application for waiver may be filed at any time.
- (f) Standard generator interconnection procedures and agreements. (1) Every public utility that is required to have on file a non-discriminatory open access transmission tariff under this section must amend such tariff by adding the standard interconnection procedures and agreement contained in Order No. 2003, FERC Stats. & Regs. & 31,146 (Final Rule on Generator Interconnection), as amended by the Commission in Order No. 661, FERC Stats. & Regs. ¶31,186 (Final Rule on Interconnection for Wind Energy), and the standard small generator interconnection procedures and agreement contained in Order No. 2006, FERC Stats. &

Regs. ¶31,180 (Final Rule on Small Generator Interconnection), or such other interconnection procedures and agreements as may be approved by the Commission consistent with Order No. 2003, FERC Stats. & Regs. & 31,146 (Final Rule on Generator Interconnection) and Order No. 2006, FERC Stats. & Regs. ¶31,180 (Final Rule on Small Generator Interconnection).

- (i) The amendment to implement the Final Rule on Generator Interconnection required by the preceding subsection must be filed no later than January 20, 2004.
- (ii) The amendment to implement the Final Rule on Small Generator Interconnection required by the preceding subsection must be filed no later than August 12, 2005.
- (iii) The amendment to implement the Final Rule on Interconnection for Wind Energy required by the preceding subsection must be filed no later than December 30, 2005.
- (iv) Any public utility that seeks a deviation from the standard interconnection procedures and agreement contained in Order No. 2003. FERC Stats. & Regs. & 31,146 (Final Rule on Generator Interconnection), as amended by the Commission in Order No. 661, FERC Stats. & Regs. ¶31,186 (Final Rule on Interconnection for Wind Energy), or the standard small generator interconnection procedures and agreement contained in Order No. 2006, FERC Stats. & Regs. ¶31.180 (Final Rule on Small Generator Interconnection), must demonstrate that the deviation is consistent with the principles of either Order No. 2003, FERC Stats. & Regs. & 31,146 (Final Rule on Generator Interconnection) or Order No. 2006, FERC Stats. & Regs. ¶31,180 (Final Rule on Small Generator Interconnec-
- (2) The non-public utility procedures for tariff reciprocity compliance described in paragraph (e) of this section are applicable to the standard interconnection procedures and agreements.
- (3) A public utility subject to the requirements of this paragraph pertaining to the Final Rule on Generator Interconnection may file a request for waiver of all or part of the requirements of this paragraph, for good cause

shown. An application for waiver must be filed either:

- (i) No later than January 20, 2004, or
- (ii) No later than 60 days prior to the time the public utility would otherwise have to comply with the requirements of this paragraph.
- (4) A public utility subject to the requirements of this paragraph pertaining to the Final Rule on Small Generator Interconnection may file a request for waiver of all or part of the requirements of this paragraph, for good cause shown. An application for waiver must be filed either:
 - (i) No later than August 12, 2005, or
- (ii) No later than 60 days prior to the time the public utility would otherwise have to comply with the requirements of this paragraph.
- (g) Tariffs and operations of Commission-approved independent system operators and regionaltransmission organizations.
 - (1) Demand response and pricing.
- (i) Ancillary services provided by demand response resources.
- (A) Every Commission-approved independent system operator or regional transmission organization that operates organized markets based on competitive bidding for energy imbalance, spinning reserves, supplemental serves, reactive power and voltage control, or regulation and frequency response ancillaryservices (or its functional equivalent in the Commissionapproved independent system operator's orregional transmission organization's tariff) must accept bids from deresponse resources thesemarkets for that product on a basis comparable to any other resources, if the demand response resourcemeets the necessary technical requirements under the tariff, and submits a bid under the Commission-approved independent system operator's or regional transmission organization's bidding rules at or belowthe marketclearing price, unless not permitted by the laws or regulations of the relevant electric retailregulatory authority.
- (B) Each Commission-approved independent system operator or regional transmission organization mustallow providers of a demand response resource to specify the following in their bids:

- (1) A maximum duration in hours that the demand response resource may be dispatched;
- (2) A maximum number of times that the demand response resource may be dispatched during a day; and
- (3) A maximum amount of electric energy reduction that the demand response resource may be required toprovide either daily or weekly.
- (ii) Removal of deviation charges. A Commission-approved independent system operator or regionaltransmission organization with a tariff that contains a day-ahead and a real-time market may not assess acharge to a purchaser of electric energy in its day-ahead market for purchasing less power in the real-timemarket during a real-time market period for which the Commission-approved independent system operatoror regional transmission organization declares an operating reserve shortage or makes a generic requestto reduce load to avoid an operating reserve shortage.
- (iii) Aggregation of retail customers. Each Commission-approved pendent system operator and regional transmission organization must accept bids from an aggregator of retail customers that aggregates the demand response of the customers of utilities that distributed more than 4 million megawatt-hours in the previous fiscal year, and the customers of utilities that distributed 4 million megawatthours or less in the previous fiscal year, where the relevant electric retail regulatory authority permits such customers' demand response to be bid into organized markets by an aggregator of retail customers. An independent system operator or regional transmission organization must not accept bids from an aggregator of retail customers that aggregates the demand response of the customers of utilities that distributed more than 4 million megawatt-hours in the previous fiscal year, where the relevant electric retail regulatory authority prohibits such customers' demand response to be bid into organized markets by an aggregator of retail customers, or the customers of utilities that distributed 4 million megawatthours or less in the previous fiscal year, unless the relevant electric retail

regulatory authority permits such customers' demand response to be bid into organized markets by an aggregator of retail customers.

- (iv) Price formation during periods of operating reserve shortage.
- (A) Each Commission-approved independent system operator or regional transmission organization mustmodify its market rules to allow the market-clearing price during periods of operating reserve shortage toreach a level that rebalances supply and demand so as to maintain reliability while providing sufficientprovisions for mitigating market power.
- (B) A Commission-approved independent system operator or regional transmission organization may phase in this modification of its market rules.
- (2) Long-term power contracting in organized markets. Each Commission-approved independent systemoperator or regional transmission organization must provide a portion of its Web site for marketparticipants to post offers to buy or sell power on a long-term basis.
 - (3) Market monitoring policies.
- (i) Each Commission-approved independent system operator or regional transmission organization mustmodify its tariff provisions governing its Market Monitoring Unit to reflect the directives provided in OrderNo. 719, including the following:
- (A) Each Commission-approved independent system operator or regional transmission organization mustinclude in its tariff a provision to provide its Market Monitoring Unit access to Commission-approvedindependent system operator andregional transmission organization market data, resources and personnel to enable the MarketMonitoring Unit to carry out its functions.
- (B) The tariff provision must provide the Market Monitoring Unit complete access to the Commission-approved independent system operator's and regional transmission organization's databases of market information.
- (C) The tariff provision must provide that any data created by the Market Monitoring Unit, including, but not limited to, reconfiguring of the Commission-approved independent system operator's and regional transmission organization's data, will be kept within

the exclusive control of the Market Monitoring Unit.

- (D) The Market Monitoring Unit must report to the Commission-approved independent system operator's or regional transmission organization's board of directors, with its management members removed, or to an independent committee of the Commissionapproved independent system operator's or regional transmission organization's board of directors. A Commission-approved independent system operator or regional transmission organization that has both an internal Market Monitoring Unit and an external Market Monitoring Unit may permit the internal Market Monitoring Unit to report to management and the external Market Monitoring Unit to report to the Commission-approved independent system operator's or regional transmission organization's board of directors with its management members removed, or to an independent committee of the Commission-approved independent system operator or transmission organization regional board of directors. If the internal market monitor is responsible for carrying out any or all of the core Market Monitoring Unit functions identified in paragraph (g)(3)(ii) of this section, the internal market monitor must report to the independent system operator's or regional transmission organization's board of directors.
- (E) A Commission-approved independent system operator or regional transmission organization may not alter the reports generated by the Market Monitoring Unit, or dictate the conclusions reached by the Market Monitoring Unit.
- (F) Each Commission-approved independent system operator or regional transmission organization must consolidate the core Market Monitoring Unit provisions into one section of its tariff. Each independent system operator or regional transmission organization must include a mission statement in the introduction to the Market Monitoring Unit provisions that identifies the Market Monitoring Unit's goals, including the protection of consumers and market participants by the identification and reporting of market design flaws and market power abuses.

- (ii) Core Functions of Market Monitoring Unit. The Market Monitoring Unit must perform the following core functions:
- (A) Evaluate existing and proposed market rules, tariff provisions and market design elements and recommend proposed rule and tariff changes to the Commission-approved independent system operator or regional transmission organization, to the Commission's Office of Energy Market Regulation staff and to other interested entities such as state commissions and market participants, provided that:
- (1) The Market Monitoring Unit is not to effectuate its proposed market design itself, and
- (2) The Market Monitoring Unit must limit distribution of its identifications and recommendations to the independent system operator or regional transmission organization and to Commission staff in the event it believes broader dissemination could lead to exploitation, with an explanation of why further dissemination should be avoided at that time.
- (B) Review and report on the performance of the wholesale markets to the Commission-approved independent system operator or regional transmission organization, the Commission, and other interested entities such as state commissions and market participants, on at least a quarterly basis and submit a more comprehensive annual state of the market report. The Market Monitoring Unit may issue additional reports as necessary.
- (C) Identify and notify the Commission's Office of Enforcement staff of instances in which a market participant's or the Commission-approved independent system operator's or regional transmission organization's behavior may require investigation, including, but not limited to, suspected Market Violations.
- $\begin{array}{ccc} \hbox{(iii)} \ \ Tariff \ \ administration \ \ and \ \ mitigation \end{array}$
- (A) A Commission-approved independent system operator or regional transmission organization may not permit its Market Monitoring Unit, whether internal or external, to participate in the administration of the Commission-approved independent sys-

- tem operator's or regional transmission organization's tariff or, except as provided in paragraph (g)(3)(iii)(D) of this section, to conduct prospective mitigation.
- (B) A Commission-approved independent system operator or regional transmission organization may permit its Market Monitoring Unit to provide the inputs required for the Commission-approved independent system operator or regional transmission organization to conduct prospective mitigation, including, but not limited to, reference levels, identification of system constraints, and cost calculations.
- (C) A Commission-approved independent system operator or regional transmission organization may allow its Market Monitoring Unit to conduct retrospective mitigation.
- (D) A Commission-approved independent system operator or regional transmission organization with a hybrid Market Monitoring Unit structure may permit its internal market monitor to conduct prospective and/or retrospective mitigation, in which case it must assign to its external market monitor the responsibility and the tools to monitor the quality and appropriateness of the mitigation.
- (E) Each Commission-approved independent system operator or regional transmission organization must identify in its tariff the functions the Market Monitoring Unit will perform and the functions the Commission-approved independent system operator or regional transmission organization will perform.
- (iv) Protocols on Market Monitoring Unit referrals to the Commission of suspected violations.
- (A) A Market Monitoring Unit is to make a non-public referral to the Commission in all instances where the Market Monitoring Unit has reason to be elieve that a Market Violation has occurred. While the Market Monitoring Unit need not be able to prove that a Market Violation has occurred, the Market Monitoring Unit is to provide sufficient credible information to warrant further investigation by the Commission. Once the Market Monitoring Unit has obtained sufficient credible information to warrant referral to the Commission, the Market Monitoring

Unit is to immediately refer the matter to the Commission and desist from independent action related to the alleged Market Violation. This does not preclude the Market Monitoring Unit from continuing to monitor for any repeated instances of the activity by the same or other entities, which would constitute new Market Violations. The Market Monitoring Unit is to respond to requests from the Commission for any additional information in connection with the alleged Market Violation it has referred.

- (B) All referrals to the Commission of alleged Market Violations are to be in writing, whether transmitted electronically, by fax, mail, or courier. The Market Monitoring Unit may alert the Commission orally in advance of the written referral.
- (C) The referral is to be addressed to the Commission's Director of the Office of Enforcement, with a copy also directed to both the Director of the Office of Energy Market Regulation and the General Counsel.
- (D) The referral is to include, but need not be limited to, the following information.
- (1) The name[s] of and, if possible, the contact information for, the entity[ies] that allegedly took the action[s] that constituted the alleged Market Violation[s]:
- (2) The date[s] or time period during which the alleged Market Violation[s] occurred and whether the alleged wrongful conduct is ongoing;
- (3) The specific rule or regulation, and/or tariff provision, that was allegedly violated, or the nature of any inappropriate dispatch that may have occurred:
- (4) The specific act[s] or conduct that allegedly constituted the Market Violation;
- (5) The consequences to the market resulting from the acts or conduct, including, if known, an estimate of economic impact on the market:
- (6) If the Market Monitoring Unit believes that the act[s] or conduct constituted a violation of the anti-manipulation rule of Part 1c, a description of the alleged manipulative effect on market prices, market conditions, or market rules:

- (7) Any other information the Market Monitoring Unit believes is relevant and may be helpful to the Commission.
- (E) Following a referral to the Commission, the Market Monitoring Unit is to continue to notify and inform the Commission of any information that the Market Monitoring Unit learns of that may be related to the referral, but the Market Monitoring Unit is not to undertake any investigative steps regarding the referral except at the express direction of the Commission or Commission Staff.
- (v) Protocols on Market Monitoring Unit Referrals to the Commission of Perceived Market Design Flaws and Recommended Tariff Changes.
- (A) A Market Monitoring Unit is to make a referral to the Commission in all instances where the Market Monitoring Unit has reason to believe market design flaws exist that it believes could effectively be remedied by rule or tariff changes. The Market Monitoring Unit must limit distribution of its identifications and recommendations to the independent system operator or regional transmission organization and to the Commission in the event it believes broader dissemination could lead to exploitation, with an explanation of why further dissemination should be avoided at that time.
- (B) All referrals to the Commission relating to perceived market design flaws and recommended tariff changes are to be in writing, whether transmitted electronically, by fax, mail, or courier. The Market Monitoring Unit may alert the Commission orally in advance of the written referral.
- (C) The referral should be addressed to the Commission's Director of the Office of Energy Market Regulation, with copies directed to both the Director of the Office of Enforcement and the General Counsel.
- (D) The referral is to include, but need not be limited to, the following information.
- (1) A detailed narrative describing the perceived market design flaw[s];
- (2) The consequences of the perceived market design flaw[s], including, if known, an estimate of economic impact on the market;
- (3) The rule or tariff change(s) that the Market Monitoring Unit believes

could remedy the perceived market design flaw:

- (4) Any other information the Market Monitoring Unit believes is relevant and may be helpful to the Commission.
- (E) Following a referral to the Commission, the Market Monitoring Unit is to continue to notify and inform the Commission of any additional information regarding the perceived market design flaw, its effects on the market, any additional or modified observations concerning the rule or tariff changes that could remedy the perceived design flaw, any recommendations made by the Market Monitoring Unit to the regional transmission organization or independent system operator, stakeholders, market participants or state commissions regarding the perceived design flaw, and any actions taken by the regional transmission organization or independent system operator regarding the perceived design flaw.
- (vi) Market Monitoring Unit ethics standards. Each Commission-approved independent system operator or regional transmission organization must include in its tariff ethical standards for its Market Monitoring Unit and the employees of its Market Monitoring Unit. At a minimum, the ethics standards must include the following requirements:
- (A) The Market Monitoring Unit and its employees must have no material affiliation with any market participant or affiliate.
- (B) The Market Monitoring Unit and its employees must not serve as an officer, employee, or partner of a market participant.
- (C) The Market Monitoring Unit and its employees must have no material financial interest in any market participant or affiliate with potential exceptions for mutual funds and non-directed investments.
- (D) The Market Monitoring Unit and its employees must not engage in any market transactions other than the performance of their duties under the tariff.
- (E) The Market Monitoring Unit and its employees must not be compensated, other than by the Commission-approved independent system operator or regional transmission organi-

- zation that retains or employs it, for any expert witness testimony or other commercial services, either to the Commission-approved independent system operator or regional transmission organization or to any other party, in connection with any legal or regulatory proceeding or commercial transaction relating to the Commission-approved independent system operator or regional transmission organization or Commission-approved indethe to pendent system operator's or regional transmission organization's markets.
- (F) The Market Monitoring Unit and its employees may not accept anything of value from a market participant in excess of a *de minimis* amount.
- (G) The Market Monitoring Unit and its employees must advise a supervisor in the event they seek employment with a market participant, and must disqualify themselves from participating in any matter that would have an effect on the financial interest of the market participant.
- (4) Offer and bid data. (i) Unless a Commission-approved independent system operator or regional transmission organization obtains Commission approval for a different period, each Commission-approved independent system operator and regional transmission organization must release its offer and bid data within three months.
- (ii) A Commission-approved independent system operator or regional transmission organization must mask the identity of market participants when releasing offer and bid data. The Commission-approved independent system operators and regional transmission organization may propose a time period for eventual unmasking.
- (5) Responsiveness of Commission-approved independent system operators and regional transmission organizations. Each Commission-approved independent system operator or regional transmission organization must adopt business practices and procedures that achieve Commission-approved independent system operator and regional transmission organization board of directors' responsiveness to customers and other stakeholders and satisfy the following criteria:
- (i) Inclusiveness. The business practices and procedures must ensure that

any customer or other stakeholder affected by the operation of the Commission-approved independent system operator or regional transmission organization, or its representative, is permitted to communicate the customer's or other stakeholder's views to the independent system operator's or regional transmission organization's board of directors:

- (ii) Fairness in balancing diverse interests. The business practices and procedures must ensure that the interests of customers or other stakeholders are equitably considered, and that deliberation and consideration of Commission-approved independent system operator's and regional transmission organization's issues are not dominated by any single stakeholder category;
- (iii) Representation of minority positions. The business practices and procedures must ensure that, in instances where stakeholders are not in total agreement on a particular issue, minority positions are communicated to the Commission-approved independent system operator's and regional transmission organization's board of directors at the same time as majority positions; and
- (iv) Ongoing responsiveness. The business practices and procedures must provide for stakeholder input into the Commission-approved independent system operator's or regional transmission organization's decisions as well as mechanisms to provide feedback to stakeholders to ensure that information exchange and communication continue over time.
- (6) Compliance filings. All Commission-approved independent system operators and regional transmission organizations must make a compliance filing with the Commission as described in Order No. 719 under the following schedule:
- (i) The compliance filing addressing the accepting of bids from demand response resources in markets for ancillary services on a basis comparable to other resources, removal of deviation charges, aggregation of retail customers, shortage pricing during periods of operating reserve shortage, long-term power contracting in organized markets, Market Monitoring Units, Commission-approved independent sys-

tem operators' and regional transmission organizations' board of directors' responsiveness, and reporting on the study of the need for further reforms to remove barriers to comparable treatment of demand response resources must be submitted on or before April 28, 2009.

(ii) A public utility that is approved as a regional transmission organization under §35.34, or that is not approved but begins to operate regional markets for electric energy or ancillary services after December 29, 2008, must comply with Order No. 719 and the provisions of paragraphs (g)(1) through (g)(5) of this section before beginning operations.

[Order 888, 61 FR 21693, May 10, 1996, as amended by Order 2003, 68 FR 49929, Aug. 19, 2003; Order 2006, 70 FR 34240, June 13, 2005; Order 661, 70 FR 75014, Dec. 19, 2005; Order 676, 71 FR 26212, May 4, 2006; Order 890, 72 FR 12492, Mar. 15, 2007; 73 FR 64167, Oct. 28, 2008; Order 719-A, 74 FR 37801, July 29, 2009]

EFFECTIVE DATE NOTE: At 76 FR 16678, Mar. 24, 2011, § 35.28 was amended by adding a new paragraph (g)(1)(v), effective Apr. 25, 2011. For the convenience of the user, the added text is set forth as follows:

§ 35.28 Non-discriminatory open access transmission tariff.

* * * * *

- (g) * * * (1) * * *
- (v) Demand response compensation in energy markets. Each Commission-approved independent system operator or regional transmission organization that has a tariff provision permitting demand response resources to participate as a resource in the energy market by reducing consumption of electric energy from their expected levels in response to price signals must:
- (A) Pay to those demand response resources the market price for energy for these reductions when these demand response resources have the capability to balance supply and demand and when payment of the market price for energy to these resources is cost-effective as determined by a net benefits test accepted by the Commission;
- (B) Allocate the costs associated with demand response compensation proportionally to all entities that purchase from the relevant energy market in the area(s) where the demand response reduces the market price for energy at the time when the demand response resource is committed or dispatched.

* * * * *

§ 35.29 Treatment of special assessments levied under the Atomic Energy Act of 1954, as amended by Title XI of the Energy Policy Act of 1992.

The costs that public utilities incur relating to special assessments under the Atomic Energy Act of 1954, as amended by the Energy Policy Act of 1992, are costs that may be reflected in jurisdictional rates. Public utilities seeking to recover the costs incurred relating to special assessments shall comply with the following procedures.

- (a) Fuel adjustment clauses. In computing the Account 518 cost of nuclear fuel pursuant to §35.14(a)(6), utilities seeking to recover the costs of special assessments through their fuel adjustment clauses shall:
- (1) Deduct any expenses associated with special assessments included in Account 518:
- (2) Add to Account 518 one-twelfth of any payments made for special assessments within the 12-month period ending with the current month; and
- (3) Deduct from Account 518 one-twelfth of any refunds of payments made for special assessments received within the 12-month period ending with the current month that is received from the Federal government because the public utility has contested a special assessment or overpaid a special assessment.
- (b) Cost of service data requirements. Public utilities filing rate applications under §§35.12 or 35.13 (regardless of whether the utility elects the abbreviated, unadjusted Period I, adjusted Period I, or Period II cost support requirements) must submit cost data that is computed in accordance with the requirements specified in paragraphs (a) (1), (2) and (3) of this section.
- (c) Formula rates. Public utilities with formula rates on file that provide for the automatic recovery of nuclear fuel costs must reflect the costs of special assessments in accordance with the requirements specified in paragraphs (a) (1), (2) and (3) of this section.

[Order 557, 58 FR 51221, Oct. 1, 1993. Redesignated by Order 888, 61 FR 21692, May 10, 1996]

Subpart D—Procedures and Requirements for Public Utility Sales of Power to Bonneville Power Administration Under Northwest Power Act

AUTHORITY: Federal Power Act, 16 U.S.C. 792-828c (1976 and Supp. IV 1980) and Pacific Northwest Electric Power Planning and Conservation Act, 16 U.S.C. 830-839h (Supp. IV (1980)).

§35.30 General provisions.

- (a) Applicability. This subpart applies to any sales of electric power subject to the Commission's jurisdiction under Part II of the Federal Power Act from public utilities to the Administrator of the Bonneville Power Administration (BPA) at the average system cost (ASC) of that utility's resources (electric power generation by the utility) pursuant to section 5(c) of the Pacific Northwest Electric Power Planning and Conservation Act, 16 U.S.C. 830–839h. The ASC is determined by BPA in accordance with 18 CFR part 301.
- (b) Effectiveness of rates. (1) During the period between the date of BPA's determination of ASC and the date of the final order issued by the Commission, the utility may charge the rate based on the ASC determined by BPA, subject to §35.31(c) of this part.
- (2) Except as otherwise provided under this section, the ASC ordered by the Commission will be deemed in effect from the beginning of the relevant exchange period, as defined in §301.1(b)(95) of this chapter. For any initial exchange period after the Commission approves a new ASC methodology, the ASC will be effective retroactively under this paragraph only if the utility files its new ASC within the time allowed under BPA procedures. Any utility that files a revised ASC with BPA in accordance with this paragraph must promptly file with the Commission a notice of timely filing of the new ASC.
- (c) Filing requirements. Within 15 business days of the date of issuance of the BPA report on a utility's ASC, the utility must file with the Commission the ASC determined by BPA, the BPA written report, the utility's ASC schedules, material necessary to comply

with 18 CFR 35.13(c), and any other material requested by the Commission or its staff.

[Order 337, 48 FR 46976, Oct. 17, 1983, as amended by Order 400, 49 FR 39300, Oct. 5, 1984]

§35.31 Commission review.

- (a) *Procedures*. Filings under this subpart are subject to the procedures applicable to other filings under section 205 of the Federal Power Act, as the Commission deems appropriate.
- (b) Commission standard. With respect to any filing under this subpart, the Commission will determine whether the ASC set by BPA for the applicable exchange period was determined in accordance with the ASC methodology set forth at 18 CFR 301.1. If the ASC is not in accord with the methodology, the Commission will order that BPA amend the ASC to conform with the methodology. If the ASC is in accord with the methodology, the rate is deemed just and reasonable.
- (c) Refunds and adjustments. (1) Any ASC-based rate charged by a public utility under this subpart pending Commission order is subject to refund or to adjustment that increases the ASC-based rate.
- (2) Any interest on refunds ordered by the Commission under this subpart is computed in accordance with 18 CFR 35.19a. Interest on any increase ordered by the Commission will be at the rate charged to BPA by the U.S. Treasury during that period, unless the Commission orders another interest rate.

(Approved by the Office of Management and Budget under control number 1902–0096)

[Order 337, 48 FR 46976, Oct. 17, 1983, as amended at 49 FR 1177, Jan. 10, 1984]

Subpart E—Regulations Governing Nuclear Plant Decommissioning Trust Funds

§35.32 General provisions.

- (a) If a public utility has elected to provide for the decommissioning of a nuclear power plant through a nuclear plant decommissioning trust fund (Fund), the Fund must meet the following criteria:
- (1) The Fund must be an external trust fund in the United States, estab-

lished pursuant to a written trust agreement, that is independent of the utility, its subsidiaries, affiliates or associates. If the trust fund includes monies collected both in Commission-jurisdictional rates and in non-Commission-jurisdictional rates, then a separate account of the Commission-jurisdictional monies shall be maintained

- (2) The utility may provide overall investment policy to the Trustee or Investment Manager, but it may do so only in writing, and neither the utility nor its subsidiaries, affiliates or associates may serve as Investment Manager or otherwise engage in day-to-day management of the Fund or mandate individual investment decisions.
- (3) The Fund's Investment Manager must exercise the standard of care, whether in investing or otherwise, that a prudent investor would use in the same circumstances. The term "prudent investor" means a prudent investor as described in Restatement of the Law (Third), Trusts §227, including general comments and reporter's notes, pages 8-101. St. Paul, MN: American Law Institute Publishers, (1992). ISBN 0-314-84246-2. This incorporation by reference was approved by the Director of the Federal Register in accordance with 5 U.S.C. 552(a) and 1 CFR part 51. Copies may be obtained from the American Law Institute, 4025 Chestnut Street, Philadelphia, PA 19104, and are also available in local law libraries. Copies may be inspected at the Federal Energy Regulatory Commission's Library, Room 95-01, 888 First Street, NE. Washington, DC or at the National Archives and Records Administration (NARA). For information on the availability of this material at NARA, call 202–741–6030, or 20 to: http:// www.archives.gov/federal register/ code of federal regulations/ ibr locations.html.
- (4) The Trustee shall have a net worth of at least \$100 million. In calculating the \$100 million net worth requirement, the net worth of the Trustee's parent corporation and/or affiliates may be taken into account only if such entities guarantee the Trustee's responsibilities to the Fund.
- (5) The Trustee or Investment Manager shall keep accurate and detailed

accounts of all investments, receipts, disbursements and transactions of the Fund. All accounts, books and records relating to the Fund shall be open to inspection and audit at reasonable times by the utility or its designee or by the Commission or its designee. The utility or its designee must notify the Commission prior to performing any such inspection or audit. The Commission may direct the utility to conduct an audit or inspection.

- (6) Absent the express authorization of the Commission, no part of the assets of the Fund may be used for, or diverted to, any purpose other than to fund the costs of decommissioning the nuclear power plant to which the Fund relates, and to pay administrative costs and other incidental expenses, including taxes, of the Fund.
- (7) If the Fund balances exceed the amount actually expended for decommissioning after decommissioning has been completed, the utility shall return the excess jurisdictional amount to ratepayers, in a manner the Commission determines
- (8) Except for investments tied to market indexes or other mutual funds, the Investment Manager shall not invest in any securities of the utility for which it manages the funds or in that utility's subsidiaries, affiliates, or associates or their successors or assigns.
- (9) The utility and the Fiduciary shall seek to obtain the best possible tax treatment of amounts collected for nuclear plant decommissioning. In this regard, the utility and the Fiduciary shall take maximum advantage of tax deductions and credits, when it is consistent with sound business practices to do so.
- (10) Each utility shall deposit in the Fund at least quarterly all amounts included in Commission-jurisdictional rates to fund nuclear power plant decommissioning.
- (b) The establishment, organization, and maintenance of the Fund shall not relieve the utility or its subsidiaries, affiliates or associates of any obligations it may have as to the decommissioning of the nuclear power plant. It is not the responsibility of the Fiduciary to ensure that the amount of monies that a Fund contains are adequate to

pay for a nuclear unit's decommissioning.

- (c) A utility may establish both qualified and non-qualified Funds with respect to a utility's interest in a specific nuclear plant. This section applies to both "qualified" (under the Internal Revenue Code, 26 U.S.C. 468A, or any successor section) and non-qualified Funds.
- (d) A utility must regularly supply to the Fund's Investment Manager, and regularly update, essential information about the nuclear unit covered by the Trust Fund Agreement, including its description, location, expected remaining useful life, the decommissioning plan the utility proposes to follow, the utility's liquidity needs once decommissioning begins, and any other information that the Fund's Investment Manager would need to construct and maintain, over time, a sound investment plan.
- (e) A utility should monitor the performance of all Fiduciaries of the Fund and, if necessary, replace them if they are not properly performing assigned responsibilities.

[Order 580–A, 62 FR 33348, June 19, 1997, as amended at 69 FR 18803, Apr. 9, 2004]

§35.33 Specific provisions.

- (a) In addition to the general provisions of §35.32, the Trustee must observe the provisions of this section.
- (b) The Trustee may use Fund assets only to:
- (1) Satisfy the liability of a utility for decommissioning costs of the nuclear power plant to which the Fund relates as provided by §35.32; and
- (2) Pay administrative costs and other incidental expenses, including taxes, of the Fund as provided by §35.32.
- (c) To the extent that the Trustee does not currently require the assets of the Fund for the purposes described in paragraphs (b)(1) and (b)(2) of this section, the Investment Manager, when investing Fund assets, must exercise the same standard of care that a reasonable person would exercise in the same circumstances. In this context, a "reasonable person" means a prudent investor as described in Restatement of the Law (Third), Trusts § 227, including general comments and reporter's notes,

pages 8-101. St. Paul, MN: American Law Institute Publishers, 1992. ISBN 0-314-84246-2. This incorporation by reference was approved by the Director of the Federal Register in accordance with 5 U.S.C. 552(a) and 1 CFR part 51. Copies may be obtained from the American Law Institute, 4025 Chestnut Street, Philadelphia, PA 19104, and are also available in local law libraries. Copies may be inspected at the Federal Energy Regulatory Commission, 888 First Street, NE. Washington, DC or at the National Archives and Records Administration (NARA). For information on the availability of this material at NARA, call 202-741-6030, or go to: http:// www.archives.gov/federal-register/cfr/ibrlocations.html.

- (d) The utility must submit to the Commission by March 31 of each year, one original and three conformed copies of the financial report furnished to the utility by the Fund's Trustee that shows for the previous calendar year:
- (1) Fund assets and liabilities at the beginning of the period;
- (2) Activity of the Fund during the period, including amounts received from the utility, a summary amount for purchases of fund investments and a summary amount for sales of fund investments, gains and losses from investment activity, disbursements from the Fund for decommissioning activity and payment of Fund expenses, including taxes; and
- (3) Fund assets and liabilities at the end of the period. The report should not include the liability for decommissioning.
- (4) Public utilities owning nuclear plants must maintain records of individual purchase and sales transactions until after decommissioning has been completed and any excess jurisdictional amounts have been returned to ratepayers in a manner that the Commission determines. The public utility need not include these records in the financial report that it furnishes to the Commission by March 31 of each year.
- (e) The utility must also mail a copy of the financial report provided to the Commission pursuant to paragraph (d) of this section to anyone who requests it.
- (f) If an independent public accountant has expressed an opinion on the re-

port or on any portion of the report, then that opinion must accompany the report.

[Order 580-A, 62 FR 33348, June 19, 1997, as amended at 69 FR 18803, Apr. 9, 2004; Order 658, 70 FR 34343, June 14, 2005; Order 737, 75 FR 43404, July 26, 2010]

Subpart F—Procedures and Requirements Regarding Regional Transmission Organizations

§ 35.34 Regional Transmission Organizations.

- (a) Purpose. This section establishes required characteristics and functions for Regional Transmission Organizations for the purpose of promoting efficiency and reliability in the operation and planning of the electric transmission grid and ensuring non-discrimination in the provision of electric transmission services. This section further directs each public utility that owns, operates, or controls facilities used for the transmission of electric energy in interstate commerce to make certain filings with respect to forming and participating in a Regional Transmission Organization.
- (b) Definitions. (1) Regional Transmission Organization means an entity that satisfies the minimum characteristics set forth in paragraph (j) of this section, performs the functions set forth in paragraph (k) of this section, and accommodates the open architecture condition set forth in paragraph (l) of this section.
 - (2) Market participant means:
- (i) Any entity that, either directly or through an affiliate, sells or brokers electric energy, or provides ancillary services to the Regional Transmission Organization, unless the Commission finds that the entity does not have economic or commercial interests that would be significantly affected by the Regional Transmission Organization's actions or decisions; and
- (ii) Any other entity that the Commission finds has economic or commercial interests that would be significantly affected by the Regional Transmission Organization's actions or decisions.
- (3) Affiliate means the definition given in section 2(a)(11) of the Public

Utility Holding Company Act (15 U.S.C. 79b(a)(11)).

- (4) Class of market participants means two or more market participants with common economic or commercial interests.
- (c) General rule. Except for those public utilities subject to the requirements of paragraph (h) of this section, every public utility that owns, operates or controls facilities used for the transmission of electric energy in interstate commerce as of March 6, 2000 must file with the Commission, no later than October 15, 2000, one of the following:
- (1) A proposal to participate in a Regional Transmission Organization consisting of one of the types of submittals set forth in paragraph (d) of this section; or
- (2) An alternative filing consistent with paragraph (g) of this section.
- (d) Proposal to participate in a Regional Transmission Organization. For purposes of this section, a proposal to participate in a Regional Transmission Organization means:
- (1) Such filings, made individually or jointly with other entities, pursuant to sections 203, 205 and 206 of the Federal Power Act (16 U.S.C. 824b, 824d, and 824e), as are necessary to create a new Regional Transmission Organization;
- (2) Such filings, made individually or jointly with other entities, pursuant to sections 203, 205 and 206 of the Federal Power Act (16 U.S.C. 824b, 824d, and 824e), as are necessary to join a Regional Transmission Organization approved by the Commission on or before the date of the filing; or
- (3) A petition for declaratory order, filed individually or jointly with other entities, asking whether a proposed transmission entity would qualify as a Regional Transmission Organization and containing at least the following:
- (i) A detailed description of the proposed transmission entity, including a description of the organizational and operational structure and the intended participants;
- (ii) A discussion of how the transmission entity would satisfy each of the characteristics and functions of a Regional Transmission Organization specified in paragraphs (j), (k)and (l) of this section;

- (iii) A detailed description of the Federal Power Act section 205 rates that will be filed for the Regional Transmission Organization; and
- (iv) A commitment to make filings pursuant to sections 203, 205 and 206 of the Federal Power Act (16 U.S.C. 824b, 824d, and 824e), as necessary, promptly after the Commission issues an order in response to the petition.
- (4) Any proposal filed under this paragraph (d) must include an explanation of efforts made to include public power entities and electric power cooperatives in the proposed Regional Transmission Organization.
 - (e) [Reserved]
- (f) Transfer of operational control. Any public utility's proposal to participate in a Regional Transmission Organization filed pursuant to paragraph (c)(1) of this section must propose that operational control of that public utility's transmission facilities will be transferred to the Regional Transmission Organization on a schedule that will allow the Regional Transmission Organization to commence operating the facilities no later than December 15, 2001.

NOTE TO PARAGRAPH (f): The requirement in paragraph (f) of this section may be satisfied by proposing to transfer to the Regional Transmission Organization ownership of the facilities in addition to operational control.

- (g) Alternative filing. Any filing made pursuant to paragraph (c)(2) of this section must contain:
- (1) A description of any efforts made by that public utility to participate in a Regional Transmission Organization;
- (2) A detailed explanation of the economic, operational, commercial, regulatory, or other reasons the public utility has not made a filing to participate in a Regional Transmission Organization, including identification of any existing obstacles to participation in a Regional Transmission Organization; and
- (3) The specific plans, if any, the public utility has for further work toward participation in a Regional Transmission Organization, a proposed timetable for such activity, an explanation of efforts made to include public power entities in the proposed Regional Transmission Organization, and any factors (including any law, rule or regulation) that may affect the public

utility's ability or decision to participate in a Regional Transmission Organization.

- (h) Public utilities participating in approved transmission entities. Every public utility that owns, operates or controls facilities used for the transmission of electric energy in interstate commerce as of March 6, 2000, and that has filed with the Commission on or before March 6, 2000 to transfer operational control of its facilities to a transmission entity that has been approved or conditionally approved by the Commission on or before March 6, 2000 as being in conformance with the eleven ISO principles set forth in Order No. 888, FERC Statutes and Regulations, Regulations Preamble January 1991-June 1996 ¶31,036 (Final Rule on Open Access and Stranded Costs; see 61 FR 21540, May 10, 1996), must, individually or jointly with other entities, file with the Commission, no later than January 15, 2001:
- (1) A statement that it is participating in a transmission entity that has been so approved;
- (2) A detailed explanation of the extent to which the transmission entity in which it participates has the characteristics and performs the functions of a Regional Transmission Organization specified in paragraphs (j) and (k) of this section and accommodates the open architecture conditions in paragraph (l) of this section; and
- (3) To the extent the transmission entity in which the public utility participates does not meet all the requirements of a Regional Transmission Organization specified in paragraphs (j), (k), and (l) of this section,
- (i) A proposal to participate in a Regional Transmission Organization that meets such requirements in accordance with paragraph (d) of this section,
- (ii) A proposal to modify the existing transmission entity so that it conforms to the requirements of a Regional Transmission Organization, or
- (iii) A filing containing the information specified in paragraph (g) of this section addressing any efforts, obstacles, and plans with respect to conformance with those requirements.
- (i) Entities that become public utilities with transmission facilities. An entity that is not a public utility that owns,

operates or controls facilities used for the transmission of electric energy in interstate commerce as of March 6, 2000, but later becomes such a public utility, must file a proposal to participate in a Regional Transmission Organization in accordance with paragraph (d) of this section, or an alternative filing in accordance with paragraph (g) of this section, by October 15, 2000 or 60 days prior to the date on which the public utility engages in any transmission of electric energy in interstate commerce, whichever comes later. If a proposal to participate in accordance with paragraph (d) of this section is filed, it must propose that operational control of the applicant's transmission system will be transferred to the Regional Transmission Organization within six months of filing the proposal.

- (j) Required characteristics for a Regional Transmission Organization. A Regional Transmission Organization must satisfy the following characteristics when it commences operation:
- (1) Independence. The Regional Transmission Organization must be independent of any market participant. The Regional Transmission Organization must include, as part of its demonstration of independence, a demonstration that it meets the following:
- (i) The Regional Transmission Organization, its employees, and any non-stakeholder directors must not have financial interests in any market participant.
- (ii) The Regional Transmission Organization must have a decision making process that is independent of control by any market participant or class of participants.
- (iii) The Regional Transmission Organization must have exclusive and independent authority under section 205 of the Federal Power Act (16 U.S.C. 824d), to propose rates, terms and conditions of transmission service provided over the facilities it operates.

Note to Paragraph (j)(1)(iii): Transmission owners retain authority under section 205 of the Federal Power Act (16 U.S.C. 824d) to seek recovery from the Regional Transmission Organization of the revenue requirements associated with the transmission facilities that they own.

- (iv)(A) The Regional Transmission Organization must provide:
- (1) With respect to any Regional Transmission Organization in which market participants have an ownership interest, a compliance audit of the independence of the Regional Transmission Organization's decision making process under paragraph (j)(1)(ii) of this section, to be performed two years after approval of the Regional Transmission Organization, and every three years thereafter, unless otherwise provided by the Commission.
- (2) With respect to any Regional Transmission Organization in which market participants have a role in the Regional Transmission Organization's decision making process but do not have an ownership interest, a compliance audit of the independence of the Regional Transmission Organization's decision making process under paragraph (j)(1)(ii) of this section, to be performed two years after its approval as a Regional Transmission Organization.
- (B) The compliance audits under paragraph (j)(1)(iv)(A) of this section must be performed by auditors who are not affiliated with the Regional Transmission Organization or transmission facility owners that are members of the Regional Transmission Organization.
- (2) Scope and regional configuration. The Regional Transmission Organization must serve an appropriate region. The region must be of sufficient scope and configuration to permit the Regional Transmission Organization to maintain reliability, effectively perform its required functions, and support efficient and non-discriminatory power markets.
- (3) Operational authority. The Regional Transmission Organization must have operational authority for all transmission facilities under its control. The Regional Transmission Organization must include, as part of its demonstration of operational authority, a demonstration that it meets the following:
- (i) If any operational functions are delegated to, or shared with, entities other than the Regional Transmission Organization, the Regional Transmission Organization must ensure that this sharing of operational authority

- will not adversely affect reliability or provide any market participant with an unfair competitive advantage. Within two years after initial operation as a Regional Transmission Organization, the Regional Transmission Organization must prepare a public report that assesses whether any division of operational authority hinders the Regional Transmission Organization in providing reliable, non-discriminatory and efficiently priced transmission service.
- (ii) The Regional Transmission Organization must be the security coordinator for the facilities that it controls.
- (4) Short-term reliability. The Regional Transmission Organization must have exclusive authority for maintaining the short-term reliability of the grid that it operates. The Regional Transmission Organization must include, as part of its demonstration with respect to reliability, a demonstration that it meets the following:
- (i) The Regional Transmission Organization must have exclusive authority for receiving, confirming and implementing all interchange schedules.
- (ii) The Regional Transmission Organization must have the right to order redispatch of any generator connected to transmission facilities it operates if necessary for the reliable operation of these facilities.
- (iii) When the Regional Transmission Organization operates transmission facilities owned by other entities, the Regional Transmission Organization must have authority to approve or disapprove all requests for scheduled outages of transmission facilities to ensure that the outages can be accommodated within established reliability standards.
- (iv) If the Regional Transmission Organization operates under reliability standards established by another entity (e.g., a regional reliability council), the Regional Transmission Organization must report to the Commission it hese standards hinder it from providing reliable, non-discriminatory and efficiently priced transmission service.
- (k) Required functions of a Regional Transmission Organization. The Regional Transmission Organization must perform the following functions. Unless otherwise noted, the Regional Transmission Organization must satisfy

these obligations when it commences operations.

- (1) Tariff administration and design. The Regional Transmission Organization must administer its own transmission tariff and employ a transmission pricing system that will promote efficient use and expansion of transmission and generation facilities. As part of its demonstration with respect to tariff administration and design, the Regional Transmission Organization must satisfy the standards listed in paragraphs (k)(1)(i) and (ii) of this section, or demonstrate that an alternative proposal is consistent with or superior to satisfying such standards.
- (i) The Regional Transmission Organization must be the only provider of transmission service over the facilities under its control, and must be the sole administrator of its own Commission-approved open access transmission tariff. The Regional Transmission Organization must have the sole authority to receive, evaluate, and approve or deny all requests for transmission service. The Regional Transmission Organization must have the authority to review and approve requests for new interconnections.
- (ii) Customers under the Regional Transmission Organization tariff must not be charged multiple access fees for the recovery of capital costs for transmission service over facilities that the Regional Transmission Organization controls
- (2) Congestion management. The Regional Transmission Organization must ensure the development and operation of market mechanisms to manage transmission congestion. As part of its demonstration with respect to congestion management, the Regional Transmission Organization must satisfy the standards listed in paragraph (k)(2)(i) of this section, or demonstrate that an alternative proposal is consistent with or superior to satisfying such standards.
- (i) The market mechanisms must accommodate broad participation by all market participants, and must provide all transmission customers with efficient price signals that show the consequences of their transmission usage decisions. The Regional Transmission Organization must either operate such

markets itself or ensure that the task is performed by another entity that is not affiliated with any market participant.

- (ii) The Regional Transmission Organization must satisfy the market mechanism requirement no later than one year after it commences initial operation. However, it must have in place at the time of initial operation an effective protocol for managing congestion.
- (3) Parallel path flow. The Regional Transmission Organization must develop and implement procedures to address parallel path flow issues within its region and with other regions. The Regional Transmission Organization must satisfy this requirement with respect to coordination with other regions no later than three years after it commences initial operation.
- (4) Ancillary services. The Regional Transmission Organization must serve as a provider of last resort of all ancillary services required by Order No. 888, FERC Statutes and Regulations, Regulations Preamble January 1991-June 1996 ¶31,036 (Final Rule on Open Access and Stranded Costs; see 61 FR 21540, May 10, 1996), and subsequent orders. As part of its demonstration with respect to ancillary services, the Regional Transmission Organization must satisfy the standards listed in paragraphs (k)(4)(i) through (iii) of this section, or demonstrate that an alternative proposal is consistent with or superior to satisfying such standards.
- (i) All market participants must have the option of self-supplying or acquiring ancillary services from third parties subject to any restrictions imposed by the Commission in Order No. 888, FERC Statutes and Regulations, Regulations Preamble January 1991–June 1996 ¶31,036 (Final Rule on Open Access and Stranded Costs), and subsequent orders.
- (ii) The Regional Transmission Organization must have the authority to decide the minimum required amounts of each ancillary service and, if necessary, the locations at which these services must be provided. All ancillary service providers must be subject to direct or indirect operational control by the Regional Transmission Organization. The Regional Transmission

Organization must promote the development of competitive markets for ancillary services whenever feasible.

- (iii) The Regional Transmission Organization must ensure that its transmission customers have access to a real-time balancing market. The Regional Transmission Organization must either develop and operate this market itself or ensure that this task is performed by another entity that is not affiliated with any market participant.
- (5) OASIS and Total Transmission Capability (TTC) and Available Transmission Capability (ATC). The Regional Transmission Organization must be the single OASIS site administrator for all transmission facilities under its control and independently calculate TTC and ATC.
- (6) Market monitoring. To ensure that the Regional Transmission Organization provides reliable, efficient and not unduly discriminatory transmission service, the Regional Transmission Organization must provide for objective monitoring of markets it operates or administers to identify market design flaws, market power abuses and opportunities for efficiency improvements, and propose appropriate actions. As part of its demonstration with respect to market monitoring, the Regional Transmission Organization must satisfy the standards listed in paragraphs (k)(6)(i) through (k)(6)(iii) of this section, or demonstrate that an alternative proposal is consistent with or superior to satisfying such standards.
- (i) Market monitoring must include monitoring the behavior of market participants in the region, including transmission owners other than the Regional Transmission Organization, if any, to determine if their actions hinder the Regional Transmission Organization in providing reliable, efficient and not unduly discriminatory transmission service.
- (ii) With respect to markets the Regional Transmission Organization operates or administers, there must be a periodic assessment of how behavior in markets operated by others (e.g., bilateral power sales markets and power markets operated by unaffiliated power exchanges) affects Regional Transmission Organization operations and how Regional Transmission Organiza-

tion operations affect the efficiency of power markets operated by others.

- (iii) Reports on opportunities for efficiency improvement, market power abuses and market design flaws must be filed with the Commission and affected regulatory authorities.
- (7) Planning and expansion. The Regional Transmission Organization must be responsible for planning, and for directing or arranging, necessary transmission expansions, additions, and upgrades that will enable it to provide efficient, reliable and non-discriminatory transmission service and coordinate such efforts with the appropriate state authorities. As part of its demonstration with respect to planning and expansion, the Regional Transmission Organization must satisfy the standards listed in paragraphs (k)(7)(i) and (ii) of this section, or demonstrate that an alternative proposal is consistent with or superior to satisfying such standards.
- (i) The Regional Transmission Organization planning and expansion process must encourage market-driven operating and investment actions for preventing and relieving congestion.
- (ii) The Regional Transmission Organization's planning and expansion process must accommodate efforts by state regulatory commissions to create multi-state agreements to review and approve new transmission facilities. The Regional Transmission Organization's planning and expansion process must be coordinated with programs of existing Regional Transmission Groups (See § 2.21 of this chapter) where appropriate.
- (iii) If the Regional Transmission Organization is unable to satisfy this requirement when it commences operation, it must file with the Commission a plan with specified milestones that will ensure that it meets this requirement no later than three years after initial operation.
- (8) Interregional coordination. The Regional Transmission Organization must ensure the integration of reliability practices within an interconnection and market interface practices among regions.
- (1) Open architecture. (1) Any proposal to participate in a Regional Transmission Organization must not contain

any provision that would limit the capability of the Regional Transmission Organization to evolve in ways that would improve its efficiency, consistent with the requirements in paragraphs (j) and (k) of this section.

(2) Nothing in this regulation precludes an approved Regional Transmission Organization from seeking to evolve with respect to its organizational design, market design, geographic scope, ownership arrangements, or methods of operational control, or in other appropriate ways if the change is consistent with the requirements of this section. Any future filing seeking approval of such changes must demonstrate that the proposed changes will meet the requirements of paragraphs (j), (k) and (l) of this section.

[Order 2000-A, 65 FR 12110, Mar. 8, 2000, as amended by Order 679, 71 FR 43338, July 31, 2006]

Subpart G—Transmission Infrastructure Investment Provisions

\$35.35 Transmission infrastructure investment.

- (a) Purpose. This section establishes rules for incentive-based (including performance-based) rate treatments for transmission of electric energy in interstate commerce by public utilities for the purpose of benefiting consumers by ensuring reliability and reducing the cost of delivered power by reducing transmission congestion.
- (b) Definitions. (1) Transco means a stand-alone transmission company that has been approved by the Commission and that sells transmission services at wholesale and/or on an unbundled retail basis, regardless of whether it is affiliated with another public utility.
- (2) Transmission Organization means a Regional Transmission Organization, Independent System Operator, independent transmission provider, or other transmission organization finally approved by the Commission for the operation of transmission facilities.
- (c) General rule. All rates approved under the rules of this section, including any revisions to the rules, are subject to the filing requirements of sections 205 and 206 of the Federal Power

Act and to the substantive requirements of sections 205 and 206 of the Federal Power Act that all rates, charges, terms and conditions be just and reasonable and not unduly discriminatory or preferential.

- (d) Incentive-based rate treatments for transmission infrastructure investment. The Commission will authorize any incentive-based rate treatment, as discussed in this paragraph (d), for transmission infrastructure investment, provided that the proposed incentivebased rate treatment is just and reasonable and not unduly discriminatory or preferential. A public utility's request for one or more incentive-based rate treatments, to be made in a filing pursuant to section 205 of the Federal Power Act, or in a petition for a declaratory order that precedes a filing pursuant to section 205, must include a detailed explanation of how the proposed rate treatment complies with the requirements of section 219 of the Federal Power Act and a demonstration that the proposed rate treatment is just, reasonable, and not unduly discriminatory or preferential. The applicant must demonstrate that the facilities for which it seeks incentives either ensure reliability or reduce the cost of delivered power by reducing transmission congestion consistent with the requirements of section 219, that the total package of incentives is tailored to address the demonstrable risks or challenges faced by the applicant in undertaking the project, and that resulting rates are just and reasonable. For purposes of this paragraph (d), incentive-based rate treatment means any of the following:
- (1) For purposes of this paragraph (d), incentive-based rate treatment means any of the following:
- (i) A rate of return on equity sufficient to attract new investment in transmission facilities:
- (ii) 100 percent of prudently incurred Construction Work in Progress (CWIP) in rate base;
- (iii) Recovery of prudently incurred pre-commercial operations costs:
- (iv) Hypothetical capital structure;
- (v) Accelerated depreciation used for rate recovery;
- (vi) Recovery of 100 percent of prudently incurred costs of transmission

facilities that are cancelled or abandoned due to factors beyond the control of the public utility;

- (vii) Deferred cost recovery; and
- (viii) Any other incentives approved by the Commission, pursuant to the requirements of this paragraph, that are determined to be just and reasonable and not unduly discriminatory or preferential.
- (2) In addition to the incentives in §35.35(d)(1), the Commission will authorize the following incentive-based rate treatments for Transcos, provided that the proposed incentive-based rate treatment is just and reasonable and not unduly discriminatory or preferential:
- (i) A return on equity that both encourages Transco formation and is sufficient to attract investment; and
- (ii) An adjustment to the book value of transmission assets being sold to a Transco to remove the disincentive associated with the impact of accelerated depreciation on federal capital gains tax liabilities.
- (e) Incentives for joining a Transmission Organization. The Commission will authorize an incentive-based rate treatment, as discussed in this paragraph (e), for public utilities that join a Transmission Organization, if the applicant demonstrates that the proposed incentive-based rate treatment is just and reasonable and not unduly discriminatory or preferential. Applicants for the incentive-based rate treatment must make a filing with the Commission under section 205 of the Federal Power Act. For purposes of this paragraph (e), an incentive-based rate treatment means a return on equity that is higher than the return on equity the Commission might otherwise allow if the public utility did not join Transmission Organization. The Commission will also permit transmitting utilities or electric utilities that join a Transmission Organization the ability to recover prudently incurred costs associated with joining Transmission Organization, through transmission rates charged by transmitting utilities or electric utilities or through transmission rates charged by the Transmission Organization that provides services to such util-

- (f) Approval of prudently-incurred costs. The Commission will approve recovery of prudently-incurred costs necessary to comply with the mandatory reliability standards pursuant to section 215 of the Federal Power Act, provided that the proposed rates are just and reasonable and not unduly discriminatory or preferential.
- (g) Approval of prudently incurred costs related to transmission infrastructure development. The Commission will approve recovery of prudently-incurred costs related to transmission infrastructure development pursuant to section 216 of the Federal Power Act, provided that the proposed rates are just and reasonable and not unduly discriminatory or preferential.
- (h) FERC-730, Report of transmission investment activity. Public utilities that have been granted incentive rate treatment for specific transmission projects must file FERC-730 on an annual basis beginning with the calendar year incentive rate treatment is granted by the Commission. Such filings are due by April 18 of the following calendar year and are due April 18 each year thereafter. The following information must be filed:
- (1) In dollar terms, actual transmission investment for the most recent calendar year, and projected, incremental investments for the next five calendar years;
- (2) For all current and projected investments over the next five calendar years, a project by project listing that specifies for each project the most upto-date, expected completion date, percentage completion as of the date of filing, and reasons for delays. Exclude from this listing projects with projected costs less than \$20 million; and
- (3) For good cause shown, the Commission may extend the time within which any FERC-730 filing is to be filed or waive the requirements applicable to any such filing.
- (i) Rebuttable presumption. (1) The Commission will apply a rebuttable presumption that an applicant has demonstrated that its project is needed to ensure reliability or reduces the cost of delivered power by reducing congestion for:
- (i) A transmission project that results from a fair and open regional

planning process that considers and evaluates projects for reliability and/or congestion and is found to be acceptable to the Commission; or

- (ii) A project that has received construction approval from an appropriate state commission or state siting authority.
- (2) To the extent these approval processes do not require that a project ensures reliability or reduce the cost of delivered power by reducing congestion, the applicant bears the burden of demonstrating that its project satisfies these criteria.
- (j) Commission authorization to site electric transmission facilities in interstate commerce. If the Commission pursuant to its authority under section 216 of the Federal Power Act and its regulations thereunder has issued one or more permits for the construction or modification of transmission facilities in a national interest electric transmission corridor designated by the Secretary, such facilities shall be deemed to either ensure reliability or reduce the cost of delivered power by reducing congestion for purposes of section 219(a).

[Order 679, 71 FR 43338, July 31, 2006, as amended by Order 679-A, 72 FR 1172, Jan. 10, 2007, Order 691, 72 FR 5174, Feb. 5, 2007]

Subpart H—Wholesale Sales of Electric Energy, Capacity and Ancillary Services at Market-Based Rates

Source: Order 697, 72 FR 40038, July 20, 2007, unless otherwise noted.

§35.36 Generally.

- (a) For purposes of this subpart:
- (1) Seller means any person that has authorization to or seeks authorization to engage in sales for resale of electric energy, capacity or ancillary services at market-based rates under section 205 of the Federal Power Act.
- (2) Category 1 Sellers means wholesale power marketers and wholesale power producers that own or control 500 MW or less of generation in aggregate per region; that do not own, operate or control transmission facilities other than limited equipment necessary to connect individual generating facilities

to the transmission grid (or have been granted waiver of the requirements of Order No. 888, FERC Stats. & Regs. ¶ 31,036); that are not affiliated with anyone that owns, operates or controls transmission facilities in the same region as the seller's generation assets; that are not affiliated with a franchised public utility in the same region as the seller's generation assets; and that do not raise other vertical market power issues.

- (3) Category 2 Sellers means any Sellers not in Category 1.
- (4) Inputs to electric power production means intrastate natural gas transportation, intrastate natural gas storage or distribution facilities; sites for generation capacity development; physical coal supply sources and ownership of or control over who may access transportation of coal supplies.
- (5) Franchised public utility means a public utility with a franchised service obligation under State law.
- (6) Captive customers means any wholesale or retail electric energy customers served by a franchised public utility under cost-based regulation.
- (7) Market-regulated power sales affiliate means any power seller affiliate other than a franchised public utility, including a power marketer, exempt wholesale generator, qualifying facility or other power seller affiliate, whose power sales are regulated in whole or in part on a market-rate basis.
- (8) Market information means non-public information related to the electric energy and power business including, but not limited to, information regarding sales, cost of production, generator outages, generator heat rates, unconsummated transactions, or historical generator volumes. Market information includes information from either affiliates or non-affiliates.
- (9) Affiliate of a specified company means:
- (i) Any person that directly or indirectly owns, controls, or holds with power to vote, 10 percent or more of the outstanding voting securities of the specified company:
- (ii) Any company 10 percent or more of whose outstanding voting securities are owned, controlled, or held with power to vote, directly or indirectly, by the specified company;

- (iii) Any person or class of persons that the Commission determines, after appropriate notice and opportunity for hearing, to stand in such relation to the specified company that there is liable to be an absence of arm's-length bargaining in transactions between them as to make it necessary or appropriate in the public interest or for the protection of investors or consumers that the person be treated as an affiliate; and
- (iv) Any person that is under common control with the specified company.
- (v) For purposes of paragraph (a)(9), owning, controlling or holding with power to vote, less than 10 percent of the outstanding voting securities of a specified company creates a rebuttable presumption of lack of control.
- (b) The provisions of this subpart apply to all Sellers authorized, or seeking authorization, to make sales for resale of electric energy, capacity or ancillary services at market-based rates unless otherwise ordered by the Commission.

[Order 697, 72 FR 40038, July 20, 2007, as amended by Order 697–A, 73 FR 25912, May 7, 2008; Order 697–B, 73 FR 79627, Dec. 30, 2008]

§ 35.37 Market power analysis required.

- (a) (1) In addition to other requirements in subparts A and B, a Seller must submit a market power analysis in the following circumstances: when seeking market-based rate authority; for Category 2 Sellers, every three years, according to the schedule contained in Order No. 697, FERC Stats. & Regs. ¶ 31,252; or any other time the Commission directs a Seller to submit one. Failure to timely file an updated market power analysis will constitute a violation of Seller's market-based rate tariff.
- (2) When submitting a market power analysis, whether as part of an initial application or an update, a Seller must include an appendix of assets in the form provided in Appendix B of this subpart.
- (b) A market power analysis must address whether a Seller has horizontal and vertical market power.
- (c) (1) There will be a rebuttable presumption that a Seller lacks horizontal

- market power if it passes two indicative market power screens: a pivotal supplier analysis based on the annual peak demand of the relevant market, and a market share analysis applied on a seasonal basis. There will be a rebuttable presumption that a Seller possesses horizontal market power if it fails either screen.
- (2) Sellers and intervenors may also file alternative evidence to support or rebut the results of the indicative screens. Sellers may file such evidence at the time they file their indicative screens. Intervenors may file such evidence in response to a Seller's submissions.
- (3) If a Seller does not pass one or both screens, the Seller may rebut a presumption of horizontal market power by submitting a Delivered Price Test analysis. A Seller that does not rebut a presumption of horizontal market power or that concedes market power, is subject to mitigation, as described in §35.38.
- (4) When submitting a horizontal market power analysis, a Seller must use the form provided in Appendix A of this subpart and include all supporting materials referenced in the form.
- (d) To demonstrate a lack of vertical market power, a Seller that owns, operates or controls transmission facilities, or whose affiliates own, operate or control transmission facilities, must have on file with the Commission an Open Access Transmission Tariff. as described in §35.28; provided, however, that a Seller whose foreign affiliate(s) own, operate or control transmission facilities outside of the United States that can be used by competitors of the Seller to reach United States markets must demonstrate that such affiliate either has adopted and is implementing an Open Access Transmission Tariff as described in §35.28, or otherwise offers comparable, non-discriminatory access to such transmission facilities.
- (e) To demonstrate a lack of vertical market power in wholesale energy markets through the affiliation, ownership or control of inputs to electric power production, such as the transportation or distribution of the inputs to electric power production, a Seller must provide the following information:

- (1) A description of its ownership or control of, or affiliation with an entity that owns or controls, intrastate natural gas transportation, intrastate natural gas storage or distribution facilities:
- (2) Sites for generation capacity development; and
- (3) Physical coal supply sources and ownership or control over who may access transportation of coal supplies.
- (4) A Seller must ensure that this information is included in the record of each new application for market-based rates and each updated market power analysis. In addition, a Seller is required to make an affirmative statement that it has not erected barriers to entry into the relevant market and will not erect barriers to entry into the relevant market.
- (f) If the seller seeks to protect any portion of the application, or any attachment thereto, from public disclosure pursuant to §388.112 of this chapter, the seller must include with its request for privileged treatment a proposed protective order under which the parties to the proceeding will be able to review any of the data, information, analysis or other documentation relied upon by the seller for which privileged treatment is sought. A seller must grant access to privileged data to any party that signs a protective order within 5 days from the date that the party executes the protective order.

[Order 697, 72 FR 40038, July 20, 2007, as amended by Order 697–B, 73 FR 79627, Dec. 30, 2008]

§35.38 Mitigation.

- (a) A Seller that has been found to have market power in generation or that is presumed to have horizontal market power by virtue of failing or foregoing the horizontal market power screens, as described in §35.37(c), may adopt the default mitigation detailed in paragraph (b) of this section or may propose mitigation tailored to its own particular circumstances to eliminate its ability to exercise market power. Mitigation will apply only to the market(s) in which the Seller is found, or presumed, to have market power.
- (b) Default mitigation consists of three distinct products:

- (1) Sales of power of one week or less priced at the Seller's incremental cost plus a 10 percent adder;
- (2) Sales of power of more than one week but less than one year priced at no higher than a cost-based ceiling reflecting the costs of the unit(s) expected to provide the service; and
- (3) New contracts filed for review under section 205 of the Federal Power Act for sales of power for one year or more priced at a rate not to exceed embedded cost of service.

§35.39 Affiliate restrictions.

- (a) General affiliate provisions. As a condition of obtaining and retaining market-based rate authority, the conditions provided in this section, including the restriction on affiliate sales of electric energy and all other affiliate provisions, must be satisfied on an ongoing basis, unless otherwise authorized by Commission rule or order. Failure to satisfy these conditions will constitute a violation of the Seller's market-based rate tariff.
- (b) Restriction on affiliate sales of electric energy or capacity. As a condition of obtaining and retaining market-based rate authority, no wholesale sale of electric energy or capacity may be made between a franchised public utility with captive customers and a market-regulated power sales affiliate without first receiving Commission authorization for the transaction under section 205 of the Federal Power Act. All authorizations to engage in affiliate wholesale sales of electric energy or capacity must be listed in a Seller's market-based rate tariff.
- (c) Separation of functions. (1) For the purpose of this paragraph, entities acting on behalf of and for the benefit of a franchised public utility with captive customers (such as entities controlling or marketing power from the electrical generation assets of the franchised public utility) are considered part of the franchised public utility. Entities acting on behalf of and for the benefit of the market-regulated power sales affiliates of a franchised public utility with captive customers are considered part of the market-regulated power sales affiliates.

- (2) (i) To the maximum extent practical, the employees of a market-regulated power sales affiliate must operate separately from the employees of any affiliated franchised public utility with captive customers.
- (ii) Franchised public utilities with captive customers are permitted to share support employees, and field and maintenance employees with their market-regulated power sales affiliates. Franchised public utilities with captive customers are also permitted to share senior officers and boards of directors with their market-regulated power sales affiliates; provided, however, that the shared officers and boards of directors must not participate in directing, organizing or executing generation or market functions.
- (iii) Notwithstanding any other restrictions in this section, in emergency circumstances affecting system reliability, a market-regulated power sales affiliate and a franchised public utility with captive customers may take steps necessary to keep the bulk power system in operation. A franchised public utility with captive customers or the market-regulated power sales affiliate must report to the Commission and disclose to the public on its Web site, each emergency that resulted in any deviation from the restrictions of section 35.39, within 24 hours of such deviation.
- (d) Information sharing. (1) A franchised public utility with captive customers may not share market information with a market-regulated power sales affiliate if the sharing could be used to the detriment of captive customers, unless simultaneously disclosed to the public.
- (2) Permissibly shared support employees, field and maintenance employees and senior officers and board of directors under §\$35.39(c)(2)(ii) may have access to information covered by the prohibition of §35.39(d)(1), subject to the no-conduit provision in §35.39(g).
- (e) Non-power goods or services. (1) Unless otherwise permitted by Commission rule or order, sales of any non-power goods or services by a franchised public utility with captive customers, to a market-regulated power sales affiliate must be at the higher of cost or market price.

- (2) Unless otherwise permitted by Commission rule or order, sales of any non-power goods or services by a market-regulated power sales affiliate to an affiliated franchised public utility with captive customers may not be at a price above market.
- (f) Brokering of power. (1) Unless otherwise permitted by Commission rule or order, to the extent a market-regulated power sales affiliate seeks to broker power for an affiliated franchised public utility with captive customers:
- (i) The market-regulated power sales affiliate must offer the franchised public utility's power first;
- (ii) The arrangement between the market-regulated power sales affiliate and the franchised public utility must be non-exclusive; and
- (iii) The market-regulated power sales affiliate may not accept any fees in conjunction with any brokering services it performs for an affiliated franchised public utility.
- (2) Unless otherwise permitted by Commission rule or order, to the extent a franchised public utility with captive customers seeks to broker power for a market-regulated power sales affiliate:
- (i) The franchised public utility must charge the higher of its costs for the service or the market price for such services;
- (ii) The franchised public utility must market its own power first, and simultaneously make public (on the Internet) any market information shared with its affiliate during the brokering; and
- (iii) The franchised public utility must post on the Internet the actual brokering charges imposed.
- (g) No conduit provision. A franchised public utility with captive customers and a market-regulated power sales affiliate are prohibited from using anyone, including asset managers, as a conduit to circumvent the affiliate restrictions in §§ 35.39(a) through (g).
- (h) Franchised utilities without captive customers. If necessary, any affiliate restrictions regarding separation of functions, power sales or non-power goods and services transactions, or brokering involving two or more franchised public utilities, one or more of whom has

captive customers and one or more of whom does not have captive customers, will be imposed on a case-by-case basis.

[Order 697, 72 FR 40038, July 20, 2007, as amended by Order 697-A, 73 FR 25912, May 7, 2008]

§ 35.40 Ancillary services.

A Seller may make sales of ancillary services at market-based rates only if it has been authorized by the Commission and only in specific geographic markets as the Commission has authorized.

§35.41 Market behavior rules.

(a) Unit operation. Where a Seller participates in a Commission-approved organized market, Seller must operate and schedule generating facilities, undertake maintenance, declare outages, and commit or otherwise bid supply in a manner that complies with the Commission-approved rules and regulations of the applicable market. A Seller is not required to bid or supply electric energy or other electricity products unless such requirement is a part of a separate Commission-approved tariff or is a requirement applicable to Seller through Seller's participation in a Commission-approved organized mar-

(b) Communications. A Seller must provide accurate and factual information and not submit false or misleading information, or omit material information, in any communication with the Commission-approved Commission. Commission-apmarket monitors, proved regional transmission organiza-Commission-approved tions. independent system operators, or jurisdictional transmission providers, unless Seller exercises due diligence to prevent such occurrences.

(c) Price reporting. To the extent a Seller engages in reporting of transactions to publishers of electric or natural gas price indices, Seller must provide accurate and factual information, and not knowingly submit false or misleading information or omit material information to any such publisher, by reporting its transactions in a manner consistent with the procedures set forth in the Policy Statement issued by the Commission in Docket No. PL03–3–000 and any clarifications

thereto. Unless Seller has previously provided the Commission with a notification of its price reporting status, Seller must notify the Commission within 15 days of the effective date of this regulation or within 15 days of the date it begins making wholesale sales, whichever is earlier, whether it engages in such reporting of its transactions. Seller must update the notification within 15 days of any subsequent change in its transaction reporting status. In addition, Seller must adhere to such other standards and requirements for price reporting as the Commission may order.

(d) Records retention. A Seller must retain, for a period of five years, all data and information upon which it billed the prices it charged for the electric energy or electric energy products it sold pursuant to Seller's market-based rate tariff, and the prices it reported for use in price indices.

§ 35.42 Change in status reporting requirement.

(a) As a condition of obtaining and retaining market-based rate authority, a Seller must timely report to the Commission any change in status that would reflect a departure from the characteristics the Commission relied upon in granting market-based rate authority. A change in status includes, but is not limited to, the following:

(1) Ownership or control of generation capacity that results in net increases of 100 MW or more, or of inputs to electric power production, or ownership, operation or control of transmission facilities, or

(2) Affiliation with any entity not disclosed in the application for market-based rate authority that owns or controls generation facilities or inputs to electric power production, affiliation with any entity not disclosed in the application for market-based rate authority that owns, operates or controls transmission facilities, or affiliation with any entity that has a franchised service area.

(b) Any change in status subject to paragraph (a) of this section, other than a change in status submitted to report the acquisition of control of a site or sites for new generation capacity development, must be filed no later

Federal Energy Regulatory Commission

Pt. 35, Subpt. H, App. A

than 30 days after the change in status occurs. Power sales contracts with future delivery are reportable 30 days after the physical delivery has begun. Failure to timely file a change in status report constitutes a tariff violation.

- (c) When submitting a change in status notification regarding a change that impacts the pertinent assets held by a Seller or its affiliates with market-based rate authorization, a Seller must include an appendix of assets in the form provided in Appendix B of this subpart.
- (d) A Seller must report on a quarterly basis the acquisition of control of a site or sites for new generation capacity development for which site control has been demonstrated in the interconnection process and for which the potential number of megawatts that are reasonably commercially feasible on the site or sites for new generation capacity development is equal

to 100 megawatts or more. If a Seller elects to make a monetary deposit so that it may demonstrate site control at a later time in the interconnection process, the monetary deposit will trigger the quarterly reporting requirement instead of the demonstration of site control. A notification of change in status that is submitted to report the acquisition of control of a site or sites for new generation capacity development must include:

- (1) The number of sites acquired;
- (2) The relevant geographic market in which the sites are located; and
- (3) The maximum potential number of megawatts (MW) that are reasonably commercially feasible on the sites reported.
- (e) For the purposes of paragraph (d) of this section, "control" shall mean "site control" as it is defined in the Standard Large Generator Interconnection Procedures (LGIP).

[Order 697–D, 75 FR 14351, Mar. 25, 2010]

APPENDIX A TO SUBPART H OF PART 35

APPENDIX A

Standard Screen Format

 $(Data\ provided\ for\ Illustrative\ Purposes\ only)$

PART I-PIVOTAL SUPPLIER ANALYSIS

Row	Generation	MW	Reference
Seller and	Affiliate Capacity		
A B C D	Installed Capacity Long-Term Firm Purchases Long-Term Firm Sales Imported Power	19,500 500 1,000 0	Workpaper. Workpaper. Workpaper. Workpaper.
Non-Affilia	ate Capacity		
E	Installed Capacity Long-Term Firm Purchases Long-Term Firm Sales Imported Power Balancing Authority Area Reserve Requirement Amount of Line I Attributable to Seller, if any Total Uncommitted Supply (SUM A,B,C,D,E,F,G,H,I,M)	8,000 500 -2,500 3,500 -2,160 -2,160 9,840	Workpaper. Workpaper. Workpaper. Workpaper. Workpaper. Workpaper.
L M N O P	Balancing Authority Area Annual Peak Load Average Daily Peak Native Load in Peak Month Amount of Line M Attributable to Seller, if any Wholesale Load (SUM L,M) Net Uncommitted Supply (K–O) Seller's Uncommitted Capacity (SUM A,B,C,D,J,N) Result of Pivotal Supplier Screen (Pass if Line Q < Line P), (Fail if Line Q > Line P)	18,000 - 16,500 - 16,500 1,500 8,340 340	Workpaper. Workpaper. Workpaper.

[Order 697, 72 FR 40038, July 20, 2007, as amended by Order 697-A, 73 FR 25913, May 7, 2008]

Pt. 35, Subpt. H, App. B

APPENDIX B TO SUBPART H OF PART 35

This is an example of the required appendix listing the filing entity and all its energy affiliates and their associated assets which should be submitted with all market-based rate filings.

MARKET-BASED RATE AUTHORITY AND GENERATION ASSETS

						Location			
Filing enti- ty and its energy affiliates	Docket No. where MBR au- thority was grant- ed	Genera- tion name	Owned by	Con- trolled by	Date control transferred	Balancing authority area	Geo- graphic region (per Ap- pendix D)	In-service date	Nameplate and/or seasonal rating
ABC Corp	ER05-23X-000	ABC falls plant #1.	ABC Corp.	ABC Corp.	NA*	ABC bal- ancing authority area.	Central	8/12/1981	153.5 MW (sea- sonal).
xyz Inc	ER94-79XX-000	NA	NA	NA	NA	NA	NA	NA	NA.
RST LLC	ER01-2XX5-000	Green CoGen.	WWW Corp.	RST LLC.	5/23/2005	New York ISO.	North- east.	12/20/2003	2000 MW (name- plate).
Sample Co	ER03-XX45-000	Sample Co. 3.	Sample Co.	YYY Corp.	2/1/1982	Sample Co. bal- ancing authority.	South- west.	5/13/1973	10 MW (sea- sonal).

^{*}If an entity has no assets or the field is not applicable please indicate so by inputting (NA).

ELECTRIC TRANSMISSION ASSETS AND/OR NATURAL GAS INTRASTATE PIPELINES AND/OR GAS STORAGE FACILITIES

				I			
Filing enti- ty and its energy affiliates	Asset name and use	Owned by	Controlled by	Date control trans- ferred	Location		
					Balancing authority area	Geo- graphic region (per Ap- pendix D)	Size
ABC Corp	CBA Line, used to interconnect Green Cogen to New York ISO transmission sys- tem.	ABC Corp	ABC Corp	NA*	New York ISO	Northeast	approximately five- mile, 500 kV line.
Etc. LP	Nowhere Pipeline, used to connect Storage LLC's— Longway Pipeline to ABC falls plant #1.	Etc. LP	Etc. LP	NA	ABC balancing authority area.	Central	approximately 14 miles of natural gas pipeline and related equipment with 50 MMcf/d capacity.

 $^{{}^\}star If$ the field is not applicable please indicate so by inputting (NA).

Subpart I—Cross-Subsidization Restrictions on Affiliate Transactions

Source: 73 FR 11025, Feb. 29, 2008, unless otherwise noted.

$\S 35.43$ Generally.

- (a) For purposes of this subpart:
- (1) Affiliate of a specified company means:
- (i) For any person other than an exempt wholesale generator:
- (A) Any person that directly or indirectly owns, controls, or holds with power to vote, 10 percent or more of the outstanding voting securities of the specified company;
- (B) Any company 10 percent or more of whose outstanding voting securities are owned, controlled, or held with

power to vote, directly or indirectly, by the specified company;

- (C) Any person or class of persons that the Commission determines, after appropriate notice and opportunity for hearing, to stand in such relation to the specified company that there is liable to be an absence of arm's-length bargaining in transactions between them as to make it necessary or appropriate in the public interest or for the protection of investors or consumers that the person be treated as an affiliate; and
- (D) Any person that is under common control with the specified company.
- (E) For purposes of paragraph (a)(1)(i) of this section, owning, controlling or holding with power to vote, less than 10 percent of the outstanding voting securities of a specified company creates a rebuttable presumption of lack of control.
- (ii) For any exempt wholesale generator (as defined under §366.1 of this chapter), consistent with section 214 of the Federal Power Act (16 U.S.C. 824m), which provides that "affiliate" will have the same meaning as provided in section 2(a) of the Public Utility Holding Company Act of 1935 (15 U.S.C. 79b(a)(11)):
- (A) Any person that directly or indirectly owns, controls, or holds with power to vote, 5 percent or more of the outstanding voting securities of the specified company:
- (B) Any company 5 percent or more of whose outstanding voting securities are owned, controlled, or held with power to vote, directly or indirectly, by the specified company;
- (C) Any individual who is an officer or director of the specified company, or of any company which is an affiliate thereof under paragraph (a)(1)(ii)(A) of this section; and
- (D) Any person or class of persons that the Commission determines, after appropriate notice and opportunity for hearing, to stand in such relation to the specified company that there is liable to be an absence of arm's-length bargaining in transactions between them as to make it necessary or appropriate in the public interest or for the protection of investors or consumers that the person be treated as an affiliate

- (2) Captive customers means any wholesale or retail electric energy customers served by a franchised public utility under cost-based regulation.
- (3) Franchised public utility means a public utility with a franchised service obligation under state law.
- (4) Market-regulated power sales affiliate means any power seller affiliate other than a franchised public utility, including a power marketer, exempt wholesale generator, qualifying facility or other power seller affiliate, whose power sales are regulated in whole or in part on a market-rate basis.
- (5) Non-utility affiliate means any affiliate that is not in the power sales or transmission business, other than a local gas distribution company or an interstate natural gas pipeline.
- (b) The provisions of this subpart apply to all franchised public utilities that have captive customers or that own or provide transmission service over jurisdictional transmission facilities.

§ 35.44 Protections against affiliate cross-subsidization.

- (a) Restriction on affiliate sales of electric energy. No wholesale sale of electric energy may be made between a franchised public utility with captive customers and a market-regulated power sales affiliate without first receiving Commission authorization for the transaction under section 205 of the Federal Power Act. This requirement does not apply to energy sales from a qualifying facility, as defined by 18 CFR 292.101, made under market-based rate authority granted by the Commission.
 - (b) Non-power goods or services.
- (1) Unless otherwise permitted by Commission rule or order, and except as permitted by paragraph (b)(4) of this section, sales of any non-power goods or services by a franchised public utility that has captive customers or that owns or provides transmission service over jurisdictional transmission facilities, including sales made to or through its affiliated exempt wholesale generators or qualifying facilities, to a market-regulated power sales affiliate or non-utility affiliate must be at the higher of cost or market price.

- (2) Unless otherwise permitted by Commission rule or order, and except as permitted by paragraphs (b)(3) and (b)(4) of this section, a franchised public utility that has captive customers or that owns or provides transmission service over jurisdictional transmission facilities, may not purchase or receive non-power goods and services from a market-regulated power sales affiliate or a non-utility affiliate at a price above market.
- (3) A franchised public utility that has captive customers or that owns or provides transmission service over jurisdictional transmission facilities, may only purchase or receive nonpower goods and services from a centralized service company at cost.
- (4) A company in a single-state holding company system, as defined in §366.3(c)(1) of this chapter, may provide general administrative and management non-power goods and services to, or receive such goods and services from, other companies in the same holding company system, at cost, provided that the only parties to transactions involving these non-power goods and services are affiliates or associate companies, as defined in §366.1 of this chapter, of a holding company in the holding company system.
- (c) Exemption for price under fuel adjustment clause regulations. Where the price of fuel from a company-owned or controlled source is found or presumed under §35.14 to be reasonable and includable in the adjustment clause, transactions involving that fuel shall be exempt from the affiliate price restrictions in §35.44(b).

[73 FR 11025, Feb. 29, 2008, as amended by Order 707–A, 73 FR 43083, July 24, 2008]

Subpart J—Credit Practices In Organized Wholesale Electric Markets

SOURCE: Order 741, 75 FR 65962, Oct. 27, 2010, unless otherwise noted

§ 35.45 Applicability.

This subpart establishes credit practices for organized wholesale electric markets for the purpose of minimizing risk to market participants.

§ 35.46 Definitions.

As used in this subpart:

- (a) Market Participant means an entity that qualifies as a Market Participant under §35.34.
- (b) Organized Wholesale Electric Market includes an independent system operator and a regional transmission organization.
- (c) Regional Transmission Organization means an entity that qualifies as a Regional Transmission Organization under 18 CFR 35.34.
- (d) Independent System Operator means an entity operating a transmission system and found by the Commission to be an Independent System Operator.

§ 35.47 Tariff provisions regarding credit practices in organized whole-sale electric markets.

Each organized wholesale electric market must have tariff provisions that:

- (a) Limit the amount of unsecured credit extended by an organized whole-sale electric market to no more than \$50 million for each market participant; where a corporate family includes more than one market participant participating in the same organized wholesale electric market, the limit on the amount of unsecured credit extended by that organized wholesale electric market shall be no more than \$50 million for the corporate family.
- (b) Adopt a billing period of no more than seven days and allow a settlement period of no more than seven days.
- (c) Eliminate unsecured credit in financial transmission rights markets and equivalent markets.
- (d) Establish a single counterparty to all market participant transactions, or require each market participant in an organized wholesale electric market to grant a security interest to the organized wholesale electric market in the receivables of its transactions, or provide another method of supporting netting that provides a similar level of protection to the market and is approved by the Commission. In the alternative, the organized wholesale electric market shall not net market

participants' transactions and must establish credit based on market participants' gross obligations.

- (e) Limit to no more than two days the time period provided to post additional collateral when additional collateral is requested by the organized wholesale electric market.
- (f) Require minimum participation criteria for market participants to be eligible to participate in the organized wholesale electric market.
- (g) Provide a list of examples of circumstances when a market administrator may invoke a "material adverse change" as a justification for requiring additional collateral; this list does not limit a market administrator's right to invoke such a clause in other circumstances.

[Order 741, 75 FR 65962, Oct. 27, 2010, as amended by Order 741–A, 76 FR 10498, Feb. 25, 2011]

PART 36—RULES CONCERNING AP-PLICATIONS FOR TRANSMISSION SERVICES UNDER SECTION 211 OF THE FEDERAL POWER ACT

AUTHORITY: 5 U.S.C. 551–557; 16 U.S.C. 791a–825r; 31 U.S.C. 9701; 42 U.S.C. 7107–7352.

§ 36.1 Notice provisions applicable to applications for transmission services under section 211 of the Federal Power Act.

- (a) Definitions. (1) Affected party means each affected electric utility, each affected State regulatory authority, and each affected Federal power marketing agency.
- (2) Affected electric utility means each electric utility that has made arrangements for the sale or purchase of electric energy to be transmitted pursuant to the particular application for transmission services, and each transmitting utility, as defined in section 3(23) of the Federal Power Act, 16 U.S.C. 796(23), being requested to transmit such electric energy.
- (3) Affected State regulatory authority means a State regulatory authority, as defined in section 3(21) of the Federal Power Act, 16 U.S.C. 796(21), regulating the rates and charges of each affected electric utility.

- (4) Affected Federal power marketing agency means a Federal power marketing agency that operates in the service area of each affected electric utility.
- (b) Additional filing requirements. Any person filing an application for transmission services pursuant to section 211 of the Federal Power Act, 16 U.S.C. 824j, shall include the following:
- (1) The applicant must include a form of notice of the application suitable for publication in the FEDERAL REGISTER in accordance with the specifications §385.203(d) of this chapter. The form of notice shall be on electronic media as specified by the Secretary.
- (2) A sworn statement that actual notice, including the applicant's name, the date of the application, the names of the affected parties, and a brief description of the transmission services sought (including the proposed dates for initiating and terminating the requested transmission services, the total amount of transmission capacity requested, a brief description of the character and nature of the transmission services being requested, and whether the transmission services requested are firm or non-firm) has been served, pursuant to Rule 2010 of the Commission's Rules of Practice and Procedure, §385.2010 of this chapter, on each affected party. Such statement shall enumerate each person so served.
- (c) Other filing requirements. All other filing requirements of the Commission's Rules of Practice and Procedure remain in effect for applications under this section.

[Order 560, 58 FR 57737, Oct. 27, 1993, as amended by Order 593, 62 FR 1283, Jan. 9, 1997; Order 647, 69 FR 32438, June 10, 2004]

EFFECTIVE DATE NOTE: By Order 560, 58 FR 57737, Oct. 27, 1993, §36.1 was added. The section contains information collection and recordkeeping requirements and will not become effective until approval has been given by the Office of Management and Budget.

PART 37—OPEN ACCESS SAME-TIME INFORMATION SYSTEMS

Sec.

37.1 Applicability.

37.2 Purpose.

37.3 Definitions. 37.4 [Reserved]

§ 37.1

- 37.5 Obligations of Transmission Providers and Responsible Parties.
- 37.6 Information to be posted on the OASIS.37.7 Auditing Transmission Service Information.
- 37.8 Obligations of OASIS users.

SOURCE: Order 889, 61 FR 21764, May 10, 1996, unless otherwise noted.

§ 37.1 Applicability.

This part applies to any public utility that owns, operates, or controls facilities used for the transmission of electric energy in interstate commerce and to transactions performed under the *pro forma* tariff required in part 35 of this chapter.

§37.2 Purpose.

- (a) The purpose of this part is to ensure that potential customers of open access transmission service receive access to information that will enable them to obtain transmission service on a non-discriminatory basis from any Transmission Provider. These rules provide standards of conduct and require the Transmission Provider (or its agent) to create and operate an Open Access Same-time Information System (OASIS) that gives all users of the open access transmission system access to the same information.
- (b) The OASIS will provide information by electronic means about available transmission capability for point-to-point service and will provide a process for requesting transmission service. OASIS will enable Transmission Providers and Transmission Customers to communicate promptly requests and responses to buy and sell available transmission capacity offered under the Transmission Provider's tariff.

§ 37.3 Definitions.

- (a) Transmission Provider means any public utility that owns, operates, or controls facilities used for the transmission of electric energy in interstate commerce.
- (b) Transmission Customer means any eligible customer (or its designated agent) that can or does execute a transmission service agreement or can or does receive transmission service.

- (c) Responsible party means the Transmission Provider or an agent to whom the Transmission Provider has delegated the responsibility of meeting any of the requirements of this part.
- (d) Reseller means any Transmission Customer who offers to sell transmission capacity it has purchased.
- (e) Wholesale merchant function means the sale for resale of electric energy in interstate commerce.
 - (f) Affiliate means:
- (1) For any exempt wholesale generator, as defined under section 32(a) of the Public Utility Holding Company Act of 1935, as amended, the same as provided in section 214 of the Federal Power Act; and
- (2) For any other entity, the term *affiliate* has the same meaning as given in §161.2(a) of this chapter.

[Order 889, 61 FR 21764, May 10, 1996, as amended by Order 889-A, 62 FR 12503, Mar. 14, 1997]

§37.4 [Reserved]

§ 37.5 Obligations of Transmission Providers and Responsible Parties.

- (a) Each Transmission Provider is required to provide for the operation of an OASIS, either individually or jointly with other Transmission Providers, in accordance with the requirements of this Part. The Transmission Provider may delegate this responsibility to a Responsible Party such as another Transmission Provider, an Independent System Operator, a Regional Transmission Group, or a Regional Reliability Council.
- (b) A Responsible Party must provide access to an OASIS providing standardized information relevant to the availability of transmission capacity, prices, and other information (as described in this part) pertaining to the transmission system for which it is responsible.
- (c) A Responsible Party may not deny or restrict access to an OASIS user merely because that user makes automated computer-to-computer file transfers or queries, or extensive requests for data.
- (d) In the event that an OASIS user's grossly inefficient method of accessing an OASIS node or obtaining information from the node seriously degrades

the performance of the node, a Responsible Party may limit a user's access to the OASIS node without prior Commission approval. The Responsible Party must immediately contact the OASIS user to resolve the problem. Notification of the restriction must be made to the Commission within two business days of the incident and include a description of the problem. A closure report describing how the problem was resolved must be filed with the Commission within one week of the incident.

- (e) In the event that an OASIS user makes an error in a query, the Responsible Party can block the affected query and notify the user of the nature of the error. The OASIS user must correct the error before making any additional queries. If there is a dispute over whether an error has occurred, the procedures in paragraph (d) of this section apply.
- (f) Transmission Providers must provide "read only" access to the OASIS to Commission staff and the staffs of State regulatory authorities, at no cost, after such staff members have complied with the requisite registration procedures.

[Order 889, 61 FR 21764, May 10, 1996, as amended by Order 605, 64 FR 34124, June 25, 1999; Order 638, 65 FR 17400, Mar. 31, 2000; Order 676, 71 FR 26212, May 4, 2006]

§ 37.6 Information to be posted on the OASIS.

- (a) The information posted on the OASIS must be in such detail and the OASIS must have such capabilities as to allow Transmission Customers to:
- (1) Make requests for transmission services offered by Transmission Providers, Resellers and other providers of ancillary services, request the designation of a network resource, and request the termination of the designation of a network resource;
- (2) View and download in standard formats, using standard protocols, information regarding the transmission system necessary to enable prudent business decision making;
- (3) Post, view, upload and download information regarding available products and desired services;

- (4) Clearly identify the degree to which transmission service requests or schedules were denied or interrupted;
- (5) Obtain access, in electronic format, to information to support available transmission capability calculations and historical transmission service requests and schedules for various audit purposes; and
- (6) Make file transfers and automated computer-to-computer file transfers and queries as defined by the Standards and Communications Protocols Document.
- (b) Posting transfer capability. The available transfer capability on the Transmission Provider's system (ATC) and the total transfer capability (TTC) of that system shall be calculated and posted for each Posted Path as set out in this section.
- (1) *Definitions*. For purposes of this section the terms listed below have the following meanings:
- (i) Posted path means any control area to control area interconnection; any path for which service is denied, curtailed or interrupted for more than 24 hours in the past 12 months; and any path for which a customer requests to have ATC or TTC posted. For this last category, the posting must continue for 180 days and thereafter until 180 days have elapsed from the most recent request for service over the requested path. For purposes of this definition, an hour includes any part of an hour during which service was denied, curtailed or interrupted.
- (ii) Constrained posted path means any posted path having an ATC less than or equal to 25 percent of TTC at any time during the preceding 168 hours or for which ATC has been calculated to be less than or equal to 25 percent of TTC for any period during the current hour or the next 168 hours.
- (iii) *Unconstrained posted path* means any posted path not determined to be a constrained posted path.
- (iv) The word *interconnection*, as used in the definition of "posted path", means all facilities connecting two adjacent systems or control areas.
- (v) Available transfer capability or ATC means the transfer capability remaining in the physical transmission network for further commercial activity over and above already committed

§ 37.6

uses, or such definition as contained in Commission-approved Reliability Standards.

(vi) Total transfer capability or TTC means the amount of electric power that can be moved or transferred reliably from one area to another area of the interconnected transmission systems by way of all transmission lines (or paths) between those areas under specified system conditions, or such definition as contained in Commissionapproved Reliability Standards.

(vii) Capacity Benefit Margin or CBM means the amount of TTC preserved by the Transmission Provider for load-serving entities, whose loads are located on that Transmission Provider's system, to enable access by the load-serving entities to generation from interconnected systems to meet generation reliability requirements, or such definition as contained in Commission-approved Reliability Standards.

(viii) Transmission Reliability Margin or TRM means the amount of TTC necessary to provide reasonable assurance that the interconnected transmission network will be secure, or such definition as contained in Commission-approved Reliability Standards.

(2) Calculation methods, availability of information, and requests. (i) Information used to calculate any posting of ATC and TTC must be dated and timestamped and all calculations shall be performed according to consistently applied methodologies referenced in the Transmission Provider's transmission tariff and shall be based on Commission-approved Reliability Standards as well as current industry practices, standards and criteria.

(ii) On request, the Responsible Party must make all data used to calculate ATC, TTC, CBM, and TRM for any constrained posted paths publicly available (including the limiting element(s) and the cause of the limit (e.g., thermal, voltage, stability), as well as load forecast assumptions) in electronic form within one week of the posting. The information is required to be provided only in the electronic format in which it was created, along with any necessary decoding instructions, at a cost limited to the cost of reproducing the material. This information is to be

retained for six months after the applicable posting period.

- (iii) System planning studies, facilities studies, and specific network impact studies performed for customers or the Transmission Provider's own network resources are to be made publicly available in electronic form on request and a list of such studies shall be posted on the OASIS. A study is required to be provided only in the electronic format in which it was created, along with any necessary decoding instructions, at a cost limited to the cost of reproducing the material. These studies are to be retained for five years.
- (3) Posting. The ATC, TTC, CBM, and TRM for all Posted Paths must be posted in megawatts by specific direction and in the manner prescribed in this subsection.
- (i) Constrained posted paths—(A) For firm ATC and TTC.
- (1) The posting shall show ATC, TTC, CBM, and TRM for a 30-day period. For this period postings shall be: by the hour, for the current hour and the 168 hours next following; and thereafter, by the day. If the Transmission Provider charges separately for on-peak and off-peak periods in its tariff, ATC, TTC, CBM, and TRM will be posted daily for each period.
- (2) Postings shall also be made by the month, showing for the current month and the 12 months next following.
- (3) If planning and specific requested transmission studies have been done, seasonal capability shall be posted for the year following the current year and for each year following to the end of the planning horizon but not to exceed 10 years.
- (B) For non-firm ATC and TTC. The posting shall show ATC, TTC, CBM and TRM for a 30-day period by the hour and days prescribed under paragraph (b)(3)(i)(A)(I) of this section and, if so requested, by the month and year as prescribed under paragraph (b)(3)(i)(A) (2) and (3) of this section. The posting of non-firm ATC and TTC shall show CBM as zero.
- (C) Updating posted information for constrained paths.

- (1) The capability posted under paragraphs (b)(3)(i)(A) and (B) of this section must be updated when transactions are reserved or service ends or whenever the estimate for the path changes by more than 10 percent.
- (2) All updating of hourly information shall be made on the hour.
- (3) When the monthly and yearly capability posted under paragraphs (b)(3)(i)(A) and (B) of this section are updated because of a change in TTC by more than 10 percent, the Transmission Provider shall post a brief, but specific, narrative explanation of the reason for the update. This narrative should include, the specific events which gave rise to the update (e.g., scheduling of planned outages and occurrence of forced transmission outages, de-ratings of transmission facilities, scheduling of planned generation outages and occurrence of forced generation outages, changes in load forecast, changes in new facilities' in-service dates, or other events or assumption changes) and new values for ATC on the path (as opposed to all points on the network).
- (4) When the monthly and yearly capability posted under paragraphs (b)(3)(i)(A) and (B) of this section remain unchanged at a value of zero for a period of six months, the Transmission Provider shall post a brief, but specific, narrative explanation of the reason for the unavailability of ATC.
 - (ii) Unconstrained posted paths.
- (A) Postings of firm and nonfirm ATC, TTC, CBM, and TRM shall be posted separately by the day, showing for the current day and the next six days following and thereafter, by the month for the 12 months next following. If the Transmission Provider charges separately for on-peak and offpeak periods in its tariff, ATC, TTC, CBM, and TRM will be posted separately for the current day and the next six days following for each period. These postings are to be updated whenever the ATC changes by more than 20 percent of the Path's TTC.
- (B) If planning and specific requested transmission studies have been done, seasonal capability shall be posted for the year following the current year and for each year following until the end of the planning horizon but not to exceed 10 years.

- (iii) Calculation of CBM.
- (A) The Transmission Provider must reevaluate its CBM needs at least every
- (B) The Transmission Provider must post its practices for reevaluating its CBM needs.
- (iv) Daily load. The Transmission Provider must post on a daily basis, its load forecast, including underlying assumptions, and actual daily peak load for the prior day.
- (c) Posting Transmission Service Products and Prices. (1) Transmission Providers must post prices and a summary of the terms and conditions associated with all transmission products offered to Transmission Customers.
- (2) Transmission Providers must provide a downloadable file of their complete tariffs in the same electronic format as the tariff that is filed with the Commission. Transmission Providers also must provide a link to all of the rules, standards and practices that relate to transmission services posted on the Transmission Providers' public Web sites.
- (3) Any offer of a discount for any transmission service made by the Transmission Provider must be announced to all potential customers solely by posting on the OASIS.
- (4) For any transaction for transmission service agreed to by the Transmission Provider and a customer, the Transmission Provider (at the time when ATC must be adjusted in response to the transaction), must post on the OASIS (and make available for download) information describing the transaction (including: price; quantity; points of receipt and delivery; length and type of service; identification of whether the transaction involves the Transmission Provider's wholesale merchant function or any affiliate; identification of what, if any, ancillary service transactions are associated with this transmission service transaction; and any other relevant terms and conditions) and shall keep such information posted on the OASIS for at least 30 days. A record of the transaction must be retained and kept available as part of the audit log required in § 37.7.
- (5) Customers choosing to use the OASIS to offer for resale transmission

§ 37.6

capacity they have purchased must post relevant information to the same OASIS as used by the Transmission Provider from whom the Reseller purchased the transmission capacity. This information must be posted on the same display page, using the same tables, as similar capability being sold by the Transmission Provider, and the information must be contained in the same downloadable files as the Transmission Provider's own available capability.

- (d) Posting Ancillary Service Offerings and Prices. (1) Any ancillary service required to be provided or offered under the pro forma tariff prescribed by part 35 of this chapter must be posted with the price of that service.
- (2) Any offer of a discount for any ancillary service made by the Transmission Provider must be announced to all potential customers solely by posting on the OASIS.
- (3) For any transaction for ancillary service agreed to by the Transmission Provider and a customer, the Transmission Provider (at the time when ATC must be adjusted in response to an associated transmission service transaction, if any), must post on the OASIS (and make available for download) information describing the transaction (including: date and time when the agreement was entered into; price; quantity; length and type of service; identification of whether the transaction involves the Transmission Provider's wholesale merchant function or any affiliate; identification of what, if any, transmission service transactions are associated with this ancillary service transaction; and any other relevant terms and conditions) and shall keep such information posted on the OASIS for at least 30 days. A record of the transaction must be retained and kept available as part of the audit log required in §37.7.
- (4) Any other interconnected operations service offered by the Transmission Provider may be posted, with the price for that service.
- (5) Any entity offering an ancillary service shall have the right to post the offering of that service on the OASIS if the service is one required to be offered by the Transmission Provider under the *pro forma* tariff prescribed by part

35 of this chapter. Any entity may also post any other interconnected operations service voluntarily offered by the Transmission Provider. Postings by customers and third parties must be on the same page, and in the same format, as postings of the Transmission Provider.

- (e) Posting specific transmission and ancillary service requests and responses— (1) General rules.(i) All requests for transmission and ancillary service offered by Transmission Providers under the pro forma tariff, including requests for discounts, and all requests to designate or terminate a network resource, must be made on the OASIS and posted prior to the Transmission Provider responding to the request, except as discussed in paragraphs (e)(1)(ii) and (iii) of this section. The Transmission Provider must post all requests for transmission service, for ancillary service, and for the designation or termination of a network resource comparably. Requests for transmission service, ancillary service, and to designate and terminate a network resource, as well as the responses to such requests, must be conducted in accordance with the Transmission Provider's tariff, the Federal Power Act, and Commission regulations.
- (ii) The requirement in paragraph (e)(1)(i) of this section, to post requests for transmission and ancillary service offered by Transmission Providers under the *pro forma* tariff, including requests for discounts, prior to the Transmission Provider responding to the request, does not apply to requests for next-hour service made during Phase I.
- (iii) In the event that a discount is being requested for ancillary services that are not in support of basic transmission service provided by the Transmission Provider, such request need not be posted on the OASIS.
- (iv) In processing a request for transmission or ancillary service, the Responsible Party shall post the same information as required in paragraphs (c)(4) and (d)(3) of this section, and the following information: the date and time when the request is made, its place in any queue, the status of that

request, and the result (accepted, denied, withdrawn). In processing a request to designate or terminate the designation of a network resource, the Responsible Party shall post the date and time when the request is made.

- (v) For any request to designate or terminate a network resource, the Transmission Provider (at the time when the request is received), must post on the OASIS (and make available for download) information describing the request (including: name of requestor, identification of the resource, effective time for the designation or termination, identification of whether the transaction involves the Transmission Provider's wholesale merchant function or any affiliate; and any other relevant terms and conditions) and shall keep such information posted on the OASIS for at least 30 days. A record of the transaction must be retained and kept available as part of the audit log required in §37.7.
- (vi) The Transmission Provider shall post a list of its current designated network resources and all network customers' current designated network resources on OASIS. The list of network resources should include the name of the resource, its geographic and electrical location, its total installed capacity, and the amount of capacity to be designated as a network resource.
- (2) Posting when a request for transmission service is denied. (i) When a request for service is denied, the Responsible Party must provide the reason for that denial as part of any response to the request.
- (ii) Information to support the reason for the denial, including the operating status of relevant facilities, must be maintained for five years and provided, upon request, to the potential Transmission Customer and the Commission's Staff.
- (iii) Any offer to adjust operation of the Transmission Provider's System to accommodate the denied request must be posted and made available to all Transmission Customers at the same time
- (3) Posting when a transaction is curtailed or interrupted. (i) When any transaction is curtailed or interrupted, the Transmission Provider must post notice of the curtailment or interruption

on the OASIS, and the Transmission Provider must state on the OASIS the reason why the transaction could not be continued or completed.

- (ii) Information to support any such curtailment or interruption, including the operating status of the facilities involved in the constraint or interruption, must be maintained and made available upon request, to the curtailed or interrupted customer, the Commission's Staff, and any other person who requests it, for five years.
- (iii) Any offer to adjust the operation of the Transmission Provider's system to restore a curtailed or interrupted transaction must be posted and made available to all curtailed and interrupted Transmission Customers at the same time.
- (f) Posting Transmission Service Schedules Information. Information on transmission service schedules must be recorded by the entity scheduling the transmission service and must be available on the OASIS for download. Transmission service schedules must be posted no later than seven calendar days from the start of the transmission service.
- (g) Posting Other Transmission-Related Communications. (1) The posting of other communications related to transmission services must be provided for by the Responsible Party. These communications may include "want ads" and "other communications" (such as using the OASIS as a Transmission-related conference space or to provide transmission-related messaging services between OASIS users). Such postings carry no obligation to respond on the part of any market participant.
- (2) The Responsible Party is responsible for posting other transmission-related communications in conformance with the instructions provided by the third party on whose behalf the communication is posted. It is the responsibility of the third party requesting such a posting to ensure the accuracy of the information to be posted.
- (3) Notices of transfers of personnel shall be posted as described in §358.4(c). The posting requirements are the same as those provided in §37.7 for audit data postings.
- (4) Logs detailing the circumstances and manner in which a Transmission

§ 37.6

Provider or Responsible Party exercised its discretion under any terms of the tariff shall be posted as described in §358.5(c)(4). The posting requirements are the same as those provided in §37.7 for audit data postings.

- (h) Posting information summarizing the time to complete transmission service request studies. (1) For each calendar quarter, the Responsible Party must post the set of measures detailed in paragraph (h)(1)(i) through paragraph (h)(1)(vi) of this section related to the Responsible Party's processing transmission service request system impact studies and facilities studies. The Responsible Party must calculate and post the measures in paragraph (h)(1)(i) through paragraph (h)(1)(vi) of this section for requests for short-term firm point-to-point transmission service, requests for long-term firm pointto-point transmission service, and requests to designate a new network resource or network load. When calculating the measures in paragraph (h)(1)(i) through paragraph (h)(1)(iv) of this section, the Responsible Party may aggregate requests for short-term firm point-to-point service and requests for long-term firm point-topoint service, but must calculate and post measures separately for transmission service requests from Affiliates and transmission service requests from Transmission Customers who are not Affiliates. The Responsible Party is required to include in the calculations of the measures in paragraph (h)(1)(i) through paragraph (h)(1)(vi) of this section all studies the Responsible Party conducts of transmission service requests on another Transmission Provider's OASIS.
- (i) Process time from initial service request to offer of system impact study agreement.
- (A) Number of new system impact study agreements delivered during the reporting quarter to entities that request transmission service,
- (B) Number of new system impact study agreements delivered during the reporting quarter to entities that request transmission service more than thirty (30) days after the Responsible Party received the request for transmission service.

- (C) Mean time (in days), for all requests acted on by the Responsible Party during the reporting quarter, from the date when the Responsible Party received the request for transmission service to when the Responsible Party changed the transmission service request status to indicate that the Responsible Party could offer transmission service or needed to perform a system impact study,
- (D) Mean time (in days), for all system impact study agreements delivered by the Responsible Party during the reporting quarter, from the date when the Responsible Party received the request for transmission service to the date when the Responsible Party delivered a system impact study agreement, and
- (E) Number of new system impact study agreements executed during the reporting quarter.
- (ii) System impact study processing time.
- (A) Number of system impact studies completed by the Responsible Party during the reporting quarter,
- (B) Number of system impact studies completed by the Responsible Party during the reporting quarter more than 60 days after the Responsible Party received an executed system impact study agreement,
- (C) For all system impact studies completed more than 60 days after receipt of an executed system impact study agreement, average number of days study was delayed due to transmission customer's actions (e.g., delays in providing needed data),
- (D) Mean time (in days), for all system impact studies completed by the Responsible Party during the reporting quarter, from the date when the Responsible Party received the executed system impact study agreement to the date when the Responsible Party provided the system impact study to the entity who executed the system impact study agreement, and
- (E) Mean cost of system impact studies completed by the Responsible Party during the reporting quarter.
- (iii) Transmission service requests withdrawn from the system impact study queue.

- (A) Number of transmission service requests withdrawn from the Responsible Party's system impact study queue during the reporting quarter,
- (B) Number of transmission service requests withdrawn from the Responsible Party's system impact study queue during the reporting quarter more than 60 days after the Responsible Party received the executed system impact study agreement, and
- (C) Mean time (in days), for all transmission service requests withdrawn from the Responsible Party's system impact study queue during the reporting quarter, from the date the Responsible Party received the executed system impact study agreement to date when request was withdrawn from the Responsible Party's system impact study queue.
- (iv) Process time from completed system impact study to offer of facilities study.
- (A) Number of new facilities study agreements delivered during the reporting quarter to entities that request transmission service,
- (B) Number of new facilities study agreements delivered during the reporting quarter to entities that request transmission service more than thirty (30) days after the Responsible Party completed the system impact study,
- (C) Mean time (in days), for all facilities study agreements delivered by the Responsible Party during the reporting quarter, from the date when the Responsible Party completed the system impact study to the date when the Responsible Party delivered a facilities study agreement, and
- (D) Number of new facilities study agreements executed during the reporting quarter.
 - (v) Facilities study processing time.
- (A) Number of facilities studies completed by the Responsible Party during the reporting quarter,
- (B) Number of facilities studies completed by the Responsible Party during the reporting quarter more than 60 days after the Responsible Party received an executed facilities study agreement,
- (C) For all facilities studies completed more than 60 days after receipt of an executed facilities study agreement, average number of days study was delayed due to transmission cus-

- tomer's actions (e.g., delays in providing needed data).
- (D) Mean time (in days), for all facilities studies completed by the Responsible Party during the reporting quarter, from the date when the Responsible Party received the executed facilities study agreement to the date when the Responsible Party provided the facilities study to the entity who executed the facilities study agreement.
- (E) Mean cost of facilities studies completed by the Responsible Party during the reporting quarter, and
- (F) Mean cost of upgrades recommended in facilities studies completed during the reporting quarter.
- (vi) Service requests withdrawn from facilities study queue.
- (A) Number of transmission service requests withdrawn from the Responsible Party's facilities study queue during the reporting quarter,
- (B) Number of transmission service requests withdrawn from the Responsible Party's facilities study queue during the reporting quarter more than 60 days after the Responsible Party received the executed facilities study agreement, and
- (C) Mean time (in days), for all transmission service requests withdrawn from the Responsible Party's facilities study queue during the reporting quarter, from the date the Responsible Party received the executed facilities study agreement to date when request was withdrawn from the Responsible Party's facilities study queue.
- (2) The Responsible Party is required to post the measures in paragraph (h)(1)(i) through paragraph (h)(1)(vi) of this section for each calendar quarter within 15 days of the end of the calendar quarter. The Responsible Party will keep the quarterly measures posted on OASIS for three calendar years.
- (3) The Responsible Party will be required to post on OASIS the measures in paragraph (h)(3)(i) through paragraph (h)(3)(iv) of this section in the event the Responsible Party, for two consecutive calendar quarters, completes more than twenty (20) percent of the studies associated with requests for transmission service from entities that are not Affiliates of the Responsible Party more than sixty (60) days after

§ 37.7

the Responsible Party delivers the appropriate study agreement. The Responsible Party will have to post the measures in paragraph (h)(3)(i) through paragraph (h)(3)(iv) of this section until it processes at least ninety (90) percent of all studies within 60 days after it has received the appropriate executed study agreement. For the purposes of calculating the percent of studies completed more than sixty (60) days after the Responsible Party delivers the appropriate study agreement, the Responsible Party should aggregate all system impact studies and facilities studies that it completes during the reporting quarter.

- (i) Mean, across all system impact studies the Responsible Party completes during the reporting quarter, of the employee-hours expended per system impact study the Responsible Party completes during reporting period:
- (ii) Mean, across all facilities studies the Responsible Party completes during the reporting quarter, of the employee-hours expended per facilities study the Responsible Party completes during reporting period;
- (iii) The number of employees the Responsible Party has assigned to process system impact studies;
- (iv) The number of employees the Responsible Party has assigned to process facilities studies.
- (4) The Responsible Party is required to post the measures in paragraph (h)(3)(i) through paragraph (h)(3)(iv) of this section for each calendar quarter within 15 days of the end of the calendar quarter. The Responsible Party will keep the quarterly measures posted on OASIS for five calendar years.
- (i) Posting data related to grants and denials of service. The Responsible Party is required to post data each month listing, by path or flowgate, the number of transmission service requests that have been accepted and the number of transmission service requests that have been denied during the prior month. This posting must distinguish between the length of the service request (e.g., short-term or long-term requests) and between the type of service requested (e.g., firm point-to-point, non-firm point-to-point

or network service). The posted data must show:

- (1) The number of non-Affiliate requests for transmission service that have been rejected,
- (2) The total number of non-Affiliate requests for transmission service that have been made.
- (3) The number of Affiliate requests for transmission service, including requests by the transmission provider's merchant function to designate a network resource or to procure secondary network service, that have been rejected, and
- (4) The total number of Affiliate requests for transmission service, including requests by the transmission provider's merchant function to designate, or terminate the designation of, a network resource or to procure secondary network service, that have been made.
 - (j) Posting redispatch data.
- (1) The Transmission Provider must allow the posting on OASIS of any third party offer to relieve a specified congested transmission facility.
- (2) The Transmission Provider must post on OASIS (i) its monthly average cost of planning and reliability redispatch, for which it invoices customers, at each internal transmission facility or interface over which it provides redispatch service and (ii) a high and low redispatch cost for the month for each of these same transmission facilities. The transmission provider must post this data on OASIS as soon as practical after the end of each month, but no later than when it sends invoices to transmission customers for redispatch-related services.

[Order 889, 61 FR 21764, May 10, 1996, as amended by Order 889–A, 62 FR 12503, Mar. 14, 1997; Order 605, 64 FR 34124, June 25, 1999; Order 2004, 68 FR 69157, Dec. 11, 2003; Order 890, 72 FR 12493, Mar. 15, 2007; Order 890–A, 73 FR 3111, Jan. 16, 2008]

§ 37.7 Auditing Transmission Service Information.

- (a) All OASIS database transactions, except other transmission-related communications provided for under §37.6(g)(2), must be stored, dated, and time stamped.
- (b) Audit data must remain available for download on the OASIS for 90 days,

except ATC/TTC postings that must remain available for download on the OASIS for 20 days. The audit data are to be retained and made available upon request for download for five years from the date when they are first posted in the same electronic form as used when they originally were posted on the OASIS.

[Order 889, 61 FR 21764, May 10, 1996, as amended by Order 889-A, 62 FR 12504, Mar. 14, 1997; Order 890, 72 FR 12496, Mar. 15, 2007]

§ 37.8 Obligations of OASIS users.

Each OASIS user must notify the Responsible Party one month in advance of initiating a significant amount of automated queries. The OASIS user must also notify the Responsible Party one month in advance of expected significant increases in the volume of automated queries.

[Order 605, 64 FR 34124, June 25, 1999]

PART 38—BUSINESS PRACTICE STANDARDS AND COMMUNICATION PROTOCOLS FOR PUBLIC UTILITIES

Sec.

38.1 Applicability.

38.2 Incorporation by reference of North American Energy Standards Board Wholesale Electric Quadrant standards.

AUTHORITY: 16 U.S.C. 791–825r, 2601–2645; 31 U.S.C. 9701; 42 U.S.C. 7101–7352.

Source: Order 676, 71 FR 26212, May 4, 2006, unless otherwise noted.

§38.1 Applicability.

This part applies to any public utility that owns, operates, or controls facilities used for the transmission of electric energy in interstate commerce or for the sale of electric energy at wholesale in interstate commerce and to any non-public utility that seeks voluntary compliance with jurisdictional transmission tariff reciprocity conditions.

[Order 698, 72 FR 38767, July 16, 2007]

§ 38.2 Incorporation by reference of North American Energy Standards Board Wholesale Electric Quadrant standards.

(a) All entities to which §38.1 is applicable must comply with the fol-

lowing business practice and electronic communication standards promulgated by the North American Energy Standards Board Wholesale Electric Quadrant, which are incorporated herein by reference:

- (1) Open Access Same-Time Information Systems (OASIS), Version 1.5 (WEQ-001, Version 002.1, March 11, 2009, with minor corrections applied May 29, 2009 and September 8, 2009, with the exception of Standards 001-0.1, 001-0.9 through 001-0.13, 001-1.0, 001-9.7, 001-14.1.3, and 001-15.1.2);
- (2) Open Access Same-Time Information Systems (OASIS) Standards & Communication Protocols, Version 1.5 (WEQ-002, Version 002.1, March 11, 2009, with minor corrections applied May 29, 2009 and September 8, 2009);
- (3) Open Access Same-Time Information Systems (OASIS) Data Dictionary, Version 1.5 (WEQ-003, Version 002.1, March 11, 2009, with minor corrections applied May 29, 2009 and September 8, 2009):
- (4) Coordinate Interchange (WEQ-004, Version 002.1, March 11, 2009, with minor corrections applied May 29, 2009 and September 8, 2009);
- (5) Area Control Error (ACE) Equation Special Cases (WEQ-005, Version 002.1, March 11, 2009, with minor corrections applied May 29, 2009 and September 8, 2009);
- (6) Manual Time Error Correction (WEQ-006, Version 001, Oct. 31, 2007, with minor corrections applied on Nov. 16, 2007):
- (7) Inadvertent Interchange Payback (WEQ-007, Version 002.1, March 11, 2009, with minor corrections applied May 29, 2009 and September 8, 2009);
- (8) Transmission Loading Relief—Eastern Interconnection (WEQ-008, Version 002.1, March 11, 2009, with minor corrections applied May 29, 2009 and September 8, 2009);
- (9) Gas/Electric Coordination (WEQ-011, Version 002.1, March 11, 2009, with minor corrections applied May 29, 2009 and September 8, 2009);
- (10) Public Key Infrastructure (PKI) (WEQ-012, Version 002.1, March 11, 2009, with minor corrections applied on May 29, 2009 and September 8, 2009);
- (11) Open Access Same-Time Information Systems (OASIS) Implementation Guide, Version 1.5 (WEQ-013, Version

Pt. 39

002.1, March 11, 2009, with minor corrections applied on May 29, 2009 and September 8, 2009); and

(12) Business Practices for Measurement and Verification of Wholesale Electricity Demand Response (WEQ-015, 2008 Annual Plan Item 5(a), March 16, 2009).

(b) This incorporation by reference was approved by the Director of the Federal Register in accordance with 5 U.S.C. 552(a) and 1 CFR part 51. Copies of these standards may be obtained from the North American Energy Standards Board, 801 Travis Street, Suite 1675, Houston, TX 77002, Tel: (713) 356-0060. NAESB's Web site is at http:// www.naesb.org/. Copies may be inspected at the Federal Energy Regulatory Commission, Public Reference and Files Maintenance Branch, 888 First Street, NE., Washington, DC Tel: (202) 502-8371. http:// 20426. www.ferc.gov, or at the National Archives and Records Administration (NARA). For information on the availability of this material at NARA, call 202-741-6030, or g_0 to: http:// www.archives.gov/federal register/ code of federal regulations/ ibr locations.html.

[Order 676, 71 FR 26212, May 4, 2006 as amended by Order 676-B, 72 FR 21099, Apr. 30, 2007; Order 698, 72 FR 38767, July 16, 2007; Order 676-C, 73 FR 43860, July 29, 2008; Order 676-E, 74 FR 63306, Dec. 3, 2009; Order 676-F, 75 FR 20908, Apr. 22, 2010]

PART 39—RULES CONCERNING CERTIFICATION OF THE ELECTRIC RELIABILITY ORGANIZATION; AND PROCEDURES FOR THE ESTABLISHMENT, APPROVAL, AND ENFORCEMENT OF ELECTRIC RELIABILITY STANDARDS

Sec.

- 39.1 Definitions.
- 39.2 Jurisdiction and applicability.
- 39.3 Electric Reliability Organization certification.
- 39.4 Funding of the Electric Reliability Organization.
- 39.5 Reliability Standards.
- 39.6 Conflict of a Reliability Standard with a Commission order.
- 39.7 Enforcement of Reliability Standards.
- 39.8 Delegation to a Regional Entity.
- 39.9 Enforcement of Commission rules and orders.

- 39.10 Changes to an Electric Reliability Organization Rule or Regional Entity Rule.
- 39.11 Reliability reports.39.12 Review of state action.
- 39.13 Regional Advisory Bodies.

AUTHORITY: 16 U.S.C. 8240.

SOURCE: Order 672, 71 FR 8736, Feb. 17, 2006, unless otherwise noted.

§ 39.1 Definitions.

As used in this part:

Bulk-Power System means facilities and control systems necessary for operating an interconnected electric energy transmission network (or any portion thereof), and electric energy from generating facilities needed to maintain transmission system reliability. The term does not include facilities used in the local distribution of electric energy.

Cross-Border Regional Entity means a Regional Entity that encompasses a part of the United States and a part of Canada or Mexico.

Cybersecurity Incident means a malicious act or suspicious event that disrupts, or was an attempt to disrupt, the operation of those programmable electronic devices and communications networks including hardware, software and data that are essential to the Reliable Operation of the Bulk-Power System.

Electric Reliability Organization or "ERO" means the organization certified by the Commission under §39.3 the purpose of which is to establish and enforce Reliability Standards for the Bulk-Power System, subject to Commission review.

Electric Reliability Organization Rule means, for purposes of this part, the bylaws, a rule of procedure or other organizational rule or protocol of the Electric Reliability Organization.

Interconnection means a geographic area in which the operation of Bulk-Power System components is synchronized such that the failure of one or more of such components may adversely affect the ability of the operators of other components within the system to maintain Reliable Operation of the facilities within their control.

Regional Advisory Body means an entity established upon petition to the Commission pursuant to section 215(j)

of the Federal Power Act that is organized to advise the Electric Reliability Organization, a Regional Entity, or the Commission regarding certain matters in accordance with \$39.13.

Regional Entity means an entity having enforcement authority pursuant to §39.8.

Regional Entity Rule means, for purposes of this part, the bylaws, a rule of procedure or other organizational rule or protocol of a Regional Entity.

Reliability Standard means a requirement approved by the Commission under section 215 of the Federal Power Act, to provide for Reliable Operation of the Bulk-Power System. The term includes requirements for the operation of existing Bulk-Power System facilities, including cybersecurity protection, and the design of planned additions or modifications to such facilities to the extent necessary to provide for Reliable Operation of the Bulk-Power System, but the term does not include any requirement to enlarge such facilities or to construct new transmission capacity or generation capacity.

Reliable Operation means operating the elements of the Bulk-Power System within equipment and electric system thermal, voltage, and stability limits so that instability, uncontrolled separation, or cascading failures of such system will not occur as a result of a sudden disturbance, including a Cybersecurity Incident, or unanticipated failure of system elements.

Transmission Organization means a regional transmission organization, independent system operator, independent transmission provider, or other transmission organization finally approved by the Commission for the operation of transmission facilities.

$\S 39.2$ Jurisdiction and applicability.

(a) Within the United States (other than Alaska and Hawaii), the Electric Reliability Organization, any Regional Entities, and all users, owners and operators of the Bulk-Power System, including but not limited to entities described in section 201(f) of the Federal Power Act, shall be subject to the jurisdiction of the Commission for the purposes of approving Reliability Standards established under section 215 of the Federal Power Act and enforcing

compliance with section 215 of the Federal Power Act.

- (b) All entities subject to the Commission's reliability jurisdiction under paragraph (a) of this section shall comply with applicable Reliability Standards, the Commission's regulations, and applicable Electric Reliability Organization and Regional Entity Rules made effective under this part.
- (c) Each user, owner and operator of the Bulk-Power System within the United States (other than Alaska and Hawaii) shall register with the Electric Reliability Organization and the Regional Entity for each region within which it uses, owns or operates Bulk-Power System facilities, in such manner as prescribed in the Rules of the Electric Reliability Organization and each applicable Regional Entity.
- (d) Each user, owner or operator of the Bulk-Power System within the United States (other than Alaska and Hawaii) shall provide the Commission, the Electric Reliability Organization and the applicable Regional Entity such information as is necessary to implement section 215 of the Federal Power Act as determined by the Commission and set out in the Rules of the Electric Reliability Organization and each applicable Regional Entity. The Electric Reliability Organization and each Regional Entity shall provide the Commission such information as is necessary to implement section 215 of the Federal Power Act.

§ 39.3 Electric Reliability Organization certification.

- (a) Any person may submit an application to the Commission for certification as the Electric Reliability Organization no later than April 4, 2006. Such application shall comply with the requirements for filings in proceedings before the Commission in part 385 of this chapter.
- (b) After notice and an opportunity for public comment, the Commission may certify one such applicant as an Electric Reliability Organization, if the Commission determines such applicant:
- (1) Has the ability to develop and enforce, subject to §39.7, Reliability Standards that provide for an adequate

§ 39.4

level of reliability of the Bulk-Power System, and

- (2) Has established rules that:
- (i) Assure its independence of users, owners and operators of the Bulk-Power System while assuring fair stakeholder representation in the selection of its directors and balanced decisionmaking in any Electric Reliability Organization committee or subordinate organizational structure;
- (ii) Allocate equitably reasonable dues, fees and charges among end users for all activities under this part;
- (iii) Provide fair and impartial procedures for enforcement of Reliability Standards through the imposition of penalties in accordance with §39.7, including limitations on activities, functions, operations, or other appropriate sanctions or penalties;
- (iv) Provide reasonable notice and opportunity for public comment, due process, openness, and balance of interests in developing Reliability Standards, and otherwise exercising its duties; and
- (v) Provide appropriate steps, after certification by the Commission as the Electric Reliability Organization, to gain recognition in Canada and Mexico.
- (c) The Electric Reliability Organization shall submit an assessment of its performance three years from the date of certification by the Commission, and every five years thereafter. After receipt of the assessment, the Commission will establish a proceeding with opportunity for public comment in which it will review the performance of the Electric Reliability Organization.
- (1) The Electric Reliability Organization's assessment of its performance shall include:
- (i) An explanation of how the Electric Reliability Organization satisfies the requirements of §39.3(b);
- (ii) Recommendations by Regional Entities, users, owners, and operators of the Bulk-Power System, and other interested parties for improvement of the Electric Reliability Organization's operations, activities, oversight and procedures, and the Electric Reliability Organization's response to such recommendations; and
- (iii) The Electric Reliability Organization's evaluation of the effectiveness of each Regional Entity, recommenda-

- tions by the Electric Reliability Organization, users, owners, and operators of the Bulk-Power System, and other interested parties for improvement of the Regional Entity's performance of delegated functions, and the Regional Entity's response to such evaluation and recommendations.
- (2) The Commission will issue an order finding that the Electric Reliability Organization meets the statutory and regulatory criteria or directing the Electric Reliability Organization or a Regional Entity to come into compliance with or improve its compliance with the requirements of this part. If the ERO fails to comply adequately with the Commission order, the Commission may institute a proceeding to enforce its order, including, if necessary and appropriate, a proceeding to consider decertification of the ERO consistent with §39.9. The Commission will issue an order finding that each Regional Entity meets the statutory and regulatory criteria or directing the Regional Entity to come into compliance with or improve its compliance with the requirements of this part. If a Regional Entity fails to comply adequately with the Commission order, the Commission may institute a proceeding to enforce its order, including, if necessary and appropriate, a proceeding to consider rescission of its approval of the Regional Entity's delegation agreement.

§ 39.4 Funding of the Electric Reliability Organization.

- (a) Any person who submits an application for certification as the Electric Reliability Organization shall include in its application a formula or method for the allocation and assessment of Electric Reliability Organization dues, fees and charges. The certified Electric Reliability Organization may subsequently file with the Commission a request to modify the formula or method.
- (b) The Electric Reliability Organization shall file with the Commission its proposed entire annual budget for statutory and any non-statutory activities, including the entire annual budget for statutory and any non-statutory activities of each Regional Entity, with supporting materials, including the

ERO's and each Regional Entity's complete business plan and organization chart, explaining the proposed collection of all dues, fees and charges and the proposed expenditure of funds collected in sufficient detail to justify the requested funding collection and budget expenditures 130 days in advance of the beginning of each Electric Reliability Organization fiscal year. The annual Electric Reliability Organization budget shall include line item budgets for the activities of each Regional Entity that are delegated or assigned to each Regional Entity pursuant to §39.8.

- (c) The Commission, after public notice and opportunity for hearing, will issue an order either accepting, rejecting, remanding or modifying the proposed Electric Reliability Organization budget and business plan no later than sixty (60) days in advance of the beginning of the Electric Reliability Organization's fiscal year.
- (d) On a demonstration of unforeseen and extraordinary circumstances requiring additional funds prior to the next Electric Reliability Organization fiscal year, the Electric Reliability Organization may file with the Commission for authorization to collect a special assessment. Such filing shall include supporting materials explaining the proposed collection in sufficient detail to justify the requested funding, including any departure from the approved funding formula or method. After notice and an opportunity for hearing, the Commission will approve, disapprove, remand or modify such reauest.
- (e) All entities within the Commission's jurisdiction as set forth in section 215(b) of the Federal Power Act shall pay any Electric Reliability Organization assessment of dues, fees and charges as approved by the Commission, in a timely manner reasonably as designated by the Electric Reliability Organization.
- (f) Any person who submits an application for certification as the Electric Reliability Organization may include in the application a plan for a transitional funding mechanism that would allow such person, if certified as the Electric Reliability Organization, to continue existing operations without

interruption as it transitions from one method of funding to another. Any proposed transitional funding plan should terminate no later than eighteen (18) months from the date of Electric Reliability Organization certification.

(g) The Electric Reliability Organization or a Regional Entity may not engage in any activity or receive revenues from any person that, in the judgment of the Commission represents a significant distraction from, or a conflict of interest with, its responsibilities under this part.

§39.5 Reliability Standards.

- (a) The Electric Reliability Organization shall file each Reliability Standard or modification to a Reliability Standard that it proposes to be made effective under this part with the Commission. The filing shall include a concise statement of the basis and purpose of the proposed Reliability Standard, either a summary of the Reliability Standard development proceedings conducted by the Electric Reliability Organization or a summary of the Reliability Standard development proceedings conducted by a Regional Entity together with a summary of the Reliability Standard review proceedings of the Electric Reliability Organization, and a demonstration that the proposed Reliability Standard is just, reasonable, not unduly discriminatory or preferential, and in the public interest.
- (b) The Electric Reliability Organization shall rebuttably presume that a proposal for a Reliability Standard or a modification to a Reliability Standard to be applicable on an Interconnection-wide basis is just, reasonable, not unduly discriminatory or preferential, and in the public interest, if such proposal is from a Regional Entity organized on an Interconnection-wide basis.
- (c) The Commission may approve by rule or order a proposed Reliability Standard or a proposed modification to a Reliability Standard if, after notice and opportunity for public hearing, it determines that the proposed Reliability Standard is just, reasonable, not unduly discriminatory or preferential, and in the public interest.
- (1) The Commission will give due weight to the technical expertise of the Electric Reliability Organization with

§ 39.6

respect to the content of a proposed Reliability Standard or a proposed modification to a Reliability Standard,

- (2) The Commission will give due weight to the technical expertise of a Regional Entity organized on an Interconnection-wide basis with respect to a proposed Reliability Standard or a proposed modification to a Reliability Standard to be applicable within that Interconnection, and
- (3) The Commission will not defer to the Electric Reliability Organization or a Regional Entity with respect to the effect of a proposed Reliability Standard or a proposed modification to a Reliability Standard on competition.
- (d) An approved Reliability Standard or modification to a Reliability Standard shall take effect as approved by the Commission.
- (e) The Commission will remand to the Electric Reliability Organization for further consideration a proposed Reliability Standard or modification to a Reliability Standard that the Commission disapproves in whole or in part.
- (f) The Commission may, upon its own motion or a complaint, order the Electric Reliability Organization to submit a proposed Reliability Standard or modification to a Reliability Standard that addresses a specific matter if the Commission considers such a new or modified Reliability Standard appropriate to carry out section 215 of the Federal Power Act.
- (g) The Commission, when remanding a Reliability Standard to the Electric Reliability Organization or ordering the Electric Reliability Organization to submit to the Commission a proposed Reliability Standard or proposed modification to a Reliability Standard that addresses a specific matter may order a deadline by which the Electric Reliability Organization must submit a proposed or modified Reliability Standard.

§ 39.6 Conflict of a Reliability Standard with a Commission Order.

(a) If a user, owner or operator of the transmission facilities of a Transmission Organization determines that a Reliability Standard may conflict with a function, rule, order, tariff, rate schedule, or agreement accepted, ap-

- proved, or ordered by the Commission with respect to such Transmission Organization, the Transmission Organization shall expeditiously notify the Commission, the Electric Reliability Organization and the relevant Regional Entity of the possible conflict.
- (b) After notice and opportunity for hearing, within sixty (60) days of the date that a notice was filed under paragraph (a) of this section, unless the Commission orders otherwise, the Commission will issue an order determining whether a conflict exists and, if so, resolve the conflict by directing:
- (1) The Transmission Organization to file a modification of the conflicting function, rule, order, tariff, rate schedule, or agreement pursuant to section 206 of the Federal Power Act, as appropriate, or
- (2) The Electric Reliability Organization to propose a modification to the conflicting Reliability Standard pursuant to §39.5 of the Commission's regulations.
- (c) The Transmission Organization shall continue to comply with the function, rule, order, tariff, rate schedule, or agreement accepted, approved, or ordered by the Commission until the Commission finds that a conflict exists, the Commission orders a change to such provision pursuant to section 206 of the Federal Power Act, and the ordered change becomes effective.

[Order 672, 71 FR 8736, Feb. 17, 2006, as amended at 71 FR 11505, Mar. 8, 2006; Order 672–A, 71 FR 19823, Apr. 18, 2006]

§ 39.7 Enforcement of Reliability Standards.

- (a) The Electric Reliability Organization and each Regional Entity shall have an audit program that provides for rigorous audits of compliance with Reliability Standards by users, owners and operators of the Bulk-Power System.
- (b) The Electric Reliability Organization and each Regional Entity shall have procedures to report promptly to the Commission any self-reported violation or investigation of a violation or an alleged violation of a Reliability Standard and its eventual disposition.

- (1) Any person that submits an application to the Commission for certification as an Electric Reliability Organization shall include in such application a proposal for the prompt reporting to the Commission of any self-reported violation or investigation of a violation or an alleged violation of a Reliability Standard and its eventual disposition.
- (2) Any agreement for the delegation of enforcement authority to a Regional Entity shall include a provision for the prompt reporting through the Electric Reliability Organization to the Commission of any self-reported violation or investigation of a violation or an alleged violation of a Reliability Standard and its eventual disposition.
- (3) Each report of a violation or alleged violation by a user, owner or operator of the Bulk-Power System shall include the user's, owner's or operator's name, which Reliability Standard or Reliability Standards were violated or allegedly violated, when the violation or alleged violation occurred, and the name of a person knowledgeable about the violation or alleged violation to serve as a point of contact with the Commission.
- (4) Each violation or alleged violation shall be treated as nonpublic until the matter is filed with the Commission as a notice of penalty or resolved by an admission that the user, owner or operator of the Bulk-Power System violated a Reliability Standard or by a settlement or other negotiated disposition. The disposition of each violation or alleged violation that relates to a Cybersecurity Incident or that would jeopardize the security of the Bulk-Power System if publicly disclosed shall be nonpublic unless the Commission directs otherwise.
- (5) The Electric Reliability Organization, and each Regional Entity through the ERO, shall file such periodic summary reports as the Commission shall from time to time direct on violations of Reliability Standards and summary analyses of such violations.
- (c) The Electric Reliability Organization, or a Regional Entity, may impose, subject to section 215(e) of the Federal Power Act, a penalty on a user, owner or operator of the Bulk-Power System for a violation of a Reliability

- Standard approved by the Commission if, after notice and opportunity for hearing:
- (1) The Electric Reliability Organization or the Regional Entity finds that the user, owner or operator has violated a Reliability Standard approved by the Commission; and
- (2) The Electric Reliability Organization files a notice of penalty and the record of its or a Regional Entity's proceeding with the Commission. Simultaneously with the filing of a notice of penalty with the Commission, the Electric Reliability Organization shall serve a copy of the notice of penalty on the entity that is the subject of the penalty.
- (d) A notice of penalty by the Electric Reliability Organization shall consist of:
- (1) The name of the entity on whom the penalty is imposed;
- (2) Identification of each Reliability Standard violated;
- (3) A statement setting forth findings of fact with respect to the act or practice resulting in the violation of each Reliability Standard;
- (4) A statement describing any penalty imposed;
 - (5) The record of the proceeding;
- (6) Other matters the Electric Reliability Organization or the Regional Entity, as appropriate, may find relevant.
- (e) A penalty imposed under this section may take effect not earlier than the thirty-first (31st) day after the Electric Reliability Organization files with the Commission the notice of penalty and the record of the proceedings.
- (1) Such penalty will be subject to review by the Commission, on its own motion or upon application by the user, owner or operator of the Bulk-Power System that is the subject of the penalty filed within thirty (30) days after the date such notice is filed with Commission. In the absence of the filing of an application for review or motion or other action by the Commission, the penalty shall be affirmed by operation of law upon the expiration of the thirty (30)-day period for filing of an application for review.
- (2) An applicant filing an application for review shall comply with the requirements for filings in proceedings

§ 39.7

before the Commission. An application shall contain a complete and detailed explanation of why the applicant believes that the Electric Reliability Organization or Regional Entity erred in determining that the applicant violated a Reliability Standard, or in determining the appropriate form or amount of the penalty. The applicant may support its explanation by providing information that is not included in the record submitted by the Electric Reliability Organization.

- (3) Application to the Commission for review, or the initiation of review by the Commission on its own motion, shall not operate as a stay of such penalty unless the Commission otherwise orders upon its own motion or upon application by the user, owner or operator that is the subject of such penalty.
- (4) Any answer, intervention or comment to an application for review of a penalty imposed under this part must be filed within twenty (20) days after the application is filed, unless otherwise ordered by the Commission.
- (5) In any proceeding to review a penalty imposed under this part, the Commission, after public notice and opportunity for hearing (which hearing may consist solely of the record before the Electric Reliability Organization or Regional Entity and the opportunity for the presentation of supporting reasons to affirm, modify, or set aside the penalty), will by order affirm, set aside, or modify the penalty or may remand the determination of a violation or the form or amount of the penalty to the Electric Reliability Organization for further consideration. The Commission may establish a hearing before an administrative law judge or initiate such further procedures as it determines to be appropriate, before issuing such an order. In the case of a remand to the Electric Reliability Organization, the Electric Reliability Organization may remand the matter to a Regional Entity for further consideration and resubmittal through the Electric Reliability Organization to the Commission.
- (6) The Commission will take action on an application for review of a penalty within sixty (60) days of the date the application is filed unless the Com-

mission determines on a case-by-case basis that an alternative expedited procedure is appropriate.

- (7) A proceeding for Commission review of a penalty for violation of a Reliability Standard will be public unless the Commission determines that a nonpublic proceeding is necessary and lawful, including a proceeding involving a Cybersecurity Incident. For a nonpublic proceeding, the user, owner or operator of the Bulk-Power System that is the subject of the penalty will be given timely notice and an opportunity for hearing and the public will not be notified and the public will not be allowed to participate.
- (f) On its own motion or upon complaint, the Commission may order compliance with a Reliability Standard and may impose a penalty against a user, owner or operator of the Bulk-Power System, if the Commission finds, after public notice and opportunity for hearing, that the user, owner or operator of the Bulk-Power System has engaged or is about to engage in any acts or practices that constitute or will constitute a violation of a Reliability Standard.
- (g) Any penalty imposed for the violation of a Reliability Standard shall bear a reasonable relation to the seriousness of the violation and shall take into consideration efforts of such user, owner or operator of the Bulk-Power System to remedy the violation in a timely manner.
- (1) The penalty imposed may be a monetary or a non-monetary penalty and may include, but is not limited to, a limitation on an activity, function, operation, or other appropriate sanction, including being added to a reliability watch list composed of major violators that is established by the Electric Reliability Organization, a Regional Entity or the Commission.
- (2) The Electric Reliability Organization shall submit for Commission approval penalty guidelines that set forth a range of penalties for the violation of Reliability Standards. A penalty imposed by the Electric Reliability Organization or a Regional Entity must be

within be within the range set forth in the penalty guidelines.

[Order 672, 71 FR 8736, Feb. 17, 2006, as amended by Order 737, 75 FR 43404, July 26, 2010]

§39.8 Delegation to a Regional Entity.

- (a) The Electric Reliability Organization may enter into an agreement to delegate authority to a Regional Entity for the purpose of proposing Reliability Standards to the Electric Reliability Organization and enforcing Reliability Standards under § 39.7.
- (b) After notice and opportunity for comment, the Commission may approve a delegation agreement. A delegation agreement shall not be effective until it is approved by the Commission.
- (c) The Electric Reliability Organization shall file a delegation agreement. Such filing shall include a statement demonstrating that:
- (1) The Regional Entity is governed by an independent board, a balanced stakeholder board, or a combination independent and balanced stakeholder board;
- (2) The Regional Entity otherwise satisfies the provisions of section 215(c) of the Federal Power Act; and
- (3) The agreement promotes effective and efficient administration of Bulk-Power System reliability.
- (d) The Commission may modify such delegation.
- (e) The Electric Reliability Organization shall and the Commission will rebuttably presume that a proposal for delegation to a Regional Entity organized on an Interconnection-wide basis promotes effective and efficient administration of Bulk-Power System reliability and should be approved.
- (f) An entity seeking to enter into a delegation agreement that is unable to reach an agreement with the Electric Reliability Organization within 180 days after proposing a delegation agreement to the Electric Reliability Organization may apply to the Commission to assign to it the Electric Reliability Organization's authority to enforce Reliability Standards within its region. The entity must demonstrate in its application that it meets the requirements of paragraph (c) of this section and that continued negotiations with the Electric Reli-

ability Organization would not likely result in an appropriate delegation agreement within a reasonable period of time. After notice and opportunity for hearing, the Commission may designate the entity as a Regional Entity and assign enforcement authority to it.

- (g) An application pursuant to paragraph (f) of this section must state:
- (1) Whether the Commission's Dispute Resolution Service, or other alternative dispute resolution procedures were used, or why these procedures were not used; and
- (2) Whether the Regional Entity believes that alternative dispute resolution under the Commission's supervision could successfully resolve the disputes regarding the terms of the delegation agreement.

§ 39.9 Enforcement of Commission Rules and Orders.

- (a) The Commission may take such action as is necessary and appropriate against the Electric Reliability Organization or a Regional Entity to ensure compliance with a Reliability Standard or any Commission order affecting the Electric Reliability Organization or a Regional Entity, including, but not limited to:
- (1) After notice and opportunity for hearing, imposition of civil penalties under the Federal Power Act.
- (2) After notice and opportunity for hearing, suspension or decertification of the Commission's certification to be the Electric Reliability Organization.
- (3) After notice and opportunity for hearing, suspension or rescission of the Commission's approval of an agreement to delegate certain Electric Reliability Organization authorities to a Regional Entity.
- (b) The Commission may periodically audit the Electric Reliability Organization's performance under this part.

§ 39.10 Changes to an Electric Reliability Organization Rule or Regional Entity Rule.

(a) The Electric Reliability Organization shall file with the Commission for approval any proposed Electric Reliability Organization Rule or Rule change. A Regional Entity shall submit a Regional Entity Rule or Rule change

§ 39.11

to the Electric Reliability Organization and, if approved by the Electric Reliability Organization, the Electric Reliability Organization shall file the proposed Regional Entity Rule or Rule change with the Commission for approval. Any filing by the Electric Reliability Organization shall be accompanied by an explanation of the basis and purpose for the Rule or Rule change, together with a description of the proceedings conducted by the Electric Reliability Organization or Regional Entity to develop the proposal.

- (b) The Commission, upon its own motion or upon complaint, may propose a change to an Electric Reliability Organization Rule or Regional Entity Rule.
- (c) A proposed Electric Reliability Organization Rule or Rule change or Regional Entity Rule or Rule change shall take effect upon a finding by the Commission, after notice and opportunity for public comment, that the change is just, reasonable, not unduly discriminatory or preferential, is in the public interest, and satisfies the requirements of § 39.3.

§39.11 Reliability reports.

- (a) The Electric Reliability Organization shall conduct assessments as determined by the Commission of the reliability of the Bulk-Power System in North America and provide a report to the Commission and provide subsequent reports of the same to the Commission
- (b) The Electric Reliability Organization shall conduct assessments of the adequacy of the Bulk-Power System in North America and report its findings to the Commission, the Secretary of Energy, each Regional Entity, and each Regional Advisory Body annually or more frequently if so ordered by the Commission.

§39.12 Review of state action.

(a) Nothing in this section shall be construed to preempt any authority of any state to take action to ensure the safety, adequacy, and reliability of electric service within that state, as long as such action is not inconsistent with any Reliability Standard, except that the State of New York may establish rules that result in greater reli-

ability within that state, as long as such action does not result in lesser reliability outside the state than that provided by the Reliability Standards.

- (b) Where a state takes action to ensure the safety, adequacy, or reliability of electric service, the Electric Reliability Organization, a Regional Entity or other affected person may apply to the Commission for a determination of consistency of the state action with a Reliability Standard.
 - (1) The application shall:
 - (i) Identify the state action;
- (ii) Identify the Reliability Standard with which the state action is alleged to be inconsistent:
- (iii) State the basis for the allegation that the state action is inconsistent with the Reliability Standard; and
- (iv) Be served on the relevant state agency and the Electric Reliability Organization, concurrent with its filing with the Commission.
- (2) Within ninety (90) days of the application of the Electric Reliability Organization, the Regional Entity, or other affected person, and after notice and opportunity for public comment, the Commission will issue a final order determining whether the state action is inconsistent with a Reliability Standard, taking into consideration any recommendation of the Electric Reliability Organization and the state.
- (c) The Commission, after consultation with the Electric Reliability Organization and the state taking action, may stay the effectiveness of the state action, pending the Commission's issuance of a final order.

§39.13 Regional Advisory Bodies.

- (a) The Commission will establish a Regional Advisory Body on the petition of at least two-thirds of the states within a region that have more than one-half of their electric load served within the region.
- (b) A petition to establish a Regional Advisory Body shall include a statement that the Regional Advisory Body is composed of one member from each participating state in the region, appointed by the governor of each state, and may include representatives of agencies, states and provinces outside the United States.

Federal Energy Regulatory Commission

- (c) A Regional Advisory Body established by the Commission may provide advice to the Electric Reliability Organization or a Regional Entity or the Commission regarding:
- (1) The governance of an existing or proposed Regional Entity within the same region;
- (2) Whether a Reliability Standard proposed to apply within the region is just, reasonable, not unduly discriminatory or preferential, and in the public interest:
- (3) Whether fees for all activities under this part proposed to be assessed within the region are just, reasonable, not unduly discriminatory or preferential, and in the public interest; and
- (4) Any other responsibilities requested by the Commission.
- (d) The Commission may give deference to the advice of a Regional Advisory Body established by the Commission that is organized on an Interconnection-wide basis.

PART 40—MANDATORY RELIABILITY STANDARDS FOR THE BULK-POWER SYSTEM

Sec.

40.1 Applicability.

40.2 Mandatory Reliability Standards.

40.3 Availability of Reliability Standards.

AUTHORITY: 16 U.S.C. 824o.

Source: Order 693, 72 FR 16598, Apr. 4, 2007, unless otherwise noted.

§ 40.1 Applicability.

- (a) This part applies to all users, owners and operators of the Bulk-Power System within the United States (other than Alaska or Hawaii), including, but not limited to, entities described in section 201(f) of the Federal Power Act.
- (b) Each Reliability Standard made effective by §40.2 must identify the subset of users, owners and operators of the Bulk-Power System to which a particular Reliability Standard applies.

§ 40.2 Mandatory Reliability Standards.

(a) Each applicable user, owner or operator of the Bulk-Power System must comply with Commission-approved Re-

liability Standards developed by the Electric Reliability Organization.

(b) A proposed modification to a Reliability Standard proposed to become effective pursuant to §39.5 of this Chapter will not be effective until approved by the Commission.

§ 40.3 Availability of Reliability Standards.

The Electric Reliability Organization must post on its Web site the currently effective Reliability Standards as approved and enforceable by the Commission. The effective date of the Reliability Standards must be included in the posting.

PART 41—ACCOUNTS, RECORDS, MEMORANDA AND DISPOSITION OF CONTESTED AUDIT FINDINGS AND PROPOSED REMEDIES

DISPOSITION OF CONTESTED AUDIT FINDINGS AND PROPOSED REMEDIES

Sec.

41.1 Notice to audited person.

41.2 Response to notification.

41.3 Shortened procedure.

41.4 Form and style.

41.5 Verification.

41.6 Determination.

41.7 Assignment for oral hearing.

41.8 Burden of proof.

CERTIFICATION OF COMPLIANCE WITH ACCOUNTING REGULATIONS

41.10 Examination of accounts.

41.11 Report of certification.

41.12 Qualifications of accountants.

AUTHORITY: 16 U.S.C. 791a–825r, 2601–2645; 42 U.S.C. 7101–7352.

SOURCE: Order 141, 12 FR 8500, Dec. 19, 1947, unless otherwise noted.

CROSS REFERENCE: For rules of practice and procedure, see part 385 of this chapter.

DISPOSITION OF CONTESTED AUDIT FINDINGS AND PROPOSED REMEDIES

§41.1 Notice to audited person.

(a) Applicability. This part applies to all audits conducted by the Commission or its staff under authority of the Federal Power Act except for Electric Reliability Organization audits conducted pursuant to the authority of part 39 of the Commission's regulations.

§41.2

(b) Notice. An audit conducted by the Commission's staff under authority of the Federal Power Act may result in a notice of deficiency or audit report or similar document containing a finding or findings that the audited person has not complied with a requirement of the Commission with respect to, but not limited to, the following: A filed tariff or tariffs, contracts, data, records, accounts, books, communications or papers relevant to the audit of the audited person; matters under the Standards of Conduct or the Code of Conduct; and the activities or operations of the audited person. The notice of deficiency, audit report or similar document may also contain one or more proposed remedies that address findings of noncompliance. Where such findings, with or without proposed remedies, appear in a notice of deficiency, audit report or similar document, such document shall be provided to the audited person, and the finding or findings, and any proposed remedies, shall be noted and explained. The audited person shall timely indicate in a written response any and all findings or proposed remedies, or both, in any combination, with which the audited person disagrees. The audited person shall have 15 days from the date it is sent the notice of deficiency, audit report or similar document to provide a written response to the audit staff indicating any and all findings or proposed remedies, or both, in any combination, with which the audited person disagrees, and such further time as the audit staff may provide in writing to the audited person at the time the document is sent to the audited person. The audited person may move the Commission for additional time to provide a written response to the audit staff and such motion shall be granted for good cause shown. Any initial order that the Commission subsequently may issue with respect to the notice of deficiency, audit report or similar document shall note, but not address on the merits, the finding or findings, or the proposed remedy or remedies, or both, in any combination, with which the audited person disagreed. The Commission shall provide the audited person 30 days to respond to the initial Commission order concerning a notice of defi-

ciency, audit report or similar document with respect to the finding or findings or any proposed remedy or remedies, or both, in any combination, with which it disagreed.

[Order 675-A, 71 FR 29784, May 24, 2006]

§41.2 Response to notification.

Upon issuance of a Commission order that notes a finding or findings, or proposed remedy or remedies, or both, in any combination, with which the audited person has disagreed, the audited person may: Acquiesce in the findings and/or proposed remedies by not timely responding to the Commission order, in which case the Commission may issue an order approving them or taking other action; or challenge the finding or findings and/or any proposed remedies, with which it disagreed by timely notifying the Commission in writing that it requests Commission review by means of a shortened procedure or, if there are material facts in dispute which require cross-examination, a trial-type hearing.

[Order 675, 71 FR 9706, Feb. 27, 2006]

§41.3 Shortened procedure.

If the audited person subject to a Commission order described in §41.1 notifies the Commission that it seeks to challenge one or more audit findings, or proposed remedies, or both, in any combination, by the shortened procedure, the Commission shall thereupon issue a notice setting a schedule for the filing of memoranda. The person electing the use of the shortened procedure, and any other interested entities, including the Commission staff, shall file, within 45 days of the notice, an initial memorandum that addresses the relevant facts and applicable law that support the position or positions taken regarding the matters at issue. Reply memoranda shall be filed within 20 days of the date by which the initial memoranda are due to be filed. Only participants who filed initial memoranda may file reply memoranda. Subpart T of part 385 of this chapter shall apply to all filings. Within 20 days after the last date that reply memoranda under the shortened procedure may be timely filed, the audited person who elected the shortened procedure

may file a motion with the Commission requesting a trial-type hearing if new issues are raised by a party. To prevail in such a motion, the audited person must show that a party to the shortened procedure raised one or more new issues of material fact relevant to resolution of a matter in the shortened procedure such that fundamental fairness requires a trial-type hearing to resolve the new issue or issues so raised. Parties to the shortened procedure and the Commission staff may file responses to the motion. In ruling upon the motion, the Commission may determine that some or all of the issues be litigated in a trial-type hearing.

[Order 675, 71 FR 9706, Feb. 27, 2006]

§41.4 Form and style.

Each copy of such memorandum must be complete in itself. All pertinent data should be set forth fully, and each memorandum should set out the facts and argument as prescribed for briefs in §385.706 of this chapter.

[Order 141, 12 FR 8500, Dec. 19, 1947, as amended by Order 225, 47 FR 19056, May 3, 1982]

§41.5 Verification.

The facts stated in the memorandum must be sworn to by persons having knowledge thereof, which latter fact must affirmatively appear in the affidavit. Except under unusual circumstances, such persons should be those who would appear as witnesses if hearing were had to testify as to the facts stated in the memorandum.

§41.6 Determination.

If no formal hearing is had the matter in issue will be determined by the Commission on the basis of the facts and arguments submitted.

§41.7 Assignment for oral hearing.

Except when there are no material facts in dispute, when a person does not consent to the shortened procedure, the Commission will assign the proceeding for hearing as provided by subpart E of part 385 of this chapter. Notwithstanding a person's not giving consent to the shortened procedure, and instead seeking assignment for hearing as provided for by subpart E of

part 385 of this chapter, the Commission will not assign the proceeding for a hearing when no material facts are in dispute. The Commission may also, in its discretion, at any stage in the proceeding, set the proceeding for hearing.

[Order 575, 60 FR 4854, Jan. 25, 1995]

§41.8 Burden of proof.

The burden of proof to justify every accounting entry shall be on the person making, authorizing, or requiring such entry.

CERTIFICATION OF COMPLIANCE WITH ACCOUNTING REGULATIONS

§41.10 Examination of accounts.

(a) All Major and Nonmajor public utilities and licensees not classified as Class C or Class D prior to January 1, 1984 shall secure, for the year 1968 and each year thereafter until December 31, 1975, the services of an independent certified public accountant, or independent licensed public accountant, certified or licensed by a regulatory authority of a State or other political subdivision of the United States, to test compliance in all material respects of those schedules as are indicated in the General Instructions set out in the Annual Report, Form No. 1, with the Commission's applicable Uniform System of Accounts and published accounting releases. The Commission expects that identification of questionable matters by the independent accountant will facilitate their early resolution and that the independent accountant will seek advisory rulings by the Commission on such items. This examination shall be deemed supplementary to periodic Commission examinations of compliance.

(b) Beginning January 1, 1976, and each year thereafter, only independent certified public accountants, or independent licensed public accountants who were licensed on or before December 31, 1970, will be authorized to conduct annual audits and to certify to compliance in all material respects, of those schedules as are indicated in the General Instructions set out in the Annual Report, Form No. 1, with the Commission's applicable Uniform System of Accounts, published accounting

§41.11

releases and all other regulatory matters.

[Order 462, 37 FR 26005, Dec. 7, 1972, as amended by Order 390, 49 FR 32505, Aug. 14, 1984]

§41.11 Report of certification.

Each Major and Nonmajor (including those companies classified as nonoperating under Part 101, General Instruction 1(A)(3) of this chapter) public utility or licensee operating on a calendar year and not classified as Class C or Class D prior to January 1, 1984 must file with the Commission a letter or report of the independent accountant certifying approval, together with or within 30 days after the filing of the Annual Report, Form No. 1, covering the subjects and in the form prescribed in the General Instructions of the Annual Report. For such utility or licensee operating on a non-calendar fiscal year, the letter or report of the independent accountant certifying approval must be filed within 150 days of the close of the company's fiscal year; the letter or report must also identify which, if any, of the examined schedules do not conform to the Commission's requirements and shall describe the discrepancies that exist. The Commission will not be bound by a certification of compliance made by an independent accountant pursuant to this paragraph.

[73 FR 58736, Oct. 7, 2008]

§41.12 Qualifications of accountants.

The Commission will not recognize any certified public accountant or public accountant through December 31, 1975, who is not in fact independent. Beginning January 1, 1976, and each year thereafter, the Commission will recognize only independent certified public accountants, or independent licensed public accountants who were licensed on or before December 31, 1970, who are in fact independent. For example, an accountant will not be considered independent with respect to any person or any of its parents or subsidiaries in whom he has, or had during the period of report, any direct financial interest. The Commission will determine the fact of independence by considering all the relevant circumstances including evidence bearing on the relationships between the accountant and that person or any affiliate thereof.

[Order 462, 37 FR 26006, Dec. 7, 1972]

PART 42—LONG-TERM FIRM TRANS-MISSION RIGHTS IN ORGANIZED ELECTRICITY MARKETS

Sec

42.1 Requirement that Transmission Organizations with Organized Electricity Markets Offer Long-Term Firm Transmission Rights.

AUTHORITY: 16 U.S.C. 791a-825r and section 217 of the Federal Power Act, 16 U.S.C. 824q.

SOURCE: Order 681, 71 FR 43619, Aug. 1, 2006, unless otherwise noted.

§ 42.1 Requirement that Transmission Organizations with Organized Electricity Markets Offer Long-Term Firm Transmission Rights.

- (a) Purpose. This section requires a transmission organization with one or more organized electricity markets (administered either by it or by another entity) to make available longterm firm transmission rights, pursuant to section 217(b)(4) of the Federal Power Act, that satisfy each of the guidelines set forth in paragraph (d) of this section. This section does not require that a specific type of long-term firm transmission right be made available, and is intended to permit transmission organizations flexibility in satisfying the guidelines set forth in paragraph (d) of this section.
- (b) Definitions. As used in this section:
- (1) Transmission Organization means a Regional Transmission Organization, Independent System Operator, independent transmission provider, or other independent transmission organization finally approved by the Commission for the operation of transmission facilities.
- (2) Load serving entity means a distribution utility or an electric utility that has a service obligation.
- (3) Service obligation means a requirement applicable to, or the exercise of authority granted to, an electric utility under Federal, State, or local law or under long-term contracts to provide electric service to end-users or to a distribution utility.

- (4) Organized Electricity Market means an auction-based day ahead and real time wholesale market where a single entity receives offers to sell and bids to buy electric energy and/or ancillary services from multiple sellers and buyers and determines which sales and purchases are completed and at what prices, based on formal rules contained in Commission-approved tariffs, and where the prices are used by a transmission organization for establishing transmission usage charges.
- (c) General rule. (1) Every public utility that is a transmission organization and that owns, operates or controls facilities used for the transmission of electric energy in interstate commerce and has one or more organized electricity markets (administered either by it or by another entity) must file with the Commission, no later than January 29, 2007, one of the following:
- (i) Tariff sheets and rate schedules that make available long-term firm transmission rights that satisfy each of the guidelines set forth in paragraph (d) of this section; or
- (ii) An explanation of how its current tariff and rate schedules already provide for long-term firm transmission rights that satisfy each of the guidelines set forth in paragraph (d) of this section.
- (2) Any transmission organization approved by the Commission for operation after January 29, 2007 that has one or more organized electricity markets (administered either by it or by another entity) will be required to satisfy this general rule.
- (3) Filings made in compliance with this paragraph (c) must explain how the transmission organization's transmission planning and expansion procedures will accommodate long-term firm transmission rights, including but not limited to how the transmission organization will ensure that allocated long-term firm transmission rights remain feasible over their entire term.
- (4) Each transmission organization subject to this general rule must also make its transmission planning and expansion procedures and plans publicly available, including (but not limited to) both the actual plans and any underlying information used to develop the plans.

- (d) Guidelines for Design and Administration of Long-term Firm Transmission Rights. Transmission organizations subject to paragraph (c) of this section must make available long-term firm transmission rights that satisfy the following guidelines:
- (1) The long-term firm transmission right should specify a source (injection node or nodes) and sink (withdrawal node or nodes), and a quantity (MW).
- (2) The long-term firm transmission right must provide a hedge against day-ahead locational marginal pricing congestion charges or other direct assignment of congestion costs for the period covered and quantity specified. Once allocated, the financial coverage provided by a financial long-term right should not be modified during its term (the "full funding" requirement) except in the case of extraordinary circumstances or through voluntary agreement of both the holder of the right and the transmission organization.
- (3) Long-term firm transmission rights made feasible by transmission upgrades or expansions must be available upon request to any party that pays for such upgrades or expansions in accordance with the transmission organization's prevailing cost allocation methods for upgrades or expansions.
- (4) Long-term firm transmission rights must be made available with term lengths (and/or rights to renewal) that are sufficient to meet the needs of load serving entities to hedge long-term power supply arrangements made or planned to satisfy a service obligation. The length of term of renewals may be different from the original term. Transmission organizations may propose rules specifying the length of terms and use of renewal rights to provide long-term coverage, but must be able to offer firm coverage for at least a 10 year period.
- (5) Load serving entities must have priority over non-load serving entities in the allocation of long-term firm transmission rights that are supported by existing capacity. The transmission organization may propose reasonable limits on the amount of existing capacity used to support long-term firm transmission rights.

Pt. 45

- (6) A long-term transmission right held by a load serving entity to support a service obligation should be re-assignable to another entity that acquires that service obligation.
- (7) The initial allocation of the longterm firm transmission rights shall not require recipients to participate in an auction.

PART 45—APPLICATION FOR AUTHORITY TO HOLD INTER-LOCKING POSITIONS

Sec.

- 45.1 Applicability; who must file.
- 45.2 Positions requiring authorization.
- 45.3 Time of filing application.
- 45.4 Supplemental applications.
- 45.5 Supplemental information.
- 45.6 Termination of authorization.
- 45.7 Form of application; filing procedure.
- 45.8 Contents of application.
- 45.9 Automatic authorization of certain interlocking positions.

AUTHORITY: 16 U.S.C. 791a-825r, 2601-2645; 31 U.S.C. 9701; 42 U.S.C. 7101-7352; 3 CFR 142.

SOURCE: Order 141, 12 FR 8501, Dec. 19, 1947, unless otherwise noted.

CROSS REFERENCES: For rules of practice and procedure, see part 385 of this chapter. For forms under rules of practice and regulations under the Federal Power Act, see part 131 of this chapter.

§45.1 Applicability; who must file.

- (a) This part applies to any person seeking to hold the following interlocking positions:
- (1) Officer or director of more than one public utility;
- (2) Officer or director of a public utility and of any bank, trust company, banking association, or firm that is authorized by law to underwrite or participate in the marketing of securities of a public utility; or
- (3) Officer or director of a public utility and of any company supplying electrical equipment to a public utility.
- (b) Any person seeking to hold any interlocking position described in §45.2 of this chapter must do the following:
- (1) Apply for Commission authorization under § 45.8 of this chapter; or
- (2) If qualified, comply with the requirements for automatic authorization under §45.9 of this chapter.

[Order 446, 51 FR 4904, Feb. 10, 1986]

§ 45.2 Positions requiring authorization.

- (a) The positions subject to this part shall include those of any person elected or appointed to perform the duties or functions ordinarily performed by a president, vice president, secretary, treasurer, general manager, comptroller, chief purchasing agent, director or partner, or to perform any other similar executive duties or functions, in any corporation 1 within the purview of section 305(b) of the Act. With respect to positions not herein specifically mentioned which applicant holds and which are invested with executive authority, applicant shall state in the application the source of such executive authority, whether by bylaws, action of the board of directors, or otherwise.
- (b) Corporations 1 within the purview of section 305(b) of the Act include:
- (1) Any public utility under the Act, which means any person who owns or operates facilities for the transmission of electric energy in interstate commerce, or any person who owns or operates facilities for the sale at wholesale of electric energy in interstate commerce.
- (2) Any bank, trust company, banking association, or firm that is authorized by law to underwrite or participate in the marketing of public utility securities; this includes any corporation when so authorized whether or not same may also be a public utility and/or a holding company. (See 12 U.S.C. 378)
- (3) Any company that supplies electrical equipment to a public utility in which applicant seeks authorization to hold a position, whether the supplying company be a manufacturer, or dealer, or one supplying electrical equipment pursuant to a construction, service, agency, or other contract.
- (c) Regardless of any action which may have been taken by the Commission upon a previous application under

¹Corporation means any corporation, jointstock company, partnership, association, business trust, organized group of persons, whether incorporated or not, or a receiver or receivers, trustee or trustees of any of the foregoing. It shall not include municipalities as defined in the Federal Power Act (sec. 3, 49 Stat. 838: 16 U.S.C. 796).

section 305(b) of the Act, an application for approval under such section is required with reference to any position or positions not previously authorized which are within the purview of said section.

§45.3 Timing of filing application.

(a) The holding of positions within the purview of section 305(b) of the Act shall be unlawful unless the holding shall have been authorized by order of the Commission. Nothing in this part shall be construed as authorizing the holding of positions within the purview of section 305(b) of the Act prior to order of the Commission on application therefor. Applications must be filed and authorization must be granted prior to holding any interlocking positions within the purview of section 305(b) of the Act; late-filed applications will be denied. The term "holding", as used in this part, shall mean acting as, serving as, voting as, or otherwise performing or assuming the duties and responsibilities of officer or director within the purview of section 305(b) of the Act.

(b) Absent Commission action within 60 days of a completed application to hold interlocking positions, an application will be deemed granted. Such authorization is subject to revocation by the Commission after due notice to applicant and opportunity for hearing. In any such proceeding, the burden of proof shall be upon the applicant to show that neither public nor private interests will be adversely affected by the holding of such positions.

[Order 664, 70 FR 55723, Sept. 23, 2005]

§45.4 Supplemental applications.

(a) New positions. In the event of a change or changes in the information set forth in an application, by the applicant's election or appointment to another position or other positions in corporations within the purview of section 305(b) of the Act, the application shall be supplemented by the applicant's setting forth all the data with respect to the new position or positions in accordance with the requirements of this part.

(b) Old positions. After applicant has been authorized to hold a particular position, further application in connec-

tion with each successive term so long as he continues in uninterrupted tenure of such position will not be required except as ordered by the Commission. If the term of office or the holding of any position for which authorization has been given shall be interrupted and the applicant shall subsequently be reelected or reappointed thereto, further authorization will be required.

§ 45.5 Supplemental information.

(a) Required by Commission. Applicants under this part shall upon request of the Commission and within such time as may be allowed, supplement any application or any supplemental application with any information required by the Commission.

(b) Notice of changes. In the event of the applicant's resignation, withdrawal, or failure of reelection or appointment in respect to any of the positions for which authorization has been granted by the Commission, or in the event of any other material or substantial change therein, the applicant shall within 30 days after any such change occurs, give notice thereof to the Commission setting forth the position corporation, and date of termination therewith, or other material or substantial change.

(c) Reports. All persons holding positions by authorization of the Commission under section 305(b) of the Act may be required to file such periodic or special reports as the Commission may deem necessary.

§45.6 Termination of authorization.

(a) By the Commission. Orders of authorization under section 305(b) of the Act are subject to revocation by the Commission after due notice to applicant and opportunity for hearing. In any such proceeding the burden of proof shall be upon the applicant to show that neither public nor private interests will be adversely affected by the holding of such positions.

(b) Without action of the Commission. Whenever a person shall cease to hold a position theretofore authorized to be held by the Commission or such position shall cease to be within the purview of section 305(b) of the Federal

§ 45.7

Power Act, the Commission's authorization to hold such position shall terminate without further action by the Commission. If upon such termination of authorization as aforesaid, such person does not continue to hold at least two positions authorized and then requiring authorization pursuant to said section 305(b) of the Act, all authorization theretofore given by the Commission shall thereupon terminate.

§ 45.7 Form of application; filing procedure.

Applications, supplemental applications, statements of supplemental information, notices of change, and reports should be filed with the Secretary of the Commission in accordance with filing procedures posted on the Commission's Web site at http://www.ferc.gov. Each filing must be dated, signed by the applicant, and verified under oath in accordance with § 385.2005(b) and (c).

[Order 737, 75 FR 43404, July 26, 2010]

§45.8 Contents of application.

Each application shall state the following:

- (a) Identification of applicant. (1) Full name, business address and state of residence.
- (2) Major business or professional activity.
- (3) If former application or applications under section 305(b) of the Act have been made by the applicant, give date and docket number of the last application filed.
- (b) List of positions within the purview of section 305(b) of the Act for which authorization is sought. (Indicate by asterisk positions which were the subjects of previous authorizations.)

Position Name of corporation

Classification: (1) Public utility, (2) authorized by law to underwrite, (3) supplying electrical equipment

(c) Data as to positions with each public utility mentioned in paragraph (b) of this section. (The format should be adapted to the information submitted, in keeping with completeness and conciseness. In the case of public utilities of the same holding company system,

brevity will generally be promoted by submitting the information for all of the utilities involved under each subsection progressively in the order of the subsections, utilizing tables when feasible.)

- (1) Name of utility.
- (2) Date elected or appointed, or anticipated date of election or appointment, to each position not previously authorized.
- (3) Names of officers and directors; number of vacancies, if any, on Board of Directors.
- (4) Description of applicant's duties: Approximate amount of time devoted thereto; and, if applicant seeks authorization as a director, the percentage of directors meetings held during the past 12 months that were attended by the applicant.
- (5) All other professional, contractual, or business relationships of applicant with the public utility, either directly or through other corporations or firms
- (6) Extent of applicant's direct or indirect ownership, control of, or beneficial interest in the public utility or the securities thereof. If ownership or interest is held in a name other than that of applicant, state name and address of the holder.
- (7) Extent of applicant's indebtedness to the public utility, how and when incurred, and consideration therefor.
- (8) All money or property received by applicant from the public utility or any affiliate (i) during the past 12 months, and expected during the ensuing 12 months, or (ii) during the public utility's most recently ended fiscal year, and expected during the public utility's current fiscal year, or (iii) during the past and current calendar years, whether for services, reimbursement for expenses, or otherwise. Specify in detail the amount thereof and the basis therefor. If applicant's compensation for services to the public utility is not paid directly by the public utility, give name of the corporation that does pay same, the amount allocated or allocable to the public utility or any affiliate, and the basis or reason for such allocation.
- (9) Whether during the past 5 years the public utility or any affiliate

thereof or any security holders of either have commenced any suit against the officers or directors thereof for alleged waste, mismanagement or violation of duty, to which suit applicant was a party defendant. If so, give date of commencement of suit, court in which commenced, and present status.

- (d) Data as to positions with each bank, trust company, banking association or firm, mentioned in paragraph (b) of this section, that is authorized by law to underwrite or participate in the marketing of a public utility. (The applicant shall use a separate sheet for each corporation.)
- (1) Name of corporation and address of principal place of business.
- (2) Positions which applicant holds or seeks authorization to hold therein and when and by whom elected or appointed to each position.
- (3) Description of applicant's duties in each position and the approximate amount of time devoted thereto, and, if applicant seeks authorization as director, the percentage of directors meetings held during the past 12 months that were attended by the applicant.
- (4) Extent of applicant's direct or indirect ownership, or control of, or beneficial interest in, the company or in the securities thereof, including common stock, preferred stock, bonds, or other securities. If such ownership or interest is held in a name other than that of applicant, state name and address of such holder.
- (5) All money or property received by applicant from the company (i) during the past 12 months, and expected during the ensuing 12 months, or (ii) during the company's most recently ended fiscal year, and expected during the company's current fiscal year, or (iii) during the past and current calendar years, whether for services, reimbursement for expenses, or otherwise. Specify in detail the amount thereof and the basis therefor.
- (6) Names and titles of directors, officers, or partners.
- (7) Whether the corporation is now engaged in underwriting or participating in the marketing of the securities of a public utility; if so, to what extent.
- (8) Whether the corporation, during applicant's connection therewith, has

- underwritten or participated in the marketing of the security issue of any public utility with which applicant was also connected; if so, the details with respect to every such transaction that occurred during the past 36 months.
- (9) (If the answer to paragraph (d)(7) of this section is in the negative.) Give excerpts from the charter, declaration of trust, or articles of partnership that authorize the underwriting or participating in the marketing of securities of a public utility.
- (10) (If the answer to paragraph (d)(7) of this section is in the negative.) Give general requirements of and appropriate reference to, the laws of the State of organization and of States in which corporation is doing business or has qualified to do business, with which it must comply in order to engage in the business of underwriting or participating in the marketing of the securities of a public utility.
- (11) What steps, if any, have been taken to comply with laws mentioned in paragraph (d)(10) of this section.
- (12) In lieu of paragraphs (d)(9), (10), and (11) of this section, an opinion by counsel to the same effect and including the information in respect thereto may be filed with the application.
- (13) Whether the corporation has registered with the Securities and Exchange Commission; if so, when and under what section of what act.
- (e) Data as to positions with each company, mentioned in paragraph (b) of this section, supplying electrical equipment to a public utility in which applicant holds a position. (Applicant shall use a separate sheet for each company.)
- (1) Name of company and address of principal place of business.
- (2) Positions which applicant holds or seeks authorization to hold therein and when and by whom elected or appointed to each position.
- (3) Description of applicant's duties in each position and approximate amounts of time devoted thereto, and, if applicant seeks authorization as director, the percentage of directors meetings held during the past 12 months that were attended by the applicant.
- (4) Names and titles of directors or partners.

§ 45.8

- (5) Name of each public utility, with which applicant holds or seeks authorization to hold a position, to which the company supplies electrical equipment; the frequency of such transactions; the approximate annual dollar volume of such business; and the type of equipment supplied.
- (6) Nature of relationship between the company supplying electrical equipment and the public utility:
- (i) Whether company manufactures such electrical equipment or is a dealer therein.
- (ii) Whether company supplies electrical equipment to the public utility pursuant to construction, service, agency, or other contract with the public utility or an affiliate thereof, and, if so, furnish brief summary of the terms of such contract.
- (7) Extent of applicant's direct or indirect ownership, or control of, or beneficial interest in, the company or in the securities thereof, including common stock, preferred stock, bonds, or other securities. If such ownership or interest is held in a name other than that of applicant, state name and address of such holder.
- (8) All money or property received by applicant from the company (i) during the past 12 months, and expected during the ensuing 12 months, or (ii) during the company's most recently ended fiscal year, and expected during the company's current fiscal year, or (iii) during the past and current calendar years, whether for services, reimbursement for expenses, or otherwise. Specify in detail the amount thereof and the basis therefor.
- (f) Data as to positions with public utility holding companies. (Do not include here data as to corporations listed in paragraph (b) of this section which are also holding companies. A holding company as herein used means any corporation which directly or indirectly owns, controls, or holds with power to vote, 10 per centum or more, of the outstanding voting securities of a public utility.)
- (1) Name of holding company and address of principal place of business.
- (2) Positions which applicant holds therein, when and by whom elected or appointed to each position.

- (3) Extent of applicant's direct or indirect ownership, or control of, or beneficial interest in, the holding company or in the securities thereof, including common stock, preferred stock, bonds, or other securities. If such ownership or interest is held in a name other than that of applicant, state name and address of such holder.
- (4) All money or property received by applicant from the holding company (i) during the past 12 months, and expected during the ensuing 12 months, or (ii) during the holding company's most recently ended fiscal year, and expected during the holding company's current fiscal year, or (iii) during the past and current calendar years, whether for services, reimbursement for expenses, or otherwise. Specify in detail the amount thereof and the basis therefor.
- (g) Positions with all other corporations. (Do not include here data that have been filed within the past 12 months in FERC-561, pursuant to part 46 of this chapter, or data as to any corporations listed in paragraph (b) or (f) of this section.)
- (1) All other corporations and positions therein, including briefly the information required in parallel columns as below:

Name of corporation	Address: Kind of business	Position held therein
oo.po.ao	01 000111000	

- (2) Any corporate, contractual, financial, or business relationships between any of the corporations listed in paragraph (g)(1) of this section and any of the public utilities listed in paragraph (b) of this section.
- (h) Data as to the public utility holding company system. The names of the public utility holding company systems of which each public utility listed in paragraph (b) of this section is a part, with a chart showing the corporate relationships existing between and among the corporations within the holding company systems.

[Order 246, 27 FR 4912, May 25, 1962, as amended by Order 427, 36 FR 5598, Mar. 25, 1971; Order 374, 49 FR 20479-20480, May 15, 1984; Order 435, 50 FR 40358, Oct. 3, 1985; Order 737, 75 FR 43404, July 26, 2010]

CROSS REFERENCE: For rules and regulations of the Securities and Exchange Commission, see 17 CFR, chap, II.

§ 45.9 Automatic authorization of certain interlocking positions.

- (a) Applicability. Subject to paragraphs (b) and (c) of this section, the Commission authorizes any officer or director of a public utility to hold the following interlocking positions:
- (1) Officer or director of one or more other public utilities if the same holding company owns, directly or indirectly, that percentage of each utility's stock (of whatever class or classes) which is required by each utility's bylaws to elect directors;
- (2) Officer or director of two public utilities, if one utility is owned, wholly or in part, by the other and, as its primary business, owns or operates transmission or generating facilities to provide transmission service or electric power for sale to its owners; and
- (3) Officer or director of more than one public utility, if such officer or director is already authorized under this part to hold different positions as officer or director of those utilities where the interlock involves affiliated public utilities.
- (b) Conditions of authorization. As a condition of authorization, any person authorized to hold interlocking positions under this section must submit, prior to performing or assuming the duties and responsibilities of the position, an informational report in accordance with paragraph (c) of this section, unless that person is already authorized to hold interlocking positions of the type governed by this section. Failure to timely file the informational report will constitute a failure to satisfy this condition, and will constitute automatic denial.
- (c) *Informational report*. An informational report required under paragraph (b) of this section must state:
- (1) The full name and business address of the person required to submit this report;
- (2) The names of all public utilities with which the person holds or will hold the positions of officer or director and a description of those positions;
- (3) The names of any entity, other than those listed in paragraph (c)(2) of this section, of which the person is an

officer or director and a description of those positions; and

- (4) An explanation of the corporate relationship between or among the public utilities listed in paragraph (c)(2) of this section which qualifies the person for automatic authorization under this section.
- (5) A statement or an affirmation that the applicant has not yet performed or assumed the duties or responsibilities of the position which necessitated the filing of this informational report.

[Order 446, 51 FR 4905, Feb. 10, 1986, as amended by Order 664, 70 FR 55723, Sept. 23, 2005]

PART 46—PUBLIC UTILITY FILING RE-QUIREMENTS AND FILING RE-QUIREMENTS FOR PERSONS HOLDING INTERLOCKING POSI-TIONS

Sec.

- 46.1 Purpose.
- 46.2 Definitions.
- 46.3 Purchaser list.
- 46.4 General rule.
- 46.5 Covered entities.
- 46.6 Contents of the statement and procedures for filing.

AUTHORITY: 16 U.S.C. 792-828c; 16 U.S.C. 2601-2645; 42 U.S.C. 7101-7352; E.O. 12009, 3 CFR 142.

SOURCE: 45 FR 23418, Apr. 7, 1980, unless otherwise noted.

§46.1 Purpose.

The purpose of this part is to implement section 305(c) of the Federal Power Act, as amended by section 211 of the Public Utility Regulatory Policies Act of 1978.

[Order 67, 45 FR 3569, Jan. 18, 1980]

§ 46.2 Definitions.

For the purpose of this part:

(a) Public utility has the same meaning as in section 201(e) of the Federal Power Act and further includes any company which is part of a holding company system which includes a registered holding company unless no company in such system is an electric utility within the meaning of section 3 of the Federal Power Act. Such term

§ 46.3

does not include any rural electric cooperative which is regulated by the Rural Electrification Administration of the Department of Agriculture or any other entities covered in section 201(f) of the Federal Power Act.

- (b) The following terms have the same meaning as in the Public Utility Holding Company Act of 1935:
 - (1) Holding company system; and
 - (2) Registered holding company.
- (c) Purchaser means any individual or corporation within the meaning of section 3 of the Federal Power Act who purchases electric energy from a public utility. Such term does not include the United States or any agency or instrumentality of the United States or any rural electric cooperative which is regulated by the Rural Electrification Administration of the Department of Agriculture.
- (d) Control and controlled mean the possession, directly or indirectly, of the power to direct the management or policies of an entity whether such power is exercised through one or more intermediary companies or pursuant to an agreement, written or oral, and whether such power is established through ownership or voting of securities, or common directors, officers, or stockholders, or voting trusts, holding trusts, or debt holdings, or contract, or any other direct or indirect means. A rebuttable presumption that control exists arises from the ownership or the power to vote, directly or indirectly, ten percent (10%) or more of the voting securities of such entity.
- (e) Entity means any firm, company, or organization including any corporation, joint-stock company, partnership, association, business trust, organized group of persons, whether incorporated or not, or a receiver or receivers, trustee or trustees of any of the foregoing. Such term does not include municipality as defined in section 3 of the Federal Power Act and does not include any Federal, State, or local government agencies or any rural electric cooperative which is regulated by the Rural Electrification Administration of the Department of Agriculture.
- (f) Electrical equipment means any apparatus, device, integral component, or integral part used in an activity which is electrically, electronically, mechani-

- cally, or by legal prescription necessary to the process of generation, transmission, or distribution of electric energy. ¹
- (g) Produces or supplies means any transaction including a sale, lease, sale-leaseback, consignment, or any other transaction in which an entity provides electrical equipment, coal, natural gas, oil, nuclear fuel, or other fuel to any public utility either directly or through an entity controlled by such entity.
- (h) Appointee means any person appointed on a temporary or permanent basis to perform any duties or functions described in §46.4(a).
- (i) Representative means any person empowered, through oral or written agreement, to transact business on behalf of an entity and any person who serves as an advisor regarding policy or management decisions of the entity. The term does not include attorneys, accountants, architects, or any other persons who render a professional service on a fee basis.

§ 46.3 Purchaser list.

(a) Compilation and filing list. On or before January 31 of each year, each public utility shall compile a list of the purchasers described in paragraph (b) of this section and shall identify each

¹Guidance in applying the definition of electrical equipment may be obtained by examining the items within the following accounts described in part 101, title 18 of the Code of Federal Regulations: Boiler/Reactor plant equipment (Accounts 312 and 322); Engines and engine driven generators (313); Turbogenerator units (314 and 323); Accessory electrical equipment (315, 324, 334 and 345); Miscellaneous power plant equipment (316, 325, 335 and 346); Water wheels, turbines and generators (333); Fuel holders, producers, and accessories (342); Prime movers (343); Generators (344); Station equipment (353 and 362); Poles, towers and fixtures (354, 355 and 364); Overhead conductors and devices (356 and 365); Underground conduit (357 and 366); Underground conductors and devices (358 and 367); Storage battery equipment (363); Line transformers (368); Services (369); Meters (370); Installation on customers' premises (371); Street lighting and signal systems (373): Leased property on customers' premises (372); and Communication equipment (397). Excepted from these accounts, are vehicles, structures, foundations, settings, and services.

purchaser by name and principal business address. The public utility must submit the list to the Secretary of the Commission in accordance with filing procedures posted on the Commission's Web site at http://www.ferc.gov and make the list publicly available through its principal business office.

- (b) Largest purchasers. The list required under paragraph (a) of this section shall include each purchaser who, during any of the three (3) preceding calendar years, purchased (for purposes other than resale) from a public utility one of the twenty (20) largest amounts of electric energy measured in kilowatt hours sold (for purposes other than resale) by such utility during such year.
- (c) Special rules. If data for actual annual sales (for purposes other than resale) are not available in the records of the public utility, the utility may use estimates based on actual data available to it. If one purchaser maintains several billing accounts with the public utility, the kilowatt hours purchased in each account of that purchaser shall be aggregated to arrive at the total for that purchaser.
- (d) Notification of largest purchasers. Each public utility shall notify by January 31 of each year each purchaser which has been identified on the list of largest purchasers under paragraph (b) of this section.
- (e) Revision of the list. Each public utility relying upon any estimates for its January 31st filing, shall revise the list compiled under paragraph (b) of this section no later than March 1 of the year in which the list was originally filed to reflect actual data not available to the utility prior to that time. Any revised list shall be filed with the Commission and made publicly available through the utility's principal business office no later than March 1. A utility filing a revised list shall indicate thereon the changes made to the list previously filed under paragraph (b) of this section. On or before the filing and publication of the revised list, the public utility shall notify the newly-listed purchasers and any purchasers whose names were removed from the list.

[Order 67, 45 FR 3569, Jan. 18, 1980; 45 FR 6377, Jan. 28, 1980, as amended by Order 737, 75 FR 43404, July 26, 2010]

§46.4 General rule.

- A person must file with the Secretary of the Commission a statement in accordance with §46.6, and in the form specified in §131.31 of this chapter (except that with respect to calendar year 1980, no filings in the form specified in §131.31 is required if such person has previously filed the statement required for calendar year 1980 in a different form than specified in §131.31), if such person:
- (a) Serves for a public utility in any of the following positions: A director or a chief executive officer, president, vice president, secretary, treasurer, general manager, comptroller, chief purchasing agent, or any other position in which such person performs similar executive duties or functions for such public utility; and
- (b) Serves for any entity described in §46.5 in any of the positions described in paragraph (a) of this section or is a partner, appointee, or representative of such entity.

[45 FR 23418, Apr. 7, 1980, as amended by Order 140, 46 FR 22181, Apr. 16, 1981; Order 737, 75 FR 43404, July 26, 2010]

§ 46.5 Covered entities.

Entities to which the general rule in §46.4(b) applies are the following:

- (a) Any investment bank, bank holding company, foreign bank or subsidiary thereof doing business in the United States, insurance company, or any other organization primarily engaged in the business of providing financial services or credit, a mutual savings bank, or a savings and loan association;
- (b) Any entity which is authorized by law to underwrite or participate in the marketing of securities of a public utility;
- (c) Any entity which produces or supplies electrical equipment or coal, natural gas, oil, nuclear fuel, or other fuel, for the use of any public utility;
 - (d) Any entity specified in §46.3;
- (e) Any entity referred to in section 305(b) of the Federal Power Act; and
- (f) Any entity which is controlled by any entity referred to in this section.

§46.6

§ 46.6 Contents of the statement and procedures for filing.

Each person required to file a written statement under the general rule in §46.4 shall comply with the following requirements:

- (a) Each person shall provide the following information: full name and business address; identification of the public utilities and the covered entities in which such person holds executive positions described in §46.4; and identification of the interlock described in §46.4;
- (b) If the interlock is between a public utility and an entity described in §46.5(c), which produces or supplies electrical equipment for use of such public utility, such person shall provide the following information:
- (1) The aggregate amount of revenues received by such entity from producing or supplying electrical equipment to such public utility in the calendar year specified in paragraph (d) of this section, rounded up to the nearest \$100,000; and
- (2) The nature of the business relationship between such public utility and such entity.
- (c) If the person is authorized by the Commission to hold the positions of officer or director in accordance with part 45, such person shall identify the authorization by docket number and shall give the date of authorization.
- (d)(1) Each person shall file an original and one copy of such written statement with the Office of Secretary of the Commission on or before April 30 of each year immediately following the calendar year during any portion of which such person held a position described in §46.4. The original of such statement shall be dated and signed by such person. The copy shall bear the date that appeared on the original; the signature on the copy may be stamped or typed on the copy.
- (2) Instead of submitting changes to the Commission on the pre-printed Form No. 561 sent annually by the Commission, a person may choose to make changes to the pre-filled electronic version provided by the Commission. This electronic version, along with the signed original and one copy (as required by Paragraph (d)(c)) shall also be filed with the Commission.

(3) Such statement shall be available to the public through the Commission's eLibrary system on http://www.ferc.gov and shall be made publicly available through the principal business offices of the public utility and any entity to which it applies on or before April 30 of the year the statement was filed with the Commission.

(Pub. L. 96–511, 94 Stat. 2812 (44 U.S.C. 3501 *et seg.*))

[45 FR 23418, Apr. 7, 1980, as amended by Order 601, 63 FR 72169, Dec. 31, 1998; Order 737, 75 FR 43404, July 26, 2010]

PART 50—APPLICATIONS FOR PER-MITS TO SITE INTERSTATE ELEC-TRIC TRANSMISSION FACILITIES

Sec.

50.1 Definitions.

50.2 Purpose and intent of rules.

50.3 Applications/pre-filing; rules and format.

50.4 Stakeholder participation.

50.5 Pre-filing procedures.

50.6 Applications: general content.

50.7 Applications: exhibits.

50.8 Acceptance/rejection of applications.

50.9 Notice of application.

50.10 Interventions.

50.11 General conditions applicable to permits.

AUTHORITY: 16 U.S.C. 824p, DOE Delegation Order No. 00-004.00A.

SOURCE: 71 FR 69465, Dec. 1, 2006, unless otherwise noted.

§ 50.1 Definitions.

As used in this part:

Affected landowners include owners of property interests, as noted in the most recent county/city tax records as receiving the tax notice, whose property:

- (1) Is directly affected (i.e., crossed or used) by the proposed activity, including all facility sites, rights-of-way, access roads, staging areas, and temporary workspace; or
- (2) Abuts either side of an existing right-of-way or facility site owned in fee by any utility company, or abuts the edge of a proposed facility site or right-of-way which runs along a property line in the area in which the facilities would be constructed, or contains a residence within 50 feet of a proposed construction work area.

Director means the Director of the Office of Energy Projects or his designees.

Federal authorization means permits, special use authorization, certifications, opinions, or other approvals that may be required under Federal law in order to site a transmission facility.

National interest electric transmission corridor means any geographic area experiencing electric energy transmission capacity constraints or congestion that adversely affects consumers, as designated by the Secretary of Energy.

Permitting entity means any Federal or State agency, Indian tribe, multistate, or local agency that is responsible for issuing separate authorizations pursuant to Federal law that are required to construct electric transmission facilities in a national interest electric transmission corridor.

Stakeholder means any Federal, State, interstate, Tribal, or local agency, any affected non-governmental organization, affected landowner, or interested person.

Transmitting utility means an entity that owns, operates, or controls facilities used for the transmission of electric energy in interstate commerce for the sale of electric energy at wholesale.

§ 50.2 Purpose and intent of rules.

(a) The purpose of the regulations in this part is to provide for efficient and timely review of requests for permits for the siting of electric transmission facilities under section 216 of the Federal Power Act. The regulations ensure that each stakeholder is afforded an opportunity to present views and recommendations with respect to the need for and impact of a facility covered by the permit. They also coordinate, to the maximum extent practicable, the Federal authorization and review processes of other Federal and State agencies, Indian tribes, multistate, and local entities that are responsible for conducting any separate permitting and environmental reviews of the proposed facilities.

(b) Every applicant shall file all pertinent data and information necessary for a full and complete understanding of the proposed project.

- (c) Every requirement of this part will be considered as an obligation of the applicant which can only be avoided by a definite and positive showing that the information or data called for by the applicable rules is not necessary for the consideration and ultimate determination of the application.
- (d) The burden of assuring that all applications and information submitted under this part is in an intelligible form and any omission of data is justified rests with the applicant.

§ 50.3 Applications/pre-filing; rules and format.

- (a) Filings are subject to the formal paper and electronic filing requirements for proceedings before the Commission as provided in part 385 of this chapter.
- (b) Applications, amendments, and all exhibits and other submissions required to be furnished by an applicant to the Commission under this part must be submitted in an original and 7 conformed copies.
- (c) When an application considered alone is incomplete and depends vitally upon information in another application, it will not be accepted for filing until the supporting application has been filed. When applications are interdependent, they must be filed concurrently.
- (d) All filings must be signed in compliance with §385.2005 of this chapter.
- (e) The Commission will conduct a paper hearing on applications for permits for electric transmission facilities.
- (f) Permitting entities will be subject to the filing requirements of this section and the prompt and binding intermediate milestones and ultimate deadlines established in the notice issued under §50.9.
- (g) Any person submitting documents containing critical energy infrastructure information must follow the procedures specified in §388.113 of this chapter.

§ 50.4

§ 50.4 Stakeholder participation.

A Project Participation Plan is required to ensure stakeholders have access to accurate and timely information on the proposed project and permit application process.

(a) Project Participation Plan. An applicant must develop a Project Participation Plan and file it with the pre-filing materials under §50.5(c)(7) that:

- (1) Identifies specific tools and actions to facilitate stakeholder communications and public information, including an up-to-date project Web site and a readily accessible, single point of contact within the company;
- (2) Lists all central locations in each county throughout the project area where the applicant will provide copies of all their filings related to the proposed project; and
- (3) Includes a description and schedule explaining how the applicant intends to respond to requests for information from the public as well as Federal, State, and Tribal permitting agencies, and other legal entities with local authorization requirements.
- (b) Document Availability. (1) Within three business days of the date the prefiling materials are filed or application is issued a docket number, an applicant must ensure that:
- (i) Complete copies of the pre-filing or application materials are available in accessible central locations in each county throughout the project area, either in paper or electronic format; and
- (ii) Complete copies of all filed materials are available on the project Web site.
- (2) An applicant is not required to serve voluminous or difficult to reproduce material, such as copies of certain environmental information, on all parties, as long as such material is publicly available in an accessible central location in each county throughout the project area and on the applicant's project website.
- (c) Project notification. (1) The applicant must make a good faith effort to notify: all affected landowners; landowners with a residence within a quarter mile from the edge of the construction right-of-way of the proposed project; towns and communities; permitting agencies; other local, State, Tribal, and Federal governments and

agencies involved in the project; electric utilities and transmission owners and operators that are or may be connected to the application's proposed transmission facilities; and any known individuals that have expressed an interest in the State permitting proceeding. Notification must be made:

- (i) By certified or first class mail, sent:
- (A) Within 14 days after the Director notifies the applicant of the commencement of the pre-filing process under §50.5(d);
- (B) Within 3 business days after the Commission notices the application under \$50.9: and
- (ii) By twice publishing a notice of the pre-filing request and application filings, in a daily, weekly, and/or tribal newspaper of general circulation in each county in which the project is located, no later than 14 days after the date that a docket number is assigned for the pre-filing process or to the application.
 - (2) Contents of participation notice
- (i) The pre-filing request notification must, at a minimum, include:
- (A) The docket number assigned to the proceeding;
- (B) The most recent edition of the Commission's pamphlet *Electric Transmission Facilities Permit Process*. The newspaper notice need only refer to the pamphlet and indicate that it is available on the Commission's website;
- (C) A description of the applicant and a description of the proposed project, its location (including a general location map), its purpose, and the timing of the project;
- (D) A general description of the property the applicant will need from an affected landowner if the project is approved, how to contact the applicant, including a local or toll-free phone number, the name of a specific person to contact who is knowledgeable about the project, and a reference to the project website. The newspaper notice need not include a description of the property, but should indicate that a separate notice is being mailed to affected landowners and governmental entities;
- (E) A brief summary of what rights the affected landowner has at the Commission and in proceedings under the

eminent domain rules of the relevant State. The newspaper notice does not need to include this summary;

- (F) Information on how to get a copy of the pre-filing information from the company and the location(s) where copies of the pre-filing information may be found as specified in paragraph (b) of this section;
- (G) A copy of the Director's notification of commencement of the pre-filing process, the Commission's Internet address, and the telephone number for the Commission's Office of External Affairs; and
- (H) Information explaining the prefiling and application process and when and how to intervene in the application proceedings.
- (ii) The application notification must include the Commission's notice issued under §50.9.
- (3) If, for any reason, a stakeholder has not yet been identified when the notices under this paragraph are sent or published, the applicant must supply the information required under paragraphs (c)(2)(i) and (ii) of this section when the stakeholder is identified.
- (4) If the notification is returned as undeliverable, the applicant must make a reasonable attempt to find the correct address and notify the stake-holder
- (5) Access to critical energy infrastructure information is subject to the requirements of §388.113 of this chapter.

§ 50.5 Pre-filing procedures.

- (a) Introduction. Any applicant seeking a permit to site new electric transmission facilities or modify existing facilities must comply with the following pre-filing procedures prior to filing an application for Commission review.
- (b) *Initial consultation*. An applicant must meet and consult with the Director concerning the proposed project.
- (1) At the initial consultation meeting, the applicant must be prepared to discuss the nature of the project, the contents of the pre-filing request, and the status of the applicant's progress toward obtaining the information required for the pre-filing request described in paragraph (c) of this section.
- (2) The initial consultation meeting will also include a discussion of wheth-

- er a third-party contractor is likely to be needed to prepare the environmental documentation for the project and the specifications for the applicant's solicitation for prospective third-party contractors.
- (3) The applicant also must discuss how its proposed project will be subject to the Commission's jurisdiction under section 216(b)(1) of the Federal Power Act. If the application is seeking Commission jurisdiction under section 216(b)(1)(C) of the Federal Power Act, the applicant must be prepared to discuss when it filed its application with the State and the status of that application.
- (c) Contents of the initial filing. An applicant's pre-filing request will be filed after the initial consultation and must include the following information:
- (1) A description of the schedule desired for the project, including the expected application filing date, desired date for Commission approval, and proposed project operation date, as well as the status of any State siting proceedings.
- (2) A detailed description of the project, including location maps and plot plans to scale showing all major components, including a description of zoning and site availability for any permanent facilities.
- (3) A list of the permitting entities responsible for conducting separate Federal permitting and environmental reviews and authorizations for the project, including contact names and telephone numbers, and a list of local entities with local authorization requirements. The filing must include information concerning:
- (i) How the applicant intends to account for each of the relevant entity's permitting and environmental review schedules, including its progress in DOE's pre-application process; and
- (ii) When the applicant proposes to file with these permitting and local entities for the respective permits or other authorizations.
- (4) A list of all affected landowners and other stakeholders (include contact names and telephone numbers) that have been contacted, or have contacted the applicant, about the project.

§ 50.5

- (5) A description of what other work already has been done, including, contacting stakeholders, agency and Indian tribe consultations, project engineering, route planning, environmental and engineering contractor engagement, environmental surveys/studies, open houses, and any work done or actions taken in conjunction with a State proceeding. This description also must include the identification of the environmental and engineering firms and sub-contractors under contract to develop the project.
- (6) Proposals for at least three prospective third-party contractors from which Commission staff may make a selection to assist in the preparation of the requisite NEPA document, if the Director determined a third-party contractor would be necessary in the Initial Consultation meeting.
- (7) A proposed Project Participation Plan, as set forth in §50.4(a).
- (d) Director's notice. (1) When the Director finds that an applicant seeking authority to site and construct an electric transmission facility has adequately addressed the requirements of paragraphs (a), (b), and (c) of this section, and any other requirements determined at the Initial Consultation meeting, the Director will so notify the applicant.
- (i) The notification will designate the third-party contractor, and
- (ii) The pre-filing process will be deemed to have commenced on the date of the Director's notification.
- (2) If the Director determines that the contents of the initial pre-filing request are insufficient, the applicant will be notified and given a reasonable time to correct the deficiencies.
- (e) Subsequent filing requirements. Upon the Director's issuance of a notice commencing an applicant's pre-filing process, the applicant must:
- (1) Within 7 days, finalize and file the Project Participation Plan, as defined in §50.4(a), and establish the dates and locations at which the applicant will conduct meetings with stakeholders and Commission staff.
- (2) Within 14 days, finalize the contract with the selected third-party contractor, if applicable.
 - (3) Within 14 days:

- (i) Provide all identified stakeholders with a copy of the Director's notification commencing the pre-filing process:
- (ii) Notify affected landowners in compliance with the requirements of $\S50.4(c)$; and
- (iii) Notify permitting entities and request information detailing any specific information not required by the Commission in the resource reports required under §380.16 of this chapter that the permitting entities may require to reach a decision concerning the proposed project. The responses of the permitting entities must be filed with the Commission, as well as being provided to the applicant.
- (4) Within 30 days, submit a mailing list of all stakeholders contacted under paragraph (e)(3) of this section, including the names of the Federal, State, Tribal, and local jurisdictions' representatives. The list must include information concerning affected landowner notifications that were returned as undeliverable.
- (5) Within 30 days, file a summary of the project alternatives considered or under consideration.
- (6) Within 30 days, file an updated list of all Federal, State, Tribal, and local agencies permits and authorizations that are necessary to construct the proposed facilities. The list must include:
- (i) A schedule detailing when the applications for the permits and authorizations will be submitted (or were submitted):
- (ii) Copies of all filed applications; and
- (iii) The status of all pending permit or authorization requests and of the Secretary of Energy's pre-application process being conducted under section 216(h)(4)(C) of the Federal Power Act.
- (7) Within 60 days, file the draft resource reports required in §380.16 of this chapter.
- (8) On a monthly basis, file status reports detailing the applicant's project activities including surveys, stakeholder communications, and agency and tribe meetings, including updates on the status of other required permits or authorizations. If the applicant fails

to respond to any request for additional information, fails to provide sufficient information, or is not making sufficient progress towards completing the pre-filing process, the Director may issue a notice terminating the process.

(f) Concluding the pre-filing process. The Director will determine when the information gathered during the pre-filing process is complete, after which the applicant may file an application. An application must contain all the information specified by the Commission staff during the pre-filing process, including the environmental material required in part 380 of this chapter and the exhibits required in §50.7.

§ 50.6 Applications: general content.

Each application filed under this part must provide the following information:

- (a) The exact legal name of applicant; its principal place of business; whether the applicant is an individual, partnership, corporation, or otherwise; the State laws under which the applicant is organized or authorized; and the name, title, and mailing address of the person or persons to whom communications concerning the application are to be addressed.
- (b) A concise description of applicant's existing operations.
- (c) A concise general description of the proposed project sufficient to explain its scope and purpose. The description must, at a minimum: Describe the proposed geographic location of the principal project features and the planned routing of the transmission line; contain the general characteristics of the transmission line including voltage, types of towers, origin and termination points of the transmission line, and the geographic character of area traversed by the line; and be accompanied by an overview map of sufficient scale to show the entire transmission route on one or a few 8.5 by 11-inch sheets.
- (d) Verification that the proposed route lies within a national interest electric transmission corridor designated by the Secretary of the Department of Energy under section 216 of the Federal Power Act.
 - (e) Evidence that:

- (1) A State in which the transmission facilities are to be constructed or modified does not have the authority to approve the siting of the facilities or consider the interstate benefits expected to be achieved by the proposed construction or modification of transmission facilities in the State;
- (2) The applicant is a transmitting utility but does not qualify to apply for a permit or siting approval of the proposed project in a State because the applicant does not serve end-use customers in the State; or
- (3) A State commission or other entity that has the authority to approve the siting of the facilities has:
- (i) Withheld approval for more than one year after the filing of an application seeking approval under applicable law or one year after the designation of the relevant national interest electric transmission corridor, whichever is later; or
- (ii) Conditioned its approval in such a manner that the proposed construction or modification will not significantly reduce transmission congestion in interstate commerce or is not economically feasible.
- (f) A demonstration that the facilities to be authorized by the permit will be used for the transmission of electric energy in interstate commerce, and that the proposed construction or modification:
- (1) Is consistent with the public interest;
- (2) Will significantly reduce transmission congestion in interstate commerce and protects or benefits consumers:
- (3) Is consistent with sound national energy policy and will enhance energy interdependence; and
- (4) Will maximize, to the extent reasonable and economical, the transmission capabilities of existing towers or structures.
- (g) A description of the proposed construction and operation of the facilities, including the proposed dates for the beginning and completion of construction and the commencement of service.
- (h) A general description of project financing.
- (i) A full statement as to whether any other application to supplement or

§ 50.7

effectuate the applicant's proposals must be or is to be filed by the applicant, any of the applicant's customers, or any other person, with any other Federal, State, Tribal, or other regulatory body; and if so, the nature and status of each such application.

(j) A table of contents that must list all exhibits and documents filed in compliance with this part, as well as all other documents and exhibits otherwise filed, identifying them by their appropriate titles and alphabetical letter designations. The alphabetical letter designations specified in §50.7 must be strictly adhered to and extra exhibits submitted at the volition of applicant must be designated in sequence under the letter Z (Z1, Z2, Z3, etc.).

(k) A form of notice suitable for publication in the FEDERAL REGISTER, as provided by §50.9(a), which will briefly summarize the facts contained in the application in such a way as to acquaint the public with its scope and purpose. The form of notice also must include the name, address, and telephone number of an authorized contact person.

§ 50.7 Applications: exhibits.

Each exhibit must contain a title page showing the applicant's name, title of the exhibit, the proper letter designation of the exhibit, and, if 10 or more pages, a table of contents, citing by page, section number or subdivision, the component elements or matters contained in the exhibit.

- (a) Exhibit A—Articles of incorporation and bylaws. If the applicant is not an individual, a conformed copy of its articles of incorporation and bylaws, or other similar documents.
- (b) Exhibit B—State authorization. For each State where the applicant is authorized to do business, a statement showing the date of authorization, the scope of the business the applicant is authorized to carry on and all limitations, if any, including expiration dates and renewal obligations. A conformed copy of applicant's authorization to do business in each State affected must be supplied upon request.
- (c) Exhibit C—Company officials. A list of the names and business addresses of the applicant's officers and directors,

or similar officials if the applicant is not a corporation.

- (d) Exhibit D—Other pending applications and filings. A list of other applications and filings submitted by the applicant that are pending before the Commission at the time of the filing of an application and that directly and significantly affect the proposed project, including an explanation of any material effect the grant or denial of those other applications and filings will have on the application and of any material effect the grant or denial of the application will have on those other applications and filings.
- (e) Exhibit E—Maps of general location of facilities. The general location map required under §50.5(c) must be provided as Exhibit E. Detailed maps required by other exhibits must be filed in those exhibits, in a format determined during the pre-filing process in §50.5.
- (f) Exhibit F—Environmental report. An environmental report as specified in §§380.3 and 380.16 of this chapter. The applicant must submit all appropriate revisions to Exhibit F whenever route or site changes are filed. These revisions must identify the locations by mile post and describe all other specific differences resulting from the route or site changes, and should not simply provide revised totals for the resources affected. The format of the environmental report filing will be determined during the pre-filing process required under §50.5.
 - (g) Exhibit G—Engineering data.
- (1) A detailed project description including:
- (i) Name and destination of the project;
 - (ii) Design voltage rating (kV);
 - (iii) Operating voltage rating (kV);
- (iv) Normal peak operating current rating:
- (v) Line design features for minimizing television and/or radio interference cause by operation of the proposed facilities; and
- (vi) Line design features that minimize audible noise during fog/rain caused by operation of the proposed facilities, including comparing expected audible noise levels to the applicable Federal, State, and local requirements.

- (2) A conductor, structures, and substations description including:
 - (i) Conductor size and type;
 - (ii) Type of structures;
 - (iii) Height of typical structures;
- (iv) An explanation why these structures were selected;
- (v) Dimensional drawings of the typical structures to be used in the project; and
- (vi) A list of the names of all new (and existing if applicable) substations or switching stations that will be associated with the proposed new transmission line.
- (3) The location of the site and rightof-way including:
- (i) Miles of right-of-way;
- (ii) Miles of circuit;
- (iii) Width of the right-of-way;
- (iv) A brief description of the area traversed by the proposed transmission line, including a description of the general land uses in the area and the type of terrain crossed by the proposed line;
- (4) Assumptions, bases, formulae, and methods used in the development and preparation of the diagrams and accompanying data, and a technical description providing the following information:
- (i) Number of circuits, with identification as to whether the circuit is overhead or underground;
- (ii) The operating voltage and frequency; and
- (iii) Conductor size, type and number of conductors per phase.
- (5) If the proposed interconnection is an overhead line, the following additional information also must be provided:
- (i) The wind and ice loading design parameters;
- (ii) A full description and drawing of a typical supporting structure including strength specifications;
- (iii) Structure spacing with typical ruling and maximum spans;
- (iv) Conductor (phase) spacing; and
- (v) The designed line-to-ground and conductor-side clearances.
- (6) If an underground or underwater interconnection is proposed, the following additional information also must be provided:
 - (i) Burial depth:
- (ii) Type of cable and a description of any required supporting equipment,

- such as insulation medium pressurizing or forced cooling;
 - (iii) Cathodic protection scheme; and
- (iv) Type of dielectric fluid and safeguards used to limit potential spills in waterways.
- (7) Technical diagrams that provide clarification of any of the above items should be included.
- (8) Any other data or information not previously identified that has been identified as a minimum requirement for the siting of a transmission line in the State in which the facility will be located.
- (h) Exhibit H—System analysis data. An analysis evaluating the impact the proposed facilities will have on the existing electric transmission system performance, including:
- (1) An analysis of the existing and expected congestion on the electric transmission system.
- (2) Power flow cases used to analyze the proposed and future transmission system under anticipated load growth, operating conditions, variations in power import and export levels, and additional transmission facilities required for system reliability. The cases must:
- (i) Provide all files to model normal, single contingency, multiple contingency, and special protective systems, including the special protective systems' automatic switching or load shedding system; and
- (ii) State the assumptions, criteria, and guidelines upon which they are based and take into consideration transmission facility loading; first contingency incremental transfer capability (FCITC); normal incremental transfer capability (NITC); system protection; and system stability.
- (3) A stability analysis including study assumptions, criteria, and guidelines used in the analysis, including load shedding allowables.
- (4) A short circuit analysis for all power flow cases.
- (5) A concise analysis to include:
- (i) An explanation of how the proposed project will improve system reliability over the long and short term;
- (ii) An analysis of how the proposed project will impact long term regional transmission expansion plans;

§ 50.8

- (iii) An analysis of how the proposed project will impact congestion on the applicant's entire system; and
- (iv) A description of proposed high technology design features.
- (6) Detailed single-line diagrams, including existing system facilities identified by name and circuit number, that show system transmission elements, in relation to the project and other principal interconnected system elements, as well as power flow and loss data that represent system operating conditions.
- (i) Exhibit I—Project Cost and Financing. (1) A statement of estimated costs of any new construction or modification.
- (2) The estimated capital cost and estimated annual operations and maintenance expense of each proposed environmental measure.
- (3) A statement and evaluation of the consequences of denial of the transmission line permit application.
- (j) Exhibit J—Construction, operation, and management. A concise statement providing arrangements for supervision, management, engineering, accounting, legal, or other similar service to be rendered in connection with the construction or operation of the project, if not to be performed by employees of the applicant, including reference to any existing or contemplated agreements, together with a statement showing any affiliation between the applicant and any parties to the agreements or arrangements.

§ 50.8 Acceptance/rejection of applications.

- (a) Applications will be docketed when received and the applicant so advised
- (b) If an application patently fails to comply with applicable statutory requirements or with applicable Commission rules, regulations, and orders for which a waiver has not been granted, the Director may reject the application as provided by §385.2001(b) of this chapter. This rejection is without prejudice to an applicant's refiling a complete application. However, an application will not be rejected solely on the basis that the environmental reports are incomplete because the company has not

been granted access by affected landowners to perform required surveys.

(c) An application that relates to a proposed project or modification for which a prior application has been filed and rejected, will be docketed as a new application.

§ 50.9 Notice of application.

- (a) Notice of each application filed, except when rejected in accordance with §50.8, will be issued and subsequently published in the FEDERAL REGISTER.
- (b) The notice will establish prompt and binding intermediate milestones and ultimate deadlines for the coordination, and review of, and action on Federal authorization decisions relating to, the proposed facilities.

§ 50.10 Interventions.

Notices of applications, as provided by §50.9, will fix the time within which any person desiring to participate in the proceeding may file a petition to intervene, and within which any interested regulatory agency, as provided by §385.214 of this chapter, desiring to intervene may file its notice of intervention.

$\S 50.11$ General conditions applicable to permits.

- (a) The following terms and conditions, among others as the Commission will find are required by the public interest, will attach to the issuance of each permit and to the exercise of the rights granted under the permit.
- (b) The permit will be void and without force or effect unless accepted in writing by the permittee within 30 days from the date of the order issuing the permit. Provided that, when an applicant files for rehearing of the order in accordance with FPA section 313(a), the acceptance must be filed within 30 days from the issue date of the order of the Commission upon the application for rehearing or within 30 days from the date on which the application may be deemed to have been denied when the Commission has not acted on such application within 30 days after it has been filed. Provided further, that when a petition for review is filed in accordance with the provisions of FPA section 313(b), the acceptance shall be

filed within 30 days after final disposition of the judicial review proceedings thus initiated.

- (c) Standards of construction and operation. In determining standard practice, the Commission will be guided by the provisions of the American National Standards Institute, Incorporated, the National Electrical Safety Code, and any other codes and standards that are generally accepted by the industry, except as modified by this Commission or by municipal regulators within their jurisdiction. Each electric utility will construct, install, operate, and maintain its plant, structures, equipment, and lines in accordance with these standards, and in a manner to best accommodate the public, and to prevent interference with service furnished by other public or non-public utilities insofar as practical.
- (d) Written authorization must be obtained from the Director prior to commencing construction of the facilities or initiating operations. Requests for these authorizations must demonstrate compliance with all terms and conditions of the construction permit.
- (e) Any authorized construction or modification must be completed and made available for service by the permitee within a period of time to be specified by the Commission in each order issuing the transmission line construction permit. If facilities are not completed within the specified timeframe, the permittee must file for an extension of time under §385.2008 of this chapter.
- (f) A permittee must file with the Commission, in writing and under oath, an original and four conformed copies, as provided in §385.2011 of this chapter, of the following:

- (1) Within ten days after the bona fide beginning of construction, notice of the date of the beginning; and
- (2) Within ten days after authorized facilities have been constructed and placed in service, notice of the date of the completion of construction and commencement of service.
- (g) The permit issued to the applicant may be transferred, subject to the approval of the Commission, to a person who agrees to comply with the terms, limitations or conditions contained in the filing and in every subsequent Order issued thereunder. A permit holder seeking to transfer a permit must file with the Secretary a petition for approval of the transfer. The petition must:
- (1) State the reasons supporting the transfer;
- (2) Show that the transferee is qualified to carry out the provisions of the permit and any Orders issued under the permit;
- (3) Be verified by all parties to the proposed transfer;
- (4) Be accompanied by a copy of the proposed transfer agreement;
- (5) Be accompanied by an affidavit of service of a copy on the parties to the permit proceeding; and
- (6) Be accompanied by an affidavit of publication of a notice concerning the petition and service of such notice on all affected landowners that have executed agreements to convey property rights to the transferee and all other persons, municipalities or agencies entitled by law to be given notice of, or be served with a copy of, any application to construct a major electric generation facility.